

Durham E-Theses

System modelling and control

Gill, Kenneth Fred

How to cite:

Gill, Kenneth Fred (1990) System modelling and control, Durham theses, Durham University. Available at Durham E-Theses Online: http://etheses.dur.ac.uk/6193/

Use policy

 $The full-text\ may\ be\ used\ and/or\ reproduced,\ and\ given\ to\ third\ parties\ in\ any\ format\ or\ medium,\ without\ prior\ permission\ or\ charge,\ for\ personal\ research\ or\ study,\ educational,\ or\ not-for-profit\ purposes\ provided\ that:$

- a full bibliographic reference is made to the original source
- a link is made to the metadata record in Durham E-Theses
- the full-text is not changed in any way

The full-text must not be sold in any format or medium without the formal permission of the copyright holders.

Please consult the full Durham E-Theses policy for further details.

Academic Support Office, The Palatine Centre, Durham University, Stockton Road, Durham, DH1 3LE e-mail: e-theses.admin@durham.ac.uk Tel: +44 0191 334 6107 http://etheses.dur.ac.uk The copyright of this thesis rests with the author. No quotation from it should be published without his prior written consent and information derived from it should be acknowledged.

UNIVERSITY OF DURHAM

APPLICATION FOR THE DEGREE OF D.Sc.

1**990**

KENNETH FRED GILL

B.Sc. (Dunelm) M.Sc., Ph.D. (Birmingham) C.Eng., M.I.Mech.E., M.I.E.E.

Part 2.



- 6 JUL 1992

*

CONTENTS

Part 1.

- 1. CURRICULUM VITAE.
- 2. STATEMENT OF CONTRIBUTION AND ORIGINALITY.
- 3. PUBLICATIONS ON GAS TURBINE MODELLING AND CONTROL.
- 4. PUBLICATIONS ON SATELLITE MODELLING AND CONTROL.
- 5. PUBLICATIONS ON NON-STEADY GAS FLOW IN DUCTS.
- 6. PUBLICATIONS ON GAS FLOW MEASUREMENT.
- 7. PUBLICATIONS OF A GENERAL NATURE.

Part 2.

- 8. TEXTBOOK ON MODELLING AND CONTROL.
- 9. PUBLICATIONS ON MEASUREMENT AND CONTROL IN MANUFACTURE.
- **10. PAPERS UNDER REVIEW.**

8. Textbook

.

••



COPY OF TEXT BOOK IN POCKET

9. MEASUREMENT AND CONTROL IN MANUFACTURE

Publication	Page
[50] Identification of surface roughness.	1
[53] A characterization of surface texture profiles.	7
[54] Identification of surface texture signals.	19
[56] A comparison of orthogonal transforms in their application to surface texture analysis.	23
[60] Knowledge representation database for the development of a fixture design expert system.	31
[61] Use of fuzzy logic in robotics.	44
[62] A complete description of surface texture profiles.	56
[64] An experimental evaluation of normalized Fourier descriptors in the identification of simple engineering objects.	66
[65] Experimental evaluation of 'shape from shading' for engineering componment profile measurement	77

PAPER 50

Heat Transfer Through a Piston by Electrical Analogy

By J. E. R. CONEY and K. F. GILL

A well established theoretical method for evaluating the heat transfer by conduction within a solid body is used to estimate steady and transient heat transfer through a typical diesel engine piston. The method is that of using an electrical passive network to simulate the mathematical finite difference representation of the temperature variation in a solid body.

ANY important problems in enginmeering and physics require the solution of partial differential equations of the form $\nabla^2 f = g$ (Poissons Equation) where f is an unknown scalar function of the space coordinates x, y, and z. In most practical cases no rigorous solution can be found and either a numerical solution⁽¹⁾ or an experimental analogy⁽²⁾ must be used. The former is not normally suitable unless computing facilities are available because of the amount of tedious computational work necessary to obtain reasonably accurate results. Of the latter, the most adaptable technique, requiring only a small initial capital outlay to obtain a solution, is the electrical resistance network analogue⁽³⁾.

An important problem which can be investigated by use of such a network analogue is that of heat transfer within the piston of an internal combustion engine. In the design of pistons for engines, it is essential that adequate allowance be made for thermal stresses, that heat dissipation is adequate to ensure acceptable piston temperatures and that the temperature distribution does not produce undesirable distortion. Hence, it is essential to have a knowledge of how the temperature varies throughout the piston, especially as engine working temperatures are continually increasing due to the perpetual demand for higher powerweight ratios and efficiencies.

From the need to determine these temperature distributions experimentally, the following methods have been used:

(a) Temperature-sensitive paints: this method relies upon the irreversible colour change of these paints on reaching certain known temperatures.

(b) Hardness recovery: it is known that the hardness of certain alloys varies with prolonged temperature subjection. Hence, by determining the change in hardness of the piston material after a prescribed period of running, an estimate of the temperature

Notation: a-Mesh size. C-Electrical capacitance. c-Specific heat (volume basis). k-Thermal conductivity. Q-Heat transferred. R-Electrical resistance. t, T-Time. V-Voltage. x, y, z-Spatial co-ordinates.

α—Thermal diffusivity. μ-Density. θ-Temperature.



Fig. 1 Flow diagram of the program used to acquire spectral estimates of surface roughness data

affected by the smoothing techniques used in its calculation. This requirement is satisfied by each of the smoothing techniques used here. Segment averaging is equivalent to averaging the transforms of a number of separate traces, and the application of an appropriate spectral window is not normally considered to affect appreciably the area under the spectral curve unless a significant peak of bandwidth less than that of the window bandwidth is present.

A computer program was specifically developed to enable a comparison to be made of the spectral estimates of the surface data with the form of the power spectrum predicted by equation (1). A flow diagram of the program is shown in Fig. 1 (7).

4 MACHINED SPECIMEN SURFACES

A range of mild steel end-milled, fly-milled, slab-milled, shaped and turned surface specimens was prepared. The machining parameters were speed of cut, feed rate and depth of cut. By maintaining in turn two of these parameters constant, three specimens were produced for each of three selected values of the remaining parameter, resulting in a group of 27 different specimens for each of the machining parameters were purposely selected in this manner so as to aid the identification of the cause of features in the spectral estimates.

The machining speeds were greater than those normally used in most of the series of investigative trials described in the literature; those adopted here are, in fact, more typical of industrial machining speeds. A range of ground mild steel specimens was also prepared (Fig. 3).

Proc Instn Mech Engrs Vol 199 No C4

Slab-milled											
Speed r/min 25					72			206			
<u>Feed rate</u> mm/min		32	104	254	32	104	254	32	104	254	
crt	0.13	1	2	3	10	11	12	19	20	21	
oth of mm	0.25	4	5	6	13	14	15	22	23	24	
Det	0.51	7	8	9	16	17	18	25	26	27	

	Shaped											
Speed r/min			10			20			30			
Feed rate mm/min		0.30	0.63	0.94	0·30	0.63	0.94	0·30	0.63	0.94		
IJ		0.13	1	2	3	10	11	12	19	20	21	
oth of	E E	0 · 25	4	5	6	13	14	15	22	23	24	
Der		0.51	7	8	9	16	17	18	25	26	27	

	End-milled											
Speed r/min			105			230			530			
Feed rate mm/min		12.7	31.8	63•4	12.7	31.8	63·4	12.7	31.8	63•4		
E		0.13	1	2	3	10	11	12	19	20	21	
th of	E	0.25	4	5	6	13	14	15	22	23	24	
Ğ		0.51	7	8	9	16	17	18	25	26	27	

	Fly-milled											
S r	ipeed /min	230			530			810				
	Feed rate nm/min	12.7	31 · 8	88.9	12.7	31.8	88.9	12.7	31.8	88·9		
cut	0.13	1	2	3	10	11	12	19	20	21		
ath of	0.25	4	5	6	13	14	15	22	23	24		
Del	0-51	7	8	9	16	17	18	25	26	27		

	Turned												
S r	ipeed /min	125			260			540					
<u>I</u>	² eed rate nm/min	20	40	80	40	80	160	80	160	320			
Tu	0.13	1	2	3	10	11	12	19	20	21			
ath of mm	0.25	4	5	6	13	14	15	22	23	24			
Der	0.51	7	8	9	16	17	18	25	26	27			

Fig. 2 Machined specimens

© IMechE 1985

\square	Feed mm	0.25	0.5	0.1
3	0.005	1	2	3
Depth of mm	0.01	4	5	6
	0.04	7	8	9

Fig. 3 Ground specimens

5 RESULTS OBTAINED FOR COMPARISON WITH THE THEORY OF VAN DEUSEN

The data acquired from the specimens were obtained relative to a straight-line datum attached to a Talysurf 4 profilometer and were unfiltered, except that they were re-evaluated with respect to the least-squares line fitted to the data. All the power spectral estimates closely approximated to a straight line of slope -2 over the greater part of the frequency band available for investigation. Examples of the spectral estimates are shown in Fig. 4; the broken line is the best-fitting least-squares line, having a slope of -2, which can be applied to the spectral estimate curve in each figure. Compared with the form of the results predicted by equation (1), those shown in Fig. 4f are the worst encountered in this group



Fig. 4 Fourier spectral estimates for the roughness band © 1MechE 1985

of trials. For the frequency band investigated, it was noted firstly that the slopes of a number of the spectral estimates had a smaller negative value at surface frequencies less than of the order of 20 kHz (corresponding to a wavelength of 0.05 mm), for example Fig. 4e. Secondly, it was observed that 16 of the 144 specimens investigated displayed significant machining peaks, for example as shown in Fig. 4b and f. Machining peaks are defined here as those peaks which repeatedly occurred during successive calculations of spectral estimates of data obtained from separate, parallel traces of the same surface. The term 'significant' is used to mean that the peaks were of greater magnitude than that of the confidence interval plotted.

A logarithmic power axis permits a confidence interval to be plotted (8) which is independent of the frequency at which the individual spectral lines are drawn, and is calculated from a chi-square distribution having a number of degrees of freedom dependent upon the number of data transform blocks used and also upon the bandwidth of any spectral window applied.

6 CALCULATION OF A NEW SURFACE TEXTURE PARAMETER

Since the parameter k is defined as the intercept of a Fourier spectral curve on a logarithmic power axis (5), anyone wishing to comprehend how the estimated values for k are obtained would need to possess an understanding of random data processing theory. Equation (4) predicts that the variance of the profile, which is the area under the power spectral curve at wavelengths less than or equal to L, is directly proportional to the length of the trace, and hence the standard deviation of the data, for a fixed traverse length, is a parameter which would also locate the straight line described by equation (5). In order to facilitate the comparison of results which may be computed from a range of trace lengths, a standard length of 1 mm was chosen and results calculated from different traverse lengths suitably scaled to provide the root mean square (r.m.s.) value which, by assuming equation (1) applies to surface texture data, would have been obtained had a 1 mm trace been taken. Because roughness data have a zero mean value and r.m.s. values are more common in surface texture analysis, an r.m.s. value is employed in preference to a standard deviation value. Three methods are now described which enable estimates of the r.m.s. value per millimetre, denoted by $\Psi_{1 \text{ mm}}$, to be computed:

1. If the spectral estimates are of the form predicted by equation (1), then, letting f_{max} and f_{min} be the maximum and minimum surface frequencies respectively of the frequency band currently being investigated, the estimated variance in this band, $\hat{\sigma}^2$, is

$$\hat{\sigma}^{2} = \hat{k} \int_{f_{\min}}^{f_{\max}} \frac{1}{f^{2}} df$$
$$= \hat{k} \left(\frac{1}{f_{\min}} - \frac{1}{f_{\max}} \right)$$

where \hat{k} is an estimate of k. Since f_{\max} is approximately two orders of magnitude greater than f_{\min} , and letting $f_{\min} = f_{1 \text{ mm}}$, where $f_{1 \text{ mm}} = 1$ kHz, then

Proc Instn Mech Engrs Vol 199 No C4

the total variance in the wavelengths equal to and less than 1 mm, that is the square of $\Psi_{1 \text{ mm}}$, is

$$\hat{\Psi}_{1\ mm}^{2} = \frac{\hat{k}}{f_{1\ mm}} \tag{6}$$

If $\hat{P}_{1 \text{ mm}}$ is the estimated power at a wavelength of 1 mm, letting f_a and \hat{P}_a be respectively the arithmetic mean of the logarithms of the surface frequency values and the arithmetic mean of the logarithms of the estimated power values in a spectral estimate, and since equation (5) describes a straight line, then

$$\log_{10} P_{1 \text{ mm}} = 2(f_a - \log_{10} f_{1 \text{ mm}}) + P_a$$

and, using equation (1), this yields

$$\hat{\Psi}_{1\,\text{mm}} = 10^{[f_a + (1/2)\hat{P}_a - (1/2)\log_{10}f_{1\,\text{mm}}]} \tag{7}$$

enabling an estimate of $\Psi_{i mm}$ to be obtained from a combination of the estimated spectral values for surface data which exhibit a zero mean value.

- 2. If equation (5) were to apply at all frequencies of engineering interest, then, substituting one logarithmic power value and the logarithm of the surface frequency at which it occurs for \hat{P}_a and f_a respectively in equation (7), this would enable an estimate of $\Psi_{1\mbox{ mm}}$ to be obtained. The computation of an estimate of a single spectral value is performed mathematically more efficiently by a discrete Fourier transform, which requires 2N complex products and additions, than by the fast Fourier transform (method 1), which requires as many operations in the calculation of a single transform coefficient as in the computation of the entire transform algorithm (9). The major disadvantage of using the estimate of a single spectral value is that it may coincide with a machining peak, resulting in an overestimate of the value of $\Psi_{1 m}$
- 3. The estimated mean value \hat{P}_m of the estimates of the power spectral values $\hat{P}(i)$ for an N-point transform is

$$\hat{P}_{m} = \frac{2}{N} \sum_{i=1}^{N/2} \hat{P}(i)$$

since $\hat{P}(0) = 0$ for data exhibiting a zero mean value. The estimated variance of the N data values f(i) is

$$\hat{\sigma}^2 = \frac{1}{N} \sum_{i=0}^{N-1} f^2(i)$$
 (8)

or, alternatively,

$$\hat{\sigma}^2 = \sum_{i=1}^{N/2} \hat{P}(i)$$

(

$$\hat{P}_{m} = \frac{2\hat{\sigma}^{2}}{N}$$

Equation (1) yields

$$\hat{k} = \frac{2\hat{\sigma}^2}{N\Upsilon_{\rm m}}$$

where Υ_m is the mean value of the inverse squares of the frequency values at which the N/2 spectral lines

Proc Instn Mech Engrs Vol 199 No C4

are plotted, and is

$$\Upsilon_{\rm m} = \frac{2}{N} \int_{f_{\rm min}}^{f_{\rm max}} \frac{{\rm d}f}{f^2}$$

By neglecting the line representing zero frequency and acknowledging that $f_{max} \ge f_{min}$, this gives

$$\Upsilon_{\rm m} = \frac{2}{N f_{\rm min}}$$

Applying equations (6) and (8) yields

$$\hat{\Psi}_{1 \text{ mm}} = \sqrt{\left\{\frac{f_{\min}}{Nf_{1 \text{ mm}}} \sum_{i=0}^{N-1} f^{2}(i)\right\}}$$
(9)

enabling an estimate of the r.m.s. value per millimetre to be obtained without the need for a transformation of the data.

Due to segment averaging, the frequency band of the spectral estimates does not coincide with that affected by the use of a skid datum in preference to a straightline datum (7), and hence the use of a skid is optional if an estimate of $\Psi_{1 mm}$ alone is to be calculated. If the sampled data are obtained relative to a skid datum, the estimation of $\Psi_{1 mm}$ by method 3 will require, as is the case prior to segment averaging, that the data are first re-evaluated with respect to least-squares lines fitted to the individual data blocks, which are of shorter length than those wavelengths affected by the use of a skid datum. A segment length of 0.5 mm is used in the computer program developed.

7 ADVANTAGES OF THE R.M.S. VALUE PER MILLIMETRE COMPARED WITH OTHER SURFACE PARAMETERS

In the comparison with k, the r.m.s. value per millimetre has the following advantages as a surface texture parameter:

- 1. The r.m.s. value per millimetre is defined in a manner which is comprehensible without the need for an understanding of the spectral estimates.
- 2. The calculation time of an estimate of $\Psi_{1 \text{ mm}}$ using equation (9) is similar to the computation time of the roughness parameters, such as Ra, and involves $\log_2 N$ times fewer operations than the evaluation of the transform algorithm used in the estimation of k (5).
- 3. The expected variance of the r.m.s. value per millimetre, where $\Psi_{1 \text{ mm}}$ is obtained using method 3, is of the same numerical order as the expected variance of an r.m.s. value, yet smaller than the expected variance of k, which is of a magnitude dependent upon the mean square value of the data. This is emphasized in the results obtained for a number of the machined surface specimens as presented in Table 1.

The two main advantages of the r.m.s. value per millimetre as a surface texture parameter compared with the currently used roughness parameters are as follows:

1. In contrast to the calculation of the roughness parameter estimates, the calculation of $\hat{\Psi}_{l \ mm}$ does not require that a standard filter is applied to the data

C IMechE 1985

285

Type of specimen and specimen	Parameter means and coefficients of variance								
number	Ψι """	(× 10 ⁶)	$k(\times 10^{14})$						
End-milled, 2	3.92	0.028	1.96	0.071					
Fly-milled, 5	5.36	0.03	3.51	0.065					
Slab-milled, 15	1.01	0.033	0.36	0.065					
Shaped, 6	11.2	0.071	9.44	0.11					
Turned, 20	11.1	0.032	10.2	0.084					
	0.10	0.14	0.051	0.42					

(10). The difficulty arising in surface measurement because of the necessity to choose one of the range of standard filters available, and because of the change in the expected values of the parameters this involves, is thus avoided. The shape of the spectral estimates calculated in Section 5 will be affected by those standard filters of cut-off wavelength less than the segment length, and hence these should not be employed when obtaining an estimate of the r.m.s. value per millimetre. The variations with cut-off wavelength of the r.m.s. value per millimetre were compared with those of the Ra value and the results are presented in Table 2.

2. Except when it relates to profiles having spectral estimates which display relatively large machining peaks, the r.m.s. value per millimetre is able to give a complete description of roughness profiles. This arises since it is a parameter derived from the measurable properties of the surface data, whereas the roughness parameters have been frequently used in attempts to describe some aspect of a surface profile without first acquiring this knowledge. Present

Table 2 A comparison of the variation of the parameters $\Psi_{1 \text{ mm}}$ and Ra with cut-off wavelength for traverses of length 10 mm

Type of	Filter	Parameter values			
specimen and specimen number	wavelength	$\Psi_{1 mm}$ (× 10 ⁶)	Ra μm		
End-milled, 8	0.25	2.14	1.09		
	0.80	3.17	1.71		
	2.50	3.21	1.92		
	Unfiltered	3.30	-		
Fly-milled, 24	0.25	2.19	0.92		
	0.80	2.41	1.24		
	2.50	2.43	1.47		
	Unfiltered	2.51			
Slab-milled, 4	0.25	0.53	0.23		
	0.80	0.66	0.62		
	2.50	0.70	1.28		
	Unfiltered	0.79	_		
Shaped, 5	0.25	6.12	3.43		
•	0.80	9.28	7.27		
	2.50	10.0	10.7		
	Unfiltered	9.95			
Turned, 8	0.25	6.28	3.27		
	0.80	9.02	5.49		
	2.50	9.96	6.92		
	Unfiltered	9.61	—		
Ground, 1	0.25	0.087	0.037		
	0.80	0.091	0.039		
	2.50	0.092	0.040		
	Unfiltered	0.093			

© IMechE 1985

analysis is based on the premise that certain aspects of roughness profiles can be uniquely identified for a particular surface, yet the results presented have shown that the form of the frequency distribution of the roughness data is independent of the machining process, except where machining peaks are present. The traverse length dependent r.m.s. value uniquely determines the 'smoothness' of the surface texture of the machined specimens in the roughness band.

8 SUMMARY

This paper has shown that the results of Van Deusen are applicable to a range of machined surface specimens and that there is an underlying form to the spectral estimates of the surface data in the roughness band. In addition, a new parameter, namely the r.m.s. value per millimetre, has been derived and developed which desicribes the r.m.s. value of the surface roughness data for a 1 mm trace and which is also able to describe completely, apart from the machining peaks, the spectral estimates of the machined surface data at all roughness wavelengths. The new parameter overcomes the problems associated with the parameter proposed by Sayles and Thomas for two reasons: firstly, it is obtained without the need to transform the data and, secondly, its variance is similar to that of the commonly available roughness parameters. In addition, the description of the longer wavelengths introduced by Sayles and Thomas has now been extended to the shorter, roughness wavelengths of the surface profile. In contrast to the means of derivation of current roughness parameters, the new analysis does not involve a subjective assessment of the characteristics of the data, and, moreover, the r.m.s. value per millimetre is obtained without the requirement to select a standard cut-off filter.

If the form of the estimates is the same for all wavelengths which are useful in the engineering sense, then the derived parameter, apart from the machining peaks, will be able to describe completely the form of machined surface data. The data would therefore have been proved to be non-stationary, as predicted by equation (4) and the assessments of Thomas (11).

9 CONCLUSIONS

In surface texture analysis, the maximum wavelength in the band analysed is normally decided by the use of skids and the application of standard filters. All longer wavelengths are generally regarded as being nonstationary, but, prior to this investigation, it was not clear whether this was justified from the observed variations of parameter values with traverse length or because of the inherent nature of the surface texture. At present, surface data processing generally involves the computation of roughness parameters which are capable of describing only certain aspects of the roughness texture. In attempts to remedy this situation, several parameters have been defined which describe certain properties of statistical functions, which are themselves able to characterize the surface roughness. Since these parameters do not account completely for the form of the functions for a sufficiently wide range of surfaces, the usefulness of these parameters in engineering is limited.

Proc Instn Mech Engrs Vol 199 No C4

An advance in the investigation of surface texture arrived with the analysis formulated by Sayles and Thomas (5) based on work initiated by Van Deusen (4). They found that a general spectral form was apparent for the non-stationary longer wavelengths of a surface profile and that this form can be completely described by a single parameter.

The results presented in this paper have verified that the analysis propounded by Sayles and Thomas is also applicable to a range of machined specimens in the roughness band. This work has also resulted in the development of a new parameter, namely the r.m.s. value per millimetre, which is also able to describe completely the form of the spectral estimates predicted by Van Deusen. This parameter has three principal advantages over that of Sayles and Thomas. Firstly, its expected variance is less and is of the same order as the roughness height parameters. Secondly, its estimation can be performed directly from the surface data and hence its calculation time is less than that required for the estimation of the Sayles and Thomas parameter, which is computed from the spectral estimate of the data. Thirdly, in contrast to the Sayles and Thomas parameter, to understand the description provided by the r.m.s. value per millimetre, a knowledge of spectral analysis is not required.

Since the current surface texture band extends to wavelengths as short as 5 μ m, the acquisition of surface data from the stylus instrument requires that a stylus of this order of radius is used when tracing the specimen. This in turn determines the magnitude of the maximum permissible stylus force and the maximum permissible speed of traverse, while ensuring that an adequate transient response is provided. Because the slope of the spectral estimates of the machined specimens is constant and known at this order of wavelength, and because machining errors are unlikely to occur at these wavelengths, a stylus of such a small radius will generally not be required. The use of a stylus which has a radius of the order of 0.1 mm would be adequate for most specimens, while still permitting the estimation of the new parameters developed here, thereby enabling a complete description of the texture at all surface texture wavelengths to be obtained. The tracing of the surface could then be performed at a higher speed and with a larger stylus force, enabling a consequent shortening of the data acquisition time.

REFERENCES

- 1 Peklenik, J. New developments in surface characterisation and measurements by means of random process analysis. Proc. Instn Mech. Engrs, 1968, 182, Pt 3K, 108.
- 2 Spragg, R. C. Surface texture measurements. Metrology Conference, National Engineering Laboratory, East Kilbride. Glasgow, Oct.-Nov. 1971.
- 3 Spragg, R. C. and Whitehouse, D. J. A new unified approach to surface metrology. Proc. Instn Mech Engrs, 1970-1, 185, 697.
- 4 Van Deusen, B. D. A statistical technique for the dynamic analysis of vehicles traversing rough yielding and non-yielding surfaces. NASA Report CR-659, 1967.
- 5 Sayles, R. S. and Thomas, T. R. Surface topography as a nonstationary random process. *Nature*, 1978, 271, 431.
- 6 Thomas, T. R. Personal communication, 1982.
- Mulvaney, D. J. Identification of surface texture signals by orthogonal transform analysis. PhD thesis, Leeds University, 1983.
 Jenkins, G. M. and Watts, D. G. Spectral analysis and its applica-
- tions, 1968 (Holden-Day). 9 Yuen, C. K. and Fraser, D. Diaital Spectral Analysis 1979 (Pitman
- 9 Yuen, C. K. and Fraser, D. Digital Spectral Analysis, 1979 (Pitman, London).
- 10 BS 1134: 1972 (British Standards Institution).
- 11 Thomas, T. R. Defining the microtopography of surfaces in thermal contact. Wear, 1982, 79, 73.

Proc Instn Mech Engrs Vol 199 No C4

PAPER 53

A characterization of surface texture profiles

D J Mulvaney, BSc. PhD. Department of Electronic Engineering, University of Hull

D E Newland. BSc. PhD. CEng. MIMechE and **K F Gill.** BSc. MSc. PhD. CEng. MIMechE Department of Mechanical Engineering, University of Leeds

The spectral analysis of surface texture data is extended so as to include the longer wavelength 'waviness' features in addition to the 'roughness' aspects. This results in the identification of an underlying form in the spectral estimates of engineering surfaces whose complete description requires only two parameters. Also, by showing that the spectral estimates converge to a final steady value at the longer wavelengths, the need to select and apply a standard cut-off filter is avoided.

1 INTRODUCTION

The previous investigation of surface texture by transform analysis, described in (1), was confined to a band of surface wavelengths within the approximate range 10 μ m to 1 mm. This range is approximately that of the roughness band normally used in surface analysis. A more complete investigation will now be described, involving the inclusion of surface wavelengths outside this band.

It is not common engineering practice to analyse surface data of wavelengths less than about 10 μ m. This is because surface undulations having wavelengths of the same order or shorter than the physical dimensions of the type of stylus normally used in surface instruments cannot be accurately reproduced. However, this is not considered to be a disadvantage by the authors, since, in this wavelength region, the slope of the spectral curves is approximately -40 dB/decade and, moreover, machining peaks do not occur in the region (2). Hence the magnitudes of the spectral values obtained for wavelengths shorter than about 10 μ m are, for practical purposes, negligibly small when analysed as part of the roughness profile.

A characterization of surface data in the roughness band is considered complete by most engineers involved in the field of surface measurement. However, the analysis of wavelengths in the 'waviness band', which consists of wavelengths longer than those in the roughness band, has recently begun to receive increased attention and is of interest to the authors for the following reasons:

1. It has not as yet been shown whether the power spectrum $G(f) = k/f^2$, predicted experimentally by Van Deusen (3), applies at wavelengths longer than those in the roughness band (1). The slopes of a number of the spectral curves in the longer wavelength region of the roughness band have smaller negative values than those predicted by Van Deusen, these values becoming decreasingly negative as the wavelength increases. A further investigation is necessary to determine whether this trend continues into the waviness band.

The MS was received on 31 May 1985 and was accepted for publication on 11 October 1985. 2. The expected surface frequencies (1) of some machining peaks, defined below, are less than those in the roughness band and, for completeness, it is necessary to include these in the analysis. Machining peaks (2) are defined here as those peaks which repeatedly occurred during successive calculations of spectral estimates of data obtained from separate parallel traces of the same surface. The surface frequency at which these peaks occur depends on both the speed and feed rate of the machining process. For example, the machining peak in Fig. 1 occurs at a surface frequency of approximately 8 kHz, which can be calculated by dividing the machining speed in revolutions per minute by the feed rate in metres per minute.

The spectral estimate in Fig. 1 was calculated for turned specimen number 25 (Table 1) which displayed features termed 'pick-up marks' on its surface. These marks probably arise when, due to the high temperatures generated during machining, a piece of metal previously cut from the surface becomes momentarily attached to the cutting tool, tempo-



0263-7154/86 S2.00 + .05

Proc Instn Mech Engrs Vol 200 No C3



7

167

39/86 © IMechE 1986

					Slab	-mille	d				
	r r	ipeed /min	25			72			206		
\backslash	<u>Feed rate</u> mm/min		32	104	254	32	104	254	32	104	254
5		0.13	1	2	3	10	11	12	19	20	21
oth of	a B B	0.25	4	5	6	13	14	15	22	23	24
ß		0.51	7	8	9	16	17	18	25	26	27

Table 1 Machined specimens

					Sh	aped					
Speed r/min				10			20		30		
$\left[\right]$	/ 1	Feed rate nm/min	0.30	0.63	0.94	0·30	0.63	0.94	0·30	0.63	0.94
Ğ		0.13	1	2	3	10	11	12	19	20	21
th of	E	0.25	4	5	6	13	14	15	22	23	24
D	Dep	0.51	7.	8	9	16	17	18	25	26	27

				End	-milled	d		_		
<u>-</u>	Speed /min		105		230			530		
Feed rate mm/min		12.7	31.8	63 • 4	12.7	31.8	63 • 4	12.7	31.8	63·4
E	0.13	1	2	3	10	11	12	19	20	21
jo Hin	0.25	4	5	6	13	14	15	22	23	24
ð	0.51	7	8	9	16	17	18	25	26	27

				Fly-	milled					
<u>-</u>	Speed /min		230			530			810	
Feed rate mm/min		12.7	31.8	88-9	12.7	31.8	88.9	12.7	31.8	88.9
E	0.13	1	2	3	10	11	12	19	20	21
nn of	0.25	4	5	6	13	14	15	22	23	24
å	0.51	7	8	9	16	17	18	25	26	27

			<u></u>	n	irned					
	Speed /min		125		ŀ	260			540	
<u>Feed rate</u> mm/min		20	40	80	40	80	160	80	160	320
IJ.	0.13	1	2	3	10	11	12	19	20	21
un of	0.25	4	5	6	13	14	15	22	23	24
ă	0.51	7	8	9	16	17	18	25	26	27

Proc Instn Mech Engrs Vol 200 No C3

1 mm

Fig. 2 Photograph of end-milled specimen number 27 displaying grooves due to 'pick-up' during machining

rarily causing the depth and other aspects of the nature of the maching to be changed (Fig. 2). Such marks were found only on the turned, end-milled and fly-milled specimens. Machining peaks were absent from the spectral estimates of those surfaces with no pick-up marks. A joint investigation of machining parameters and surface profiles revealed that the high temperatures necessary to cause pick-up marks were probably the result of a combination of high cutting speed and a relatively deep cut. For example, a prominent machining peak does not occur in the spectral estimate of turned specimen number 9 in Fig. 3, although the expected frequency of any machining peak is approximately the same as that expected for turned specimen number 25 in Fig. 1. This is probably due to the fact that the cutting speed for specimen number 9 was one-quarter of that

30.0 25.0 20.0 Ē 15.0 10.0 5.0 0.0L 0.0 4.0 8.0 12.0 16.0 $f \times 10^{-4}$ Hz

Fig. 3 Spectral estimate of turned specimen number 9

©IMechE 1986





than at the lower speeds adopted in most of the previously published experimental trials.

- 3. Several cut-off wavelengths in the range 0.08-8 mm are recommended by the British Standards Institution (4) and the selection of the appropriate filter for a particular surface specimen rests on the experience of the stylus instrument operator. Spectral analysis of the waviness band taken in conjunction with the roughness band would determine whether the selection of the cut-off wavelength could be made on a more sound engineering basis.
- 4. The spectral analysis referred to in (3) would also establish whether there is a valid engineering reason for restricting the analysis to the roughness band. This would be confirmed if the form of the spectral estimates in the roughness band described in (1) is found to continue unchanged into the waviness band, since the form of the estimates at all engineering frequencies would then be known. However, if as reported in the literature, the form of the spectral estimates is dictated by the particular path traced by the stylus on a surface to a greater extent than would be expected by statistical sampling variations, then the investigation of surface texture by the present stationary analysis techniques will not be acceptable at these longer wavelengths. Consequently, it would be necessary to continue to confine the range of investigation to the stationary shorter wavelengths until an instrument capable of analysing a significant proportion of the available surface is developed.

2 A NEW SPECTRAL SMOOTHING TECHNIQUE WHICH PERMITS THE ANALYSIS OF THE LONGER WAVELENGTHS OF THE SURFACE TEXTURE

As far as the authors are aware, waviness has not been previously investigated using spectral analysis owing to the difficulty in formulating an appropriate spectral smoothing procedure. Although in (1) traces of length up to 60 mm were taken, the application of a spectral window and segment averaging of an appropriate length reduced the maximum wavelength present in the spectral estimates to 0.5 mm. To smooth the spectral estimate curves (that is to reduce the variance of the spectral estimates) without truncation of the lower frequency band, the Fourier transform of the data was performed in a single segment and a non-recursive second-order filter applied to the spectral data. The smoothed spectral estimate is computed by forming a weighted sum of the discrete, unsmoothed spectral values. For a point which is a distance n points in the unsmoothed spectral estimate from the point for which the smoothed spectral value is to be calculated, the weighting value α is given by

$$\alpha = \frac{2\pi}{N}, \qquad n = 0$$

$$\alpha = \frac{4\pi}{N} \left(1 - \pi \frac{n}{N} \right) e^{-2\pi n/N}, \qquad n = 1, 2, \dots, 2N$$

where N is the cut-off length in terms of the number of spectral points. Two smoothed spectral values are then computed and plotted for each cut-off length. By employing the technique of 'window closing', a smooth-

©IMechE 1986

ing filter of the same cut-off length was found to be appropriate for all the specimens tested, for a given traverse length.

There are two main disadvantages in analysing this extended wavelength band of waviness and roughness combined, compared with analysing the roughness band only. The first disadvantage is that the computation time is longer for the same number of sampled data values, because the data are transformed in a single block rather than in a number of successive, shorter blocks. For N data values, where N is a positive integral power of 2, the calculation of the Fourier transform in a single segment requires that $2N \log_2 N$ computations be performed. However, if the transform is performed in blocks each of length M data values, where M is a positive integral power of 2 and N > M, only 2N $\log_2 M$ computations need to be carried out. In practice. using the filtering technique described above, the calculation times of the smoothed spectral estimates. obtained using a VAX 11/780 computer running FORTRAN, were approximately 35 seconds for a 60 mm trace, 25 seconds for a 30 mm trace and 11 seconds for traces of 15 mm or less. The second disadvantage is that a skid cannot be employed to simplify the setting-up procedure when collecting the data, since the band of wavelengths to be analysed coincides with those wavelengths affected when using a skid datum.

3 RESULTS OF THE SPECTRAL ANALYSIS OF COMBINED ROUGHNESS AND WAVINESS PROFILES

Figure 4 shows typical spectral estimates of skidless traces of length 60 mm computed from a number of the specimens listed in Tables 1 and 2. The digital filter outlined in Section 2 was used to smooth the spectral estimates and its application truncated the surface frequency content to a maximum of approximately 70 kHz (corresponding to a surface wavelength of about 14 μ m), but this is not considered to be a disadvantage. At the higher surface frequencies in the band investigated, the spectral estimates displayed the slope predicted by Van Deusen (1), but the slopes of the spectral curves displayed a smaller negative value than that predicted by Sayles and Thomas (5) at lower frequencies. In agreement with the predictions of Reason (6) for the majority of the surface specimens, this slope apparently tended to zero with increasing wavelength, and hence the area under the spectral curve, that is the variance of the data, converges to a finite value and does not diverge as predicted in (1), which was derived as a consequence of Van Deusen's investigations. For these surface specimens, the surface texture data are consequently stationary, contrary to the findings in the liter-

Table 2 Ground specimens

\setminus	Feed mm	0·25	0.5	0.1		
cri	0.005	1	2	3		
pth of mm	0.01	4	5	6		
മ്	0.04	7	8	9		

Proc Instn Mech Engrs Vol 200 No C3





ature (5, 7 and 8). However, the spectral estimates of a few of the specimens did not follow this trend, for example that illustrated in Fig. 4e, and the reasons for this behaviour are investigated in the subsequent section.

Since for most of the specimens there is a difference between the shapes of the estimates at low and at high surface frequencies in the band investigated, it could be argued that there is also a difference between the forms of roughness and waviness. An exact surface frequency at which a physical discontinuity exists between waviness and roughness is, however, not obvious in any of the spectral estimates.

4 INVESTIGATION OF THE ERRORS OF FORM

In the previous section, it was pointed out that the values of the slopes of the spectral estimates of a few of



Fig. 5 Surface profiles of two fly-milled specimens

Proc Instn Mech Engrs Vol 200 No C3

©IMechE 1986

171

A CHARACTERIZATION OF SURFACE TEXTURE PROFILES

the specimens did not appear to be tending to zero at the longer wavelengths. For a number of these particular specimens, the depth of cut of the machining changed visibly along the length of the specimen in the direction traced. An examination of the profiles acquired from the remainder of these specimens, whose spectral estimates did not converge, revealed that they had also been machined in a non-uniform manner. For example, Fig. 5a shows the profile of a specimen whose spectral estimate diverged at longer wavelengths and Fig. 5b shows the profile of a specimen whose spectral estimate tended to zero slope at longer wavelengths. The occurrence of a non-convergent spectral estimate could always be directly linked to a non-uniformly machined specimen. The possible causes of such machining errors, termed 'errors of form', are discussed in (2). Algorithms for detecting and assessing these machining errors are investigated in Section 7.2.

5 SPECTRAL ESTIMATE MODELS WHOSE DESCRIPTION INVOLVES A SMALL NUMBER OF PARAMETERS

Since the spectral estimates of those specimens manufactured without errors of form are nominally of identical shape, several models of these estimates which can be described by a small set of parameters are considered. One method suggested in the literature for obtaining a small set of parameters from data with known spectral estimates is that of fitting an autoregressive moving-average (ARMA) process (9) to the data. However, the power spectra of those ARMA processes of low order do not correspond closely to the shape of the spectral estimates found from this work. The authors believe, therefore, that these models are of little practical use in surface texture measurement. A study of the frequency characteristics of linear systems resulted in the identification of the shape of the spectral estimates as closely resembling that of the gainfrequency response of a linear second-order system with unity damping factor. This model had an equation of the form

$$G(f) = \frac{K}{1 + (f/f_e)^2}$$
(1)

where K is the intercept on the vertical power axis and f_c is the cut-off surface frequency. Equation (1) extends the analysis of (1) to lower surface frequencies, since for $f \ge f_c$ this is of the same form as the equation predicted experimentally by Van Deusen. Note that equation (1) is the Fourier transform of the exponential autocorrelation function used by Whitehouse and Archard (10) to describe certain aspects of roughness data.

Since equation (1) can be written in the form

$$\frac{K}{G(f)} = 1 + \left(\frac{f}{f_c}\right)^2 \tag{2}$$

a linear least-squares line could be used to obtain estimates of the values of the parameters K and f_c from the spectral estimates of the surface data. However, because this curve fitting involved the calculation of frequency values to the power 4, and the frequency and power values in the spectral estimates vary over a range of 3-4 decades, this method proved to be ill-conditioned. Instead, a least-squares successive approximation was employed which involved fitting the logarithmic version of equation (1) to the logarithmic spectral estimate. Figure 6 shows typical curves of the form of equation (1) fitted to the spectral data of specimens manufactured with no machining errors. The fitted curves approximated to the spectral estimates within the confidence interval at all wavelengths in the band investigated.



Fig. 6 Fourier spectral estimates

@IMechE 1986

Proc Instn Mech Engrs Vol 200 No C3

Hence, for a sufficiently long traverse length, the underlying form of the spectral estimates could be described completely by only two parameters, namely K and f_c in equation (1).

6 MEANINGFUL PARAMETERS

Although K and f_e are able to describe completely the form of surface data in the frequency domain, current engineering practice requires the following alternative description to guarantee a sound understanding. An estimate of the variance of the acquired data, $\hat{\sigma}^2$, can be obtained from the area under a curve represented by equation (1); hence

$$\hat{\sigma}^{2} = \hat{K} \int_{f_{\min}}^{f_{\max}} \frac{df}{(1 + f/\hat{f_{c}})^{2}}$$
(3)

where \hat{K} and \hat{f}_c are the estimated values of K and f_c respectively and f_{max} and f_{min} are the maximum and minimum surface frequencies in the spectral estimate. Integrating equation (3) yields.

$$\hat{\sigma}^2 = \hat{K}\hat{f}_{c} \left\{ \tan^{-1} \left(\frac{f_{max}}{\hat{f}_{c}} \right) - \tan^{-1} \left(\frac{f_{min}}{\hat{f}_{c}} \right) \right\}$$
(4)

enabling the variance of the data to be estimated from the parameters \hat{K} and \hat{f}_e . If $f_{\max} \to \infty$ and $f_{\min} \to 0$, equation (4) becomes

$$\hat{\sigma}_0^2 = \frac{\pi \hat{K} \hat{f}_c}{2} \tag{5}$$

where $\hat{\sigma}_0^2$ is the total area under a curve of the form of equation (1). Dividing equation (5) by equation (4) yields

$$A = \frac{\pi}{2} \left\{ \tan^{-1} \left(\frac{f_{\max}}{\hat{f}_c} \right) - \tan^{-1} \left(\frac{f_{\min}}{\hat{f}_c} \right) \right\}^{-1}$$
(6)

where A is an adjustment factor. By forming the product of the variance of the data with A, an estimate of the total area under a curve of the form of equation (1) can be obtained.

Two parameters can now be defined. The first parameter is termed the 'surface r.m.s. value', denoted by ψ_s , and is the r.m.s. value that would be computed from a trace of infinite length. The value of the parameter is obtained from $\hat{\sigma}_{\sqrt{A}}$. Such a parameter was chosen since r.m.s. values are already established in surface texture analysis. The second parameter is termed the 'surface cut-off wavelength', denoted by λ_c , and is the wavelength corresponding to the surface frequency f_c . The estimate of the surface cut-off wavelength λ_{e} is the wavelength at which the only discernible change in the form of the spectral estimates occurs, and is the half-power point of the spectral estimates. This parameter is purposely termed similarly to that of 'cut-off wavelength', which is employed to define the amplitude response characteristic of the standard filter used in the surface roughness analysis to remove the longer wavelengths on the surface texture. This is no coincidence. Since there are no other wavelengths at which an identifiable change occurs between roughness and waviness, λ_e is the value which the standard filter cut-off wavelengths are used to simulate.

Proc Instn Mech Engrs Vol 200 No C3

Note that if the integral in equation (3) is performed between the limits of f_c and f_{max} , the area under the spectral curve between these limits is

$$\frac{\pi \hat{K} \hat{f}_{c}}{4} \tag{7}$$

and hence the line $f = f_c$ divides two regions of equal area. Thus, the root mean square value of the data at surface frequencies greater than f_c , that is the root mean square value of the roughness data, is half the surface r.m.s. value.

7 IDENTIFICATION OF MACHINING ERRORS AND INSUFFICIENT TRACE LENGTHS

The two sources of machining errors, namely machining peaks and errors of form, which have been identified, cause the spectral estimates to depart significantly from the form of equation (1). For a complete identification of the surface texture of the machined specimens, algorithms for the detection of the machining errors must be employed.

The methods which enable the two types of machining error to be detected are now discussed with a view to including them in an identification program developed for the automatic assessment of surface texture. To verify the presence of errors of form, a check must be incorporated to ensure that the trace is of adequate length, since it is possible under certain circumstances to identify incorrectly non-uniform machining when in reality the traverse length was insufficient.

7.1 Identification of the machining peaks

The physical cause of the machining peaks has been discussed, and such peaks were found to be present in only fly-milled, end-milled and turned surface specimens. It was also found that machining peaks could not be sustained at surface frequencies greater than about 50 kHz. In addition, due to the physical dimensions of the cutting tool, these marks are unlikely to occur at wavelengths longer than a few millimetres. Hence, in order to detect these machining errors, it is necessary only to measure the magnitude of the spectral estimates relative to the fitted curve described in Section 5 in this wavelength band, and to record the maximum difference. If this is of a magnitude greater than that of the confidence interval for that spectral estimate, a machining peak is noted as being identified.

7.2 Identification of the errors of form and insufficient traverse lengths

Errors of form were discussed in Section 4 and resulted in the spectral estimates of the surface texture not converging within the wavelength band investigated. The method used to identify this phenomenon required the estimation of an additional r.m.s. value. This involves splitting the acquired data record into two halves, recalculating the data relative to least-squares lines fitted to each half individually and then computing the mean of the two r.m.s. values obtained from each of these two sample records. An estimate of the surface r.m.s. value obtained from a trace half the length of the one actually

173

A CHARACTERIZATION OF SURFACE TEXTURE PROFILES

collected can then be computed. The change in the estimated surface r.m.s. value by halving the apparent trace length can then be used as a measure of the relative convergence of the estimates as the wavelength increases. If the spectral estimates are such that they are of the form given by equation (1), then the two r.m.s. values should not differ significantly. However, because the estimates of the parameters become worse with decreasing traverse lengths (see Section 10), this method could incorrectly identify errors of form when short trace lengths are used. The method for distinguishing between the errors of form and insufficient trace length was to calculate the ratio of the trace length to the surface cut-off wavelength. If this is less than 2, a reasonable estimate of the surface texture parameters cannot be obtained (see Section 10), and the traverse length is insufficient for an analysis of the form errors.

8 SURFACE IDENTIFICATION PROGRAM

A flow diagram of the program used for the identification of the machined surface specimens is shown in Fig. 7. The surface data must be acquired unfiltered and collected relative to a straight line datum rather than to a skid datum. The program is able to compute estimates of the two parameters described in Section 6. Machin-



Fig. 7 Surface identification program

©IMechE 1986

ing errors due to pick-up marks and non-uniform machining are detected as described in Section 7 and a warning given if these are present. In addition, if by the method described in Section 7.2, the trace is found to be of insufficient length, a further warning is given that a longer trace is necessary for a proper assessment of the surface texture. If the traverse is of insufficient length and/or machining errors are present, values are estimated for the parameters, but warnings are given that the values are likely to be poor estimates.

9 A WRITTEN ASSESSMENT

In addition to providing estimators of the surface texture parameters, the surface identification program also supplies a written assessment of the surface texture. The comments are based on the values of the various parameters calculated for all the machined specimens described in Tables 1 and 2. For the surface r.m.s. and surface cut-off wavelength parameters, these comments depend on the grade of membership of the parameters as given in Tables 3 and 4. Comments which depend on the magnitude of the machining errors, as discussed in Section 7, are also provided by the program and these are given in Tables 5 and 6.

 Table 3
 Comments provided by the surface identification program depending on the grade of membership of the surface r.m.s. value

Surface r.m.s. range	Comment on the magnitude of the surface undulations				
μm					
> 20	Extremely large				
10-20	Very large				
5-10	Large				
2-5	Quite large				
1-2	Greater than average				
0.5-1.0	Average				
0.2-0.5	Less than average				
0.1-0.2	Quite small				
0.05-0.1	Small				
0.02-0.05	Very small				
0.01-0.02	Extremely small				
< 0.01	Out of range				

Table 4	Comments provided by the surface identification
	program depending on the grade of membership of
	the surface cut-off wavelength value

Surface cut-off wavelength range	Comment on the spacing
mm	of the texture peaks
>4	Extremely widely spaced
3–4	Very widely spaced
2-3	Widely spaced
1.5-2.0	Quite widely spaced
1.0-1.5	Above average spacing
0.7-1.0	Average spacing
0.5-0.7	Below average spacing
0.4-0.5	Quite closely spaced
0.3-0.4	Closely spaced
0.2-0.3	Very closely spaced
< 0.02	Extremely closely spaced

Proc Instn Mech Engrs Vol 200 No C3

Table 5	Comments provided by the surface identification
	program depending on the grade of membership of
	the machining peaks

- •	
Maximum deviation of the spectral values from the fitted curve relative to the	
95 per cent confidence	Comment on the
interval	pick-up marks
>4	Badly marked by pick-up
3-4	Very significant pick-up marks
2-3	Significant pick-up marks
1-2	Possible pick-up marks
<1	(No comment given)

 Table 6
 Comments provided by the surface identification program depending on the grade of membership of the errors of form

Percentage increase in the estimated r.m.s. value of the data compared with that of a traverse of half the length employed	Comment on the errors of form
>40	Extremely large errors of form
30-40	Very significant errors of form
20-30	Significant errors of form
10-20	Possible errors of form
0-10	(No comment given)

length. Hence, in order to assess the applicability of the results when traverses of only a few millimetres are available, several traverses of various lengths were collected and the variation of the parameter estimates with traverse length assessed. Figures 8 and 9 show the results for a smooth ground specimen (number 2) and a slab-milled surface (number 15) respectively. For the ground specimen, the surface cut-off wavelength was about 0.2 mm, and hence traverses as short as 0.4 mm were permitted by the surface identification program. For the slab-milled specimen, $\hat{\lambda}_c$ was approximately 6 mm, but the estimates of ψ_s and $\hat{\lambda}_c$ were only significantly affected (by 5 per cent or more compared with the longest traverse) for traverses shorter than twice the surface cut-off wavelength (Table 7).

Table 7	Parameter values and coefficients of variance for a
	number of traverses of different length for a ground
	specimen and for a slab-milled specimen

		Parameter estimates and coefficients of variance averaged over five traverses					
Type of specimen and specimen number	length mm	<u></u>	<u>//s</u> m	 m	im		
Ground, 2	60.0	0.34	0.054	0.12	0.043		
	30.0	0.35	0.071	0.13	0.069		
	15.0	0.36	0.12	0.12	0.12		
	7.5	0.35	0.16	0.14	0.15		
	3.75	0.31	0.21	0.16	0.18		
	1.87	0.38	0.24	0.11	0.22		
Slab-milled, 15	60.0	6.52	0.031	6.07	0.032		
	30.0	6.72	0.042	5.97	0.036		
	15.0	6.91	0.052	5.67	0.051		
	7.5	8.01	0.061	5.64	0.052		
	3.75	8.87	0.087	3.32	0.076		
	1.87	4.77	0.11	1.64	0.092		

10 LIMITATIONS OF THE PRESENT ANALYSIS

The analysis formulated has resulted from gathering data from traverses of length 60 mm, but the variations in the parameter estimates will be related to the traverse



©IMechE 1986

A CHARACTERIZATION OF SURFACE TEXTURE PROFILES



Fig. 9 Fourier spectral estimates of slab-milled specimen number 15

The surface r.m.s. value is obtained from the surface texture data values and hence its expected variance is similar to that of other surface texture parameters, such as the r.m.s. value per millimetre, whose variances are, in turn, similar to that expected of the surface texture height parameters, such as Ra. Because the coefficients of variance of the surface r.m.s. value and the surface cut-off wavelength were generally similar in the tests performed. (Table 7), the variances of both parameters are of the order expected of the currently used roughness parameters.

Since the program permits a number of traces to be analysed simultaneously, data acquired from suitable parallel traces can be used in order to smooth the spectral estimates and hence reduce the expected variance of the estimated parameter values.

11 A SIMPLE THREE-DIMENSIONAL ASSESSMENT

The specimens were purposely manufactured such that the machined face was square and of side 60 mm, and hence a traverse of at least this length could be obtained in any direction on the surface. Figure 10 shows a typical result of taking traces at various angles with respect to that usually chosen, namely at right angles to the direction in which the surface was machined by the tool. The underlying form of the spectral estimates, as described by equation (1), was obtained independently of the direction of the traverse. In addition, it was found that the value of the surface cut-off wavelength for a traverse taken at an angle θ with respect to the usual direction of traverse could be estimated by

$$\hat{\lambda}_{\rm c} = \sqrt{\hat{\lambda}_0^2 \cos^2 \theta} + \hat{\lambda}_{90}^2 \sin^2 \theta \tag{8}$$

@1MechE 1986

where $\hat{\lambda}_0$ and $\hat{\lambda}_{90}$ are the estimated values of the surface cut-off wavelength for traces taken at angles of 0° and 90° respectively to that usually traced. Hence only two traverses are required in order to obtain an assessment of the surface cut-off wavelength for a three-dimensional surface texture. However, no similar relationship could be derived from the surface r.m.s. value, since its maximum value did not often lie in the usual direction of traverse and nor did it vary uniformly with the angle of traverse.

12 GENERAL APPLICABILITY OF THE RESULTS

To determine whether the results collected here are applicable only to the mild-steel machined surface specimens, several other materials were tested. The spectral estimates shown in Fig. 11 were obtained from traces taken from an unmachined rolled bar, a reamed hole, a turned brass bar and a stylus instrument standard specimen which is used for calibrating the Talysurf instrument. The form of these spectral estimates is thus apparently not restricted to machined specimens and further investigations are necessary to determine the extent to which the present analysis applies.

Figure 12a shows the spectral estimate of a ground specimen and Fig. 12b shows the spectral estimate of the same specimen after it has been subjected to wear. The form of the spectral estimates both before and after the wear test conformed to equation (1), and hence the surface texture parameters, whose values changed significantly, were able to describe completely the changes which took place in the surface texture.

Proc Instn Mech Engrs Vol 200 No C3







Proc Instn Mech Engrs Vol 200 No C3

©IMechE 1986



Fig. 12 Spectral estimates of a ground specimen

13 CONCLUSIONS

It has been demonstrated how a complete description of the surface texture of machined specimens can be formulated. If no machining errors are present, two parameters uniquely describe the shape of the spectral estimates and hence completely identify the surface texture. The first parameter is termed the 'surface r.m.s. value' and can be defined as that r.m.s. value of a surface profile obtained from a certain traverse length, greater lengths than which would yield a value of no significant practical difference. The second parameter is termed the 'surface cut-off wavelength' and can be defined as that wavelength which is intermediate between the waviness and the roughness bands of the surface texture.

More precisely, the surface r.m.s. value is the asymptotic value of the r.m.s. value of the data as the traverse length is increased, and the surface cut-off wavelength is the wavelength at which the power in the spectral estimate has fallen to half its maximum value. The surface cut-off wavelength is the approximate wavelength at which the form of the spectral estimate changes, the

©IMechE 1986

half-power point being the most common way of describing such a transition. Since there is necessarily only one such transition in these spectral estimates, this is the only wavelength at which a change from roughness to waviness could conceivably occur.

The derivation of these parameters does not depend on a subjective assessment. They are able to describe spectral estimates of surface profiles both within the 95 per cent confidence interval and throughout the wavelength range investigated of almost four decades. extending from a wavelength of 10 μ m to a wavelength of 60 mm. This result was made possible by developing a digital spectral smoothing technique which did not truncate the low-frequency content of the spectral estimates. By showing that the spectral estimate values converge to a final steady value at the longer wavelengths, the investigations revealed that surface texture data are stationary.

If machining errors are present, then the values of the estimates for these two new parameters will be affected; the surface identification program developed will give a warning of the presence of the errors and provide a written assessment of their significance. For users not familiar with the expected magnitudes of the parameter values, a written account of the magnitude of estimated values is given.

The surface texture description provided by the two new parameters derived here can be seen as having distinct advantages over that provided by the two parameters normally employed, namely the cut-off wavelength value and the Ra value. Although the latter have been widely applied without experimental justification of the accuracy of the surface texture characterization provided, their prevalence has confirmed the usefulness of the description they produce. The cut-off wavelength, which has five possible values (4), is an approximation to that wavelength at which occurs the only discernible change in the form of the spectral estimates, defined here as the surface cut-off wavelength. The value of the cut-off wavelength is estimated solely by the stylus instrument operator from an inspection of the surface, a task requiring considerable skill. The machine computation of the surface cut-off wavelength eliminates the need for a skilled operator and consequently produces a desirable simplification of the surface texture measurement procedure. It is important to point out that since the cut-off wavelength value determines the time constant of the standard cut-off filter, its selection has a significant influence on the values of the roughness parameters, including Ra (11). This problem is overcome in the computation of the surface r.m.s. value, when the need to select and apply a filter to the data is avoided.

The general applicability of these results has yet to be proved, but investigations initiated in Section 12 suggest that there will be a wide application for the analysis formulated here.

REFERENCES

- 1 Mulvaney, D. J., Newland, D. E. and Gill, K. F. Identification of surface roughness. Proc. Instn Mech. Engrs, 1985, 199 (C4), 281– 286.
- 2 Mulvaney, D. J. Identification of surface texture signals by orthogonal transform analysis. PhD thesis, 1983, University of Leeds
- 3 Van Deusen, B. D. A statistical technique for the dynamic analysis of vehicles traversing rough yielding and non-yielding surfaces. NASA Report CR-659, 1967.

Proc Instn Mech Engrs Vol 200 No C3

4 BS 1134: 1972 (British Standards Institution, London).

- B 1134: 19/2 (British Standards Institution, London).
 Sayles, R. S. and Thomas, T. R. Surface topography as a non-stationary random process. *Nature*, 1978, 271, 431.
 Reason, R. G. Report on the measurement of surface finish by stylus methods. Rank Taylor Hobson, 1944.
 Whitehouse, D. J. and Phillips, M. J. The assessment of engineering surfaces using random process analysis. SERC Vacation School, 1981, University of Warwick.
 Benergy D. E. Engengraphic the appreciated of surface topography.
- 8 Reason, R. E. Progress in the appraisal of surface topography during the first half century of instrument development. Presented
- at the International Conference on Metrology and properties of engineering surfaces, Leicester, 18-20 April 1979.
 9 Box, G. E. P. and Jenkins, G. M. Time series analysis, forecasting and control, 1970 (Holden-Day Inc.).
 10 Whitehouse, D. J. and Archard, J. F. The properties of random surfaces of significance in their contact. Proc. R. Soc., 1970, A316, 97, 121 97-121.
- 11 Thomas, T. R. and Charlton, G. Variation of roughness parameters on some typical manufactured surfaces. Department of Mechanical Engineering, Teeside Polytechnic.

PAPER 54

oc. IASTED Int. Symp. Measurement, Signal Processing d Control, MEEO'86, Taormina, Italy, Sept. 3-5, 1986. 's. G. Messina & M.H. Hamza, ACTA Press, P.O.Box 2481, aheim, CA, 92804 USA, ISBN 0-88986-096-3

THE IDENTIFICATION OF SURFACE TEXTURE SIGNALS

David Mulvaney Department of Electronic Engineering University of Hull U.K. David Newland and Ken Gill Department of Mechanical Engineering University of Leeds U.K.

Abstract

The industrial analysis of surface texture entails the processing of data acquired from a stylus which traverses a single track across a surface. Until recently, the sole design criterion for surface texture was that the mean amplitude of the measured surface undulations should be within specified tolerance limits and consequently it was only necessary to obtain a single roughness parameter to indicate the "av-erage height" of the surface undulations. Al-though this is the only parameter normally quoted, an additional parameter is involved during the pre-processing of the data, namely the "cut-off wavelength". This parameter must be estimated by the stylus instrument operator in order that an appropriate filter can be app-lied to the surface texture data before the roughness parameter is computed. However, with the increasingly stringent demands being placed the increasingly stringent demands being placed on the components of mechanical systems, the necessity has arisen for a more precise descrip-tion of the form of surface texture. The work described in this paper is aimed at meeting this need by considering methods of improving the characterization of surface texture data. In particular, an analysis of the complete profile, rather than an analysis restricted to the surface roughness only, is shown to permit the identification of an underlying Fourier spectral shape, whose complete description requires only snape, whose complete description requires only two parameters. The first parameter, termed the "surface r.m.s. value", is an estimate of the root-mean-square value of the texture amplitude for a profile of infinite length. The second garameter, termed the "surface cut-off wave-ength", is the result of automating the computation of the cut-off wavelength. Consequent-ly, the analysis eliminates the need for the stylus instrument operator to select an appro-priate filter, thereby significantly simplifying the data acquisition procedure. The analysis has been incorporated into a surface identification program, which enables these two new para-meters to be calculated and also permits the presence of the two types of surface machining error isolated in this work to be detected.

1. Introduction

The purpose of investigating surface texture is the determination of engineering properties. For example, under the same operating conditions, a smooth surface will generally wear at a slower rate than a rough surface, yet a certain degree of roughness is required in order to maintain an oil film which is of sufficient thickness for lubrication purposes. Other important properties which are dependent upon

105-069

surface texture include friction, corrosion, fatigue and both thermal and electrical conductivity.

The stylus instrument is the most popular device available for the assessment of surface texture. This instrument produces an analogue of the surface texture by amplifying the vertical deflections of a stylus of radius approximately 2 μ m traversed across the surface at a speed of about 1 mm s⁻¹. A typical surface texture signal produced by such an instrument is shown in figure 1.

A variety of methods exist for processing surface texture signals; many of these methods are provided by stylus instruments. The most common procedure is to first high-pass filter the signal in order to reduce the effect of the overall shape of the surface on any subsequent analysis which is performed. Secondly, a single roughness parameter dependent upon the average height of the signal, usually a root-mean-square (r.m.s.) parameter, or a similar quantity, is calculated. There are three major disadvantages associated with this procedure and these are now considered.

(1) The cut-off wavelength (wavelength is used in preference to frequency in surface texture analysis) of the high-pass filter is selected by the stylus instrument operator from a set of five standard filters[1]. The cut-off wavelength values of these standard filters cover a range of two decades of wavelength. Because of this limited number of possible cutoff values, an estimation error of 50% will commonly result, thereby contributing to a significant error in the value of the roughness parameter which is subsequently calculated. Moreover, there is also the possibility of human error in the selection procedure itself. Consequently, there are two important improvements which could be made to the pre-processing of surface texture signals, namely to automate the procedure of filter selection and to permit the cut-off wavelength to be selected from a continuous range of cut-off values.

(2) There is no sound engineering basis for the need to filter the surface texture signal. Hence, a detailed analysis is needed in order to determine on what particular physical attribute of the surface texture such filtering is based.

(3) With the increased use of surface texture analysis, the description provided by a single average height parameter is now proving inadequate. The need for a more detailed description of surface texture signals has been reflected in the introduction in recent years of stylus instruments capable of calculating a large number, often more than 30, new parameters[2]. Rather than providing a solution, this profusion has bred confusion[3]. In particular, there are no general agreements regarding the use of any one parameter in any given set of circumstances, or in what way physical phenomena are reflected in parameter values. Consequently, there exists a requirement for a new parameter, or small number of parameters, which can describe completely the form of surface texture signals.

2. Analysis procedure and results

A large range of analysis methods is available to aid the identification of signals[4]. In this work, various orthogonal transforms were compared as to their ability to characterize surface texture signals[5]. From this initial comparison, the Fourier transform was selected for further investigation. The following processing methods were adopted.

(1) The normal procedure of high-pass filtering was avoided, but a best-fitting line in the least-squares sense was computed and the data values recalculated relative to this line.

(2) To assist in the detection of any overall pattern in the spectral estimates of the signal, traces of over 50 mm in length were acquired. This is of an order of magnitude longer than that normally used for surface texture analysis.

(3) The raw spectral estimates were smoothed by the oplication of non-recursive second-order dici al filter. This was applied in such a manner that the lowest frequency present in the spectral e timates remained the same as that in the raw estimates.

(4) A total of 144 mild-steel test specimens were prepared by a range of processes, producing end-milled, fly-milled, slab-milled, shaped, turned and ground surfaces[6]. The untroken lines shown in the spectral estimates of figure 2 are typical of the results obtained from the test specimens. In this figure, frequency is the recirrocal of the wavelength actually present on the surface; for example 1 kHz corresponds to a wavelength of 1 mm. These results demonstrate that there is a common form to the spectral estimates of surface texture data.

3. Identification of the spectral form

To characterize the spectral estimates, a number of mathematical functions and models, for example autoregressive moving-average processes, were investigated. The most consistent results were produced by the function

$$G(f) = \frac{K}{1 + (f/f_c)^2},$$
 (1)

where G(f) is the spectrum, f is frequency and K and f_c are parameters determined from the surface under investigation. The broken lines in figure 3 are curves of this form which give the best fit in a least squares sense to the spectral estimate of the surface under investigation. In fact, such a fit lies inside the 95% confidence interval, except where one, or both, of the conditions outlined below occurred.

(1) The spectral estimates of a few of the test specimens did not converge to a constant value with decreasing frequency, for example, figure 3. Inspection of the surfaces of such specimens revealed that they displayed "errors of form", that is, they had not been uniformly prepared. In fact, it is probable that during production either the machining parameters had changed, or the position of the machining bed had altered.

(2) A number of test specimens displayed "machining peaks" in their spectral estimates, figure 4. Such peaks occurred only for those specimens which displayed "pick-up marks" on their surface, resulting in a texture which, over relatively short lengths, was almost sinusoidal in nature.

4. Surface identification program

The two main requirements for a surface identification program based on the new results are seen to be as follows.

(1) The parameters which describe the spectral estimates need to be comprehensible in terms of parameters already in common use in surface texture analysis. The parameters K and f_c in equation 1 completely describe the spectral estimates, but to avoid difficulty in understanding of the meaning of the parameter K, the following alternative description is adopted. Integrating equation 1 between the limits f_{min} and f_{max} , the minimum and maximum frequencies in the spectral estimate respectively, yields

$$\sigma^{2} = Kf_{c}[tan^{-1}(f_{max}/f_{c}) - tan^{-1}(f_{min}/f_{c})]$$
, (2)

where σ^2 is variance. If $f_{max} \neq \infty$ and $f_{min} \neq 0,$ the right-hand side of equation 2 becomes

$$\pi K f_c/2$$
 (3)

Dividing equation 3 by equation 2 gives

$$A = \frac{\pi}{2} [\tan^{-1}(f_{max}/f_c) - \tan^{-1}(f_{min}/f_c)]^{-1}, \quad (4)$$

where A is an adjustment factor.

Two parameters can now be defined which are not only meaningful to those conversant with surface texture analysis, but also completely describe the form of the spectral estimates obtained from surface texture data. The first parameter is termed the "surtace r.m.s. value", denoted by Ψ_s . Ψ_s is the r.m.s. value that would be computed from a trace of infinite length and its value can be estimated by forming ∂/\overline{A} , where the circumflex denotes an estimated value. The second parameter is termed the "surface cut-off wavelength", λ_c , and is the reciprocal of f_c. The wavelength λ_c , being the halfpower point of the spectrum, is the wavelength at which the only discernible change in the form of the spectral estimate uccurs. Consequently, following the argument presented in the introduction, this is the value which is estimated by the use of standard filters. (2) The program will also need to identify the errors produced during the machining of the surface. The errors of form are identified by detecting whether there is a significant difference between the estimates of $\Psi_{\rm g}$ obtained from two traces, one twice the length of the other. The pick-up marks are recorded as identified if, at any frequency, the difference in power between that indicated by the spectral estimate curve and that shown by the curve fitted to it is greater than the 95% confidence interval.

A flow diagram of the surface identification program developed in this work is shown in figure 5.

5. Conclusions

The proposals outlined in the introduction have been investigated in the work presented in this paper.

(1) The need to select a cut-off wavelength has been circumvented by the new analysis proposed in this paper. The cut-off wavelength, now re-termed the surface cut-off wavelength, is now not only calculated automatically by the surface identification program, but is also obtained from a continuous range of values.

(2) There is only one wavelength in the spectral estimates at which any discernible change in the form of the estimates occurs. The most usual way of defining such a change is the half-power wavelength. This corresponds to the value of the surface cut-off wavelength computed by the surface identification program. Consequently, there does exist a physical attribute exhibited by the surface itself on which the cut-off wavelength is based. However, there is no need to filter the data; indeed the complete analysis presented here has been produced without needing to resort to the manual selection of a filter.

(3) A surface manufactured without machining errors can be completely described by just two parameters. This is a significant improvement on the present method of analysis, which does not offer a complete description and in which the value of the first parameter, used to describe the average height of the texture, is significantly affected by the value of the manually-selected second parameter. Moreover, in the new analysis presented here, a surface identification program is now available which is able to identify certain types of error introduced in the machining process.

References

1. BS 1134: 1972 (British Standards Institution).

2. Dagnell,H. Exploring surface texture. Rank Taylor Hobson, U.K., 1980.

3. Whitehouse,D.J. The parameter rash - is there a cure? Presented at the Second International Conference on Metrology and Properties of Engineering Surfaces, Leicester, U.K., April 14-16, 1982.

4. Oppenheim.A.V. and Shafer,R.W. Digital signal processing. Prentice-Hall Inc., 1975.

5. Mulvaney, D.J. Identification of surface texture signals by orthogonal transform analysis. Ph.D. thesis, University of Leeds, U.K., 1983.

6. Mulvaney, D.J., Newland, D.E. and Gill, K.P. Identification of surface roughness. Proc. Instn Mech. Engnrs, 199, C4, 1985, pp. 281-286.





PAPER 56

A comparison of orthogonal transforms in their application to surface texture analysis

D J Mulvaney, BSc, PhD Department of Electronic Engineering, University of Hull

D E Newland, BSc, PhD, CEng, MIMechE and **K F Gill**, BSc, MSc, PhD, CEng, MIMechE Department of Mechanical Engineering, University of Leeds

The orthogonal transforms most commonly used in engineering applications are compared in their ability to characterize surface texture data. The results suggest that, for this application, the Fourier and cosine transforms are to be preferred because of their ability to model the data more precisely and because of their relatively rapid rates of convergence.

1 INTRODUCTION

The stylus instrument used in the acquisition of the surface data was a Talysurf 4 (Rank Taylor Hobson, Leicester) and the stylus deviations were measured relative to a straight line datum. The stylus was a general-purpose conical type, with a truncated flat tip of radius 2.5 μ m; use of the skid and standard filter were both avoided.

For the data acquired from a range of machined surface specimens, the orthogonal transforms most commonly used in engineering applications, namely the Fourier, Walsh, phase-shift-invariant Walsh, BIFORE, Haar and cosine, are computed. These orthogonal transforms are defined, in discrete terms, in the Appendix. In the transform computations, fast transform algorithms (1) are implemented because they require at least $2^{N}/N$ times fewer operations, where N is the number of sampled data values. The spectral estimates $\bar{G}(f)$ are calculated from the transform coefficients, as shown in the Appendix. In the spectral estimates to be presented, the upper frequency limit is imposed by the physical dimensions of the monitoring stylus and the lower frequency limit results from the application of an appropriate spectral window.

As a result of its more extensive use, greater knowledge exists about the effects of windowing on the Fourier spectral estimates than on the spectral estimates of the remaining orthogonal transforms under consideration. Consequently, only segment averaging was implemented (which is equivalent to averaging the spectral estimates of a number of shorter traverses), and spectral windows were not applied in these investigations due to the unduly favourable advantage this would have conferred on the Fourier estimates. A technique known as window closing was employed. This involves firstly using the version of the window which is of length N, where N is the number of samples in the data record collected, and then successively halving the window length until it is found that there are no significant differences in the spectral shapes when comparing the spectral estimates obtained from consecutive calculations. The process produces successive discrete

The MS was received on 30 October 1985 and was accepted for publication on 5 March 1986. increases in the bias and successive discrete decreases in the variance of the spectral estimate. The chosen spectral estimate is normally considered to display the optimum 'bias/variance trade-off' for that particular window and data record. A window length of approximately 0.5 mm was normally considered appropriate for the spectral estimates, and traverse lengths longer than 20 mm generally gave adequate smoothing, although this was dependent upon the particular orthogonal transform being computed. The computer program developed enables a number of traverses to be analysed simultaneously, and hence the data for analysis need not have been collected entirely from the same trace, but could have been obtained from parallel traces of the same specimen, provided that the traverses were of a length at least equal to that of the spectral window.

The sampling rate adopted to provide the results was approximately 200 Hz which, at the traverse speed of 1 mm/s employed, corresponds to a surface frequency of 200 kHz. Surface frequency is defined here as the reciprocal of the corresponding wavelength on the surface. Approximately 200 sampled data values were therefore collected for each millimetre of surface traversed, although the program ensures that the number of data values collected was always a positive integral power of 2, in order that all the data can be transformed by the fast algorithms. The maximum number of sampled data values which can be collected in the program is 16 384, and hence traverses of lengths up to about 80 mm were permitted at this sampling rate.

A comparison of the performances of the orthogonal transforms is undertaken in the following two sections. Firstly, a subjective comparison is made based on an inspection of the spectral estimates of all the machined specimens listed in Tables 1 and 2. Secondly, the rate of convergence of the spectral estimates is considered, which is important in assessing the quantity of data required in order to characterize the surface texture band being investigated.

2 THE SUBJECTIVE PERFORMANCE OF THE ORTHOGONAL TRANSFORMS

Although all of the machined specimens listed in Tables 1 and 2 were assessed, two specimens are selected to

116/86 C IMechE 1986

0263-7154/86 \$2.00 + .05

Proc Instn Mech Engrs Vol 200 No C6

	Slab-milled									
Speed r/min 25 72				206						
Feed rate mm/min		32	104	254	32	104	254	32	104	254
Ē	0-13	1	2	3	10	11	12	19	20	21
th of	0.25	4	5	6	13	14	15	22	23	24
å	0.51	7	8	9	16	17	18	25	26	27

Lable L Machined Specificity	Table	1	Machined	specimens
------------------------------	-------	---	----------	-----------

				Sh	aped					
	Speed /min		10			20			30	
\setminus	Feed rate mm/min	0.30	0.63	0.94	0-30	0.63	0.94	0.30	0.63	0.94
E	0.13	1	2	3	10	11	12	19	20	21
Ч С Е Е	0.25	4	5	6	13	14	15	22	23	24
ð	0-51	7	8	9	16	17	18	25	26	27

	End-milled									
Speed r/min 105 230 530										
	Feed rate mm/min	12.7	31.8	63·4	12.7	31-8	63•4	12.7	31.8	63·4
Ħ	0-13	1	2	3	10	11	12	19	20	21
nn of	0.25	4	5	6	13	14	15	22	23	24
Å.	0.51	7	8	9	16	17	18	25	26	27

	Fly-milled										
Speed r/min 2			230	230 530					810		
\setminus	<u>I</u> V	Feed rate nm/min	12.7	31.8	88-9	12.7	31-8	88-9	12.7	31.8	88-9
Ē		0·13	1	2	3	10	11	12	19	20	21
th of	EEE	0.25	4	5	6	13	14	15	22	23	24
ă		0-51	7	8	9	16	17	18	25 ·	26	27

	Turned										
-	Speed /min	125			260			540			
	Feed rate mm/min	20	40	80	40	80	160	80	160	320	
ž	0.13	1	2	3	10	11	12	19	20	21	
un of	0.25	4	5	6	13	14	15	22	23	24	
5	0.51	7	8	9	16	17	18	25	26	27	

Proc Instn Mech Engrs Vol 200 No C6

r	•	C	
Ladie	4	Ground	specimens

\square	Feed mm	0.25	0.5	0.1				
cut	0-005	1	2	3				
pth of mm	0.01	4	5	6				
ല്പ	0.04	7	8	9				

display the features which distinguish the spectral estimates of the various transforms. These two specimens are an end-milled specimen (number 11) and a turned specimen (number 19). Figures 1 and 2 show the spectral estimates of the orthogonal transforms obtained from traces of these specimens. Peaks occurred in a number of the spectral estimates at a surface frequency of 7 kHz for the end-milled specimen and at a surface frequency of 6 kHz for the turned specimen. The surface frequency at which these peaks arise can be predicted from the machining speed and feedrate used during the manufacture of the specimen. The term used for such peaks arising in the spectral estimates is 'machining peaks' (2), and several surface profiles were investigated in order to identify the surface features which give rise to these machining peaks. The majority of the machined specimens, however, did not display these machining peaks.

Because of the familiarity gained by the abundant use of Fourier transform analysis in many branches of engineering, the performance of other orthogonal transforms can reasonably be assessed by comparison with the results produced by the application of the Fourier transform.

The Fourier estimates of most of the machined specimens contained spectral coefficients which had their largest values in the lower frequency part of the band investigated, with the spectral coefficients becoming gradually smaller in magnitude with increasing frequency. The only departure from this trend occurred in those specimens which contained a machining peak. The cosine spectral estimates closely approximated to those of the Fourier estimates for all the specimens, although a comparatively larger variance was occasionally present in the cosine estimates, for example in Fig. 1.

The N-point Haar and BIFORE transforms provide only $1 + \log_2 N$ logarithmically spaced spectral coefficients. Because the spectral coefficients become more widely spaced in sequency as the sequency increases, the spectral shapes (Fig. 2c and d) are unable to follow that of the corresponding Fourier estimate. However, the Haar and BIFORE estimates were mostly similar to the underlying shape of the corresponding Fourier and the Walsh spectral estimates. The performance of the BIFORE transform was generally superior to that of the Haar transform in this series of trials, and an example of this can be seen in Fig. 1 where the BIFORE estimate (Fig. 1d) resolved the machining peak which remained undetected by the Haar estimate (Fig. 1c). Since their calculation time was at least one order of magnitude less than that of the other orthogonal transforms in these trials, the Haar, and particu-

C IMechE 1986

A COMPARISON OF ORTHOGONAL TRANSFORMS IN SURFACE TEXTURE ANALYSIS



Fig. 1 Spectral estimates obtained from end-milled specimen number 11

C IMechE 1986

Proc Instn Mech Engrs Vol 200 No C6

.





Proc Insta Mech Engrs Vol 200 No C6

© IMechE 1986
larly the BIFORE, transform could be of use in some applications, including surface texture analysis, should only a quick check rather than a detailed investigation of the spectral content of the data be required.

The Walsh spectral estimates were of the same shape as the corresponding Fourier spectral estimates for most of the surface data, although the Walsh transform normally required a longer traverse length than the Fourier transform in order to provide similar spectral smoothing (Section 3). In contrast to the results of Weide et al. (3), the respective spectral peaks in the Walsh and Fourier spectral estimates were of similar amplitudes, but the Walsh spectral estimates containing machining peaks often exhibited odd-sequency harmonics of the peaks which were not present in the other spectral estimates, for example Fig. 1e. Apart from these odd-sequency harmonics, the problems associated with the variation of the Walsh transform with circular time shifts of the data appear to be of little importance, since their effects are reduced by segment averaging and also since periodic signals are seldom encountered in surface texture analysis.

The circular phase-shift-invariant Walsh spectral estimates were generally of different shapes from those of the other spectral estimates, and many of the spectral coefficients had larger values than the corresponding coefficients encountered in the other estimates. In addition, any machining peaks which occurred were distributed over wider frequency ranges than those in the Walsh estimates. An explanation for these phenomena is now developed by considering the respective algorithms of the Walsh and phase-shift-invariant Walsh transforms. Applying a phase-shift-invariant Walsh transform is equivalent to averaging the Walsh transforms of all the possible time-shifted versions of the data. Each segment in a phase-shift-invariant Walsh spectral estimate will normally contain a less diverse range of leaked sequency components than is found in the individual segments which constitute the Walsh estimates. Hence, on averaging the segments, the leaked values present in the individual segments are reduced more significantly in the Walsh spectral estimates than in the phase-shift-invariant estimates.

The non-orthogonal R transform estimates followed the general shape of the Fourier and Walsh estimates at low sequencies. The computation time is similar to that of the Walsh transform, yet the R transform was unable to detect the machining peaks, which limits its use in surface texture analysis.

3 RATE OF CONVERGENCE OF THE ORTHOGONAL TRANSFORMS

A less subjective assessment of one aspect of the performance of the orthogonal transforms was developed, namely that of the rate of convergence of the spectral estimates. This test consists of taking a series of successively longer traverses until, for a fixed window length, there appeared no engineeringly significant difference between consecutive spectral estimates.

One of the tests carried out is illustrated in Figs 3 and 4. It was found that, in general, those transforms invariant to circular time shifts of the data, namely the Fourier, cosine, BIFORE and the phase-shift-invariant Walsh transform, converged more quickly than the noncircular phase-shift-invariant transforms, such as the Walsh, Haar and R transforms.

4 CONCLUSIONS

Various orthogonal transforms have been compared insofar as they apply to surface texture analysis. The advantages in computation time of a number of the digital orthogonal transforms are offset by their phaseshift invariance, which dictates that a longer sample record is necessary in order to provide a convergence of the spectral estimates similar to that exhibited by the Fourier transform. The orthogonal transforms which produced a smaller number of spectral coefficients than the Fourier transform, namely the BIFORE and the Haar transforms, cannot be used to classify a number of data records whose frequency content differs only in relatively minor aspects. However, none of the spectral estimates of the data acquired from the machined specimens contained more than one machining peak, and hence the effect of the phase-shift invariance of the Walsh, Haar and R transforms was probably not as great as in some other engineering applications. Further investigations would be necessary to reveal whether there exists an underlying form in the spectral estimates of surface texture and, in particular, to determine the extent to which the results obtained are related to the frequency band analysed.

The performances of the Fourier and cosine transforms were generally similar, and in surface texture analysis it suffices, therefore, to calculate either one of these transforms rather than both. The performance of the Walsh transform was similar to that of the Fourier transform except that it exhibited a lower rate of convergence and also, if machining peaks were present, some additional odd-sequency harmonics were occasionally evident. In contrast to the results of Beauchamp (4), the circular phase-shift-invariant Walsh transform did not appear to be a useful alternative to the Walsh transform in an engineering sense, since the shape of the spectral estimates was different from that of the remainder of the spectral estimates. The Haar and BIFORE transforms were normally unable to detect the machining peaks, but nevertheless detected the underlying spectral shape found in the Fourier, cosine and Walsh estimates. However, a new windowing method (5) which permits spectral analysis at longer wavelengths than those within the band investigated in this paper requires that a greater number of spectral coefficients is contained in the spectral estimates than the $1 + \log N$ values present in the Haar and BIFORE spectra. At present, therefore, the analysis by the Haar and BIFORE transforms is limited to the roughness band investigated.

The orthogonal transforms which can be recommended for use in surface texture analysis are the Fourier and cosine transforms, in view of their rapid rates of convergence and also because of their ability to characterize the data and the machining peaks in particular. If, for example, it is known that the machining parameters employed will not produce machining peaks in the Fourier spectral estimates, then either the Walsh transform or the BIFORE transform, both of which have a shorter computation time, can be used, although the

C IMechE 1986



Fig. 4 Walsh spectral estimates obtained from shaped specimen number 17

Proc Instn Mech Engrs Vol 200 No C6

© IMechE 1986

413

A COMPARISON OF ORTHOGONAL TRANSFORMS IN SURFACE TEXTURE ANALYSIS

BIFORE transform would be limited to an analysis of the roughness data. In the authors' opinion, however, for an appropriate characterization of the data, the orthogonal functions need to be of a similar nature to the signal to which the corresponding transform is applied. Consequently, the Walsh, Haar and BIFORE transforms have few applications in the analysis of analogue signals and do not raise a significant challenge to the dominance of the Fourier transform in analogue signal identification. Orthogonal transform coding of digital signals is increasing in popularity, particularly for use in noise-contaminated digital communication channels, and this is the most likely field in which the digital orthogonal transforms will be found in the future.

REFERENCES

- 1 Cooley, J. W. and Tukey, J. W. An algorithm for the machine calculation of complex Fourier series. *Math. Comput.*, 1965, 19(90), 297
- 2 Mulvaney, D. J., Newland, D. E. and Gill, K. F. Identification of surface roughness. Proc. Instn Mech. Engrs, Part C, 1985, 199(C4) 281-286.
- 3 Wiede, B. W., Andres, L. T. and Iannone, A. M. Real time analysis of EEG using Walsh transforms. *Comput. Biol. Med.*, 197, 8, 225.
- 4 Beauchamp, K. G. Walsh functions and their applications, 1975 (Academic Press London).
- (Academic Press, London).
 5 Mulvaney, D. J., Newland, D. E. and Gill, K. F. Description of surface texture profiles. *Proc. Instn Mech. Engrs*, Part C, 1986, 200(C3), 167-178.
- 6 Harmuth, H. F. Transmission of information by orthogonal functions, 1972 (Springer-Verlag, Berlin).
- 7 Ahmed, N., Natarajan, T. and Rao, K. R. Discrete cosine transform. IEEE Trans. Comput., 1974, C-23, 90.
- 8 Hear, A. Zur Theorie der Orthogonalen Funktionensysteme. Math. Annln, 1910, 69, 331.
- 9 Walsh, J. H. A closed set of orthogonal functions. Am. J. Math., 1923, 45, 5.
- 10 Yuen, C. K. A fast algorithm for computing Walsh power spectrum. Application of Walsh functions, 1973, p. 284 (Washington DC).
- 11 Ulman, L. J. Computation of the Hadamard transform and the R-transform in ordered form. *IEEE Trans. Comput.*, 1970, C19(4), 359.
- 12 Ahmed, N. and Rao, K. R. The complex BIFORE transform. Elect. Lett., 1970, 6(8), 256.
- 13 Ohnsorg, F. R. Spectral modes of the Walsh-Hadamard transform. Applications of Walsh functions, 1971, pp. 55-59, IEEE Trans., EMC-13-14 (Washington DC).

APPENDIX

Definition of transforms

With the application of digital techniques in many branches of engineering, interest has developed in the investigation of orthogonal transforms which more closely relate to digital signals and whose transform values represent a series of square waves. The term 'frequency' is not generally applicable to these series of square waves since the function crossings of the time axis are not always regularly spaced. The term 'sequency', which was defined by Harmuth (6) as one half of the number of time axis crossings in the interval over which the functions are defined, is normally used to classify the square wave orthogonal function series. The unit of sequency is 'Zps', an abbreviation of 'zero crossings per second'.

© IMechE 1986

1

(a) The Fourier transform

The Fourier transform is the most popular of the orthogonal transforms and this is mainly due to the elegant properties of the frequency domain representation of a large class of engineering signals. The Fourier function series, FOR(p, n), is given by

FOR(0, n) = 1
FOR(p, n) =
$$sin{\pi(p + 1)n/N}$$
, $p = 1, 3, 5, ...$
FOR(p, n) = $cos(\pi pn/N)$, $p = 2, 4, 6, ...$

where n = 0, 1, ..., N - 1. The discrete Fourier transform of f(n) is given by

$$F_{f}(0) = \frac{1}{N} \sum_{n=0}^{N-1} f(n)$$

$$F_{f}(p) = \frac{1}{N} \sum_{n=0}^{N-1} f(n) \{\cos(2\pi pn/N) - j \sin(2\pi pn/N)\},$$

$$p = 1, 2, ..., N - 1$$

or by

$$F_{\rm f}(p) = \frac{1}{N} \sum_{n=0}^{N-1} f(n) \exp(-j \ 2\pi p n/N),$$

$$p = 0, 1, \dots, N-1$$

and the Fourier power spectral values $P_f(q)$, $q = 0, 1, \dots, N/2$, can be obtained from the transform coefficients as follows:

 $P_{\rm f}(q) = F_{\rm f}^2(q), \qquad q = 0 \ 1, \ \dots, \ N/2$

(b) The cosine transform

The cosine transform has been developed from the Fourier transform (7) and is defined as the real part of the Fourier transform. Since it is a sub-series of the Fourier series, the cosine function series must also form an orthogonal set of functions. The discrete cosine transform can be stated as

$$F_{c}(0) = \frac{\sqrt{2}}{N} \sum_{n=0}^{N-1} f(n)$$

$$F_{c}(p) = \frac{2}{N} \sum_{n=0}^{N-1} f(n) \cos\{(2n+1)p\pi/2N\},$$

$$p = 1, 2, ..., N-1$$

and the cosine power spectrum as

$$P_c(q) = F_c^2(q), \qquad q = 0, 1, ..., N/2$$

(c) The Haar transform

The Haar function series (8) was the first complete set of digital orthogonal functions to be described, each function taking no more than three values in the interval over which it is defined, except at a finite number of discontinuities. The Haar functions can be expressed as

$$HAR(0, n) = 1$$

$$HAR(p, n) = r, \qquad \frac{(p-q)N}{q} \le n \le \frac{(p-q)N}{q} + \frac{N}{2q}$$
$$= -r, \qquad \frac{(p-q)N}{q} + \frac{N}{2q} \le n \le \frac{(p-q)N}{q} + \frac{N}{q}$$
$$= 0, \qquad \text{otherwise}$$

Proc Instn Mech Engrs Vol 200 No C6

where p = 1, 2, ..., N-1, $q = 2^{(INT(\log_2 p))}$, $r = 2^{(INT(\log_2 p)/2)}$ and INT denotes the integer part of a numerical variable. The discrete Haar transform $F_h(p)$ is given by

$$F_{h}(p) = \frac{1}{N} \sum_{n=0}^{N-1} f(n) \text{HAR}(p, n), \qquad p = 0, 1, \dots, N-1$$

The Haar power spectrum

$$P_{h}(0) = F_{h}^{2}(0)$$

$$P_{h}(q) = \frac{1}{2^{(q-1)}} \sum_{s=2^{(q-1)}}^{2^{s}-1} F_{h}^{2}(q), \quad q = 1, 2, \dots, \log_{2} N$$

consists of only $1 + \log_2 N$ coefficients which are logarithmically spaced in sequency.

(d) The Walsh transform

The most popular of the square wave transforms is the Walsh transform. This is because computationally the transform is similar to the more familiar Fourier transform, which has enabled the Walsh transform algorithm to be substituted for that of the Fourier transform in a number of applications.

The Walsh functions were originally defined in 1923 (9) and take only two values, namely +1 and -1, except at a finite number of discontinuities. The Walsh functions are defined as follows:

WAL
$$(u_{r-1}, u_{r-2}, ..., u_0; v_{r-1}, v_{r-2}, ..., v_0)$$

= $\prod_{s=0}^{r-1} (-1)^{[u_{r-1}-s(v_s-v_{s-1})]}$

where u and v are the Walsh function arguments expressed in binary notation. The Walsh transform is given by

$$F_{w}(p) = \frac{1}{N} \sum_{n=0}^{N-1} f(n) \text{WAL}(p, n),$$

$$n = 0, 1, 2, \dots, N-1$$

and the Walsh power spectrum by

$$P_{w}(0) = F_{w}^{2}(0)$$

$$P_{w}(q) = F_{w}^{2}(2q - 1) + F_{w}^{2}(2q),$$

$$q = 1, 2, \dots, N/2 - N$$

$$P_{w}(N/2) = F_{w}^{2}(N-1)$$

Unlike the power spectra of the Fourier and cosine transforms, the Walsh and Haar power spectra are not invariant to circular time shifts of the data. This has led to the development of a number of phase-shift-invariant square wave transforms from, in particular, the Walsh transform. These are:

- 1. The orthogonal phase-shift-invariant Walsh transform, which involves deriving the Walsh transform from the autocorrelation function by means of a series of translation matrices (10), which is equivalent to summing and averaging the Walsh transforms of the N possible circular time-shifted versions of the data.
- 2. The non-orthogonal Ulman's R transform (11), which is formed by taking the absolute values of the terms obtained at each stage in the calculation of the fast Walsh transform.
- 3. The orthogonal BIFORE transform (12) which has proved the most popular of these methods.

(e) The BIFORE transform

-

1

The BIFORE (binary Fourier representation) power spectrum is obtained from the Walsh power spectrum as follows:

$$P_{b}(0) = P_{w}(0)$$

$$P_{b}(q) = \frac{N}{2^{(q+1)}} \sum_{n=1}^{N/2^{*}} P_{w}(2^{q}n - 2^{(q-1)}),$$

$$q = 1, 2, \dots, \log_{2} N$$

Hence, in common with the Haar transform, the BIFORE power spectrum contains only $1 + \log_2 N$ spectral values spaced logarithmically in sequency. In order to ensure that the BIFORE spectrum is circular phase-shift invariant, a property of the Walsh transform is utilized. This property is that if a circular time shift is performed on the data, then power is always conserved in restricted groups of Walsh power coefficients, each of which contains a coefficient of 'fundamental sequency' $2^{(p-1)}$, p = 1, 2, ..., $\log_2 N - 1$, together with $N \times 2^{-(p+1)} - 1$ odd-sequency harmonics. The BIFORE spectrum is a graphical representation of the sum of the power in each of these spectral groupings plotted as a function of their fundamental sequency.

Although a fast BIFORE transform is available which enables the BIFORE spectrum to be obtained directly from the data (13), to the authors' knowledge there has been no previous definition of a series of BIFORE functions. This series can be expressed as

BIF(0, n) = 1
BIF(p, n) =
$$\frac{N}{2q}$$
 INT[cos{2q\pi(n - p + 1)/N}],
p = 0, 1, ..., N - 1

where n = 0, 1, ..., N - 1, $q = 2^{\lfloor \log_2 N - \ln T(\lfloor \log_2 N - p \rfloor) - 1 \rfloor}$ and INT denotes the integer part of a variable. The BIFORE series can be shown to be both orthonormal and complete.

Proc Instn Mech Engrs Vol 200 No C6

PAPER 60

Knowledge representation database for the development of a fixture design expert system

A R Darvishi, BSc, MSc, PhD and K F Gill, BSc, MSc, PhD, CEng, MIMechE Department of Mechanical Engineering, University of Leeds

To meet the demands of the flexible manufacturing system, an exploratory approach to the design of fixtures using an expert system is presented. A small part of the range of knowledge existing is utilized to create a database framework that can be used to develop a computerized approach to fixture design.

The method allows for an expansion to a relatively large database that, when established, would encompass all of the important practical aspects of fixture design.

1 INTRODUCTION

Modern manufacturing practices demand medium to small batch production runs and this requirement has established the need for a computer integrated manufacturing (CIM) system. A variant of this is often referred to as a flexible manufacturing system (FMS), a concept that cannot be successfully implemented until significant hardware and software development is undertaken.

A major factor limiting the capability of the FMS is the component fixturing requirement. In the hierarchy of computer aided design and manufacture (CADCAM) a fixture design system is the interface between product design (CAD) and process planning on the one hand and part programming and manufacture (CAM) on the other.

Fixture design is a complex task with many variants that must be satisfied simultaneously to achieve the optimum result.

The results of this work reveal that the principal problem is one of knowledge representation and the authors believe that a technique based on artificial intelligence (AI), particularly a rule-based expert system, offers a most promising solution. An attempt is made to adopt and utilize established expertise in the evolution of a logical automated approach to fixture design for a part family or a part in hand.

By combining a knowledge of manufacturing methods and machine information with a specialpurpose computer language, developed primarily for list processing and symbolic manipulation, ground rules for a fixture design approach are suggested in this paper. The ideas presented are illustrated by application to a number of simple examples.

2 WORKHOLDING

Whatever the production method, sophistication of the manufacturing process, a solution to how best to grip workpieces in the manufacturing phase must be found at the planning stage. For most components, the question of 'how to grip' must be answered many times as they progress through the machining, inspection and

The MS was received on 18 February 1987 and was accepted for publication on 5 November 1987.

assembly stages, indicating clearly that workholding is of major importance to the metal-working industry.

Unfortunately, industry considers workholding only of secondary importance and as a result it is left to the production engineer, process planner, or tool designer to improvise as and when necessary. In answering the simplest question on workholding, there is a variety of approaches that will ultimately lead to a range of costs for the machining operation, whether for a single prototype or a special fixture for a long production run. In the first case, a few minutes may not be significant, in the latter, a few seconds lost may be economically undesirable.

The general trends that affect workholding and its economic effectiveness can be broken down into three broad categories:

- (a) changes in component design, method of manufacture and plant operation,
- (b) developments in machine tool design and application, and
- (c) evolution of tooling components and practices.

Although the infinite variety of workpieces machined makes it impossible to generalize, some trends can nevertheless be perceived. The market life of products tend now to be shorter, which means that tooling must be amortized over shorter periods. Proliferation of models and variants of these emphasize the need for modular design and, now, attention to the *family of parts* (group technology) concept. These all contribute to the demand for greater versatility in fixture characteristics (1).

Unfortunately design references (2, 3) do not give a clear indication of principles involved or specify a logical design approach. The philosophy is still one of trial and observation based on established company practice.

2.1 Fixturing for NC machining centres

A machining centre with either horizontal or vertical spindle can replace many separate machine tools and operational practice with these machines has directly affected fixture design. For optimum fixture design, information flow as depicted in Fig. 1 must exist and be integrated into the philosophy adopted by a manufac-

20/88 C IMechE 1988

0263-7146/88 \$2.00 + .05

Proc Instn Mech Engrs Vol 202 No B1

A R DARVISHI AND K F GILL



Fig. 1 Information flow for optimum tool and fixture design

turing team of product designer, process planner, tool designer and machine programmer.

Gouldson (4) suggests for numerically controlled (NC) machine fixture design:

- (a) from a knowledge of the tool motion axis (usually the z axis) and employing the shortest tool consistent with the machining operations to be performed, the best position of the fixture in one axis should be established;
- (b) the fixture allows access to difficult regions in the component;
- (c) permits the (majority) component reference datums and measurement directions to be orthogonal with the axis of NC machine;
- (d) gives access to the maximum number of surfaces for the least number of changes in set-up.

Smith (5) developed a clamping method which he maintains largely eliminates the distortion problem that generally occurs with castings and forgings by providing 'floating ball' supports at points of contact.

A ball is incorporated in each fixture element at every workpiece contact point and these are free to move within their sockets to adapt automatically to the irregular workpiece under-surface. This adaptation eliminates the need for preliminary machining at clamping points.

Tuffentsmmer (6), developed a numerically controlled clamping machine, in which position determination, supporting and clamping mechanisms are designed separately, to make the independent movement of each system mechanism possible. In small batch production (or FMS) this approach is economically unsound because of component variations and the frequent change in machining sequence required. The NC clamping system is technologically sophisticated and requires significant capital investment.

Reduced product life cycles and batch size make it essential that demanded variation to machining sequence, tooling and changes in composite fixture layout can readily be accommodated. Lewis (7) suggests the use of standard fixture elements. The system is

Proc Instn Mech Engrs Vol 202 No B1

based on T-slot locations arranged in three groups for the clamping of small, medium and large components respectively.

Graham, Woodwark and Neads (8) have explored the use of flexible fixturing kits. They argue that the advent of FMS and adoption of palletized workpieces have increased the demand for more flexibility to be introduced into fixtures design and fabrication, and that machining fixtures are not economically sound for use with FMS.

A solution would be the introduction of fixturing kits, the modular elements of which would allow a range of composite fixtures to be assembled.

3 USE OF AI AND AI LANGUAGES IN FIXTURE DESIGN

Markus et al. (9) have proposed a prototype fixture design approach, for a family of box-type workpieces from a fixture kit, using the artificial intelligence language for logic programming named 'Prolog'. The input data describe the shape of the workpiece, the machining required and the coordinates of the supports and clamps to be realized. The computer produces a sequence of draft fixture arrangements and the engineer has the overall authority to guide the computer decision-making towards a specific fixture style. Unfortunately, the paper leaves many important questions on workpiece shape, attributes and machining unanswered.

Ingrand and Latombe (10) discussed an expert system for automatic fixture design, which defines fixture elements by their operational function and class, an approach adopted in this work.

The system applies expert rules to select the resting surfaces and resting points that will eliminate six degrees of freedom from the component, taking into account the attributes of quality of surfaces, dimensions and geometrical relations of the surfaces. Generic elements are substituted for the resting and clamping points and finally, standard elements are chosen that inherit the constraints of the generic elements.

The system is implemented in Maclisp, a dilect of LISP (11) and includes only fixing and positioning functions of the fixture.

3.1 Production systems

Progress towards solving the artificial intelligence related problems of how to represent and use knowledge in computer programs has been made. Several different models of knowledge representation and use have been developed. The popular terms used for these language types are production systems, with OPS5 and EXPERT (12) as specific languages.

The production system paradigm has been used successfully to solve a wide variety of problems such as medical diagnosis (13) and the automatic configuring of computers (14).

Many of the application systems created with the production system paradigm are of a class known as expert systems or knowledge-based systems. The term 'expert system' is used to refer to a computer program that is able to perform within a specific and limited task domain at the level of a human expert. An expert system has a large component of domain-specific

© IMechE 1988



Fig. 2 Architecture of a production system model

knowledge embedded within it. When that knowledge is represented in an identifiable, separate part of the system rather than being dispersed throughout it, the implementation is referred to as a knowledge-based system.

The basic architecture of a computational production-system model is shown in Fig. 2 and includes three major components:

- (a) a data store, called *data memory* or working memory which serves as a global database of symbols representing facts and assertions about the problem;
- (b) a set of rules constitutes the program: each rule has a condition part, usually indicated by the keyword 'IF' and an action part, indicated by the keyword 'THEN', which gives instructions for changing the data configuration;
- (c) an inference engine is needed to execute the rules.

The preference for writing rule-based expert systems is that the human expert finds it intuitively appealing to express domain knowledge in terms of condition-action pairs. This work has been implemented on a VAX 11/780 computer using the OPS5 production system language. The language OPS5 was developed for application in the areas of AI and expert systems (12) and its architecture includes the three major components of a production system.

For the reader not familiar with OPS5, reference (12) should be consulted.

4 FIXTURE DESIGN OBJECTIVE

The main objective in any tool design is to reduce the overall manufacturing costs, maintain quality and increase productivity within a framework of maximum flexibility. One approach to accomplish this is:

- (a) design simple, easy to operate tools;
- (b) adhere to basic principles;
- (c) avoid unnecessary complexity in design;

© IMechE 1988

- (d) design to avoid improper usage (include foolproofing features);
- (e) select tool materials which are known to give adequate life and incorporate safety features at initial design stage.

It is extremely difficult to find a precise and generally accepted definition for design. Dieter (15) adopts the definition: 'design establishes and defines solutions to and pertinent structures for problems not solved before, or new solutions to problems which have previously been solved in a different way'.

The ability to design is both a science and an art. The science can be taught, the art is acquired through trial and observation and is more commonly referred to as expertise.

The traditional industrial fixture design approach is wasteful of resources and prolongs lead time, because of the tenuous links that exist between design, development and production. A strengthening of these links will inevitably:

- (a) lead to simplification;
- (b) facilitate ease of fixturing (including locating, supporting and clamping);
- (c) make use of less expensive and easier to operate tools;
- (d) reduce handling between machines;
- (e) require less complex assembly procedures.

4.1 Fixture selection

The important criteria in the choice of an engineering fixture, as with any industrial product, are quality, quantity and cost of the product to be manufactured. Unfortunately these requirements can be forgotten when personal preference, excessive inventiveness and engineering dogma are not restrained. The use of a knowledge-based computer design scheme will help to introduce the essential philosophy to ensure that the optimum design is achieved.

Proc Instn Mech Engrs Vol 202 No B1

A R DARVISHI AND K F GILL



Fig. 3 A free body in space and its six degrees of freedom (12 directions of movement)

In a production cycle, a fixture is required to fulfil a variety of functions, typically, locating, supporting, centralizing and clamping in addition to facilitating the operational requirements of loading and machining.

Locating, centralizing and clamping assumes a special role in relation to the accuracy of the workpiece produced. To guarantee the exact relationship between the cutting tool and the workpiece, the relationship linking machine, machine sub-assemblies, fixture and workpiece must be unambiguously defined.

5 LOCATING AND SUPPORTING RULES

Locating refers to the establishment of a desired relationship between the workpiece and the fixture, and this in turn establishes a relationship between the workpiece and the cutting tool. To guarantee the desired accuracy the workpiece must be precisely located and rigidly supported.

A free body in space (Fig. 3) has a maximum of six degrees of freedom (6DOF), a linear and rotational movement for each of the three axes x, y and z.

The most widely used method to ensure full location of a component is the three-two-one or six points principle. The interpretation of this principle is three locating points are required for the first plane, two for the second and one for the third, assuming the planes are not parallel and are preferably at 90° to each other (see Fig. 4). This method when correctly applied will reduce the number of DOF to zero for all rigid workpieces without inclusion of redundant location features. For non-rigid workpieces extra support will always be needed.

The number of faces or type of feature to be machined dictates the number of dimensions to be controlled and hence the type of location required.

Locating alternatives have been illustrated in Fig. 5. To machine one face the dimension 'a' (see illustration) is adequate and only one location plane is necessary, that is the exact location of the component is not

Proc Instn Mech Engrs Voi 202 No B!



Fig. 4 Six points location (3-2-1)

important. The machining of a simple slot requires dimensions 'a' and 'b' and two locating planes. All three dimensions 'a', 'b' and 'c' are required to mill a blind slot (three planes) and the component will now have zero DOF.

5.1 Supporting and clamping elements

These can be subdivided into three classification types of solid, adjustable and equalizing supports. The type of support employed must accommodate the surface shape and condition, and be able to withstand the clamping and cutting forces applied.

The solid support is machined (or cast) or rigidly mounted in the fixture base and is not compatible with the modular design concept. The adjustable support is most suitable with uneven surfaces, (that is cast parts) and allows the workpiece to be supported at the required orientation for machining. Equalizing supports function as connected units: if one point is depressed a second will rise to maintain contact with the workpiece and are appropriate for use with uneven surfaces.

The clamping method adopted, whether simple or complex, should be the result of operational analysis and not prejudice. In selecting clamping elements for a given part, its shape and size must be carefully considered, for some clamps are of such design that they are more suitable for a particular size of workpiece. Clamping elements must be looked at with due regard to ease of operation and simplicity in design. The level of clamping force required may restrict certain design features on the workpiece as well as choice of clamping element. Surface finish or condition of the surface on which the clamps are to be positioned, whether finished or rough, limits the selection, because rough work surfaces require greater clamp travel in the clamping range.

6 REPRESENTATION

In the design of a new fixture, the tool or fixture designer relies heavily on past experience, an understanding of practices, manufacturing methods and machine information, on which the fixture is to be used. To use this knowledge most effectively within the timescale normally available to him, the design engineer does require computer assistance.

@ IMechE 1988

41

DATABASE FOR THE DEVELOPMENT OF A FIXTURE DESIGN EXPERT SYSTEM



Fig. 5 Degrees of freedom and number of locating planes for differing production requirements (14)

To encourage the engineering use of such a facility does demand that the following criteria be met:

- (a) each component feature and characteristic fixture element should be known by an appropriate identifier;
- (b) the designer must be able to represent uniquely the physical, geometrical and technological attributes of a component and
- (c) the system must be sufficiently flexible to allow a part with many attributes to be precisely represented.

The approach adopted by these authors to the development of a fixture design expert system follows and outlines a representation of the essential facilities required in the manufacturing environment.

6.1 Overall representation of the component

This describes in detail the component to be machined and is central to a fixture design. It contains the overall physical, geometrical and technological information on the workpiece. The data structure that contains this information is called PART. 'LITERALIZE', an OPS5 statement, is used to declare the attributes associated with PART and these are:

(a) Addressing attributes IS-A

Part-name

Type of object For identification of component

© IMechE 1988

Classification code

(b) Physical attributes Type-of-material

> Magnetization-status Overall-weight Number-of-throughholes Number-of-throughpockets Number-of-slots

(c) Geometric attributes Overall-shape Overall-size Relative-size

Dimensional-variation

Code generated by a classification method used in group technology (GT)

Engineering material or suppliers code Magnetic properties if any Integer value in any unit Integer value or NIL

Integer value or NIL

Integer value (all types of slot) or NIL

Geometric form Major dimensions Subjective size description or code from group technology [helpful in selection of machine size or fixture and other related equipment; Table 1 from reference (16) shows size classes, dimensions and gives examples]

Maximum range of dimensional variations Surfaces enveloping the component

Proc Instn Mech Engrs Vol 202 No BI

•

Surface-list

 Table 1
 Part family size classification

C:	dimension			
cod	e in	mm	Description	Examples
1	0.5	10	Sub-miniature	Capsules
2	2	50	Miniature	Paper clip box
3	4	100	Small	Large match box
4	10	250	Medium-small	Shoe box
5	20	500	Medium	Bread box
6	40	1 000	Medium-large	Washing machine
7	100	2 500	Large	Pickup truck
8	400	10 000	Extra-large	Moving van
9	1 000	25 000	Giant	Railroad box-car
(d)	Feature-list Technologica	al attribute	Features ex compone and thos machine	kisting on ent (machined) e to be d.
` ´	Is-the-part-a-family-		Boolean or	perator (ves or
	member		no): influencing the level of capital that should be invested for design and construction of fixture	
	Initial-state		State of con CAST, A	mponent (that is, S FORGED etc)
	Batch-size		A positive of parts	integer (number in a batch)
	Batch-repeat	ts-in-12	Number of	yearly batch
	months		repeats	
	Part-life-cycl	le	Life of pro	duct (years),
	Prior-machining- operation		Machining operation(s) completed.	
	Initial-surfac	e-quality	Rough, fini	sh etc
	Heat-treatme	ent	Work com	pleted
	Part-function	n-in-the-	Location a	nd purpose (for
	assembly		example	base plate)
	Spindle-dire	ction	Rectangula	r Cartesian
	Spinale-diffe		coordina hand)	ite system (left
	Required-cut direction	tting-	Cartesian o	coordinates
	Part-details		Simple deta Table 2)	ails,, (s ce

All of these attributes and their respective values must be known to the engineer at the initial stage of the fixture design. Careful study of the component drawing(s) will reveal the pertinent information and the process plan. Figure 6 shows a data structure developed in terms of OPS5 (using the system command LITER-ALIZE) for an overall representation of a component named 'PART'.

Having declared the required data structure for representing a 'PART', instances of PART can easily be created.

6.2 Machine tool representation

The fixture designer must have available to him the 'machine reference sheet', or equivalent.

The data structure representing a machine tool here, primarily aims at a machining centre with three-axis control (NC or CNC) or a typical horizontal (vertical) milling machine.

Proc Instn Mech Engrs Vol 202 No Bl

 Table 2
 The most common details seen on engineering components and their classification

Part details	Examples
Simple detail	Slots, holes, plane surfaces and pockets
Evenly spaced details	Gears, splines, squares, hexagons, polygons, etc.
Complex details	Other than above

The attributes employed to represent a machine tool are:

IS-A	Describes class, a production tool in this case
Classification	Machine code or company code to identify a particular machine
Generic-type	Machine tool to be employed, for example three-axis machining centre
Main-axis-direction	z direction for a vertical spindle machining centre
Max-main-axis-thrust	Force due to cutting action on component (fixture must absorb this together with clamping forces, typical value 15 600 N)
Milling-capacity	Maximum value, for example 180 cm ³ /min
Drilling-capacity	Maximum value, for example 50 mm diameter hole
Tapping-capacity	Maximum value for example M30
Spindle-approach-relative- to-table	The spindle stationary and table moves along the 'z' axis, or spindle moves vertically and approaches the table
Max-spindle-distance- from-table	For vertical spindle machine distance is normally measured from the tip of the spindle to the machine table; for horizontal axis machines the distance is measured from the centre-line of the spindle to the table, typical value 600 mm

Separate data structures have been developed to represent machine tool sub-assemblies. Typical of these are machine table, spindle, tool magazine, machinery operations and suggested fixture. To illustrate these, the machining operation and the suggested fixture will be presented.

6.2.1 Machine table representation

The machine table is mounted on the machine bed and carries the fixtures, workpieces, indexing unit and tailstock. Table movement in the horizontal plane is labelled as the x-y directions and the vertical movement as the z direction.

•

C IMechE 1988

DATABASE FOR THE DEVELOPMENT OF A FIXTURE DESIGN EXPERT SYSTEM

.

.

DATABASE FOR THE D	EVELOTMENT OF A FIATORE DESIGN EXPERT STSTEM
(LITERALIZE PART	; ELEMENT CLASS REPRESENTING A COMPONENT OR A PART
PART NAME	; A UNIQUE NAME FOR EACH PART
CLASSIFICATION CODE	A UNIQUE CODE GENERATED BY ONE OF THE CLASSIFICATION METHODS COMMONLY USED IN "GROUP TECHNOLOGY"
IS-THE-PART-A-FAMILY-MEMBER	; YES OR NO THIS IS IMPORTANT IN CHOOSING KIND OF FIXTURE TO BE USED
PART-LIFE-CYCLE	; AS FOR THE ABOVE PURPOSE
OVERALL-SIZE	: X-DIM, Y-DIM & Z-DIM
RELATIVE-SIZE	; SMALL, MEDIUM OR LARGE
OVERALL-SHAPE	; PRISMATIC OR CYLINDRICAL
SPECIFIC-WEIGHT	; GRAMS PER CUBIC CENTIMETRE
OVERALL-WEIGHT	; A positive integer the unit of which could be in "Ib" OR "kg"
TYPE-OF-MATERIAL	; AN ENGINEERING MATERIAL NAME LIKE "GREY-CAST-IRON" OR A CLASSIFICATION CODE "ASTM" "AIS!", ETC
MATERIAL-MAGNETISATION-STATUS	; MAGNETISABLE OR NONMAGNETISABLE
INITIAL-STATE	; FOR INSTANCE "AS-CAST" OR "FORGED" ETC
DIMENSIONAL-VARIATION	; THE RANGE OF DIMENSIONAL VARIATION THAT CAN BE EXPECTED
INITIAL-LENGTH	; DIMENSIONS OF THE BOX
INITIAL-WIDTH	; ENVELOPING THE INITIAL PART
INITIAL-HEIGHT	·
INITIAL-ENCLOSING-BOX-VOLUME	
FINAL-LENGTH	; FINAL DIMENSIONS OF THE COMPONENT
FINAL-WIDTH	: THESE DIMENSIONS WILL BE CHECKED AGAINST THE INITIAL
FINAL-HEIGHT	; DIMENSIONS TO MAKE SURE THAT THE BOX ENVELOPING THE INITIAL COMPONENT IS NOT SMALLER THAN THE BOX ENVELOPING THE FINAL COMPONENT AND ALSO TO MAKE SURE THAT THE INITIAL BOX IS NOT GREATER THAN THE BOX THE MACHINE TOOL CAN ACCOMMODATE
FINAL-ENCLOSING-BOX-VOLUME	
INITIAL-SURFACE-QUALITY	; ROUGH FINISH, ETC
BRINEL-HARDNESS	; FOR SELECTION OF MACHINING DATA
BATCH-SIZE	; A POSITIVE INTEGER USUALLY THE BATCH SIZE, 200, 500 ETC
BATCH-REPS-IN-12 MONTHS	; NO OF TIMES THAT A BATCH MUST BE REPEATED IN A YEAR
SPINDLE-DIRECTION	; USUALLY IN "2" DIRECTION
PRIOR-MACHINING-OPERATION	; NIL FOR "NONE" OR FOR INSTANCE "SLAB-MILLING" AND SO ON
HEAT TREATMENT	; NONE OR KIND OF HEAT TREATMENT DONE
0	
0	
0)	

Fig. 6 Data structure developed in terms of OPS5 for component representation

The attributes of the data structure corresponding to a machine tool are:		Max-longitudinal-traverse	x direction, for example 1000 mm,
IS-A	Object represented, for example machine element	Max-cross-traverse	y direction, for example 500 mm,
Machine-classification	Classification code of the machine to which the table	Max-vertical-traverse	z direction, for example 600 mm,
	belongs	Longitudinal-traverse-	for example 6–90 mm/min
Table-size	Area of machine table	feed-range	
Max-cutting-area	Most efficient machining area	Cross-traverse-feed-	for example 6–90 mm/min
No-of-T-slots	Number depends on table size,	range	
	usually three or four slots	Vertical-traverse-feed-	for example 3-45 mm/min
Distance-between-T-slots	Solid table portion between	range	
	two adjacent T-slots	Table-position-accuracy	for example 0.005–0.013 mm
T-slot-throat-width	These attributes are the most	Table-axes-drive	for example d.c. motor, range
T-slot-throat-depth	important dimension of the		4–12 kW
T-slot-headspace-width	T-slot, and dictate the bolt	Table-swivel-right	for example 45° for horizontal
T-slot-headspace-depth	selection that can be used to		table
	hold the fixture	Table-swivel-left	for example 45°, swivel is
Max-table-load	Component plus fixture, pallet		about 'z' axis
	or indexing unit	Swivel-about-y-axis-right	This feature is not common or
	(overloading table will lead	Swivel-about-y-axis-left	most machines
	to poor repeatability and machine bed may be deformed)	Direction-of-external- normal	For a vertical axis machining centre this direction is opposite to spindle direction

© IMechE 1988

,

4

,

2

-

Proc Instn Mech Engrs Vol 202 No BI

(LITEMALIZE SUGGESTED-FIX	IUNE
CLASSIFICATION	
TYPE	; GIVEN FIXTURE NAME
PARTNAME	; PART-NAME FOR WHICH THE FIXTURE IS USED
PRODUCTION-SPEED	; THE DESIRED PRODUCTION SPEED EX: NORMAL, HIGH, ETC.
NO-OF-FIXTURING-STATIONS	; EX: 1 OR 2 FOR SINGLE OR DOUBLE FIXTURING STATIONS
PART-LOADING-METHOD	; EITHER MANUAL OR AUTOMATIC
USED-TO-MAKE	; GIVES THE KIND OF PART THAT THE FIXTURE IS SUPPOSED TO HOLD
PRODUCTION CYCLE	;
PART-POSITION	; GIVES THE PART POSITION WITH REGARD TO THE SUPPORTS OR LOCATORS
TOOL-POSITION-STATUS	; WHETHER THE TOOL IS STATIONARY OR NOT. FOR INSTANCE WHEN AN EVENLY-SPACED-DETAILS IS PRODUCED THE TOOL IS NORMALLY STATIONARY
PART-SIZE	; THE RELATIVE PART SIZE FOR WHICH WE WANT A FIXTURE
JAW-TYPE)	; TYPE OF JAW REQUIRED IF ANY.

Fig. 7 The data structure for representing a requested/suggested fixture

6.3 Requested and suggested fixtures

The designed fixture must fulfil its principal functions of supporting, locating and clamping; and ensures that unclamping and unloading are physically realizable. To account for these, production speed, number of fixturing stations, loading and unloading methods and component details must be established. The data structure holding the relevant knowledge for a fixture is shown in Fig. 7. For a fixture design to be optimum, these attributes must match those of the requested fixture.

CON UTE ON OCEATED EN

All engineering components both simple and complex are bounded by surfaces. A geometric surface is the area generated by the motion of a geometric line, either straight or curved. Reference (17) classifies the surface into two major categories, namely planar and curved.

Most machined components are bounded by either plane or single-curved surfaces; cubes, prisms and pyramids are bounded by plane surfaces while singlecurved surfaces (generated by a moving straight line), bound cylinders and cones. This work is concerned with these two categories of surface. When the term workpiece, component or part is used, a component bounded by planar or single-curved surfaces is defined.

6.4 Surface representation

A data structure has been developed for representation of surfaces. It includes surface identification attributes, size (dimensions) attributes and their respective tolerances, form tolerances and surface quality parameters. Surface representation attributes, their interpretation and constraints are described below:

I	S-A

Part-name

States what the element class is, for example a feature Name referring to the part on which the surface resides Specifying whether the feature (surface) is external or internal

Proc Instn Mech Engrs Vol 202 No B1

Features-generic-type

Feature-type

Surface-label Surface-type Surface-length Surface-length-tol Surface-width Surface-width-tol Dimensional-tol

Surface-position

Status

Heat-treatment Surface-quality

- - -

Direction-of-externalnormal

Required-cuttingdirection

Is-parallel-to-surfaces

Is-parallel-to-features

Parellelism

- -----

Is-perpendicular-tosurfaces

Perpendicularity

Features-starting-fromthis-surface Features-opening-tothis surface Name specifying the type of common feature

Identification and reference for example, planar

- This and the following three attributes represent major dimensions of the surface and their respective tolerances
- If no specific tolerance has been given for individual dimensions, these attributes would represent the overall tolerance of the surfaces
- For example, distance from a parallel surface
- Matching status of the surface, that is machined
- Type of heat treatment done or NIL
- Initial quality of the surface, for example rough
- Direction of external-normal to the specific surface, for example Z
- Can be specified for features such as holes, counterbores and countersinks; for other features (for example slots) is the direction of first cut
- List of surfaces to which the surface to be represented is parallel
- List of features to which the surface to be represented is parallel
- A form of tolerance, for example 0.01 mm
- List of surfaces to which the surface in question is perpendicular, for example S2, S3, S4, S5
- A form of tolerance, for example 0.01 mm
- For example H1 (denoting hole-1)

For example slot-3

© IMechE 1988

_

DATABASE FOR THE DEVELOPMENT OF A FIXTURE DESIGN EXPERT SYSTEM



(LITERALIZE T-SLOT	
FEATURE-TYPE	; e.g. SLOT
SLOT-TYPE	; e.g. T-SLOT
PART-NAME	: e.g. P1
STATUS	; e.g. MACHINED
T-SLOT-LABEL	; e.g. T-SLOT1
MAJOR-AXIS-DIRECTION	; e.g. X
STARTING-FROM-FACE	; e.g. S3
OPENING-TO-FACE	; e.g. 55
T-SLOT-LENGTH	; e.g. F
T-SLOT-THROAT-WIDTH	; e.g. A
T-SLOT-THROAT-DEPTH	; e.q, E
T-SLOT-HEADSPACE-WIDTH	; e.q. B
T-SLOT-HEADSPACE-DEPTH	; e.g. D
T-SLOT-OVERALL-DEPTH	; e.g. C
T-SLOT-DIMENSIONAL-TOL)	: e.a. 0.01 mm

Fig. 8 Attributes for a T-slot representation

The interpretation of the required-cutting-direction attribute, is true for the last two attributes. For holes, the surface from which drilling starts, is the surface from which the feature (hole) has started, and in the case of a through hole, the surface into which the drill tip breaks, is the surface into which the feature (hole) opens. For slots the surface from which the first cut starts is the surface from which the feature starts by convention and the surface to which cutting action ends, is the surface to which the feature opens.

6.5 T-slot representation

T-slots are available on most machine tool tables for clamping purposes. A T-slot can be decomposed into two separate features, a slot and a groove.

The important attributes of a T-slot are throat width and depth, headspace width and depth and their respective tolerances. These attributes together with the identifying attributes are included in the data structure shown in Fig. 8.

6.6 Starting and opening features

One or more features may start from or open into a surface. Holes, countersinks and counterbores can be defined without ambiguity; however, grooves, slots and notches are less precise and for this reason the surface from which the first cut starts is defined as the surface from which the feature starts. The surface to which the first cutting ends is defined as opening surfaces. The attributes representation (opening feature) are:

Surface-label	Surface from which the feature	
	starts or opens into	
Part-name	Part on which the feature exists	
Feature-type	Starting or opening feature, for	
••	example T-slot	

C IMechE 1988

Feature-label Feature in qu T-slot	estion, for example
Status For example to-be-macl	machined or nined
Major-axis-direction Cutting direct counterbor or direction	tion (for holes, res and countersinks n of first cut
Major-axis-is-parallel-to A surface or	other feature
Location Relative loca for exampl surface fro or open ini	tion of the feature, e below or above m which they start to
Feature-width/surface- width For example attribute in standard ei when supp heights are a hole or n componen	0.25, an important a selection of lements (that is orts with adjustable e needed to contact eccess in a t)
Hole-dia-surface-width For example	0.2

6.7 Reference planes

The reference planes are faces on the component used for locating it on the fixture and it is required that all six degrees of freedom are specified. The most widely used method for locating a component is the three-twoone or six points location strategy. That is, to position a component at least three locating points are required in the first plane, two in the second and one in the third. Usually, stability is satisfactory if the three base buttons are widely spaced and the resultant cutting force hits the base plate well within the triangular area between the buttons. If it hits outside of this area, then it generates a moment which tends to tilt or overturn the part.

No redundant support should be incorporated in a fixture, but this is violated on occasions to improved stability, specially on the first reference plane. By the addition of a fourth locator in the base, the shape of the supporting area can be changed from a triangle to a rectangle and this provides the extra stability. This can be described the '4-2-1 locating principle'. For rough castings, one of the four base locators may require to be adjustable.

Redundant or intermediate supports are only employed when the part does not have sufficient rigidity to withstand the operating forces without distortion (18).

The three reference planes must have functional attributes and separate data structures have been developed for each. The reference plane data structure attributes are:

Part-name	Part on which the reference plane is selected
Surface-type	Usually planar
Surface-label	Label of the selected reference plane
Status	For example machined or to-be-machined
Surface-quality	For example rough-machined
Direction-of-external-normal	For example z
Is-parallel-to-surfaces	Surface to which the reference plane is parallel
Parallelism	Form tolerance, for example 0.01 mm
Is-perpendicular-to-surfaces	Surfaces to which reference plane is perpendicular

Proc Instn Mech Engrs Vol 202 No B1

A R DARVISHI AND K F GILL

Perpendicularity	Form tolerance, for example 0.01 mm	Stem-length	Must be lo positive
Features-starting-from-FRP	Features starting from first reference plane	Ease-of-use-factor	in fixture Ease with
Features-opening-to-FRP	Features opening into the first reference plane		locator o some ele
			manipul

6.8 Representation of generic and standard elements

A generic element is an assumed conceptualized fixture element that can fulfil in part or completely one of the functions of a required fixture.

Constraints may be imposed on each or all generic elements associated with a particular reference plane. For instance, if three generic elements are to support a planar surface, they must not be co-linear.

Standard elements are real world fixture elements that are capable of fulfilling all required functions. Expert rules map all properties of generic elements into standard element, that is qualitative, quantitative and constraints.

Each data structure developed to represent standard fixture elements includes qualitative attributes describing functional properties, quantitative attributes listing the important dimensions, total number of each element stocked, number of elements in use and those available for use.

Since generic elements will be mapped into standard elements, both are represented by similar attributes and the two separate data structures include the following attributes:

IS-A	Points to an object, for example generic element
Part-name	For example, component
Gen-el-code	Identifies generic element's code
Associated-with	Reference plane code (or
	acronym) for generic or standard elements
AKO	Abbreviation of A-KIND-OF,
Turne	Installing nature of the support
Туре	instaning nature of the support
	or locator for example
	inx-machined-support
	(permanent support machined
a :	into the fixture body)
Generic-name	Group to which the element
	belongs, for example solid
	supports
Function	For example supporting
Principle-of-action	Intended function, for example fixed-in-position
Nature of contact	Nature of contact that will be established between support and surface of the component, for exempt a point to plane
Classification-code	Standard elements only
Man die	This is required if the locator is
wax-ua	to locate the component from a specific hole
Max-dia-tol	For example 0.01 mm
Max-height	Measured from the surface of
U	base plate and includes the
	top portion of the support
Stem-dia	Support stem has to correspond
	to diameter of hole in fixture body
Stem-dia-tol	For example 0.01 mm

Proc Instn Mech Engrs Vol 202 No BI

Stem-length	Must be long enough to ensure positive support installation in fixture body
Ease-of-use-factor	Ease with which a support or a locator can be positioned; some elements need less manipulative movement and effort
Economic-factor	Relative cost of a particular element that can fulfil the required function
Type-of-height	Adjustable or fixed
Adjustment-range	Depending upon component's dimensional variation, it may be necessary to choose an adjustable height locator or support, supports must accommodate the variation expected
Used-to-support-from	External feature, for example a planar surface
Used-to-locate-from	Locate from an external profile or internal feature, for example hole
Friction-and-jamming- factor	A relative factor, for example low: the greater the area of contact between the component and support the more likely the component will iam
Load-unload-factor	A relative factor: a locator that engages in an internal profile (for example a hole) resists loading and unloading more than a relieved pin
No-of-elements-required	Both standard and generic elements
Total-no-of-elements- stocked	Standard element stocked
Total-no-of elements-	Available for use

6.9 Clamping element representation

A clamp rigidly holds, without damage and distortion, a component during the machining cycle. Two major groups exist, the mechanical and non-mechanical devices. The selection made from either group is dictated by the shape, size and material properties to be held. The most common types of mechanical clamp are strap, screw, swing, cam, wedge and chuck.

The non-mechanical clamp would be magnetic (electromagnetic or permanent magnetic) or nonmagnetic (vacuum) and the two data structures developed for these are as follows:

States what the element to be represented is, for example CLAMP
Class type, for example strap clamp
Component name
Identifying the main group to which the element belongs, for example mechanical-clamp
Primary function of element
Material characteristic (ferrous or non-ferrous)

C IMechE 1988

Clamp-length

Clamp-width Clamp-slot-length Clamp-slot-width

Clamp-thickness Clamp-weight Suitable-for-part-size

Nature-of-contact Clamping-surfacequality

Clamping-force-range

Clamp-complexityfactor

Operating-method Operating-speed

Ease-of-use-factor

Economical-factor

Friction-and-jammingfactor Load-unload-factor

No-of-elementsrequired

ţ

Overall length, for example 110

mm For example 32 mm For example 30 mm Largest diameter of the bolt that

can be used Measured at rear end

For example 0.5 kg

Some clamps are so designed to be used for certain size of work, for example small, medium or large

For example point-to-plane The quality of surface on which the clamps will be set to some

extent limits the clamp selection; normally rough surfaces require greater clamp travel in working range

- The level of clamping force required in a particular situation (depending on the component size, feeds, speeds, type of material) may eliminate certain design and makes of clamps; the level of vibration involved may also dictate selection of a clamp with higher clamping force range
- A relative factor pertaining to the complexity of the clamp's design

Manually or automatic Speed with which the clamp can be used

- Relative ease with which a clamp can be positioned and
- tightened at point of contact Most economical element given the priority

For example low

Assigned to each clamp in a design group For example 4

6.10 Fastening devices

There are a great number of fastening elements used in a fixture assembly, typically screws, nuts and bolts, lock-rings, keys and pins. It is essential in the selection of these that non-standard elements be avoided whenever possible. Cost, reliability and life expectancy are the important attributes.

Two separate data structures have been written for representing the nut and the bolt and, since both items are very closely related, the attributes uses are the same and include the following:

- (a) identification attributes;
- (b) thread characteristics: class, series and form;
- (c) major dimensions: diameters and their respective tolerances;
- (d) thread particulars: thickness, depth, helix-angle;
- (e) operating ease, speed and economic factor.

© IMechE 1988

7 APPLICATIONS

To demonstrate the use of the knowledge representation database a number of typical examples are presented:

- 1. Overall representation of a typical medium size prismatic component identified as COMPONENT (Fig. 9).
- 2. A machine tool representation: a typical machining centre with three axes under NC control (courtesy of Wadkin Machine Tools plc) (Fig. 10).
- 3. Machining centres table representation (Fig. 11).
- Surface S1 of the prismatic component in Fig. 9 (Fig. 12).
- 5. A strap clamp (Fig. 13).

Inspection of these examples do highlight the following requirements of a database:

- 1. It should include most facilities of a manufacturing environment including details on the machine tool and its sub-assembly classifications, machining operations, workpiece and fixture specifications.
- 2. It must be organized into modules to allow direct



Fig. 9 Example illustrating the representation of a prismatic component

Proc Instn Mech Engrs Voi 202 No B1

(MAKE MACHINE-TOOL MACHINE-CLASSIFICATION GENERIC-NAME MACHINE-ID-CODE

MAIN-AXIS-DIRECTION MAX-MAIN-AXIS-THRUST MILLING-CAPACITY DRILLING-CAPACITY TAPPING-CAPACITY MAX-SPINDLE-HEIGHT-FROM-TABLE 85 ; mm

3-AXIS-MACHINING CENTRE V5-10-CNC z 15600; N 180 ; cm³/min 50 ; mm M30 Fig. 10 Representation of a machine tool

MILLING

(MAKE MACHINE-TABLE **IS-AN-ELEMENT-OF** V5-10-CNC MACHINE-ID-CODE V5-10-CNC TABLE-SIZE 1150 600 ; mm 1200 ; kg MAX-TABLE-LOAD-CAPACITY MAX-CROSS-TRAVERSE 500 ; mm (Y-AXIS) MAX-LONGITUDINAL-TRAVERSE 1000 ; mm (X-AXIS) MAX-VERTICAL-TRAVERSE 600 ; mm (Z-AXIS) TABLE-POSITION-ACCURACY 0.025; mm +/-TABLE-REPEATABILITY +/- 0.013NO-OF-T-SLOTS 4 DISTANCE-BETWEEN-T-SLOTS 130; mm)

Fig. 11 Representation of machine tool sub-assembly

(MAKE SURFACE	
IS-A	FEATURE
PART-NAME	COMPONENT 1
FEATURES-GENERIC-TYPE	EXTERNAL
FEATURE-TYPE	SURFACE
SURFACE-LABEL	St
SURFACE-TYPE	PLANAR
SURFACE-LENGTH	400 ; mm
SURFACE-LENGTH-TOL	.001 ; mm
SURFACE-WIDTH	300
SURFACE-WIDTH-TOL	.001
DIMENSIONAL-TOL	.002
SURFACE-POSITION	200
STATUS	MACHINED
HEAFTREATMENT	NIL
SURFACE-QUALITY	FINISH
DIRECTION-OF-EXTERNAL-NORMAL	-Z
REQUIRED-CUTTING-DIRECTION	х
IS-PARALLEL-TO-SURFACES	S6
PARALLELISM	.001
IS-PERPENDICULAR-TO-SURFACES	S2 S3 S4 S5
PERPENDICULARITY	.001
FEATURES-STARTING-FROM-THIS-SURFACE	NIL
FEATURES-OPENING TO-THIS-SURFACE)

Fig. 12 Typical surface (belonging to part component 1) representation

access to existing data structures, add data structures for new components and make modifications as experience with and knowledge from the system is gained.

- 3. Each individual data structure developed should include a sufficient number of attributes to ensure an unambiguous representation.
- Workpiece description should include physical, geometrical and technological attributes.
- To improve readability, mnemonic and informative file identifiers must be chosen to indicate their functions

Proc Instn Mech Engrs Vol 202 No B1



(MAKE CLAMPING-ELEMENT PART-NAME ID-CODE AKO TYPE GENERIC-NAME FUNCTION CLAMP-LENGTH CLAMP-SLOT-LENGTH CLAMP-SLOT-WIDTH CLAMP-THICKNESS CLAMP-WEIGHT SUITABLE-FOR-PART-SIZE PRINCIPLE-OF-ACTION NATURE-OF-CONTACT CLAMP-COMPLEXITY-FACTOR CLAMPING-SURFACE-QUALITY

OPERATING-SPEED

OPERATING-METHOD

EASE-OF-USE-FACTOR

ADJUSTMENT-RANGE

FRICTION-AND-JAMMING-FACTOR LOAD-UNLOAD-FACTOR

NO-OF-ELEMENTS-REQUIRED

COMPONENT1 STRAP-5 ; USER SUPPLIED CODE MECHANICAL-CLAMP FORK-TYPE STRAP-CLAMP CLAMPING 120 ; mm L 40 ; mm l 12-50 ; mm 16 : mm H -50 kg MEDIUM FIXED-IN-POSITION PLANE-TO-PLANE SIMPLEST ROUGH NORMAL MANUAL EASY ECONOMICAL 26 ; mm LOW LOW 4

- Fig. 13 Illustrates a fork-type strap clamp represented using the developed data structure
- 6. All rules in a grouping associated with a specific module must be prefixed by an identifier unique to the function fulfilled.

8 CONCLUSION

A significant amount of knowledge on the part to be machined is required to permit a fixture design to be chosen for a given manufacturing environment. In addition knowledge on the process plan, machine tool and its sub-assemblies is necessary together with detailed information on the fixture elements available if an optimum fixture choice is to be made.

The defined data structures presented should be utilized to ensure that the best fixture design is arrived at within the constraints imposed. Many of the attributes defined are self-evident; however, several attributes, of which friction is one, must be defined subjectively because of the difficulty of evaluating a precise numerical value.

The application of this concept requires that a production system language be available and the representation tailored to the needs of the individual user. The rules governing the interrelationship of the represented objects and those governing the design function must be developed, with particular attention being paid to the parts at hand and the concept of part families.

Progressive expansion, by the practitioner, of the limited knowledge base presented in this paper will ensure the database that evolves is applicable to the requirements of industry.

49

REFERENCES

- 1 Metal cutting: to-day's techniques for engineers and shop personnel, (Editors of American Machinist) 1979 (McGraw-Hill, New York).
- 2 Henrikson, E. K. Jig and fixture design manual, 1973 (Industrial
- Press, New York). 3 Storey, W. H. The fundamentals of jig design, 1957 (The Draughts-
- 4 Gouldson, C. J. Principles and concepts of fixturing for NC machining centers, in jigs and fixtures, (2nd edition W. Boyes) 1982 (Society of Manufacturing Engineers, USA).
- 5 Smith, W. F. A new approach to positioning and holding workpieces. Presented at Tool and Manufacturing Engineering Con-ference, Ohio Valley, October 1981 (Society of Manufacturing Engineers, USA).
- 6 Tuffentsmmer, K. Automatic loading of machining systems and automatic clamping of workpieces. Ann. CIRP, 1981, 30, 2.
- 7 Lewis, G. Modular fixturing systems. Proceedings of 2nd International Conference on Flexible Manufacturing Systems, 26-28 October 1983, London (Elsevier).
- 8 Neads, S. J., Graham, D. and Woodwark, J. R. Towards a fixturebuilding robot, School of Engineering, University of Bath, 1985.

- 9 Markus, A., Markusz, J., Farkas, J. and Filemon, H. Fixture design by prolog. Computer and Automation Institute, Hungarian Academy of Sciences, Budapest Technical University, 1982.
- Ingrand, F. and Latombe, J. Functional reasoning for automatic fixture design, 1980. (Lifia (IMAG) France).
 Winston, P. H. and Klaus, B. Lisp, 2nd edition, 1984 (Addison
- Wesley, London).
- Brownston, L. and Farrell, R. Programming expert systems in OPS5, 1985 (Addison Wesley, London).
 Buchanan, B. G. and Duda, R. O. Principles of rule based expert
- systems. In Advances in Computers, Vol. 22 (Ed. M. Yovitts) 1983
- (Academic Press, New York).
 14 McDermott, D. V. and Doyle, J. Non-monotonic logic 1. Artificial Intelligence, 1980, 13(1, 2), 41-72.
 15 Dieter, G. Engineering design a material and processing approach, 1982 (McCore) (1981). New York).
- 1983 (McGraw-Hill, New York).
- Warren, J. and Luzadder, P. E. Basic graphics for design, analysis, 16 communication, and the computer, 1968 (Prentice-Hall, New York).
- 17 Allen, D. K. and Smith, P. R. Part family classification and coding. Computer Aided Manufacturing Laboratory, Brigham Young University Provo, Utah, USA, 1982. Henrikson, E. K. Jig and fixture design manual, 1973 (Industrial
- Press, New York).

PAPER 61

Use of Fuzzy Logic in Robotics

B.A.M. Wakileh and K.F. Gill

Department of Mechanical Engineering, University of Leeds, Leeds LS2 9JT, U.K.

An investigation is described that attempts to demonstrate the benefits that can be gained by the use of a fuzzy logic control law. Employing such an algorithm avoids the need for a detailed mathematical description of a manipulator link and the algorithm is inherently more robust than a conventional controller

By adopting a multi-valued parameter mode of operation it is shown that good dynamic and steady state response for a wide range of input demands can be achieved.

Keywords: Fuzzy logic, Controller, Robot control



Dr. Gill obtained his engineering training with a company manufacturing large steam turbines and alternating equipment for the electrical power supply industry. His academic training was gained at the University of Durham and the University of Birmingham. He was awarded the degrees of B.Sc. in 1955, M.Sc. in 1958 and Ph.D. in 1961. From 1961 Dr. Gill was a Senior Scientific Officer at the Admiralty Weapons Establishment, Portland. For the last 25 years

has been a Lecturer, then Senior Lecturer at the University of Leeds, teaching control engineering and engineering dynamics to undergraduate students reading for an honours degree in Mechanical Engineering.

North-Holland Computers in Industry 10 (1988) 35-46

1. Introduction

It has been recognised in the published work that real time control for a manipulator based on a detailed dynamic model is difficult to achieve if not impossible [1-3]. The equations used to represent a manipulator mathematically are both complex and non-linear, difficult to handle computationally and require a relatively long computer "run time" for their solution. The requirement exists, therefore, for an alternative approach.

It has been reported [4-16] that fuzzy logic controllers are successful when applied to plants that are difficult to model precisely. The general opinion is that its use should only be considered when conventional control design techniques have proven inadequate, yet a human operator has been shown to cope adequately in similar situations. If this recommendation was followed, the range of fuzzy logic application would be extremely limited. It has been reported in the application studies that the fuzzy logic controller is more robust to plant parameter value changes than a classical control algorithm and has better noise rejection capabilities. By the very nature of the fuzzy logic controller design, one would expect it to be robust, which is clearly a very desirable characteristic to have and this alone may be sufficient justification for its wider use. It is therefore an area worthy of investigation that may lead to a more general





Dr. Wakileh then returned to Jordan and is currently completing his military service.

0166-3615/88/\$3.50 © 1988 Elsevier Science Publishers B.V. (North-Holland)

36 Forum on Fuzziness

design philosophy in robotic control.

A basic familarity with the 'simple' fuzzy logic controller is assumed; the unfamiliar reader is referred to [17,18].

2. Description of Fuzzy Logic Control Law

The fuzzy logic controller has been implemented with little comment on its origin other than it reflected the linguistic control policy of an experienced operator. This is evident by the similarity of approach adopted in the various published applications employing a fuzzy logic control strategy. A typical structure would be as illustrated in Fig. 1.

Error or sum of errors in conjunctions with change in error are the most used antecedent fuzzy variables. Whereas, an absolute output fuzzy variable or a change in the output is usually employed as a consequent. A combination of both is used in some cases as an attempt to optimize the response.

Similar labels are adopted to quantify the fuzzy variables and these are typically, positive big PB, negative medium NM, etc. The discrete quantisation used in the fuzzy set operation, Q, in Fig. 1, range between 5 to 15 discrete levels in the majority of cases. Only Kickert and Van Nauta Lemke [4] selected a functional form to describe the discretisation of a fuzzy set.

The most frequently used methods to generate a continuous output, from a controller operation M in Fig. 1, are:

- 1. The mean of a maxima procedure in which the discrete element with the maximum membership function value is selected or the average value of the maxima in the case of multiple maxima.
- 2. The centre of area procedure in which the selected deterministic output has a value that divides the area under a fuzzy set into two equal halves.

In all applications non fuzzy measurements are presented as fuzzy measurements are presented as fuzzy singletons. This enables the algorithm to be pre-calculated and expressed as a multilevel relay.

3. Description of the Process

To test the proposed control strategy the Stanford manipulator, *Fig. 2*, was adopted for the study. This manipulator comprises six joints in which the third is prismatic, thus the structure has six degrees of freedom.

The first three links of the manipulator called the post, shoulder and boom, with associated variables θ_1 , θ_2 and d_3 respectively, form the manipulator's positional section and is the heaviest part of its structure. The last three joints form the end



Fig. 1. General form of the fuzzy logic controller.

Computers in Industry



Fig. 2. The Slanford Manipulator with reference frames of joint-link subassemblies.

effector with the associated motion variables θ_4 , θ_5 and θ_6 respectively. The axis system employed was defined by Denavit and Hartenberg [19] and the associated initial condition kinematic parameter values for the manipulator configuration shown in *Fig. 2* are listed in *Table 1* for reference.

Each manipulator joint is actuated by a separate D.C. armature controlled electric motor; torque amplification being achieved by an appropriate speed reduction gearbox. The supply voltage used is 90V and the pertinent dynamic equation is listed here for completeness, i.e.

$$K_1 V = J(\theta) \ddot{\theta} + C(\theta, \dot{\theta}) + G(\theta) + K_2 \dot{\theta}, \qquad (1)$$

where

- V : 6×1 vector of actuators supply voltage
- K_1 : 6 × 6 diagonal matrix defining actuators gains
- K_2 : 6×6 diagonal matrix defining actuator feedback terms
- $J(\theta)$: 6 × 6 inertia matrix
- $C(\theta, \dot{\theta})$: vector defining Coriolis and Centrifugal terms

Table 1				
Kinematic parameter	values	for	initialisation	configuration

Joint number	θ degrees	d metres	a metres	α degrees	Range degrees
1	0	0.54	0	- 90	- 180 to 180
2	90	0.162	0	- 90	-180 to 180
3	- 90	-	0	0	0 to 1.1m
4	0	0	0	- 90	-180 to 180
5	0	0	0	90	- 90 to 90
6	0	0.25	0	0	-180 to -180

 $G(\theta)$: vector defining gravity terms.

All the relevant details of Eq. (1) can be found in [2] and in this study the design values for matrices K_1 and K_2 are:

	1.02	0	0	0	0	0]
:	0	3.37	0	0	0	0
<i>v</i> –	0	0	1.1	0	0	0
$n_1 - 1$	0	0	0	0.113	0	0
	0	0	0	0	0.113	0
	LO	0	0	0	0	0.08
	69.5	0	0	0	0	0]
	0	229.5	0	0	0	0
K -	0	0	74.7	0	0	0
$\Lambda_2 -$	0	0	0	5.5	0	0
	0	0	0	0	6.5	0
	LO	0	0	0	0	7.9

since these numerical values are not given in [2].

4. Design of a Fuzzy Logic Controller

The design strategy to be adopted for a robot link is to actuate point to point movements of a single joint of the manipulator with all other joints in a temporary state of rest. By actuating individual joints sequentially along a pre-programmed displacement path, the end effector will eventually be located at a desired position in its own working environment.

This sequential movement is adopted because of the difficulty of obtaining sufficient information from which a "rule algorithm" can be constructed. This algorithm, if available, would allow corrections to be made for dynamic coupling and the non-linear properties of the manipulator when the joints are in motion simultaneously. A fuzzy logic controller is presented that will actuate joint

Table	2		
Fuzzy	logic	lingustic	algorithm

Computers in Industry

2 of the Stanford manipulator and the extension to the other joints is a relatively straightforward modification to introduce. Joint 2 is chosen because it demonstrates more clearly the impact on controller design of:

1. significant gravitational and inertial forces,

- 2. positional accuracy constraints, and
- 3. major inertia variations with loading.

It is assumed that displacement and speed data are available from sensors mounted at the joint. A fuzzy logic linguistic control algorithm is written using engineering intuition to describe a one input monotonic undamped process. The control algorithm is intended to provide fast convergence consistent with adequate damping to achieve a high angular rotational accuracy.

The proposed algorithm is represented by *Table 2* and is intended to be read as:

IF joint displacement is *PB* and joint speed is *PB* THEN voltage input to joint actuator is *NB* ELSE...

The variable identifiers used to describe the fuzzy values have the following meanings;

PB Positive Big

- PM Positive Medium
- PS Positive Small
- PO Positive Zero
- ZO Zero
- NO Negative Zero
- NS Negative Small
- NM Negative Medium
- *NB* Negative Big

and are consistent with those reported in most of the application studies.

The terms NO and PO are introduced in accordance with Daley [16] to give finer tuning around the equilibrium state. Term PO defines a region that is slightly above zero and NO a region

Joint	Joint speed										
displacement	PB	PM	PS	ZO	NS	NM	NB				
PB	NB	NB	NB	NB	NM	NM	NM				
РМ	NB	NB	NB	NB	NM	NM .	NS				
PS	NM	NS	NS	NS	NS	PS	PM				
PO	NM	NM	NS	ZO	ZO	PS	PM				
NO	NM	NS	ZO	ZO	PS	PM	PM				
NS	NM	NS	PS	PS	PS	PS	PM				
NM	PS	PM	PM	PB	PB	PB	PB				
NB	PM	PM	PM	PB	PB	PB	PB				

Computers in Industry

that is slightly below zero in the state space.

Fuzzy sets formed to represent the discrete support universes consist of 13 elements for the joint speed, 14 elements for the joint displacement and 15 elements for the voltage input to an actuator. Appropriate membership functions are assigned to each element of the support set using the definitions by Mamdani and Assilian [14], an approach employed in preference to a functional method [4] because of its manipulative simplicity. This yields the fuzzy set values listed in *Table 3*. It should be noted that +0 and -0 discrete support elements are the values chosen to ensure finer control around the equilibrium state.

The non-fuzzy measurements are expressed as fuzzy singletons in the algorithm whose output represents an absolute voltage and not a change in voltage. The magnitude of this voltage is generated using the mean of the maxima procedure.

The flow diagram presented in Fig. 3 and the fuzzy sets in Table 3, allow the linguistic control algorithm to be converted into a table of discrete

.

Table 3 The fuzzy set definition

B.A.M. Wakileh and K	.F. Gill	/ Fuzzy Logic in Robotics	39
----------------------	----------	---------------------------	----

controller outputs when the compositional rule of inference is used. These resultant outputs are computed off line and used in a multilevel relay, *Table* 4, to dramatically reduce the number of control algorithm manipulations required. The basic set operations to calculate the multilevel relay values using fuzzy calculus can be defined as follows:

i. The union of two fuzzy sets A and B of the universe of discourse E denoted $A \cup B$ has the membership function defined by

$$\mu_{A\cup B}(e) = \max[\mu_A(e); \mu_B(e)] \ e \in E.$$

e is a generic element of E and expresses the membership value of e in A and B and taking a value in the interval [0,1].

ii. Intersection of A and B is denoted $A \cap B$ and has a membership function defined by

 $\mu_{A\cap B}(e) = \min \left[\mu_A(e), \, \mu_B(e) \right] \, e \in E.$

iii. Complement of a fuzzy set A is denoted a 7A and has a membership function defined by

$$\mu_{7A}(e) = 1.0 - \mu_A(e) \ e \in E.$$

The fu	izzy set d	efinitions													
	-6	-5	-4	- 3	-2	-1	-0	+0	1	2	3	4	5	6	
PB	0	0	0	0	0	0	0	0	0	0	0.1	0.4	0.8	1.0	
PM	0	0	0	0	0	0	0	0	0	0.2	0.7	1.0	0.7	0.2	
PS	0	0	0	0	0	0	0	0.3	0.8	1.0	0.5	0.1	0	0	
PO	0	0	0	0	0	0	0	1.0	0.6	0.1	0	0	0	0	
NO	0	0	0	0	0.1	0.6	1.0	0	0	0	0	0	0	0	
NS	0	0	0.1	0.5	1.0	0.8	0.3	0	0	0	0	0	0	0	
NM	0.2	0.7	1.0	0.7	0.2	0	0	0	0	0	0	0	0	0	
NB	1.0	0.8	0.4	0.1	0	0	0	0	0	0	0	0	0	0	
a) Join	t displace	ement θ_2													
	-6	-5	-4	- 3	-2	-1	0'	1	2	3	4	5	6		
PB	0	0	0	0	0	0	0	0	0	0.1	0.4	0.8	1.0		
PM	0	0	0	0	0	0	0	0	0.2	0.7	1.0	0.7	0.2		
PS	0	0	0	0	0	0	0	0.9	1.0	0.7	0.2	0	0		
NO	0	0	0	0	0	0.5	1.0	0.5	0	0	0	0	0		
NS	0	0	0.2	0.7	1.0	0.9	0	0	0	0	0	0	0		
NM	0.2	0.7	1.0	0.7	0.2	0	0	0	0	0	0	0	0		
NB	1.0	0.8	0.4	0.1	0	0	0	0	0	0	0	0	0		
b) Joir	nt speed b	2													
	-7	-6	- 5	-4	- 3	-2	-1	0	1	2	3	4	5	6	7
РВ	0	0	0	0	0	0	0	0	0	0	0	0.1	0.5	0.8	1.0
PM	0	0	0	0	0	0	0	0	0	0.2	0.7	1.0	0.7	0.2	0
PS	0	0	0	0	0	0	0	0.4	1.0	0.8	0.4	0.1	0	0	0
NO	0	· 0	0	0	0	0	0.2	1.0	0.2	0	0	0	0	0	0
NS	0	0	0	0.1	0.4	0.8	1.0	0.4	0	0	0	0	0	0	0
NM	0	0.2	0.7	1.0	0.7	0.2	0	0	0	0	0	0	0	0	0
NB	1.0	0.8	0.4	0.1	0	0	0	0	0	0	0	0	0	0	0
c) The	process in	nput, V				-									
	p1000331				·										

40 Forum on Fuzziness

iv. A fuzzy relation R from $U = \{x\}$ to $V = \{y\}$ is a fuzzy set on the Cartesian product $U \times V$ characterised by a function $\mu_R(x, y)$ by which each pair (x, y) is assigned a membership value indicating to what extent the relation is true in (x, y). The linguistic statement "If A Computers in Industry

then B" is the fuzzy relation $R = A \times B$. This has a membership function defined by

$$\mu_R(x, y) = \mu_{A \times B}(x, y)$$
$$= \min[\mu_A(x); \mu_B(y)]$$

 $x \in U$ and $y \in V$.



Fig. 3. Flow chart for calculation of multilevel relay.

Computers in Industry

v. Given the fuzzy relation $R = A \times B$ and a given fuzzy subset value A' of A, the corresponding value B' is inferred from the relation using the compositional rule of inference as

 $B' = A' \circ R = A' \circ (A \times B),$

• denoting the composition operation, and the membership function is defined by

 $\mu_{B'}(y) = \max \min [\mu_{A'}(x); \mu_{R}(x, y)].$

The mean of maxima procedure selects the discrete element of a fuzzy set with the maximum membership function value or the average value of the maxima in the case of multiple maxima.

5. Selection of Scaling Factors

To permit the multilevel relay output to be used as the manipulated variable it must be scaled to the real universe of the process.

The scaling of the controller output has been initially selected to provide the maximum chosen attainable actuator input of 90 volts. Therefore the output scaling factor GU is

GU = 90/7 = 12.86

Trial and observation procedure is adopted during controller development to choose the joint displacement scaling factor GE and the joint speed scaling factor GC. The choice of these factors is made on line, so as to tune the controller to the B.A.M. Wakileh and K.F. Gill / Fuzzy Logic in Robotics 41

discrete computation interval of 0.01 second that is used in the study.

The robustness of the fuzzy controller algorithm, applied to a robot, is tested with joint 2 subjected to the maximum permissible change in parameter values. The joint sustains its maximum inertial and gravitational forces when the manipulator boom (link 3) is horizontal and at full extension. The arrangement of the robot links is:

joint angles $\theta_1, \theta_4, \theta_5, \theta_6 = 0$

oint angle
$$\theta_2 = 90$$
 degrees,

link $d_3 = 1 m$, the initial condition for the robot.

To obtain the maximum physically realizable joint displacement (radians) with maximum joint speed (radians/second), the values of GE and GC are constrained to be:

$$GE = \frac{6}{(\text{maximum required joint displacement})}$$

and

$$GC = 6/1.25 = 4.8$$

where the maximum joint speed attainable is 1.25 rad/sec.

Employing these values, the robot arm, Fig. 2, is emulated on a Amdahl (470) digital computer and is incorporated in the block diagram shown in Fig. 4. The scheme assumes that the joint sensors are adjusted to zero for the starting configuration (initial condition); a practice normally followed with a robotic system.



Fig. 4. An outline of the manipulator joint and fuzzy logic controller.

42 Forum on Fuzziness

Table 4

Computers in Industry

Fuzzy	logic con	trol law n	nodelled a	as multile	vel relay								
	-6	-5	-4	-3	-2	-1	0	1	2	3	4	5	6
-6	7	6	7	6	7	7	7	4	4	4	4	4	4
- 5	6	6	6	6	6	6	6	4	4	4	4	4	4
- 4	7	6	7	6	7	7	7	4	4	4	4	2	1
-3	6	6	6	6	6	6	6	4	4	4	4	3	2
- 2	4	4	1	2	1	1	1	1	1	0	-1	-4	- 4
-1	4	4	2	1	2	2	2	2	2	0	-1	-4	- 4
-0	4	4	3	2	1	1	0	-1	0	- 3	-1	-4	- 4
0	4	4	3	2	0	1	0	-1	-1	- 3	-1	-4	- 4
1	4	4	2	0	-1	-1	-1	-1	-1	-1	-1	- 4	4
2	4	4	1	0	-1	-1	-1	-1	-1	-1	-1	-4	-4
3	-1	- 3	- 4	- 4	-4	-4	-6	-6	-6	-6	-6	-6	- 6
4	-1	-2	-4	- 4	- 4	- 4	-7	-7	-7	-6	-7	-6	-7
5	-4	-4	-4	-4	-4	-4	-6	-6	-6	-6	-6	-6	-6
6	- 4	-4	4	-4	4	-4	7	-7	-7	-6	-7	-6	-7
Table GE va	5 llues for d	lifferent	simulated	joint disp	lacement	s							
Joint o	lisplacem	ent (degre	ees)	30		15		7	4		2	1	
GE va	lues			11.5	5	23	4	46	85		171	343	3

Simulation runs made for a range of joint displacements, result in the GE values given in *Table* 5 being selected.

The time responses of Fig. 5 show a steady convergence to the equilibrium state for the range of input demands applied, unfortunately the steady state error in each case is relatively large. This error can be reduced by increasing the value of GE. The error magnitude ER/GE depends upon the discrete joint displacement value ER such that the actual steady state error is given in the range of 0.02 < ER/GE < 15 (degrees) for ER and GE constrained to the ranges

1 < ER < 3 and 3000 < GE < 11.

RESPONSE	GE	GC	GU	INITIAL CONDITION DEG.	FINAL CONDITION DEG
а	85	4.8	12.85	90.0	86.00
b	46	4.8	12.85	90.0	83.00
C	23	4.8	12.85	90.00	75.00
d	11.5	4.8	12.85	90.00	60.00



Fig. 5. Responses - simple fuzzy logic controller.

RESPONSE	GE	GC	GU	INITIAL CONDITION DEG	FINAL CONDITION DEG
a	343	4.8	12.85	90.0	89.00
b	171	4.8	12.85	90.0	88.00



Fig. 6. Responses - simple fuzzy logic controller.

Computers in Industry

RESPONSE	GE	GC	GU	INITIAL CONDITION DEG	FINAL CONDITION DEG
a	685	4.8	12.85	90.0	89.50
b	685	8.8	6.6	90.0	89.50
c	685	17.8	.3.4	90.0	89.50



Fig. 7. Responses - simple fuzzy logic controller.

Steady state error is inherent with the fuzzy logic system adopted due to the coarseness of the support set (14 elements) used to represent the full range of joint displacement. For instance, the discrete error value +0 in the fuzzy set is equivalent to a real displacement value somewhere in the range 0 to 5 degrees for GE = 11.5.

To study the positional accuracy in detail, the input disturbances will be restricted to small angular movements. Using these small disturbances, *Fig.* 6, with values of GE in excess of 170 leads to oscillatory responses.

A limiting condition is reached for a value of GE = 685, thereafter oscillatory motion results, *Fig. 7a*.

In an attempt to minimise the steady state error the value of the scaling factor GC was varied. Increasing values of GC requires that the maximum permissable joint speed be reduced. This leads to a reduction in the process input scaling factor GU because of the assumed steady state operation relationship between maximum actuator supply voltage (V_{max}) and the attainable joint speed $(\dot{\theta}_{max})$ at that input, i.e.

$$V_{\max} \cong K_e \dot{\theta}_{\max}, \tag{2}$$

where K_e is the actuator constant of proportionality.

Initially, the maximum allowable joint speed is reduced to half its original value. This allows values of GE = 685, GC = 8.8 and GU = 6.6 to be used and Fig. 7 shows that the initial overshoot is

B.A.M. Wakileh and K.F. Gill / Fuzzy Logic in Robotics 43

RESPONSE	GE	GC	GU	INITIAL CONDITION DEG	FINAL CONDITION DEG
а	3440	18	3.5	90.0	89.90
Ь	3440	24	2.5	90.0	89.90
C	3440	32	1. 9	90.0	89, 90



Fig. 8. Responses - simple fuzzy logic controller.

reduced, but the response is still oscillatory. Improvement is made when the joint actuator speed is further reduced, Fig. 7c, using values for the scaling factor of GE = 685, GC = 17.8 and GU = 3.4.

Clearly using the above settings an improvement in time response is achieved, but the steady state errors are still unacceptably large. It is also noted that improved response for large values of GE are achieved using larger GC values. The effect of increasing GC for small joint displacement is tested for a joint displacement of order 0.1 degrees, for the three different joint rate scaling factors GE = 3440, GC = 32, 24 and 18 and GU =1.9, 2.5 and 3.5.

The responses, Fig. 8, show that there is no well defined optimum for constant scaling factor values at which the steady state error is deemed acceptable. The response rapidly becomes unstable at higher or lower speed values.

6. An Improved Fuzzy Logic Control Algorithm Operating in a Multi-Valued Parameter Mode

A design conflict between positional accuracy and transient behaviour has been demonstrated. Modification to the control law is possible that will combine both the superior transient response with steady state precision for a range of input demands.

Table 6					
Operation	range of	the multi-va	lued para	meter con	troller

optimite				
Angle (degrees)	GE	GC	GU	
$\theta_2 > 1$	6/(maximum joint displacement)	4.8	12.85	
$1 > \theta_2 > 0.1$	340	8.8	6.6	
$\theta_2 < 0.1$	3440	32	1.9	

To maintain this performance over a wide range of input demands, the controller is made to operate in a multi-valued parameter mode. A switch is introduced that will permit changes to be made in the scaling factor values at the boundaries of predefined displacement speed regions and these are listed in *Table 6*.

Before a switch can be actuated at a parameter value boundary, the inequalities

abs (real joint displacement) $< Js/GE_b$ and

abs (real joint speed) $< JRs/GC_b$

must be satisfied. The parameters GE_b and GC_b are the scaling factors before the switch and Jsand JRs are scaled discrete values of the joint displacement and speed, respectively, when the switch is operated. In the event of divergence in the displacement speed responses, the switch has the capability to revert back to a convergent mode when

abs (real joint displacement) > Jr/GE_b and

abs (real joint speed) > JRr/GC_b .

The parameters Jr and JRr are the scaled discrete joint displacement and speed, respectively, after the switch is made.

The values chosen for Js, JRs, Jr and JRrshould be in the range 0 to 1 to allow the use of the full range of the multi-level relay. The actual values used for JRs, Jr and JRr was unity, however, for joint displacements greater than 1 degree, a better value of Js was found to be 3. A more reliable approach because of this Js value would be to observe the controller output and switch on the conditions

The scaling factor switch defined by Eq. (3) is tested for the two arbitrarily input demands of 5

CONDITION	CONDITION	LOAD MASS KG
90. 0 90. 0	89.0 85.0	0.0
	CONDITION DEG 90. 0 90. 0	CONDITION CONDITION DEG DEG 90.0 89.0 90.0 85.0



Fig. 9. Responses - multi-mode controller.

and 1 degree, respectively. The performance shown in *Fig. 9* illustrates the improvement attained adopting this alternative control algorithm; the responses converge to an equilibrium region of width better than 0.06 degrees. When the end

RESPONSE	INITIAL RESPONSE CONDITION DEG		LOAD MASS KG		
6	90.0	89.00	1.8		
Ь	90.0	85.00	1,8		
C C	90.0	80.00	1.8		



Fig. 10. Responses - multi-mode controller.

Computers in Industry

Computers in Industry

joi Num	INT 18ef	2	1	2	3	4	5	6
ISE (a)	INITIAL	N DEG	60.0	30.0	0.0	50.0	60.0	0.0
RESPON	FINAL	CONDITIC	60. 0	31.0	0.0	50.0	60.0	0.0
ISE (b)	INITIAL	IN DEG	60.00	30.0	0.0	50.0	0.0	3.0
RESPON	FINAL	CONDITIO	60.0	34.0	0.0	50.0	0.0	3.0
ISE (c)	INITIAL	ON DEG	50.0	60.0	0.0	30.0	0.0	0.0
RESPON	FINAL	CONDITI	50.0	70.0	0.0	30.0	0.0	0.0



Fig. 11. Responses - multi-mode controller.

effector is carrying the maximum allowable load of 1.8 kg, the transient response of Fig. 10 shows some deterioration in positional accuracy. However, this is directly attributable to the changed value of effective link inertia with the loaded condition. An inertia variation of 5:1 exists between loaded and unloaded states.

Operating the link from different initial conditions, Fig. 11, produced no significant variation in the quality of the output response. The variations in performance with both changes in initial conditions and input disturbance is sufficient to demonstrate that the operating region to be used is an important feature and should be included at the design stage to ensure that the most satisfactory overall performance is achieved.

7. Conclusions

The basic idea behind the introduction of a fuzzy logic controller is to avoid a design strategy

B.A.M. Wakileh and K.F. Gill / Fuzzy Logic in Robotics 45

based on a detailed dynamic model by employing the approach of a human operator to an ill-defined system. The application is presented in more general terms to observe any advantages it may possess over a conventional controller.

The authors believe the study does demonstrate that fuzzy logic offers a novel approach to robot control that avoids the two major problems areas, accurate dynamic modelling over the whole of the operating environment and computational time constraints. The results have confirmed that fuzzy logic does allow the design of a robust control algorithm which is likely to find engineering application with the more intelligent robot system.

References

- Paul, R.: Modelling Trajectory Calculation and Serving of a Computer Controlled Arm. Stanford Artifical Intelligence Laboratory, A.I. Memo 177, Nov. 1972.
- [2] Bejczy, A.K.: Robot Arm Dynamics and Control. Technical Memo 33-669, Jet Propulsion Laboratory, Feb. 1974.
- [3] Lee, C.S.G., Chung, M.J., Turney, T.L., Mudge, T.N.: On the Control of Mechanical Manipulators. Proceedings of the Sixth IFAC Conference in Estimation and Parameter Identification, Washington D.C., June 1982, pp. 1454-1459.
- [4] Kickert, W.J.M. and Van Nauta Lemke, H.R.: Application of a Fuzzy Controller in a Warm Water Plant. *Automatica*, 1976, 12, 301-208.
- [5] Kickert, W.J.M.: Further Analysis and Application of Fuzzy Logic Control. Internal report F/WK22/75, 1975. Queen Mary College, London.
- [6] Østergaard, J.J.: Fuzzy Logic Control of a Heat Exchanger Process. Internal report 7601, 1976. Electrical Power Engineering Department, Technical University of Denmark.
- [7] Rutherford, D.A. and Carter, G.A.: A Heuristic Adaptive Controller for a Sinter Plant. Proceedings of the Second IFAC Symposium on Automation in Mining, Mineral and Metal Processing, Johannesburg, 1976.
- [8] Tong, R.M.: Some Problems with the Design and Implementation of Fuzzy Controllers. Internal report CUED/F-CAMS/TR 127, 1976, Cambridge University.
- [9] Van Amerongen, J., Van Nauta Lemke, H.R. and Van der Veen, J.C.T.: An Autopilot for Ships Designed with Fuzzy Sets. In: Digital Computer Applications to Process Control (Ed. H.R. Van Nauta Lemke), 1977 (IFAC and North-Holland).
- [10] Larsen, P.M.: Industrial Applications of Fuzzy Logic Control. Int. J. Man-Mach. Stud., 1979, 12, 3-10.
- [11] Tong, R.M., Beck, M.B. and Latten, A.: Fuzzy Control of the Activated Sludge Wastewater Treatment Process. Automatica, 1980, 16, 659-701.
- [12] Sheridan, S.E. and Skjoth, P.: Automatic Kiln Control at Oregon Portland Cement Company's Durkee Plant Utilising Fuzzy Logic. Proceedings of the 25th IEEE Technical Conference on Cement industry, San Antonio, 1983.

46 Forum on Fuzziness

- [13] Mamdani, E.H.: Application of Fuzzy Algorithms for Control of Simple Dynamic Plant. Proc. IEE, 1974, 121, 1585-1588.
- [14] Mamdani, E.H. and Assilian, S.: An Experiment in Linguistic Synthesis with a Fuzzy Logic Controller. Int. J. Man-Mach. Stud., 1975, 7, 1-13.
- [15] King, P.J. and Mamdani, E.H.: The Application of Fuzzy Control Systems to Industrial Processes. *Automatica*, 1977, 13, 235-242.
- [16] Daley, S.: Analysis of Fuzzy Logic Control Algorithms

and their Application to Engineering Systems. Ph.D. Thesis, Mech. Eng. Dept., University of Leeds, Sept. 1984.

- [17] Mamdani, E.H.: Advances in the Linguistic Synthesis of Fuzzy Controllers. Int. J. Man-Mach. Stud., 1976, 669-678.
- [18] Tong, R.M.A.: Control Engineering Review of Fuzzy Systems. Automatica, 1977, 13, 559-569.
- [19] Depavit, J. and Hartenberg, R.S.: A Kinematic Notation for Lower-pair Mechanisms Based on Matrices. Journal of Applied Mechanics, June 1955, pp. 215-221.

Computers in Industry

PAPER 62

A COMPLETE DESCRIPTION OF SURFACE TEXTURE PROFILES

D. J. MULVANEY

Cambridge Consultants, Cambridge (U.K.) D. E. NEWLAND and K. F. GILL Department of Mechanical Engineering, University of Leeds, Leeds (U.K.) (Received November 23, 1988; accepted February 9, 1989)

Summary

Previous assessments of the surfaces of machined components have failed to identify an underlying form to the texture and hence no method of completely characterizing a wide range of such surfaces has been devised. In this paper, the reason for this failure is shown to be the current standard practice of confining analysis to the shorter wavelengths of the texture. The extension of analysis so as to include the longer wavelengths permits the derivation of a complete description of the surface texture of machined components drawn from a large sample population and production processes. In addition, this approach enables surface texture to be fully described by only two parameters, both of which are defined so as to be easily understood in terms of present methods of analysis.

1. Introduction

Surface texture data can be readily shown to have a random nature [1 - 3] and hence statistical methods of investigation are normally employed. Descriptions of previous attempts to produce characterizations of surface texture based on such investigations can be found in the literature [4 - 6]. Much of the work related to surface texture analysis has assumed that at the longer wavelengths the data are non-stationary and that only the shorter wavelengths are stationary, for example Whitehouse and Phillips [7]. This practice has become widely established; the British Standard [8] recommends the use of one of five standard filters to attenuate the longer wavelengths in the data to be included for analysis. However, as a filter is selected by an operator according to a personal assessment of the nature of the surface texture, the selection of an inappropriate filter would adversely affect the values of the parameters calculated [9]. In the work presented here, it will be shown that such filtering is unnecessary and consequently subjective surface analysis can be avoided.

0043-1648/89/\$3.50

© Elsevier Sequoia/Printed in The Netherlands

2. Analysis of the longer wavelengths of the surface texture

The only previous attempt to extend the analysis to longer wavelengths relied on the texture being non-stationary. This investigation was performed by Sayles and Thomas [6] and their description was based on

$$G(f) = \frac{k}{f^2} \tag{1}$$

where f is the surface frequency and G(f) is the power spectrum. The surface frequency is the reciprocal of the corresponding wavelength on the surface, for example a wavelength of 1 mm corresponds to a surface frequency of 1 kHz. The term "topothesy" was used for k, a parameter whose value is determined by the nature of the surface and which, if the spectral estimate of the surface is of the form given in eqn. (1), will describe completely the form of the surface texture.

To test the hypothesis of Sayles and Thomas, 144 specimens were prepared so as to include examples of surface textures produced by end-milling, fly-milling, slab-milling, shaping, turning and grinding. The machinery parameters were speed of cut, feed rate and depth of cut. By maintaining in turn two of these parameters constant, three specimens were produced for each of three selected values of the remaining parameter, resulting in a group of 27 different specimens for each of the machining processes. Tables 1 and 2 present this in detail. The machinery parameters were purposely selected in this manner so as to aid the identification of the cause of features in the spectral estimates. In the normal wavelength range for surface texture analysis [8], eqn. (1) can be seen to apply for a number of sample surfaces [10], Fig. 1. However, for the longer wavelengths termed "waviness", the results from a number of sample surfaces did not agree with the findings of Sayles and Thomas, for example Fig. 1(c), and this prompted the investigation now described.

First, it is necessary to develop a smoothing technique which will preserve the lowest surface frequencies present in the data for display in the spectral estimates. The method used involved smoothing the periodogram with a non-recursive second-order digital filter. As far as the authors are aware, waviness has not been previously investigated using spectral analysis owing to the difficulty in formulating an appropriate spectral smoothing procedure. Although in ref. 10 traces of length up to 60 mm were taken, the application of a spectral window and segment averaging of an appropriate length reduced the maximum wavelength present in the spectral estimates to 0.5 mm. To smooth the spectral estimate curves (that is to reduce the variance of the spectral estimates) without truncation of the lower frequency band, the Fourier transform of the data was performed in a single segment and a non-recursive second-order digital filter applied to the spectral data. The smoothed spectral estimate is computed by forming a weighted sum of the discrete, unsmoothed spectral values. For a point which is a distance n points in the unsmoothed spectral estimate from the

TABLE 1

,

.

.

.

Machined specimens

(a) Slab milled

Speed (rev min ⁻¹)	25			72			206		
Feed rate (mm min ^{-1})	32	104	254	32	104	254	32	104	254
Depth of cut 0.13 mm	1	2	3	10	11	12	19	20	21
0.25 mm	4	5	6	13	14	15	22	23	24
0.51 mm	7	8	9	16	17	18	25	26	27
(b) Shaped									
Speed (m min ⁻¹)	10			20			30		
Feed (mm)	0.30	0.63	0.94	0.30	0.63	0.94	0.30	0.63	0.94
Depth of cut 0.13 mm	1	2	3	10	11	12	19	20	21
0.25 mm	4	5	6	13	14	15	22	23	24
0.51 mm	7	8	9	16	17	18	25	26	27
(c) End milled									
Speed (rev min ⁻¹)	105			230			530		
Feed rate (mm min ^{-1})	12.7	31.8	63.4	12.7	31.8	63.4	12.7	31.8	63.4
Depth of cut 0.13 mm	1	2	3	10	11	12	19	20	21
0.25 mm	4	5	6	13	14	15	22	23	24
0.51 mm	7	8	9	16	17	18	25	26	27
(d) Fly milled					•				
Speed (rev min ^{-1})	230			530			810		
Feed rate (mm min ⁻¹)	12.7	31.8	88.9	12.7	31.8	88.9	12.7	31.8	88.9
Depth of cut 0.13 mm	1	2	3	10	11	12	19	20	21
0.25 mm	4	5	6	13	14	15	22	23	24
0.51 mm	7	8	9	16	17	18	25	26	27
(e) Turned									
Speed (rev min ⁻¹)	125			260			540		
· · · · · · · · · · · · · · · · · · ·									
Feed rate (mm min ⁻¹)	20	40	80	40	80	160	80	160	320
Feed rate (mm min ⁻¹) Depth of cut 0.13 mm	20 1	40 2	80 3	40 10	80 11	160 12	80 19	160 20	320 21
Feed rate (mm min ⁻¹) Depth of cut 0.13 mm 0.25 mm	20 1 4	40 2 5	80 3 6	40 10 13	80 11 14	160 12 15	80 19 22	160 20 23	320 21 24

176

TABLE 2

Ground specimens

Depth of cut 0.005 mm 1 0.01 mm 4 0.04 mm 7 $\hat{G}(f)$ I 95 Per Cent $\hat{G}(f)$ (μm^2) 10 ⁻⁶ 10 ⁻⁶	2 5 8	0.1
$ \hat{G}(f) = \begin{bmatrix} 95 & Per & Cent \\ (\mu m^2) \end{bmatrix} $ $ 10^{-6} = \begin{bmatrix} 1 & Confidence & Interval \\ 10^{-6} \end{bmatrix} $ $ 10^{-6} = \begin{bmatrix} 10^{-6} & 10^{-6} \\ 10^{-7} & 10^{-4} \end{bmatrix} $ $ 10^{-6} = \begin{bmatrix} 10^{-6} & 10^{-6} \\ 10^{-6} & 10^{-6} \end{bmatrix} $ $ 10^{-6} = \begin{bmatrix} 10^{-6} & 10^{-6} \\ 10^{-6} & 10^{-6} \end{bmatrix} $ $ 10^{-6} = \begin{bmatrix} 10^{-6} & 10^{-6} \\ 10^{-6} & 10^{-6} \end{bmatrix} $ $ 10^{-6} = \begin{bmatrix} 10^{-6} & 10^{-6} \\ 10^{-6} & 10^{-6} \end{bmatrix} $ $ 10^{-6} = \begin{bmatrix} 10^{-6} & 10^{-6} \\ 10^{-6} & 10^{-6} \end{bmatrix} $ $ 10^{-6} = \begin{bmatrix} 10^{-6} & 10^{-6} \\ 10^{-6} & 10^{-6} \end{bmatrix} $ $ 10^{-6} = \begin{bmatrix} 10^{-6} & 10^{-6} \\ 10^{-6} & 10^{-6} \end{bmatrix} $ $ 10^{-6} = \begin{bmatrix} 10^{-6} & 10^{-6} \\ 10^{-6} & 10^{-6} \end{bmatrix} $ $ 10^{-6} = \begin{bmatrix} 10^{-6} & 10^{-6} \\ 10^{-6} & 10^{-6} \end{bmatrix} $ $ 10^{-6} = \begin{bmatrix} 10^{-6} & 10^{-6} \\ 10^{-6} & 10^{-6} \end{bmatrix} $ $ 10^{-6} = \begin{bmatrix} 10^{-6} & 10^{-6} \\ 10^{-6} & 10^{-6} \end{bmatrix} $ $ 10^{-6} = \begin{bmatrix} 10^{-6} & 10^{-6} \\ 10^{-6} & 10^{-6} \end{bmatrix} $ $ 10^{-6} = \begin{bmatrix} 10^{-6} & 10^{-6} \\ 10^{-6} & 10^{-6} \end{bmatrix} $ $ 10^{-6} = \begin{bmatrix} 10^{-6} & 10^{-6} \\ 10^{-6} & 10^{-6} \end{bmatrix} $ $ 10^{-6} = \begin{bmatrix} 10^{-6} & 10^{-6} \\ 10^{-6} & 10^{-6} \end{bmatrix} $ $ 10^{-6} = \begin{bmatrix} 10^{-6} & 10^{-6} \\ 10^{-6} & 10^{-6} \end{bmatrix} $ $ 10^{-6} = \begin{bmatrix} 10^{-6} & 10^{-6} \\ 10^{-6} & 10^{-6} \end{bmatrix} $ $ 10^{-6} = \begin{bmatrix} 10^{-6} & 10^{-6} \\ 10^{-6} & 10^{-6} \end{bmatrix} $ $ 10^{-6} = \begin{bmatrix} 10^{-6} & 10^{-6} \\ 10^{-6} & 10^{-6} \end{bmatrix} $ $ 10^{-6} = \begin{bmatrix} 10^{-6} & 10^{-6} \\ 10^{-6} & 10^{-6} \end{bmatrix} $ $ 10^{-6} = \begin{bmatrix} 10^{-6} & 10^{-6} \\ 10^{-6} & 10^{-6} \end{bmatrix} $ $ 10^{-6} = \begin{bmatrix} 10^{-6} & 10^{-6} \\ 10^{-6} & 10^{-6} \end{bmatrix} $ $ 10^{-6} = \begin{bmatrix} 10^{-6} & 10^{-6} \\ 10^{-6} & 10^{-6} \end{bmatrix} $ $ 10^{-6} = \begin{bmatrix} 10^{-6} & 10^{-6} \\ 10^{-6} & 10^{-6} \end{bmatrix} $ $ 10^{-6} = \begin{bmatrix} 10^{-6} & 10^{-6} \\ 10^{-6} & 10^{-6} \end{bmatrix} $ $ 10^{-6} = \begin{bmatrix} 10^{-6} & 10^{-6} \\ 10^{-6} & 10^{-6} \end{bmatrix} $ $ 10^{-6} = \begin{bmatrix} 10^{-6} & 10^{-6} \\ 10^{-6} & 10^{-6} \end{bmatrix} $ $ 10^{-6} = \begin{bmatrix} 10^{-6} & 10^{-6} \\ 10^{-6} & 10^{-6} \end{bmatrix} $ $ 10^{-6} = \begin{bmatrix} 10^{-6} & 10^{-6} \\ 10^{-6} & 10^{-6} \end{bmatrix} $ $ 10^{-6} = \begin{bmatrix} 10^{-6} & 10^{-6} \\ 10^{-6} & 10^{-6} \end{bmatrix} $ $ 10^{-6} = \begin{bmatrix} 10^{-6} & 10^{-6} \\ 10^{-6} & 10^{-6} \end{bmatrix} $ $ 10^{-6} = \begin{bmatrix} 10^{-6} & 10^{-6} \\ 10^{-6} & 10^{-6} \end{bmatrix} $ $ 10^{-6} = \begin{bmatrix} 10^{-6} & 10^{-6} \\ 10^{-6} & 10^{-6} \end{bmatrix} $ $ 10^{-6} & 10^{-6} \end{bmatrix} $ $ 10^{-6} & 10^{-6}$		3 6 9
$ \begin{array}{c} \hat{G}(f) \\ (\mu m^2) \\ 10^{-5} \end{array} \begin{array}{c} 95 \text{ Per Cent} \\ Confidence \\ Interval \\ 10^{-1} \end{array} \begin{array}{c} \hat{G}(f) \\ (\mu m^2) \\ 10^{-1} \end{array} $	I 95 Per Cent Confidence Interval 2 4 10 ⁵ f (Hz) Fly-milled specimen	
10^{-5}	95 Per Cent I Confidence Interval	

Fig. 1. Spectral estimates and fitted curves of the form of eqn. (1).

point for which the smoothed spectral value is to be calculated, the weighting value α is given by

$$\alpha = \frac{2\pi}{N} \qquad n = 0$$

$$\alpha = \frac{4\pi}{N} \left(1 - \pi \frac{n}{N} \right) \exp(-2\pi n/N) \qquad n = 1, 2, \dots, 2N$$

where N is the cut-off length in terms of the number of spectral points. Two smoothed spectral values are then computed and plotted for each cut-off length. By employing the technique of "window closing", a smoothing filter of the same cut-off length was found to be appropriate for all the specimens tested, for a given traverse length.
3. Spectral estimate models

The results for a number of sample surfaces for data obtained from a trace length of 60 mm and acquired relative to a straight line horizontal datum, adjusted with respect to a mean square line are shown in Fig. 2. From this figure, it can be seen that a common spectral shape is present. Figure 2 adequately demonstrates the effect is not by any means confined or dependent upon the long wavelength content of the estimates. The characteristic is in many cases clear by inspecting wavelengths well over an order of magnitude shorter than the 60 mm traverse length. The fitted curves in Fig. 2, shown as a broken line, are of the form

$$G(f) = \frac{K}{1 + (f/f_c)^2}$$
(2)

where K and f_c are constants for a particular surface specimen. Inherent in the form of this equation is the hypothesis that the surface data are both random and stationary. By computing the best-fitting least-squares curve, of the form of eqn. (2), to the spectral estimates of the surface data, values of K and f_c can be obtained for any sample surface. These two parameters are able to describe the spectral estimates of all but nine of the surface specimens investigated within the 95% confidence interval from a range of 10 μ m - 60 mm. Further investigation with these nine specimens revealed that the cause of the deviation from the form given by eqn. (2) could arise for two possible reasons.



Fig. 2. Spectral estimates and fitted curves of the form of eqn. (2).

(1) The specimen exhibited an error of form owing to an inadvertent change in one or more machining parameters along the length of the specimen during production. Errors of form produced spectral estimates which did not converge to a constant value at low frequencies. An example of the spectral estimate of such a specimen is shown in Fig. 3.

(2) The surface of the specimen exhibited "pick-up marks". These marks probably arise when, because of the high temperatures generated during machining, a piece of material becomes temporarily attached to the cutting tool, thereby increasing the depth of cut. An example of the peak in the spectral estimate resulting from the presence of such a machining error is shown in Fig. 4.

Methods for automatically detecting the presence of such machining errors were implemented and are summarized by the flow diagram Fig. 5. The surface identification program developed gives a warning of the presence of the errors and provides a written assessment of their significance. The comments are based on the values of the various parameters calculated for all machined specimens described. For the surface r.m.s. and surface cut-off wavelength parameters, the nature of comments depends on the grade of membership of the parameters as given in Tables 3 and 4. Comments relating to the magnitude of the machining errors are also provided by the program and these are given in Tables 5 and 6.

To ensure a sound understanding in terms of current engineering practice, the following alternative description of the characterization of surface texture provided by eqn. (2) was developed. An estimate of the variance of the acquired data $\hat{\sigma}^2$ can be obtained from the area under a curve represented by eqn. (2), hence

$$\hat{\sigma}^2 = \hat{K} \int_{f_{\min}}^{f_{\max}} \frac{\mathrm{d}f}{1 + (f/\hat{f}_{\mathrm{c}})^2}$$
(3)

where \hat{K} and \hat{f}_{c} are the estimated values of K and f_{c} respectively and f_{max} and



Fig. 3. Spectral estimate indicating an error of form. Fig. 4. Spectral estimate displaying a machining peak.



Fig. 5. Surface identification.

TABLE 3

180

Comments provided on the grade of membership of the surface r.m.s. value

Surface r.m.s. range (μm)	Comment on the magnitude of the surface undulations	
> 20	Extremely large	
10 - 20	Very large	
5 - 10	Large	
2 - 5	Quite large	
1 - 2	Greater than average	
0.5 - 1.0	Average	
0.2 - 0.5	Less than average	
0.1 - 0.2	Quite small	
0.05 - 0.1	Small	
0.02 - 0.05	Very small	
0.01 - 0.02	Extremely small	
< 0.01	Out of range	

TABLE 4

Comments provided on the grade of membership of the surface cut-off wavelength value

Surface cut-off wavelength range (mm)	Comment on the space of the texture peaks		
>4	Extremely widely spaced		
3 - 4	Very widely spaced		
2 - 3	Widely spaced		
1.5 - 2.0	Quite widely spaced		
1.0 - 1.5	Above average spacing		
0.7 - 1.0	Average spacing		
0.5 - 0.7	Below average spacing		
0.4 - 0.5	Quite closely spaced		
0.3 - 0.4	Closely spaced		
0.2 - 0.3	Very closely spaced		
< 0.02	Extremely closely spaced		

TABLE 5

Comments provided on the grade of membership of the machining peaks

Maximum deviation of the spectral values from the fitted curve relative to the 95% confidence interval	Comment on the pick-up marks	
>4	Badly marked by pick-up	
3 • 4	Very significant pick-up marks	
2 - 3	Significant pick-up marks	
1 • 2	Possible pick-up marks	
<1	(no comment given)	

TABLE 6

Comments provided on the grade of membership of the errors of form

Percentage increase in the estimated r.m.s. value of the data compared with that of a traverse of half the length employed	Comment on the errors of form		
>40	Extremely large errors of form		
30 - 40	Very significant errors of form		
20 - 30	Significant errors of form		
10 - 20	Possible errors of form		
0 - 10	(no comment given)		

 f_{\min} are the maximum and minimum surface frequencies in the spectral estimate. Integrating eqn. (3) yields

$$\hat{\sigma}^2 = \hat{K}\hat{f}_{c} \left\{ \tan^{-1}(f_{\max}/\hat{f}_{c}) - \tan^{-1}(f_{\min}/\hat{f}_{c}) \right\}$$
(4)

enabling the variance of the data to be estimated from the parameters \hat{K} and \hat{f}_c . If $f_{\max} \to \infty$ and $f_{\min} \to 0$, eqn. (4) becomes

$$\hat{\sigma}_0^2 = \frac{\pi K f_c}{2} \tag{5}$$

where $\hat{\sigma}_0^2$ is the total area under a curve of the form of eqn. (2). Dividing eqn. (5) by eqn. (4) yields

$$A = \frac{\pi}{2} \{ \tan^{-1}(f_{\max}/f_c) - \tan^{-1}(f_{\min}/f_c) \}^{-1}$$
(6)

where A is an adjustment factor. By forming the product of the variance of the data with A, an estimate of the total area under a curve of the form of eqn. (2) can be obtained.

4. Discussion

Two parameters, both readily comprehensible in terms of current surface texture analysis, can now be defined. The first parameter is termed the "surface r.m.s. value", denoted by Ψ_s , and is the r.m.s. value of the surface data which would be computed from a trace of infinite length. The value of the parameter is obtained from $\hat{\sigma}(A)^{1/2}$. Such a parameter was chosen since the concept of r.m.s. values is already established in surface texture analysis. The second parameter is termed the "surface cut-off wavelength", denoted by λ_c , and is the wavelength corresponding to the surface frequency f_c . The surface cut-off wavelength is that wavelength at which occurs the only discernible change in the form of the spectral estimates, namely the "half-power point" of the spectral estimates. The parameter is purposely named similarly to the term "cut-off wavelength", whose value

defines the amplitude response characteristic of the standard filters. Since there are no other wavelengths at which an identifiable change occurs in the form of the spectral estimates, λ_c is the value which the standard filter cut-off wavelengths are used to simulate. The machine computation of the surface cut-off wavelength thus produces both a desirable simplification of the data acquisition process and a more consistently accurate method of surface texture measurement. Moreover, in contrast to present analysis techniques, the description of the surface texture provided by the two new parameters is complete.

References

- 1 M. S. Selvam and K. Balakrishnan, The study of machined surface roughness by random analysis, Wear, 41 (1977) 287 293.
- 2 D. J. Whitehouse and M. J. Phillips, Discrete properties of random surfaces, *Philos.* Trans. R. Soc., A290 (1978) 267 - 297.
- 3 N. Patir, A numerical procedure for random generation of rough surfaces, Wear, 47 (1978) 273 296.
- 4 R. C. Spragg and D. J. Whitehouse, A new, unified approach to surface metrology, Proc. Inst. Mech. Eng., London, 185 (47) (1970 - 1971) 697 - 707.
- 5 A. G. M. Hunter and E. A. Smith, Measurement of surface roughness, Wear, 59 (1980) 383 386.
- 6 R. C. Sayles and T. R. Thomas, Surface topography as a non-stationary random process, *Nature*, 271 (1978) 431 434.
- 7 D. J. Whitehouse and M. J. Phillips, The assessment of engineering surfaces using random process analysis, SERC Vacation School, University of Warwick, 1981.
- 8 British Standard 1134, 1972.
- 9 T. R. Thomas and G. Charlton, Variation of roughness parameters on some typical manufactured surfaces, Department of Mechanical Engineering, Teesside Polytechnic.
- 10 D. J. Mulvaney, D. E. Newland and K. F. Gill, Identification of surface roughness, Proc. Inst. Mech. Eng., London, 199 (1985) C3.

PAPER 64

Applications

An Experimental Evaluation of Normalised Fourier Descriptors in the Identification of Simple Engineering Objects

B. Nikravan, R.M. Baul and K.F. Gill

Department of Mechanical Engineering, University of Leeds, Leeds LS2 9JT, U.K.

Proprietary vision systems based on Fourier transforms have already reached the prototype stage and will almost certainly be employed on high-speed production lines for the inspection of components in the near future. In a related area, the development of a robotic system for picking unsorted components from bins can only be achieved with the aid of a fast vision system. This paper examines the potential of a Fourierbased analysis for the fast shape recognition of randomly positioned and orientated components. Two approaches are presented; the first based on the coordinate points of the object boundary and the second on the information contained in the Freeman chain code. Both methods suggest, for the shapes considered, that the recognition process can be achieved reliably with a relatively small number of normalised Fourier descriptors. Each method has a particular advantage depending on the accuracy and the processing speed required.

Keywords: Fourier descriptors. Object recognition.



B. Nikravan was born in 1959. He received a BSc with Honours in Engineering Systems and Control from Huddersfield Polytechnic. He started his research in Digital Image Processing and Computer Vision in the Department of Mechanical Engineering, University of Leeds, in 1983 after completing his postgraduate studies in Electrical and Electronic at the University of Aston in Birmingham. He was awarded the de gree of PhD in 1987. He is currently

employed at Rank Xerox Research (UK) Limited, as a Motion Control consultant. His research interests are high-precision digital control, document and image processing and object recognition.

Elsevier Science Publishers B.V. Computers in Industry 13 (1989) 37-47

0166-3615/89/\$3.50 © 1989 Elsevier Science Publishers B.V.

1. Introduction

The method most frequently used to describe the closed boundary of an object when viewed from a particular angle is either based on the chain code of Freeman [2] or the polygon approximation of Pavlidis [7]. Theoretical and experimental evidence available in the literature. however, indicates that the Fourier Descriptor (FD) is a more powerful way of classifying closed contours [4,8]. Features can be uniquely represented by a normalised FD and this can lead to a significant reduction in object recognition data



R.M. Baul was born in Knaresborough. England in 1935. He obtained his engineering training, as an indentured student apprentice. with Cooke Troughton & Simms Ltd. York. His academic training was gained initially at York Technical College and later at the University of Leeds where he was awarded a BSc Honours Degree in 1960 and PhD in 1966. Dr. Baul was an assistant engineer at P.E.R.A. Melton Mowbray during the period 1960-63 and is currently a Lecturer at

the University of Leeds, teaching Robotics, Design and Computing in the Mechanical Engineering Department.



K.F. Gill obtained his engineering training with a company manufacturing large steam turbines and alternating equipment for the electrical power supply industry. His academic training was gained at the University of Durham and the University of Birmingham. He was awarded the degrees of BSc in 1955. MSc in 1958 and PhD in 1961. From 1961 Dr Gill was a Senior Scientific Officer at the Admirally Weapons Establishment, Portland. For the last 25 years he has been

a Lecturer, then Senior Lecturer at the University of Leeds, teaching control engineering and engineering dynamics to undergraduate students reading for an honours degree in Mechanical Engineering.

Computers in Industry

38 Applications

over the chain code and polygon approximation methods. Furthermore the normalised FD is invariant with respect to translation, rotation and scale of the object and images of similar shape should have identical descriptors.

To demonstrate the benefit of this object classification procedure, existing data records, for a number of objects, have been adopted for computing FD values. Two methods have been considered. In the first the coordinate values of the contour pixels are used to obtain the Fourier transforms and in the second the orientation vectors of the Freeman chain code are employed.

The quality of the contour mapping will be dependent on the lighting system employed. For reference, Fig. 1 presents a typical histogram which shows the contrast between background and object for the image processing carried out in this investigation.

2. Fourier Descriptors (Review of First Method)

The development of slope and curvature codes for use in boundary description has led to the more general concept of the intrinsic equation [3]. The nature of contour data satisfies the mathematical constraints for shape representation by Fourier descriptors since any single-valued periodic function may be expressed as a Fourier transform.

In the first approach, using cartesian coordinates, the two discrete series x(m), y(m), m = 0, 1, ..., L-1, are obtained by image segmentation and edge tracking; and since the boundaries are closed curves x(L) = x(0) and y(L) = y(0).

The analysis involves the derivation of the "descriptors" based on the Fourier series for each of the two series x(m) and y(m) defined as

$$x(m) = \sum_{n=-\infty}^{\infty} X(n) e^{jn\omega_0 m}, \qquad (1)$$

$$y(m) = \sum_{n=-\infty}^{\infty} Y(n) e^{jn\omega_0 m}, \qquad (2)$$

where $\omega_0 = 2\pi/L$ and X(n) and Y(n) are the complex Fourier coefficients, e.g. $X(n) = a_n - jb_n$. The data interval used in the evaluation of the



$$X(n) = \frac{1}{L} \sum_{m=0}^{L-1} x(m) e^{-jn\omega_0 m},$$
 (3)

$$Y(n) = \frac{1}{L} \sum_{m=0}^{L-1} y(m) e^{-jn\omega_0 m},$$
 (4)

where the parameter L is the number of pixels representing the closed contour. It is clear from the analytical derivation that the Fourier coefficients (X(n), Y(n)) contain no information relating to the translation and orientation of the object. Therefore the descriptor defined as

$$R(n) = \left[|X(n)|^{2} + |Y(n)|^{2} \right]^{1/2}$$
(5)

is independent of orientation and this is confirmed by data given in Table 1 for a simple triangular object.

The descriptors defined by (5), although invariant to object position and rotation, are influenced by object size. In order to compensate for this weakness a simple and direct normalisation procedure has been adopted and can be defined as

$$S(n) = R(n)/R(1), \qquad (6)$$

where R(1) is the first descriptor value computed. R(1) is available without further computation, unlike the normalisation techniques described in [5], and computation time is therefore kept to a minimum. Results to support the use of (6) are



Computers in Industry

Table 1 Influence of rotation on Fourier coefficient

$\overline{X_n}$		Y _n		R _n
Real part	Imaginary part	Real part	Imaginary part	
(a) Triang	le			
- 29.36	-0.42	- 2.30	23.15	37.46
- 6.62	-0.32	-1.12	- 5.75	8.85
- 0.60	0.11	- 1.50	0.43	1.67
- 2.56	-0.17	- 1.63	1.41	3.35
-1.27	0.06	-1.31	-0.84	2.01
- 0.64	0.07	-1.48	0.27	1.64
-1.18	-0.09	- 1.54	0.50	2.00
0.77	0.01	- 1.34	-0.21	1.56
-0.58	0.05	- 1.46	0.24	1.59
- 0.86	-0.03	- 1.51	0.28	1.76
- 0.60	0.04	- 1.40	- 0.01	1.52
- 0.57	0.07	- 1.45	0.26	1.58
-0.71	-0.02	- 1.44	0.25	1.62
-0.54	0.09	-1.38	0.08	1.49
- 0.56	0.05	- 1.44	0.24	1.57
(b) Triang	le rotated by 60°			
- 20.33	- 12.73	- 16.31	24.26	37.81
-6.10	2.37	- 3.46	- 5.63	9.29
- 0.92	0.42	- 0.90	- 0.07	1.35
- 1.91	- 0.89	- 2.29	1.55	3.47
- 1.73	0.26	- 1.17	-0.70	2.22
-0.90	0.24	- 0.85	0.02	1.27
- 1.14	-0.27	- 1.47	0.47	1.94
- 1.17	0.07	- 1.01	-0.19	1.56
-0.88	0.20	- 0.90	0.07	1.28
- 0.93	- 0.09	-1.24	0.26	1.58
- 0.99	0.10	- 0.94	- 0.01	1.37
-0.90	0.19	-0.88	0.10	1.28
0.89	0.00	- 1.13	0.17	1.45
- 0.91	0.10	-0.95	0.08	1.32
- 0.88	0.16	-0.88	0.15	1.26

 $R_n = (|X_n|^2 + |Y_n|^2)^{1/2}.$

given in Table 2 for triangular and L-shaped objects.

3. Reconstruction of Contours Using Fourier Coefficients

A sufficient condition for the existence of a set of Fourier descriptors that unambiguously represents an object is that the object contour can be reconstructed by using the Fourier inverse formulae. If the set of FDs is not unique, a reconstruction of the contour could not take place.

It is not obvious from Table 1 and Table 2, however, just how many Fourier coefficients would

be necessary to enable an object to be identified with a high degree of confidence. In an attempt to clarify this last point, equations (1) and (2), employing a finite number of coefficients, are used to create varying approximations to the original which can be seen in the reconstruction of the contour's data. Although these expressions do not provide enough information to allow a continuous series to be obtained, all the discrete values of x(m) and y(m) could be regained exactly if all the Fourier coefficients were employed.

Figure 2 illustrates the reconstruction of a "triangle" for different numbers of terms in the expressions (1) and (2) using the first method. These figures suggest that no sensible improvement will result if more than two coefficients are used for this shape. A subjective assessment of Table 1 would support this because of the dominance of the first two terms over the remainder listed. The results demonstrate very clearly the valuable compression of the boundary data that can be expected and illustrates the power of the descriptor as an aid to simple shape recognition. In this example six hundred pairs of contour coordinate values have been compressed to two descriptor values.

4. Similarity Measurement and Recognition Algorithm

The normalised Fourier descriptor vector S for a particular boundary shape can be expected to change marginally as a result of scaling and rotation, and to accommodate this likely variation an



Fig. 2. Reconstruction of triangle using the first method.

40 Applications

 Table 2

 Effect of scaling or contour dilation on Fourier coefficients

Triangle Tria		Triangle	Triangle rotated		L-shaped		Half-sized L		ized L
$\overline{R_n}$	S _n	R _n	S _n	$\overline{R_n}$		$\overline{R_n}$	S _n	$\overline{R_n}$	
37.46	1.00	37.81	1.00	55.06	1.00	28.22	1.00	15.42	1.00
8.85	0.24	9.29	0.25	15.60	0.28	7.92	0.28	4.30	0.28
1.67	0.04	1.35	0.04	7.46	0.14	4.30	0.15	3.28	0.21
3.35	0.09	3.47	0.09	5.41	0.10	2.79	0.10	2.16	0.14
2.01	0.05	2.22	0.06	2.20	0.04	1.81	0.06	2.16	0.14
1.64	0.04	1.27	0.03	0.36	0.01	0.86	0.03	1.66	0.11
2.00	0.05	1.94	0.05	1.63	0.03	1.42	0.05	1.90	0.12
1.56	0.04	1.56	0.04	0.68	0.01	0.89	0.03	1.62	0.11
1.59	0.04	1.28	0.03	1.02	0.02	1.16	0.04	1.74	0.11
1.76	0.05	1.58	0.04	0.99	0.02	0.97	0.03	1.58	0.10
1.52	0.04	1.37	0.04	0.70	0.01	1.04	0.04	1.63	0.11
1.58	0.04	1.28	0.03	0.37	0.01	0.84	0.03	1.50	0.10
1.62	0.04	1.45	0.04	0.60	0.01	0.98	0.03	1.54	0.10
1.49	0.04	1.32	0.03	0.34	0.01	0.82	0.03	1.43	0.09
1.57	0.04	1.26	0.03	0.52	0.01	0.92	0.03	1.44	0.09

improved reference feature vector can be obtained by averaging the S descriptor vector, that is

$$F_{\rm r} = \frac{1}{N} \sum_{i=1}^{N} S_i,$$
 (7)

where F_r is the reference feature vector whose elements are the normalised Fourier Descriptors averaged over N nominally identical objects having different sizes and orientations. Identifying the vector of the object to be inspected as S_t , the similarity between the two independent vectors, F_r and S_t , can be estimated from the error

$$E = \frac{1}{k} \left[\sum_{i=1}^{k} \left(F_{ri} - S_{ti} \right)^2 \right]^{1/2},$$
 (8)

where k is the number of descriptors in each vector. In an environment where more than one shape is to be tested for recognition, the parameter E must be evaluated for all the test vectors, the minimum value found yielding the recognised shape.

In the test study conducted, 10 descriptors in each feature vector were adopted and the results are shown in Fig. 3. In Fig. 3(a) a reference triangle is compared with other simple shapes listed in the key and the minimum value of Eclearly indicates the presence of a triangle in the group considered. In Fig. 3(b) the outcome of a trial with an L-shaped contour shows it to be a member of a family of L-shaped objects labeled 1 and 2 in the key. The corresponding FDs are listed in Table 2 in the raw and normalised form.



Fig. 3. Object recognition.

Computers in Industry

5. Fourier Descriptors (Review of Second Method)

To take into account all the information stored in a Freeman chain code contour representation, allowance is made, in the second approach, for the perimeter length between contour pixels, a method first used by Kuhl et al. [6].

Consider the chain code C with k elements:

$$C = a_1 a_2 a_3 a_4 \cdots a_k,$$

where each element a_i is an integer number between 0 and 7. The direction of the vector labelled a_i is given by:

$$\angle(a_i)=(\pi/4)a_i.$$

If the perimeter length of a contour is t, the length of each code vector, i.e. link, Δt_i is:

$$\Delta t_i = 1, \quad \text{if } a_i \text{ is an even number,} \\ \Delta t_i = \sqrt{2}, \quad \text{if } a_i \text{ is an odd number.}$$
(9)

From the definition of the chain code, the "time" required to traverse the first q links at "constant unit speed" as proposed in [5] is given by

$$\Delta t_q = \sum_{i=1}^{q} \Delta t_i \tag{10}$$

and the period of the chain code is defined as t_k . The changes in the x, y coordinate values as the chain elements a_i are traversed are:

$$\Delta x_i = \text{Sgn}(6 - a_i) \text{ Sgn}(2 - a_i)$$

$$\Delta y_i = \text{Sgn}(4 - a_i) \text{ Sgn}(a_i),$$

where

Sgn(Z) =
$$\begin{cases} 1, & Z > 0, \\ 0, & Z = 0, \\ -1, & Z < 0. \end{cases}$$

If the first chain code element is positioned at the origin of the x - y axes, the projections on the x and y axis of the first q links are

$$x_q = \sum_{\substack{i=1\\q}}^{q} \Delta x_i, \tag{11}$$

$$y_q = \sum_{i=1}^{7} \Delta y_i, \qquad (12)$$

respectively.

Adopting the Fourier representation for a discrete series, the derivative of x(t) is given by:

$$\dot{x}(t) = \sum_{n=1}^{\infty} \left(\alpha_n \cos \frac{2\pi n}{t_k} t + \beta_n \sin \frac{2\pi n}{t_k} t \right), \quad (13)$$

where

$$\alpha_n = \frac{2}{t_k} \int_0^{t_k} \dot{x}(t) \cos \frac{2\pi n}{t_k} t \, \mathrm{d}t,$$
$$\beta_n = \frac{2}{t_k} \int_0^{t_k} \dot{x}(t) \sin \frac{2\pi n}{t_k} t \, \mathrm{d}t.$$

Then for the interval $t_{q-1} \leq t \leq t_q$

$$\alpha_n = \frac{2}{t_k} \sum_{q=1}^k \frac{\Delta x_q}{\Delta t_q} \int_{t_{q-1}}^{t_q} \cos \frac{2\pi n}{t_k} t \, dt$$
$$= \frac{-1}{\pi n} \sum_{q=1}^k \frac{\Delta x_q}{\Delta t_q} \left(\sin \frac{2\pi n}{t_k} t_q - \sin \frac{2\pi n}{t_k} t_{q-1} \right)$$

B. Nikravan et al. / Normalised Fourier Descriptors

and similarly

$$\beta_n = \frac{1}{\pi n} \sum_{q=1}^k \frac{\Delta x_q}{\Delta t_q} \left(\cos \frac{2\pi n}{t_k} t_q - \cos \frac{2\pi n}{t_k} t_{q-1} \right).$$

The time derivative of the Fourier series for x(t) can (by definition) be written

$$\dot{x}(t) = \frac{2\pi n}{t_k} \sum_{n=1}^{\infty} \left(a_n \sin \frac{2\pi n}{t_k} t - b_n \cos \frac{2\pi n}{t_k} t \right).$$
(14)

Equating like terms in (13) and (14) yields the terms of the complex Fourier coefficients a_n and b_n

$$\beta_n = \frac{2\pi n}{t_k} a_n,$$



Fig. 4. Reconstruction of triangle using the second method.



Fig. 5. Reconstruction of wrench.

and on rearrangement this gives

$$a_{n} = \frac{t_{k}}{2\pi^{2}n^{2}} \sum_{q=1}^{k} \frac{\Delta x_{q}}{\Delta t_{q}}$$
$$\times \left(\cos \frac{2\pi n}{t_{k}} t_{q} - \cos \frac{2\pi n}{t_{k}} t_{q-1} \right).$$
(15)

Similarly,

$$b_n = \frac{t_k}{2\pi^2 n^2} \sum_{q=1}^k \frac{\Delta x_q}{\Delta t_q} \times \left(\sin \frac{2\pi n}{t_k} t_q - \sin \frac{2\pi n}{t_k} t_{q-1} \right).$$
(16)

Expressions having the same form can be obtained for the y(t) series.

Applying this second method to a "triangle' and a "wrench" generated the results shown in Figs. 4 and 5, respectively. These results clearly demonstrate that a progressive improvement in "fit" can be achieved with an increasing number of descriptors. The degree of similarity between objects, required to be identified or inspected, will dictate the number of Fourier descriptors to be employed. The associated descriptor values are given in Table 3 for information.

6. Error Approximation as a Function of the Number of Fourier Coefficients Employed

To obtain an indication of the sub-optimum number of descriptors necessary to ensure object recognition, an error function has been developed. Let

$$\hat{x}_{N}(t) = a_{0} + \sum_{n=1}^{N} a_{n} \cos \frac{2n\pi}{T} t + b_{n} \sin \frac{2n\pi}{T} t,$$
$$\hat{y}_{N}(t) = c_{0} + \sum_{n=1}^{N} c_{n} \cos \frac{2n\pi}{T} t + d_{n} \sin \frac{2n\pi}{T} t$$

be the Fourier series truncated after N terms for the x(t) and y(t) series and let the actual error ΔE_a be defined as:

$$\Delta E_{a} = \max_{t=1}^{k} \left[|x(t) - \hat{x}_{N}(t)|, |y(t) - \hat{y}_{N}(t)| \right],$$
(17)

Computers in Industry

Table 3 A list of 30 Fourier descriptors for the triangle and wrench raw and normalised versions

	Triangle		Wrench		
n	$\overline{R_n}$		R _n	S _n	
1	73.562	1	99.407	1	
2	16.528	0.225	12.643	0.127	
3	2.696	0.037	15.536	0.156	
4	4.648	0.063	6.603	0.066	
5	2.023	0.028	2.559	0.026	
6	1.162	0.016	1.549	0.016	
7	1.414	0.020	2.254	0.023	
8	0.651	0.009	0.810	0.008	
9	0.696	0.009	1.854	0.019	
10	0.643	0.009	1.163	0.018	
11	0.226	0.003	0.302	0.003	
12	0.528	0.007	0.768	0.008	
13	0.283	0.004	0.337	0.003	
14	0.096	0.001	0.056	0.000	
15	0.387	0.005	0.342	0.003	
16	0.142	0.002	0.478	0.005	
17	0.062	0.001	0.437	0.004	
18	0.240	0.003	0.082	0.000	
19	0.118	0.002	0.112	0.001	
20	0.073	0.001	0.290	0.003	
21	0.161	0.002	0.297	0.003	
22	0.062	0.001	0.164	0.002	
23	0.115	0.002	0.142	0.001	
24	0.085	0.001	0.109	0.001	
25	0.055	0.001	0.160	0.002	
26	0.093	0.001	0.155	0.002	
27	0.036	0.000	0.220	0.002	
28	0.059	0.001	0.047	0.000	
29	0.060	0.001	0.084	0.000	
30	0.061	0.001	0.172	0.002	

where k is the number of chain code elements available.

To predict the error directly from the chain code data it has been shown by Kuhl and Giardina [5] that ΔE can be approximated by the expression:

$$\Delta E_{p} \leq \frac{T}{2\pi^{2}N} \max \left[V_{0}^{\mathsf{T}}(\dot{x}(t)), V_{0}^{\mathsf{T}}(\dot{y}(t)) \right], \quad (18)$$

where the total variation of the "time" derivative $\dot{x}(t)$ has been symbolised as $V_0^{\mathsf{T}}(\dot{x}(t))$ and of the derivative $\dot{y}(t)$ as $V_0^{\mathsf{T}}(\dot{y}(t))$. In this context the time period T is equal to t_k . These derivative values are estimated from:

$$\dot{x}_i = \frac{\Delta x_i}{\Delta t_i}, \qquad \dot{y}_i = \frac{\Delta y_i}{\Delta t_i}$$

for the Freeman code element a_i .

ESTIMATED ERROR ∆Ep (2™ METHOD OF ANALYSIS)

B. Nikravan et al. / Normalised Fourier Descriptors

30

25

20



Fig. 6 Actual and estimated error for the reconstruction of the triangle. (a) Second method. (b) First method.



Fig. 7. Actual and estimated error curves for the reconstruction of the wrench.

73

The total variations of $\dot{x}(t)$ and $\dot{y}(t)$ are

$$\begin{aligned} V_0^{\mathsf{T}}(\dot{x}(t)) &= \left(\sum_{2}^{k} |\dot{x}_i - \dot{x}_{i-1}|\right) + |\dot{x}_k - \dot{x}_1|, \\ V_0^{\mathsf{T}}(\dot{y}(t)) &= \left(\sum_{2}^{k} |\dot{y}_i - \dot{y}_{i-1}|\right) + |\dot{y}_k - \dot{y}_1|, \end{aligned}$$

respectively. The actual error ΔE_a , for the triangle, evaluated using (17), is shown in Fig. 6(a). The predictor error, ΔE_p estimated from (18), is superimposed on the same plot and for completeness a similar error estimation based on the first method has been made for the results of Section 2, and this is shown in Fig. 6(b).

An error curve has also been generated for the wrench, Fig. 7, and both figures indicate that the primary shape can be recognised by 10 descriptors with a high degree of confidence.

7. Engineering Applications of FD Recognition

The principal characteristics which need to be identified in a reliable application of an FD based method are:

(1) The FD vector derived from object image spatial data must adequately reconstruct the object when used in the inverse formulae.

- (2) The hyperbolic shaped error curves, typically Fig. 6, must fall rapidly in the interval 0-10 descriptors for high-speed recognition.
- (3) The simple normalisation procedure presented in this paper, which has the advantage of speed, must be shown to adequately compensate for object size.
- (4) The error measure, obtained by means of (8), must exhibit good selectivity i.e. the standard deviation of the reference feature vector must be small compared to its mean value.

To highlight the above points further tests were undertaken for two typical engineering components, photographs of which are given in Fig. 8.

The reconstructed contours of the 4- and 7bladed fans shown in Figs. 9 and 10 for both methods considered, demonstrate the creditability of the FD vector as a means of object identification. The quality of the reconstruction can be seen from the comparison with the original contour, shown dotted, and this gives an indication of the necessary size of the FD vector to be employed in the recognition process.

The error curves shown in Fig. 11 for the fan blades fall rapidly in the 0-10 interval and consequently a small finite number of FDs would be required and these would give fast recognition of the component. The error measures listed in the



Fig. 9. Reconstruction of the 7-bladed fan using the first method.

upper part of Table 4 give likely variations due to orientation and size of the fan blades and are relatively small. The lower half of Table 4 shows the degree of selectivity that can be attained by the use of (8), indicating the potential of the method for component inspection, missing blades or part blades, and for component sorting for use in robotic handing.

8. Discussion and Conclusions

The results demonstrate the value of the normalised Fourier descriptor as a method for classifying objects by virtue of their silhouette. It has been demonstrated that extensive contour data can be condensed to a relatively small number of numerical values which are virtually independent



Fig. 10. Reconstruction of the 4- and 7-bladed fans using the second method.

46 Applications

Both simple 2-D shapes and more complex engineering components have been studied and in the latter case the silhouette has been shown to be adequate for inspection and identification purposes.

In the derivation of the Fourier descriptors two alternative approaches have been taken and comparisons have been made between the two methods based on the reconstructed contours. The first method based on the coordinates of the object boundary gives a rapid test approximation to the object shape but does not possess the convergence and higher accuracy of reconstruction exhibited by the second method where a larger number of Fourier descriptors need to be employed. The second method based on the Freeman chain code can, therefore, be recommended particularly for the classification and recognition of more complex shapes, although computational times will be higher.

In order to judge the degree of similarity between objects for inspection and sorting operations the concept of a reference feature vector has been established. The lack of similarity is

Table 4

Similarity measured for the two engineering objects, i.e. the seven bladed and four bladed fans, for different sizes and positions with parts being masked (invisible to camera)

First experiment		Second experiment		
Nature of comparison	Error measure ^a	Nature of comparison	Error measure ^a	
Fan with 7 blades compared to its rotated version	0.7	Fan with 4 blades compared to its scaled version	0.2	
Fan with 7 blades compared to its rotated version with a different size	0.8	Fan with 4 blades compared to its rotated version with a different size	0.3	
Fan with 7 blades compared to its scaled version, i.e. much smaller version	0.8	Fan with 4 blades compared to its rotated version	0.3	
2/3 of two of the 7-bladed fan masked and compared with a version where 2 of the blades were completely masked	1.0	1/3 of two of the four blades masked out and compared to the original 4 bladed fan	0.6	
1/3 of two of the blades of fan with 7 blades masked and compared to its original version	1.1	2/3 of 2 of the 4-bladed fan masked and compared to a version where $1/3$ of 2 of the blades were masked	0.6	
1/3 of two of the 7 blades masked and compared to a version where $2/3$ of 2 of the 7 blades were masked.	1.1	2/3 of 2 of the 4 blades masked and compared to a version where 2 of the 4 blades were completely masked	0.7	
1/3 of 2 of the 7 blades masked compared to a version where 2 blades were completely masked	1.9	2/3 of 2 of the 4 blades masked and compared to the original 4-bladed fan	1.1	
Original 7 bladed fan compared to a version where $2/3$ of 2 of the blades were masked	2.0	1/3 of 2 of the 4 blades masked and compared to a version where 2 blades were completely masked	1.2	
Original fan with 7 blades compared to a version where 2 blades were completely masked	2.6	Original 4-bladed fan compared to a version where 2 of its blades completely masked	1.6	
Fan with 4 blades compared to 7 bladed fan when two of the 7 blades were masked	3.7	Original 4-bladed fan compared to a version of 7-bladed fan where 4 of its blades are 1/3 masked	3.6	
7-bladed fan compared to the 4 bladed fan	3.4	4-bladed fan compared to the 7-bladed fan	3.4	

^aError measure as calculated by (8).

Computers in Industry



Fig. 11. Actual and estimated error curves. (a) 4-bladed fan. (b) 7-bladed fan.

determined by an error function based on the differences in the normalised Fourier descriptor vectors for the reference and test object. This has been successfully employed in the identification of a number of different geometrical shapes.

To predict in advance the sub-optimum number of Fourier descriptors necessary for reliable recognition and identification, a theoretical error function based on the truncated Fourier series for

B. Nikravan et al. / Normalised Fourier Descriptors 47

the boundary data has been established and tested for a range of objects. The error magnitude has been shown to decrease rapidly with increasing number of Fourier descriptors before finally levelling out. This indicates the possibility of establishing an optimum gradient for automatically assessing the required size of the Fourier descriptor vector. The CPU time, when using the fast Fourier transform algorithm given in [1], for the evaluation of the 50 descriptors for the 4- and 7-bladed fans was 10 and 15 ms respectively on a DEC (VAX 8600) computer. These are well within the normal manufacturing and assembly time scales for most industrial applications.

References

- J.W. Cooley and J.W. Tukey, "An algorithm for the machine calculation of complex Fourier series", *Math. Comput.*, Vol. 19, No. 90, April 1965, pp. 297-301.
- [2] H. Freeman, "Boundary encoding and processing" in: Lipkin and Rosenfeld, eds., *Picture Processing and Psychopictorics*, Academic Press, New York, 1970, pp. 241-306.
- [3] H. Freeman, "Computer processing of line-drawings images", Comput. Surv., Vol. 6, 1974, pp. 57-97.
- [4] G.H. Granlund, "Fourier preprocessing for hand print character recognition", *IEEE Trans. Comput.*, Vol. C-21, 1972, pp. 195-201.
- [5] F.P. Kuhl and C.R. Giardina, "Elliptic Fourier features of a closed contour", *Comput. Graphics Image Process.*, Vol. 18, 1982, pp. 236-258.
- [6] F.P. Kuhl, C.R. Giardina, M.J. Webster and D. O'Connor. "Fourier Series approximation of chain-encoded contours". *Proc. Electro-Optics / Laser 80 Conf. & Expos.* Boston, MA, November 19-21, 1980.
- [7] T. Pavlidis and F. Ali, "Computer recognition of handwritten numerals by polygonal approximations", *IEEE Trans Syst. Man Cybernet.*, Vol. SMC-6, 1975, pp. 610-614.
- [8] C.T. Zahn and R.Z. Roskies, "Fourier descriptors for plane closed curves", *IEEE Trans. Comput.*, Vol. C-21, 1972. pp. 269-281.

PAPER 65

.

Experimental evaluation of 'shape from shading' for engineering component profile measurement

G A H Al-Kindi, BSc, R M Baul, BSc, PhD, CEng, MIMechE, and K F Gill, BSc, MSc, PhD, CEng, MIMechE Department of Mechanical Engineering, University of Leeds

This paper examines the application of the 'shape from shading technique' for three-dimensional surface geometry measurements and object inspection. The results for diffuse, specular and combined models are presented for a range of components exhibiting flat to highly curved surfaces. A more representative relationship between surface radiance and image intensity has been developed which, when incorporated in the models, is shown to change the accuracy of profile measurement.

1 INTRODUCTION

In object recognition it has been established (1) that a minimum number of data points are required to achieve an adequate surface description. In the production of three-dimensional computer images, two principal calculations are involved. The first and the most frequently discussed in the literature is the determination of visible attributes, their location and the gradient of the normal vector at each location. The second is to predict the image light intensity at a point (2, 3) from the gradient of the normal vector and the position and intensity of the light source.

In the earliest work on image processing (4-9), a reflectance map was developed and the image intensity gave a measure of the surface contouring. Strat (10) and Smith (11) improved on the reflectance map concept by introducing light reflection models and developed an image intensity equation to enable surface contouring to be defined more readily. Phong (12), Cook and Torrance (13), Brassel (14), Blinn (2) and Woodham (15) proposed alternative image intensity models to improve contour estimation and extended the work of these earlier authors.

By using analytical expressions, symbolic models of the image features can be established and shape from shading is one such analytical approach. Application of this allows a representation of shape suitable for surface contour measurement and object recognition (10).

2 SURFACE SHADING

The shading of a surface point depends on the reflection characteristics, the local geometry and the lighting conditions (16). The surface geometry is identified by measurement of a normal vector at a point of intersection of the incident light (17). This type of representation for shape is favoured for machine vision (18) because surface normals undergo simple transformation with rotation while distance measurements from viewer to surface point changes in a less symmetrical fashion (19).

In the treatment of surface reflection (16), three light components can be identified, which are ambient, diffuse and specular. The ambient component represents light assumed uniformly incident from the environment and is reflected equally in all directions by the surface.

The MS was received on 28 November 1988 and was accepted for publication on 18 August 1989.

B04488 C IMechE 1989

The diffuse and specular components are associated with illumination from a specific light source (13). The model used to describe the diffuse component is based on Lambert's law. This law implies that a surface will diffuse incident light proportional to the cosine of the angle between the normal vector at the surface and the vector to the light source (θ) (Fig. 1). For values of $-\pi/2 > \theta > \pi/2$ the light source will not illuminate the surface and the value of the radiance must be zero (2). The Lambertian model is

$$I = \begin{cases} \cos \theta, & -\pi/2 < \theta < \pi/2 \\ 0, & \text{otherwise} \end{cases}$$

where I is the radiance of the object surface at each point.

The concept of specular light component introduced by Phong (12) proposes that the angles of the incident and reflected light are the same but on opposite sides of the normal vector, and unlike the ideal Lambertian model the radiance of a point on a specular surface is dependent on the viewing angle. If the surface was a perfect reflector, e.g. a mirror, light would only reach the eye if the surface normal vector was mid-way between the light source vector and the viewing vector. For a less perfect surface the radiance would fall as the angle ϕ between the viewing vector and reflection vector increases. The relationship (16) incorporating this feature is the radiance model (Fig. 1):

$$I = \begin{cases} \cos^{m} \phi, & -\pi/2 < \phi < \pi/2 \\ 0, & \text{otherwise} \end{cases}$$





0954-4054/89 \$2.00 + .05

Proc Instn Mech Engrs Vol 203

4 IMPROVED RADIANCE ESTIMATION

Using the object-lens-image arrangement shown in Fig. 3, the apparent area of the image patch (δa_i) as seen from the centre of the lens is $\delta a_i \cos \varepsilon$ and its distance is $g/\cos \varepsilon$ from this centre. The solid angle (21) subtended by the patch from the centre of the lens is

$$\Omega = \frac{\delta a_i \cos \varepsilon}{(g/\cos \varepsilon)^2} \tag{3}$$

Similarly, the solid angle of the object patch (δa_0) is

$$\Omega = \frac{\delta a_{\rm o} \cos \tau}{\left(h/\cos \varepsilon\right)^2} \tag{4}$$

and since these two angles are equal, equating equations (3) and (4) yields

$$\frac{\delta a_{o}}{\delta a_{i}} = \frac{\cos \varepsilon}{\cos \tau} \left(\frac{h}{g}\right)^{2} \tag{5}$$

The power of the light originating on the object patch and passing through the lens is

$$\delta p = I \delta a_0 \Omega \cos \tau$$

where I is the radiance of the surface in the direction towards the lens. This power will be concentrated in the image, assuming losses in the lens are negligible; hence if no light from other areas reaches this image patch, then

$$E = \frac{\delta P}{\delta a_{i}} = I \frac{\delta a_{o}}{\delta a_{i}} \frac{\pi}{4} \left(\frac{d}{h}\right)^{2} \cos^{2} \varepsilon \cos \pi$$

Substituting equation (5) gives the intensity of the image patch (22):

$$E = \frac{\pi}{4} I \left(\frac{d}{g}\right)^2 \cos^4 \varepsilon \tag{6}$$

where d is the lens diameter.

If the area of the image patch is chosen to be that of one pixel in the image plane, equation (6) becomes

$$E = \frac{\pi}{4} I \left(\frac{d}{g}\right)^2 \cos^4\left(\tan^{-1}\frac{r}{g}\right)$$
(7)



(1)

(2)

Fig. 3 The relationship between the image radiance and the object radiance

Part B: Journal of Engineering Manufacture

@ IMechE 1989

nearer to unity.

 $\beta = \alpha \pm \cos^{-1} I$

since by definition $\beta = \frac{1}{2}(\psi + \alpha)$

the radiance I.

and the specular model gives

 $\beta = \frac{1}{2}(\alpha + \gamma \pm \cos^{-1} I^{1/m})$



(horizontal datum for shape measurement)

where the coefficient m is a measure of the shininess of the surface. A bright surface would correspond to a

large value of m and a dull surface would exhibit values

3 SHAPE MEASUREMENT

To estimate shape, the inclination of surface normals

must be evaluated from surface radiance. If the light

source angle is α and the surface normal is inclined at

Earlier work (2, 16, 20, 13) assumes the over-

simplification that the intensity (E) of an image is

directly proportional to the radiance (1) of the object.

Improved shape estimations should be possible if a more accurate relationship is developed for evaluating

an angle β (Fig. 2) the Lambertian model gives

213

EVALUATION OF 'SHAPE FROM SHADING' FOR ENGINEERING COMPONENT PROFILE MEASUREMENT

where r is the distance of each image pixel from the centre of the image plane.

For a given physical arrangement, the diameter (d) of the lens and the distance from the lens to the image plane (a) are both known and of fixed value. Therefore, equation (7) indicates that the numerical value of the image intensity (E) at a pixel is not directly proportional to object radiance (1) but is influenced by the distance (r) to the pixel measured from the optical axis through the image plane.

5 MONITORING EQUIPMENT

All experimental results were obtained using a monochrome camera, Link 109A, and a Matrox QFG01 frame grabber. The image obtained in each case is a 256×256 pixel matrix with each pixel having 256 possible grey level values.

A DEC (LSI 11/23 (16 bit)) microcomputer was used for data collection and a DEC (VAX 8600) computer for image data processing.

The light source and camera position, angles α and γ , were chosen to achieve a good-quality image by avoiding unnecessary shadows and to simplify the algebra of the 'combined model' given below. The diagrammatic arrangement is shown, for reference purposes, in Fig. 4.

6 COMBINED MODEL

On the assumption that surfaces will not, in general, behave in a pure diffuse or specular manner the models



(a) Camera and lighting arrangement



Fig. 4

@ IMechE 1989

previously given were combined to give

$$I = a \cos (\alpha - \beta) + b \cos^{m} (2\beta - \alpha - \gamma) + c \qquad (8)$$

The value of m = 1 is adopted because the objects used in the trials can be sensibly classified as dull. Therefore, for the arrangement shown in Fig. 4, equation (8) reduces to

$$I = a \sin \beta + 2b \sin (\beta - b + c)$$

If both α and γ are chosen to be 90°, this equation can be rearranged to yield the surface normal

$$\beta = \sin^{-1} \left[-a \pm 0.25b \sqrt{\{a^2 + 8b(I+b-c)\}} \right]$$
(9)

where a + b + c = 1 for the general case (4).

7 RESULTS

To test the performance of the three models described, several objects were selected so as to give a wide range of surface curvature. These objects are shown in Fig. 5 and consist of a flat surface, two turbine compressor blades of varying curvature and a beaker. The components were sprayed with matt white paint to eliminate errors due to the discoloration present on the surfaces of the objects.

Results are presented for the models represented by equations (1), (2) and (9) and for the same models incorporating the radiance estimation of equation (7). For the specular model, the small variation in the angle γ (Fig. 2) resulting from the pixel distribution within the image is ignored.



Proc Instn Mech Engrs Vol 203



Fig. 6 Model results for compressor blades

Typical compressor blade results are given in Fig. 6 including, for comparison, the profiles obtained by a well-proven stylus instrument at the sections under examination. To facilitate comparison the surface



Part B: Journal of Engineering Manufacture

normal estimates have been mapped to surface profiles to give an improved visual indication of the relative accuracy achieved.

The diffuse model incorporating equation (7) yields the most accurate surface profiles and a significant reduction in accuracy is shown in the case of the specular model, when the value of m equals 1. Clearly the value of m will influence the results obtained from the specular model and the magnitude of the change to be expected is illustrated in Fig. 7. It is evident from these results that by selecting a value of m less than unity the accuracy of the specular model can also be improved when equation (7) is applied.

In the measurement results presented, the most appropriate values of a and b were determined by trial and observation from all the data available. The values of (a + b) obtained experimentally from trials on the diffuse and specular models were found to be sensibly unity; hence the value of c was set to zero in the combined model trials. As would be expected the results from the combined model are good in all cases since the dominance of either the diffuse or specular model can be charged by adjusting the ratio of a and b values.

The most severe test of the models is the measurement of the cylindrical beaker because of the large changes in surface normal values. The results displayed in Fig. 8 favour the diffuse model used in conjunction with equation (7), provided the value of m in the specular model is retained at unity. These are in agreement with the results from the compressor blade trials.

To obtain some idea of how surface texture and colour influences model performance, trials were conducted on three different flat surfaces. The results (Fig. 9) follow the same pattern as before with one exception,

@ IMechE 1989



namely that the specular model (m = 1) now gives the superior results. The results from the diffuse model without the advantage of the radiance correction [equation (7)] were abysmal and no clear explanation for this could be found other than surface texture effects. Profilometer measurements of the flat surfaces are provided in Fig. 10 for comparison.

8 CONCLUSIONS

In this paper three mathematical models for shape representation using data from two-dimensional images have been used to measure surface profiles of varying curvature. An accurate datum for comparison of the experimental results was provided by conventional stylus measurement and these reference profiles are referred to as 'actual' profiles.

The results obtained indicate that the accuracy of the

C IMechE 1989



EVALUATION OF 'SHAPE FROM SHADING' FOR ENGINEERING COMPONENT PROFILE MEASUREMENT

Proc Instn Mech Engrs Vol 203

profiles obtained by 'shape from shading' is dependent on:

(a) the model chosen.

(b) the radiance reflection law adopted,

(c) a range of combinations of the above (a and b) and

(d) surface texture and colour variations,

but would appear to be independent of surface curvature.

In any practical engineering application, therefore, preliminary results from trials and observation on actual components would need to be considered in order to establish the relative importance of the factors listed.

Provided the combinations of model and parameter values are optimized the technique of 'shape from shading' would adequately distinguish components of varying curvature, and the accuracy of the results suggest that components could be detected reliably as part of an automatic sorting facility. The compressor blade cross-section shown in Fig. 11 provides a suitable illustration where the shaded area represents a small percentage of the actual cross-sectional area, for example profiles (b), (d) and (e). Such data would also allow major imperfections to be identified automatically in engineering component manufacture.

Tests carried out on nominally flat surfaces provide some initial information on texture and colour, and



Part B: Journal of Engineering Manufacture

clearly indicate the need for a more specific investigation in this area. Again stylus measurements of the surface profiles have been included for reference purposes and for a visual indication of the accuracies achieved with varying texture and colour.

82

In this case both the diffuse and specular model surface normals show improvements in accuracy when the correction [equation (7)] is applied and the normals of the smooth surface are positioned, as would be expected, closest to the ideal 90° normals.

The uniformly coloured surface of the wooden block compared to the natural surfaces gives the better result, implying that in any application random discoloration of the surface could adversely affect the results.

REFERENCES

- 1 McPherson, C. A., Tio, J. B. K., Hill, E. L. and Sadjadi, F. A. Curved surface representation for image recognition. Proceedings of Conference on Pattern recognition and image processing, June 1982, pp. 370-378.
- 2 Blinn, J. F. Models of light reflection for computer synthesized pictures. Computer Graphics, 1977, 11(2).
- Brady, J. M. Computer vision, 1981, pp. 117-139 (North-Holland Elsevier Science Publishers).
- Horn, B. K. P. Shape from shading-a method of obtaining the shape of a smooth opaque object from one view. MAC internal report TR-70 and MIT AI Lab technical report 232, 1970.
- Wiston, P. H. Obtaining shape from shading information. In Psy-chology of computer vision, 1975, pp. 115-155 (McGraw-Hill, New 5 York)
- 6 Horn, B. K. P. Hill shading and reflectance map. Proc. IEEE, January 1981, 69(1) pp. 414-417.
- Abort, B. K. P. and Sjoberg, R. W. Calculating the reflectance map. Appl. Optics, June 1979, 18(11), 1770–1779.
- Ikeuchi, K. and Horn, B. K. P. Numerical shape from shading and occluding boundaries. In Artificial intelligence, Vol. 17, 1981, pp. 141-185 (North-Holland, Amsterdam).
- Horn, B. K. P. and Brooks, M. J. The variational approach to shape from shading. MIT A.I. Lab. memo 813, March 1985.
- 10 Strat, T. M. A numerical method for shape from shading. MS thesis, 1979, MIT, Cambridge, Mass.
- Smith, G. The recovery of surface orientation from image irradiance. Proceedings of DARPA Workshop on Image understanding, Palo Alto, September 1982.
- Phong, B. T. Illumination for computer generated pictures. Com-12 munication ACM18 No. 6, June 1975, pp. 311-317.
- 13 Cook, R. L. and Torrance, K. E. A reflectance model for computer graphics. Computer Graphics, August 1981, 15(3), 307-316.
- 14 Brassel, K. A model for automatic shading. The American Cartographer, 1974, 1(1), 15-27.
- 15 Richards, W. (Ed.) Photometric method for determining shape from shading. In Image understanding, 1984, pp. 97-125 (Ablex, Norwood, New Jersey).
- 16 McPherson, C. A. 3-D robot vision. Proceedings of Third International Conference on Robot vision and sensory controls, Cambridge, Mass., November 1983, pp. 116-126.
- Mackworth, A. K. Interpreting pictures of polyhedral scenes. In Artificial intelligence, Vol. 4, 1973, pp. 121-137 (North-Holland, Amsterdam).
- 18 Horn, B. K. P. Understanding image intensities. In Artificial intelligence, Vol. 8, 1977, pp. 201-231 (North-Holland, Amsterdam). Horn, B. K. P. and Woodham, R. I. Determining shape and reflec-
- 19 tance using multiple images. MIT AI Lab. memo 490, 1978.
- 20 Nikravan, B. Image data acquisition, processing and object recognition for use in a machine vision system. PhD thesis, 1987, Mechanical Engineering Department, University of Leeds.
- 21 James, R. and James, G. Mathematics dictionary, 1959 (Strand Company, New Jersey). 22 Horn, B. K. P. Robot vision, 1986 (MIT Press, McGraw-Hill
- Company, New York).

10. PAPERS UNDER REVIEW

.

Publication		
[66]	Robot control using self-organizing fuzzy logic.	1
[67]	A contribution to fixture design rules: An expert system.	35
[68]	A comparison of orthogonal transforms in engineering computer vision.	69
[69]	Monitoring and assessment of engineering surface texture using computer vision.	91

•

PAPER 66

.

ROBOT CONRTOL USING SELF-ORGANISING FUZZY LOGIC

WAKILEH, B.A.M. and GILL, K.F.

ABSTRACT

A theoretical investigation is described that attempts to demonstrate fuzzy logic is an effective alternative algorithm for use in robot manipulator control. Employing such a control law avoids the need for a mathematical description of the robot and the algorithm is still able to compensate for the changing process characteristics that occur during operation.

It is shown by employing a motion strategy, that synchronous motion of the manipulator joints can be achieved.

1. INTRODUCTION

It has been recognised in the published work that real time control of a manipulator based on a detailed dynamic model is difficult to achieve if not impossible [1-3]. The equations used to represent a manipulator mathematically are both complex and non-linear, difficult to handle computationally and require a relatively long computer "run time" for their solution. The requirement exists, therefore, for an alternative approach and these authors adopted fuzzy logic, a control strategy that has been successfully applied to processes difficult to model.

Since its introduction by Mamdani and Assilian [4], the 'simple' fuzzy logic controller has been implemented in many test cases and in actual industrial applications [5-12]. Its performance, however, is dependant upon the availability of a reliable linguistic control plan which is not always easily formulated [13,14]. An attractive solution to this problem is provided by the self-organising fuzzy logic controller (SOC) proposed by Procyk and Mamdani [15,16] which uses closed loop performance data to generate and modify the control rules.

This paper describes an application study of the SOC algorithum, Fig. 1, and highlights some of the design problems most likely to be encountered. The results represent a further contribution to the published work on fuzzy logic and indicate that the SOC is an adaptive control approach worthy of consideration for use in robotics.

2. SIMULATION

In developing a computer emulation for a robot mechanism, it is necessary to formulate the rigid body equations of motion. The two principal methods adopted for the solution of these equations

are the numeric and symbolic techniques. Normally the preference is for a numeric solution as emphasis is placed on computational efficiency, because of its use in real time control. The symbolic technique, however, yields state equations that provide a better insight into the dynamic and control problems associated with manipulators and therefore is more suited to control system analysis and synthesis. The symbolic equations are complex and difficult to derive for a manipulator with more than three degrees of freedom, however, computer programs for the derivations of these equations have been developed [17]

To test the proposed control strategy the Stanford manipulator [2] was adopted for the study. This manipulator comprises six joints, in which the third is prismatic, giving the structure six degrees of freedom.

The first three links of the manipulator called the post, shoulder and boom, with associated variables θ_1 , θ_2 and d_3 respectively, form the manipulator's positional section and is the heaviest part of its structure. The last three joints form the end effector with variables θ_4 , θ_5 and θ_6 respectively. The axis system employed was defined by Denavit and Hartenberg [18] and the associated initial condition kinematic parameter values for the manipulator configuration shown in Fig. 2 are listed on Table 1 for reference.

Each manipulator joint is actuated by a separate d.c. armature controlled electric motor; torque amplification being achieved by an appropriate speed reduction gearbox.

The symbolic technique selected is the Lagrauge-Euler formulation presented by Bejecy [2] and Paul [19] incorporating a 4 x 4 homogeneous transformation matrix. The formulation is simple, systematic and generates highly structured equations of the form:

$$DM (\theta) \ddot{\theta} = AM\dot{\theta} + BMV + DGM (\theta) + DVM (\theta, \dot{\theta})$$
(1)

where $DM(\theta)$ = effective link inertia matrix,

AM = motor feedback gain matrix,

BM = input matrix,

 $DGM(\theta)$ = gravity force vector,

and $DVM(\underline{\theta}, \underline{\dot{\theta}}) = Corious$ and centrifugal force vector. This latter term is only significant at high manipulator speeds [1] and was equated to zero in this study.

The manipulator model, Eq. 1, is simulated using the state variable notation

$$\mathbf{x}_{i} = \mathbf{\theta}_{i}, \mathbf{x}_{i+6} = \mathbf{\dot{\theta}}_{i}, \mathbf{x}_{3} = \mathbf{d}_{3}$$
 and $\mathbf{x}_{9} = \mathbf{\dot{d}}_{3}$

and for this nomenclature, Eq. 1 becomes

$$\begin{bmatrix} \underline{E} & \underline{O} \\ 0 & DM(\underline{\theta}) \end{bmatrix} \stackrel{\cdot}{\underline{X}} = \begin{bmatrix} \underline{O} & \underline{E} \\ 0 & AM \end{bmatrix} \stackrel{\cdot}{\underline{X}} + \begin{bmatrix} \underline{O} \\ DGM(\underline{\theta}) \end{bmatrix} + \begin{bmatrix} \underline{O} \\ BM \end{bmatrix} \stackrel{\vee}{\underline{W}}$$
$$\stackrel{\cdot}{\underline{X}} = DD^{-1} \begin{bmatrix} AA\underline{X} + DG + BB\underline{V} \end{bmatrix}$$
(2)

All the relevant details of Eq. 1 can be found in [2] and the meaning of the additional terms in Eq. 2 are:-

$$\underline{\mathbf{x}} = [\theta_1, \theta_2, d_3, \dots \theta_6, \dot{\theta}_1, \dot{\theta}_2, \dot{d}_3, \dots, \dot{\theta}_6]^{\mathrm{T}},$$
$$\underline{\mathbf{v}} = [\mathbf{v}_1, \mathbf{v}_2, \mathbf{v}_3, \mathbf{v}_4, \mathbf{v}_5, \mathbf{v}_6]^{\mathrm{T}},$$

E is a unity matrix and 0 is a null matrix.

A discrete representation for Eq. 2 can be written as:-

$$\underline{\mathbf{x}}(\mathbf{k}\mathbf{T}+\mathbf{T}) = \mathbf{A}_1 \underline{\mathbf{x}}(\mathbf{k}\mathbf{T}) + \mathbf{Q}_1 + \mathbf{B}_1 \underline{\mathbf{v}}(\mathbf{k}\mathbf{T})$$

where

and

$$A_{1} = e^{FT}$$

$$B_{1} = F^{-1} [e^{FT} - E] DD^{-1} BB$$

$$Q_{1} = F^{-1} [e^{FT} - E] DD^{-1} DG$$

$$F = DD^{-1} AA$$

$$T = sample time.$$

The term A_1 , B_1 , and Q_1 are variable and have to be computed at each sampling instant. Hence a series solution to e^{FT} that converges rapidly is required. In the approach suggested by Franklin and Powell [20], the exponential is computed for $T/2^k$ and the value of k chosen by ensuring the largest element of $FT/2^k$ is less than unity. Employing this method, A_1 is approximated to a value better than 0.001 in 20 interations.

The model proposed is a sound representation and should ensure that the results of the control study can be used with confidence.

3. DETAILS OF THE SELF ORGANISING CONTROLLER

3.1 The Performance Index

The SOC is an extension of the simple fuzzy logic controller that incorporates performance feedback (Fig. 1) The performance index measures the system output deviation from a desired trajectory and issues appropriate correcting commands at the controller output, the controller inputs chosen are joint position error and speed. The performance index is formulated linguistically using the same fuzzy terms and linguistic statement form as presented in [21] to describe a solution path during one sample interval. The same algorithm will be used for all manipulator joints although the real attainable trajectory for each joint may be different. This is possible because the performance index value is dependent on scaling factor values which can be different for each joint.

The performance index rules, Table 2 (a), are written to generate control rules starting from a controller containing no rules and does make for rapid rule generation as indicated by the few ZO terms employed. The weakness of an active performance index rule table is the continuous rule modification when optimal conditions are attained.

The linguistic rules are transformed into a look up table of output commands using the standard techniques of fuzzy calculus [22] and the results are as shown in Table 2 (b).

3.2 SELECTION OF PROCESS MODEL

It has been shown by [15,21] that the SOC performance is insensitive to the accuracy of the model used. Therefore, to reduce the computational time in evaluating the SOC output at each sample interval, these authors have adopted the unit matrix for the process model.

3.3 RULE MODIFICATION AND OUTPUT SET CALCULATION

The six jointed manipulator requires a general rule that is written in implication form as:-

$$A_{1}^{+}CA_{1}^{+}A_{2}^{+}CA_{2}^{+}A_{3}^{+}CA_{3}^{+}A_{4}^{+}CA_{4}^{+}A_{5}^{+}CA_{5}^{+}A_{6}^{+}CA_{6}^{+}V_{1}^{+}V_{2}^{+}V_{3}^{+}V_{4}^{+}V_{5}^{+}V_{6}$$
(3)

where A_i , CA_i and V_i are fuzzy sets representing position error, joint speed and voltage supply respectively. The system is designed to generate rules for an empty rule store. Thus the initial rule is formulated by fuzzifying the initial conditions

$$(a_1^{i}, ca_1^{i}, a_2^{i}, ca_2^{i}, ..., a_6^{i}, ca_6^{i}, P_{i1}^{i}, P_{i2}^{i}, ..., P_{i6}^{i})$$
 where

 $(v_1^{i} = v_2^{i} = v_3^{i} = v_4^{i} = v_5^{i} = v_6^{i} = 0)$. The fuzzification process is done by providing a symmetrical spread of membership function values around the single support elements, thus creating fuzzy sets that are the same for all the joints. These are:

$$A = \{(a - x), \mu_{K} (a - x)\}$$

$$CA = \{(ca - x), \mu_{K} (ca - x)\}$$

$$V = \{(v + P_{i} - x), \mu_{K} (v + P_{i} - x)\}$$
(4)

where

$$\mu_{K} (a - x) \mu_{K} (ca - x) \mu_{K} (v + P_{i} - x)$$
 =

$$\begin{cases} 1.0 & x = 0 \\ 0.7 & x = \pm 1 \\ 0.2 & x = \pm 2 \\ 0.0 & 3 \le x \le -3 \end{cases}$$

At each sampling instant the element values of input and output from the controller are stored as a new rule to be used for control and modification purposes. As the general manipulator rule requires 12 antecedents and 6 consequents for a rule, each rule is stored using its single elements (a ,ac ,..etc) in an 18 element array while limiting the rule store to a capacity of 1000 rules. This arrangement is selected as a compromise between computer time required for the manipulation of the output set and storage.

If the present sample time is kT and the modification is made to the controller output rT samples earlier, then the rules to be included result from the fuzzification of the single elements $a_1(kT-rT)$, $ca_1(kT-rT)$, $a_2(kT-rT)$,, $v_6(kT-rT)+P_{16}(kT)$ and the inclusion of new rules into the store may result in the presence of contradicting rules.

Daley and Gill [21] overcame this by the deletion of rules that have identical antecedents to the rules to be included. This is implemented by comparing directly the coincident support sets of the respective rules. To reduce the number of rules generated (especially for a multi-input multi-output process) the procedure deletes rules unless the antecedents are displaced along the universe of discourse by more than one support value relative to the rule to be included. This is expressed linguistically as:

"Deletes all rules that are about the same as the one to be included"

To keep computation to a minimum, the modification and removal procedure is applied at each sample instant. The removal procedure checks the antecedents of the most recent rule with the rest allowing only those rules with dissimilar antecedents to be stored. The most recent rule can be contradicted by at most, one of the remaining rules. An existing contradictory rule will be deleted if it constitutes one of the recently non modified rules, otherwise the most recent rule is excluded.

The controller output set is obtained using the compositional rule of inference. This will be demonstrated for two antecedents and one consequent rule to ease the presentation. It is readily extended to the higher dimensional case. The controller rule

The membership function is defined by

$$\mu_{R}^{}$$
 (a, ca, v) = min { $\mu_{A}^{}(a)$, $\mu_{CA}^{}$ (ca), $\mu_{V}^{}(v)$ }

If the measured fuzzy sets at some instant are \hat{A} and \hat{AC} , the implied output fuzzy set \hat{V} is obtained using the compositional rule of inference,

$$V = (A^{\circ} (CA^{\circ} (A \times CA \times V)))$$

and the membership function is

$$\mu_{\hat{V}}(v) = \max \min \{ [\max \min\{\mu_{R}(a, ca, v), \mu_{C\hat{A}}(ca)\} \}, \mu_{\hat{A}}(a) \}$$
(5)

For a control algorithm that is composed of several control rules, the output fuzzy set $\sqrt[V]{}$ is defined by membership function
The deterministic output (control action) is obtained by use of the mean of maxima procedure. Clearly, the computation time to estimate the output set depends on the form of the fuzzy sets A and CA used in the algorithm. For the general rule Eq. 3 and fuzzy sets formed using the process detailed in Eq. 4, it would require 4.1×10^{20} operations to generate \hat{V}_1 to \hat{V}_6 .

The computation time needed for these operations is excessive and drastic reduction in the number of these operations is required if the SOC is to offer a viable alternative to classical control methods.

One approach to reducing the number of operations is to replace Eq. 3 by six separate expressions,

$$A_{1} \rightarrow CA_{1} \rightarrow A_{2} \rightarrow CA_{2} \rightarrow A_{3} \rightarrow CA_{3} \rightarrow A_{4} \rightarrow CA_{4} \rightarrow A_{5} \rightarrow CA_{5} \rightarrow A_{6} \rightarrow CA_{6} \rightarrow V_{1}$$

$$A_{1} \rightarrow CA_{1} \rightarrow A_{2} \rightarrow CA_{2} \rightarrow A_{3} \rightarrow CA_{3} \rightarrow A_{4} \rightarrow CA_{4} \rightarrow A_{5} \rightarrow CA_{5} \rightarrow A_{6} \rightarrow CA_{6} \rightarrow V_{2}$$

$$A_{1} \rightarrow CA_{1} \rightarrow A_{2} \rightarrow CA_{2} \rightarrow A_{3} \rightarrow CA_{3} \rightarrow A_{4} \rightarrow CA_{4} \rightarrow A_{5} \rightarrow CA_{5} \rightarrow A_{6} \rightarrow CA_{6} \rightarrow V_{3}$$

$$A_{1} \rightarrow CA_{1} \rightarrow A_{2} \rightarrow CA_{2} \rightarrow A_{3} \rightarrow CA_{3} \rightarrow A_{4} \rightarrow CA_{4} \rightarrow A_{5} \rightarrow CA_{5} \rightarrow A_{6} \rightarrow CA_{6} \rightarrow V_{4}$$

$$A_{1} \rightarrow CA_{1} \rightarrow A_{2} \rightarrow CA_{2} \rightarrow A_{3} \rightarrow CA_{3} \rightarrow A_{4} \rightarrow CA_{4} \rightarrow A_{5} \rightarrow CA_{5} \rightarrow A_{6} \rightarrow CA_{6} \rightarrow V_{4}$$

$$A_{1} \rightarrow CA_{1} \rightarrow A_{2} \rightarrow CA_{2} \rightarrow A_{3} \rightarrow CA_{3} \rightarrow A_{4} \rightarrow CA_{4} \rightarrow A_{5} \rightarrow CA_{5} \rightarrow A_{6} \rightarrow CA_{6} \rightarrow V_{5}$$

$$A_{1} \rightarrow CA_{1} \rightarrow A_{2} \rightarrow CA_{2} \rightarrow A_{3} \rightarrow CA_{3} \rightarrow A_{4} \rightarrow CA_{4} \rightarrow A_{5} \rightarrow CA_{5} \rightarrow A_{6} \rightarrow CA_{6} \rightarrow V_{5}$$

$$A_{1} \rightarrow CA_{1} \rightarrow A_{2} \rightarrow CA_{2} \rightarrow A_{3} \rightarrow CA_{3} \rightarrow A_{4} \rightarrow CA_{4} \rightarrow A_{5} \rightarrow CA_{5} \rightarrow A_{6} \rightarrow CA_{6} \rightarrow V_{6}$$

for which \hat{V}_1 to \hat{V}_6 can be computed using 2.3×10^{12} operations. A further reduction is possible by limiting the compositional rule evaluation to the non zero membership function values. For example, the fuzzy sets A and \hat{A} defined

	-6	-5	-4	-3	-2	-1	0	1	2	3	4	5	6
A	0	0	0	0.2	0.7	1	0.7	0.2	0	0	0	0	0
Â	0	0	0	0	0	0.2	0.7	1	0.7	0.2	0	0	0

yields for the minimum operation, min ($\mu_{\hat{A}}(a)\,,\,\mu_{\hat{A}}(a)\,)\,,$ the results

|--|

Only the part of the fuzzy set A that overlaps with the fuzzy set \hat{A} produces non zero membership function values and thus contribute to the max-min operation. By defining the overlapping region between a fuzzy set of an antecedent with its corresponding fuzzy measurements in a rule and considering the fuzzy kernel form of Eq. 4, the formulation of \hat{V}_1 to \hat{V}_6 will need a maximum number of 2.4 x 10⁶ operations. The same technique is used to exclude rules that do not contribute to the output set. These rules have resultant zero membership function values for all the support sets values signifying that a fuzzy set and its corresponding measurement do not overlap. Despite this modification, the number of operations is still high and the problem is aggravated with an increasing number of stored rules.

An alternative approach is to consider elements of A and CA as fuzzy singletons in which case Eq. 5 can be written as:

$$\mu_{\hat{\mathbf{U}}}(\mathbf{v}) = \min \left[\mu_{\lambda}(\hat{\mathbf{a}}), \, \mu_{C\lambda}(\hat{\mathbf{ca}}), \, \mu_{\mathbf{V}}(\mathbf{v}) \right]$$
(6)

Using Eq. 6, it would require 30 operations to formulate the fuzzy sets \hat{V}_1 to \hat{V}_6 , however, the use of fuzzy singletons reduces the impact of accurate measurement.

3.4 SELECTION OF CONTROLLER PARAMETERS

An alternative approach to that described by Daley and Gill [21] is employed for the initial selection of the scaling factors. The maximum real joint positional error and speed are mapped to the maximum descrete support value of the respective fuzzy sets, thus maximising the sensitivity of the response during rise time. The values obtained using this procedure are shown in Table 3.

3.5 FORM OF PROCESS INPUT

Given the discrete controller output $\underline{u}^*(kT)$ at sample instant kT, two forms of process input $\underline{u}(kT)$ are used, that is the incremental process input given by

$$\underline{u}(kT) = \underline{u} [(k-1)T] + \widehat{GV} \underline{u}(kT)$$

and an absolute form given by

$$\underline{u}(kT) = \widehat{GV} \underline{u}^{*}(kT)$$

The controller output scaling factor value GV is calculated for both forms to allow for the maximum voltage supply to be used. The maximum voltage supply is mapped to the maximum discrete support set value of the controller output universe of discourse, via

 $GV_i = 90/7 = 12.85$ for $i=1, 2, \dots, 6$.

4. MOTION CONTROL

Improvements to manipulator motion is made when the manipulator joints are actuated simultaneously along pre-defined motion paths. However, manipulator motion of this type induces significant dynamic cross coupling at the joints and gives rise to a more severe control problem.

A motion trajectory can be described by line segments created by time dependent input disturbances, where position, velocity and acceleration is defined along every segment of the trajectory. This defined motion allows the manipulator joints to operate synchronously, each joint completing that fraction of the motion allotted to it at the same time. Consequently, the manipulator can execute all movements more quickly and has the capability to track a moving object.

To execute a motion path, the manipulator is made to travel from one location to another in a pre-determined time. The trajectory is created by two segments, a constant velocity segment to execute the principle motion and a transitional segment that interpolates between successive operations, Fig. 3, thus providing the necessary continuity in trajectory position, velocity and acceleration components to avoid eratic motion.

To generate the trajectory, Paul [19] selected a polynomial function of time to describe the transitional trajectory segment over the time interval- $t_{acc} < t < t_{acc}$, Fig 3, where t_{acc} is the time allowed for the manipulator to change its velocity.

The function provides the necessary motion continuity for the trajectory and uses three boundary conditions at each end of a segment. This would require the formulation of a 5th order polynomial, however, because of symmetry of the transitional segment a 4th order polynomial

suffices, i.e.

$$q(t) = a_4 t^4 + a_3 t^3 + a_2 t^2 + a_1 t + a_0$$
(6)

where q(t) is the generalised position. This expression, Eq. 6, does define both translational and rotational movements of a joint.

Differentiation with respect to time yields generalised velocity and acceleration expressions $(\dot{q}(t) \text{ and } \ddot{q}(t) \text{ respectively})$ which Paul [19] showed to be:

$$q = \left[(\Delta C \frac{t_{acc}}{T_1} + \Delta B) (2 - h) h^2 - 2\Delta B \right] h + B + \Delta B$$

$$\dot{\mathbf{q}} = \begin{bmatrix} (\Delta C \frac{t_{acc}}{T_1} + \Delta B) (1.5 - h) 2h^2 - \Delta B \end{bmatrix} \frac{1}{t_{acc}}$$
(7)

$$\ddot{q} = \left[(\Delta C \frac{t_{acc}}{T_1} + \Delta B) (1 - h) \frac{3h}{t_{acc}^2} \right]$$

where $\Delta C = C-B$ $\Delta B = A-B$

$$h = \frac{t + t_{acc}}{2t_{acc}} \text{ (see Fig.3)}$$

After the transitional period (time t \ge t_{acc}) the general trajectory equations for the constants velocity segment become:

 $q = \Delta Ch + B$

}

$$\dot{q} = \frac{\Delta C}{T_{1}}$$

$$\ddot{q} = 0$$

$$h = t/T_{1}$$
(8)

Employing a fixed acceleration time t_{acc} and a travel time T_1 , it is not necessary to plan the entire task A to D, but to look ahead for the next operation in sequence (C) once the manipulator has reached the start of a transitional segment (A).

To apply this procedure to all manipulator joints, the following information must be available. If J is an array containing the current manipulator joint variable values at time $t = T_1 - t_{acc}$ and J_C the joint variables corresponding to point C in Fig. 3, then the joint variables at point D can be evaluated as J_D . The time needed to move to points J_D , for each joint, can be estimated from:

$$t_{i} = \frac{J_{Di} - J_{Ci}}{\dot{\theta}_{maxi}}$$

where θ_{maxi} is maximum velocity of joint i.

The travel time value (T_1) is selected to be either t_i or $2t_{acc}$ depending which is larger. Having estimated these values, the synchronised trajectories are generated for each joint from Eq. 7 and Eq. 8.

5. SIMULATION RESULTS

To rigorously test the SOC algorithm, a motion path was selected to ensure that a diverse and large number of rules would be created

during each trial. A transition segment period of 0.1 seconds was chosen between the path segments to allow 'smooth' movement and use was made of the whole actuator velocity range from maximum positive to maximum negative value.

All the results presented are obtained starting from an empty controller store for each trial conducted. Repeated trials under the same conditions along the same trajectory add a similar number of rules, but these are only marginally different from those of the first trial. The responses to a preplanned datum trajectory for the parameter values of Table 3 are shown in Fig. 4.

The controller output is incremental in form and the rewards are directly assigned to the rules created in the previous sample. As indicated, only a slight change on the initial response is obtained after the 6th run. The number of rules stored in the initial run is 56 rules and these increased to 321 rules after the 6th run.

The two most critial parameter values that dictate the system performance are found to be the delay in reward parameter and the form of controller output. Distributing the reward over several samples in accordance with previous SOC applications [15,21] leads to a poor manipulator response as shown in Fig. 5. For this figure the reward is distributed over the rules created in the previous five samples. When the reward is distributed over several samples the individual value to be added to each rule becomes too small to be effective in initiating rule modification. Consequently, fewer unmodified rules are stored and this will in general lead to a deterioration in system performance.

To determine the impact on system behaviour of eliminating the delay in reward parameter, the imminent action to be taken is rewarded and the changes that result can be seen by comparison of Fig. 4 and Fig. 6.

Both the absolute and incremental forms of the process input have been tested successfully in the previous simple fuzzy logic controller applications cited. The absolute form of the process input, however, does give poorer responses for this particular application and Fig. 7 shows larger response deviations from the set point over certain trajectory segments. This form of process input, coupled with a low resolution of the discrete support set values that are mapped to 90 volts reduces the ability of the manipulator to follow the set-point trajectories. To avoid this weakness, the approach adopted has been to use the incremental form of process input because of its superior sensitivity in the vicinity of the varying set point trajectory.

The behaviour of the system based on error and change in error universes of discourse is obtained for the scaling factors listed in Table 3 for completeness.

The responses, Fig. 8, show that this arrangement yields less sensitive responses in comparision with that of Fig. 4 in which speed is used instead of change in error. The results indicate that benefits can be gained by employing different controller input terms to those previously used in SOC applications. Moreover, speed measurement is directly accessible and avoids the computation to evaluate the change in error.

Responses using the controller parameter values of Fig. 4 for different manipulator configurations are shown in Fig. 9 and Fig. 10 and these help to confirm that the SOC algorithm is robust to process parameter change.

Observation of joint motion for certain manipulator configurations has shown large sustained positional errors in parts of the trajectory. The joints driven at maximum actuators speeds are most affected, i.e. joints 1 and 3 in Fig. 4. It is known that certain manipulator

configurations and or loading increases the effective joint inertia and the associated increased power demand on the actuators leads to an increased positional error. The problem is resolved by placing an upper limit on the speed at which set-point variations can be demanded. Employing the controller parameter values of Fig. 4 and limiting the trajectory speeds to 75% of the maximum actuator speed available, gives a marked improvement in system response, Fig. 11.

6. CONCLUSIONS

The point to point movement for individual manipulator joints is considered inefficient by present robotic standards. A better motion strategy is to move all the manipulator joints synchronously, thus reducing travel times by excluding stoppage delays. The inability to represent fully the detailed manipulator dynamic equations linguistically does prevent the use of synchronous motion with the simple fuzzy logic controller [4]. This can be overcome by adopting the self organising fuzzy logic algorthm [15,16]. The on-line SOC learning capability allows the rule algorithm to be generated and updated simultaneously as the manipulator motion proceeds and the manipulator dynamic properties are implicitly reproduced within the control system. The capacity of the controller to learn from its own environment makes it a viable algorithm to control the synchronous motion of manipulator joints.

Existing SOC algorithms were designed to respond to 'time invariant' system input disturbances. The requirement for synchronous motion of manipulator joints is for time varying input disturbances. To overcome this problem, an additional assumption is made that the working range of the system is considered to be restricted to a sample interval. This leaves the SOC structural design intact. Scaling

factor selection is more simple and direct, in comparison with other reported SOC applications. The delay in reward, however, must be accredited to recent actions unlike the earlier reported work [15,21]. In this application, an incremental form of process input is essential if the best performance is to be achieved.

The majority of motion control strategies employed in robotics utilise some form of manipulator dynamic model to maintain an acceptable motion control and this inevitably leads to computational time problems. The principal reason for adopting the SOC algorithm in this work was to avoid the need for a detailed dynamic model. In the first attempt to use the SOC it was found that the general rule Eq. 3 and fuzzy set form Eq. 4 required many minutes of computational time. The initial ideas of section (3.3) did not give sufficient time reduction and this led to all fuzzy measurements being represented as fuzzy singletons. With this algorithm modification, it was estimated that 0.01 second of CPU time is needed to calculate the manipulated variables over 250 rules.

The SOC arrangement finally used incorporates the unit matrix as the process model and is shown to perform well on the complex process of this study. The algorithm is computationally efficient, needs the minimum of computer storage and could be implemented on a microcomputer.

7. REFERENCES

1. PAUL, R.

"Modelling, Trajectory Calculation and Servoing of A Computer Controlled Arm", Stanford Artifical Intelligence Laboratory, Memo AIM - 177, Nov 1972.

BEJCZY, A.K. "Robot Arm Dynamics and Control", Technical Memo 33 - 669, Jet Propulsion Laboratory, Feb 1974.

3. LEE, C.S.G., CHUNG, M.J., TURNEY, T.L., AND MUDGE, T.N. "On the Control of Mechanical Manipulators, Proceedings of the Sixth IFAC Conference in Estimation and Identification, Washington D.C., June 1982, pp. 1454-1459.

MAMDANI, E.H. AND ASSILIAN, S. "An Experiment in Linguistic Synthesis with a Fuzzy Logic Controller", Int. J. Man-Machine Stud., Volume 7, 1975, pp. 1-13.

5. KICKERT, W.J.M. AND VAN NAUTA LEMKE, H.R. "Application of a Fuzzy Controller in a Warm Water Plant", Automatica, Volume 23, 1976, pp. 201-208.

6. OSTERGAARD, J.J.

"Fuzzy Logic Control of a Heat Exchanger Process", Internal report 7601, Tech. Univ. Denmark, Elec. Power Eng. Dept., 1976.

7. RUTHERFORD, D.A. AND CARTER, G.A.

"A Heuristic Adaptive Controller for a Sinter Plant", Proc. 2nd IFAC Symp. Automation in Mining, Mineral and Metal Processing, Johannesburg, 1976.

8. TONG, R.M.

"Some Problems with the Design and Implementation of Fuzzy Controllers", Internal report CUED/F-CAMS/TR 127, Cambridge University, 1976.

9. VAN AMERONGEN. J, VAN NAUTA LEMKE, H.R. AND VAN DER VEEN, J.C.T.

"An Autopilot for Ships Designed with Fuzzy Sets", Digital Computer Applications to Process Control, H.R. Van Naute Lemke (ed.), IFAC and North-Holland, 1977.

10. LARSON, P.M.

"Industrial Applications of Fuzzy Logic Control", Int. J. Man-Machine Stud., Volume 12, 1979, pp. 3-10.

11. TONG, R.M., BECK, M.B. AND LATTEN, A.

"Fuzzy Control of the Activated Sludge Wastewater Treatment Process", Automatica, Volume 16, 1980, pp. 659-701.

12. DALEY, S. AND GILL, K.F.

"A Justification for the Wider Use of Fuzzy Logic Control Algorithms". Proc. Instn. Mech. Engrs., Volume 199. No. Cl, 1985, pp.43-49.

13. BAINBRIDGE, L.

"The Process Controller", The Analysis of Practical Skills, W.T. Singleton (ed.), Academic Press, 1975.

14. UMBERS, I.G. AND KING, P.J.

"An Analysis of Human Decision-making in Cement Kiln Control and the Implications of Automation", Fuzzy Reasoning and its Applications, E.H. Mamdani and B.R. Gaines (eds.), Academic Press, 1981.

PROCYK, T.J. "A Self-organising Controller for Dynamic Processes", Ph.D. Thesis, Queen Mary College, London, 1977.

- 16. PROCYK, T.J. AND MAMANI, E.H. "A Linguistic Self-organising Process Controller", Automatica, Volume 15, 1979, pp. 15-30.
- 17. DILLON, S.P. AND HEMAMI, H. "Automated Equation Generation and its Application to Problems in Control", Proc. of JACC Conf., Austin-USA, June 1974, pp. 572-580
- DEPAVIT, J. AND HARTENBERG, R.S. "A Kinematic Notation for Lower-pair Mechanisms Based on Matrices, Journal of Applied Mechanics, June 1955, pp. 215-221.
- PAUL, R.P. "Robot Manipulator : Mathematics, Programming and Control, MIT Press, 1981.
- 20. FRANKLIN, G.F. AND POWELL, J.D. "Digital Control of Dynamic Systems", Addison-Wesley, 1980.

21. DALEY, S. AND GILL, K.F.

"A Design Study of a Self-organising Fuzzy Logic Controller. Proc. Instn. Mech. Engrs., Volume 200, No Cl, 1986, pp. 59-69.

22. TONG, R.M.

"A Control Engineering Review of Fuzzy Systems. Automatica, 1977, 13, pp. 559-569.



٤.

FIG. 1 SCHEMATIC ARRANGEMENT FOR SOC ALGORITHM



Figure 2 The Slanford Manipulator with reference frames of joint-link subassemblies

23

:



Fig.3 Displacement/time trajectory arrangement







FIG. 5 RESPONSE WITH REWARDS DISTRIBUTION OVER FIVE RULES



FIG. 6 RESPONSE WITH REWARD ON IMMINENT ACTION



FIG. 7 RESPONSE WITH ABSOLUTE FORM OF CONTROLLER OUTPUT



FIG. 8 RESPONSE WITH ERROR AND CHANGE IN ERROR UNIVERSE OF DISCOURSE

. .



FIG. 9 RESPONSE WITH CHANGE IN ROBOT GEOMETRY



FIG. 10 RESPONSE WITH CHANGE IN ROBOT GEOMETRY



32

......

		PB	PM	PS	joint speed ZO	i NS	NM	NB
	PB	zo	PS	PM	PB	PB	PS	PB
	PM	NS	20	PS	PM	PB	PB	PB
<u> </u>	PS	NM	NS	zo	PS	PM	PB	PB
erro	PO	NB	NM	NS	zo	zo	PM	PB
lon	NO	NB	NM	żo	zo	PS	PM	PB
osit	NS	NB	NB	NM	NS	20	PS	PM
<u> </u>	NM	NB	NB	NB	NM	N5	20	PS
	NB	NB	NB	NB	NB	NM	NS	zo
L								

a)

							joint	speed						
		-6	-5	-4	-3	-2	-1	٥	1	2	: 3	4	5	6
	-6	7.0	6.5	7.0	6.5	7.0	7.0	4.0	4.0	4.0	3.0	1.0	0.0	0.0
	-5	6.5	6.5	6.5	5.0	6.5	6.5	4.0	4.0	4.0	2.5	1.5	0.0	0.0
	-4	7.0	6.5	7.0	5.0	4.0	4.0	4.0	1.0:	1.0	1.0	0.0	-1.5	-1.0
	-3	6.5	6.5	6.5	5.0	4.0	4.0	4.0	1.5	1.5	1.0	0.0	-1.0	-1.5
	-2	7.0	6.5	7.0	4.0	1.0	1.0	1.0	0.0	0.0	-1.0	-1.0	-4.0	-4.0
	-1	6.5	6.5	6.5	4.0	1.5	1.5	1.5	0.0	0.0	-1.0	-1.5	-4.0	-4.0
LOL	-0	7.0	6.5	4.0	3.0	1.0	بە	0.0	0.0	0.0	-3.0	-4.0	-6.5	-7.0
n er	+0	7.0	6.5	4.0	3.0	0.0	0.0	۵.۵	-1.0	-1.0	-3.0	-4.0	-6.5	-7.0
Itio	1	4.0	4.0	1.5	1.0	0.0	0.0	-1.5	-1.5	-1.5	-4.0	-6.5	-6.5	-6.5
505	. 2	4.0	4.0	1.0	1.0	0.0	0.0	-1.0	-1.0	-1.0	-4.0	-7.0	-6.5	-7.0
	3	1.5	1.0	0.0	-1.0	-1.5	-1.5	-4.0	-4.0	-4.0	-5.0	-6.5	-6.5	-6.5
	4	1.0	1.5	0.0	-1.0	-1.0	-1.0	-4.0	-4.0	-4.0	-5.0	-7.0	-6.5	-7.0
	5	0.0	0.0	-1.5	-2.5	-4.0	-4.0	-4.0	-6.5	6.5	-5.0	-6.5	-6.5	-6.5
	6	0.0	0.0	-1.0	-3.0	-4.0	-4.0	-4.0	-7.0	-7.0	-6.5	-7.0	-6.5	-7.0

b)

-

Table \mathcal{Z} Performance index rule and look up tables

- - -

Joint number	θ degrees	d metres	a metres	α degrees	Range degrees
1	0	0.54	0	-90	-180 to 180
2	90	0.162	0	90	-180 to 180
3	-0	-	0	0	0 to 1.1m
4	0	0	0	-90	-180 to 180
5	0	0	0	90	-90 to 90
6	0	0.25	0	0	-180 to-180

Table 1 Kinematic parameter values for initialisation configuration

		C	GV	
Joint number	GX positional error	joint speed	change in error	controller output
1	100		160	10.05
	460	4.0	460	12.85
2	460	4.6	460	12.85
3	460	4.6	460	12.85
4	330	3.3	330	12.85
5	330	3.3	330	12.85
6	660	6.6	660	12.85

Table 3 Scaling factor values used in SOC studies

PAPER 67

A CONTRIBUTION TO FIXTURE DESIGN RULES: AN EXPERT SYSTEM A.R. DARVISHI AND K.F. GILL

The capability of a Flexible Manufacturing System to meet the demands for small to medium size batch production can be limited by component fixturing requirements. The rule based method illustrated could lead to an optimum solution for the fixture design problem.

To illustrate the approach proposed, the system has been used to design a fixture for a real, although simple, prismatic engineering component.

1. INTRODUCTION

Modern manufacturing demands medium to small batch production runs and this requirement has established the need for the flexible manufacturing system (FMS) [1]. An essential preliminary to the creation of a FMS is fixture design, itself a complex task with many variants. Each design variant must be satisified if optimum results are to be attained. No single design philosophy has been established in the literature to achieve an optimum fixture design and the results of this work suggest that a principal difficulty is one of knowledge representation. The authors believe that the use of an 'Expert System' offers a promising solution to the fixture design problem.

For readers not familiar with the expert system, a list of the most recent papers are references [2-18] and these indicate the type of data required to construct expert system tools.

In the authors opinion the most relevant of these publications are [1-3]. Markus et al. [1] proposed a prototype fixture design using a family of box-type pieces from a fixture kit, Miller and Haunam [2] tried a CAD/CAM fixture design procedure and developed a knowledge base for subsequent use with an expert system. Ingrand and Latpmke [3] suggested the use of an expert system for automatic fixture design, which defines fixture elements by their operation function.

An appropriate database for the development of a fixture design expert system is presented in [19] and looks at rule development for such a system. The example used to illustrate the approach adopted was chosen because the solution is understood and known to many engineers. The authors believe the example given, although simple, does allow the important ideas to be clearly described and effectively demonstrated.

The work was implemented on a VAX 11/780 computer using the OPS5 production system language [20], a language available for use in expert systems.

2. EXPERT SYSTEM

The fixture design expert system (FDES) proposed is divided into four modules, each intended to achieve a particular goal. The methodology employed is based on examining the design goals to be achieved and then creating rules to satisfy these imposed specifications. From these rules the fixture design evolves together with a list of all standard constructional fixture elements required to satisfy the production and manufacturing process needs.

MODULE 1 : FIXTURE SELECTION RULES

A fixture name usually emanates from the manufacturing process associated with the component to be machined and the machine application is used to identify the fixture type classification. Fixtures are also identified by a sub-classification, typically one designed for use only on a milling machine would be called a "milling fixture" [21].

The rules presented in the four selection modules of the FDES have been written for a milling fixture and expert rules have been developed for the more common types, that is, plate, angle-plate, modified, vise-jaw, indexing, duplex and the electromagnetic chuck.

In addition to the physical and geomtric attributes associated with a component, manufacturing operations require other attributes must be used in the fixture selection process. Typically, production rate, machining cycle, fixture complexity and each of these attributes can be assigned values, for example:-

Production rate

high	:	short lead time (high throughout),
normal	:	production demands readily met,
low	:	low throughput;

Machining cycle

continuous	:	high volume flow,
one-by-one	:	normal operation,
discrete	:	single item (infrequent requirement),

Fixture complexity

simplest	:	plate-fixture,
simple	:	duplex-fixture,
complex	:	special purpose fixture.

Separate rules can be developed to govern the selection of each type of fixture. In the example selected, the reasoning behind the rule construction is the same for all rules, and only those rules dealing with the plate and multi-station fixture, together with the electromagnetic chuck will be presented to illustrate the method proposed. Expansion of the approach outlined will lead to a system of use to the practitioner.

2.1.1 Plate Fixtures

The most versatile and rudimentary form of fixture is the plate fixture. This device comprises a flat plate which allows a component to be supported, located and clamped using a variety of standard elements. The first rule governing the selection of a plate fixture is based on the component's physical and geometrical attributes and the production requirements. The left-hand-side or CONDITION part of a rule is expressed linguistically as follows:

```
IF
```

THE GIVEN PART HAS THE FOLLOWING CHARACTERISTICS:

	-	part is not a member of a	part-family
	-	relative size of the part is	medium
	_	overall shape of the part is	prismatic
	-	part details are	simple
	-	batch size or number of the part to be	
		machined in each run is not	> N and $<$ M
	-	part life cycle is	24 months
	-	batch is repeated every	12 months
AND	IF	- •	

THE REQUESTED FIXTURE MUST HAVE THE FOLLOWING CHARACTERISTICS:

-	operation production speed is	milling normal
	no of fixuring station (s) is/are	one
-	it is used to make	simple-details
-	part size that can be accommodated is	medium
-	part position is	at-90-deg-to-support
-	production cycle is	one-by-one

THEN

(This is the "ACTION" part that must take place if the above conditions are simultaneously satisfied. When a match for the entire left hand side has been found, the rule is said to be instantiated).

- suggest "plate-fixture" is a "milling" fixture
- create a frame representing the suggested fixture
- ask for a name to be given to the suggested fixture
- create the suggested fixture
- put the supplied name in its "name" slot
- inform the user that the suggested fixture has been created
- put the created fixture in the "working-memory" (WM).

Rules are initiated using the command "P" (for production) in the OPS5 language. P takes as arguments the name of the Rule, e.g. FIXTURE==SELECTION=RULE1. The sequence of condition elements are partitioned from the action statements by the language separator '-->'. Employing the above procedure, the coding instructions are as shown in Table 1.

2.1.2 Multistation Fixtures

Multistation fixtures are primarily used for high-speed, high-volume production where the machining cycle is continuous. Duplex fixtures (2 station) are the simplest form of multistation fixture and allow component loading and unloading to take place without impeding the maching operation in progress. For example, once the machining operation is completed at one station, the cycle is repeated at a second station and simultaneously the component at station one is replaced by the next unit to be machined.

The selection of a multistation fixture, duplex in this case, is determined from similar condition elements to those presented in Section 2.1.1. The additional constraints include higher batch sizes, increased life-cycle, more frequent batch repetition. The rule construction and coding will be similar to those given in Table 1.

2.1.3 Magnetic Chuck

Most surface ground ferrous components are held during machining by this form of non-mechanical clamp fixed in a conventional manner to the machine table [22].

The Fixture-Selection-Rule which governs the selection of a magnetic chuck has more condition elements than the previous rule, however, the requirements of each condition element are more relaxed - i.e. a number of solutions are possible that will satisfy the required conditions because of the versatility and ease of use a magnetic chuck affords. The OPS5 coding for this rule is given in Table 2 and a description of its condition statements is as follows:-

a) First condition element PART :

As it is seen in the condition element PART, some attributes can have more than one value. This kind of value set is called "Disjunctions". A disjunction specifies a set of values, only one of which must correspond for the LHS-value to match. Disjunctions are denoted by twin angle brackets

typically:(PART RELATIVE-SIZE {<R-SIZE> <<SMALL MEDIUM LARGE>>})

The geometry and size of a magnetic chuck make it possible for it to accommodate small, medium and large components. Grinding operations can start from any point, in any direction, because interference with cutter movments can usually be eliminated. For optimal holding conditions, surface irregularities must be a minimum and preferably the mating surface should be machined.

b) Second condition element requested - fixture:

Naturally the unit will be classified as a grinding fixture. The facility of manual or automatic operation of the magnetic chuck allows the whole of the machining method spectrum to be accommodated, i.e. high to low production speeds, continuous to one-by-one production cycle, including automatic component placement.

c) Third condition element - machining operation:

Conditions in this element dictate the machining operation employed (grinding or lapping) and controls the depth of cut value. No constraints have been imposed on the range of attribute values available and traditional values can be modified to improve manufacturing efficiency.

To conclude this module, a node-and-link representation of the production tool is given in Fig. 1 in which the emphasis is on the fixtures functional attributes. Node-and-link representation is sometimes referred to as "Semantic Net" which can virtually be used to represent any concept.

2.2 MODULE 2 : RESTING OR REFERENCE PLANES SELECTION RULES

An attempt should always be made to develop at least one rule for each combination of the component's physical and geometrical attributes.

This enables rule construction to be more easily understood, facilitates rule modification and the inclusion of additional rules when desirable. The latter might be deemed necessary at a later stage, as experience with the system grows.

The first rule of module 2 requires the least number of condition elements and each additional rule becomes progressively more complex as the number of condition elements are increased.

The condition elements associated with the selection of the first reference plane, (1STRP) shown in Table 3 and illustrated in Fig. 3, can be expressed as follows:

IF

- there is a request to select the 1STRP, i.e. there is a WM (working memory) element matching the first condition element whose SURFACE-LABEL attribute has the value of NIL.

AND

- the feature is a SURFACE, and its label is S1 (say)
- part name to which S1 belongs is P1
- S1 is planar
- S1 is machined
- the quality of S1 is one of FINAL-FINISH, ROUGH-FINISH or
 ROUGH-MACHINED

- direction of external normal to S1 is opposite to direction of external normal of the machine table
- S1 form geonetry tolerances (parallelism and perpendicularity) = 0.01 mm.
- no features start from S1
- no features open into S1

THEN

- display the message:
 - "1STRP==SELECTION=RULE=" WAS FIRED AND SURFACE <surface-label> IS CHOSEN AS FIRST-REFERENCE-PLANE

AND

modify the WM matching the first condition element, i.e. map all attributes of surface S1 into the 1STRP and put it in WM

AND

- display :

THE "1STRP" WAS CREATED AND ASSIGNED TO WM

Rules governing the selection of the second and third reference planes follow the same pattern as those for the first. Employing the 3-2-1 principle, the second and third reference planes must eliminate the remaining degrees of freedom (DOF) associated with the component.

2.3 MODULE 3 : GENERIC ELEMENT SELECTION

This module includes three groups of rules, each dealing with the selection of a class of element for the 1st, 2nd and 3rd reference planes respectively. A member of this class will be referred to as a generic element.

In the selection of the generic elements (GE) associated with each reference plane, the important properties of STARTING and OPENING

features are presented. The constraints imposed are:

- i) no feature starts from the 1STRP, 2NDRP or 3RDRP,
- ii) features starting from the reference planes will be machined during subsequent machining phases,
- iii) features starting from the reference planes are already machined,
- iv) whenever a feature opens into a surface chosen as the reference plane, provisions must be made to avoid damaging the locating and supporting elements when the tool breaks into that surface, and
 - v) whenever a generic or standard-element is selected and the height is not specified, a fixed-height is assumed by default, unless otherwise stated.

2.3.1 Selection of Generic Elements for (1STRP)

The first reference plane is the most important and must eliminate at least three of the component's degrees of freedom. Therefore at least three generic elements are associated with it.

The first rule will satisfy the most basic requirements and contains the least number of condition elements. Each further rule will become progressively more complex, employing an increasing number of condition elements.
RULE1 : GENERIC==ELEMENT=SELECTION=FOR=1STRP=RULE1

IF

there is a need to select a generic element for the 1STRP
 i.e. SELECTION-STATUS of a WM element type 1STRP=GEN=EL is NIL.

AND

- 1STRP surface label is S1 (say)
- the part-name is P1 (say)
- S1 is planar
- S1 is machined
- feature opens into S1

AND

- suggested fixtures classification is either milling or grinding
- suggested fixture type is electromagnetic-chuck
- part is P1

AND

- type of operation i.e. lapping, grinding or finishing
 (or other operations, having shallow depths of cut)
- depth of cut is less than 1 mm (arbitrarily chosen, can be set to any value)
- machining phase could be any phase

AND

- If part is P1
 - component material is magnetizable
 - component's relative size is small or medium

THEN

- display the message

; BECAUSE FIXTURE-TYPE IS : <TYPE> : AND THE: TYPE-OF-OPERATION IS : <T-O-OP> DEPTH-OF-CUT IS SHALLOW PART-MATERIAL IS : MAGNETIZABLE AND THE 1SFRP IS PLANAR: MACHINED AND NO FEATURE OPENS INTO : IT.....THEN : THE THREE RESTING POINTS ASSOCIATED WITH : THE 1STRP ELIMINATES 3DOF AND CAN BE : REPLACED BY 3 GENERIC-ELEMENTS HAVING : ;;; FUNCTION CURRORTING OF DOCT TONTNO

FUNCTION	SUPPORTING OR POSITIONING	:
PRINCIPLE-OF-ACTION	FIX-BY-ATTRACTION	:
NATURE-OF-CONTACT	PLANE-TO-PLANE	:

AND

 modify the first condition element and set the SELECTION-STATUS attribute to SELECTED

AND

- associate the three generic elements GE11, GE12 and GE13 with the 1STRP, the attributes and values are:

IS-A	SUPPORT
PART-NAME	<pre><part-name></part-name></pre>
GEN-EL-CODE	GE11
ASSOCIATED-WITH	1STRP

FUNCTION	SUPPORT	;	POSITIONING	
PRINCIPLE-OF-ACTION	FIX-BY-ATTRACTION			
NATURE-OF-CONTACT	PLANE-TO-PLANE			
FRICTION-AND-JAMMING-FACTOR	VERY-LOW			
LOAD-UNLOAD-FACTOR	HIGH			
NO-OF-ELEMENTS-REQUIRED	1			
N.B. GE12 and GE13 will have	the same attributes	an	d values as	
GE11				

AND

display the message
 ;REQUIRED GENERIC-ELEMENTS WERE DETERMINED:
 AND ASSIGNED TO WM:
 GE11, GE12 and GE13 HAVE THE SAME ATTRIBUTES:
 AND CAN BE REPLACED BY THREE STANDARD ELEMENTS:
 OF THE SAME TYPE AND SIZE :

RULE 2: deals with a situation where the 1STRP into which a feature opens is a machined planar surface. The opening feature(s) could be one or any combination of common features, i.e. holes, slots and must have been machined in a previous phase of operation. The fixture type is an electromagnetic-chuck and the other condition elements are as in Rule 1. In the following, since the type and status of the opening features play a major role in determining the generic elements, checks are made to ensure that: IF THE OPENING FEATURE(S) IS ALREADY MACHINED:

- can the generic elements be replaced by the fixture's base plate or the machine-tool table,
- ii) if the opening feature is a hole and it has already been machined, checks are made to see whether the hole could be used for locating the component and
- iii) if a hole is to be machined and it opens into the 1STRP, the use of generic elements (which could be replaced by the support elements) is inevitable. Otherwise the drill tip will penetrate the machine table.

Rule 3: accommodates the situation where a plate-fixture (milling fixture) is suggested as the required fixture, the machining operation is one of drilling, boring or counterboring, the component material is either ferrous or non-ferrous, there is a feature(s) opening into the 1STRP and the opening feature is TO-BE-MACHINED.

The remaining rules in this group each deal with a specific situation eg. component attributes and manufacturing requirements.

2.3.2 Generic-elements Selection for 2NDRP AND 3RDRP

The second reference plane is perpendicular to the 1STRP, eliminates at least two component degrees of freedom and requires at least two generic-elements.

The third reference plane eliminate at least one degree of freedom of the component. Hence, there is at least one generic element associated with it. The 3RDRP is perpendicular to both the first and the second reference planes. The reasoning behind the selection of generic elements for the second and third reference planes is similar to those stated for the first reference plane and the only difference being the number required.

2.4 MODULE 4 : SELECTION OF STANDARD ELEMENTS

Rules governing the selection of standard elements, are presented in this module; typically supports, locators and clamps.

Standard elements are real physical objects, capable of fulfilling certain functions and the rules to be developed will map the attributes of the generic element into a standard element capable of fulfilling the desired function.

All constraints applicable to the generic element are transferred to the standard element,

- a) standard support elements of the 1STRP must not be collinear and
- b) standard support and locating elements of 2NDRP must be positioned as far apart as possible and collinear.

The rules of this module are divided into four groups, selection of standard elements for the 1STRP, 2NDRP and 3RDRP, and creating a list of required elements needed for the construction of the suggested fixture.

. . .

2.4.1 Selection of Standard Elements for 1STRP

RULE 1:

IF

```
- there is a 1STRP, standard elements are to be selected
```

AND

- if suggested fixtures classification is GRINDING or MILLING
- fixture type is electromagnetic-chuck
- part-name is P1 (say)

AND

- part material is magnetisable (ferrous)
- part relative size is small or medium

AND

- if the 3 generic elements of the 1STRP are the same, with attributes:

GENERIC-ELEMENT

IS-A	SUPPORT
GEN-EL-CODE	GE11
ASSOCIATED-WITH	1STRP
FUNCTION	- SUPPORTING
	- POSITIONING
	- LOCATING
PRINCIPLE-OF-ACTION	FIX-BY-ATTRACTION
PART-NAME	P1
NATURE OF CONTACT	PLANE-BY-PLANE
FRICTION-AND-JAMMING-FACTOR	LOW OR VERY-LOW
LOAD-UNLOAD-FACTOR	HIGH
NO-OF-ELEMENTS-REQUIRED	1

 if machining operation is grinding, finishing, lapping or finish-cut

AND

AND

- 1STRP belongs to P1

- and there is no feature opening into 1STRP

THEN

- display the message: : THE THREE STANDARD-ELEMENTS ASSOCIATED : WITH THE 1STRP ARE REPLACED BY THE: TOP-SURFACE OF THE ELECTROMAGNETIC: CHUCK. THIS CHUCK WOULD SERVE AS THE BASE : PLATE OF THE FIXTURE THE SUPPORT : ELEMENTS OF THE 2NDRP : AND 3RDRP WILL BE POSITIONED ALONG THE : 2NDRP AND 3RDRP :
- write

SUPPLY AN IDENTIFICATION-CODE OR A NAME FOR CHUCK :

- read the supplied name or ID-code:

modify the WM element matching the first condition element:

- create the required standard elements by mapping generic element attributes into standard element with attributes:

PART-NAME	<part-name></part-name>
FUNCTION	<func></func>
ASSCIATED-WITH	1STRP
PRINCIPLE-OF-ACTION	FIX-BY-ATTRACTION

NATURE-OF-CONTACTPLANE-TO-PLANEFRICTION-AND-JAMMING-FACTOR<F-A-J-F>LOAD-UNLOAD-FACTORHIGHIDENTIFICATION-CODE<ID-CODE>NO-OF-ELEMENTS-REQUIRED1

- display message: REQUIRED STANDARD-ELEMENT DETERMINED : AND ASSIGNED TO WM:

RULE 2: (Feature opening into the 1STRP.)

First to seventh condition elements of Rule 2 are identifical to Rule 1, the eighth and nineth condition elements are:

IF

- 1STRP belongs to Part P1
- and there are features opening into it

AND

- opening features (into 1STRP) are of type hole, simple-slot,
 T-slot or dovetail-slot
- and opening features are already machined

THEN

- actions as in Rule 1.

RULE 3:

If the suggested fixture is either of type Plate or Duplex and if there is a 1STRP for which standard elements must be selected and suggested fixture classification is milling or grinding and there are three generic-elements for the 1STRP with attributes:

	IS-A	SUE	PORT
	GEN-EL-CODE	GE1	.1
	ASSOCIATED-WITH	151	RP
	FUNCTION	-	SUPPORTING
		-	POSITIONING
		-	LOCATING
	PRINCIPLE-OF-ACTION	FIX	KED-IN-POSITION
	PART-NAME	P1	
	NATURE-OF-CONTACT	PLA	NE-TO-PLANE
	FRICTION-AND-JAMMING-FACTOR	-	VERY LOW
		-	LOW
	LOAD-UNLOAD-FACTOR	HIC	H
	NO-OF-ELEMENTS-REQUIRED	1	
THEN			
	- display:		
	THE THREE STANDARD-ELEMENTS ASSOC	IATI	ED:
	WITH THE 1STRP ARE REPLACED BY TH	E:	

53

THREE SUPPORT ELEMENTS HAVING THE FOLLOWING:

ATTRIBUTES:

AND

FUNCTION	SUPPORTING OR POSITIONING :
PRINCIPLE-OF-ACTION	FIXED-IN-POSITION :
NATURE-OF-CONTACT	PLANE-TO-PLANE :
- modify the WM element matching t	the first condition element

AND - create standard element with the following attributes and assign to WM. STANDARD-ELEMENT SUPPORT TYPE FLAT-TOP-SUPPORT

PART-NAME	<part-name></part-name>
ST-EL-CODE	ST-EL-777
ASSOCIATED-WITH	1STRP
FUNCTION	<func></func>
PRINCIPLE-OF-ACTION	FIXED-IN-POSITION
NATURE-OF-CONTACT	PLANE-TO-PLANE
FRICTION-AND-JAMMING-FACTOR	<f-a-j-f></f-a-j-f>
LOAD-UNLOAD-FACTOR	HIGH
NO-OF-ELEMENTS-REQUIRED	3

AND - display message:

REQUIRED STANDARD-ELEMENTS: WHICH ARE THREE FLAT-TOP-SUPPORTS WERE DETERMINED : AND ASSIGNED TO WM.

2.4.2 Selection of Standard Elements for the 2ND AND 3RDRP:

Rules governing the selection of standard elements for these two planes follow the same reasoning as those of generic elements for the respective planes. Some rules may differ, however, in one important respect that of type-of-height, i.e. the generic element(s) have adjustable height(s).

2.4.3 Selection of Clamps and Listing of Elements Used

When selection of standard support and locating elements for the 1STRP, 2NDRP and 3NDRP is finalised, the appropriate rules fire and a description is created of the required clamping elements. Illustrated in Fig. 2 is a node and link representation of these clamp elements and their functional attributes. Finally, when the appropriate clamping elements are determined, a complete list of standard elements is displayed for the suggested fixture. The list includes type, identification and quantities required from each standard element, together with the overall number of elements selected.

3. USING THE SYSTEM

To execute the program the working memory (WM) must be initialised after the declaration and rule section blocks of the OPS5 programming language have been loaded. The sequential steps in the execution of the program are:-

- Loading and declarations i.e. data structures developed [23] for representation of individual objects.
- ii) Loading the four production rule blocks that contain the action files for the selection of:
 - a) required fixture,
 - b) reference or resting planes,
 - c) generic elements and
 - d) standard elements and output the list of standard elements required to construct the fixture,
- iii) initialise working memory and
- iv) execute program.

4. OBJECTS REPRESENTATION

As a prerequisite to fixture design, the component surfaces and fixtures must be defined, the machine tool to be used identified and the process plan specified including at least one phase of the machining operation to be performed. The list below itemizes the information required by the system for the represention of:

i) component,

- ii) surfaces enveloping the component and features opening into and starting from each surface,
- iii) machine tool and sub-assemblies i.e. machine table, tool magazine,
- iv) phases of machining operations and
- v) requested fixture.

Having defined and saved these files, execution yields:

- Computations for determining significant machining environmental volumes and weights.
 - i. boxes containing the initial and final component (i.e.
 volume),
 - ii. the maximum working volume of the machine tool,

iii. overall weight of the component,

iv. overall weight of the component and fixture assembly.

- Machining operations required are checked to determine if they are within the machine capabilities. If not, warning messages are displayed.
- Fixture selection to hold the component for defined machining environment.
- Selection of three reference planes for component to eliminate six degrees of freedom.

- Conceptualised generic elements associated with the three reference planes are chosen.
- Mapping of the generic element attributes into standard fixture elements for intended functions.
- Generation of standard element list (with identification attributes) for construction of required fixture.

5. **RESULTS**

The results obtained using the system for an engineering example is given below. The workpiece chosen is a medium sized prismatic component manufactured in low carbon steel. The machining details include a simple slot and three through holes starting from the uppermost surface S6 and opening into surface S1. The process plan indicates the machining of the simple slot by end milling, followed by the drilling of the three holes; shown in Fig. 3 is the finished component.

The fixture to be designed must satisfy the conditions

- a) machining speed is to be normal,
- b) single station fixture with manual load and unload capability and
- c) the preferred component placement direction is the z axis.

With these engineering decisions taken the fixture design program would be executed [23] and a list of all standard construction fixture elements required to satisfy the production and manufacturing process would be available on a monitor screen. A typical display would be as shown in Table 4 in which all the necessary elements have been itemized.

6. CONCLUSIONS

The above program execution demonstrates very clearly the power of this novel system for fixture design. It will allow greater versatility than is now economically possible employing the traditional approach to fixture design. It must be emphasized this is a knowledge based approach and the system will be unable to make subjective judgments.

The practising engineer would generally welcome a three dimensional display of recommended elements in preference to the itemised list given. It is the view of these authors that in parallel with the future developments of the fixture design expert system proposed, CAD software should be utilized to create a library of standard fixture elements that can be referenced by assigned features.

The rules embodied within the expert system would be used to create a graphic representation for each selected element and the style recommended would be as shown in Fig. 4. The graphics facility would allow the system's user to generate a perspective assembly arrangement, as illustrated in Fig. 4e, to aid the actual machine set-up.

It is appreciated that the expert system software that incorporates the proposed graphic displays would require a significant computing facility. The best estimate that can be made on the size, at this stage of development, would be to support a hardware capability that can handle a currently available commercial CAD package.

7. REFERENCES

- MARKUS, A., MARKUSZ, Z., FARKAS, J.& FILEMON, J. Fixture Design Using Prolog : An Expert System. Rob. Comput. Integr. Manuf., Vol. 1, No. 2, p.167-172, 1984.
- [2] MILLER, A.S. & HANNAM, R.G. Computer Aided Design Using a Knowledge Base Approach and its Application to the Design Of Jigs and Fixtures. Proc. Instn. Mech. Engrs. Vol. 199, No B4, p. 227-234, 1985.
- [3] INGRAND, F., & LATOBE, J. Fundamental Reasoning for Automotive Fixture Design. Lifia (IMAG) France.
- [4] NEE, A.Y.C., BHATTACHARYYA, N. & POO, A.N. Knowledge-Based CAD of Jigs and Fixtures. Tech. Pap. Soc. Manuf. Eng. TE85-902, p.9, 1985.
- [5] RAMALINGAM, S. Expert Systems for Manufacturing : Examples of Tools to Assess Manufacturability. Manufacturing Engineering Transactions, p.411-417, 1985.
- [6] ENGBERT, P. & WRIGHT, P.K. Application of Artificial Intelligence and the Design of Fixtures for Automated Manufacturing IEEE, International Conference on Robotics and Automation, p.345-351, 1986.
- HARTLEY, P., STURGESS, C.E.N. & ROWE, G.W.
 Expert Systems for Design and Manufacture. Proceedings of the 12th All India Machine Tool Design and Research Conference, Delhi, p.21-27, 1986.
- [8] HAYES, C. & WRIGHT, P. Automated Planning in the Machining Domain. ASME Production Engineering Division, Vol. 2A, p.221-232, 1986.
- [9] SCHAFFER, G.H. Artificial Intelligence : A Tool for Smart Manufacturing. Am Mach Autom Manuf., Vol. 130, No. 8, p.83-94, 1986.
- [10] VAN'T, ERVE, A.H. & KALS, H.J.J. XPANE, A Generative Computer Aided Process Planning System for Part Manufacture. CIRP Annals, Vol. 35, No. 1, p.325-329, 1986.
- [11] HATRANY, J. (Ed) International Conference on Intelligent Manufacturing Systems. Robotics and Computer - Integrated Manufacturing, Vol. 3, No. 2, p.141-272, 1987.
- [12] LIM, B.S. & KNIGHT, J.A.G. A Foundation for a Knowledge Base Computer Integrated Manufacturing System. Int. J. Artf. Intell. Eng. (GB) Vol. 2, No. 1, p.11-22, 1987.
- [13] NEE, A.Y.C., BHALLASHORYYA, N. & POO, A.N. Applying AI in Jigs and Fixtures Design, Robotics and Compt – Integrated Manuf. (GB) Vol. 3, No.2, p.195-200, 1987.

- [14] LIM, B.S. IKBS for Integrateding Component Design to Tool Engineering. Expert Syst. Vol. 4, No.4, p.252-291, 1987.
- [15] NEVILL, G.E. & PAUL, G.H. Knowledge-Based Spatial Reasoning for Designing Structural Configurations. Computers in Engineering Conference and Exhibit, Vol. 1, ASME, New York, p.155-160, 1987.
- [16] de SAM LAZORA, A., PHAM, D.T. & ROBINSON, F.J. Automated Design for Fixtures for FMS. IEEE Comput. Soc. Press, p.201-207, 1988.
- [17] HANNAM, R.G. & LIU, C.L. An Assessment of Knowledge Bases and Data Bases in CAD/CAM Applications, Int. J. Adv. Manuf. Technolg. (UK) Vol. 3, No. 3, p.123-31, 1988.
- [18] NNAJI, B.O., ALLADIN, S. & LYU, P. A Framework for a Rule Based Expert Fixturing System for Face Milling Planar Surfaces on a CAD System Using Flexible Fixtures. J. Manuf. Syste. (USA) Vol. 7, No. 3. p.193-207, 1988.
- [19] DARVISHI, A.R. & GILL, K.F. Knowledge Representation Database for the Development of a Fixture Design Expert System. Proc. Instn Mech. Engrs, Vol 202, No B1, p.37-49, 1988.
- [20] BROWNSTON, L., FARRELL, R., Programming Expert Systems in OPS5, Addison Wesley, 1985.
- [21] BOYES, W., Jigs and Fixture, 2nd ed., Society of Manufacturing Engineers, (USA), 1982.
- [22] KRAR, OSWALD & ST. AMAND, Technology of Machine Tools, 2nd ed. McGraw-Hill Book Co, 1977.
- [23] DARVISHI, A.R., Philosophy of Fixture Design for Flexible Manufacturing Systems, Ph.D. Thesis, Leeds University, Mechanical Engineering Department, January 1987.

P FIXTURE==SELECTION=RULE1 (PART PART-NAME <ANY-NAME> ^IS-THE-PART-A-FAMILY-MEMBER NO **^RELATIVE-SIZE** MEDIUM **^OVERALL-SHAPE** PRISMATIC **^PART-DETAILS** SIMPLE-DETAILS **^BATCH-SIZE** $\{ \langle NO \rangle > 40 \langle = 200 \rangle \}$ PART-LIFE-CYCLE { <MONTHS> <= 24 } ≥ 1 $\hat{}$ **^BATCH-REPS-IN-12 MONTHS** {<REPS< Start of and if block ie. continuation of LHS (REQUESTED-FIXTURE MILLING **^CLASSIFICATION** ^ PRODUCTION-SPEED NORMAL **^NO-OF-FIXTURING-STATIONS** ONE PART-LOADING-METHOD MANUAL [^]USED-TO-MAKE SIMPLE-DETAILS **^ PRODUCTION-CYCLE** ONE-BY-ONE AT-90-DEG-TO-SUPPORTS ^PART-POSITION ^PART-SIZE MEDIUM end of LHS: Start of the block ie. RHS (CRLF) (TABTO 10) : "RULE1 WAS FIRED AND :) (OPS::WRITE (CRLF) (TABTO 10) : "PLATE-FIXTURE" WHICH IS AKO : (OPS::WRITE (TABTO 10) : "MILLING-FIXTURE' IS SUGGESTED :(CRLF) (TABTO 10) : FOR PART ***: <ANY-NAME> :***: (CRLF)) (CRLF) (TABTO 10) : WHAT DO YOU WISH TO CALL (OPS::WRITE (TABTO 10) : THE SUGGESTED FIXTURE : (CRLF)) the end part of the RHS (MAKE SUGGESTED-FIXTURE **CLASSIFICATION** MILLING **TYPE** PLATE-FIXTURE ^PART-NAME <ANY-NAME> **^NO-OF-FIXTURING-STATIONS** ONE MANUAL ^PART-LOADING-METHOD [^]USED-TO-MAKE SIMPLE-DETAILS ^ PRODUCTION-CYCLE ONE-BY-ONE **PART-POSITION** AT-90-DEG-TO-SUPPORTS **^PART-SIZE** MEDIUM **^NAME (ACCEPT)** (OPS::WRITE (CRLF) : THE SUGGESTED FIXTURE IS CREATED AND PUT INTO "WM" : (CRLF)) ; end of RHS

TABLE 1. CODING INSTRUCTIONS FOR PLATE FIXTURE

(P FIXTURE==SELECTION=RULE2 (PART ^PART-NAME <PART--NAME> <R-SIZE> <<SMALL MEDIUM LARGE>> } **RELATIVE-SIZE** { **^OVERALL-SHAPE** PRISMATIC SIMPLE DETAILS ^PART-DETAILS <T-O-MAT> <<FERROUS **^TYPE-OF-MATERIAL** { CAST-IRON STEEL >> } ^DIMENSIONAL-VARIATION <DIM-VAR> <0.1 }</pre> ^SPINDLE-DIRECTION <ANY-DIR> ^REQUIRED-CUTTING-DIRECTIONS { <R-C-DIR> << ANY-DIR > }
^PRIOR-MACHINING-OPERATION { <P-M-OP> << MILLING</pre> SLAB-MILLING FACE-MILLING>> } **^INITIAL-SURFACE-QUALITY** { <SURF-QUL> <<ROUGH **ROUGH-MACHINED FINISH** FINAL_FINISH>> }) (REQUESTED-FIXTURE { <CLA> <<GRINDING LAPPING>> } ^CLASSIFICATION { <P-S> <<HIGH NORMAL>> } ^ PRODUCTION-SPEED { <P-C> <<CONTINUOUS ^ PRODUCTION-CYCLE ONE-BY-ONE>> } ^PART_POSITION AT-90-DEG-TO-SUPPORTS **^PART-SIZE** { <P-SIZE> <<MEDIUM SMALL>> } [^]USED-TO-MAKE SIMPLE-DETAILS ^NO-OF-FIXTURING-STATIONS { <N-O-F-S> <<ONE TWO>> } PART-LOADING-METHOD { <P-L-M> <<AUTOMATIC MANUAL>> }) (MACHINING-OPERATION <P-NAME> ^ PART-NAME ^PHASE-NO <ANY-PHASE> **TYPE-OF-OPERATION** GRINDING ^DEPTH-OF-CUT $\{ \langle D-0-C \rangle \langle = 1 \}$; (OPS::WRITE (CRLF) (TABTO 10) "RULE7" WAS FIRED AND :) : "ELECTROMAGNETIC CHUCK" WHICH IS AKO : (OPS::WRITE (CRLF) (TABTO 10) : (TABTO 10) : "NON-MECHANICAL-CHUCK OR CLAMP" : (TABTO 10) : IS SUGGESTED : (CRLF) (CRLF) FOR PART ***: <P-NAME> :***:(CRLF) (TABTO 10) : (OPS::WRITE (CRLF) (TABTO 10) WHAT DO YOU WISH TO CALL: : (TABTO 10) THE SUGGESTED FIXTURE : (CRLF) : (MAKE SUGGESTED-FIXTURE ^CLASSIFICATION GRINDING TYPE ELECTROMAGNETIC-CHUCK ^PART-NAME <PART-NAME> ^NAME (ACCEPT) (OPS::WRITE (CRLF) : THE SUGGESTED FIXTURE IS CREATED AND PUT INTO "WM" : (CRLF))) TABLE 2 CODING INSTRUCTIONS FOR MAGNETIC CHUCK

(P 1STRP==SELECT==RULE 1 The following condition element checks that the "1STRP" does not already exist in the "WM" {(1STRP ^ SURFACE-LABEL NIL) $\langle 1 \rangle$ (SURFACE ^FEATURE_TYPE SURFACE **^PART-NAME** <PART--NAME> **^FEATURES-GENERIC-TYPE** EXTERNAL ^SURFACE-LABEL <SURF-LABEL> [^]SURFACE-TYPE PLANAR **STATUS** { <STATUS-VAL> << MACHINE AS-CAST>>} SURFACE-QUALITY { <S-QUL-VAL> << FINAL-FINISH FINISH ROUGH-MACHINED>> } ^DIRECTION-OF-EXTERNAL-NORMAL $-\mathbf{Z}$ **^IS-PARALLEL-TO-SURFACES** <PARA-SURFS> ^PARALLELISM $\{ \langle PAR-VAL \rangle \langle = .01 \rangle \}$ ^IS-PERPENDICULAR-TO-SURFACES {<PERP-SURFS> <> <PARA-SURFS> } ^ PERPENDICULARITY $\{ \langle PER-VAL \rangle < = .01 \}$ FEATURES-STARTING-FROM THIS-NIL SURFACE ^FEATURES-OPENING-TO-THIS-NIL) SURFACE ; ; (OPS::WRITE (CRLF) (TABTO 10) : 1STRP==SELECTION=RULE=WAS FIRED : (CRLF) (TABTO 10) : AND SURFACE : <SURF-LABEL> : IS CHOSEN : (CRLF) (TABTO 10) : AS "FIRST-REFENCE-PLANE : (CRLF)) (MODIFY <1> SURFACE-LABEL <SURF-LABEL> **^PART-NAME** <PART-NAME> **^SURFACE-TYPE** PLANAR [^]STATUS <STATUS-VAL> SURFACE-OUALITY <S-QUL-VAL> DIRECTION-OF-EXTERNAL-NORMAL -Z **^IS-PARALLEL-TO-SURFACES** <PARA-SURFS> [^]PARALLELISM <PAR-VAL> **^IS-PERPENDICULAR-TO-SURFACES** <PERP-SURFS> [^]PERPENDICULARITY <PER-VAL>) THE 1STRP WAS CREATED AND ASSIGNED OPS::WRITE (CRLF) (TABTO 10) : TO "WM" (CRLF) (TABTO 20) (CRLF))) : :

TABLE 3. FIRST REFERENCE PLANE SELECTION RULE AND ITS CONDITION ELEMENTS

PLATE FIXTURE			
ITEM-	-NAME	QUANTITY	
SUPPORT	FOR 1STRP	3	
ST-EL-CUDE	ST - EL - / / /		
	FLAI-IUF-SUFFURI	2	
ST_FL_CODE	ST_EL_2288	2	
TYPE	FLAT_TOP_SUPPORT		
SUPPORT	FOR 3RDRP	1	
ST-EL-CODE	ST-EL-3144	-	
TYPE	ADJUSTABLE-SUPPORT		
ELEMENTS F(OR CLAMPING		
CLAMP-TYPE	SLOTED-STRAP	4	
NUT-TYPE	SPHERICAL-NUT	4	
WASHER-TYPE	SPHERICAL	4	
SCREW-TYPE	CAP-SCREW	4	
SPRING-TYPE	COMPRESSION-SPRING	4	
FIXTURE-BA	SE*		
BASE-ID-CODE	B-458		
BASE-MADE-OF	CAST-IRON		
NO-OF-BASES-REG	QUIRED	1	
TOTAL-ITEMS		27	

TABLE 4 TYPICAL OUTPUT DISPLAY

-- -- --



FIG. 1 NODE-AND-LINK REPRESENTATION OF PRODUCTION TOOL.



FIG.2 NODE-AND-LINK REPRESENTATION OF CLAMP.







A bracket-support assembly



b A strap clamp









A clamping unit assembly



Fig. 4 GRAPHICAL REPRESENTATION OF TYPICAL STANDARD FIXTURE ELEMENTS USING A CAD SYSTEM

PAPER 68

.

A COMPARISON OF ORTHOGONAL TRANSFORMS IN ENGINEERING COMPUTER VISION

AL-KINDI, G.A.H., BAUL, R.M. AND GILL, K.F.

A comparison of a number of commonly used orthogonal transforms, when applied to the recognition and visual inspection of engineering components, has been made.

The impact on the performance and computational time for the machine vision process, due to varying numbers of transform coefficients is assessed.

Department of Mechanical Engineering The University of Leeds 1. INTRODUCTION

Orthogonal transforms are well established mathematical tools that have been successfully employed in engineering, typically in the fields of surface texture estimation, process identification and communications.

More recently, image data processing has employed orthogonal transforms for the purposes of object recognition and feature extraction. In particular, published work [1] based on the Fourier transform has shown that only a small number of normalized descriptors are necessary for reliable object identification. This paper examines a number of alternative transforms and a comparison has been made in order to assess their relative suitability for machine vision applications.

It is not always clear how many transform coefficients are necessary for an object to be identified with confidence. In this paper contours are reconstructed from inverse formula using varying numbers of coefficients. The reconstruction of a contour is a sufficient condition for the existence of a finite set of transform coefficients that unambiguously represent a specific object boundary.

2. REPRESENTATION OF OBJECT'S BOUNDARY AS A DISCRETE SERIES

The vision system used in this work contains a standard monochrome camera type LINK 109A, and a, MATROX Qfg-01/8, frame grabber which is capable of digitizing images into a 256 square array whose elements can take one of a possible 256 numerical values (grey levels).

A PDP 11/23 microcomputer is used to control the frame grabber and to send the digitized images to a VAX 11/780 computer which is used to carry out the analysis. Figure (1) shows a block diagram of the system hardware.

Because it is possible to recognise many engineering components from their silhouette, establishing the x,y-coordinate values of object boundary pixels allows a contour to be readily extracted from image data.

The usual preprocessing operations performed on image data prior to the extraction of boundary coordinate values are illustrated in Figure (2).

3. DISCRETE ORTHOGONAL TRANSFORMS

Transform theory has played a key role in image processing and it continues to be a topic of interest in theoretical as well as applied work in this field [2].

To use the transforms in the processing of digital signals, transform pairs must be defined in discrete form and these are presented in the Appendix.

In general, orthogonal transform pairs can be described by the following expressions:

$$F(u) = \sum_{\substack{m=0 \\ m=0}}^{L-1} f(m) FTK(m,u)$$

$$f(m) = \sum_{\substack{u=0 \\ u=0}}^{L-1} F(u) ITK(m,u)$$

where F(u) is the transform of f(m), FTK(m,u) is the forward transform kernel, ITK(m,u) is the inverse kernel and L is the number of pixels on a boundary.

4.

OBJECT RECOGNITION, COMPARISON AND INSPECTION USING ORTHOGONAL TRANSFORMS

The theoretical and experimental evidence available [3,4,5] indicates that a set of normalized Fourier descriptors is an effective abstract representation of an engineering object. This is because the numerical values are invariant with object rotation and size and give a unique description. Other transform descriptors exist, however, which could provide alternatives that may improve object recognition for use in machine vision. The method employed for the evaluation of the Fourier transform descriptors in reference [1] has been applied to five of the better known transforms, i.e. Cosine [6], Haar [7], Slant [8], Walsh [9] and Hadamard [10].

Fourier transforms have also been evaluated to provide a basis for comparison.

If X(u) and Y(u) represent the transform coefficients of the object's boundary coordinates, a normalized descriptor is defined as:

$$S(u) = \frac{[X(u)^{2} + Y(u)^{2}]^{\frac{1}{2}}}{[X(o)^{2} + Y(o)^{2}]^{\frac{1}{2}}}$$
(1)

Object recognition, comparison and inspection can be made with confidence using these normalized descriptors. If S1(u) represents the normalized descriptor series of one object and S2(u) that of a second object, then an "error measure" can be defined as:

$$E = \frac{1}{N} \left[\sum_{u=0}^{N-1} (S1(u)^2 - S2(u))^2 \right]^{\frac{1}{2}}$$
(2)

where N is the number of descriptors in a set and E is a measure of the similarity of any two objects. Small values of E are associated with good matching.

Comparing reconstructed objects for the range of orthogonal transform inverses listed, the error measure can be used to assess the relative effectiveness of each transform in a given vision application.

5. RESULTS AND DISCUSSIONS

Results for the two very different objects shown in Figure (3) (objects 1 and 2) were initially used for the evaluation of the effectiveness of the different transforms investigated. One is a part of a knee joint prosthesis and the other is the more common gear wheel, the latter having a more complex boundary shape. All boundary pixel coordinates are employed in the evaluation of the forward transform formulae presented in the Appendix.

In order to select the optimum number of descriptors needed for object recognition and comparison, reconstructions have been made (inverse transforms) with between 4 and 1024 coefficients from each object examined. A typical result is presented in Figure 4 for the knee joint and in Figure 5 for the gear wheel.

From the reconstructions made of the knee joint it was observed that for all the transforms examined, with the exception of the Hadamard transform, Figure 6, 128 coefficients are sufficient for recognition. In the case of Hadamard transform, 256 coefficients were required to give results that were comparable and could be used with confidence. The CPU relative times required to compute the coefficients of the transforms tested are given in Table (1). For each reconstruction the 'relative time' required for computation is viver. It should be noted that the computing time for the calculations of the coefficients obtained from the Hadamard transform is the smaller on each occasion.

In the case of the gear wheel the reconstructions indicate that 256 coefficients are necessary for acceptable recognition for all but the Hadamard transform, Figure 7, where a full set of coefficients (1024) must be used for satisfactory recognition.

Equation (2) has been applied for all similarity measurements for the transforms investigated. The changes in the value of the error measure, with variations in object angular position, are similar for all the transforms tested. Typical results with increasing numbers of descriptors to represent the object are shown in Figure 8. The exception is the Hadamard transform and the results of this transform is shown separately in Figure 9. To strengthen the results presented, additional tests were

conducted on the other objects shown in Figure 3 and these fully support the results of Figures 4 to 9 inclusive.

6. CONCLUSIONS

In this paper six orthogonal transforms have been tested for silhouette recognition and inspection. It has been shown that all transforms give adequate information and can recognise objects even when few coefficients have been employed except, however, in the case of Hadamard transform where a higher number of descriptors is needed for the recognition of an object.

In the case of Hadamord transform, the error value behaves erratically and gives no clear indication of the number of descriptors to be used in the recognition process, Figure 9. In all of the other transforms employed the error variation shows a consistent trend, typically Figure 8, and indicates the benefits to be gained with increasing numbers of descriptors.

The time for the computation of transform descriptors indicates that the Slant, Haar, Walsh and Fourier require almost the same amount of time while the Cosine transform requires twice as much time. The Hadamard transform has been shown to require the least time but the speed advantage is lost since a greater number of descriptors have been shown to be necessary for satisfactory recognition. It is left to the user to choose the right transform for each application.

REFERENCES

NIKRAVAN, B., BAUL, R.M. and GILL, K.F. 1. "The Fourier Descriptor in Closed Curve Representation for Simple Engineering Object Recognition". Faper submitted to "Computers in Industry, Elsevier Science Publication", 1989. 2. GONZALEZ, R.C., and WINYZ, P. "Digital Image Processing". Addison-Wesley Co., Reading, MA, 1977. 3. PERSOON, E. and FU, K. "Shape Discrimination Using Fourier Descriptors". IEEE Tran. PAMI-8, No.3, pp. 388-397, May 1986. 4. NIKRAVAN, B. "Image Data Acquisition, Processing and Object Recognition". Ph.D. Thesis, Mech. Eng Dept., University of Leeds, August, 1987. 5. DROSSAKIS, M. "Two Dimensional Image Processing Algorithms for use in Computer Vision Systems". MSc. Thesis, Mech. Eng Dept., University of Leeds, February 1988. 6. AHMED, N., NATARAKAN, T. AND RAO, K.R. "Discrete Cosine Transform", IEEE Tran. Comp. 1074, C-23, 90. 7. HAAR, A. "Zur Theorie der Orthogonalen Funktionsysteme" Math. Annal., 1910, 69, 331-371. 8. PRATT, W.K., CHEN, W., AND WELCH, C.K. "Slant Transform Image Coding", IEEE Tran. on Comm., Vol. Com-22, No.8, pp.1075-1093, August 1974. 9. WALSH, J.H. "A Closed Set of Orthogonal Functions", Am. J. Math. 1923, 45, 5-24. 10. PRATT, W.K., ANDREWS, H.C. AND KANE, J. "Hadamard Transform Image Coding", Proc. IEEE, Vol. 57, pp. 58-68, 1969. 11. MULVANEY, D.J., NEWLAND, D.E. AND GILL, K.F. "A Comparison of Orthogonal Transforms in their Application to Surface Texture Analysis", Proc. of the Inst. of Mech. Eng., Part C, Vol. 200. No. 6, pp.407-414, 1986. PRATT, W.K. 12. "Digital Image Processing", John Wiley and Sons, A

Wiley-Interscience Pub., 1978.

APPENDIX

Discrete Orthogonal Transform

1. FOURIER TRANSFORM

The Fourier transform is the most popular of the orthogonal transforms and this is mainly due to the elegant properties of the frequency domain representation of a large class of engineering signals (11).

The forward transform can be represented by:

$$F(u) = \frac{1}{N} \sum_{m=0}^{N-1} f(m) e \qquad u = 0, 1, 2, N-1$$

where F(u) is the transform of f(m).

The inverse of such a transforms can be calculated from:

$$f(m) = \sum_{\substack{\nu=0 \\ \nu=0}}^{N-1} F(u) e^{\frac{N}{N}}$$

2. COSINE TRANSFORM

It is known that the Fourier series representation of any continuous real and symmetric function contains only real coefficients corresponding to the cosine term of the series (12). This leads to the definition of a Cosine transform which is a sub-series of the Fourier transform. This transform can be stated as:

$$F(u) = \sqrt{\frac{2}{N}}C(u) \qquad \sum_{m=0}^{N-1} f(m) \cos\left[\frac{(2m+1)u\pi}{2N}\right]$$

where

 $C(u) = \frac{1}{\sqrt{2}} \text{ when } u = o$ and C(u) = 1 otherwise.

The inverse is:

$$f(m) = \sqrt{\frac{2}{N}} \quad \sum_{u=0}^{N-1} C(u)F(u)\cos\left[\frac{(2m+1)u\pi}{2N}\right]$$

3. HAAR TRANSFORM

The Haar function series was the first complete set of digital orthogonal functions to be described, each function taking no more than three values in the interval over which it is defined, except at a finite number of discontinuities (11). The Haar function can be expressed as

HAR(u) = 1 when u = 0

 $HAR(u) = r \text{ when } \frac{(u-q)N}{q} \le m \le \frac{(u-q)N}{q} + \frac{N}{2q}$ $HAR(u) = -r \text{ when } \frac{(u-q)N}{q} + \frac{N}{2q} \le m \le \frac{(u-q)N}{q} + \frac{N}{q}$

HAR(u) = o otherwise

where
$$u = 0, 1, 2, \dots N-1$$
 $q = 2^{INT(1 \circ q_2 u)}$

and $r = 2^{INT(\log_2 u)/2}$

The forward Haar transform is:

$$F(u) = \frac{1}{N} \sum_{m=0}^{N-1} f(m) HAR(u)$$

and the inverse is:

$$f(m) = \sum_{\substack{u=0}}^{N-1} F(u) HAR(u)$$

SLANT TRANSFORMS

4.

The Slant transform is an orthogonal transform that is designed to

have the following properties [12]:

- (i) Constant basis vector,
- (ii) slant basis vector,
- (iii) sequency property,
- (iv) fast computation algorithm and
- (v) high energy compaction.

The forward Slant transform can be computed from the formulae [12]:

$$F(u) = \sum_{\substack{m=1}}^{N} f(m)SL(u,m)$$

and the inverse formulae is:

$$f(m) = \sum_{u=1}^{N} F(u)SL(m,u)$$

where SL is the Slant matrix which is expressed as:

order 2
$$\Rightarrow$$
 SL₂ = $\sqrt{2}$ $\begin{bmatrix} 1 & 1 \\ 1 & -1 \end{bmatrix}$

$$= \frac{1}{\sqrt{4}} \begin{bmatrix} 1 & 1 & 1 & 1 \\ a_4 + b_4 & a_4 - b_4 & -a_4 + b_4 & -a_4 - b_4 \\ 1 & -1 & -1 & 1 \\ a_4 - b_4 - a_4 - b_4 & a_4 + b_4 & -a_4 + b_4 \end{bmatrix}$$



where ${\rm I}_{\rm K}$ is K*K $% {\rm M}$ identity matrix

$$a_{2} = 1$$

 $a_{2N} = \left(\frac{3N^{2}}{4N^{2}-1}\right)^{\frac{1}{2}}$ and $b_{2N} = \left(\frac{N^{2}-1}{4N^{2}-1}\right)^{\frac{1}{2}}$

79 -
5.

WALSH TRANSFORM

The Walsh transform is the most popular of the square wave transforms, because it is computationally similar to the more familiar Fourier transform [11].

The Walsh function takes only two values, except at a finite number of discontinuities. It is defined as:

$$WAL (m,u) = \frac{1}{N} \prod_{i=0}^{n-1} (-1)^{i(m)b} n - 1 - i^{(u)}$$

where $b_k(z)$ is the kth bit in the binary representation of z and $N = 2^n$.

The Walsh transform is given by:

$$F(u) = \frac{1}{N} \int_{m=0}^{N-1} f(m) WAL(m,u)$$

and the inverse is:

$$f(u) = \frac{1}{N} \begin{array}{c} N-1 \\ \Sigma \\ u=0 \end{array} F(u) WAL(u,m)$$

6. HADAMARD_TRANSFORM

The Hadamard transform is based on the Hadamard matrix which is an array of plus and minus ones, whose rows and columns are orthogonal. The forward Hadamard transform is given by:

$$F(u) = \frac{1}{N} \begin{array}{c} N-1 \\ \Sigma \\ m=0 \end{array} \begin{array}{c} n \overline{\Sigma}^{1} \\ b_{i}(m)b_{i}(u) \\ \vdots \\ m \\ m \\ m \\ n \end{array}$$

and the inverse by:

$$f(m) = \frac{1}{N} \begin{array}{c} N-1 \\ \Sigma \\ u=0 \end{array} F(u) (-1) \\ n \overline{\Sigma}^{1} \\ i \overline{\Sigma}^{0} \\ i \overline{\Sigma}^$$

where

 $b_k(z)$ is the kth bit in the binary representation of z and N=2ⁿ.

NUMBER	EVALUATION TIMES					
COEFFICIENTS	FOURIER	COSINE	HAAR	SLANT	HADAMARD	WALSH
512	0.16	0.26	0.15	0.16	0.08	0.15
256	0.07	0.13	0.08	0.08	0.04	0.08
128	0.04	0.06	0.03	0.04	0.03	0.04
64	0.02	0.03	0.02	0.02	0.02	0.02
32	0.01	0.02	0.01	0.01	0.01	0.01
16	<0.01	0.01	0.01	0.01	0.01	0.01
8	<0.01	0.01	0.01	<0.01	<0.01	0.01
4	<0.01	<0.01	<0.01	<0.01	<0.01	<0.01

.

Table 1. RELATIVE CPU TIMES FOR COMPUTING TRANSFORM COEFFICIENTS



•

,

.





Fig. (2) Preprocessing operation









Fig. (4) Reconstruction of simple engineering component using finite Haar transform coefficients

. .



Fig. (5) Reconstruction of a gear wheel using Walsh transform



0.08 unit time

128 coefficients 0.12 unit time

16 coefficients 0-04 unit time

o 32 coefficients 0∙04 unit time

4 coefficients less than 0.04 unit time

8 coefficients less than 0.04 unit time

. -

Fig. (6) Reconstruction of simple engineering component using finite Hadamard transform coefficients



Fig. (7) Reconstruction of a gear wheel using Hadamard transform

ţ







Fig. (9) Error Measure-Hadamard transform

PAPER 69

MONITORING AND ASSESSMENT OF ENGINEERING SURFACE TEXTURES USING COMPUTER VISION

ALKINDI G.A., BAUL. R.M. AND GILL K. F.

ABSTRACT

A number of problems associated with texture measurement using vision systems are highlighted and some progress towards a model for surface texture measurement has been made. It is shown that the examination of engineering surfaces using machine vision can give a reliable measure of surface texture based on space. Parameters based on amplitude are less certain.

Mechanical Engineering Department, University of Leeds

1 Introduction

The ability to carry out on-line measurement and real time assessment of a component surface would enhance the control of a wide range of manufacturing processes and allow current performance to be estimated.

Currently the most popular methods of surface assessment involve stylus techniques [1], which have known limitations. Other methods available include comparator specimens, pneumatic gauging, interference microscopy and electrical resistance measurements [2].

The application of digital techniques and random process analysis has led to a significant increase in the number of parameters employed in surface measurement based on classical statistics [3].

An engineering surface will, in general, depart from the ideal envisaged because of limitations in the geometrical accuracy of machine tools, the nature of the process involved, material and environmental factors. Errors of form and surface texture are viewed as two dimensional space varying quantities, which can contain periodic and random elements in varying proportions [4]. In most surface measurements the information is separated into two components namely waviness and roughness; usually achieved by filtering techniques. In addition the generation of a reference line is basic to all approaches, Table(1).

Three basic types of texture parameter exist and each roughness parameter has an equivalent in the waviness regime. These are summarized in Table(1)

1. Roughness Amplitude parameters which measure the vertical characteristics of surface topography, e.g. R_a and R_z values.

2. Roughness Spacing parameters which measure the horizontal characteristics of surface topography, e.g. S and S_m values.

3. Roughness Proportion parameters which, indirectly, combine both amplitude and spacing characteristics of a surface, e.g. δq .

2 Machine Vision Approach

Although many researchers have published material in the area of texture using machine vision [5], few have measured texture parameters for manufactured components. The main area of interest has been the texture of geographical features of the countryside [6].

The statistical methods employed in the vision approach are based on classical properties e.g. mean, variance and skewness of the data collected [7,8]. Fourier series and autocorrelation function have also been used, but less frequently [9,10]. Recent surveys on texture models and experimental techniques using vision can be found in references [5,11,12,13].

In this paper the capability of a machine vision system to monitor texture is explored experimentally because surface colour, lighting conditions, surface texture, surface geometry and tarnishing will all create measurement problems.

A block diagram for the system used in this work is presented in Figure(1). Images are stored in a 256×256 pixel array, having a grey level value in the range $0 \rightarrow 255$.

Specimen surfaces are aligned so that the lay is vertical. A horizontal row of pixels therefore corresponds to the track normally employed in stylus measurements.

A camera position of 90 degrees from the horizontal, and controlled diffused lighting were found to give the best illumination for the engineering surfaces studied.

3 Initial Assessment of The Vision System

The initial tests conducted by these authors was for a range of surface roughnesses normally encountered in high precision manufactured components (R_a in the range 0.4 to 1.0 μ m). Comparisons between vision grey level data and conventional stylus results, however, in both the spatial and frequency domain were severely limited because of the magnification of the camera employed and the resolution of surface detail.

Attention was therefore focused on rougher surfaces (R_a 6 to 100 μm), where the available magnification was adequate.

Four sets of standard tactile comparison specimens (Microsurf), Figure(2), each containing four samples with specified R_a values, were selected and examined using computer vision.

Preliminary tests were carried out to investigate the effect of lighting. Results showed clearly that light intensity and direction has a significant effect on grey level values, but not on the spacing between peaks. Changing the position of the sample within the field of view had only minimal effect on both the amplitude and spacing between peaks. These preliminary results indicated that texture parameter data could be obtained by vision and that this data might be useful to the practitioner if correctly interpreted and employed.

To investigate the degree of similarity and to determine the possibility of establishing a calibration between vision and stylus data, a series of test was undertaken that would allow a comparison to be made between vision and stylus data.

The computer images of the selected surfaces were examined and the number of peaks per unit length were estimated for both the vision and stylus techniques. A good correlation for this spacing parameter was observed. In the case of amplitude, however, an increase in roughness or stylus amplitude did not necessarily lead to a corresponding increase in vision amplitude. Typical results are presented in Figure (3) for a range of camera gains.

The tests confirm the feasibility of a calibration for the spacing parameters, but not for the amplitude parameters. This does not rule out the possibility of employing a control specimen with the vision method for specific manufacturing processes. Aliassing is avoided by ensuring four pixels per cycle are monitored at the highest frequencies. This is confirmed in Figure(4) for cylindrical turned surfaces.

4 Data Processing and Analysis

Earlier work on shape from shading by these authors [14] has shown that image intensity is influenced by the distance from the pixel to the optical axis taken through the image plane. All other effects, e.g. colour differences, lighting distribution speckles and tarnishing, were small and could be neglected. In this present study, however, these effects can not be ignored because of the higher magnification required for texture assessment.

The principal difference between the current vision approach and the stylus method is that the later monitors almost pure amplitude data whereas vision data can be corrupted by other surface properties unique to vision, e.g. colour and lighting. In addition the stylus method employs a filter, a skid and a cut-off value to separate waviness from roughness. The data collected and the reference lines calculated are therefore based on filtered signals. To allow a fair comparison, it is necessary to filter the raw vision data to isolate roughness from waviness and to remove the corruption caused by the non texture features.

The normal expression used in the calculation of the base line vector employed in stylus studies is

$$\overline{x} = \frac{1}{N} \sum_{j=1}^{N} x(j) \tag{1}$$

where x(j) is a sampled value and N is the number of samples. Typical results

for vision using Eq.1 are shown in Figure(4.a) for a range of specimens. The reference line obtained does not always follow the direction of peaks or their envelope because of the corruption in surface data caused by the non texture features.

The authors therefore propose an alternative approach in order to compute a more reliable reference line for the vision data.

4.1 Reference Line Computation for vision data

This approach averages the grey level values, in groups of 5 pixels and the result is centered on the middle pixel. The initial filtering expression employed is:

$$\overline{x}(j) = \frac{1}{5} \sum_{k=j-2}^{j+2} x(k) \tag{2}$$

where x is the sample vector containing the grey level values and j = 3, 4, 5, ..., N-2.

Repeated application of Eq.2 for each grey level value, is made for a total of $\frac{N}{4}$ applications. In the successive applications the filtered values are used in preference to measured values. Equation 2 can then be rewritten as:

$$\overline{x_i}(j) = \frac{1}{5} \sum_{k=j-2}^{j+2} x_{i-1}(j)$$
(3)

where $i = 1, 2, 3, \dots, \frac{N}{4}$ and $j = 3, 4, 5, \dots, N-2$.

The raw image can be corrected to the new datum by application of

$$\overline{x_c}(j) = x(j) - x_{\underline{N}}(j) \tag{4}$$

where $\overline{x_c}$ is the corrected data, x is original data and $x_{\frac{N}{2}}$ is the filtered data.

Final processing to produce equal areas on both sides of the reference line was carried out to produce a reference line equivalent to that normally obtained from stylus measurements, Figure(4.b).

A comparison between normalized stylus and corrected vision data, Figure(5), shows a mean error of less than 5.7% with a standard deviation of 7.5%of the mean value. The maximum mean error found for all surfaces examined was 14.5% and maximum standard deviation was 7.5%

The slopes of the amplitude probability density function (APDF), Figure(6), for both the stylus and vision approach are very similar for each roughness considered. The estimation of absolute amplitude using vision is not possible because the grey level range, as shown in Figure(6), does not follow a linear relationship over the range of roughness considered. Parameters outside normal control, typically variations in colour and tarnishing, could be significant factors influencing this non-linearity.

To test the suitability of vision for on-line application, four samples were prepared using the same material and manufacturing process and R_a values were computed from both vision and stylus data. For each sample, 6 tracks separated by 2mm, were measured and the results are shown in Figure(7). Although no linear calibration can exist between stylus and vision results, as shown earlier, the scatter in vision was found to be no more serious than that of the stylus. This suggests that vision amplitude data could be usefully employed to detect changes occurring in a specific metal cutting process, where machining parameters should remain sensibly constant. Noting that the vision method provides field data as opposed to serial data from a stylus traverse, the on-line texture monitoring using vision could provide a fast and reliable assessment of a workpiece surface over a given area rather than a single track.

4.2 A Proposal for a Visual Texture Measurement Parameter

It has been shown that a reliable measure for the spacing parameter can be obtained using computer vision. However, the spacing parameter gives no information about peak shapes, slope, or amplitude, and consequently its use in practical applications is limited if it is not supported by additional information.

A vision parameter which combines both space and amplitude features could provide a solution. A parameter of this type can be defined as:

$$S_v = \frac{C \times h}{d} \tag{5}$$

where C is the number of reference line crossings, h is the normalized height of peaks and d is the traverse distance.

To gain some positive evidence to support the use of the S_v parameter, it was evaluated for both vision and stylus data. The results showed variations of 15% or less, demonstrating S_v could give an absolute measure compatible with that obtain by stylus.

To maintain the performance indicated above, the raw vision data should be filtered to minimize colour differences and waviness. The estimation approach adopted is summarized in Figure(8) for the evaluation of parameter S_v .

4.2.1 Real Application of Parameter S_v

A possible application of the S_v parameter is in monitoring surface texture of samples used in standard tribology tests. The tribologist employs the tests to



determine the load carrying capacity of lubricating fluids and is widely used for the specification of lubricants [15].

4.2.2 Visual Inspection

Figure (2) shows a number of standard pieces after testing. Inspection of the surface wear is usually done by human eye and the operator must decide, by experience, whether the lubricant has passed or failed by visual assessment of the texture in the wear zone.

4.2.3 Machine Inspection

Figure (9) show images taken for a number of standard test pieces after wear. It can be noticed that the affected areas can be easily distinguished and a percentage measurement of that area can be obtained. Results of area measurement are presented in Table(2). Although the area is an important parameter, it does not give any indication about the resulting surface texture and possible scoring. Hence an estimation of the S_v parameter has been carried out, and the results are also included in Table(2). Although the parameter S_v is obtained numerically its significance can only be determined if available subjective information is also supplied. If the results are representative and a correlation can be established, the test procedure may be worthy of automation and operator judgement minimized in the testing.

5 CONCLUSIONS

This experimental work, in the field of machine vision, extends that already published by the authors for the evaluation of component profiles. The results presented are based on measurements of more than 30 different specimens and therefore represent a reliable indicator of the importance of machine vision in engineering texture measurement. Although the tests have been limited to a range of surface roughness which exceed those normal found in precision manufacturing, surface monitoring employing computer vision has been shown to have potential as an on-line proximity technique. The examination of smooth surfaces will be possible when higher camera gain and increased pixel resolution are commercially available.

One important observation made is that the scatter for the vision surface measurements is no worse or no better than that of the equivalent stylus measurements and reflects the inherent variations in surface topography from one part of the surface to another. The roughness proportion parameter for the vision system S_v has been shown to have the most merit, since it combines both space and amplitude data. The application of the technique to the objective assessment of a wear track in the evaluation of lubricants has been encouraging. The vision system has been shown capable of measuring both the area of the wear scar and the surface texture using S_v . Deep scratches can also be detected if present thus enabling a complete automation of the test.

REFERENCES

- [1] RICHARDS, P. J. 'A Review of Methods of Measurement and Assessment of Surface from Texture' Conf. on Properties and Metrology, Oxford Instn. Mech. Engrs., Paper 41, 1968, pp. 1-13.
- [2] MULVANEY, D. J. 'Identification of Surface Texture by Orthogonal Transform Analysis' Ph.D thesis Mechanical Engineering Department, University of leeds, 1983.
- [3] MULVANY, D.S, NEWLAND, D.W. and GILL, K.F. "Identification of Surface Roughness" Proc. Instn. Mech. Engrs., Vol.199, C4, 1985, pp. 281-286.
- [4] HUME, K.J. "Engineering Metrology" McDonalds and Co.Publisher, 1963.
- [5] PRATT, W. K. 'Digital Image Processing' Wiley-Interscience Publisher, 1978.
- [6] BALLARD, D.H. and BROWN, C.M. "Computer Vision" PrenticeHall Inc. Publisher, 1982.
- [7] COHEN, F. S. and FAN, Z. "Rotation and Scale Invariant
 Texture Classification" Proc. of IEEE. Int. Conf. on
 Robotics and Automation, 1988, pp. 1394-1399.
- [8] NIKRAVAN, B. "Image Data Acquisition, Processing and Object

Recognition for use in Machine Vision System" Ph.D thesis, Mech. Eng. Dept., Univ. of Leeds, 1987.

- [9] ROSENFELD, A. and TROY, E. 'Visual Texture Analysis' TR.70-116, Univ. of Maryland, College Park, Maryland, June, 1970.
- [10] LENDARIS, G. G. and STANLEY, G. L. 'Diffraction Pattern
 Sampling for Automatic Pattern Recognition' Proc. IEEE, 58,
 2, pp. 198-216, 1970.
- [11] EOM, K. B. and KASHYAP, R. L. 'Texture and Intensity Edge Detection With Random Field Models' Proc. Comp. Soc., Workshop on Computer Vision, 1987, pp. 29-34.
- [12] Van Gool, L., Dewaele, P., and Dosterlinck, A. 'Texture Analysis anno 1983' Comp. vision, Graphics and image processing, Proc. 29, 1985, pp. 336-357.
- [13] TEAGUE, E. C., VORBURGER, T. V. and MAYSTRE, D. ''Light Scattering from Manufactured Surfaces'' Annals of CIRP, vol.30, 2, 1981, pp.563-569.
- [14] ALKINDI, G. A. H., BAUL, R. M. and GILL, K. F. 'Experimental Evaluation of Shape from Shading for engineering component Profile Measurement' Proc. Instn. Mech. Engrs., vol.203, B4, 1989, pp. 211-216.
- [15] Inst. of Petrolium, London, 'Method for Analysis and Testing, IP Standards for Petrolium and its Products' Inst. of Petrolium, London, Designation IP 240/84, 1984.



Table 1 Surface texture parameters.

107

Test-Piece	Results of Operator	Area%	Numerical value	
No.	Judgement		of S_v	
1	Failed	30.6%	4.795	
2	Failed	44.6%	4.688	
3	Passed	10.1%	6.180	
4	Failed	20.2%	4.832	
5	Passed	8.2%	6.512	
6	Failed	27.4%	4.582	
7	Passed	12.2%	6.316	
8	Failed	23.1%	4.911	
9	Failed	33.4%	4.612	
10	Failed	39.2%	4.331	
11	Failed	22.8%	4.112	

Table 2 Evaluation of the parameter S_{v}

in real application







Figure 2 Specimens used in this study.



Figure 3 Comparison between Vision and Stylus R, values using different camera gains.





112

•



Figure(5) Typical Stylus and Vision data

113

Stylus data

.............

Vision data Comparison error




Figure 7 Comparison between Stylus and Vision R. Values.







c. Specimen No.3 (Passed)

j.

d. Specimen No.4 (Failed)

Figure 9 Computer images of specimens used.



System Modelling and Control

Second Edition

J. Schwarzenbach K. F. Gill

Mechanical Engineering Department University of Leeds





6 JUL 1992

C J. Schwarzenbach and K. F. Gill 1984

First published in Great Britain 1978 by Edward Arnold (Publishers) Ltd, 41 Bedford Square, London WC1B 3DQ

Edward Arnold, 3 East Read Street, Baltimore, Maryland 21202, U.S.A.

Edward Arnold (Australia) Pty Ltd, 80 Waverley Road, Caulfield East, Victoria 3145, Australia

Reprinted with corrections 1979 Second edition 1984 Reprinted 1986

ISBN 07131 35182

All rights reserved. No part of this publication may be reproduced, stored in a retrieval system, or transmitted in any form or by any means, electronic, mechanical, photocopying, recording or otherwise, without the prior permission of Edward Arnold (Publishers) Ltd.

Typeset and printed in Great Britain at The Bath Press, Avon

Preface to First Edition

Since the early 1940s, the number of practical applications of the principle of feedback has grown rapidly and the range of application has become very wide, with the consequence that an increasing proportion of engineers, scientists, and technologists require a basic appreciation of the fundamentals of automatic control theory. As the requirements for system dynamic performance have become more exacting so also have the demands on the engineer. Many who are new to the subject find feedback control theory difficult to understand, largely because of the rather abstract nature of some of the concepts involved. Of the textbooks available many are so comprehensive in their coverage that they are more suited to the reader who already has some understanding of the subject rather than to the beginner. The sheer volume of theoretical material tends to discourage the latter type of reader, and the detail often obscures the significance of the main principles.

The primary objective in writing this book has therefore been to sift from the large volume of literature on control theory the material believed to be most pertinent to industrial practice, and to present it in such a way that the student or practising engineer can attain a sound physical understanding of the basic principles of control. Familiarity with the material presented in the book will enable the reader to converse with specialists in the field, to design simple control loops adequate for many industrial applications, and with the aid of more advanced texts to design more complex control schemes. The aim throughout has been to present the fundamental theory in such a way that the reader can see the practical relevance of the material and that he can build up a clear mental picture to aid understanding. The mathematical manipulations can readily be mastered with practice; understanding the significance of the procedures and of their results is the real problem.

The principles of feedback control theory are very general; thus the topic is broadly based and is of relevance for a wide range of dynamic systems. The main variation in the potential areas of application lies in the differing characteristics and complexity of the systems to be controlled. Electrical engineers probably have the least difficulty in understanding control theory since many of the concepts are relevant to their other areas of study. They probably also face the fewest problems of application since the systems with which they deal, although often complex, are well defined because of the discrete lumped nature of most components and of the ease of measurement of system variables.

Many existing textbooks are intended primarily for the electrical engineer. The method of approach used in this book should make it particularly useful

iv Preface

for mechanical engineers, chemical engineers, and other technologists and scientists (and in part also for life scientists, economists and others with an interest in the dynamic behaviour of systems and in the concept of feedback). The main problem in designing a control loop for non-electrical systems normally arises when attempting to obtain an adequate mathematical model for the system since, in general, components cannot readily be represented by simple discrete ideal elements; often non-linearities are dominant, measurement is difficult, and noise is significant.

In our teaching of control to undergraduates, postgraduates, industrial engineers and non-engineers we have experimented with the method of approach. the topics included, and the order of presentation. The approach that has evolved, and appears to be the most effective, forms the basis for this book. We have found that the 'classical' approach based on the transfer function and associated techniques of analysis is more easily comprehended and related to practice by the beginner than is 'modern control theory' which is based on a state space approach. The emphasis in this book is accordingly on classical linear control theory. Some understanding of the ideas of the state space approach and its relationship to the classical approach is nevertheless highly desirable; hence a chapter is included to introduce the reader to the more advanced theoretical procedures which have been developed over the last decade or so and which are particularly useful for the mathematical analysis of multivariable systems. The material is presented in such a way as to make the transition from the classical to the modern approach as smooth as possible. With regard to the order of presentation of material we have found definite advantage in analysing in some detail the dynamic behaviour of components of systems in both the time and the frequency domain prior to any detailed consideration of a closed loop system. This gives the student a clear awareness of the nature of the dynamic response of a system component and how the response varies with the form of the transfer function and the input excitation. It shows him how the response to any given input can be calculated from a knowledge of the transfer function and conversely how a transfer function can be determined by practical testing of a system component. The latter, the process of system identification, is used for verifying mathematical models derived theoretically and may be the only means, or the easiest means, of obtaining a transfer function representation where theoretical derivation is difficult. When this foundation has been laid the principle of feedback can be introduced and rapid progress made in analysing the dynamic behaviour of closed loop systems and, in particular, how accuracy and stability are affected by components within the loop. This then leads logically and easily to the most important stage, consideration of the design of feedback control systems to meet specific dynamic performance specifications.

The material presented in this book should cover the control engineering content of most undergraduate degree schemes which include the subject of automatic control. The book should be of equal value to the engineer in industry who did not include control in his studies but who is now faced with having to deal with some aspect of control or to communicate with others working in the field. It is suggested that at a first reading (or where a minimum of time is available) Chapters 5 and 7 can be omitted without detriment to the

understanding of later material. Where time is not available for these chapters it is nevertheless recommended that the reader tries to gain a general idea of the contents.

We wish to express our thanks to those who have contributed most to the development of this book—the many students of differing backgrounds who by their attempts at learning control theory have highlighted points of particular difficulty in understanding. We are especially indebted to our colleague Mr. J. L. Douglas who has endeavoured to learn the fundamentals of the subject by using this book for self-teaching. In doing so he has made valuable suggestions which have enabled us to remove some of our errors and ambiguities, and to make minor additions where our steps have been rather large. We thank also Mrs M. Fernando for her valuable contribution of a neatly typed text.

Leeds 1978 J. Schwarzenbach K. F. Gill

Preface to Second Edition

It is encouraging to find that this book appears to a large extent to have attained its primary objectives and that it is therefore helping to meet the need for textbooks which explain simply and clearly the basic fundamentals of control engineering. In the period since it was first published digital computers and microprocessors have come to play a very prominent role both as control system components and as tools for analysis, and the main purpose of this second edition is to introduce supplementary material to reflect this change. Sections have been incorporated describing digital simulation and simulation languages, Section 7.5 has been rewritten to include digital computation of correlation functions and power spectra, and a new chapter has been included to deal with the analysis of discrete data systems. The important topic of design has also been given greater emphasis by expanding the final chapter. Solutions to the problems have been included, and the bibliography has been updated.

Leeds 1984 J. Schwarzenbach K. F. Gill

Contents

Preface	-	
List of	principal	symbols

1	Introduction	1
1.1	What is a system?	1
1,2	System control	3
1.3	The need for analysis	4
1.4	Methods of system representation	4
1.5	Methods of analysis and design	6
2	Mathematical Description of System Components	10
2.1	Linearity of systems	10
2.2	Laplace transforms and their significance	14
2.3	Transfer functions and the characteristic equation	16
2.4	Transfer functions for some simple elements	18
2.5	Effect of secondary factors on transfer functions	23
2,6	State equations	27
3	Analogue Computers and System Simulation	29
3.1	Analogue and digital computers	29
3.2	Basic analogue computer elements (linear)	30
3.3	Production of circuit diagrams to solve differential equations	33
3.4	Computer operating modes	38
3.5	Non-linear analogue computer components	40
3.6	Differentiation	41
3.7	Problem scaling	42
3.8	Digital continuous system simulation	47
3.9	Simulation languages	53
4	Transient Response of Systems	60
4.1	Response of first order system to step, ramp, or impulse function	61
4.2	Response of second order system to step, ramp, or impulse function	64
4.3	Transient response of third and higher order systems	68
4:4	Performance characteristics (time domain)	69
4.5	Step response testing of practical systems	72

viii	Contents	
4.6	Comparison of transient forcing functions	76
4.7	The convolution integral	77
5	State Space Representation and Analysis	80
5.1	State variable diagrams	81
5.2	Generalized state equations	86
5.3	State relations in the s-domain	88
5.4	Solution of the state vector differential equation	90
5.5	Discrete time model	94
6	Frequency Response of Systems	99
6.1	The transfer function in the frequency domain	100
6.2	Polar plots	101
6.3	Bode plots	106
6.4	Frequency response testing of practical systems	114
6.5	Frequency domain performance criteria	117
7	Statistical Methods for System Identification	119
7.1	Correlation functions	120
7.2	Dynamic testing using correlation techniques	126
7.3	Power spectral density	130
7.4	System identification using spectral density functions	134
7.5	Digital evaluation of correlation functions and power spectral density	137
7.6	Pseudo random binary sequences (PRBS)	142
7.7	Illustrative example	147
8	Feedback Systems—Accuracy and Stability	152
8.1	Closed loop or feedback control	152
8.2	Steady state error	155
8.3	Routh-Hurwitz stability criterion	157
8.4	Nyquist stability criterion	161
8.5	Gain margin and phase margin	163
8.6	Loci of constant closed loop magnitude and phase	165
9	The Root Locus Method	170
9.1	Root locus plots	171
9.2	Construction of root loci	174
9.3	Aids to construction of root locus diagram	177
9.4	Interpretation of the root locus diagram	182
9.5	Root contours	189
10	The Sampled-Data Process	194
10.1	Mathematical description of sampling process	195
10.2	Transfer function of sampled-data element	196
10.3	Closed-loop transfer function	197
10.4	Polar plots	199
10.5	Pulse transfer function	201

			Contents	ix
10.6	Block	diagrams		204
10.7	Inverse	e operation		206
10.8	Data r	econstruction		207
10.9	The z-	plane		213
10.10	Routh	Hurwitz stability test	3	214
10.11	Freque	ency response of sampled-data system	12	218
11	Design	of Closed Loop Systems		222
11.1	The ge	neral approach to design		222
11.2	Propor	rtional control		224
11.3	Integra	and derivative action		226
11.4	Selecti	ng controller settings on existing process plant		232
11.5	System	compensation	. 13	234
11.6	Phase	lead series compensation	1.2	236
11.7	Phase	lag, and lag-lead series compensation		241
11.8	Pole ca	ancellation, and feedforward compensation	19	247
11.9	Comp	ensation of sampled-data system	1.0	249
11.10	State v	ector feedback control		254
11.11	Relay	13	263	
11.12	Case st	udy of electrohydraulic servomechanism		265
Apper	dix A	Problems		279
	B	Bibliography		297
	C	Introduction to matrix algebra		300
	D	Answers to problems		304
Index				317

List of Principal Symbols

Symbols which appear in one part of the book only, and whose meaning is clear from the accompanying text or figures, have been omitted from this list.

Variables which are functions of time are normally represented by lower case letters, and the Laplace transforms of the variables are normally represented by the corresponding capital letters e.g. F(s) is the Laplace transform of f(t), i.e. $F(s) = \mathcal{L}[f(t)]$.

Subscripts are used where more than one of a given variable is used, e.g. $f_1(t), f_2(t), \ldots, F_1(s), F_2(s), \ldots$

Variables are assumed to be relative to appropriate datum or design values (system components are normally assumed to be linear).

Starred symbols are used to indicate time functions that are in sampled form e.g. $f^*(t)$, the sampled version of f(t), is the series of values f(0), f(T), f(2T), $f(3T) \dots$; it has Laplace transform $F^*(s)$ and z transform $F(z) = \mathscr{Z}[f(t)]$.

u(t) = U(s)	Input signal to system or system component
r(t) = R(s)	Reference input to (or set point of) feedback system
c(t) = C(s) v(t) = Y(s)	Output signal from system or system component
e(t) = E(s)	Error signal
n(t) N(s)	Noise signal
m(t) M(s)	Manipulated variable, output from controller
$s = \sigma + j\omega$	Laplace operator, (real part σ , imaginary part ω)
G(s)	Transfer function of component of a system
H(s)	Transfer function of component in a feedback path
$\delta(t)$	Unit impulse function
g(t), or $w(t)$	Unit impulse response, or weighting function, $\mathscr{L}^{-1}[G(s)]$
<i>G</i> (jω)	Transfer function with $s = j\omega$; gives harmonic characteristics of system, i.e. $ G(j\omega) = \text{magnitude}, \ \angle G(j\omega) = \text{phase angle } \varphi$ of output relative to input for input frequency ω
p_1, p_2, p_3, \ldots	Poles of a transfer function (factors of denominator) i.e. roots of the characteristic equation
21, 221 23,	Zeros of a transfer function (factors of numerator)
K	Gain constant
r	Time constant of a first order component (in Chapter 7, τ represents time shift of a signal)
ζ and ω_n	Damping factor and undamped natural frequency, respec-

	tively, for a second order system component (or associated with a pair of complex conjugate roots)
m.	Sampling frequency
T	Sampling interval, $T = 2\pi/\omega$.
z	Alternative to Laplace operator used with sampled signals, $z = e^{sT}$
G(z)	Pulse transfer function
$K_{\rm p}, K_{\rm v}, K_{\rm a}$	Positional, velocity, and acceleration error coefficients respec- tively
M_p and ω_p	Peak magnification of a closed loop system, and the frequency at which it occurs
M	Closed loop magnification
k_1, k_2, k_3 k_1, T_1, T_1	Coefficients of P + I + D controller
$\{\mathbf{x}(t)\}$	State vector, comprising the <i>n</i> state variables $x_1(t)$, $x_2(t)$, $x_3(t) = x_1(t)$
{v(t)}	Response or output vector
(u(t))	Control or input vector
A	Coefficient matrix
B	Driving matrix
С	Output matrix
D	Transmission matrix
1	Unit matrix
$\varphi(t)$	Solution matrix or transition matrix
$\varphi_{xx}(\tau)$	Autocorrelation function of a variable $x(t)$
$\varphi_{xy}(\tau), \varphi_{yx}(\tau)$	Cross correlation function of a pair of variables $x(t)$, $y(t)$
$\Phi_{xx}(\omega)$	Power spectral density of a variable $x(t)$, i.e. the Fourier transform of $\varphi_{-x}(\tau)$
$\Phi_{xy}(\omega)$	Cross spectral density, Fourier transform of $\varphi_{xy}(\tau)$

1 Introduction

As technological processes increase in complexity, and the required performance specifications become more severe, analytical design procedures assume great importance. It has become essential for engineers to have an understanding of the nature of the dynamic behaviour of systems, and of the methods available for analysing and improving dynamic performance.

These requirements are making the use of mathematical modelling techniques an essential part of design. The nature of the model and the methods employed in obtaining it are dependent on the depth of understanding needed at a particular stage of the design study, and on the use to which the model will be put.

It is hoped that this book will give to both the student and the practising engineer a clear insight into the main facets of system modelling, linear control theory, and control system design, and that it will form a sound foundation for practical application or more advanced study. The level of mathematical knowledge assumed is a familiarity with simple differential equations and with complex numbers.

1.1 What is a system?

It is desirable first to define what is meant by a system, a word which is frequently used in conversation. Broadly, a system can be thought of as a collection of interacting components, although sometimes interest might lie just in one single component. These components will often be discrete physical elements of hardware, but can equally well be functional parts of such physical components. The system of interest might be a power station, a steam turbine in the power station, or a control valve on the turbine; it might be an aeroplane, its air conditioning, an engine, or part of an engine; a process plant for the production of a chemical, or a large or small part of the plant; a human being, or some part of the body such as the muscle control mechanism for a limb; or it might be the economic system of a country, or any other from a wide range of fields.

The system would normally be considered conceptually as being that part of the universe in which interest lay. There would be interaction between the system and certain parts of the surroundings known as the environment. The two would be separated by an imaginary boundary. In defining the system and its environment it is necessary to decide where this boundary should be

2 Introduction

placed; this decision depends both on the physical entities involved and on the purpose of the investigation.

In studying a power station, interest might lie primarily in the relationship between the power station and the community, in which case the system and its environment might be envisaged as in Fig. 1.1. There might, however, be a more specific interest in the speed control system of the turbogenerator, in which case the system could be as in Fig. 1.2.



Fig. 1.1 Power station system and its relationship to the environment



Fig. 1.2 Turbogenerator speed control system

In abstracting from the whole the system of interest, it is necessary to consider carefully where the boundary shall be placed, and closely allied is the need to decide what relevant signals cross the system boundary. In addition, there will be signals of interest within the system boundary, variables which help to describe and define the detailed system behaviour. Some of these signals will be measurable, some not or only indirectly; some will be useful from the viewpoint of analysis, and some not.

The signals which pass to the system from the environment will be termed the system inputs, while those passing out across the boundary will be the system outputs. Often there will be only one system input that is varying and one system output which is affected. The systems to be considered in this book will be predominantly single-input-single-output systems, the type which occurs most frequently in practice.

1.2 System control

The aim of studying dynamic system behaviour is generally one of gaining an understanding of the system, with a view to controlling it to give specific values of certain important variables, to satisfy a required specification.

For the purpose of controlling the system it is necessary to adjust the values of one or more of the inputs to the system. Only certain of the inputs will be available for adjustment and these are referred to as the controlled inputs, whereas others will be disturbance inputs over which no control exists. In the heating of a room, for example, the heat input from the heating device can be altered as required, but the heat flow to or from the environment cannot be controlled in the same way. The variable chosen to be a measure of the desired system output may or may not give a true indication that the control is satisfactory. In the room heating control, the temperature of interest is probably the average temperature or the temperature in the part of the room where people sit, whereas the temperature measured is that at one specific point, the location of the thermostat. This may not even be positioned in the same room, so that appropriate allowances must be made in the design and utilization of the heating system.

Sometimes the incentive for studying a system will be purely one of seeking an understanding of the way in which it functions. In this category come some physiological systems, for which possibilities for designing control loops or improving system behaviour are rather limited.

Two broad classes of control system are available, open loop control and closed loop control and these are depicted schematically in Fig. 1.3.



Fig. 1.3 Open loop and closed loop control (a) open loop (b) closed loop

(a) Open loop or scheduling control. On the basis of knowledge about the system and of past experience, a prediction is made of what the input should be to give the desired output; the input is adjusted accordingly. Familiar examples are automatic toasters, programmable washing machines, and interest rate variations as they are used to affect economic systems. Such control is frequently unsatisfactory because any unexpected disturbances to the system can cause a deviation in the output from the desired value. The quality of the toast will vary with the type of bread and the initial temperature of the toaster, the cleanliness of the clothes will depend on correct assessment of amount of soap powder and length of washing cycle required, the effectiveness of an

4 Introduction

interest rate change will depend on a host of other factors affecting the economy.

(b) Closed loop or feedback control. The system output is measured and compared with the desired value; the system continually attempts to reduce the error between the two. Familiar examples are thermostatic controls on domestic and industrial ovens and other heating systems, level controls on water cisterns, and speed regulation by means of engine governors. Frequently the loop is closed through a human being; this is the case with road vehicles, as when a car is driven along an undulating road at a steady speed, or when the car is positioned in its garage at the end of the run.

1.3 The need for analysis

There are many examples of early control systems such as a device of Hero of Alexandria which opened a set of temple doors when a ceremonial fire was lit, and closed them again when the fire died down, or the much later centrifugal governor developed by James Watt for the speed control of steam engines. These systems were produced almost entirely by a trial and observation design process and without the assistance of any theoretical analysis. Simple control loops can still often be made to operate satisfactorily in this way because the specifications have a wide tolerance. As performance requirements become more demanding it becomes necessary to resort to a more analytical approach, since without this the cost in terms of time, manpower, and unnecessary complexity of equipment is not justifiable.

Closing the loop can make the system more accurate by giving a much smaller or a zero steady state error, but it can make the system very oscillatory or even unstable. Basically, problems arise when delays occur within the system; this causes corrective action to be applied too late, leading to alternating overcorrection and undercorrection. It is necessary to achieve a satisfactory compromise between the conflicting requirements of accuracy and stability.

1.4 Methods of system representation

It has been shown that the first step in the study of a system is the important one of defining clearly what constitutes the system of interest, and in what ways the system interacts with the surrounding environment. Having drawn a conceptual boundary round the system it is necessary to represent the system in a convenient pictorial and mathematical way.

A useful and very frequently used pictorial representation of a system is the block diagram where individual blocks are used to represent separate functional parts of the system. Fig. 1.3a is a simple block diagram representing a system with a single input and a single output, the lines indicating the signal flow paths, with direction of signal transmission given by the arrows. Where signals are added to or subtracted from one another, summing points are indicated, as shown in Fig. 1.3b. Although any single-input-single-output system could be represented by a single block as in Fig. 1.3a, if the system comprises a number of interacting components it is more useful if it is represented by several blocks interconnected by the appropriate signal flow paths.



Fig. 1.4 Schematic diagram of pump and flow control on a gas turbine speed control system (with symbols for pressures, flows, springrates, diaphragm areas, flow restrictions, etc. omitted)

In arriving at a block diagram representation, an intermediate schematic diagram in which the functional parts are clearly shown would often be utilized. Consider as an illustration that the system of interest is an aircraft gas turbine speed control system. The engineering drawings, although showing the physical arrangement, are too congested with detail and would not show the type of information required for a dynamic study. A schematic diagram of the form shown in Fig. 1.4 would however show how the system components function and form the basis for an analytical study, and enable the production of a block diagram of the form shown in Fig. 1.5.

Block diagrams show only the interrelationships between the different parts of the system, and for analysis must be supplemented by a quantitative





6 Introduction

description in the form of appropriate mathematical expressions for each of the blocks on the diagram. Such a mathematical description of the system is termed a *model*. Evaluation of an accurate model is often difficult. Since dynamic conditions are being considered the equations relating the outputs to the inputs of the blocks will in general be differential equations. It will be seen in the next chapter that for control engineering purposes these are often written as *transfer functions*, defined as the ratio of the Laplace transforms of the output and input when initial conditions are zero. When these equations are combined to give an overall output-input relationship a single differential equation of high order results.

An alternative method of mathematical representation which is particularly amenable to solution by means of the digital computer is the *state space* technique. Instead of a single *n*th order equation the problem is transformed to one of solving *n* first order equations. Unlike the classical approach, the complexity of the method of solution does not increase rapidly with the complexity of the problem, and hence this approach is particularly suitable for complex systems.

An alternative pictorial representation is the *signal flow graph*. This type of representation does not have to cater for the transfer function normally associated with the block diagram and so is capable of giving a more detailed schematic representation of a complex system if needed. It illustrates the passage of signals through a system, as does the block diagram, but also includes a more basic description of the feedback paths associated with a system, and enables the effects of variation in all system parameters to be seen directly.

The analytical equations would generally be obtained by a combination of theoretical analysis and experimental testing. If the component of the system is sufficiently simple, then it is possible to write down equations which govern the variables concerned, and hence obtain both the form of the equations and the values of the parameters of the equations. This is described for a number of physical components in Chapter 2. Subsequently the component would often be tested to verify the parameter values obtained. If the system is more complex, then simplifying assumptions must be made to arrive at the likely form of the equations; these assumptions must be confirmed and parameters obtained by testing. If the contents of the block are unknown (a so-called *black box* problem) then it will be necessary to arrive at the characteristics entirely experimentally.

The resulting equations will be a parametric model of the component, and ideally the parameters will be associated with specific physical characteristics of the system. In the black box type of situation the latter would not be the case, and sometimes no attempt is made to fit equations to the experimental response. In this situation the component is described by the actual response curve, and the model would be non-parametric.

1.5 Methods of analysis and design

When the input to a system changes as a function of time, the form of the resulting response is clearly dependent both on the nature of the input func-

tion and on the equations describing the dynamic characteristics of the system and, provided that these equations are available, the output response can be calculated for any input function of known mathematical form. In practice external disturbances, often of unpredictable form, may act as additional inputs and modify the response in a random manner. Such disturbance inputs, generally referred to as noise when they consist of random fluctuations about a mean value, have negligible effect in many situations and hence are usually ignored in the earlier stages of the analysis of a system. The disturbance can also be a change in mean value of a variable which causes an alteration in the system datum operating point, and hence in the parameters of the system equations. It is unrealistic for the purposes of analysis to consider attempting to investigate the response of the system for all conceivable types of input function; thus usually only certain specific types of function are studied. These are chosen primarily for reasons of analytical simplicity and because design criteria have been developed for them. They include an instantaneous change, which is the most severe input change that a system can undergo, and certain other typically encountered input functions such as an input changing in a sinusoidal manner, or one changing at a constant rate.

The input forcing functions most commonly considered are of three types:

- (i) transient disturbances such as step changes of magnitude, ramp changes, or impulsive changes.
- (ii) sinusoidal signals,
 - (iii) statistical signals, which have random characteristics.

The output response as a function of time can be obtained for any specific forcing function by analytical or computer solution of the differential equations. Study of the effects on the response which result from alterations in the mathematical model yields an understanding of the dynamic significance of the various terms in the governing equations, and hence the significance of the corresponding parameters in a physical system. For experimental verification of theoretically obtained dynamic equations, or for a black box approach to the identification of a mathematical model, a practical system component can be tested by recording the output for one or more of these input functions. Subsequently the measured response is compared with the response from a range of mathematical models; the model with the closest fit can then be chosen to represent the tested system component.

Solution of a system differential equation can, in principle, be carried out using either a digital or an analogue computer. Chapter 3 describes the basic elements of an electronic analogue computer, and the method for deriving a circuit diagram which defines the appropriate interconnection of these elements, to produce a circuit whose governing differential equation is the same as that to be solved. The circuit is thus an analogue or simulation of the system and, if forced with a voltage input of the desired waveform, then the variation of the output voltage represents the time response of the system. The variation of any system variables other than the output can be noted by recording the voltages at appropriate points in the circuit. This is followed by a description of how the mode control buttons on the computer switch the circuit to enable the computation to be started, stopped, and otherwise

8 Introduction

controlled in a useful manner, a list of some non-linear elements which are available for solving non-linear differential equations, and a discussion about time scaling and amplitude scaling. The chapter concludes with two sections on digital computer solution of differential equations. These explain first the discrete approach which is needed and the way in which integration routines are used to obtain an approximation to the solution, and then the use of a high level simulation language to simplify program writing and often to enable the digital computer to be used in an on-line interactive manner.

Solution of the system differential equation can alternatively be carried out by hand for simple transient changes of input and for sinusoidal input changes, the first two types of forcing function listed above. Chapters 4 and 6, which deal respectively with time domain analysis and frequency domain analysis, describe how the Laplace transform technique can be utilized to determine the response of a linear system of known transfer function or differential equation to such input functions. Systems of gradually increasing complexity are analysed: in this way a clear mental picture can be built up of the nature of dynamic response and of the way in which changes in the form of the governing equation influence the response. The latter part of each chapter describes the converse process-testing an actual system component with one of these forms of input function, noting the resulting response, and determining a transfer function which would give a very similar response and which could be used as a mathematical model of the system component. Statistical signals, which have random characteristics and must be described by appropriate statistical functions rather than by analytical functions of time, form the third class of important forcing functions. They are particularly useful for this process of system identification for experimental testing in situations where the level of inherent system noise is significant; by appropriate mathematical manipulation the effect of the noise can largely be eliminated and time or frequency response information obtained. Chapter 7 defines and illustrates the significance of autocorrelation functions and power spectral densities which are respectively time and frequency domain descriptions of such signals; it also describes this method of system identification with particular reference to the most frequently used signal, the pseudo random binary sequence.

The chapters whose contents are outlined above all utilize a dynamic system description which is in the form of a transfer function or the equivalent *n*th order differential equation relating output to input. The methods of analysis and design associated with this form of description are referred to as the *classical methods*. Chapter 5 describes the alternative state space approach to system representation and analysis, an approach which is fundamental to a range of techniques of analysis and design referred to as *modern control theory*. Methods are presented for deriving the state vector differential equation, the *n* first order differential equations which constitute the state space description of a system component, a form of description which can be used equally well where there is more than one input and output. A method of solving the state vector differential equation is presented, and for a simple input function it is demonstrated that the analytical solution is the same as that obtained by the Laplace transform approach of Chapter 4. Normally however solution must be carried out by digital computer methods, and it is here, in the unified

approach that is possible, that the power of the state space description lies. The chapter concludes by giving an insight into the way in which such solution is carried out. The methods of analysis and design described in the later chapters are the classical methods, chosen because of the unrivalled understanding of system behaviour which is offered by time and frequency response techniques. Once a fundamental understanding has been gained in this way a deeper study of the modern approach is recommended particularly where design work must be carried out for systems with more than one input or output.

The final four chapters are concerned broadly with designing or modifying a system to ensure that its dynamic behaviour is acceptable. Chapter 8 returns to the topic of system control already introduced in Section 1.2, and describes the main characteristics of feedback control systems. The steady state accuracy is evaluated in terms of the transfer function and of the input function; it is also shown that an increase of accuracy is accompanied by a tendency towards. more oscillatory behaviour and might give rise to instability. The two main methods of stability analysis are described and illustrated, and the chapter concludes by showing how the overall response of the closed loop system can be evaluated from a knowledge of the dynamic response of the system components within the loop. Chapter 9 describes the root locus method of analysis, a technique which assists the engineer to gain an understanding of system behaviour by showing what effect variation of system gain or some other variable has on the transient response. Chapter 10 explains the effect on system performance of introducing a sampler within the loop, thus converting a continuous signal to a discrete data form, and describes how the analytical techniques must be modified to extend to sampled-data systems. The final chapter explains the functioning of integral action and derivative action within a controller, and describes the general approach to system improvement by the use of additional compensation networks within the control loop. To conclude, the many facets of system modelling and control described throughout the book are brought together by presenting in outline form a case study for a practical system, an electrohydraulic position control for the slideway of a milling machine, designed to be numerically controlled.

2 Mathematical Description of System Components

It has been indicated in Chapter 1 that for analysis and design of a system to give satisfactory dynamic behaviour it is necessary to obtain a suitable mathematical model to represent the system. The combination of a block diagram and the mathematical expressions relating the input and output of each block provides a pictorial and a quantitative representation of the cause-and-effect relationship between the variables of the system.

Although all systems with a single input and a single output may be denoted by a single block connecting the input and the output, the advantage of the block diagram concept lies in the fact that many systems are composed of several non-interacting elements whose output-input relationships can be determined independently. In this chapter it is shown how these relationships can be obtained for relatively simple components, and the form which the equations take. It is also shown how they can be combined to yield the overall output-input relationship for the system.

Section 2.1 considers the concept of linearity, and shows the way in which many non-linear systems can be linearized provided any perturbations from a datum are small. The Laplace transform technique is used widely in system analysis and synthesis, and for those unfamiliar with the method an introduction and an outline of the most important features are given in Section 2.2. This is followed by the definition of a transfer function, and the derivation of transfer functions for some physical components. To complete the picture, representation of a system by means of state equations is introduced in Section 2.6.

2.1 Linearity of systems

The expression relating input and output of a component of a system will in general be a differential equation. For ideal systems this is frequently linear and if it is possible to represent practical systems sufficiently closely by linear differential equations considerable analytical advantage is gained.

A system is said to be *linear* if it obeys the *principle of superposition*. This requires that if the separate application of time dependent inputs $u_1(t)$ and $u_2(t)$ produces outputs $c_1(t)$ and $c_2(t)$ respectively, then the simultaneous application of $u_1(t) + u_2(t)$ will produce the output $c_1(t) + c_2(t)$. The input $u_1(t)$ might be a step change, and $u_2(t)$ a ramp change where the input increases at a constant rate. Alternatively if $u_2(t) = u_1(t)$ the result would be that a doubling of the step size would double the output response curve, or increasing the step size

by a factor of 10 would merely scale up the response by 10. A differential equation which is linear does not contain any terms which are products of or powers of the variable and its derivatives.

No real system component is completely linear, but often the range of operation is such that linearity can be assumed. This is the case, say, for a helical spring where the input is the force acting and the output is the spring length, provided the load does not exceed that which causes the spring to compress till the coils touch, and provided any tensile force does not cause material yield to occur. Similarly an electronic amplifier will saturate for very large inputs, but for inputs within the design range the gain should be substantially constant.

Many system elements are inherently governed by a non-linear relationship. The rate of flow of fluid through an orifice is proportional to the square root of the pressure difference across it; doubling the pressure difference does not double the flow rate, and the principle of superposition does not apply. With such elements the function relating the input and output variables does not have any linear region. If the system is operating about a nominal datum condition (say point A on Fig. 2.1), and if the range of operation is such that



Fig. 2.1 Non-linear relationship

movement from this datum is small, then the departure from linearity is small. Replacing the curve by the tangent at the datum point would allow linear techniques of analysis to be used. If the range of operation is large, linear techniques intelligently used could still give a useful indication of the nature of the expected performance.

In analysing a system with such continuous types of non-linearity it is often useful to adopt a linear approach and carry out a *small perturbation analysis*. Linearization essentially consists of replacing the actual operating characteristic functions by tangents at the operating point.

Consider a non-linear relationship

$$Y = f(X_1, X_2, X_3 ...)$$
 2.1

Differentiation gives

$$dY = \frac{\partial Y}{\partial X_1} dX_1 + \frac{\partial Y}{\partial X_2} dX_2 + \frac{\partial Y}{\partial X_3} dX_3 + \dots \qquad 2.2$$

Let a subscript o be used to indicate conditions at a datum point. Considering

12 Mathematical Description of System Components

dY, dX_1 , dX_2 ... to be incremental changes from the datum $[Y]_0$, $[X_1]_0$, $[X_2]_0$... the partial derivatives will be evaluated at that datum and will be constants. Let the changes from the datum be y, x_1 , x_2 , x_3 ... The linearized equation then becomes

$$y = C_1 x_1 + C_2 x_2 + C_3 x_3 + \dots$$
where $C_1 = \left[\frac{\partial Y}{\partial X_1}\right]_0$, $C_2 = \left[\frac{\partial Y}{\partial X_2}\right]_0$, etc.

Note that the variables are now not absolute variables but variables relative to a datum point. For this one section of the book absolute variables are represented by upper case letters and variables relative to the datum by lower case letters. The original non-linear relationship between the absolute variables has been replaced by a linear relationship involving the new variables.

Example 2.1. Consider a mechanical flyweight governor as shown schematically in Fig. 2.2. A reasonable assumption would be that the force F exerted by



Fig. 2.2 Flyweight governor with force-speed relationship

the flyweights, arising from centrifugal action, will be proportional to the square of the rotational velocity N

i.e. $F = KN^2$ where K is a constant. This is a non-linear relationship.

If
$$K = \frac{1}{10\,000}$$

the force-speed curve is as shown in Fig. 2.2.

If the datum point is 200 rev/minute

$$\left[\frac{\mathrm{d}F}{\mathrm{d}N}\right]_{0} = 2K\left[N\right]_{0} = \frac{2 \times 200}{10\,000} = \frac{1}{25}$$

Hence the linearized equation is

$$f = \frac{1}{25} n$$

where force f and speed n are measured relative to the datum F = 4 and N = 200.

To assess the error arising from linearization choose N = 250 rev/minute

i.e. n = 50.

This gives f = 2 i.e. F = 6.

The true value of
$$F = \frac{(250)^2}{10\,000} = 6.25$$

The smaller the change from the datum the smaller will be the percentage error. If the datum point is 100 rev/minute

$$\left[\frac{\mathrm{d}F}{\mathrm{d}N}\right]_{0} = 2K\left[N\right]_{0} = \frac{2 \times 100}{10\,000} = \frac{1}{50}$$

and the linearized equation is

$$f = \frac{1}{50} n$$

Example 2.2. Consider the equation for the flow of fluid through an orifice:

$$Q = C_{\rm d} A \sqrt{\left(\frac{2\Delta P}{\rho}\right)}$$

where Q is the fluid flow rate, C_d the discharge coefficient, A the orifice area, ΔP the pressure difference across the orifice and p the density of the fluid.

If C_d , A and ρ are constant then this becomes

$$Q = C \sqrt{(\Delta P)}$$
 where $C = \text{constant}$

In linearized form this is

$$q = C'\Delta p$$

where q and Δp are changes relative to a datum $[Q]_0, [\Delta P]_0$

and
$$C' = \left[\frac{\mathrm{d}Q}{\mathrm{d}\Delta P}\right]_0 = \left[\frac{C}{2\sqrt{\Delta P}}\right]_0$$

If the area A is not constant, such as would be the case in a flow control valve, then the linearized equation would be

$$q = C_1 a + C_2 \Delta p$$

$$C_1 = \left[\frac{\partial Q}{\partial A}\right]_0 = \left[C_d \sqrt{\left(\frac{2\Delta P}{\rho}\right)}\right]_0$$

$$C_2 = \left[\frac{\partial Q}{\partial \Delta p}\right]_0 = \left[\frac{C_d A}{\sqrt{(2\rho\Delta P)}}\right]_0$$

where

14 Mathematical Description of System Components

If the density is also variable then the equation would be

$$q = C_1 a + C_2 \Delta p + C_3 \rho$$

Linearized equations can also be obtained directly from experimental curves. The constant coefficients are then obtained by measuring the slopes at the datum points directly from the plotted graphs.

It should be noted that certain types of non-linearity cannot be dealt with by linearization. Notable amongst these are effects such as hysteresis or backlash where decreasing the perturbation size causes the effect to be more prominent. Sometimes it is convenient to deal with such systems in a quasi-linear manner by writing down a set of linear equations for each distinct operating region.

Throughout the remainder of this book all variables will normally be considered to be relative to a datum; hence the need for maintaining a distinction between absolute and relative variables is not great. This allows use of the more usual convention that lower case letters are employed to denote functions of time and upper case letters the equivalent functions of the Laplace operator, as will be shown in the next section.

2.2 Laplace transforms and their significance

The Laplace transform technique is a very convenient method for assisting in the solution of differential equations. It is helpful at first to consider this technique as being somewhat analogous to the use of logarithms to simplify such mathematical operations as multiplication, division and raising numbers to powers. When carrying out such operations, the original numbers of the problems are transformed into the logarithmic domain by the use of log tables, the solution in the logarithmic domain is obtained by a simpler process (addition instead of multiplication, multiplication instead of raising to a power, etc.), and finally the result is transformed back into the normal number domain by use of the anti-log tables.

Similarly, when solving a differential equation using Laplace transforms, the equation is first transformed into the Laplace domain by changing the variable from time t to a new complex variable $s = \sigma + j\omega$, known as the Laplace operator. The solution of the differential equation is then effected by simple algebraic manipulations in the s domain yielding a solution which is a function of s. Finally, to obtain the desired time solution, it is necessary to invert the transform of the solution from the s domain back to the time domain. As with logarithms, tables exist and can be used to transform from one domain to another.

The Laplace transform of a function of time f(t) is written as F(s) and is defined as

$$F(s) = \mathscr{L}[f(t)] = \int_{0}^{\infty} f(t) e^{-st} dt$$

where $s = \sigma + j\omega$ is an arbitrary complex variable.

2.4

2.2 Laplace transforms and their significance 15

The transformation is thus an integration process applied to f(t). F(s) is finite even when f(t) does not tend to zero, the only requirements being that the variable f(t) must be defined for all values of time t > 0 and be zero for t < 0, and that s is sufficiently large to ensure that the integral converges. There is a unique value of F(s) in spite of there being a large range of values of s which are suitable. By applying the above integration process the Laplace transform for any function f(t) can be obtained, and once obtained it need not be derived again. Tables of Laplace transforms have thus been compiled, and appear in many control engineering and mathematics textbooks. Table 2.1 lists the Laplace transform pairs of most importance in control work, and which will be used in this book; the rather complicated f(t) describes a damped sine wave whose significance will be made clear later.

f(t)	F(s)	f(t)	F(s)
unit step	$\frac{1}{s}$	$\frac{\mathrm{d}t(t)}{\mathrm{d}t}$	sF(s) = T(0)
unit tamp I	$\frac{1}{s^2}$	$\frac{\mathrm{d}t^n(t)}{\mathrm{d}t^n}$	$s^{n}F(s) = s^{n-1}F(0) = -f^{n-1}(0)$
unit impulse $\delta(t)$	ŧ		where $t^{\alpha}(0) = \left[\frac{\mathrm{d}^{n}f(t)}{\mathrm{d}t^{\alpha}}\right]_{t=0}$
R-41	$\frac{1}{s+a}$	$\frac{\omega_n}{\sqrt{(1-\zeta^2)}} \in \frac{-2\pi i n}{2} \sin \omega_n t \sqrt{(1-\zeta^2)}$	$\frac{\omega_n^2}{s^{2^2}+2\zeta\omega_ns+\omega_n^2}$
sin wt	$\frac{\omega}{s^2+\omega^2}$	e-*t sin Wt	$\frac{\omega}{(s+a)^2+\omega^2}$
005 W t	$\frac{s}{s^2+\omega^2}$	e-** cos w r	$\frac{s+a}{(s+a)^2+\omega^2}$

Table 2.1 Common Laplace transform pairs

The following theorems are those of most frequent use:

(a) Addition and subtraction:

$$\mathscr{L}[f_1(t) \pm f_2(t)] = F_1(s) \pm F_2(s)$$
where $\mathscr{L}[f_1(t)] = F_1(s), \mathscr{L}[f_2(t)] = F_2(s)$

$$2.5$$

(b) Multiplication by a constant:

$$\mathscr{L}[Kf(t)] = KF(s) \tag{2.6}$$

(c) Final value theorem:

$$\lim_{t \to \infty} f(t) = \lim_{s \to 0} sF(s)$$
 2.7

This is useful since it gives the final value of a time function (i.e. the steady state value) by determining the value of its Laplace transform as $s \rightarrow 0$. The theorem is not valid if the denominator of sF(s) contains any root whose real part is zero or positive (which as will be seen later implies that the function tends to infinity as $t \rightarrow \infty$).

16 Mathematical Description of System Components

(d) Shifting theorem:

If
$$\mathscr{L}[f(t)] = F(s)$$
 then $\mathscr{L}[f(t-T)] = e^{-sT}F(s)$ 2.8

It can be seen from the table that the Laplace transform of the *n*th derivative of a function is given by

$$\mathscr{L}\left[\frac{\mathrm{d}^{n}f(t)}{\mathrm{d}t^{n}}\right] = s^{n}F(s) - s^{n-1}f(0) - s^{n-2}f'(0) - \dots - sf^{n-2}(0) - f^{n-1}(0) \quad 2.9$$

The second and subsequent terms are dependent on the initial conditions, the values of the function and its derivatives at t = 0. Hence, if all initial conditions are zero, transformation of a differential equation into the s-domain can be achieved by replacing $\frac{d}{dt}$ by $s, \frac{d^2}{dt^2}$ by $s^2, \frac{d^3}{dt^3}$ by s^3 etc. The linear differential equation thus becomes an algebraic equation in s

equation thus becomes an algebraic equation in s. Inverse Laplace transformation, $\mathscr{L}^{-1}[F(s)] = f(t)$, is required to obtain the time response. A transformed solution will in general be a ratio of polynomials:

$$F(s) = \frac{a_m s^m + a_{m-1} s^{m-1} + \dots + a_0}{s^n + b_{n-1} s^{n-1} + \dots + b_0}$$

When this is not in a form which can be found in the available tables it must be split by partial fraction expansion into a number of functions which are listed in the tables:

$$F(s) = F_1(s) + F_2(s) + \dots$$

therefore $f(t) = \mathcal{L}^{-1}[F_1(s)] + \mathcal{L}^{-1}[F_2(s)] + \dots$

The process of using the Laplace transform technique in this way to solve the differential equation for a given system input function is illustrated in Chapter 4. Often however in control system analysis, information in the s-plane suffices, and it is not necessary to carry out the final step of transforming the solution back into the time domain.

2.3 Transfer functions and the characteristic equation

The relationship between input and output of a dynamic system component will normally be described by a differential equation. For the purposes of system analysis the control engineer uses Laplace transform notation, and if the component is linear writes the equation in the form of a transfer function.

The *transfer function* of a linear system is defined as the ratio of the Laplace transform of the output to the Laplace transform of the input when all initial conditions are zero. Conventionally the symbol G(s) is used, but if the element appears in a feedback loop the symbol would be H(s) (see Fig. 2.3).

It is thus seen that the transfer function is a property of a system and describes the dynamic characteristics of the system but is not influenced by the state of the system. Since initial conditions are assumed to be zero for evaluating the transfer function it follows that the differential equation can be



Fig. 2.3 Examples illustrating block diagram reduction

transformed into the Laplace domain by replacing $\frac{d}{dt}$ by $s_t \frac{d^2}{dt^2}$ by s^2 etc. The resulting transfer function in the general case then takes the form

$$G(s) = \frac{P(s)}{Q(s)}$$

where P(s) and Q(s) are polynomials in s. For the system to be physically realizable the order of the numerator cannot exceed that of the denominator.

It will be seen in Chapter 4 that the form of the output response of a system when subjected to a changing input variable is determined by the values of the roots of the equation Q(s) = 0. This equation, obtained by equating the denominator of the overall transfer function to zero, is termed the *characteristic equation*. The roots of the characteristic equation are referred to as the *poles* of the overall system transfer function, since they are values of s which cause the transfer function magnitude to become infinite. The number of roots, and thus the order of the characteristic equation is termed the order of the system.

Where a complex system comprises many interconnected simple blocks the overall transfer function can be obtained by reduction of the block diagram to a simple one with a single block, which has however a complex transfer function. The process is one of simple algebraic manipulation, and is illustrated by Fig. 2.3. Where loops interlink as in Fig. 2.3d some additional manipulation is needed.

2.4 Transfer functions for some simple elements

Transfer functions will be obtained for six simple components to illustrate the form which they take and also the way in which they are derived analytically.

Example 2.3. Ideal spring. Consider a simple coil spring of negligible mass, to one end of which a force $f_s(t)$ is applied, the other end being fixed (Fig. 2.4).



Fig. 2.4 Simple spring

There will be a deflection x(t) from the unloaded position of the end of the spring. This component could be represented by a block with input $f_n(t)$ and output x(t).

The equation relating input and output is

 $f_{s}(t) = K x(t)$, where K = spring stiffness

$$\frac{x(t)}{f_s(t)} = \frac{1}{K}$$

Taking Laplace transforms yields the transfer function

$$G(s) = \frac{X(s)}{F_s(s)} = \frac{1}{K}$$
 2.10

In this case there is no time dependence, and the transfer function is a pure gain term. This relationship assumes that the spring stiffness is constant, and is obviously valid only provided the force is not so large that the spring coils come together.

Example 2.4. Ideal hydraulic damper. Consider now a piston of negligible mass sliding with some clearance in an oil-filled cylinder under the action of a force $f_d(t)$ (Fig. 2.5). Again there will be a relationship between the position of the



Fig. 2.5 Hydraulic damper

piston x(t) and the applied force. This is obtained from the basic physical considerations which are that the viscous drag on the piston is proportional to the velocity of the piston in the cylinder.

$$f_{\rm d}(t) = C \, \frac{{\rm d}x(t)}{{\rm d}t} \tag{2.11}$$

where C = viscous damping coefficient = force per unit velocity

Taking Laplace transforms gives, for zero x(0)

$$F_{\rm d}(s) = CsX(s) \tag{2.12}$$

and hence the transfer function is

$$G(s) = \frac{X(s)}{F_d(s)} = \frac{1}{Cs}$$
 2.13

Again no inertia has been included and it has been assumed that there is a free flow through the piston so that no pressure difference can arise across the piston. If either were not valid then the force balance equation (Eq. 2.11) would not be correct. Clearly these equations are only valid so long as the piston does not reach either limit of its stroke.

Example 2.5. Mass-spring-damper. A frequently occurring physical arrangement that can be represented by a force acting on a mass which is restrained by a spring and viscous damper is shown diagrammatically in Fig. 2.6. It is assumed that the mass is constrained to move in the direction of the applied force by friction-free guiding surfaces.



Fig. 2.6 Mass-spring-damper

The basic physical law relating the position of the mass to the force acting upon it is Newton's second law of motion i.e. the sum of the applied forces is equal to the rate of change of momentum.

$$f(t) - f_{s}(t) - f_{d}(t) = M \frac{d^{2}x(t)}{dt^{2}}$$

but $f_{s}(t) = Kx(t)$ and $f_{d}(t) = C \frac{dx(t)}{dt}$
$$f(t) = M \frac{d^{2}x(t)}{dt^{2}} + C \frac{dx(t)}{dt} + Kx(t) \qquad 2.14.$$

For zero initial conditions the Laplace transform is

$$F(s) = Ms^2 X(s) + Cs X(s) + KX(s)$$

Hence the transfer function is

$$\frac{X(s)}{F(s)} = \frac{1}{Ms^2 + Cs + K}$$
2.15

The most general form of equation 2.14, using the dot notation for differentiation, is

$$\ddot{x}(t) + 2\zeta \omega_n \dot{x}(t) + \omega_n^2 x(t) = \omega_n^2 u(t)$$
 2.16

where u(t) is the forcing term, $\omega_n = \sqrt{(K/M)}$ is the system natural frequency, and the symbol ζ is known as the damping factor. The full significance of these terms will be considered in Section 4.2.

For zero initial conditions, the Laplace transform of Eq. 2.16 is

 $(s^2 + 2\zeta\omega_n s + \omega_n^2)X(s) = \omega_n^2 U(s)$

and thus the transfer function is

$$\frac{X(s)}{U(s)} = \frac{\omega_n^2}{s^2 + 2\zeta\omega_n s + \omega_n^2} = G(s)$$
 2.17

Example 2.6. Liquid in glass thermometer. For a simple thermometer (or equivalent temperature measurement device) there is a relationship between the

2.4 Transfer functions for some simple elements 21

indicated temperature and the temperature being measured. Let $\theta_i(t)$, the temperature of the fluid around the bulb, be the input and $\theta_0(t)$, the temperature of the fluid in the thermometer, be the system output. The thermometer fluid volume will vary in proportion to its temperature, and the stem is graduated accordingly. The temperatures are both time varying and hence functions of time t.

The rate of heat flow q(t) into the thermometer fluid is proportional to the temperature difference across the walls

$$q(t) = \frac{\theta_{i}(t) - \theta_{0}(t)}{k_{1}}$$
 2.18

where k_1 is the thermal resistance and is determined by the coefficients of heat transfer from fluid to glass, through the glass, and from glass to inner fluid.

Also, the rate of heat flow q(t) is proportional to the rate of temperature rise of the thermometer fluid

$$q(t) = cm \, \frac{\mathrm{d}\theta_0(t)}{\mathrm{d}t} \tag{2.19}$$

where c is the specific heat and m is the mass of the thermometer fluid.

$$\theta_i(t) - \theta_0(t) = k_2 \frac{d\theta_0(t)}{dt}$$
 where $k_2 = k_1 cm$

Taking Laplace transforms yields the equation

$$\theta_{i}(s) - \theta_{0}(s) = k_{2}s\theta_{0}(s)$$

$$\therefore \quad \frac{\theta_{0}(s)}{\theta_{i}(s)} = \frac{1}{1 + k_{2}s}$$

2.20

Note that the thermal capacity of the glass has been assumed to be negligible, and the overall coefficient of heat transfer assumed to be constant.

Note also that the parameters have been considered to be *lumped*, which means that the thermometer fluid temperature has been assumed to be uniform in a spatial sense, as has the temperature of the fluid being measured. If the temperatures were to be considered as functions of both time and position, it would be necessary to describe the system by partial differential equations, and it would be termed a *distributed parameter system*.

Example 2.7. Resistance-capacitance network. Consider the circuit shown in Fig. 2.7 with input voltage $V_i(t)$ and output voltage $V_0(t)$. Assume that the output impedance is infinite.



Fig. 2.7 R C network

22 Mathematical Description of System Components

$$i(t) = \frac{V_{i}(t) - V_{0}(t)}{R} = C \frac{dV_{0}(t)}{dt}$$

..., $V_{i}(s) - V_{0}(s) = RCsV_{0}(s)$
..., $\frac{V_{0}(s)}{V_{i}(s)} = \frac{1}{1 + RCs}$
2.21

Example 2.8. Hydraulic servomechanism. Fig. 2.8 shows schematically a hydraulic servomechanism, a feedback device commonly found in practice,





whose function is to move a load of mass M to a position y(t) in response to a command signal x(t) using a hydraulic or pneumatic supply to provide the power. Let the input and output be positive in the direction of the arrows and let the position of the spool valve be given by e(t). Note that x, y, and e are functions of time t. Increase of x, say, would cause the link AC (called a *walking beam*) to pivot about C, cause e to increase and the spool valve to move upwards thus allowing fluid to flow to the space above the piston and from the space below. The piston thus moves downward, so causing AC to pivot about A and thus e to decrease again. Ideally the whole system will come to rest when the valve is again in the central position.

What is the relationship between output y(t) and input x(t)?

 $e(t) = \frac{x(t) - y(t)}{2}$

Ideally in the steady state, i.e. as $t \to \infty$, e = 0 and y = (b/a)x. Transiently however this is not true.

By geometry $e(t) = \frac{b}{a+b} x(t) - \frac{a}{a+b} y(t)$

and if a = b, then
Assuming that the pressure drop across the valve is constant, then the rate of flow through the valve q(t) is proportional to the area of opening (see Example 2.2),

say

$$q(t) = Ce(t) \tag{2.23}$$

But also this flow rate must equal the rate of change of volume of the chamber into which it is flowing

 $\therefore \quad q(t) = A \frac{dy(t)}{dt}$ $\frac{C[x(t) - y(t)]}{2} = A \frac{dy(t)}{dt}$ 2.24

Taking Laplace transforms

$$X(s) - Y(s) = \frac{2A}{C} sY(s)$$

$$\therefore \quad \frac{Y(s)}{X(s)} = \frac{1}{1 + \frac{2A}{C} s}$$

2.25

In addition to the assumption about constant pressure drop across the valve, many other gross assumptions have been made: no leakage, fluid is incompressible, no inertia, constant temperature, no clearance at pin joints etc. In the next section some of these other effects will be considered.

For each of the last three systems the transfer function has the same form, namely $\frac{1}{1+\tau s}$. This form of relationship is called a *simple time lag*; the

constant τ which is dependent on the system parameters is called the *time* constant. Although physically the systems are completely different, mathematically they are represented by the same form of model. They will thus all behave in a similar way and for analytical purposes can be considered to be identical. If the time constants are equal, and the excitation functions are of the same form, then the response functions will also be the same.

2.5 Effect of secondary factors on transfer functions

As has been shown above, in order to obtain the transfer function for a system element, it is necessary first to define clearly the boundaries of the system and the relevant signal variables. The relationships between these variables must then be determined by writing down the physical equations governing the system behaviour. Those variables which are not of direct interest must be eliminated leaving the relationship between input and output. The form of the resulting transfer function will be governed by what effects have been included. In a mechanical system, is the inertia of various parts significant? In a hydraulic system, are fluid leakage or compressibility effects of importance?

24 Mathematical Description of System Components

To illustrate the significance of secondary factors consider the effect of various factors on the transfer function of a simple hydraulic ram. Let the piston area be A, the system input be the input flow rate q(t), the system output be the piston position x(t), and the fluid pressure be p(t) (Fig. 2.9).



Fig. 2.9 Simple ram

(a) Simplest representation of ram. Neglect inertia and leakage; assume that the fluid is incompressible and the piping and cylinder are rigid. The governing equation is the flow continuity equation

$$q(t) = A \frac{\mathrm{d}x(t)}{\mathrm{d}t}$$
 2.26

Laplace transformation and rearrangement gives the transfer function

$$\frac{X(s)}{Q(s)} = \frac{1}{As}$$
 2.27

This is the relationship for an integrator, the integration occurring at rate $\frac{1}{4}$.

(b) Ram with inertia. Let the piston and whatever is connected to it have mass M.

If there is no leakage past the piston then the flow continuity equation, Eq. 2.26, is still valid and inertia has no effect.

If leakage does exist this is likely to be viscous in nature with a flow rate proportional to pressure difference. Let the leakage flow be $K_L p(t)$ where K_L is a leakage coefficient, p(t) is the fluid gauge pressure, and the pressure at the other side of the piston is atmospheric. The flow continuity equation is now

$$q(t) = A \frac{\mathrm{d}x(t)}{\mathrm{d}t} + K_L p(t)$$
 2.28

It is necessary to obtain another equation to allow p(t) to be eliminated. In this case it is obtained by applying Newton's second law.

$$p(t)A = M \frac{\mathrm{d}^2 x(t)}{\mathrm{d}t^2}$$
 2.29

$$\therefore \quad q(t) = A \, \frac{\mathrm{d}x(t)}{\mathrm{d}t} + \frac{K_{\mathrm{L}}M}{A} \frac{\mathrm{d}^2 x(t)}{\mathrm{d}t^2}$$
 2.30

Taking Laplace transforms for zero initial conditions gives

$$Q(s) = AsX(s) + \frac{K_{\rm L}M}{A} s^2 X(s)$$
$$\frac{X(s)}{Q(s)} = \frac{1}{\left(A + \frac{K_{\rm L}M}{A}s\right)s} = \frac{1}{As\left(1 + \frac{K_{\rm L}M}{A^2}s\right)}$$
2.31

It can be seen that the effect of including inertia and leakage is to introduce a simple lag of time constant $K_L M/A^2$ proportional both to the mass and the leakage coefficient.

(c) Ram with viscous load. If the resistance to motion arises not from inertia of the moving parts but from viscous drag forces then the following equations are relevant:

$$q(t) = A \frac{\mathrm{d}x(t)}{\mathrm{d}t} + K_{\mathrm{L}}p(t)$$

and

$$p(t)A = \mu \frac{\mathrm{d}x(t)}{\mathrm{d}t}$$
 where $\mu = \text{constant}$

$$q(t) = A \frac{\mathrm{d}x(t)}{\mathrm{d}t} + \frac{K_{\mathrm{L}}\mu}{A} \frac{\mathrm{d}x(t)}{\mathrm{d}t}$$

and hence

In this case the effect is still one of integration, but at a slower rate.

(d) Ram with inertia and viscous load. The equations are

$$q(t) = A \frac{\mathrm{d}x(t)}{\mathrm{d}t} + K_{\mathrm{L}} p(t)$$

and

2

$$p(t)A - \mu \frac{\mathrm{d}x(t)}{\mathrm{d}t} = M \frac{\mathrm{d}^2 x(t)}{\mathrm{d}t^2}$$

$$\therefore \quad q(t) = A \frac{\mathrm{d}x(t)}{\mathrm{d}t} + K_{\mathrm{L}} \left\{ \frac{M}{A} \frac{\mathrm{d}^2 x(t)}{\mathrm{d}t^2} + \frac{\mu}{A} \frac{\mathrm{d}x(t)}{\mathrm{d}t} \right\}$$

and
$$\frac{X(s)}{Q(s)} = \frac{1}{\left(A + \frac{K_{\rm L}\mu}{A}\right)s + \frac{K_{\rm L}M}{A}s^2} = \frac{\overline{A + \frac{K_{\rm L}\mu}{A}}}{s\left(1 + \frac{K_{\rm L}M}{A^2 + K_{\rm L}\mu}s\right)}$$
 2.33

$$\frac{X(s)}{Q(s)} = \frac{1}{\left(A + \frac{K_{\rm L}\mu}{A}\right)s}$$
2.32

26 Mathematical Description of System Components

(e) *Effect of compressibility*. In writing down the flow continuity equation above it has been assumed that the fluid is incompressible. In practice, even liquids are not completely incompressible, and so if the pressure rises the fluid volume decreases slightly. Hence some of the fluid entering goes to make up the decrease in volume due to compressibility of the fluid (and expansion of pipes, if the containing system is not completely rigid).

Compressibility is defined by the

Bulk modulus
$$K_{\rm B} = \frac{change in pressure}{change in volume per unit volume}$$

 $= \frac{\Delta p}{\Delta v}$ where $v =$ volume of fluid under pressure
 $\therefore \Delta v = \frac{v}{K_{\rm B}} \Delta p$
 $\frac{dv}{dt} = \frac{v}{K_{\rm B}} \frac{dp}{dt} = q_{\rm compressibility}$ 2.34

and

The flow continuity equation is

$$q_{\rm in} = q_{\rm velocity} + q_{\rm leakage} + q_{\rm compressibility}$$

$$q(t) = A \frac{\mathrm{d}x(t)}{\mathrm{d}t} + K_{\rm L}p(t) + \frac{v}{K_{\rm B}}\frac{\mathrm{d}p(t)}{\mathrm{d}t}$$

$$2.35$$

$$\therefore \quad q(t) = A \, \frac{dx(t)}{dt} + K_{\rm L} \left(\frac{M}{A} \, \frac{d^2 x(t)}{dt^2} + \frac{\mu}{A} \, \frac{dx(t)}{dt} \right) \\ + \frac{v}{K_{\rm B}} \left(\frac{M}{A} \, \frac{d^3 x(t)}{dt^3} + \frac{\mu}{A} \, \frac{d^2 x(t)}{dt^2} \right) \\ q(t) = \frac{vM}{K_{\rm B}A} \, \frac{d^3 x(t)}{dt^3} + \left(\frac{K_{\rm L}M}{A} + \frac{\mu v}{K_{\rm B}A} \right) \frac{d^2 x(t)}{dt^2} + \left(\frac{K_{\rm L}\mu}{A} + A \right) \frac{dx(t)}{dt} \\ \frac{X(s)}{Q(s)} = \frac{1}{s \left\{ \frac{Mv}{K_{\rm B}A} \, s^2 + \left(\frac{K_{\rm L}M}{A} + \frac{\mu v}{K_{\rm B}A} \right) s + \left(\frac{K_{\rm L}\mu}{A} + A \right) \right\}}$$
 2.36

and

It can be seen that the effect of inclusion of compressibility has been to raise by one the order of the polynomial in s in the denominator, and that the transfer function G(s) is a property of the system elements only and is independent of excitation and initial conditions. Similarly, including compressibility, inertia and leakage for the hydraulic servo would give a third order equation.

 $p(t)A - \mu \frac{\mathrm{d}x(t)}{\mathrm{d}t} = M \frac{\mathrm{d}^2 x(t)}{\mathrm{d}t^2}$

2.6 State equations

An alternative method of system representation developed since about 1960 has been the characterization of dynamic systems by means of state equations instead of transfer functions. The state equation will be recognizable as no more than a set of differential equations which define the behaviour of a dynamic system in terms of the dependent variables, at least if these equations are written in a certain form.

As a simple example to show the form taken by state equations, consider again the mechanical system shown in Fig. 2.6.

The equation of motion, written in the general form of Eq. 2.16, is

$$\ddot{x}(t) + 2\zeta \omega_n \dot{x}(t) + \omega_n^2 x(t) = \omega_n^2 u(t)$$
 2.37

where x(t) is the horizontal motion (output) and u(t) is the forcing (input) function. The equation may be solved by the application of the Laplace transform technique; this will indicate the variation of the dependent variable x(t) with respect to the independent variable time.

Eq. 2.37 can be rewritten as a pair of first order differential equations:

$$\dot{x}_1(t) = x_2(t)$$
 2.38

$$\dot{x}_{2}(t) = -2\zeta \omega_{n} x_{2}(t) - \omega_{n}^{2} x_{1}(t) + \omega_{n}^{2} u(t)$$
2.39

where $x_1(t) = x(t)$ (position) and $x_2(t) = \dot{x}(t)$ (velocity). The terms $x_1(t)$ and $x_2(t)$ are the dependent variables of a pair of differential equations, written in a certain form, which define the behaviour of a second order system and are known as *state variables*.

A two dimensional vector, having components x(t) and $\dot{x}(t)$ also defines the state of the system and is known as the state vector. If the mass M is displaced a distance x_0 from its static equilibrium position and released, the solution trajectory could be as shown in Fig. 2.10. This form of trajectory represents an oscillation which is decreasing until the system comes to rest.



Fig. 2.10 Typical solution trajectory

28 Mathematical Description of System Components

The importance of the state vector is that all future system states are completely defined if the initial state and system inputs are known. To illustrate this, Eq. 2.38 and Eq. 2.39 are rewritten as a single vector matrix equation:

$$\begin{cases} \dot{x}_1(t) \\ \dot{x}_2(t) \end{cases} = \begin{bmatrix} 0 & 1 \\ -\omega_n^2 & -2\zeta\omega_n \end{bmatrix} \begin{cases} x_1(t) \\ x_2(t) \end{cases} + \begin{bmatrix} 0 \\ \omega_n^2 \end{bmatrix} u(t)$$

or more briefly as

$${\dot{x}(t)} = A{x(t)} + Bu(t)$$
 2.40

where $\{x(t)\}\$ is the state vector, A is a 2×2 square matrix and B is a two element column vector. Equation 2.40 relates the rate of change of the state of the system to the present state of the system and the input signals. Some examples of the derivation of state equations will be given in Chapter 5.

The total space occupied by all possible values of the state vector is known as the *state space*. An *n*th order system will require a state vector having n components, and the equivalent state space will be *n*-dimensional.

It has been shown how a system can be represented schematically by a block diagram, each block having associated with it a transfer function describing the relevant output-input relationship. Also, a complex block diagram incorporating many elements with relatively simple transfer functions can be reduced to a single block with a high order transfer function relating the system output to the system input. In studying system behaviour it is desirable to be able to determine how the system would react to various input or disturbance functions.

If the form of the input function and of the transfer function are simple, then the dynamic response can be determined analytically by solving the governing differential equation, as will be shown in Chapter 4. If the response is required for a number of different input functions, or for a number of different system parameters, then the equation solution time can be excessive. It is useful to be able to obtain time response traces quickly, particularly for complex systems, or where it is necessary to investigate the effect of parameter changes. This is most satisfactorily achieved by computer solution of the governing equations, a very convenient method being by simulating the system on an analogue computer, or on a digital computer using a high level simulation language.

3.1 Analogue and digital computers

An analogue computer is a machine in which various physical components can be selected and interconnected in such a way that the equations describing the resulting computer arrangement are the same as those describing the physical behaviour of the system to be studied. The computer arrangement is then analogous to the system. It is a continuous data device which operates in a real time parallel mode, making it particularly suitable for the solution of differential equations and hence for the simulation of dynamic systems. Almost the only type now in use is the electronic analogue computer, in which voltages at various points within the programmed computer circuit represent the variable quantities of the system being simulated. The ease of use, and the direct interactive control which the engineer has over the running of such a computer, allows full scope for engineering intuition and makes it an invaluable tool for the analysis of dynamic systems and the synthesis of any associated controllers. A facility which is frequently helpful is that of being able to slow down or speed up the problem solution. The accuracy of solution, since it is dependent on analogue devices, is generally only of the order of a few

percent but, for the purposes of system analysis and design, higher accuracy is seldom necessary; also this accuracy often matches the quality of the available input data.

The digital computer, by contrast, is a discrete data machine which basically operates in a serial mode and with which real time operation is generally not possible. It is a highly sophisticated calculating machine which performs simple arithmetic operations sequentially at very high speed. It has extensive capacity for storage of information; this memory is a very important feature of the machine and makes it different from the simple calculating machine. Solution of system equations is effected by writing appropriate programs but, with the modern simulation languages which are now available, this can be a simple process very similar to programming for an analogue computer. The accuracy of solution can be made as high as required by selecting a small enough sampling interval for the discrete data being handled, but the penalty is one of long computation times, particularly with the relatively inefficient compilation associated with high level simulation languages. With the development of interactive systems utilizing visual display units the advantage of a close manmachine relationship is being extended to many digital computer installations.

Because of the ability of both digital and analogue computers to solve complicated mathematical equations at high speed, digital or analogue computational elements are often incorporated as part of the control system or as the complete control system.

A number of *hybrid* machines were developed to combine the two and enable a problem to be programmed in such a way that each machine is used for the computations which it performs most efficiently. Much of the cost of such a machine arises from the rather complicated interface equipment which is required to make the necessary conversions between analogue and digital signals, and vice versa. Specific computational needs arising with certain control problems have contributed significantly to the development of the hybrid computer as, in earlier years, they did with the analogue computer. The cost of an analogue or hybrid computer is now justified only for certain large simulations where fast computing speed attained by parallel operation is important.

3.2 Basic analogue computer elements (linear)

(a) Summing amplifier. The basic building block of the analogue computer is the high gain d.c. amplifier, represented schematically by Fig. 3.1. When the input voltage is $e_i(t)$ then the output voltage is given by

$$e_0(t) = -Ae_i(t) \qquad 3.1$$

where A, the amplifier voltage gain, is a large constant value.



Fig. 3.1 High gain d.c. amplifier

When this is used in conjunction with a resistance network, as shown in Fig. 3.2, then the resulting circuit can be used to add a number of voltages. Let V_{11}



Fig. 3.2 Circuit for summing amplifier

 V_2 , V_3 ... be voltages, relative to some base potential, which can be functions of time; also let these be applied to resistors R_1 , R_2 , R_3 ... At any given instant, applying Kirchoff's 1st Law to the summing junction, the point where the outputs of these resistors are connected,

$$i_1 + i_2 + i_3 = i_f + i_a$$
 3.2

where i_f is the current through the feedback resistor, and i_a is the input current into the high gain amplifier. If V_s and V_0 are the voltages at input and output of the amplifier respectively, then

$$\frac{V_1 - V_s}{R_1} + \frac{V_2 - V_s}{R_2} + \frac{V_3 - V_s}{R_3} = i_a + \frac{V_s - V_0}{R_f}$$

Now the amplifier voltage gain A will be of the order of 10^8 and the computer will have an operational range of ± 100 volts, or in some cases ± 10 volts. Hence, for V_0 to remain within this range, V_s must not exceed about 10^{-6} volts; it is then negligible compared to V_1 , V_2 , V_3 and V_0 and is virtually at earth potential. The summing junction is thus referred to as a virtual earth point.

Also, the input impedance of the amplifier will be of the order of 10^{10} ohms, compared to 10^{6} ohms for the feedback resistor. Hence i_{a} can also be neglected.

$$\therefore \quad \frac{V_1}{R_1} + \frac{V_2}{R_2} + \frac{V_3}{R_3} = -\frac{V_0}{R_f}$$

$$\therefore \quad V_0 = -\left(\frac{R_f}{R_1} V_1 + \frac{R_f}{R_2} V_2 + \frac{R_f}{R_3} V_3 + \dots\right) \qquad 3.3$$

If $R_1 = R_2 = R_3 = ... = R_f$ then

 $V_0 = -(V_1 + V_2 + V_3 + \dots)$ addition of voltages.

If there is only one voltage input

$$V_0 = -\frac{R_f}{R_1} V_1 \dots$$
 multiplication by a constant.

It should be noted that in all cases there is a sign inversion. Usually the available ratios R_f/R_1 etc. are standardized to 1 and 10, the appropriate gain being selectable as required. The complete circuit comprising high gain amplifier, input resistors, and feedback element is termed an *operational amplifier*, and for it to act as a summing amplifier the feedback element must be a resistor. It is given a symbol as shown in Fig. 3.3.



Fig. 3.3 Symbol for summing amplifier

(b) Coefficient potentiometer. In order to multiply a voltage by a constant other than 10, use is made of a grounded potentiometer (usually a 10-turn helical potentiometer), as shown in Fig. 3.4. This permits multiplication by a constant in the range 0 to 1. If larger values are required then one or more summing amplifiers with gain 10 must be used in series with the potentiometer.





Coefficient potentiometers are generally set to the desired value when in circuit, as will be described in Section 3.4. Since the setting is affected by the loading, the load resistance should be constant, hence two potentiometers should not be placed in series but should be buffered by placing an amplifier between them.

(c) Integrating amplifier. The circuit for the integrating amplifier shown in Fig. 3.5a is similar to that for the summing amplifier, the difference being that there





3.3 Production of circuit diagrams to solve differential equations 33

is a capacitor instead of a resistor in the feedback path. Again the summing junction is a virtual earth point and i_a is negligible.

$$\frac{V_1}{R_1} + \frac{V_2}{R_2} + \frac{V_3}{R_3} = -C \frac{dV_0}{dt}$$

$$V_0 = -\int_0^t \left(\frac{1}{R_1C} V_1 + \frac{1}{R_2C} V_2 + \dots\right) dt \qquad 3.4$$

The values $\frac{1}{RC}$, the time constants of the integration, vary the integration rate, and again are generally standardized at 1 and 10. An initial condition voltage can be applied to the capacitance, and integration would then commence from this value. This involves the inclusion of two additional resistors as shown in Fig. 3.13. Fig. 3.5b shows the diagrammatic representation for the integrating amplifier with three inputs and an initial condition value applied. As with the summing amplifier there is always a sign inversion.

3.3 Production of circuit diagrams to solve differential equations

The three basic elements described in the section above are sufficient to simulate any linear system and hence to solve the corresponding differential equation. To illustrate the technique for deriving the computer circuit diagram consider a system with differential equation

$$A\ddot{c}(t) + B\dot{c}(t) + c(t) = u(t)$$
 3.5

The transfer function corresponding to this is

$$\frac{C(s)}{U(s)} = \frac{1}{1 + Bs + As^2}$$
 3.6

The following steps are carried out:

(a) The equation is rearranged so that the highest derivative term is on the left hand side and all other terms are on the right hand side of the equation:

i.e.
$$A\ddot{c}(t) = u(t) - c(t) - B\dot{c}(t)$$
 3.7

(b) It is assumed that a voltage representing the highest derivative term is available, then lower derivative terms can be obtained by successive integrations. For a second order differential equation, as in this case, two integrators are required (Fig. 3.6).



Fig. 3.6 Successive integration

(c) Potentiometers and summing amplifiers are used to obtain the correct coefficients and signs of the lower derivative terms on the right hand side of the equation, and these signals are added to u(t) in a summing amplifier to produce $A\ddot{c}(t)$ at the point where it was assumed to be available (Fig. 3.7).



(d) For identification purposes, numbers are assigned to the amplifiers and potentiometers on the circuit diagram; the interconnections are then made externally on the patch panel of the computer by means of plugs and leads.

(e) The system can now be forced with any desired voltage waveform representing u(t) and the resulting response c(t) can be observed and recorded if required. Potentiometer values can be altered to represent variation of system parameters and the resulting system behaviour change noted. Other system quantities, in this case $-\dot{c}(t)$ and $\ddot{c}(t)$, can also be monitored if desired. If c(t) and $\dot{c}(t)$ are required to have values other than zero at the start of computation then these initial condition values are applied as voltages to the initial condition inputs of amplifiers 03 and 02.

Normally the circuit diagrams might be modified for a number of reasons:

(i) The circuit can generally be rearranged slightly to reduce the number of amplifiers used. In this example if the signal $\ddot{c}(t)$ does not require to be monitored amplifiers 01 and 02 can be combined as in Fig. 3.8, reducing the number of amplifiers needed from four to three.



Fig. 3.8 Circuit of Fig. 3.7 using one less amplifier

3.3 Production of circuit diagrams to solve differential equations 35

(ii) The circuit may be rearranged so that each variable parameter is represented by a single potentiometer. Fig. 3.7 is better than Fig. 3.8 in this respect. Care must be taken to ensure that potentiometers are not placed in series, since alteration of one potentiometer setting would then affect the setting of the other.

(iii) It may be found necessary to adjust the scaling of the problem to avoid, in certain parts of the circuit, voltages which are either so large that they exceed the linear range of the amplifiers, or so small that the poor signal to noise ratio causes significant inaccuracies. Also it may be found desirable to adjust the time scale of the problem to slow down or speed up the solution. The principles of amplitude scaling and time scaling will be described in Section 3.7.

Example 3.1. For the block diagram of Fig. 3.9 consider the derivation of an analogue computer circuit diagram which could be used to investigate the



Fig. 3.9 Block diagram of feedback control system

effect on dynamic behaviour of the system of changes in settings of the proportional plus integral governor. Assume that the operational amplifiers available have gains of 1 or 10.

It would be possible to reduce this diagram by the methods illustrated in Fig. 2.3 to one single block with input R(s), output C(s), and transfer function which is of fifth order. The fifth order differential equation which this represents could then be simulated by following the above procedure and using 5 integrators in series, from which would be obtained the 5 signals proportional to c(t), $\dot{c}(t)$, $\ddot{c}(t)$, etc. which must be summed with r(t) to produce the highest order derivative term. (There would in this case be a difficulty, namely the need also to obtain derivatives of the input signal; the method of overcoming this is described in Section 3.6.) It is, however, much more useful and also simpler to draw a circuit diagram for each system component separately, and then to combine these in the appropriate way to obtain a circuit diagram for the overall system. This then gives a true simulation where all intermediate system variables can be easily identified and monitored.

Obtain first a circuit diagram for the plant, noting that it comprises a first and a second order component which are in series, as represented by Fig. 3.10a.





For block $G_1(s)$

$$\frac{C_2(s)}{C_1(s)} = \frac{1}{s+1}$$

$$\therefore \quad sC_2(s) = C_1(s) - C_2(s)$$

$$\dot{c}_2(t) = c_1(t) - c_2(t)$$

or

and this requires the circuit diagram of Fig. 3.10b. For block $G_2(s)$

$$\frac{C(s)}{C_2(s)} = \frac{1}{s^2 + 4s + 8}$$

$$\therefore \quad s^2 C(s) = C_2(s) - 4sC(s) - 8C(s)$$

$$\ddot{c}(t) = c_2(t) - 4\dot{c}(t) - 8c(t)$$

or

which can be simulated by the circuit of Fig. 3.10c. The summer and the first integrator can be combined requiring one amplifier less. The constant 3.5



requires a potentiometer set to 0.35 in conjunction with an input into an amplifier in the gain 10 position.

The circuit for the transducer is similar to that for $G_1(s)$, except that a potentiometer is required to give the time constant of 0.2 seconds. A summing amplifier enables e(t) to be obtained from r(t) and the transducer output. The governor output is obtained by addition of e(t) and a constant *a* times $\int e(t) dt$, the whole being multiplied by the constant *K*. Combination of all of these parts, paying careful attention to ensure that the signs of signals are correct, yields the overall circuit diagram, Fig. 3.11.

3.4 Computer operating modes

If there is a voltage at the input of an integrating amplifier, then the output will increase continually until the amplifier saturates. It is necessary therefore to be able to control the operation of the amplifiers by starting and stopping the computation as required. Also it is necessary to be able to adjust potentiometers to their required settings, and to be able to set the circuit variables to the desired initial conditions. These are achieved by means of a mode control switch or push buttons which energize relays in the amplifier and potentiometer circuits in the way described below for the usual operating modes.

(a) 'POT SET'. This mode enables each potentiometer to be set while in circuit, and loaded by the normal amplifier input resistance. Switching to this mode connects the reference voltage supply, +100 volts on many machines, to the input of all potentiometers and connects the summing junction (SJ) of all amplifiers to earth potential as shown in Fig. 3.12. By switching a digital



Fig. 3.12 Potentiometer and amplifier in 'POT SET' mode

voltmeter (DVM) to display a potentiometer output voltage, each potentiometer can in turn be adjusted to its required setting.

(b) 'RESET' or 'IC'. Two relays are actuated in each integrator circuit, as shown in Fig. 3.13, thus applying initial condition voltages to the feedback capacitors and setting each integrator output to the initial value of the vari-



Fig. 3.13 Operational amplifiers in 'RESET' mode

able which it represents. (The voltage applied to the initial condition socket must be of opposite polarity to V_{ic} .) When no connection is made to the initial condition socket, the amplifier starts from an initial voltage of zero. The summing amplifiers remain in their normal operating state.

(c) 'COMPUTE', or 'OPERATE'. The initial condition voltages are disconnected and the integrator inputs are connected, the integrators start, and the computation proceeds. Any variables of interest can be observed on an oscilloscope or can be recorded.

(d) 'HOLD'. Switching the computer to HOLD causes the integrator inputs to be disconnected and the problem to be 'frozen', Fig. 3.14. The circuit voltages



Fig. 3.14 Operational amplifiers in 'HOLD' mode

can thus be measured at a given instant of time. Returning to the COMPUTE mode allows the problem solution to continue from the point at which it was frozen.

(e) '*REP OP*'. Sometimes a repetitive operation mode is available allowing automatic and continuous cycling between *RESET* and *COMPUTE*.

On analogue computers intended for use in a hybrid installation, electronic switching controlled by logic signals replaces the mechanical switches in the amplifier circuits. With electronic mode control, there is usually available fast integration, achieved by using capacitors of about 1% or so of the normal values. In the *REP OP* mode the complete response can then be viewed as a

persistent trace on an oscilloscope, and the effect of a gradually changing potentiometer setting is very clearly seen.

3.5 Non-linear analogue computer components

The analogue computer is particularly useful for simulating systems where non-linearities are present, analytical solution then generally being complex or impossible. To deal with non-linearities, several other computer elements are available, and these are described briefly below.

(a) *Multipliers*. These enable two variables to be multiplied together or divided by one another, or the square root of one obtained, and are available in three main forms. Servo multipliers operate by positioning a shaft, connected to the wiper of a potentiometer, to an angular position proportional to the first voltage, applying the second voltage to the potentiometer winding, and taking the product from the wiper. By having several potentiometers ganged together one variable can be multiplied by a number of others. Being electromechanical these are not suitable for rapidly varying signals. Quarter square multipliers make use of electronic squaring circuits to mechanize the relationship

$$\frac{1}{4}[(V_1 + V_2)^2 - (V_1 - V_2)^2] = V_1 V_2 \qquad 3.8$$

Time division multipliers alter the amplitude and mark/space ratio of a signal in proportion to the two signals and filter the resulting waveform to give a mean value representing the product.

(b) Servo resolvers. These are servo multipliers which have specially wound sine/cosine potentiometers and produce the functions $\sin V$ and $\cos V$ when the input variable is V. Their use is of importance where there is transformation between rectangular and polar coordinates.

(c) Diode function generators. The relationship $V_2 = f(V_1)$ can be simulated by approximating the curved relationship by a series of straight-line segments. The break points and the slopes of the segments must be set prior to running the simulation.

(d) Comparator relays. A relay which can be incorporated in the circuit is switched one way or the other according to whether the sign of the sum of two signals $(V_1 + V_2)$ is positive or negative. This allows different parts of circuit to be connected depending on the magnitude of a variable V_1 relative to a reference variable V_2 . For high speed repetitive operation the mechanical relay is replaced by its electronic counterpart, a D to A switch.

In addition to these components which are built into the computer, effects such as saturation, deadband, and hysteresis can be simulated using external diodes and appropriate circuits. It is also possible to include a human being or items of physical equipment as part of the simulation.

3.6 Differentiation

The method of solving a differential equation by analogue computation, which has been described in Section 3.3, requires the equation to be written in such a form that a circuit diagram can be produced which involves integration of system variables rather than differentiation. Although differentiation could be achieved by means of a high gain amplifier with a capacitor at the input and a resistor in the feedback path, it is something to be avoided since differentiation is a noise amplifying process. An unwanted noise signal $A \sin \omega t$, say, with amplitude A and frequency ω , would become a noise signal of amplitude ωA after differentiation, or $\omega^2 A$ after two stages of differentiation. Further problems arise when step changes of variables occur. Wherever possible, therefore, the equations must be manipulated so that integrators can be employed.

A form of transfer function for which additional manipulation is required is one which has a polynomial in s on the numerator in addition to that on the denominator. The corresponding differential equation thus contains terms involving derivatives of the input function in addition to the derivatives of the output function. Consider the transfer function

$$\frac{C(s)}{U(s)} = \frac{2s^2 + s + 1}{s^3 + 5s^2 + 6s + 1}$$
3.9

which represents the differential equation

$$\ddot{c}(t) + 5\ddot{c}(t) + 6\dot{c}(t) + c(t) = 2\ddot{u}(t) + \dot{u}(t) + u(t)$$
3.10

To simulate Eq. 3.9 introduce a new variable $C_1(s)$ so that

$$\frac{C_1(s)}{U(s)} = \frac{1}{s^3 + 5s^2 + 6s + 1}$$
3.11

The output C(s) is then given by

$$C(s) = (2s^2 + s + 1)C_1(s)$$
 3.12

Equation 3.11 can be tackled by the methods of Section 3.3, and the output can then be obtained by addition of the appropriate derivative terms of the new variable as given by Eq. 3.12. The computer circuit which results is shown in Fig. 3.15.

Situations sometimes occur where differentiation cannot be avoided. Satisfactory results can then usually be obtained by the use of a circuit giving an approximation to differentiation, such as that shown in Fig. 3.16 with transfer function

$$\frac{C(s)}{U(s)} = \frac{s}{1 + (1 - k)s}$$
3.13

It can be seen that the differential term is modified by the presence of a simple lag term with time constant (1 - k). It will be shown in Chapter 6 that this has the effect of attenuating high frequency components of signal input. In practice the potentiometer setting can be adjusted to be as close to unity as the noise permits.



Fig. 3.15 Circuit diagram for $\frac{C(s)}{U(s)} = \frac{2s^2 + s + 1}{s^3 + 5s^2 + 6s + 1}$





3.7 Problem scaling

Circuit diagrams obtained by the method described in Section 3.3 usually require some further modification, often of a minor nature, in order to obtain acceptable accuracy of solution by minimizing the effect of inherent physical limitations of the computer elements and of the recording device, and perhaps also to alter the speed of computation for reasons of convenience. For any given input signal the maximum values of the voltage signals within the circuit, which depend on the maximum values of each variable and its derivatives, must neither be so large that the linear range of one or more amplifiers is exceeded, with resulting errors in the solution, nor so small that poor signal-noise ratios cause unacceptable inaccuracy. An excessively slow solution is wasteful of computer and operator time and can introduce inaccuracy caused by integration of small voltage errors over a long period. At the other extreme a very short solution time can cause errors due to the limitations of the dynamic response of certain of the elements being used. Modification to a circuit diagram to minimize these problems takes two forms referred to as *time scaling*, where the solution time is altered, and *amplitude scaling*, where the variables are scaled to ensure that voltage levels throughout the circuit are as high as possible without exceeding the reference value.

Inspection of the coefficients of the system equations and the gain values in the circuit diagram often suggests whether or not scaling is likely to be needed: suitable modifications can then be determined in a number of ways. It is not inappropriate to tackle the problem by a practical trial and observation approach, first making any changes which suggest themselves by inspection of the circuit diagram, patching in the resulting circuit on the computer, observing the response of the output voltage and of the voltages elsewhere in the circuit to see whether further scaling is desirable, and then making additional modifications if required. For any circuit the computer variables are directly proportional to the problem variables in such a way that α volts, say, represent one unit of the corresponding variable. The magnitude of the forcing function voltage can be increased so that the operating voltages within the circuit become as large as possible, without anywhere exceeding the computer reference voltage, and preferably until the scaling constant α is a convenient number. Provided that the problem solution time is reasonable and that voltages are nowhere so small that unacceptable errors are likely to occur then no further action on scaling is required.

If the solution time must be increased or decreased then time scaling is required. This is effected by a change of time variable

 $t = \beta T$

where t is the problem time, T is the computer or machine time, and β is a constant which is greater than unity if the solution is to be speeded up, and less than unity if it is to be slowed down. Hence 1 problem second = β machine seconds, and when $\beta = 1$ the solution is referred to as a *real time simulation*. This change of variable causes a change in the magnitudes of the time derivatives of any problem variable such as c(t), hence

$$\frac{\mathrm{d}c(t)}{\mathrm{d}t} = \frac{\mathrm{d}c(T)}{\mathrm{d}T}\frac{\mathrm{d}T}{\mathrm{d}t} = \frac{1}{\beta}\frac{\mathrm{d}c(T)}{\mathrm{d}T}, \quad \frac{\mathrm{d}^2c(t)}{\mathrm{d}t^2} = \frac{1}{\beta^2}\frac{\mathrm{d}^2c(T)}{\mathrm{d}T^2}, \quad \text{etc.}$$

The differential equation when written in terms of T is therefore the same as the original equation but with the coefficient of each derivative term multiplied by the appropriate power of $\frac{1}{\beta}$. The effect on the circuit diagram is to require the gain of each integrator to be changed by the factor β . As illustration consider the differential equation Eq. 3.5

$$A\ddot{c}(t) + B\dot{c}(t) + c(t) = u(t)$$

which in terms of machine time T becomes

$$\frac{A}{\beta^2}\ddot{c}(T) + \frac{B}{\beta}\dot{c}(T) + c(T) = u(T)$$

This has the circuit diagram shown in Fig. 3.17 where for ease of comparison with Fig. 3.7 the integrating amplifier gains have been shown as having the



Fig. 3.17 Analogue computer circuit corresponding to Fig. 3.7 for solution speeded up by a factor β

value β . Since amplifier gains normally have values of 1 or 10 only, the time scaling would be straightforward for $\beta = 10$ but would require the introduction of additional potentiometers, and perhaps also summing amplifiers, to accommodate other values of β . If when first drawing a circuit diagram amplifier gains are consistently high, as occurs inherently in equations for fast systems, then it is likely that the solution time should be increased whilst, if potentiometer settings are consistently low, it is probable that the time scale should be altered to speed up the solution. On many machines the integrating amplifier gains and hence the time scale can be changed automatically by a switch on the operating console, enabling the system to be studied visually on an oscilloscope using repetitive operation and a fast solution time, and then slowed down to within the dynamic range of a pen recorder whenever a hard copy trace is required.

The need for amplitude scaling is indicated if gains are very high or very low in certain parts only of the circuit, since the accuracy decreases as the signal amplitude reduces. If somewhere in a loop there is a gain of 100 or an attenuation of 0.01, say, then there is at least one point in the circuit where the voltage can at most be 1% of the reference voltage, and consequently small absolute errors in this voltage could cause large percentage errors in the solution. The procedure for amplitude scaling is not easily learned by reading alone and can best be understood by practical application on a computer. It is a compromise between obtaining the highest possible accuracy of solution and having a simulation where interpretation is easy and where least confusion can arise. The procedure is as follows:

- (i) estimate the maximum values of each of the variables $x_1(t)$, $x_2(t)$, $x_3(t)$... $x_r(t)$... for the forcing function of interest
- (ii) determine scaling factors A₁, A₂, ... A_r, ... which are simple numbers and equal to or slightly less than the corresponding value of the ratio

reference voltage

maximum expected value of $x_{r}(t)$

- (iii) rewrite the differential equations in terms of scaled variables $A_1x_1(t)$, $A_2x_2(t), \ldots A_rx_r(t) \ldots$
- (iv) draw the circuit diagram by the method of Section 3.3
- (v) try the circuit, observe maximum voltages throughout, and readjust where necessary.

The result is that instead of each variable being represented by the voltage at an appropriate point in the circuit multiplied by the common scaling factor α , the variables appear in a normalized form and effectively have different scaling factors. This procedure has the effect of levelling out the maximum voltages around the various loops of the circuit, avoiding high gains being followed by large attenuation or vice versa, and ensuring that each amplifier has a range of operating voltage which is as large as possible.

For many problems, where high accuracy of solution is not of prime importance, the above procedure for amplitude scaling need not be followed fully. The aim can be simply to arrange potentiometer positions and values, and amplifier gain settings, in a sensible manner when first drawing the circuit, and then to carry out the last step of the procedure, making adjustments if they appear necessary in a logical practical manner.

Example 3.2. To illustrate the features of time and amplitude scaling consider the differential equation Eq. 3.5 in which A = 125 and B = 5

i.e.
$$125\ddot{c}(t) + 5\dot{c}(t) + c(t) = u(t)$$

and prepare a circuit diagram which could be used to investigate the response to a step change of input for varying values of A and B. Assume that the computer reference voltage is 100 volts, and that the available amplifier gains are 1 and 10.

The circuit diagram for A = 125 and B = 5 is that derived in section 3.3 (Fig. 3.7) which, with the potentiometers set to the appropriate values, is shown in Fig. 3.18a. It can be observed that the voltages at the circuit input and at the outputs of potentiometer P_{01} and amplifiers 02 and 03 are directly proportional to -u(t), $\ddot{c}(t)$, $-\dot{c}(t)$ and c(t) respectively. For the value A = 125 the setting of potentiometer P_{01} is very small, and hence to avoid errors which would arise from overloading of amplifier 01 the output voltage of P_{01} cannot be allowed to exceed 0.8 volt. The small magnitude of this voltage which is the input to the integrating amplifier 02 could lead to significant inaccuracy, and also suggests that the solution time may be long. Applying a step change of 50 volts at the input, and measuring the resulting peak voltages throughout the circuit, yields the approximate voltage values shown in the diagram. The large







on a 100 volt reference machine.

Fig. 3.18 Analogue computer circuit diagrams for simulation of differential equation $A\ddot{c}(t) + B\dot{c}(t) + c(t) = u(t)$ for A = 125, B = 5 (a) basic circuit (b) circuit modified by amplitude scaling (c) circuit further modified to decrease solution time by factor of 10

range of values suggests that amplitude scaling is desirable and study of the diagram suggests where changes could be made to avoid very small voltages. Initial experimental tests also show that the settling time is of the order of 200 seconds which is inconveniently large; this confirms the desirability of time scaling the problem. Fig. 3.18b shows the circuit and the resulting peak voltages after carrying out amplitude scaling by simple trial and observation or by the method described above. The attenuation of 0.008 is carried out in two stages with the result that the voltage output of potentiometer P_{01} now represents the signal $25\ddot{c}(t)$ rather than the signal $\ddot{c}(t)$. The loop gains remain identical at 0.04 for the inner loop and 0.008 for the outer loop. Relatively small voltages at certain points (in this case the input to amplifier 03) cannot be avoided. The solution can be speeded up, say by a factor of 10, merely by altering the inputs of integrating amplifiers 02 and 03 to gain 10 (Fig. 3.18c). The voltage levels remain unchanged. For a complex simulation such a change is particularly convenient if push button time scaling is available. To decrease the solution time by a factor of 20 say, the settings of potentiometers P_{01} and Pos are also adjusted. With time scaling the loop gains are altered, in the case of Fig. 3.18c to 0.4 and 0.8 respectively. Study of the variations in the form of the transient response for changing values of coefficients A and B can now be carried out by altering the settings of potentiometers Po1 and Po2 to the appropriate values. This is a particularly simple simulation which, however, demonstrates the method of approach which would be used to obtain an actual simulation diagram for a more complex dynamic model.

3.8 Digital continuous system simulation

An analogue computer simultaneously solves all of the differential equations which form the model for a physical system, and continuous voltage signals represent the variables of interest. This enables the machine to operate in real time which permits the incorporation within the simulation of actual items of equipment or of human beings where these form part of the system to be studied. Significant disadvantages of analogue simulation are the high cost of the computer due to the multiplicity of elements with demanding performance specifications, difficulties of problem scaling to avoid overloading of amplifiers, and relatively limited accuracy and repeatability due in part to amplifier drift. As a consequence of the very rapid development of digital computer hardware and software giving ever greater capability and flexibility at decreasing cost, system simulation is inevitably being carried out more and more on the digital computer. There is effectively no problem of overloading so very wide ranges of parameter variation can readily be accommodated, any desirable accuracy can be attained, and with the aid of appropriate high level languages program writing is straightforward. With the availability of the 'on-line' computer facility a simulation can be run interactively in a closely similar manner to that of the analogue computer.

The solution of a differential equation involves the process of integration, and for the digital computer analytical integration must be replaced by some numerical method which yields an approximation to the true solution. A continuous signal x(t) is represented by a series of numbers $x_0, x_1, x_2, x_3, \ldots, x_n$.

say, which define the signal amplitude at times t_0 , t_1 , t_2 , t_3 , ..., t_n . These sample values are normally at equally spaced time intervals, and if the sampling interval is chosen to be small enough then no information about the signal is lost (see Section 7.5). With such discrete representation of continuous signals differential equations are converted to difference equations and integration is carried out in a stepwise fashion. Integration of a signal x(t) of known form (where the sample values $x_0, x_1, x_2, \ldots x_n$ are known at the outset) can be effected by means of the trapezoidal rule

$$\int x(t) \, \mathrm{d}t = \left(\frac{x_0 + x_1}{2} + \frac{x_1 + x_2}{2} + \frac{x_2 + x_3}{2} + \dots + \frac{x_{n-1} + x_n}{2}\right) \Delta t$$

The solution of a differential equation, however, requires integration of a signal $\dot{x}(t)$ which is itself a function of x(t). Consider the first order differential equation

$$\dot{\mathbf{x}}(t) = a\mathbf{x}(t) + b\mathbf{u}(t) \tag{3.14}$$

If the value of x(t) at time t is known then the value at time $t + \Delta t$ is given by

$$x(t + \Delta t) = x(t) + \int_{t}^{t} \dot{x}(t) dt$$

10.00

but to evaluate the integral term it is necessary to know $x(t + \Delta t)!$ Numerous *integration algorithms* are available to overcome this difficulty and enable the next value of x(t) to be estimated. To solve a differential equation the unknown solution trajectory $x_1, x_2, x_3, \ldots x_n$ is built up progressively, one integration time step at a time, starting from a known value x_0 .

The simplest integration algorithm is the *Euler method* which assumes that the function to be integrated, the derivative function, remains unchanged from t to $t + \Delta t$ with the value which it has at time t, i.e. $\dot{x}(t)$. Hence

$$x(t + \Delta t) = x(t) + \Delta t \hat{x}(t)$$
3.15

which shows that the values x(t) and $\dot{x}(t)$ are used to estimate $x(t + \Delta t)$. For any specified input function u(t) starting from a known initial output value x(0)equations 3.14 and 3.15 can be alternately and repeatedly applied to calculate successive values of the output function as follows:

$$\begin{aligned} x(\Delta t) &= x(0) + \Delta t \{ ax(0) + bu(0) \} \\ x(2\Delta t) &= x(\Delta t) + \Delta t \{ ax(\Delta t) + bu(\Delta t) \} \\ x(3\Delta t) &= x(2\Delta t) + \Delta t \{ ax(2\Delta t) + bu(2\Delta t) \} \\ \text{etc.} \end{aligned}$$

To help visualize this computational procedure consider Eq. 3.14 in which a = -5 and b = 1

i.e.
$$\dot{x}(t) = u(t) - 5x(t)$$
 3.16

This corresponds to the transfer function

$$\frac{X(s)}{U(s)} = \frac{1}{s+5} = \frac{0.2}{1+0.2s}$$

3.17

which is a simple lag system with time constant 0.2 seconds and gain 0.2. For a unit step input function, u(t) = 1, an initial value x(0) = 0, and an integration step size of $\Delta t = 0.1$ seconds the output is given in discrete form by

$$x(t + 0.1) = x(t) + 0.1\{1 - 5x(t)\}$$

i.e. $x(t + 0.1) = 0.5x(t) + 0.1$

The successive output sample values are thus

 $\begin{array}{l} x(0) &= 0 \\ x(0.1) &= 0.5(0.0000) + 0.1 = 0.1000 \\ x(0.2) &= 0.5(0.1000) + 0.1 = 0.1500 \\ x(0.3) &= 0.5(0.1500) + 0.1 = 0.1750 \\ x(0.4) &= 0.5(0.1750) + 0.1 = 0.1875 \\ \text{etc.} \end{array}$

For comparison these discrete values together with the true analytical solution $x(t) = 0.2(1 - e^{-5t})$ are plotted in Fig. 3.19 and it can be seen that there is a significant difference in the solution trajectories. Reduction of the integration time step size Δt improves the accuracy of the solution at the expense of an increase in computation time. If Δt is reduced by a factor of 2 then the discrete equation becomes

$$x(t + 0.05) = 0.75x(t) + 0.05$$
 3.18



Fig. 3.19 Step response of 1st order system evaluated analytically and using discrete algorithms

from which x(0.05) = 0.0500, x(0.1) = 0.0875, x(0.15) = 0.1156, x(0.2) = 0.1367, ... There are twice as many computations, but the errors are approximately halved.

The Euler method is a helpful introduction for understanding the principles of integration algorithms but is not often used in practice since it employs a poor estimate of the mean value of the derivative function for the time interval Δt . Better accuracy for a given integration step size is achieved by making use of multistage algorithms. The *improved Euler method* is a two stage computation in which a first estimate of the next point is made using the Euler method, the derivative is calculated for this estimated point, the average of this derivative value and that at the beginning of the step is evaluated and this average value used to calculate the next point. As an illustration of the method, consider again the first order system of Eq. 3.16 with a unit step input and an integration step size of 0.1. As before

$$\dot{\mathbf{x}}(t) = 1 - 5\mathbf{x}(t)$$

A first estimate of x(t + 0.1) is given by

$$x_e(t + 0.1) = x(t) + 0.1\{1 - 5x(t)\} = 0.5x(t) + 0.1$$

The derivative value at this point is

$$\dot{x}_{s}(t+0.1) = 1 - 5\{0.5x(t) + 0.1\} = 0.5 - 2.5x(t)$$

The average derivative value is then

$$\frac{1}{2} \{\dot{x}(t) + \dot{x}_{e}(t+0.1)\} = 0.75 - 3.75x(t)$$

which is used to obtain an improved and final estimate of x(t + 0.1) as

$$x(t+0.1) = x(t) + 0.1\{0.75 - 3.75x(t)\} = 0.625x(t) + 0.075$$
3.19

The solution using Eq. 3.19 shows a marked improvement in accuracy. In this example the solution is particularly simple since u(t) is a constant and thus $\dot{x}(t)$ is a function only of x(t), resulting in one single equation to be repetitively solved. Normally $\dot{x}(t)$, $x_e(t + \Delta t)$, $\dot{x}_e(t + \Delta t)$, \dot{x}_{mean} , and $x(t + \Delta t)$ must all be computed in turn for each integration time interval Δt . A similar two stage method referred to as the modified Euler method makes a first estimate of the next point, evaluates the derivative at the estimated midpoint, and uses this as the derivative value to calculate the next point. A more complex and efficient algorithm which, like the above, uses only the current value to estimate the next value, but estimates three or more usually four derivative values to do so, is the Runge-Kutta method. A different class of integration algorithm is that of predictor-corrector methods which make use of both present and past values to predict the next value, and then correct the predicted value by an appropriate algorithm, the prediction and the correction sometimes being done iteratively. A complication is that at the start there are no past values, hence one of the Runge-Kutta class of algorithms must be used for the first one or two steps of integration. Many algorithms are available in standard software libraries and are documented in textbooks such as references 17 and 18.

3.8 Digital continuous system simulation 51

The choice of value for the time interval Δt , referred to as the integration step size or integration interval, must clearly be a sensible one. Using the basic Euler method the above example suggests that the integration step size should be of the order of one tenth of the time constant or smaller (corresponding to a sampling frequency of approximately 60 times the bandwidth, a concept which is discussed in Section 6.5). If a large step size is chosen then quite erroneous results may be obtained since not only is the error significant but oscillations can occur, as illustrated in Fig. 3.20 for the above numerical example with an



Fig. 3.20 Illustration of errors from use of large integration step

integration step size of 0.3 seconds. The integration step size can be larger, say one quarter of the time constant, if using a better algorithm. With some algorithms the integration time step is not fixed but is changed automatically so that it is increased in value when changes are slow and decreased when the variable changes more rapidly.

With the above background, together with some understanding of a scientific high level language such as Fortran or Pascal the reader will be able to write a digital computer program to evaluate the time response for any chosen simple system, and to output the results in an appropriate graphical or tabular form. A sample program is outlined in the next paragraph. A quicker and more convenient approach to simulation, if appropriate software is available, is to employ a simulation language, a special programming language for dynamic system simulation designed to require a minimum knowledge of computer programming and computer procedures. Simulation languages are discussed in Section 3.9.

Figure 3.21 lists a Fortran program which can be used to evaluate c(t) for the second order differential equation, Eq. 3.5, with a unit step forcing function u(t), and is thus a digital computer equivalent of the analogue computer circuit diagram of Fig. 3.7. The heart of the program which uses the Euler approach

to integration is the iterative application of the following three statements:

```
CDD011 = (1.0 - C(1) - 8*C001(1))/A

CDD01(1 + 1) = CD01(1) + 01*C00011

C(1 + 1) = C(1) + 01*C001(1)
```

The similarity between these (which correspond to equations 3.14 and 3.15) and the analogue computer circuit will be evident. The initial part of the program is written in a form such that the system parameters A and B, the integration interval DT, the run time, and the time interval between the solution values which are to be printed out can be chosen at run time and entered

```
C SOLUTION OF SECOND ORDER DIFFERENTIAL EQUATION
            DIMENSION C(500),CDDT(500),PRTPLT(61)
DATA BLANK/1H /,CROSS/1HX/,DASHV/1H-/,DASHH/1HI/
DATA C0/0.0/,CDDT0/0.0/.CMAX/0.0/
      REQUEST DATA INPUT FOR SIMULATION RUN
            WRITE(6,10)
FORMAT('0*** STEP RESPONSE OF SECOND ORDER SYSTEM ****/
 10
       1 ' A*CODOT+8*CDOT+1=U OR C(S)/R(S)=1/(A*S**2+B*S+1)'/
2 'Ø (INSERT NUMBERS IN HEAL FORM -I.E. WITH DEC POINT)'/
            (SCDEFFICIENT "A" = ')
           READ(5,20)A
FORMAT(F10.4)
20
            WRITE(6,30)
FORMAT('$COEFFICIENT "B" = ')
 30
            READ(5,20)8
            WRITE(5,40)
FORMAT('$INTEGRATION INTERVAL (SECONDS) = ')
48
            READ(5,20)DT
WRITE(6,50)
FORMAT('$RUN TIME (MAXIMUM 500*DT, SECONDS) = ')
 50
            READ(5,20)FINTIM
            WRITE(6,60)
FURMAT('$PRINT INTERVAL (SECONDS) = ')
-60
            READ(5,20)PI
IP=TFIX(PI/DT)
            N=IFIX(FINTIM/DT)
      COMPUTE C(T) USING EULER INTEGRATION
CODOTI=(1.0-C0-8*CDET0)/A
             C(1)=CØ+DT*CDUTØ
             CDOT(1)=Ø.Ø+DT*CODDTI
             00 100 I=1.N-1
CDD0TI=(1.0-C(I)-8*CDUT(I))/A
C(I+1)=C(I)+0T*COUT(I)
                CODT(I+1)=CODT(I)+DT*CODDTI
100
               IF(C(I+1).GT.CMAX)CMAX=C(I+1)
       PRINT AND PLOT THE TIME RESPONSE
             WRITE(6,120)
FORMAT(3X, 'TIME
  120
                                          DUTPUT')
             PRTPLT(1)=CROSS
             DD 130 H=2,61
                PRTPLT(K)=DASHV
  130
             TIME=0.0
             WRITE(6,150)TIME,C0,PRIPLT
PRTPLT(1)=DASHH
             DB 140 K=2,61
PR1PLT(K)=BLANK
DO 160 J=1,N/IP
PRTPLT(1)=DASHH
140
                TIME=J*IP*DT
                K=60.0*C(J*IP)/CMAX+1
                PRTPLT(K)=CROSS
WRITE(6,150)TIME,C(3*IP),PRTPLT
FORMAT(1X,F7.3,F10.4,2X,61A1)
PRTPLT(K)=BLANK
 150
  160
                CONTINUE
             STOP
             END
```

Fig. 3.21 Fortran program to solve Eq. 3.5

interactively. The initial values of C and CDOT are selected to be zero. The third part of the program produces a simple print and plot of the output response in the form shown in Fig. 3.22 which gives the results for A = 1.0,

TIME	DUTHUT								
0.000	0.0000	XEELEE	Sa Charles		 *******	*******	**********		
8.500	0.0976	1 4							
1.000	0.3367	1		8					
1.500	0.6155	I			× .				
2.000	0.8631	I.							
2.500	1.0426	I					× 1		
3.000	1. 1444	I						- X-	
3.500	1.1785	T						- 196	
4.000	1.1638	I						. Y .	
4.500	1.1223	I						-8-	
5.000	1.0725	T							
5.500	1,0275	I					×		
5.000	0.9944	I					- 8		
6.500	4.9752	T					×		
7.000	0.9683	I					×		
7.500	0.9703	I					3.		
8.000	0.9775	T					8		
8.500	0.9864	1					×		
9.000	0.9945	I					8		
9.500	1.0006	1					x		
10.000	1.0843	1					×		
10.500	1.0056	I					8		
11.000	1.0054	1					8.		
11.500	1.0041	I					- 8		
12.000	1.0026	I					- 8		
12.500	1.0011	I					× 1		

Fig. 3.22 Typical output from program of Fig. 3.21; A = 1.0, B = 1.0, DT = 0.05

B = 1.0, and DT = 0.05. The integration interval must be small but not every output sample value calculated is (or need be) printed out. Values of CDOT(I) are also stored and with suitable addition to the third part of the program could also be printed or plotted. If initial values other than zero are to be investigated alterations can be made so that they, too, can be supplied at run time. Such an approach can be used for any other simple simulation. The experienced programmer will be able to incorporate more efficient integration algorithms and more comprehensive output, perhaps producing a family of curves to show the effect of variation of one of the parameters, with hard copy output produced on a graph plotter.

3.9 Simulation languages

A number of special programming languages referred to as continuous system simulation languages (CSSL) or simply as *simulation languages* have been developed as analytical tools which can be used to study the behaviour of a wide range of dynamic systems without the need for a detailed knowledge of computing procedures. The languages are designed to be simple to understand and use, and they minimize programming difficulty by allowing the program to be written as a sequence of relatively self-descriptive statements. The programs may not have very high computational efficiency, but learning and writing time is minimized which can make them cost effective in an engineering design situation and free the user to concentrate on interpretation of the results rather than on computational details. Familiarity with a high level programming language and an understanding of computer graphics may well

be helpful but is not necessary. Numerous different languages with acronyms such as CSMP, CSSL, DYNAMO, DARE, MIMIC AND TELSIM have been developed by computer manufacturers, software companies, universities and others, some for specific families of machines and others for wider application. Many features have now been standardized and thus a number of the languages tend to be broadly similar. Symbolic names are used for the system variables (these names being required to follow certain conventions) and the main body of the program is written as a series of simple statements based either on the system equations or on a block diagram representation of the system. To these are added statements specifying initial parameter values and values of system constants, and simple command statements controlling the running of the program and specifying the form in which the output is required. The short user written program is then automatically translated into a Fortran (or other high level language) program which is then compiled, loaded and executed to produce a time history of the variables of interest in printed or plotted form. System constants and initial conditions can be altered and the program rerun without the need to retranslate and recompile. Many of the languages are designed to be run interactively from a graphics terminal and have the facility of displaying on the screen whichever outputs are of interest and, if the solution is not developing as desired, the capability of interrupting the program, changing parameters, and rerunning immediately.

To illustrate the general nature of a simulation language and the way in which it is used CSMP (Continuous System Modelling Program), a widely available Fortran based language developed by IBM will be described in outline. The aim is to show the contrast both with analogue computer simulation and with user written programs, and to give an appreciation of the facilities available. The ease of use, power and versatility can only be appreciated by actually using the software, and the reader is strongly recommended to seek access to a simulation language and to undertake some simulation studies. The self-descriptive nature of the program statements is evident in Fig. 3.23, a CSMP program for the feedback control system of Fig. 3.9. It can be seen that lines 7 to 12 describe almost directly the mathematical equations and physical variables of the system (the whole program is shorter than that of Fig. 3.21 which is for a much simpler system).

The basic elements which appear in the statements of such a program are (i) system variables (i.e. quantities which may change in magnitude during a program run) represented by appropriately descriptive symbolic names, (ii) numerical constants, (iii) the basic arithmetical operators +, -, *, /, **, and (), (iv) functions or functional blocks for more complex mathematical operations and (v) labels, which are key words at the start of certain statements to indicate the type of statement so that it is appropriately handled in the translation phase. A program is constructed from three classes of statement:

(a) Structural statements: These define the model by relating the system variables to one another by the appropriate mathematical relationships and form the heart of the program. They are similar to Fortran statements, make wide use of functions, in particular one specifying integration, and are executed repetitively during the running of the program. (Writing these statements is

```
    CSMP SIMULATION OF SYSTEM OF FIG 3.9
    -EFFECT OF GAIN "K' AND INTEGRAL ACTION SETTING
    'A' DN RESPONSE TO A UNIT STEP CHANGE OF INPUT

CONSTANT A-0.5
PARAM H=(0.5,2.0,3.5,5.0)
F = H - F
M=B*(E+A*INTGRL(Ø.Ø.E))
C1=3.5*M
C2=REALPL(0.0,1.0,C1)
C=CMPXPL(0.0,0.0,2/SQRT(8.0),⊴QRT(8.0),C21
F=REALPL(0.0,0.2,C)
R=STEP(0.0)
TIMER DELT=0.05,FINTIM=12.0,OUTOEL=0.2
DUTPUT D
LABEL 1.
PAGE MERGE
              .... " SERVO RESPONSE FOR UNIT STEP INPUT "
BUTPUT M
              .... " CONTROLLER OUTPUT "
LABEL
          31
PAGE MERGE
END
CONSTANT A= 1.0
FNP
STOP
ENUDOB
```

Fig. 3.23 Typical CSMP program

equivalent to drawing the circuit diagram and connecting it up in analogue simulation.)

(b) Data statements: These assign numerical values to the system constants and to the initial values of system variables. (This is analogous to adjusting potentiometer settings and amplifier gains, and to setting integrator initial condition voltages.)

(c) Control statements: These specify the conditions under which the program is to run, in particular the integration step size and the finish time for the solution, and specify the form in which the output is to be presented. (The latter is equivalent to selecting analogue computer voltages for display on oscilloscope or plotter, and the finish time corresponds to the length of time in the 'Compute' mode). Control statements also determine what is to occur when the finish time is reached, whether to stop, or to alter a parameter value and repeat the computation.

CSMP has available an extensive range of *functions* which can be grouped into the categories listed below. Certain of the most important functions are included so that the reader can understand the program of Fig. 3.23 and can himself write comparable programs. For other than simple systems reference must clearly be made to a programming manual or to a descriptive textbook such as reference 19.

(i) Standard mathematical functions: All the standard Fortran mathematical functions such as sine, SIN(X), and square root, SQRT(X), are available for use within algebraic expressions.

(ii) Integration function: The main simulation language function is INTGRL (IC, X), where IC is the initial condition and X the variable to be integrated. CSMP offers a choice of 7 or more integration methods, the default being the Runge-Kutta method with variable step size, or the user may incorporate his

own routine. The input variable X can be an algebraic expression, thus a typical statement could be

HUTPUT - INTERL IVD, ERBOR * BAINS

(iii) *Transfer functions:* To facilitate simulation of a system whose mathematical model is available in block diagram form functions are provided for a simple lag, REALPL (IC, τ , X), for a quadratic lag, CMPXPL (IC1, IC2, ζ , ω_n , X) and for a lead-lag or lag-lead element. Higher order transfer functions must be split into factors by redrawing the block diagram with primitive blocks each with named input and output.

(iv) Non-linearities: These are functions for common non-linearities such as time delay, saturation, and hysteresis, for arbitrary function generation, and for sampled data system elements.

(v) *Input functions:* A unit step at time T can be called for by the function STEP(T), and corresponding functions are available for ramp, impulse, pulse and sinusoidal inputs.

(vi) User defined functions: User written Fortran subroutines can be incorporated to define functions for specific features not already included, and groups of statements can be combined to form larger functional blocks which can be called whenever needed by a one-line statement.

The first word of a data statement or control statement is referred to as a *label* and specifies the form of action required. A data statement of the form

CONSTRAT & = 1.0, C = 5.0, XDDT = 0.0

is used to assign specific values to parameters which are constant for a program run and to initial conditions. By using variable names rather than numerical values in structural statements and assigning numbers with such a statement it is easy to alter parameter values at run time simply by a new CONSTANT statement. A statement of the form

PARAM B = 10.2, 0.5, 1.0, 2.0)

calls for separate runs in which one named parameter takes the successive values listed.

TIMER DELT = W.W1, FINTIM = 10.0, FRDEL = 0.1, DUTDEL = 0.1

is the form of the statement which is required to specify the variables which control the integration step size, the finish time for the run, the time increment for printed output, and the time increment for print-plotted output. The label PRINT followed by one or more variable names specifies which variables are to be printed out, and TITLE followed by an appropriate title allows the programmer to specify the heading for each page of printout. The label PRTPLT (or OUTPUT in CSMP III) followed by one or more variable names specifies for which variables a printer plot is required and LABEL indicates the heading for the output. CSMP III, an enhancement of earlier versions of CSMP, offers such facilities as multiple curves on the same plot (PAGE MERGE), contour plots and shaded plots. The statement END indicates the end of a run. If additional runs are required with changes of variables

9	ARIABLE	MINIMUM	MASIMUM	VAR	TABL	MINIM	IIM M	AXIMUM		
	TIMI	0.000000E+0 -2.903968E-0	0 12.0000 2 5.75765	ē.		9.99	0000E+00	1.4103	3	
1	1 N	SERVO RESPONS	E FOR UNIT S	rên în	й <u>Ш</u> т -					
Ø.	MERGED OUTF	UT PRESENTATI	ON FOR E							
ø	PARAMETER	RUN 1	RUN 7	BHIN	3	HIIN -	B 4			
	H	0.50000	2.0000	3,9	003	5.000	a			
										Course -
			0.0000E+00			"D"= RDN	4			1.500
			U. UUUUE +UU			18 = HUN	3			- 500
			0.0000E+00			• = HUN	1.1			
	TIME	BUIK F	A. AAAAAE + AA			. # . ≘ . HUW				
	a addadac . a		0 10	in Se	10.0.0.0.	1000000	0	Sec. F.		Francis,
	0.0000000.+0	1 055005.0	3 11							
	0.20000	1.0000000-0	2	1				Ť		
	0.60000	3. 122925-0	2 1	n		1		Ŷ		
	0.80000	5.82712F-0	21. 4					Ť		
	1.0000	8.9679hF-7	21 +		1.15	2. 7	0	Ť		- 0
	1.2000	1. 12248	1 +			X		0 I.		- 1
	T.4000	8. 15447	1. 91	Ĩ		1	× .	1	71	1
	1.6000	0. 18432	1 .	T		· · · · · · · · · · · · · · · · · · ·		× 1	U.	T
	1.8000	2.21147	1 .	1				X I	0	1
	2.0000	0.23584	[interactor	1	*****			X I -		in a second
	2.2000	Ø.25768	L 4	1		1	•	X 0		. t
	2.4000	2.27739	1	4. J		1	• 1	18 I.		τ
	2.6000	0.29539	1	+ I			•0 A	·		
	2.8000	0,31206	L	9.1		D 1	1. X			1
	3.0000	2.32770	1	0.03		(2) 3	- X			
	3.2000	2,34257	I.	- 24		1. 1	8			+
	3.4000	N-35683	5	- 12		1.0		- 1.4		+
	3.5000	0.37029	1	1.1			2.0			
	7.0000	W. 30503	Contraction of the second							in an
	1.20000	0 10656						n 1		
	6.6000	9.42194	E.	1				· 8 1		1.1
	9.6000	0.43399	7	1	2 - 1	1		01		- 1
	4.8000	0.44577	T	1	2	1				1
	5.0000	0.45728	î		14. C	Î	· · · · ·	1 17		- 1
	5.2000	0.46853	Ĩ.	1	51	T	· ·	0 1		1
	5.4000	0.47953	T	1	(14) I	7	· D	T		- 7
	5.6000	0.49029	L	1			4 3 8	1		1.
	9.8000	0,50082	1	- 1	- A -	1		I		1
	5.0000	0.51113	I			T	-11*-X	T-		1
	B.2000	0.52121	I	1		1	B • X	1		E
	6.4000	0.53100	I	1	1.4	1	0 • x	I		1
	6.6000	0.54075	1			1	12 • X	1		1
	6.8000	0-55021	L	1		1	D X	I		5
	7.0000	4.55947	I.	1		3	+×0	4		- 15
	7.2000	0.56855	5		+ I	1	X			
	1.44444	W+2449	1	1		- 2	- 8 -	0 I		

DUTPUT VARIABLE RANGES FOR ALL RUNS IN CASL

Fig. 3.24 Part of printer plot output from CSMP program of Fig. 3.23

this is called for by defining each change after the END statement and concluding it with another END statement. The final two statements are STOP and ENDJOB, and if there is any user defined subroutine this is interposed between the two. An asterisk at the start of any line indicates that what follows is comment.

The above outline should explain the CSMP program listed in Fig. 3.23. If there is a likelihood of wishing to change any parameters other than the two controller settings, say the plant parameters, then these would be best included as variables and assigned values in the constant statement. With A = 0.5 there will be 4 runs, each evaluating the response up to a finish time of 12 seconds for a different value of K, and then the whole will be repeated for A = 1.0. For each value of A there will be a printer plot of C and one of M, each with the 4 traces superimposed. Part of one of these is shown in Fig. 3.24 (with the




printed values for runs 2, 3 and 4 omitted for clarity). Due to the severe resolution limitations of a lineprinter the plots are rather uneven, and a much improved plot is obtained if a graphics display unit or graph plotter output is available (Fig. 3.25).

It is hoped that the above gives some idea of the nature of digital simulation languages and of the advantages that they offer: simple program writing, accuracy and reproducibility, cost effectiveness, and, where interactive facilities exist, keyboard entry of program and running, inspection of plots on graphics display screen, on-line modification and rerunning.

4 Transient Response of Systems

When concerned with dynamic systems it is of interest to know how the output of the system will change as a result of specific types of input change. On the basis of some appropriate criterion, an assessment can be made of whether or not the system behaviour is satisfactory and, if not, an attempt made to improve the response by a realizable modification to the system. With practical systems the exact form of the input excitation function may be known in advance, but most frequently the input will vary in a somewhat random and hence largely unpredictable manner (such would be the case where ambient temperature is a significant input variable for process plant or heating systems).

For analysis and design certain basic input functions are chosen, mainly to bring the analysis into regions of reasonably simple mathematics, but often they may also represent the most typical or the most severe form of disturbance to which the system is subjected. The forcing function giving the best insight for transient response studies is the unit step function-a sudden change in the input, conventionally normalized to unit magnitude and considered to occur at an arbitrarily chosen time datum t = 0 (Fig. 4.1). The step change is the most severe disturbance possible for any given signal amplitude and is a type of change which frequently occurs in practice; mathematically it is simple to handle and the resulting system response is easily assessed for practical suitability. A forcing function which is less severe and more relevant for some physical systems, such as those possessing high inertia input characteristics, is the ramp function, or step change in velocity. This would be applicable where a step function is undesirable or is not physically possible, or where a change in input velocity is the normal forcing function. A parabolic input function or step change of acceleration could be used in situations where even the ramp input is too severe.

A further type of input function which is of considerable analytical importance is the unit impulse function, the limiting case of a pulse of unit area where the pulse duration tends to zero (Fig. 4.1). Although physical systems can seldom be satisfactorily tested with an impulsive forcing function, because of the very large change in input variable required to introduce sufficient energy into the system, the concept is one of great convenience. The impulse response can be obtained indirectly using random inputs and correlation techniques as will be described in Chapter 7. Other important forms of input function are sinusoidal signals, and statistical signals such as white noise, described in Chapters 6 and 7 respectively.



Fig. 4.1 Basic system forcing functions.

In this chapter is considered the calculation of the time response of systems of known dynamic characteristics for step, ramp, and impulse forcing functions. The forcing function and system response are given the symbols u(t) and c(t) respectively. From the results can be seen the effect of variation of the form of the system equations and the magnitude of the equation coefficients on the nature of the resulting transient response. The ideas can also be applied to the identification of practical systems by experimental testing. It is shown how a physical system of unknown dynamic characteristics can be subjected to a disturbance of such a form, and the resulting response curve used to estimate the system transfer function. The chapter concludes by describing the convolution integral and its use in evaluating transient system response for a more complex form of input function.

4.1 Response of first order system to step, ramp, or impulse function

Consider any system, such as the thermometer or the simple hydraulic servomechanism of Section 2.4, which is described by the first order differential equation

$$c(t) + \tau \frac{\mathrm{d}c(t)}{\mathrm{d}t} = u(t) \tag{4.1}$$

If the initial value c(0) of the output is zero then this can be transformed into the Laplace domain by replacing d/dt by s, c(t) by C(s) and u(t) by U(s) giving

$$C(s) + \tau s C(s) = U(s) \tag{4.2}$$

In transfer function form this would be written as

$$\frac{C(s)}{U(s)} = \frac{1}{1+\tau s} \tag{4.3}$$

(a) Unit step. What would be the response c(t) of this first order system to a step change of input? In other words, for the examples given, how does the mercury level rise when the thermometer is suddenly inserted in hot water, and how does the piston position vary as a function of time if the servomechanism input position is suddenly changed to a new desired value?

62 Transient Response of Systems

The Laplace transform of the output is, from Eq. 4.3,

$$C(s) = \frac{U(s)}{1 + \tau s}$$

and the Laplace transform of the input U(s) for a unit step is $\frac{1}{r}$.

$$C(s) = \frac{1}{s(1 + \tau s)}$$
 4.4

The time response c(t) can now be obtained by seeking the Laplace inverse of this expression from tables, or alternatively the expression can be split into partial fractions before inversion,

i.e.
$$C(s) = \frac{1}{s} - \frac{\tau}{1 + \tau s}$$

 $C(s) = \frac{1}{s} - \frac{1}{s + \frac{1}{\tau}}$
4.5

or

Each of the terms on the right appears in Table 2.1 and so the solution can be written directly as

$$c(t) = 1 - e^{-t/t}$$
 4.6

This is traditionally known as a simple lag or exponential lag and the constant τ is called the *time constant*. The form of the step response curve is shown in Fig. 4.2 and is such that

- (i) c(t) = 0.63 when $t = \tau$.
- (ii) a tangent to the curve at t = 0 meets the final value line at $t = \tau$.
- (iii) a tangent drawn at any point on the curve meets the final value line τ seconds later.
- (iv) c(t) is within 2% of unity for $t > 4\tau$.

The response can be considered to be in two parts, the *transient response* $-e^{-t/t}$ which decays to zero as $t \to \infty$, and the *steady state response* 1 which



Fig. 4.2 Step response of first order system

implies that in the steady state the output is equal to the input. These are respectively the complementary function and the particular integral obtained by classical methods of solution of the differential equation. All physical systems represented by a first order transfer function will have this solution trajectory.

For a step of magnitude k the response is scaled accordingly.

.e.
$$c(t) = k(1 - e^{-t/\tau})$$
 4.7

(b) Ramp. What would be the response c(t) of the first order system to an input which is changing at a fixed rate? How does the mercury level rise when the thermometer is placed in water whose temperature starts to rise uniformly, and how does the servomechanism respond to a velocity input?

The Laplace transform of a unit ramp is $\frac{1}{2}$

$$C(s) = \frac{U(s)}{1 + \tau s} = \frac{1}{s^2(1 + \tau s)}$$
4.8

As before, the time response c(t) can be obtained directly from Laplace transform tables, or by Laplace inversion after separation into partial fractions.

i.e.
$$C(s) = \frac{1}{s^2} - \frac{\tau}{s} + \frac{\tau^2}{1 + \tau s} = \frac{1}{s^2} - \tau \left(\frac{1}{s}\right) + \tau \left(\frac{1}{s + 1/\tau}\right)$$

and hence from Table 2.1

$$c(t) = t - \tau + \tau e^{-t/\tau}$$
 4.9

The transient response is $\tau e^{-t/\tau}$ which decays to less than 2% of τ in time 4τ , and the steady state response is $t - \tau$. It can thus be seen that there is a *steady* state error of magnitude τ for a unit ramp input (Fig. 4.3). For large values of time t the output lags the input by a constant value τ .



Fig. 4.3 Response of first order system to unit ramp function

For a ramp of magnitude k' i.e. for an input which increases steadily at k' units per second

$$c(t) = k'(t - \tau + \tau e^{-t/\tau})$$
 4.10

and the steady state error is $k'\tau$.

64 Transient Response of Systems

(c) Unit impulse. The Laplace transform of the input is

$$U(s) = 1$$

 $\therefore \quad C(s) = \frac{1}{1 + \tau s} = \frac{\frac{1}{\tau}}{\frac{1}{s + \frac{1}{\tau}}}$
4.11

and

 $c(t) = \frac{1}{\tau} e^{-t/\tau}$ 4.12

The response decays exponentially to a steady state value of zero after a sudden rise to $\frac{1}{2}$ at t = 0, the time of application of the impulse (Fig. 4.4).



Fig. 4.4 Impulse response of first order system

4.2 Response of second order system to step, ramp, or impulse function

Many mechanical systems are characterized by the presence of inertia, stiffness, and viscous damping and are thus described by the second order transfer function (Eq. 2.17) derived in Section 2.4. How does such a mass-spring-damper system react to a suddenly applied force, or to a steadily increasing force, or to an impulsive force? What will be the response to these inputs of an electrical or other system with this same transfer function? All have the same form of transfer function and thus will have the same form of response.

The output response is given by

$$C(s) = \frac{U(s)\omega_{n}^{2}}{s^{2} + 2\zeta\omega_{n}s + \omega_{n}^{2}}$$
4.13

(a) For a unit step $U(s) = \frac{1}{s}$

4.2 Response of second order system to step, ramp, or impulse function 65

$$C(s) = \frac{{\omega_n}^2}{s(s^2 + 2\zeta\omega_n s + {\omega_n}^2)}$$
 4.14

$$=\frac{1}{s} + \frac{A_1}{s - p_1} + \frac{A_2}{s - p_2}$$
 4.15

where A_1 and A_2 are constants, and p_1 and p_2 are the roots of the characteristic equation $s^2 + 2\zeta \omega_n s + \omega_n^2 = 0$

$$c(t) = 1 + A_1 e^{p_1 t} + A_2 e^{p_2 t}$$
4.16

where

$$A_{1} = -\frac{\zeta}{2} - \frac{\zeta}{2\sqrt{\zeta^{2} - 1}} \text{ and } A_{2} = -\frac{1}{2} + \frac{\zeta}{2\sqrt{\zeta^{2} - 1}}$$
$$p_{1} = -\zeta\omega_{n} + \omega_{n}\sqrt{\zeta^{2} - 1} \text{ and } p_{2} = -\zeta\omega_{n} - \omega_{n}\sqrt{\zeta^{2} - 1}$$

Three distinct types of response are possible (Fig. 4.5) according to whether the roots p_1 and p_2 are real and unequal, real and equal, or complex, this being determined by the value of the damping factor ζ .



Fig. 4.5 Form of step response for second order system

- (i) $\zeta > 1$: gives two negative real unequal roots, and an overdamped response where the coefficients A_1 and A_2 are also real.
- (ii) $\zeta < 1$: gives a pair of complex conjugate roots and coefficients A_1 and A_2 which also form a complex conjugate pair. The expression for the system response can be rearranged to give

$$c(t) = 1 - \frac{e^{-\zeta \omega_n t}}{\sqrt{(1-\zeta^2)}} \sin (\omega_n \sqrt{(1-\zeta^2)}t + \varphi), \text{ where } \varphi = \cos^{-1} \zeta$$
 4.17

This is an oscillatory or underdamped response.

- (iii) ζ = 1: gives two negative real equal roots, a critically damped system corresponding to the minimum value of damping factor for which there is no overshoot.
- (b) For a unit ramp input $U(s) = \frac{1}{s^2}$

$$C(s) = \frac{\omega_n^2}{s^2(s^2 + 2\zeta\omega_n s + \omega_n^2)}$$

$$= \frac{B_1}{s^2} + \frac{B_2}{s} + \frac{A_1}{s - p_1} + \frac{A_2}{s - p_2}$$
4.18

where B_1 , B_2 , A_1 and A_2 are constants which can be evaluated in terms of ζ and ω_n . Their values are given by

$$B_{1} = 1, B_{2} = -\frac{2\zeta}{\omega_{n}}, A_{1} \text{ and } A_{2} = \frac{\zeta}{\omega_{n}} \pm \frac{2\zeta^{2} - 1}{2\omega_{n}\sqrt{(\zeta^{2} - 1)}}$$

$$\therefore \quad c(t) = t - \frac{2\zeta}{\omega_{n}} + A_{1} e^{\rho_{1}t} + A_{2} e^{\rho_{2}t} \qquad 4.19$$

There is a steady state error of $-\frac{2\zeta}{\omega_n}$. The form of the transient part of the

response described by the third and fourth terms is discussed below.

(c) For a unit impulse input U(s) = 1

$$C(s) = \frac{\omega_{n}^{2}}{s^{2} + 2\zeta\omega_{n}s + \omega_{n}^{2}}$$
 4.20

$$c(t) = A_1 e^{p_1 t} + A_2 e^{p_2 t}$$

$$4.21$$

and

$$A_1 = -A_2 = \frac{\omega_n}{2\sqrt{(\zeta^2 - 1)}}$$

Whatever the type of input, whether a step ramp or impulse, it can be seen that the transient part of the solution has the form

$$A_1 e^{p_1 t} + A_2 e^{p_2 t}$$

where p_1 and $p_2 = -\zeta \omega_n \pm \omega_n \sqrt{(\zeta^2 - 1)} = -\zeta \omega_n \pm j \omega_n \sqrt{(1 - \zeta^2)}$

are the roots of the characteristic equation. It is helpful to see how the values of these roots affect the transient portion of the response, in other words, how the position of the roots in the complex *s*-plane influences the time response.

Consider a second order system with fixed value of ω_n but with varying ζ . As ζ varies from zero to infinity the roots will trace the locus shown in Fig. 4.6.

(i) For
$$\zeta = 0$$

$$p_1$$
 and $p_2 = \pm j\omega_n$

The roots are wholly imaginary, and the transient part of the solution is

$$A_1 e^{j\omega_n t} + A_2 e^{-j\omega_n t}$$

For a unit step input $A_1 = A_2 = -\frac{1}{2}$ and $c(t) = 1 - \cos \omega_n t$ For a unit ramp input $A_1 = -A_2 = -j/\omega_n$ and $c(t) = t + (2/\omega_n) \sin \omega_n t$ For a unit impulse input $A_1 = -A_2 = -j\omega_n/2$ and $c(t) = \omega_n \sin \omega_n t$ In each case the transient part of the solution is a sinusoidal oscillation of constant amplitude.





(ii) For $0 < \zeta < 1$

 p_1 and $p_2 = -\zeta \omega_n \pm j\omega_n \sqrt{(1-\zeta^2)}$ and the locus of the roots is a semicircle of radius ω_n . If the angle subtended with the real axis at the origin is φ as shown in Fig. 4.6 then $\zeta \omega_n = \omega_n \cos \varphi$

$$\zeta = \cos \varphi \qquad 4.22$$

:
$$\varphi = \cos^{-1}\zeta$$
 or $\varphi = \tan^{-1}\frac{\sqrt{(1-\zeta^2)}}{\zeta}$ 4.23

The value $\omega = \omega_n \sqrt{(1 - \zeta^2)}$ is often referred to as the *conditional frequency* or *damped natural frequency*. It is the frequency associated with the period of successive oscillations of the damped sinusoid. As $\zeta \to 0$ then $\omega \to \omega_n$.

(iii) For $\zeta \ge 1$ the two roots lie on the negative real axis. When $\zeta = 1$ the roots are real and equal, with value $-\omega_n$, and as ζ increases one root moves along the real axis towards the origin while the other moves towards $-\infty$. The effect of the former root on the time response becomes dominant, while the effect of the distant real root decreases, and when more than about six times as far from the origin as the dominant root its influence is negligible. Hence as the value of ζ becomes very large the response becomes very similar to that of a first order system.

The effect on the step and impulse response of a second order unity gain system with varying damping factor is shown in Fig. 4.7 for a few selected values of ζ . It can be clearly seen how as ζ decreases the response becomes more oscillatory and the damped natural frequency increases. The magnitude of the first overshoot is frequently of significance and can be found either from standard curves such as these or from Eq. 4.17 or Eq. 4.21 by finding the value of t for which $\dot{c}(t) = 0$ and then inserting this in the expression for c(t).





4.3 Transient response of third and higher order systems

In the general case the transfer function can be written as

$$\frac{C(s)}{U(s)} = \frac{P(s)}{Q(s)} \tag{4.24}$$

where P(s) and Q(s) are polynomials in s. If Q(s) is of order N then the characteristic equation defined as Q(s) = 0 will have N roots $p_1, p_2, p_3, \dots, p_N$, and Q(s) can be factorized to give

$$\frac{C(s)}{U(s)} = \frac{P(s)}{(s-p_1)(s-p_2)(s-p_3)\dots(s-p_N)}$$
 4.25

Dividing this expression into partial fractions, and then taking the inverse Laplace transform gives for a unit step input

$$c(t) = 1 + A_1 e^{p_1 t} + A_2 e^{p_2 t} + A_3 e^{p_3 t} + \dots + A_N e^{p_N t}$$

$$c(t) = 1 + \sum_{n=1}^{N} A_n e^{p_n t}$$

$$4.26$$

or

where $A_1, A_2, \dots A_N$ are constants.

The transient portion of the response is thus composed of a summation of terms of the form $A_n \exp p_n t$ where the values $p_1, p_2, \ldots, p_n, \ldots, p_N$ are the roots of the characteristic equation, which are at the same time the poles of the overall transfer function. The contribution which each term makes towards the overall response is dependent on the magnitude and sign of A_n , and on the position of the pole p_n in the complex s-plane, each pole generally having a real part σ and an imaginary part j ω . The effect of pole position on the time response is shown schematically in Fig. 4.8, the magnitude A_n being assumed the same in all cases. It should be noted that complex roots always occur as conjugate pairs, and the responses shown arise from the pairs of roots.

It can be seen that the presence of any pole with a positive real part, which means any pole located in the right half of the complex s-plane, gives rise to a contribution to the time response which is increasing without limit. Thus if any pole lies in the right half plane the system will be unstable, where instability implies the fact that the output is unbounded for a bounded input. Of the poles in the left half plane, those farthest from the imaginary axis will have contributions to the transient response which decay most rapidly, and hence the system response will be influenced most by the poles closest to the imaginary axis, called the *dominant poles*. A convenient rule of thumb approximation for design purposes is to assume that the effect of any roots more than 5 or 6 times as far from the imaginary axis as the dominant roots can be neglected.

In addition to the transient response, the value of the steady state gain is generally of interest. This can be evaluated quickly and conveniently making use of the final value theorem of Laplace transform analysis (Section 2.2).

If

$$\frac{C(s)}{U(s)} = G(s)$$

then

 $\lim_{t \to \infty} c(t) = \lim_{s \to 0} sC(s)$ by the final value theorem

 $= \lim_{s \to 0} G(s) \text{ for a steady unit input, } U(s) = \frac{1}{s}$

e.g. if
$$\frac{C(s)}{U(s)} = \frac{50(1+5s)}{(s^2+3s+16)(1+s)}$$

then

 $[c(t)]_{t \to \infty} = \lim_{s \to 0} \frac{50(1+5s)}{(s^2+3s+16)(1+s)} = \frac{50}{16}$ for unit input

4.4 Performance characteristics (time domain)

What is a 'good' type of transient response? How should the transient response of a practical system, or the type of response desired for a system be described? The algebraic equation is not very helpful since the form of the response is not readily apparent. A plot of the response is not satisfactory since a numerical description is required for analysis.





A first order system can be completely described by specifying the value of the time constant. A second order system can be clearly described by specifying the two time constant values if overdamped, or the values of ζ and ω_n if underdamped. For a higher order system, values of ζ and ω_n cannot be specified since they do not exist—though values of these can be given for the dominant roots.

In general the following parameters, shown in Fig. 4.9, give an adequate description and are used to describe the step response of a system:

- (i) maximum overshoot—this is usually expressed as a percentage of the step size,
- (ii) number of oscillations,
- (iii) rise time—this is usually defined as the time taken to rise from 5% to 95% of the step size, or over some similar range; defining rise time thus avoids the practical difficulty of having to determine the exact start of the transient, and the finish, if overdamped,
- (iv) settling time—the time taken until the output falls within and remains within $\pm 5\%$, say, of the steady state value,
- (v) steady state error.



Fig. 4.9 Parameters describing unit step response

These parameters are interrelated, and requirements tend to conflict. The maximum overshoot can generally only be decreased at the expense of an increase in rise time, steady state error can generally only be reduced at the expense of making the transient more oscillatory.

It can be useful to be able to describe a complete transient response by a single numerical value. Such a requirement commonly exists in adaptive control where, to compensate for general system changes, a specific system parameter is caused to be adjusted to maintain sensibly constant dynamic performance. Several types of function have been used for this purpose, the

72 Transient Response of Systems

chosen value being called the *performance index*. This generally involves integration of the transient response, some frequently used functions being

$$\int_{0}^{\infty} |e(t)| dt \quad \text{integral of absolute error (IAE)}$$

$$\int_{0}^{\infty} e^{2}(t) dt \quad \text{integral of error squared (IES)}$$

$$\int_{0}^{\infty} t |e(t)| dt \quad \text{integral of time } \times \text{ absolute error (ITAE)}$$

In general, the value of these functions will be large if the transient is either very sluggish or very oscillatory and, as a system parameter is changed to give a resulting response which varies from one to the other, there will be a value of parameter for which the performance index is a minimum. This would be the 'best' transient response for the chosen performance index.

The IES function accentuates large errors. The ITAE function, by introducing time weighting of the error signal, has the effect of placing small emphasis on the largely unavoidable large initial errors but great emphasis on long duration transients. In addition the ITAE criterion generally gives a more sharply defined minimum and it is thus the most selective. For a second order system with fixed ω_n and varying ζ the optimum is achieved for $\zeta = 0.7$ (ITAE), $\zeta = 0.5$ (IES), $\zeta = 0.7$ (IAE). For many purposes a damping factor of 0.7 is taken to be the ideal for a second order response.

4.5 Step response testing of practical systems

Whenever an existing design requires modification or forms a basis for a future design an appropriate mathematical model must be obtained for the practical system. The system may be one whose internal structure is not understood sufficiently to allow analytical formulation of equations, a so-called *black-box* system. In this case, the dynamic characteristics would have to be obtained by practical testing of the system itself, and the information used to determine a representative mathematical model. On the other hand, by making certain assumptions it may be possible to write down governing equations for the system, in which case the equipment could be tested practically to confirm the form of the theoretically obtained dynamic relationship, and to confirm or determine the parameter values for this model.

Step response testing will usefully and relatively easily give a first idea of the general form of the transfer function and its parameters, and can also give some indication of how linear the system is. A step change of input can usually be applied fairly easily, and inspection of the resulting response for various magnitudes of step size and for step changes in a positive and a negative direction relative to the datum condition will indicate whether a linear model can realistically be assumed. If not, some idea of the type of non-linearity may

be forthcoming. If a large transportation lag or dead time is present then this is generally detectable from the transient response curve.

The dynamic information may be suitably presented in the form of the actual transient response curves—a non-parametric model—but usually a parametric model is required and it becomes necessary to fit a transfer function or state space model to the response curve. To do this it is necessary to select some error criterion to quantify the goodness of fit, and then to adjust the model and its parameter values to minimize the chosen error index.

(a) Response apparently of first order. If the step response apparently rises exponentially to the new steady state it is likely that the system is predominantly of first order. The dominant time constant can be found by any of the three measurements described in Section 4.1(a) and Fig. 4.2. Each should give the same value.

Superimposing the curve $1 - e^{-ir}$ will show qualitatively whether this is a reasonable model, or whether it is necessary to use a higher order model.

(b) Response apparently lightly damped second order. A mathematical relationship between ζ and the relative magnitudes of overshoots and undershoots can be found for the second order system with damping low enough to give more than one oscillation. This can be used to estimate ζ from an oscillatory step response.

From Eq. 4.17 and Eq. 4.23

$$c(t) = 1 - \frac{\exp(-\zeta\omega_{n}t)}{\sqrt{(1-\zeta^{2})}} \sin\left(\omega_{n}\sqrt{(1-\zeta^{2})t} + \tan^{-1}\frac{\sqrt{(1-\zeta^{2})}}{\zeta}\right) \quad 4.27$$

 $\therefore \quad \frac{\mathrm{d}c(t)}{\mathrm{d}t} = -\frac{1}{\sqrt{(1-\zeta^2)}} \left\{ \exp\left(-\zeta\omega_n t\right) \cos\left[\omega_n\sqrt{(1-\zeta^2)t} + \varphi\right] \omega_n\sqrt{(1-\zeta^2)} \right\}$

$$+\exp(-\zeta\omega_{n}t)(-\zeta\omega_{n})\sin[\omega_{n}\sqrt{(1-\zeta^{2})t+\varphi}]$$

$$= \frac{\exp\left(-\zeta \omega_n t\right)}{\left(1 - \zeta^2\right)} \left\{ \zeta \omega_n \sin\left[\omega_n \sqrt{(1 - \zeta^2)t} + \varphi\right] \right\}$$

$$-\omega_n \sqrt{(1-\zeta^2)} \cos \left[\omega_n \sqrt{(1-\zeta^2)t} + \varphi\right]$$

= 0 when $\tan \left[\omega_n \sqrt{(1-\zeta^2)t} + \varphi\right] = \frac{\sqrt{(1-\zeta^2)}}{\zeta}$

But $\varphi = \tan^{-1} \frac{\sqrt{(1-\zeta^2)}}{\zeta}$

Hence peaks and troughs occur when $\omega_n \sqrt{(1-\zeta^2)t} = n\pi$

i.e. when
$$t = \frac{n\pi}{\omega_n \sqrt{(1-\zeta^2)}} = t_n$$
 where $n = 1, 2, 3...$

74 Transient Response of Systems

n = 1 gives the 1st overshoot: $c(t_1) =$

n =

n =

$$1 - \frac{\exp\left(-\frac{\zeta\pi}{\sqrt{(1-\zeta^2)}}\right)}{\sqrt{(1-\zeta^2)}} \sin\left(\pi + \tan^{-1}\frac{\sqrt{(1-\zeta^2)}}{\zeta}\right)$$
$$= 1 + \exp\left(-\frac{\zeta\pi}{\sqrt{(1-\zeta^2)}}\right)$$
2 gives the 1st undershoot: $c(t_2) = 1 - \exp\left(-\frac{2\zeta\pi}{\sqrt{(1-\zeta^2)}}\right)$
3 gives the 2nd overshoot: $c(t_3) = 1 + \exp\left(-\frac{3\zeta\pi}{\sqrt{(1-\zeta^2)}}\right)$

Hence if x_1 and x_3 are the magnitudes of the 1st and 2nd overshoots (Fig. 4.10) the ratio of these overshoot values is:

$$\frac{x_1}{x_3} = \frac{\exp\left(-\frac{\zeta\pi}{\sqrt{(1-\zeta^2)}}\right)}{\exp\left(-\frac{3\zeta\pi}{\sqrt{(1-\zeta^2)}}\right)}$$

$$\therefore \quad \log_e\left(\frac{x_1}{x_3}\right) = \frac{2\pi\zeta}{\sqrt{(1-\zeta^2)}}$$
 4.28



Fig. 4.10 Step response of oscillatory system

Practically it is more often possible to obtain ζ using the first overshoot and first undershoot

$$\log_e\left(\frac{x_1}{x_2}\right) = \frac{\pi\zeta}{\sqrt{(1-\zeta^2)}}$$

Also, the interval between successive overshoots is $\frac{2\pi}{\omega}$ where $\omega = \omega_n \sqrt{(1-\zeta^2)}$

Hence for an experimentally obtained transient response which appears to be of a lightly damped second order system it is possible to quickly obtain

4.5 Step response testing of practical systems 75

estimates of ζ and ω_n . Again, comparison of the actual response with the second order response for the calculated values of ζ and ω_n , and taking into account the use to which the model will be put, will show whether or not it is necessary to attempt to fit a higher order model to the response.

(c) Overdamped second order response. For a second order system with damping factor greater than unity

$$c(t) = 1 + A_1 e^{p_1 t} + A_2 e^{p_2 t}$$

where p_1 and p_2 are negative real numbers.

There are two separate time constants. If these are an order of magnitude different, then the effect of the smaller time constant is only evident in the first part of the response, and the approximate value of the dominant time constant can be found from the slope of the later portion of the response curve (Fig. 4.11). For more accurate values a curve fitting procedure is needed.



Fig. 4.11 Step response of overdamped second order system.

(d) Higher order overdamped system. In the general case of an overdamped system the response is of the form

$$c(t) = 1 - \sum_{n=1}^{N} A_n e^{-t/t_n}$$

where $\tau_1 > \tau_2 > \tau_3 > ... > \tau_N$ are the time constants of the system.

.
$$1 - c(t) = A_1 e^{-t/\tau_1} + A_2 e^{-t/\tau_2} + A_3 e^{-t/\tau_3} + \dots$$

 $\simeq A_1 e^{-t/\tau_1}$ for large t

$$\log_{\mathbf{e}}[1-c(t)] = \log_{\mathbf{e}}A_1 - \frac{t}{\tau_1}$$
 for large t

By plotting $\log_{e}[1 - c(t)]$ against t (Fig. 4.12), and looking for a linear relationship at large values of t, the values of A_1 and τ_1 can be estimated. The coefficient A_1 is the value of the intercept of the straight line with the t = 0 axis. The dominant time constant τ_1 is obtained from the slope of the line.





i.e.
$$\tau_1 = \frac{t_2 - t_1}{\log_e [1 - c(t_2)] - \log_e [1 - c(t_1)]}$$

 $1 - c(t) - A_1 e^{-t/t_1} = A_2 e^{-t/t_2} + A_3 e^{-t/t_3} + \dots$
 $\simeq A_2 e^{-t/t_2}$ for large t

Now

The procedure is then to subtract the $A_1 e^{-t/\tau_2}$ line from the curve 1 - c(t) to obtain a second curve, attempt to draw a straight line to fit this for large values of t and hence find values for A_2 and τ_2 . It may then be possible to repeat this once more.

This method gives values of the time constants fairly quickly providing they differ in magnitude by, say, more than a factor of 2 or 3. The accuracies of the results for a second order system would then be about 10% and 25% for the primary and secondary time constants respectively. If the values of the time constants are closer together than this, the method fails, as it becomes impossible to detect linear portions of the curves. A severe constraint is always the value of the noise-signal ratio pertaining when making practical measurements.

4.6 Comparison of transient forcing functions

The main advantages of the step function as a test signal for a practical system are that step changes are physically easy to apply, and that the resulting response curves very readily give an idea of the general dynamic characteristics of the system. Simple observation shows whether a system is oscillatory or not, and hence by simple measurement can be determined the order of magnitude of the dominant time constant, or of the damping factor and undamped natural frequency. One can look for the presence of a transportation lag and, with different magnitudes of steps, look for signs of non-linearity. There are, however, several important disadvantages:

- (i) the size of step required to give a transient response of the system which is detectable in the presence of the inherent noise may result in an unacceptably large disturbance to the system,
- (ii) the steady state point of the system changes,
- (iii) very small variations in the shape of the time response curve have a large effect on the higher order terms of the transfer function.

An impulse forcing function does avoid the second problem above in that the steady state point before and after applying the disturbance is the same. The disturbance to the system is however even larger than for a step change, and is almost always unacceptable in the presence of noise. It is difficult to apply a true impulse with enough energy to give a transient which is detectable in the presence of noise, without saturating part of the system. A representation of the impulse response curve can, however, be obtained using correlation techniques, as will be shown in Chapter 7.

With some practical systems a step change which implies a substantially instantaneous change of a variable is physically impossible. In such cases it may be more relevant to consider a ramp input function, or a step change in velocity. An example might be an item of process plant where a motorized valve can only close at some finite rate.

4.7 The convolution integral

It has been shown above how, knowing a system transfer function, the response of the system to inputs of certain simple analytical forms can be calculated. Sometimes the response must be evaluated when the input is more complex and perhaps not even expressible in a deterministic manner. A procedure based on the convolution integral enables this to be done. In such situations the input can be considered as being made up of a series of pulses of varying amplitude, and the principle of superposition can be used to obtain an expression for the system output.

For the linear system shown in Fig. 4.13, the output response is given by Y(s) = X(s)G(s)

$$\begin{array}{c|c} X(s) & G(s) & Y(s) \\ \hline & g(t) \text{ or } w(t) & y(t) \end{array}$$

Fig. 4.13 Single-input-single-output linear system

If the input is a unit impulse $\delta(t)$

then $X(s) = \mathcal{L}[\delta(t)] = 1$ and Y(s) = G(s)

r(s) = c

 \therefore $y(t) = \mathcal{L}^{-1}[G(s)] = g(t)$ or w(t), the unit impulse response.

78 Transient Response of Systems

If the input is now an arbitrary function x(t), consider it as being made up of an infinite number of impulses of width $\delta \lambda$ and height x(t), Fig. 4.14.





The contribution $\Delta y(t)$ to the output at time t due to an impulse applied λ seconds earlier (i.e. at time $t - \lambda$) will be the value of the impulse response at time λ times the magnitude of the impulse which was applied at time $t - \lambda$.

$$\Delta y(t) = w(\lambda)x(t - \lambda)\delta\lambda$$

The impulse response function is frequently called the *weighting function* because it specifies by how much the input applied λ seconds in the past has decayed.

The total system output is the sum of terms due to all impulses which have occurred in the past prior to time t, the summation being performed in the limit by the integral

$$y(t) = \int_{-\infty}^{t} w(\lambda) x(t - \lambda) \, \mathrm{d}\lambda$$

Now w(t) = 0 for t < 0 for any real system

$$y(t) = \int_{0}^{t} w(\lambda) x(t - \lambda) d\lambda$$

This is called the *convolution integral* or *superposition integral* and is sometimes written as

$$y(t) = w(t) * x(t)$$

where the symbol * implies convolution. This has a similar form to the equivalent Laplace domain relationship

$$Y(s) = G(s)X(s)$$

Physically the process of convolution can be illustrated by Fig. 4.15.

If the system weighting function is as shown in Fig. 4.15a, and the input function as in Fig. 4.15b, then $x(t - \lambda)$ will be as Fig. 4.15d and the product for this value of t will be as Fig. 4.15e; y(t) is the area under this curve.



Fig. 4.15 Illustration of process of convolution

It can be seen that the convolution integral can be evaluated by time shifting $w(\lambda)$ rather than $x(\lambda)$ giving the alternative expression

$$y(t) = \int_{0}^{t} x(\lambda) w(t - \lambda) \, \mathrm{d}\lambda$$

Also, provided the signal is bounded, i.e. x(t) = 0 for t < 0, the upper limit of integration can be extended to infinity since $x(t - \lambda) = 0$ for $\lambda > t$, or $w(t - \lambda) = 0$ for $\lambda > t$. Such extension of the limits from $-\infty$ to $+\infty$ can simplify certain manipulations.

5

State Space Representation and Analysis

Transfer function representation, described in Chapter 2, is one of two established methods employed in the modelling and analysis of linear control system elements. The approach is based implicitly on the use of Laplace transforms, and associated design procedures for use with this method are described in Chapter 11. These design methods enable system performance to be predicted without actually solving to find the roots of the characteristic equation. It is against a design background requiring the use of trial and observation procedures, and with impetus given by the rapid development of the high speed digital computer, that the second approach to control system modelling has been developed. This method does not use a transfer function description, but replaces it by a state space representation, as introduced and described briefly in Section 2.6. The state space model representation makes possible the use of mathematical techniques that lead to a more systematic design process than is directly possible with a transfer function representation. Extensive use of the digital computer in design makes it necessary to be able to replace differential equations by difference equations, and to this end the first order state equations are easier to handle than the high-order equations associated with the transfer function.

The state space procedure basically involves transforming a single nth order differential equation into a set of n first order simultaneous differential equations, employing matrix notation as a form of technical shorthand. This requires the introduction of additional variables, the state variables; the number of state variables required to define a system completely is equal to the order of the system. These variables are not unique and several different methods of choosing them are presented to illustrate this. The actual state variables used may be chosen to suit the particular problem and an appropriate choice of variable at an early stage may well facilitate later solution.

For example, the second order differential equation which describes mathematically an electrical circuit consisting of a resistor, a capacitor, and an inductor in series, requires a two-state-variable description. The pair of variables could be the current flowing to the capacitor and the voltage across the capacitor, or alternatively the current from and voltage across the inductor. However, any other two variables such as the voltage across the resistor and the rate of change with respect to time of this voltage would be equally valid as state variables for this simple system.

The use of matrices can simplify the computational manipulations since two matrices A and B, defined later in this chapter, specify the dynamic character-

istics of a particular linear system and so the dynamic properties of the system can be studied by investigating the properties of these matrices. An additional advantage of the state space method is that the restriction to single-input single-output models, which is implicit throughout the preceding chapters, disappears and the state equation can be used directly to describe a multivariable dynamic system with many inputs and outputs.

Section 5.1 describes three methods of obtaining the state equations and writes them both as a set of first order differential equations and also more compactly in matrix equation form. The section which follows extends the thinking to multi-input multi-output systems by defining the generalized state equations, and deriving them for an illustrative 2-input-3-output system. Section 5.3 investigates the s-domain representation of the state equations and derives a matrix transfer function, a set of transfer functions, each of which relates one output to one input. Section 5.4 shows how in principle the state equations can be solved analytically to find the time response, although in practice only simple situations are amenable to such solution. The section following gives an insight into the principles of handling a discrete time model in computer solution of the state equations.

A useful background for this chapter for those not familiar with simple matrix methods is given in Appendix C.

5.1 State variable diagrams

A convenient method for deriving state equations is based on analogue computing techniques. A circuit diagram is constructed for the system equations in the manner illustrated in Chapter 3, and the integrator output signals are defined as the state variables. The integrator inputs then become the derivatives of the state variables and the state equations can be written down immediately by inspection of the circuit diagram.

The basic elements most frequently used in such diagrams are similar to those described in Section 3.2 and consist of the ideal integrator, ideal amplifier and ideal adder as shown in Fig. 5.1. These differ from the practical



Fig. 5.1 Symbols used in state variable diagrams

analogue computer symbols because the integrator and adder do not incorporate a sign change, and the amplifier achieves multiplication by any constant within the range $\pm \infty$.

To illustrate the construction of state variable diagrams for the derivation of state equations, and to give an insight into the structure of a state variable expression, consider the transfer function relationship

$$\frac{Y(s)}{U(s)} = \frac{s^2 + 3s + 1}{s(s^2 + 6s + 8)}$$
5.1

By expansion of Eq. 5.1 into partial fraction form the transfer function can be written in an alternative way as

$$\frac{Y(s)}{U(s)} = \frac{1}{8s} + \frac{5}{8(s+4)} + \frac{2}{8(s+2)}$$
5.2

or it can be written in factored form as

$$\frac{Y(s)}{U(s)} = \frac{(s+2.62)(s+0.38)}{s(s+4)(s+2)}$$
5.3

and by programming each of these equations separately three different state variable diagrams can be produced. From each diagram a different set of state variables results, which clearly indicates that no single set is unique to a particular dynamic system.

To avoid confusion later in this chapter, the symbol Y(s) or y(t) is used to define the output response, instead of the symbol C(s) or c(t) used elsewhere in the text.

(a) State variable diagram for Eq. 5.1. Since a transfer function can be treated as an algebraic expression, the numerator and denominator of Eq. 5.1 can each be divided by the highest power of s to replace all differentiating terms by integrating terms. Eq. 5.1 can thus be written as

$$\frac{Y(s)}{U(s)} = \frac{s^{-1} + 3s^{-2} + s^{-3}}{1 + 6s^{-1} + 8s^{-2}}$$
$$Y(s) = (s^{-1} + 3s^{-2} + s^{-3})V(s)$$
5.4

or where

$$V(s) = \frac{U(s)}{1 + 6s^{-1} + 8s^{-2}}$$

which transposes to

$$V(s) = U(s) - 6s^{-1}V(s) - 8s^{-2}V(s)$$
5.5

The variable V(s) must be introduced to avoid the need for differentiation, as in Section 3.6. The state variable diagram, shown in Fig. 5.2, follows from Eq. 5.4 and Eq. 5.5 by integrating V(s) three times and combining signals as required.

By designating the integrator outputs to be the system state variables, it can readily be seen by inspection of Fig. 5.2 that the state equations are:

$$\dot{x}_1(t) = x_2(t)
\dot{x}_2(t) = x_3(t)
\dot{x}_3(t) = u(t) - 6x_3(t) - 8x_2(t)
5.6$$



When written using matrix notation Eqns. 5.6 become

$$\begin{cases} \dot{x}_1(t) \\ \dot{x}_2(t) \\ \dot{x}_3(t) \end{cases} = \begin{bmatrix} 0 & 1 & 0 \\ 0 & 0 & 1 \\ 0 & -8 & -6 \end{bmatrix} \begin{cases} x_1(t) \\ x_2(t) \\ x_3(t) \end{cases} + \begin{bmatrix} 0 \\ 0 \\ 1 \end{bmatrix} u(t)$$
 5.7

or more concisely

$$\{\dot{x}(t)\} = A\{x(t)\} + Bu(t)$$
 5.8

where

$$\{\mathbf{x}(t)\} = \begin{cases} x_1(t) \\ x_2(t) \\ x_3(t) \end{cases}, \quad \mathbf{A} = \begin{bmatrix} 0 & 1 & 0 \\ 0 & 0 & 1 \\ 0 & -8 & -6 \end{bmatrix}, \quad \mathbf{B} = \begin{bmatrix} 0 \\ 0 \\ 1 \end{bmatrix}$$

The important points about this notation are:

(i) $\{x(t)\}\$ is the state vector and has *n* components $x_1(t), x_2(t), \ldots, x_n(t)$, where *n* is the order of the dynamic system. The brackets { } are used to represent a column matrix indicating a system vector.

(ii) A and B are matrices of order $n \times n$ and $n \times 1$ respectively. The brackets [] are used to indicate a matrix representing the coefficient parameters of a system.

Also, from inspection of Fig. 5.2 the system output y(t) can be seen to be

$$y(t) = x_1(t) + 3x_2(t) + x_3(t)$$

$$y(t) = C\{x(t)\} = [1 \ 3 \ 1]\{x(t)\}$$
5.9

(b) Alternative state variable diagram, using Eq. 5.2. The second and third terms

on the right-hand side of Eq. 5.2 which have the form $\frac{b}{s+a}$ can be expressed

as $Y(s) = \frac{b}{s} \left(U(s) - \frac{a}{b} Y(s) \right)$, enabling the ideas of Chapter 3 to be used. The

circuit diagram for such an equation is shown in Fig. 5.3a and the state variable diagram Fig. 5.3b for Eq. 5.2 follows immediately.

The integrator outputs are, as before, chosen to be the state variables; the set of simultaneous first-order dynamic equations can be written down by inspection of Fig. 5.3b.





(b)
$$\frac{Y(s)}{U(s)} = \frac{1}{8s} + \frac{5}{8(s+4)} + \frac{2}{8(s+2)}$$

 $\dot{x}_1(t) = u(t)$
 $\dot{x}_2(t) = u(t) - 4x_2(t)$
 $\dot{x}_3(t) = u(t) - 2x_3(t)$
5.10

 $\{\dot{\mathbf{x}}(t)\} = \begin{bmatrix} 0 & 0 & 0\\ 0 & -4 & 0\\ 0 & 0 & -2 \end{bmatrix} \{\mathbf{x}(t)\} + \begin{bmatrix} 1\\ 1\\ 1 \end{bmatrix} u(t)$ 5.11

and the output is given by

$$y(t) = \frac{1}{8}x_1(t) + \frac{5}{8}x_2(t) + \frac{2}{8}x_3(t)$$

$$y(t) = \left[\frac{1}{8}, \frac{5}{8}, \frac{2}{8}\right] \{x(t)\}$$
5.12

It can be clearly seen that these are not the same state equations or variables as those obtained by method (a).

(c) Second alternative state variable diagram, using Eq. 5.3. Consider first a general transfer function

$$\frac{Y(s)}{U(s)} = \frac{s+b}{s+a}$$

OF

As shown in case (a), this can be written

$$\frac{Y(s)}{U(s)} = \frac{1+bs^{-1}}{1+as^{-1}}$$

and introducing a variable V(s) to avoid the operation of differentiation yields the pair of equations

$$Y(s) = (1 + bs^{-1})V(s) V(s) = U(s) - as^{-1}V(s)$$

and

The diagram for these equations is shown in Fig. 5.4a; using this the state diagram for Eq. 5.3 can now be drawn, Fig. 5.4b.



Fig. 5.4 State variable diagram for (a) $\frac{Y(s)}{U(s)} = \frac{s+b}{s+a}$ (b) $\frac{Y(s)}{U(s)} = \frac{(s+2.62)}{s(s+4)} \cdot \frac{(s+0.38)}{(s+2)}$

The state equations follow from Fig. 5.4b after some simple algebraic manipulations:

$$\dot{x}_{1}(t) = -2x_{1}(t) - 1.38 \ x_{2}(t) + x_{3}(t)$$

$$\dot{x}_{2}(t) = -4x_{2}(t) + x_{3}(t)$$

$$\dot{x}_{3}(t) = u(t)$$

$$\{\dot{x}(t)\} = \begin{bmatrix} -2 & -1.38 & 1\\ 0 & -4 & 1\\ 0 & 0 & 0 \end{bmatrix} \{x(t)\} + \begin{bmatrix} 0\\ 0\\ 1 \end{bmatrix} u(t)$$

5.14

The output is

or

$$y(t) = -1.62 x_1(t) - 1.38 x_2(t) + x_3(t)$$

$$y(t) = [-1.62 - 1.38 1] \{x(t)\}$$
5.15

The above example demonstrates very clearly that it is possible to characterize the same system by different sets of state variable equations. So far as the authors are aware there is no rigorous method for determining which set is likely to prove the most useful analytically and the choice is largely dictated by the form in which the dynamic equations appear.

5.2 Generalized state equations

Up to this point only *n*th order systems having a single input and a single output have been considered. The state equations may be generalized to include the multi-input-multi-output case by writing them in the form

$$\{\dot{\mathbf{x}}(t)\} = A\{\mathbf{x}(t)\} + B\{\mathbf{u}(t)\}$$
 5.16

$$\{y(t)\} = C\{x(t)\} + D\{u(t)\}$$
5.17

The additional term $D\{u(t)\}$ introduced in Eq. 5.17 allows for possible interaction between system inputs and outputs. The terms in Eq. 5.16 and Eq. 5.17 are referred to in the literature by the following names:

- $\{x(t)\}$ is the state vector
- $\{y(t)\}$ is the response or output vector
- $\{u(t)\}$ is the control or input vector
- A is the coefficient matrix of the process
- **B** is the driving matrix
- C is the output matrix
- D is the transmission matrix.

If a system has p inputs and q outputs, then $\{u(t)\}$ is an input column vector containing the p elements $u_1(t)$, $u_2(t)$, \dots $u_j(t)$, \dots $u_p(t)$, and $\{y(t)\}$ is an output column vector containing the q elements $y_1(t)$, $y_2(t)$, \dots $y_i(t)$, \dots , $y_q(t)$. Consequently the **B** matrix must be of order $(n \times p)$, the **C** matrix of order $(q \times n)$ and the **D** matrix of order $(q \times p)$ to satisfy the rules of matrix multiplication (Appendix C). No standard notation has yet been accepted for Eq. 5.16 and Eq. 5.17 and other symbols will be found in the published literature, $\{c(t)\}$ in place of $\{y(t)\}$ and $\{r(t)\}$ in place of $\{u(t)\}$ being the most common.

Example 5.1. As an illustration of the form of the state equations for a system with more than one input and output, consider the multi-tank system shown in Fig. 5.5. The input liquid flow rates are labelled $q_1(t)$ and $q_2(t)$ and the output variables, or system response, will be assumed to be the liquid heads in the 3 tanks, designated $h_1(t)$, $h_2(t)$ and $h_3(t)$.

For simplicity of modelling, the interconnecting and outlet pipes are assumed to have a linear head-flow relationship, so that the flow through each constriction is related to the liquid head difference across the constriction by the relationship

$$q(t) = \frac{\Delta h(t)}{R}$$

where $\Delta h(t)$ is the liquid head difference and the constant R is the linear flow resistance. The rate of change of fluid volume within a tank can be described d

by $a \frac{d}{dt} h(t)$ where a is the cross-sectional area of a tank.

For each tank a flow continuity equation can be written in which the rate of change of fluid volume is equated to the rate of inflow of fluid. By inspection



Fig. 5.5 Multi-tank flow system

of Fig. 5.5, and using the symbols shown, the flow continuity equations for tanks 1, 2, and 3 are respectively

$$a_1 \frac{d}{dt} h_1(t) = q_1(t) - \frac{1}{R_1} (h_1(t) - h_2(t))$$

$$a_2 \frac{d}{dt} h_2(t) = \frac{1}{R_1} (h_1(t) - h_2(t)) - \frac{1}{R_2} (h_2(t) - h_3(t)) - \frac{1}{R_3} h_2(t) \qquad 5.18$$

and

$$a_3 \frac{d}{dt} h_3(t) = q_2(t) + \frac{1}{R_2} (h_2(t) - h_3(t))$$

Rearranging equations 5.18 gives

$$\dot{h}_{1}(t) = \left(-\frac{1}{a_{1}R_{1}}\right)h_{1}(t) + \left(\frac{1}{a_{1}R_{1}}\right)h_{2}(t) + \left(\frac{1}{a_{1}}\right)q_{1}(t)$$

$$\dot{h}_{2}(t) = \left(\frac{1}{a_{2}R_{1}}\right)h_{1}(t) - \left(\frac{1}{a_{2}R_{1}} + \frac{1}{a_{2}R_{2}} + \frac{1}{a_{2}R_{3}}\right)h_{2}(t) + \left(\frac{1}{a_{2}R_{2}}\right)h_{3}(t) - 5.19$$

$$\dot{h}_{3}(t) = \left(\frac{1}{a_{3}R_{2}}\right)h_{2}(t) - \left(\frac{1}{a_{3}R_{2}}\right)h_{3}(t) + \left(\frac{1}{a_{3}}\right)q_{2}(t)$$

The three expressions in $\hat{h}(t)$ which constitute Eqns. 5.19 are equivalent to the matrix equation 5.16, if corresponding terms are equated; i.e.

$$\{\dot{\mathbf{x}}(t)\} = A\{\mathbf{x}(t)\} + B\{\mathbf{u}(t)\}$$

here $\{\mathbf{x}(t)\} = \begin{cases} h_1(t) \\ h_2(t) \\ h_3(t) \end{cases}$, $A = \begin{bmatrix} -\frac{1}{a_1R_1} & \frac{1}{a_1R_1} & 0 \\ \frac{1}{a_2R_1} & -\frac{1}{a_2}\left(\frac{1}{R_1} + \frac{1}{R_2} + \frac{1}{R_3}\right) & \frac{1}{a_2R_2} \\ 0 & \frac{1}{a_3R_2} & -\frac{1}{a_3R_2} \end{bmatrix}$
 $\{\mathbf{u}(t)\} = \begin{cases} q_1(t) \\ q_2(t) \end{cases}$, $B = \begin{bmatrix} \frac{1}{a_1} & 0 \\ 0 & 0 \\ 0 & \frac{1}{a_3} \end{bmatrix}$

The system output is given by the matrix equation Eq. 5.17

$$\{y(t)\} = C\{x(t)\} + D\{u(t)\}$$

The outputs are $h_1(t)$, $h_2(t)$ and $h_3(t)$ which in this example are also the state variables.

Hence
$$\{y(t)\} = \{h(t)\} = \{x(t)\}$$

 $\therefore D = 0$
and $C = \begin{bmatrix} 1 & 0 & 0 \\ 0 & 1 & 0 \\ 0 & 0 & 1 \end{bmatrix}$

and

W

Such a matrix with coefficients of 1 on the principal diagonal and zero elsewhere is called a *unit matrix* and is given the symbol I.

5.3 State relations in the s-domain

It may be desirable to develop a transfer function representation from a state model since, in engineering work, it is often necessary to confirm a mathematical model experimentally. A well established experimental modelling procedure is given in Chapter 6 and it will be seen that this is based on frequency response data, making direct comparison with a state model extremely difficult. In this section the relationship between the two model representations is derived.

Ordinary differential equations are transformed into algebraic equations by using Laplace transformation to change from the time domain to the sdomain. Since Laplace transformation is a scalar operation, it can be applied

to the generalized state equations (Eq. 5.16 and Eq. 5.17) to obtain the following s-domain equations:

$$s\{X(s)\} - \{x(0)\} = A\{X(s)\} + B\{U(s)\}$$
5.20
$$S\{X(s)\} - S\{X(s)\} - S\{X(s)\} + B\{U(s)\}$$
5.21

$$\{Y(s)\} = C\{X(s)\} + D\{U(s)\}$$
 5.21

where $\{x(0)\} = \lim_{t \to 0} \{x(t)\}$ is the initial condition vector.

Eq. 5.20 can be written as

$$\{X(s)\} - A\{X(s)\} = \{x(0)\} + B\{U(s)\}$$
5.22

To combine the terms on the left-hand side of Eq. 5.22, the rules for matrix addition and subtraction must be observed, namely that such an arithmetic operation can be performed only if the two matrices have the same order. Thus, since s is a scalar quantity, the unit matrix I must be introduced and Eq. 5.22 written as

$$sI\{X(s)\} - A\{X(s)\} = \{x(0)\} + B\{U(s)\}$$

and the Laplace transformed state variable can now be expressed as

$$\{X(s)\} = (sI - A)^{-1}(\{x(0)\} + B\{U(s)\})$$
5.23

Substitution of Eq. 5.23 into Eq. 5.21, yields the system response equation in algebraic form, i.e.

$$\{Y(s)\} = C(sI - A)^{-1}(\{x(0)\} + B\{U(s)\}) + D\{U(s)\}$$
5.24

The transfer function representation as defined in Chapter 2 can be used to describe multivariable systems if the scalar variables and transfer functions are replaced by vector variables and matrix transfer functions respectively. Therefore, using vector notation, the transfer function G(s) is given by

$$G(s) = \frac{\{Y(s)\}}{\{U(s)\}}$$
5.25

Substituting Eq. 5.24 into Eq. 5.25, for zero initial conditions, gives

$$G(s) = C(sI - A)^{-1}B + D$$
 5.26

G(s) is a matrix transfer function, and each element represents one component of the single variable output-input relationship

$$G_{ij}(s) = \frac{Y_i(s)}{U_j(s)}$$
5.27

Examination of Eq. 5.26 shows that the inverse matrix $(sI - A)^{-1}$ plays an important role in the solution of the system equations and in determining the transfer function. It is shown in Appendix C that $(sI - A)^{-1}$ can be evaluated as

$$(sI - A)^{-1} = \frac{\operatorname{adj}(sI - A)}{|sI - A|}$$

Expansion of the determinant |sI - A| yields an *n*th order polynomial in *s*. This determinant appears as the denominator of the transfer function, therefore the characteristic equation is given by

$$|sI - A| = 0$$

and the roots of this *n*th order polynomial are the poles of the transfer function.

Example 5.2. To demonstrate the use of Eq. 5.26, the transfer function relating the input u(t) and the output response $y(t) = x_1(t)$ is derived for the second order system shown in Fig. 2.6. For a natural frequency of 2 rad/second and a damping factor of 0.5, it can be shown from Eq. 2.40 that the state equation is

$$\begin{cases} \dot{x}_1(t) \\ \dot{x}_2(t) \end{cases} = \begin{bmatrix} 0 & 1 \\ -4 & -2 \end{bmatrix} \begin{cases} x_1(t) \\ x_2(t) \end{cases} + \begin{bmatrix} 0 \\ 4 \end{bmatrix} u(t)$$

In matrix form the system response is

$$\{y(t)\} = \begin{bmatrix} 1 & 0 \end{bmatrix} \{x(t)\}$$

hence from Eq. 5.26

Now

$$G(s) = C(sI - A)^{-1}B + D = \begin{bmatrix} 1 & 0 \end{bmatrix} \begin{bmatrix} s & -1 \\ 4 & s+2 \end{bmatrix}^{-1} \begin{bmatrix} 0 \\ 4 \end{bmatrix}$$

adj $(sI - A) = \begin{bmatrix} s+2 & 1 \\ -4 & s \end{bmatrix}$ and $|sI - A| = s^2 + 2s + 4$
 $(s^2 + 2s + 4)G(s) = \begin{bmatrix} 1 & 0 \end{bmatrix} \begin{bmatrix} s+2 & 1 \\ -4 & s \end{bmatrix} \begin{bmatrix} 0 \\ 4 \end{bmatrix}$
 $= \begin{bmatrix} 1 & 0 \end{bmatrix} \begin{bmatrix} 4 \\ 4s \end{bmatrix} = 4$
i.e. $G(s) = \frac{4}{s^2 + 2s + 4}$

The result is verified by the results from the conventional Laplace transform technique in Chapter 2, Eq. 2.17.

5.4 Solution of the state vector differential equation

As explained earlier in this chapter, one of the benefits to be gained from use of state space representation is that it enables system performance in the time domain to be more readily computed. The aim of this section is to develop a general solution for the vector differential equation, Eq. 5.16, from which a system time response can be predicted. To simplify the mathematical development of the general solution, first consider the solution of a simple onedimensional differential equation representing the dynamic behaviour of a first-order system,

i.e.
$$\dot{x}(t) = ax(t) + bu(t)$$
 5.28

This was solved in Section 4.1 for certain specific functions u(t).

To determine the general solution of Eq. 5.28, the law of superposition is used, an inherent and unique property of a linear system. Consider x(t) to comprise two motions, a free component $x_i(t)$ with u(t) = 0 and initial condition x(0), and a forced component $x_j(t)$ for the actual input u(t) but with zero initial state, x(0) = 0.

(i) Free motion $x_i(t)$

The equation is

$$\dot{x}_{i}(t) - ax_{i}(t) = 0$$
, with $x_{i}(t) = x(0)$ at time $t = 0$

The Laplace transform of this equation is

$$(s-a)X_i(s) = x(0)$$

and from the Laplace transform table given in Section 2.2, the solution is

$$x_i(t) = x(0)e^{at}$$
 5.29

(ii) Forced motion $x_j(t)$ The equation is

$$\dot{x}_{i}(t) - ax_{i}(t) = bu(t)$$
, with $x_{i}(t) = 0$ at time $t = 0$,

and its Laplace transform is

$$(s-a)X_i(s) = bU(s)$$
5.30

The most obvious choice for a solution for $x_j(t)$ is one of similar form to that of $x_i(t)$

i.e.
$$e^{at}x_{p}(t) = x_{i}(t)$$
 5.31

where $x_{p}(t)$ is an unknown function.

Now $\frac{d}{dt} \left[e^{at} x_{p}(t) \right] = a e^{at} x_{p}(t) + e^{at} \dot{x}_{p}(t)$

Converting this to an algebraic form by use of the Laplace transform gives, since initial conditions are zero,

$$s\mathscr{L}[e^{at}x_{p}(t)] = a\mathscr{L}[e^{at}x_{p}(t)] + \mathscr{L}[e^{at}\dot{x}_{p}(t)]$$
$$(s-a)\mathscr{L}[e^{at}x_{p}(t)] = \mathscr{L}[e^{at}\dot{x}_{p}(t)]$$

Substituting for $e^{at}x_p(t)$ from Eq. 5.31 gives

$$(s-a)X_{i}(s) = \mathscr{L}[e^{at}\dot{x}_{o}(t)]$$

and on substituting bU(s) for the left-hand side (Eq. 5.30), taking the Laplace inverse yields

$$bu(t) = e^{at}\dot{x}_{p}(t)$$
$$x_{p}(t) = \int_{0}^{t} e^{-a\tau}bu(\tau) d\tau$$

or

where τ is a dummy variable of integration that will vanish when the integration limits are inserted.

Thus

$$x_{j}(t) = e^{at} \int_{0}^{t} e^{-a\tau} bu(\tau) d\tau$$
$$= \int_{0}^{t} e^{a(t-\tau)} bu(\tau) d\tau \qquad 5.32$$

and the total solution to Eq. 5.28 is

$$x(t) = x(0)e^{at} + \int_{0}^{t} e^{a(t-\tau)}bu(\tau) d\tau$$
 5.33

An integral of a product of two functions of this form, Eq. 5.32, is called a convolution integral as was explained in Section 4.7.

Since the vector equation, Eq. 5.16, is simply a collection of first order differential equations which must be solved simultaneously, then for the general case involving n independent state variables

$$\{x(t)\} = e^{At}\{x(0)\} + \int_{0}^{t} e^{A(t-\tau)} B\{u(\tau)\} d\tau \qquad 5.34$$

The exponential term e^{At} which is an $n \times n$ matrix that appears in both the free and forced motion portions of the solution is called the *solution matrix* or *transition matrix* $\varphi(t)$. The final solution to the vector differential equation is written

$$\{x(t)\} = \varphi(t)\{x(0)\} + \int_{0}^{t} \varphi(t-\tau)B\{u(\tau)\} d\tau \qquad 5.35$$

The evaluation of the solution matrix $\varphi(t)$ is possible by a series expansion of e^{At} , but this is impractical manually for all but the simplest problems. The most direct method is by taking the inverse Laplace transform of Eq. 5.23, that is

$$\{x(t)\} = \{x(0)\} \mathcal{L}^{-1}[(sI - A)^{-1}] + \mathcal{L}^{-1}[(sI - A)^{-1}B\{U(s)\}]$$
 5.36

Thus, to obtain $\varphi(t) = e^{At} = \mathscr{L}^{-1}[(sI - A)^{-1}]$, the inverse of the (sI - A) matrix is found and the inverse transform of the resulting matrix is evaluated term by term.

From the definition of the inverse matrix

$$\mathscr{L}[\boldsymbol{\varphi}(t)] = \frac{\operatorname{adj}(sI - A)}{|sI - A|}$$
5.37

Example 5.3. It will be illustrated in this example how Eq. 5.37 can be used to find all the elements of the solution matrix $\varphi(t)$.

Consider a second order system (with coefficients different to those in Example 5.2 for arithmetic simplicity) described by the transfer function

$$\frac{X(s)}{U(s)} = \frac{2}{s^2 + 3s + 2}$$

which in state vector form is

$$\{\dot{\mathbf{x}}(t)\} = \begin{bmatrix} 0 & 1 \\ -2 & -3 \end{bmatrix} \{\mathbf{x}(t)\} + \begin{bmatrix} 0 \\ 2 \end{bmatrix} u(t)$$

The characteristic equation is |sI - A| = 0 and expansion of this determinant yields

$$\begin{vmatrix} s & 0 \\ 0 & s \end{vmatrix} - \begin{bmatrix} 0 & 1 \\ -2 & -3 \end{vmatrix} = 0$$
$$\therefore \begin{vmatrix} s & -1 \\ 2 & s+3 \end{vmatrix} = 0$$
$$\therefore s^2 + 3s + 2 = 0$$
$$(s+1)(s+2) = 0$$

or

and from Eq. 5.37

$$\mathscr{L}[\varphi(t)] = \frac{1}{(s+1)(s+2)} \begin{bmatrix} s+3 & 1\\ -2 & s \end{bmatrix}$$

$$\mathcal{L}(s+1)(s+2) = \mathscr{L}^{-1} \begin{bmatrix} \frac{s+3}{(s+1)(s+2)} & \frac{1}{(s+1)(s+2)} \\ \frac{-2}{(s+1)(s+2)} & \frac{s}{(s+1)(s+2)} \end{bmatrix}$$

According to matrix theory, the inverse Laplace transform of a matrix is a matrix whose elements are the inverse transforms of the corresponding elements of the original matrix.

Now
$$\mathscr{L}^{-1}\left(\frac{s+3}{(s+1)(s+2)}\right) = \mathscr{L}^{-1}\left(\frac{2}{s+1} - \frac{1}{s+2}\right) = (2e^{-t} - e^{-2t})$$

 $\mathscr{L}^{-1}\left(\frac{1}{(s+1)(s+2)}\right) = \mathscr{L}^{-1}\left(\frac{1}{s+1} - \frac{1}{s+2}\right) = (e^{-t} - e^{-2t})$
 $\mathscr{L}^{-1}\left(\frac{-2}{(s+1)(s+2)}\right) = \mathscr{L}^{-1}\left(\frac{2}{s+2} - \frac{2}{s+1}\right) = (-2e^{-t} + 2e^{-2t})$
 $\mathscr{L}^{-1}\left(\frac{s}{(s+1)(s+2)}\right) = \mathscr{L}^{-1}\left(\frac{2}{s+2} - \frac{1}{s+1}\right) = (-e^{-t} + 2e^{-2t})$

and hence

$$\varphi(t) = \begin{bmatrix} (2e^{-t} - e^{-2t}) & (e^{-t} - e^{-2t}) \\ (-2e^{-t} + 2e^{-2t}) & (-e^{-t} + 2e^{-2t}) \end{bmatrix}$$

Note that the numerical values -1 and -2 in the exponential terms are the roots of the characteristic equation defined by |sI - A| = 0.

To apply this solution matrix, consider first the case u(t) = 0, $x_1(0) = 1$ and $x_{2}(0) = 0$. This would mean, for a simple spring mass system, that the mass is released with zero initial velocity and finite initial displacement in the absence of a forcing function.

From Eq. 5.35

$$\{x(t)\} = \varphi(t)\{x(0)\}$$
$$= \varphi(t)\begin{bmatrix} 1\\ 0 \end{bmatrix}$$
$$x_1(t) = 2e^{-t} - e^{-2t}$$
$$x_2(t) = -2e^{-t} + 2e^{-t}$$

and

Secondly, consider the response to a unit step input. If the system is initially at rest with zero displacement, $\{x(0)\} = 0$. For this particular example [0]

21

$$B = \begin{bmatrix} 0 \\ 2 \end{bmatrix} \text{ and } u(\tau) = 1 \text{ for } \tau > 0; \text{ hence from Eq. 5.35}$$
$$x_1(t) = 2 \int_0^t (e^{-(t-\tau)} - e^{-2(t-\tau)}) \, d\tau$$
$$= 1 - 2e^{-t} + e^{-2t}$$
and
$$x_2(t) = 2 \int_0^t (-e^{-(t-\tau)} + 2e^{-2(t-\tau)}) \, d\tau$$
$$= 2e^{-t} - 2e^{-2t}$$

and

This example demonstrates that from a knowledge of a system A matrix, the solution matrix, $\varphi(t)$, can be found and hence the time response evaluated for a given set of conditions. Thus the evolution of an excited system from one state to another through time may be visualized as a process of state transition.

5.5 Discrete time model

If a system response or forcing function can be observed at discrete intervals of time only, that system is termed a discrete or sampled-data system and is mathematically modelled by difference equations rather than by differential equations. A digital computer solution must use a discrete-time model of the continuous time system. In formulating the discrete-time model a sequence of numerical and logical operations, known as a computer algorithm, is devised and programmed for digital computer solution.
The use of state model formulation avoids the need for the introduction of z-transform theory, a technique which extends the transfer function model into the realms of discrete data systems. Although striking similarities exist between Laplace and z-transform representation, the step from continuous to discrete system representation is more direct using the state model, and is more readily understood.

The solution of the state equations, Eq. 5.16 and Eq. 5.17, has been derived, Eq. 5.34 or Eq. 5.35, and consists of a set of solutions of the form of Eq. 5.33.

$$x(t) = x(0)e^{at} + \int_{0}^{t} e^{a(t-\tau)}bu(\tau) d\tau$$
 (5.33)

In order to carry out the integration digitally, the signals must be considered in sampled form, and the input function $u(\tau)$ must be represented by a suitable approximation. If the sampling interval is T it is convenient to approximate $u(\tau)$ by a series of steps forming a staircase, Fig. 5.6; the choice of the value of



Fig. 5.6 Sampling of input function

T dictates the resolution accuracy possible in the evaluation of x(t), but also dictates the computing time necessary to achieve the solution. Therefore in practical engineering situations a compromise between accuracy and computing time is always necessary. If $t_k = kT$, where k is a positive integer, it will be assumed that for the time interval $t_k < t < t_{k+1}$, $u(\tau)$ is a constant u_k , and hence Eq. 5.33 can be written as

$$x(t_{k+1}) = x(t_k)e^{aT} + \int_0^T e^{a(T-\tau)}bu_k d\tau$$

$$= x(t_k)e^{aT} + bu_k \left[-\frac{1}{a} e^{a(T-\tau)} \right]_0^T$$

$$x(t_{k+1}) = e^{aT}x(t_k) + (e^{aT} - 1)\frac{b}{a}u_k$$
5.39

96 State Space Representation and Analysis

To illustrate the application of Eq. 5.39 consider a unit ramp input, $u(\tau) = t$, and for arithmetic simplicity let a = -1 and b = 1 which is equivalent to a time constant of 1 second. The continuous response, obtained by analytical solution of Eq. 5.33 or from Eq. 4.9, for zero initial conditions, is

$$x(t) = t - 1 + e^{-t}$$

which is plotted in Fig. 5.7. The discrete time solution is evaluated from Eq. 5.39, and for a value of T = 0.5 seconds, since in this case $u_k = kT$, the equation becomes

$$x(t_{k+1}) = e^{-0.5}x(t_k) - (e^{-0.5} - 1)0.5 k$$

= 0.6065x(t_k) + 0.1967 k

Successive values of the discrete time solution are thus

$$\begin{aligned} x(t_0) &= x(0) = 0\\ x(t_1) &= x(0.5) = 0 + 0 = 0\\ x(t_2) &= x(1.0) = 0 + 0.1967 = 0.1967\\ x(t_3) &= x(1.5) = (0.6065)(0.1967) + 0.1967(2) = 0.5127\\ \text{etc.} \end{aligned}$$



Fig. 5.7 Response of first order system to unit ramp evaluated discretely

The resulting response is shown in Fig. 5.7 together with the corresponding results obtained for T = 0.25. As is to be expected, as the sampling interval is decreased the resulting response tends closer to the true continuous curve. It should be noted that reasonable accuracy has been achieved with a sampling interval of 0.25 seconds; this is relatively large compared to the time constant of 1 second.

By inspection of Eq. 5.33 and Eq. 5.39, the general solution to Eq. 5.34, making use of the staircase approximation $\{u(\tau)\} = \{u_k\}$, yields

$$\{x(t_{k+1})\} = e^{AT}\{x(t_k)\} + [-e^{A(T-t)}]_0^T A^{-1} B\{u_k\}$$

= $e^{AT}\{x(t_k)\} + [e^{AT} - I]A^{-1} B\{u_k\}$
 $\{x(t_{k+1})\} = \varphi(T)\{x(t_k)\} + [\varphi(T) - I]A^{-1} B\{u_k\}$ 5.40

or

If the substitution $A_1 = \varphi(T)$ and $B_1 = [\varphi(T) - I]A^{-1}B$ is made in Eq. 5.40

$$\{x(t_{k+1})\} = A_1\{x(t_k)\} + B_1\{u_k\}$$
 5.41

The response $\{x(t_k)\}\$ at time $t_k = kT$ is computed by repeated application of Eq. 5.41:

$$\{x(t_1)\} = A_1 \{x(0)\} + B_1 \{u(0)\}$$

$$\{x(t_2)\} = A_1 \{x(t_1)\} + B_1 \{u_1\}$$

$$= A_1^2 \{x(0)\} + A_1 B_1 \{u(0)\} + B_1 \{u_1\}$$

$$\{x(t_3)\} = \dots$$

$$\dots$$

$$\dots$$

$$\dots$$

$$\{x(t_k)\} = A_1^k \{x(0)\} + \sum_{i=0}^{k-1} A_1^{k-1-i} B_1 \{u_i\}$$

$$5.42$$

The sampling interval T, involved in evaluating A_1 and B_1 , must be chosen to give the desired accuracy in the calculated time response. The most direct way to evaluate A_1 and B_1 is to use the series expansion of e^{AT} :

$$e^{AT} = I + AT + \frac{1}{2!} (AT)^{2} + \dots = A_{1}$$

and $[e^{AT} - I]A^{-1}B = \left[\left[I + AT + \frac{1}{2!} (AT)^{2} + \dots \right] - I \right] A^{-1}B$
 $= T \left[I + \frac{1}{2!} AT + \frac{1}{3!} (AT)^{2} + \dots \right] B = B_{1}$
5.43

The matrix inversion, A^{-1} , is avoided in Eqs. 5.43 and the series expansion of A_1 and B_1 can be truncated to any desired accuracy.

It has been shown in this chapter that the transfer function and state matrix constitute alternative methods for the mathematical representation of a linear dynamic system. It is now recognized that in general the theoretical results needed in system design can be obtained equally well either by vector-space methods or by algebraic methods using Laplace transforms.

The area in which the state concept is most effective is that of discrete and multivariable system studies. The state concept makes possible a systematic formulation for all systems, including those with arbitrary sampling patterns and non-linear operation. However, frequency response methods give an unrivalled engineering insight into system behaviour.

98 State Space Representation and Analysis

It must be emphasized that despite the unified approach that is possible by use of state description, the Laplace formulation of system equations remains significant with practising engineers. The particular problem to be studied, however, should always determine the preference of one approach to the other.

6 Frequency Response of Systems

Chapter 4 has described how for specific forcing functions the time response of a system of known transfer function can be evaluated, and in what ways the form of the response is dependent on the form of the transfer function. In this chapter the equivalent response in the frequency domain is presented. Frequency response methods of analysis have been very popular since the early stages of development of control theory, and are still of importance largely because of the physical understanding which they help the engineer to acquire. The methods are easy to apply, they are graphical, and they offer a good basis for synthesizing systems, in that they indicate clearly the type of change that is required to improve system dynamic behaviour.

The description of a system in the frequency domain is given in terms of the response to a sinusoidally varying input signal after all initial transients have died out. Provided the system is linear this steady state output, which is the particular integral term of the solution of the governing differential equation, is a sinusoid of the same frequency as the input, but with a shift of phase and a change of amplitude. The ratio of the amplitude of the output sine wave to the amplitude of the input sine wave is usually referred to as the magnitude (or sometimes as the magnitude ratio, amplitude ratio, or gain); the shift of phase of the output sine wave relative to the input is termed simply the phase. The magnitude and phase are dependent both on the system transfer function and on the forcing frequency but not, with a linear system, on the amplitude. The variation of magnitude and phase with frequency is traditionally known as the frequency response or the harmonic response of the system. Any non-linearity which is present in the system introduces signal components at higher frequencies, with the result that the output then contains the basic forcing frequency plus certain of its harmonics.

The main part of the chapter shows how the harmonic response information can be found from the transfer function G(s) by letting $s = j\omega$, and how it can be presented graphically by means of either a polar diagram or a Bode diagram. The physical significance of plots of various shapes is discussed to illustrate the understanding that can be obtained from a knowledge of the harmonic response characteristics of a system. Frequency response testing of practical systems, and the estimation of transfer functions from the measured frequency response, is then discussed. The chapter concludes by describing the criteria which are used to describe performance in the frequency domain.

6.1 The transfer function in the frequency domain

The object of this section is to show that the magnitude and phase, the ratio of the amplitude of the steady state output to the amplitude of the input sine wave, and the phase shift between the output and input sinusoids, are given respectively by the modulus and argument of $G(j\omega)$, the transfer function with s replaced by $j\omega$.

For a linear system, the transfer function G(s) has been shown to be a ratio of two polynomials in s, each of which can be factorized to give

$$G(s) = \frac{C(s)}{U(s)} = \frac{K(s-z_1)(s-z_2)\dots(s-z_m)}{(s-p_1)(s-p_2)(s-p_3)\dots(s-p_n)}$$

$$6.1$$

 z_1, z_2, \ldots, z_m are defined as the zeros of G(s), values of s which make the function zero, while $p_1, p_2, p_3, \ldots, p_n$ are the poles of G(s), values which make the function infinite.

If the input to this system is

$$u(t) = \sin \omega t$$

a sine wave of unit amplitude and of frequency ω rad/second, then the Laplace transform of the input (Table 2.1) is

$$U(s) = \frac{\omega}{s^2 + \omega^2}$$

$$C(s) = G(s)U(s) = \frac{K\omega(s - z_1)(s - z_2)\dots(s - z_m)}{(s^2 + \omega^2)(s - p_1)(s - p_2)\dots(s - p_m)}$$
6.2

This expression can be rewritten by a partial fraction expansion to give

$$C(s) = \frac{G(s)\omega}{s^2 + \omega^2} = \frac{A_1}{s - j\omega} + \frac{A_2}{s + j\omega} + \frac{B_1}{s - p_1} + \frac{B_2}{s - p_2} + \dots + \frac{B_n}{s - p_n} = 6.3$$

where $A_1, A_2, B_1, B_2, \dots, B_n$ are constants. Taking the Laplace inverse (using the fourth transform pair of Table 2.1) gives the time response as

$$c(t) = A_1 e^{j\omega t} + A_2 e^{-j\omega t} + B_1 e^{p_1 t} + B_2 e^{p_2 t} + \dots + B_n e^{p_n t}$$
 6.4

The first two terms describe the particular integral component of the solution, frequently referred to as the steady state response, while the remaining terms describe the complementary function, the transient response. Provided the system is stable and linear the poles $p_1, p_2, \ldots p_n$ all have negative real parts and hence all of the modes in Eq. 6.4 decay to zero with increasing time t except the first two.

Hence
$$[c(t)]_{t \to x} = A_1 e^{j\omega t} + A_2 e^{-j\omega t}$$

To determine the coefficient A_1 multiply both sides of equation 6.3 by $(s - j\omega)$, and then let $s = j\omega$:

$$A_{1} = \left[\frac{(s - j\omega)G(s)\omega}{s^{2} + \omega^{2}}\right]_{s = j\omega} = \left[\frac{G(s)\omega}{s + j\omega}\right]_{s = j\omega} = \frac{G(j\omega)}{2j} = \frac{1}{2j} |G(j\omega)| e^{j\angle G(j\omega)}$$

Similarly to obtain A_2 multiply by $(s + j\omega)$ and let $s = -j\omega$:

$$A_2 = \left[\frac{(s+j\omega)G(s)\omega}{s^2+\omega^2}\right]_{s=-j\omega} = \frac{G(-j\omega)}{-2j} = -\frac{1}{2j} |G(j\omega)| e^{-j\angle G(j\omega)}$$

$$\therefore \quad [c(t)]_{t \to \infty} = |G(j\omega)| \frac{1}{2j} \{ e^{j\omega t + j \angle G(j\omega)} - e^{-j\omega t - j \angle G(j\omega)} \}$$

i.e. $[c(t)]_{t \to x} = |G(j\omega)| \sin (\omega t + \angle G(j\omega))$

The magnitude and phase as functions of frequency can thus be obtained from the transfer function G(s) by replacing s by $j\omega$ and determining the modulus and argument of $G(j\omega)$, which for any particular frequency is generally a complex number. In the next section the magnitude and phase characteristics are evaluated for different forms of G(s).

6.2 Polar plots

The frequency response information for a system can conveniently be displayed on an Argand diagram and is then referred to by the control engineer as a polar plot. The input sinusoid is considered to be represented by a unit vector lying along the positive real axis, and for any given frequency the magnitude and phase of the output can then be defined by a corresponding output vector. By convention a phase lag is represented by rotation of the vector in a clockwise direction; the quadrants are also numbered in this direction relative to the real axis as datum.

A polar plot is a plot showing the variation of magnitude and phase of the output on polar coordinates, for a constant amplitude input, as the frequency ω is varied from zero to infinity. The curve drawn is the locus of the termini of the system output vectors, and a typical plot is shown in Fig. 6.1. The harmonic information is evaluated discretely for specific values of ω , and thus numerical values of frequency should be marked against points on this locus. The plot shown in the figure represents a third order system with a steady



Fig. 6.1 Typical polar plot



6.5

102 Frequency Response of Systems

state gain of unity. At frequencies tending to zero the output has the same amplitude as the input and is in phase with it; as frequency increases the output amplitude is seen to decrease, tending towards zero for high frequencies, while the output lags the input by an ever increasing amount tending towards 270° maximum for a third order system (180° for a second order system, etc. as will be shown later in this section).

The harmonic information for such a plot can be calculated if the transfer function is known, or if an actual system is available for test it can be determined experimentally, provided that frequency response testing is feasible.

Consider, first, plots obtained analytically for systems of known transfer function.

(a) *First order system, or simple lag.* Examples of some physical systems having first order transfer functions have been given in Section 2.4 and their transient response characteristics have been studied in Section 4.1. The transfer function for unity steady state gain is

$$G(s) = \frac{1}{1 + \tau s}$$

To obtain the frequency response characteristics replace s by $j\omega$:

$$G(j\omega) = \frac{1}{1+j\omega\tau} = \frac{1-j\omega\tau}{1+\omega^{2}\tau^{2}} = \left(\frac{1}{1+\omega^{2}\tau^{2}}\right) - j\left(\frac{\omega\tau}{1+\omega^{2}\tau^{2}}\right) \qquad 6.6$$

This gives the real and imaginary coordinates of the harmonic locus, and can be rewritten in terms of magnitude and phase as

$$G(j\omega) = |G(j\omega)| \angle G(j\omega) = \frac{1}{\sqrt{(1+\omega^2\tau^2)}} \angle (-\tan^{-1}\omega\tau)$$

$$6.7$$

For

$$\omega = 0 ||G(j\omega)| = 1 \text{ and } \angle G(j\omega) = 0$$

As

$$\omega \rightarrow \infty |G(j\omega)| \rightarrow 0 \text{ and } \angle G(j\omega) \rightarrow -90^{\circ}$$

The locus can be shown to be a semicircle as follows. In Eq. 6.6 let the real and imaginary coordinates for $G(j\omega)$ at frequency ω be x and y respectively

$$x = \frac{1}{1 + \omega^2 \tau^2}$$
$$y = -\frac{\omega \tau}{1 + \omega^2 \tau^2} = -x\omega \tau$$

and

$$y^{2} = x^{2}\omega^{2}\tau^{2} = x^{2}\left(\frac{1-x}{x}\right) = x - x^{2}$$

$$\therefore \quad (x - \frac{1}{2})^{2} + y^{2} = (\frac{1}{2})^{2}$$

6.8

The harmonic locus for a simple lag is thus a semi-circle with centre (0.5, 0) and radius 0.5 as shown in Fig. 6.2. Frequency points can be marked on this locus by determining the phase lag $\varphi = \tan^{-1} \omega \tau$.



Fig. 6.2 Polar plot for first order system, $G(j\omega) = \frac{1}{1 + i\omega\tau}$

When $\omega = 1/\tau$, $\varphi = -45^{\circ}$ which corresponds to the mid-point of the semicircular locus. For a frequency $1/\tau$ rad/second the phase lag is 45°; for higher frequencies the lag increases but it never exceeds 90°. It must be realized that this single semi-circle represents all unity gain first order systems, irrespective of the value of the time constant, but that any single point on the locus represents different frequencies for systems with different time constants.

(b) Second order system, or quadratic lag. The transfer function of this, Eq. 2.17, is, for unity gain,

$$G(s) = \frac{\omega_n^2}{s^2 + 2\zeta\omega_n s + \omega_n^2}$$

$$G(j\omega) = \frac{\omega_n^2}{\omega_n^2 - \omega^2 + j2\zeta\omega\omega_n} = \frac{1}{1 - \left(\frac{\omega}{\omega_n}\right)^2 + j\left(2\zeta\frac{\omega}{\omega_n}\right)}$$
 6.9

By inserting a range of values of $\frac{\omega}{\omega_n}$ and evaluating the real and imaginary

parts of Eq. 6.9 (or alternatively the magnitude and phase) the location of points on the polar plot can be calculated and the harmonic locus drawn to join them.

e.g. for
$$\omega = 0$$
 $G(j\omega) = 1 = 1 \angle 0^{\circ}$
 $\omega = \infty$ $G(j\omega) = -\frac{\omega_n^2}{\omega^2} = 0 \angle -180^{\circ}$
 $\omega = \omega_n$ $G(j\omega) = \frac{1}{j2\zeta} = \frac{1}{2\zeta} \angle -90^{\circ}$

104 Frequency Response of Systems

$$\omega = 2\omega_n \quad G(j\omega) = \frac{1}{-3 + j4\zeta} = \frac{-3 - j4\zeta}{9 + 16\zeta^2}$$
$$= \frac{1}{\sqrt{(9 + 16\zeta^2)}} \angle \left(\tan^{-1}\frac{4\zeta}{3} - 180^\circ\right)$$

The completed locus is one member of a family of curves of which three are shown in Fig. 6.3, the shape of the curve being dependent on the value of the



Fig. 6.3 Polar plots for second order system. $G(j\omega) = \frac{\omega_n^2}{\omega_n^2 - \omega^2 + j2\zeta\omega\omega_n}$

damping factor. For $\zeta > 1$ the magnitude is 1 for zero frequency, and decreases continuously with increase in frequency, while the phase lag increases to a maximum of 180°. For $\zeta < 1$ there is a range of frequency for which the magnitude exceeds unity, and when $\zeta \ll 1$, as illustrated by the curve for $\zeta = 0.5$, the phase lag increases very rapidly over a small range of frequency around $\omega_{\rm p}$.

It should be noted that for a second order system, for all values of damping

factor ζ , when $\omega = \omega_n$ the phase lag is 90° and the magnitude is $\frac{1}{2r}$.

(c) Integrator. Consider now a device such as an electric motor or a hydraulic ram which integrates the input signal at a constant rate. Let this rate be one unit per second.

$$G(s) = \frac{1}{s}$$

$$G(j\omega) = \frac{1}{j\omega} = -\frac{j}{\omega} = \frac{1}{\omega} \angle -90^{\circ}$$
6.10

The harmonic locus thus lies wholly on the imaginary axis (Fig. 6.4). The phase lag is 90° for all frequencies, and the magnitude is the reciprocal of the

0	1	Re
	- ω = 4 - ω = 2	
-1-	- ∸ω≑1	
-2 -	- ω = 0.5 rad/second	

Fig. 6.4 Polar plot for integrator. $G(j\omega) = \frac{1}{j\omega}$

frequency in rad/second. Hence as $\omega \to 0$ the magnitude becomes infinite, and as $\omega \to \infty$ the magnitude tends to zero.

(d) Higher order systems. The harmonic information and hence polar plot for any higher order transfer function can be obtained similarly by writing down $G(j\omega)$, rationalizing, and inserting in turn different values of frequency ω . If the transfer function appears in factorized form the magnitude and phase information is likely to be more easily obtained by thinking of the system as a number of elements in series. The overall phase for any given frequency is then obtained by adding the individual phase components, and the overall magnitude is obtained by multiplying together the individual values of magnitude.

Example 6.1. Obtain a polar plot for a system with transfer function

$$G(s) = \frac{1}{(1+2s)(s^2+s+1)}$$

Letting $s = j\omega$

$$G(j\omega) = \frac{1}{(1+j2\omega)(1-\omega^2+j\omega)}$$
$$= \frac{1}{1-3\omega^2+j(3\omega-2\omega^3)}$$
$$= \frac{(1-3\omega^2)-j\omega(3-2\omega^2)}{1+3\omega^2-3\omega^4+4\omega^6}$$

Now insert a range of numerical values of frequency in rad/second

e.g. for
$$\omega = 1$$
 $G(j\omega) = -0.4 - j0.2$
 $\omega = 0.5$ $G(j\omega) = +0.154 - j0.77$
etc.

106 Frequency Response of Systems

Such calculation is rather tedious and errors easily occur, particularly with high order transfer functions. Considering G(s) as two factors multiplied together, curves for the first and second order factors can be drawn with the relevant frequencies marked on (as Fig. 6.2 and Fig. 6.3) and for corresponding frequency points the vectors can be combined graphically. Easiest, however, is to use the graphical method as an aid to understanding, but to do the multiplication of magnitudes and addition of phase angles in tabular form, obtaining the component magnitudes and phases by calculation, or from standard curves or tables (Table 6.1, Fig. 6.5).





6.3 Bode plots

An alternative method for obtaining and presenting system frequency response data which is particularly useful where the transfer function is available in factorized form is the Bode plot. This consists of two plots normally drawn on semi-logarithmic graph paper: a magnitude plot, $\log |G(j\omega)|$, and a phase plot, $\angle G(j\omega)$, both on a linear scale, against frequency ω plotted on a logarithmic scale. The magnitude is most commonly plotted in decibels i.e. $20 \log_{10} |G(j\omega)|$. One reason for this choice of graph is that, since the magnitude is plotted in logarithmic form, the overall magnitude and phase information can both be obtained from the component parts by graphical addition. A second important advantage of such a plot is that certain approximations

Table 6.1. Tabular evaluation of harmonic response (Example 6.1)

Frequency (rad/second)	0	0.2	04	07	Ť	2
$\left \frac{1}{1+j2\omega}\right = \frac{1}{\sqrt{(1+4\omega^2)}}$	1	0.928	0 781	0 581	0.447	0 243
$\left \frac{1}{1-\omega^2+j\omega}\right = \frac{1}{\sqrt{\{(1-\omega^2)^2+\omega^2\}}}$	1	1 020	1 075	1 1 5 5	1 000	0.277
(G(j w))	1	0.947	0.840	0.671	0 447	0.067
$L\left(\frac{1}{1+j2\omega}\right) = \tan^{-1} 2\omega$ (degrees lag)	Q	21.8	38.7	54.5	63 4	76.0
$\mathcal{L}\left(\frac{1}{1-\boldsymbol{\omega}^2+\boldsymbol{\omega}}\right) = \tan^{-1}\frac{\boldsymbol{\omega}}{1-\boldsymbol{\omega}^2}$ (degrees lag)	0	11,8	25.5	53.9	90.0	146 3
∠G(jw) (degrees lag)	0	33.6	64.2	108.4	153.4	222 3

using straight line constructions can be quickly drawn and often suffice for accuracy.

It has been shown in Section 6.1 that any transfer function can be factorized into the form

$$G(s) = \frac{K(s-z_1)(s-z_2)\dots(s-z_m)}{(s-p_1)(s-p_2)\dots(s-p_n)}$$

The zeros $z_1, z_2 ... z_m$ and the poles $p_1, p_2, ..., p_n$ will each be either zero, real, or complex, and thus in general G(s) can be considered to be composed entirely of terms of the four following types appearing on the numerator or the denominator:

K, s,
$$1 + \tau s$$
, $\frac{s^2 + 2\zeta \omega_n s + \omega_n^2}{\omega_n^2}$

Hence $G(j\omega)$ is composed of multiples or quotients of terms of the form

K, j
$$\omega$$
, 1 + j $\omega\tau$, $\frac{\omega_n^2 - \omega^2 + j2\zeta\omega\omega_n}{\omega_n^2}$

Consider now the Bode plots for these types of component term, and the way in which they can be added to produce a plot for a known transfer function.

(a) Constant term (gain term)

$$G(s) = K$$
, \therefore $G(j\omega) = K$

The magnitude is $20 \log_{10} K \, dB$; the argument is zero (Fig. 6.6). Hence a gain term has a constant multiplying effect irrespective of frequency, and thus merely shifts the overall magnitude plot up or down by a certain number of dB. There is no effect on phase.

108 Frequency Response of Systems



Fig. 6.6 Bode plot for constant term

(b) Integral term or derivative term (pole or zero at origin)(i) Pole at origin:

$$G(s) = \frac{1}{s}$$

.
$$G(j\omega) = \frac{1}{j\omega} = -\frac{j}{\omega}$$

The magnitude is $\frac{1}{\omega}$, or $-20 \log_{10} \omega \, dB$, which has value 0 dB when the

frequency is 1 rad/second and decreases by 20 dB for a tenfold increase in frequency. With ω plotted on a logarithmic scale the magnitude is represented by a straight line of slope -20 dB per decade of frequency and passing through 0 dB for $\omega = 1$ rad/second (Fig. 6.7). The phase is -90° , a constant



Fig. 6.7 Bode plot for poles or zeros at origin

lag, which does not vary with frequency. If a double pole is present, $G(s) = \frac{1}{s^2}$, the magnitude line has twice the slope, and the phase is a constant -180° .

(ii) Zero at origin:

$$G(s) = s$$
$$G(j\omega) = j\omega$$

The magnitude is 20 $\log_{10} \omega$ dB, a straight line of slope + 20 dB per decade passing through 0 dB at $\omega = 1$ rad/second. The phase is 90°, which means that the output would lead the input by 90° irrespective of frequency.

(c) Simple lag or lead (real pole or zero)

(i) Real pole:

$$G(s) = \frac{1}{1 + \tau s}$$

,
$$G(j\omega) = \frac{1}{1 + j\omega\tau}$$

The magnitude is $|G(j\omega)| = \frac{1}{\sqrt{(1+\omega^2\tau^2)}}$

$$= -20 \log_{100} / (1 + \omega^2 \tau^2) dB$$

A linear asymptotic approximation is frequently used, making use of the following:

 $|G(j\omega)| dB \simeq -20 \log_{10} 1 = 0$ for $\omega \tau \ll 1$

 $|G(j\omega)| dB \simeq -20 \log_{10} \omega \tau$ for $\omega \tau \ge 1$

The latter is a straight line of slope -20 dB per decade of frequency, which intersects the zero dB line when $\omega t = 1$ i.e. at $\omega = 1/\tau$. This is termed the break point or corner frequency. The true plot rounds off the junction as shown in Fig. 6.8, the maximum error being 3 dB at the break point. On either side of

this the errors are 1 dB at ± 1 octave of frequency $\left(\frac{2}{\tau} \text{ and } \frac{0.5}{\tau}\right)$ and 0.3 dB at ± 1 decade of frequency $\left(\frac{10}{\tau} \text{ and } \frac{0.1}{\tau}\right)$. The rounding if required can generally

be done sufficiently accurately by hand through these points.

 $\angle G(i\omega) = -\tan^{-1}\omega\tau$ The phase is

A linear approximation can also be used for phase, as shown in Fig. 6.8, namely 0° for $\omega \le 0.1/\tau$, -90° for $\omega \ge 10/\tau$, and a linear variation between. The true curve is gently curving, the error being approximately $5\frac{1}{2}^{\circ}$ at ω equal to $0.1/\tau$, $0.4/\tau$, $2.5/\tau$, and $10/\tau$. The error is zero at the break point frequency since the lag is exactly 45° when $\omega = 1/\tau$.



Fig. 6.8 Bode plot for a simple lag

(ii) Real zero

$$G(s) = 1 + \tau s$$
$$G(j\omega) = 1 + j\omega\tau$$

The expressions for magnitude and phase are identical to those for a real pole except that they have the opposite sign. The curves on the Bode plot are thus mirror images about the 0 dB and 0 degree lines. The magnitude and phase therefore both increase with frequency, the latter tending towards 90° (a phase

lead) for frequencies in excess of -

For different values of τ the curves are merely shifted along the frequency axis. If curves are to be drawn frequently, templates can be produced to enable the true curves to be drawn quickly and easily.

(d) Quadratic lag or lead (pairs of complex conjugate poles or zeros)

(i) Pair of conjugate poles:

$$G(s) = \frac{\omega_n^2}{s^2 + 2\zeta\omega_n s + \omega_n^2}$$

$$G(j\omega) = \frac{1}{\left(1 - \frac{\omega^2}{\omega_n^2}\right) + j\left(2\zeta \frac{\omega}{\omega_n}\right)}$$

The magnitude is $-20 \log_{10} \sqrt{\left\{ \left(1 - \frac{\omega^2}{\omega_n^2}\right)^2 + \left(\frac{2\zeta\omega}{\omega_n}\right)^2 \right\}} dB$

 $-\tan^{-1}\left(\frac{2\zeta \frac{\omega}{\omega_{n}}}{1-\frac{\omega^{2}}{\omega_{n}^{2}}}\right)$

and the phase is

For
$$\frac{\omega}{\omega_n} \ll 1$$
 $|G(j\omega)| \simeq -20 \log_{10} 1 = 0 \, dB$

and for
$$\frac{\omega}{\omega_n} \gg 1$$
 $|G(j\omega)| \simeq -20 \log_{10} \left(\frac{\omega^2}{\omega_n^2}\right) = -40 \log_{10} \left(\frac{\omega}{\omega_n}\right)$

The straight line approximation for magnitude is thus a line at 0 dB for low frequencies, changing to a line of slope -40 dB per decade at the break point given by $\omega = \omega_n$, the undamped natural frequency. The shape of the true curve depends on the value of ζ , the error being least for $\zeta \simeq 0.5$. Curves for a range of values of ζ are shown in Fig. 6.9. (For $0.35 < \zeta < 0.7$, the error < 3 dB.)

The phase curve varies from 0° for $\frac{\omega}{\omega_n} \ll 1$ to -180° for $\frac{\omega}{\omega_n} \gg 1$, and passes through the -90° point at $\omega = \omega_n$. There is no convenient straight line approximation, the transition again being a function of ζ , and being most rapid for very small values of ζ , as shown in Fig. 6.9.

(ii) Pair of conjugate zeros:

$$G(s) = \frac{s^2 + 2\zeta \omega_n s + \omega_n^2}{\omega_n^2}$$
$$G(j\omega) = \left(1 - \frac{\omega^2}{\omega^2}\right) + j\left(2\zeta \frac{\omega}{\omega}\right)$$

The magnitude and phase are numerically the same as for a pair of poles but of opposite sign, and are thus represented on the Bode plot by families of curves which are the mirror images of those of Fig. 6.9 reflected about the 0 dB and the 0 degree lines.

Here, even more than with the simple lag, templates can prove useful. The curves for the calculated value of ζ can be quickly drawn at the position along the frequency scale corresponding to the appropriate ω_{p} .

Example 6.2. Using straight line approximations draw a Bode diagram for a system with transfer function

$$G(s) = \frac{10}{s(1+0.5s)(1+0.1s)}$$





This transfer function can be seen to be made up of 4 constituent components:

- (a) a constant gain term of 10
- (b) an integrating term 1/s
- (c) a simple lag of time constant 0.5 second
- (d) a simple lag of time constant 0.1 second

Using straight line approximations the contributions of these terms to the overall magnitude are respectively:

(a) a constant of 20 $\log_{10} 10 = 20 \text{ dB}$ for all frequencies

- (b) a line of slope -20 dB/decade, passing through 0 dB at the frequency 1 rad/second
- (c) a magnitude of 0 dB up to a break point at $\frac{1}{0.5} = 2$ rad/second, and thereafter a line of slope -20 dB/decade
- (d) a magnitude of 0 dB up to a break point at $\frac{1}{0.1} = 10$ rad/second, and thereafter a line of slope -20 dB/decade

These magnitude contributions are shown in Fig. 6.10 together with the overall magnitude curve which results from summing them. The overall curve



Fig. 6.10 Bode plot for transfer function $G(s) = \frac{10}{s(1+0.5s)(1+0.1s)}$

is in error particularly in the region of the break points. The true curves for (c) and (d) and hence for the overall curve can be drawn in with sufficient accuracy for most purposes by interpolating by eye, using the guide points of error being 3 dB at the corner frequency, 1 dB at ± 1 octave, 0.3 dB at ± 1 decade. It can be seen that the result is a rounding of the corners.

The contributions of these four components to the overall phase are respectively:

(a) no effect.

114 Frequency Response of Systems

- (b) a constant phase lag of 90° for all frequencies.
- (c) zero lag to $\omega = 0.2$ rad/second, 90° lag for $\omega > 20$ rad/second, and a linear variation between, with a lag of 45° at $\omega = 2$ rad/second, the corner frequency.
- (d) a similar curve to (c), but centred about $\omega = 10$ rad/second.

The true curves for phase vary from these straight line approximations by a maximum of $5\frac{1}{2}^{\circ}$, as described earlier in this section, and they can be drawn in by eye very easily.

Example 6.3. Draw a Bode diagram for the transfer function

$$G(s) = \frac{5}{(1+2s)(s^2+3s+25)}$$

This transfer function is made up of three components for which the Bode plots can readily be drawn:

(a) a constant gain term of $\frac{5}{25} = 0.2$. This contributes a constant magnitude of

 $20 \log_{10} 0.2 = -14 \text{ dB}$ at all frequencies, and has no effect on phase.

(b) a simple lag of time constant 2 second. This gives magnitude and phase contributions as in the previous example, but centred on a break point at 0.5 rad/second.

(c) a quadratic lag with $\omega_n = \sqrt{25} = 5$ and $\zeta = \frac{3}{2\omega_n} = 0.3$. The straight line

approximation is 0 dB to the corner frequency 5 rad/second, and falling at 40 dB per decade beyond this. With a value of $\zeta = 0.3$ the true curve peaks very close to 5 rad/second and can be drawn from Fig. 6.9. The phase curve passes steeply through 90° at $\omega = \omega_n = 5$ rad/second and can also be drawn from Fig. 6.9.

The overall magnitude and phase can now be obtained by addition, Fig. 6.11. It should be noted that the effect of the quadratic lag is felt at high frequencies where the amplitude has already been markedly attenuated by the simple lag. This is consistent with the fact that the poles at $-1.5 \pm j4.77$ are three times as far from the imaginary axis in the s-plane as the dominant root at -0.5 (see Section 4.3).

6.4 Frequency response testing of practical systems

Methods have been described in the earlier part of this chapter by means of which the frequency response information for a system can be calculated, provided that the transfer function is known. In the derivation of the transfer function from the basic physical laws certain assumptions have been made, and the validity of these should be confirmed by experimental testing of the actual system, provided that this is possible. With some physical components theoretical derivation of a transfer function is impossible and in such cases experimental testing can provide a means of determining a transfer function for the component. The form of the testing is dependent on various factors e.g.



Fig. 6.11 Bode plot for transfer function $G(s) = \frac{5}{(1+2s)(s^2+3s+25)}$

Can the system be tested off-line? What types of forcing function are physically realizable? How much disturbance to the system can be tolerated, and what is the noise level present? These aspects will be discussed further in the next chapter, particularly considerations of signal level and noise.

Clearly, provided that the input can be varied in a sinusoidal manner then the output or some variable representing it can be monitored, and its amplitude and phase measured for a range of frequencies. Any distortion of the waveform would suggest the presence of non-linearities in the system. The magnitude and phase information can be obtained in a number of ways, a convenient one being by using a transfer function analyser, a special purpose instrument which generates a sinusoidal forcing voltage for use as the system input, and which compares the system output voltage with this as reference to produce a direct readout of the harmonic information. Less accurate measurements can, however, be made in other ways not involving specialized equipment, such as by the use of an XY oscilloscope and a waveform generator. A sine wave from the signal generator is used to force the system input,

116 Frequency Response of Systems

the voltage from the transducer which measures the system output is applied to the Y input of the oscilloscope, and an auxiliary triangular waveform from the signal generator in phase with the forcing sine wave is applied to the X input of the oscilloscope. The amplitude and phase shift can be measured directly from the resulting trace.

The harmonic information would normally be presented on a polar or Bode plot. The polar plot gives by inspection a good idea of the likely form of the transfer function and, if the system approximates to one of first or second order, the values of τ or ζ and ω_n can be estimated. In principle, on the magnitude against frequency Bode plot it should be possible to draw straight lines whose slopes are multiples of 20 dB/decade and from the intersection of the asymptotes determine time constants; where complex lags are present ζ and ω_n can be estimated by comparison with standard curves. In practice, where time constants are close together it is difficult to decide where to draw asymptotes; hence prior knowledge of the theoretically expected form of the transfer function is of considerable help.

It will be clear from Section 6.3 that for a linear system there is a unique phase relationship associated with any given magnitude relationship. Study of the phase curve can assist significantly in the attempt to fit asymptotes to the magnitude curve with a view to determining factors of the transfer function. Lack of correspondence between experimentally obtained magnitude and phase curves would suggest the existence within the system of some non-linear effect. Any non-linearity increases the phase lag from that expected for a given magnitude curve, and is thus frequently referred to as a *non-minimum phase* effect.

Once an approximate transfer function has been determined an estimate of its accuracy may be made by evaluating its frequency response and comparing with the experimental data. A common criterion for 'goodness of fit' is the integral squared error

$$\frac{1}{\pi}\int\limits_{0}^{\infty}|G_{\rm cale}({\rm j}\omega)-G_{\rm exp}({\rm j}\omega)|^2\;{\rm d}\omega$$

At this stage of curve fitting a digital computer program can clearly be of great assistance when attempting to minimize this function.

Certain difficulties arise in practice indicating that harmonic testing has distinct limitations. If the input variable is not a voltage, current, position, or other easily controlled variable then variation of the input in a sinusoidal fashion of known amplitude may be very difficult or impossible. If the system is relatively slow, a long period of testing is required since each frequency in a wide range must be used in turn, and the system allowed to settle to a steady state each time before amplitude and phase measurements can be averaged over a number of cycles. The system characteristics may well change during the test period, especially when considering process or boiler plant, where ambient conditions have a significant influence. Finally, for perturbations of an amplitude that are relevant and tolerable, the response of the system may be masked by uncontrollable disturbances collectively referred to as *noise*.

6.5 Frequency domain performance criteria

Having described how to determine the harmonic characteristics of a system of known transfer function, or of a physical system which can be practically tested, it is necessary to decide what form of response is likely to be most acceptable. Consider the case where the system in question is a positional control, a measuring system, or other system where the output is intended to be equal to the input, or directly proportional to it. Ideally, the overall magnitude which in such cases is generally referred to as the *magnification* would be specified to be unity (or a constant) for all frequencies from zero to the maximum frequency component of interest in the input signal, and zero for higher frequencies which can be thought of as unwanted noise. Ideally also there should be no phase shift for the frequencies of interest. The input signal of interest would then be handled in an undistorted way and any noise at higher frequencies would be filtered out.



Such an ideal characteristic cannot be achieved in practice, but the form of curve which is typical is shown in Fig. 6.12 where the overall magnification M

Fig. 6.12 Typical frequency response characteristics for unity gain systems

has been normalized to unity for low frequencies. It can be seen that there is no sharp cut-off. The response characteristics are often described by the following parameters:

(a) bandwidth—defined as the frequency beyond which the magnification drops more than 3 dB from the low frequency value; i.e. below 0.707 for the unity gain system. All frequency components of interest should lie within the bandwidth.

118 Frequency Response of Systems

- (b) peak magnification M_p —the height of the peak, which should ideally be in the range 1.1 to 1.5 for good transient behaviour. For an overdamped system there will be no peak.
- (c) ω_p —the frequency at which the peak magnification occurs. In the case of an underdamped second order system this is close to the undamped natural frequency.
- (d) cut-off rate—the rate at which the magnification curve falls beyond the peak, and is thus a measure of the selectiveness of the filter characteristics.

For accurate measurement, say recording of a dynamic trace, it is important that the gain and phase are as close as possible to a constant and zero respectively for the frequency range being covered. Transducers not infrequently have very low damping ratio and thus a very large value of M_p ; in this case the flat region must be used otherwise there is likely to be severe distortion of the signal. In such instances the transducer chosen should have a value of ω_p which is some ten times larger than the maximum frequency of interest.

7 Statistical Methods for System Identification

System identification is the process of determining by means of practical testing the transfer function or some equivalent mathematical description for the dynamic characteristics of a system component. This of necessity requires the application to the component of some specific input signal since it relies on the analysis of input and output signals to identify the relationship between them. Traditional experimental procedures involve subjecting the system to step, ramp, pulse or sinusoidal input variations, and then carrying out relatively simple analysis of the output response curves, as outlined in Chapters 4 and 6. The advantages of these test inputs, which are typical of many normally occurring system inputs, have already been described from the point of view of the relative ease of signal generation and ease of analysis, and the physical understanding of system response which results. Unfortunately, response testing with these input functions is not always practical because of limitations imposed by the existence of system noise. Consider a component of process plant whose transfer function is to be determined, and which normally operates in a nominally steady state condition. With such a practical system component the output often varies randomly with time, even with a constant input, this variation arising from disturbances to the system component and being referred to as noise. The amplitude of the forcing functions when response testing must then be large enough to avoid the resulting output response being swamped by the noise signal, and this often requires input signals much larger than the normally occurring input variations. The results may not then be representative, as they are likely to include non-linearities and make small perturbation analysis invalid. Also, for large plant components the input variations would often need to be larger than the plant or the management can tolerate, so these methods find somewhat limited application where the component can be isolated for off-line testing.

This chapter describes a statistical method of identification applicable to the determination of transfer functions by on-line testing during normal plant operation with minimum disturbance to that operation. The method uses a non-deterministic forcing function which has random characteristics, unlike the above deterministic signals which are explicit functions of time, and which must thus be described by means of an appropriate statistical function. It can be considered as a wanted noise signal, and its amplitude can be small enough, if necessary, for it to be almost indistinguishable from the normal input signal. The effect of this forcing function on the output is not obviously noticeable from the plant operating records since it is buried in the inherent natural

120 Statistical Methods for System Identification

noise. Nevertheless, when the appropriate statistical procedure is used to analyse the signals, the effect of the inherent noise is largely eliminated and the response characteristics can be determined. The penalty of using a small amplitude forcing signal is that it must be applied for a long time period, of the order of 10 to 100 times the largest time constant of the plant but, because the effect of the test disturbance is not obviously detectable in the output, normal operation need not be interrupted. Occasionally it may be necessary to try to extract information from the inherent noise on the input and output signals, but the bandwidth of the input noise is seldom wide enough to yield adequate information by this means.

The primary purpose of this chapter is to describe this method and its advantages. As a background to understanding the method it also introduces the general ideas of correlation and spectral analysis, mathematical techniques employed widely in the handling of non-deterministic signals. The chapter can be omitted at a first reading since an understanding of the contents is not essential to the later chapters of the book.

The first section introduces the concept of correlation and defines the autocorrelation function, one of the ways available for describing a nondeterministic signal which is of particular use to the engineer. The autocorrelation functions for certain signals are given to illustrate the forms they can take; their main properties are also outlined. The section concludes by defining the cross correlation function which is used to describe the dependence of one signal upon another; in system identification these signals are the system input and output responses. Section 7.2 describes how correlation can be used to determine the impulse response of a practical system when the forcing function is chosen to be a signal which is equivalent to an ideal random signal known as white noise. Section 7.3 defines and gives examples of power spectral density, the frequency domain description of a signal and Section 7.4 discusses the determination of the harmonic response of a system by means of spectral analysis. This explanation of correlation functions and power spectral density functions and their use is followed by guidance on how they can be estimated for a given continuous signal by sampling and digital computation. Section 7.6 describes the pseudo random binary sequence, or PRBS signal, which is the forcing function most frequently used for the statistical testing of systems. The chapter concludes by illustrating graphically the forms of the various functions involved in the identification of a noisy second order system.

7.1 Correlation functions

A non-deterministic signal cannot be defined by means of an explicit function of time but must instead be described in some probabilistic manner. When undertaking system identification with non-deterministic forcing functions and carrying out the analysis in the time domain the appropriate statistical descriptions for the signals are termed correlation functions. The concept of *correlation* is a familiar one, there being for example an obvious correlation between a flash of lightning and the thunder which follows, and between the depth of water in a river or reservoir and the variation over a period of time of the rate of rainfall. Mathematically, the correlation of two random variables is the expected value of their product; it shows whether one variable depends in any way on the other. If the variables have non-zero means, it is sometimes preferable to subtract the mean values before determining the correlation; the result is usually known as the *covariance*, by analogy to the variance of a single variable. If the two variables are values of a random signal at two different time instants then the expected value of the product depends on how rapidly the signal can change. A high correlation might be expected when the two time instants are very close together, but much less correlation when the time instants are widely separated. It thus becomes appropriate to define a *correlation function* or *covariance function* in which the independent variable is the time separation of the two random variables. If the random variables come from the same signal the function is called an *autocorrelation function*, if from different signals a *cross correlation function*.

The autocorrelation function (or ACF) of a signal x(t) is given the symbol $\varphi_{xx}(\tau)$ and is defined as

$$\varphi_{xx}(\tau) = \lim_{T \to \infty} \frac{1}{2T} \int_{-\tau}^{T} x(t)x(t+\tau) dt \qquad 7.1$$

$$\varphi_{xx}(\tau) = \lim_{T \to \infty} \frac{1}{2T} \int_{-T}^{T} x(t-\tau)x(t) dt \qquad 7.2$$

or

i.e. it is the time average of the product of the values of the function τ seconds apart as τ is allowed to vary from zero to some large value, the averaging being carried out over a long period 2*T*. The process is shown graphically in Fig. 7.1 and consists of operations of displacement of the signal x(t) through a



Fig. 7.1 Calculation of ACF of x(t) for one value of time shift

122 Statistical Methods for System Identification

time τ , multiplication of x(t) by $x(t + \tau)$, integration of the product, and division by the integration time to give the value of $\varphi_{xx}(\tau)$ for that single value of τ . This process is repeated for other values of τ to yield φ_{xx} as a function of τ . The autocorrelation function is a measure of the predictability of the signal at some future time based on knowledge of the present value of the signal. If the value τ seconds from now is closely dependent on the present value then $\varphi_{xx}(\tau)$ will generally be large.

Consider first the forms which the autocorrelation function could have by evaluating it for certain signals which are functions of time.

Example 7.1. Sine wave. Consider the deterministic signal $f_1(t) = A \sin(\omega t + \psi)$, a sine wave with amplitude A, frequency ω , and phase ψ relative to the zero time datum (Fig. 7.2).



Fig. 7.2 Sine wave and its ACF

Using the above definition (Eq. 7.1) the ACF is given by

$$\varphi_{11}(\tau) = \lim_{T \to \infty} \frac{1}{2T} \int_{-T}^{T} A \sin(\omega t + \psi) A \sin(\omega t + \omega \tau + \psi) dt$$

Since the signal is periodic it is theoretically only necessary to average over one period, the time $\frac{2\pi}{\omega}$, where ω is the frequency in rad/second.

$$\therefore \quad \varphi_{11}(\tau) = \frac{\omega}{2\pi} A^2 \int_{0}^{2\pi/\omega} \sin(\omega t + \psi) \sin(\omega t + \omega \tau + \psi) dt$$

This integration can be carried out most conveniently by means of a change of variable.

Let
$$u = \omega t + \psi$$

$$\therefore \quad du = \omega \, dt \text{ or } dt = \frac{du}{\omega}, \text{ since } \psi \text{ is a constant.}$$

$$\varphi_{11}(\tau) = \frac{A^2}{2\pi} \int_{\psi}^{2\pi + \psi} \sin u \sin (u + \omega \tau) \, du$$

$$= \frac{A^2}{2\pi} \int_{\psi}^{2\pi + \psi} \left[\sin^2 u \cos \omega \tau + \sin u \cos u \sin \omega \tau \right] \, du$$

$$= \frac{A^2}{2\pi} \int_{\psi}^{2\pi + \psi} \left[\cos \omega \tau \left(\frac{1 - \cos 2u}{2} \right) + \sin \omega \tau \left(\frac{\sin 2u}{2} \right) \right] \, du$$

$$= \frac{A^2}{2\pi} \left[\cos \omega \tau \left(\frac{u}{2} - \frac{\sin 2u}{4} \right) + \sin \omega \tau \left(-\frac{\cos 2u}{4} \right) \right]_{\psi}^{2\pi + \psi}$$

$$\varphi_{11}(\tau) = \frac{A^2}{2\pi} \cos \omega \tau$$
7.3

Although a sine wave is not a random function many non-deterministic signals contain sinusoidal components and it is important to know what their effect would be on the overall autocorrelation function. This will be discussed later in this section. Note that the amplitude A and frequency ω appear in the autocorrelation function, but the phase angle ψ is absent, so movement of the sine wave function relative to the time axis has no effect on the autocorrelation function.

Example 7.2. Random binary function. Consider now a signal $f_2(t)$ which has the general form shown in the upper part of Fig. 7.3. (It will be seen in Section 7.6 that a deterministic and hence repeatable function of this form, termed a pseudo random binary sequence, is a very suitable input disturbance for the statistical testing of systems.) The signal has zero mean, only two possible values $\pm a$, and is able to change from one to the other only every Δt seconds, there being an equal probability of the signal being +a or -a in each interval Δt . What is the autocorrelation function of this signal?

Let the function $f_2(t)$ be shifted τ seconds along the time axis, and evaluate the ACF by determining the time average of the product of $f_2(t)$ and $f_2(t + \tau)$ as a function of τ . When $|\tau| > \Delta t$, then any time instant t_1 and the subsequent $t_1 + \tau$ cannot lie in the same Δt interval, hence $f_2(t_1)$ and $f_2(t_1 + \tau)$ are statistically independent; the value of the product is thus equally likely to be $+a^2$ or $-a^2$, and the time average is zero. When $|\tau| < \Delta t$, then the probability that



Fig. 7.3 Random binary function and its ACF

 t_1 and $t_1 + \tau$ lie in the same interval is $\frac{\Delta t - |\tau|}{\Delta t}$, i.e. 1 for $\tau = 0$, 0 for $\tau = \pm \Delta t$, and a linear variation between. When they are in the same interval the product $f_2(t_1)f_2(t_1 + \tau)$ is $+a^2$, when they are not in the same interval the average value of the product is again zero.

Hence the ACF is given by

$$\varphi_{22}(\tau) = \begin{cases} a^2 \left(1 - \frac{|\tau|}{\Delta t} \right) & \text{for } |\tau| \le \Delta t \\ 0 & \text{for } |\tau| > \Delta t \end{cases}$$
 7.4

This same autocorrelation function results also from certain other random time functions, such as one where the magnitude can take a random value to be held constant for each of the time intervals Δt .

Example 7.3. White noise. This is a completely random signal with defined properties which will be described in terms of its frequency characteristics in Section 7.3. Let it be referred to as the signal $f_3(t)$.

The correlation is zero for all time shifts τ except $\tau=0.$ For $\tau=0$

$$\varphi_{33}(\tau) = \lim_{T \to \infty} \frac{1}{2T} \int_{-T}^{T} f_3^{2}(t) dt = \text{mean square value} \qquad 7.5$$

The autocorrelation function thus consists of an impulse of magnitude equal to the mean square value occurring at $\tau = 0$, and is zero elsewhere. It can be seen that such a function results from letting $\Delta t \rightarrow 0$ in Example 7.2.

These examples illustrate certain important properties of autocorrelation functions and, to give a more complete understanding of the nature of the autocorrelation function $\varphi_{xx}(\tau)$ of a signal x(t), some of these properties are

outlined below.

- (i) the ACF is an even function of τ , i.e. $\varphi_{xx}(\tau) = \varphi_{xx}(-\tau)$, because the same set of product values is averaged regardless of the direction of translation in time.
- (ii) $\varphi_{xx}(0)$ is the mean square value, or average power of x(t).
- (iii) $\varphi_{xx}(0)$ is the largest value of the ACF, but if x(t) is periodic, then $\varphi_{xx}(\tau)$ will have the same maximum value when τ is an integer multiple of the period.
- (iv) if x(t) has a d.c. component or mean value, then $\varphi_{xx}(\tau)$ also has a d.c. component, the square of the mean value.
- (v) if x(t) has a periodic component, then $\varphi_{xx}(\tau)$ also has a component with the same period, but with a distorted shape resulting from the lack of discrimination between differing phase relationships of the constituent sinusoidal components.
- (vi) if x(t) has only random components, $\varphi_{xx}(\tau) \rightarrow 0$ as $\tau \rightarrow \infty$.
- (vii) a given ACF may correspond to many time functions, but any one time function has only one ACF.

A consequence of (iv), (v), and (vi) is that examination of the autocorrelation function for large values of τ shows whether any d.c. level or any periodic component is present in the signal.

Frequently there exist two signals x(t) and y(t) which are not completely independent (in system identification these would be an input and an output); a measure of the dependence of one signal on the other is given by the cross correlation function (CCF) which is defined as

$$\varphi_{xy}(\tau) = \lim_{T \to \infty} \frac{1}{2T} \int_{-\tau}^{T} x(t)y(t+\tau) dt$$
 7.6

$$\rho_{yx}(\tau) = \frac{\lim}{T \to \infty} \frac{1}{2T} \int_{-T}^{T} y(t)x(t+\tau) dt \qquad 7.7$$

or

Two analytical functions are defined, since time shifting y(t) yields a different result to that obtained by shifting x(t). If the two signals are from independent sources, and if they have zero mean values, then the cross correlation function is zero and the signals are said to be uncorrelated. If the mean values \bar{x} and \bar{y} are both non-zero, then the cross correlation function has a d.c. value equal to the product $\bar{x} \ \bar{y}$. The CCF is not an even function, though there is a type of symmetry because $\varphi_{xy}(\tau) = \varphi_{yx}(-\tau)$, since shifting one function forwards gives the same result as shifting the other backwards. $\varphi_{xy}(0)$ and $\varphi_{yx}(0)$ have no particular significance and the CCF generally does not have a maximum at $\tau = 0$. Examples of cross correlation functions are given in Fig. 7.20.

The ACF of the sum of two signals, say a sine wave plus noise, can be

126 Statistical Methods for System Identification

expressed in terms of the ACF and CCF of the individual signals thus:

$$\begin{split} \varphi_{(x+y)(x+y)}(\tau) &= \lim_{T \to \infty} \frac{1}{2T} \int_{-T}^{T} [x(t) + y(t)] [x(t+\tau) + y(t+\tau)] dt \\ &= \lim_{T \to \infty} \frac{1}{2T} \int_{-T}^{T} [x(t)x(t+\tau) + y(t)y(t+\tau) + x(t)y(t+\tau) + y(t)x(t+\tau)] dt \\ &\therefore \quad \varphi_{(x+y)(x+y)}(\tau) = \varphi_{xx}(\tau) + \varphi_{yy}(\tau) + \varphi_{xy}(\tau) + \varphi_{yx}(\tau) \quad 7.8 \end{split}$$

Provided that x(t) and y(t) are uncorrelated then the third and fourth terms each have the value $\bar{x} \bar{y}$, and hence

$$\varphi_{ix+yyx+y}(\tau) = \varphi_{xx}(\tau) + \varphi_{yy}(\tau) + 2\bar{x}\,\bar{y}$$
7.9

A noisy sine wave would thus have an ACF as in Fig. 7.2 but rising to a peak at $\tau > 0$.

The general nature of correlation functions has been explained in this section, together with the principle of evaluation using the three operations of time shifting, multiplication, and averaging. Evaluation of the correlation functions of continuous signals by analogue methods using electronic circuits, tape recorders and analogue computers has almost completely been superseded by digital methods operating on discrete data representations of the signals. Digital evaluation of correlation functions is described in Section 7.5. This principle can perhaps be best understood by hand calculation for a signal with a small number of sample values, though normally the equations are programmed for digital computer solution.

7.2 Dynamic testing using correlation techniques

First the analytical background will be considered and then the method of implementation. Assume the system to be tested is as in Fig. 7.4a with transfer function G(s), and impulse response or weighting function w(t). It has been shown in Section 4.7 that for an input x(t) the output y(t) is given by the convolution integral

$$y(t) = \int_{0}^{T} w(\lambda)x(t - \lambda) d\lambda$$
$$y(t) = \int_{-\infty}^{\infty} w(\lambda)x(t - \lambda) d\lambda$$

7.10

OT.



Fig. 7.4 Single-input-single-output system (a) without noise (b) with noise

Also, the cross correlation function between x(t) and y(t) is given (Eq. 7.6) by

$$\varphi_{xy}(\tau) = \lim_{T \to \infty} \frac{1}{2T} \int_{-T}^{T} x(t)y(t+\tau) dt$$
$$\varphi_{xy}(\tau) = \lim_{T \to \infty} \frac{1}{2T} \int_{-T}^{T} x(t) \left(\int_{-T}^{\tau} w(\lambda)x(t+\tau-\lambda) d\lambda \right) dt$$

Interchanging the order of integration gives

12

14.14

$$\varphi_{xy}(\tau) = \int_{-\infty}^{\infty} w(\lambda) \left(\lim_{T \to \infty} \frac{1}{2T} \int_{-T}^{T} x(t) x(t + \tau - \lambda) dt \right) d\lambda$$
$$\varphi_{xy}(\tau) = \int_{-\infty}^{\infty} w(\lambda) \varphi_{xx}(\tau - \lambda) d\lambda$$
7,11

It should be noted that this equation is very similar to Eq. 7.10 above. Comparing the two, it can be seen that if a signal x(t) whose autocorrelation function is $\varphi_{xx}(\tau)$ is applied to a system with weighting function w(t), then the cross correlation function of the input and output signals is equivalent to the time response of the system when subjected to an input signal $\varphi_{xx}(\tau)$. In particular if the input signal x(t) is chosen to be the idealized signal called white noise then, as has been seen in Example 7.3, the autocorrelation function $\varphi_{xx}(\tau)$ is an impulse; hence $\varphi_{xy}(\tau)$ is proportional to the impulse response or weighting function. This is the basis for statistical testing methods. A true white noise signal is not physically realizable but approximations to it, in particular a form of the random binary function of Example 7.2 (to be

128 Statistical Methods for System Identification

described more fully in Section 7.6), can be used as system forcing functions and can yield close approximations to the system impulse response.

This testing technique is at its most useful where noise is present in the system output, since the process of cross correlation extracts from the output only that part which is correlated with the test input. The noise disturbance n(t) is conventionally represented as an additive signal as shown in Fig. 7.4b; c(t) is assumed to be the system output which would exist in the absence of this noise. The cross correlation function between x(t) and y(t) is given as above by

$$\varphi_{xy}(\tau) = \frac{\lim}{T \to \infty} \frac{1}{2T} \int_{-T}^{\infty} x(t)y(t+\tau) dt$$

y(t) = c(t) + n(t)

But

$$\varphi_{xy}(\tau) = \lim_{T \to \infty} \frac{1}{2T} \int_{-T}^{T} x(t)c(t+\tau) dt + \lim_{T \to \infty} \frac{1}{2T} \int_{-T}^{T} x(t)n(t+\tau) dt$$
$$= \varphi_{xe}(\tau) + \varphi_{xn}(\tau)$$

Normally, the noise disturbance is not correlated with the test input signal, hence $\varphi_{xn}(\tau) = 0$ and then

$$\varphi_{xy}(\tau) = \varphi_{xc}(\tau)$$

This relationship shows that the cross correlation of the output and input signals is the same whether or not there is any noise present, provided the noise is uncorrelated.

The test signal, which for convenience is usually referred to as white noise, although not physically realizable and hence in practice replaced by an approximation to white noise, is added to the normal system input and cross correlation is carried out between the system output and the test signal. The configuration required for obtaining one point on the impulse response curve by analogue methods is shown in Fig. 7.5. The output y(t) of the system under test, corrupted by additive noise, is multiplied by $x(t + \tau_1)$, the test signal delayed τ_1 seconds. The multiplier output $z_1(t)$ is then integrated for a given period of time to obtain an estimate of $w(\tau_1)$, the value of the impulse response for time τ_1 . When the system component has a large bandwidth the integrator may be replaced by an averaging filter. Additional points can be obtained by adding more delay-multiplier-integrator paths and/or repeating the testing with different values of time delay. It is much more common now to use a digital computer either on-line or off-line to carry out the correlation, and in either case the signals x(t) and y(t) must be sampled to obtain time series representations as input data for the correlation program. If a special purpose correlator is available the signals are fed directly to the instrument and the resulting correlation functions are usually displayed on a screen; a choice of algorithm for averaging is common, whereby the picture may gradually be built up over a selected time period, or an approximation may appear almost immediately and then be gradually improved by continual updating as





more data becomes available. Typical traces obtained from a correlator are shown in Figs. 7.19 and 7.20.

A disadvantage of using a white noise signal is that a long averaging time is required to obtain an accurate and consistent estimate of its autocorrelation function, necessary to ensure that changes in the autocorrelation function of the input are not affecting the cross correlation function; also, if the period of testing is too long, the system characteristics may not remain constant. It is thus generally preferable to attempt to generate a white noise signal which is periodic and which has an autocorrelation function as shown in Fig. 7.6, in





which case the autocorrelation function can theoretically be computed to its full accuracy by correlation over one period of the input signal. With a noise free system the cross correlation function and hence impulse response could then theoretically be obtained to its full accuracy by correlation for one period. In a real system where noise is present correlation must be carried out for an integral number of periods of the input signal; the smaller the signalnoise ratio the longer the correlation time required to integrate out the inherent system noise. Results can be obtained with signal-noise ratios significantly less than unity.

130 Statistical Methods for System Identification

7.3 Power spectral density

The autocorrelation function describes the statistical properties of a signal in the time domain but for many purposes it is convenient to describe the signal in terms of frequency domain characteristics. This particularly applies where artificial inputs are not possible and where information must be obtained from naturally occurring signals. The function used is the *power density spectrum* or *power spectral density* $\Phi_{xx}(\omega)$ which is the Fourier transform of the autocorrelation function:

$$\Phi_{xx}(\omega) = \int_{-\infty}^{\infty} \varphi_{xx}(\tau) e^{-j\omega\tau} d\tau \qquad 7.12$$

The transformation can be seen to be similar to Laplace transformation, though in this case the real part of the power of e is zero and the integration extends from $-\infty$ to $+\infty$.

Now

$$e^{-j\omega\tau} = \cos \omega\tau - j \sin \omega\tau$$
$$\Phi_{xx}(\omega) = \int_{-\infty}^{\infty} \varphi_{xx}(\tau)(\cos \omega\tau - j \sin \omega\tau) d\tau$$

Also, $\varphi_{xx}(\tau)$ has been shown to be always an even function, hence the product $\varphi_{xx}(\tau) \sin \omega \tau$ integrates to zero

$$\Phi_{xx}(\omega) = \int_{-\infty}^{\infty} \varphi_{xx}(\tau) \cos \omega \tau \, d\tau \qquad 7.13$$

It can be seen that the power spectral density is a real and even function. Applying the inverse Fourier transformation gives

$$\varphi_{xx}(\tau) = \frac{1}{2\pi} \int_{-\infty}^{\infty} \Phi_{xx}(\omega) \mathrm{e}^{\mathrm{j}\omega\tau} \,\mathrm{d}\omega$$

or, since $\Phi_{xx}(\omega)$ has in Eq. 7.13 been shown to be an even function of ω ,

$$\varphi_{xx}(\tau) = \frac{1}{2\pi} \int_{-\infty}^{\infty} \Phi_{xx}(\omega) \cos \omega \tau \, d\omega \qquad 7.14$$

Letting $\tau = 0$ gives

$$\varphi_{xx}(0) = \frac{1}{2\pi} \int_{-\infty}^{\infty} \Phi_{xx}(\omega) \, \mathrm{d}\omega = \lim_{T \to \infty} \frac{1}{2T} \int_{-\tau}^{T} x^2(t) \, \mathrm{d}t = \overline{x^2(t)}$$

If x(t) were a voltage or current for a 1 ohm load, then the mean square value
would be the mean power taken by the load; by borrowing this terminology $\varphi_{xx}(0)$ is termed the mean power of the signal x(t).

Hence
$$\int \Phi_{xx}(\omega) d\omega = 2\pi \times \text{mean power}$$
 7.15

and the power spectral density is a measure of the energy distribution of the signal within the frequency spectrum. It is never negative, and extends over a frequency range from $-\infty$ to $+\infty$, thus being defined also for a hypothetical negative frequency region.

As with the ACF, the relative phase of the various frequency components is lost, and a given power density spectrum can correspond to a large number of different time functions. If the input signal has a periodic component such that the Fourier series for this component contains terms at frequencies $\omega_1, \omega_2, ...$ then $\Phi_{xx}(\omega)$ will have discrete values at $\pm \omega_1, \pm \omega_2, ...$

The cross power spectral density $\Phi_{xy}(\omega)$ bears the same transform relationship to the cross correlation function $\varphi_{xy}(\tau)$ as does the power spectral density to the autocorrelation function.

Example 7.4. White noise. This is defined as a signal having uniform power content at all frequencies

i.e.
$$\Phi_{11}(\omega) = \text{constant}$$
, for $-\infty < \omega < \infty$

The ACF, given by the inverse transform of a constant, is an impulse at $\tau = 0$. It can now be seen why this is only a convenient theoretical function, since it implies that the mean power (Eq. 7.15) is infinite.

Example 7.5. Band limited white noise. Here, the power spectral density is constant for all frequencies less than some given frequency ω_1 , and zero for all higher frequencies.

e.
$$\Phi_{11}(\omega) = \begin{cases} A \text{ for } -\omega_1 \le \omega \le \omega_1 \\ 0 \text{ for } \omega > \omega_1 \text{ and } \omega < -\omega_1 \end{cases}$$

Hence

$$\varphi_{11}(\tau) = \frac{1}{2\pi} \int_{-\infty}^{\infty} \Phi_{11}(\omega) \cos \omega \tau \, d\omega$$
$$= \frac{A}{\pi} \int_{0}^{\omega_{1}} \cos \omega \tau \, d\omega$$
$$= \frac{A}{\pi} \frac{\sin \omega_{1} \tau}{\omega_{1}}$$

τ

π

This has the form shown in Fig. 7.7; it can be seen that as $\omega_1 \rightarrow \infty$ the ACF will tend towards an impulse. Provided that ω_1 is an order of magnitude larger than the bandwidth of a system being tested, then such a signal is an adequate approximation to true white noise.



Fig. 7.7 Power spectral density and ACF of a band limited white noise signal $f_1(t)$

Example 7.6. Random binary function. Consider the random binary function of Example 7.2.

$$\varphi_{22}(\tau) = \begin{cases} a^2 \left(1 - \frac{|\tau|}{\Delta t}\right) \text{ for } |\tau| < \Delta t \\ 0 & \text{ for } |\tau| \ge \Delta t \end{cases}$$
$$\Phi_{22}(\omega) = \int_{-\infty}^{\infty} \varphi_{22}(\tau) \cos \omega \tau \, d\tau$$
$$\Phi_{22}(\omega) = \int_{-\Delta t}^{\Delta t} a^2 \left(1 - \frac{|\tau|}{\Delta t}\right) \cos \omega \tau \, d\tau$$
$$= 2a^2 \int_{0}^{\Delta t} \left(\cos \omega \tau - \frac{\tau \cos \omega \tau}{\Delta \tau}\right) d\tau$$

Now

$$= 2a^{2} \left[\frac{\sin \omega \tau}{\omega} - \frac{1}{\Delta t} \left(\frac{\tau \sin \omega \tau}{\omega} + \frac{\cos \omega \tau}{\omega^{2}} \right) \right]_{0}^{\Delta t}$$
$$= \frac{2a^{2}}{\omega^{2} \Delta t} \left(1 - \cos \omega \Delta t \right)$$
$$= a^{2} \Delta t \left(\frac{\sin \omega \Delta t/2}{\omega \Delta t/2} \right)^{2}$$

The form of this is shown in Fig. 7.8.



Fig. 7.8 ACF and power spectral density of random binary function

Fourier analysis allows any continuous signal x(t) which is periodic with period T to be represented by the summation of a number of sine and cosine waves of varying amplitudes with frequencies ω , 2ω , 3ω , ..., $n\omega$ where $\omega = 2\pi/T$ radians per second,

i.e.
$$x(t) = a_0 + a_1 \cos \omega t + a_2 \cos 2\omega t + a_3 \cos 3\omega t$$

 $+ \ldots + a_n \cos n\omega t + b_1 \sin \omega t + b_2 \sin 2\omega t + \ldots + b_n \sin n\omega t$

By combining cosine and sine terms of like frequency the Fourier series representation of x(t) can be written as

$$\begin{aligned} \mathbf{x}(t) &= a_0 + c_1 \cos\left(\omega t + \varphi_1\right) + c_2 \cos\left(2\omega t + \varphi_2\right) \\ &+ c_2 \cos\left(3\omega t + \varphi_2\right) + \ldots + c_n \cos\left(n\omega t + \varphi_n\right) \end{aligned}$$

The signal can thus be seen to have a mean value a_0 , and to have its energy at specific discrete frequencies ω , 2ω , 3ω , ..., $n\omega$ (the fundamental frequency ω and integer multiples of it), the energy contained in the respective constituents being given by c_1^2 , c_2^2 , c_3^2 , ..., c_n^2 . The energy or power in the signal can be represented by lines of appropriate length spaced ω apart on a plot of power against frequency (as in Fig. 7.12 say). As the period T increases ω decreases and the discrete frequency values come closer together. In the limit for a non-periodic signal $(T \rightarrow \infty)$ the discrete line spectrum tends towards a continuous spectrum. This is the power density spectrum. The power spectral density for a signal can thus be evaluated by first obtaining the autocorrelation function and then carrying out Fourier transformation, as in the above definition, or directly from the signal itself by Fourier transformation of the signal.

The approach adopted for evaluating power spectral density is dictated by the nature of the signal of interest and the equipment available. If the signal is a continuous voltage waveform of large bandwidth, a good estimate of the power spectral density can be obtained by means of a wave analyser. This is an instrument which utilizes a tuneable bandpass filter and measures the signal power contained in narrow bands of frequency, usually scanning automatically through a wide frequency range. Many signals from industrial processes have a bandwidth which is low in comparison to the smallest frequency that can be discriminated, allowing only a very approximate estimate of the power spectral density curve to be obtained by this method. A more flexible approach to determination of spectral density is by digital computation, operating on a sampled representation of the signal as outlined in Section 7.5. Direct computation of the Fourier coefficients and hence power density spectrum of a signal by means of a digital computer involves the handling of a vast amount of data, and the computation time increases very rapidly as the number of sample values increases. Prior to the development of computer algorithms known as Fast Fourier Transforms the power spectral density function was evaluated from the autocorrelation function. Now, however, the spectrum is usually evaluated directly using one of these algorithms and their computational efficiency is such that the correlation function is often evaluated from the density spectrum rather than directly from the time series data.

7.4 System identification using spectral density functions

The preceding section has shown that the power spectral density is a function of frequency and that it describes quantitatively how the energy in a signal is distributed over the frequency spectrum. From Chapter 6 it should be clear that a system affects the magnitude and phase of input signal components of different frequencies by different amounts. The amplitude of the output of a system with transfer function G(s), for a frequency ω_1 , is the input amplitude multiplied by $|G(j\omega_1)|$, while the phase of the output relative to the input is $\angle G(j\omega_1)$. For a statistical input signal it should thus be possible to determine the transfer function from a knowledge of the power density spectra of the input and output signals and the appropriate cross spectra. Consider now the form of the relationships between these for the system of Fig. 7.4.

7.4 System identification using spectral density functions 135

The output of the system can be described by the convolution integral, Eq. 7.10, in the form

$$y(t) = \int_{-\infty}^{\infty} w(\lambda_1) x(t - \lambda_1) \, \mathrm{d}\lambda_1$$

and for time $t + \tau$ the output is

$$y(t+\tau) = \int_{-\infty}^{\infty} w(\dot{\lambda}_2) x(t+\tau-\dot{\lambda}_2) \, \mathrm{d}\dot{\lambda}_2$$

The dummy variable λ_2 has been introduced in place of λ_1 so that the two integrals can be kept distinct later in the analysis.

From the definition, Eq. 7.1, the autocorrelation function of the output is

$$\varphi_{yy}(\tau) = \frac{\lim}{T \to \infty} \frac{1}{2T} \int_{-T}^{T} y(t)y(t+\tau) dt$$

Substitution for y(t) and $y(t + \tau)$ from above gives

$$\varphi_{yy}(\tau) = \lim_{T \to \infty} \frac{1}{2T} \int_{-T}^{T} dt \int_{-\infty}^{\infty} w(\lambda_1) x(t - \lambda_1) d\lambda_1 \int_{-\infty}^{\tau} w(\lambda_2) x(t + \tau - \lambda_2) d\lambda_2$$

Interchanging the order of the limit process and the integration enables this to be written as

$$\varphi_{yy}(\tau) = \int_{-\infty}^{\infty} w(\lambda_1) \, d\lambda_1 \int_{-\infty}^{\infty} w(\lambda_2) \, d\lambda_2 \left\{ \lim_{T \to \infty} \frac{1}{2T} \int_{-T}^{T} x(t-\lambda_1)x(t+\tau-\lambda_2) \, dt \right\}$$

where the expression in brackets $\{ \}$ is, by definition, Eq. 7.1, the input autocorrelation function $\varphi_{xx}(\tau + \lambda_1 - \lambda_2)$

$$\therefore \quad \varphi_{yy}(\tau) = \int_{-\infty}^{\infty} w(\dot{\lambda}_1) \, d\dot{\lambda}_1 \int_{-\infty}^{\infty} \varphi_{xx}(\tau + \dot{\lambda}_1 - \dot{\lambda}_2) w(\dot{\lambda}_2) \, d\dot{\lambda}_2 \qquad 7.16$$

Now the power spectral density of the output signal is the Fourier transform of the autocorrelation function of the output, hence

$$\Phi_{yy}(\omega) = \int_{-\infty}^{\infty} \varphi_{yy}(\tau) e^{-j\omega\tau} d\tau$$
$$= \int_{-\infty}^{\infty} e^{-j\omega\tau} \left\{ \int_{-\infty}^{\infty} w(\lambda_1) d\lambda_1 \int_{-\infty}^{\infty} \varphi_{xx}(\tau + \lambda_1 - \lambda_2) w(\lambda_2) d\lambda_2 \right\} d\tau$$

$$= \int_{-\infty}^{\infty} w(\lambda_1) e^{j\omega\lambda_1} d\lambda_1 \int_{-\infty}^{\infty} w(\lambda_2) e^{-j\omega\lambda_2} d\lambda_2 \int_{-\infty}^{\infty} \varphi_{xx}(\tau + \lambda_1 - \lambda_2) \times e^{-j\omega(\tau + \lambda_1 - \lambda_2)} d\tau \qquad 7.17$$

From the Laplace transform relationship between w(t) and G(s)

$$G(s) = \int_{0}^{\infty} w(t) \mathrm{e}^{-st} \mathrm{d}t$$

and letting $s = j\omega$, since only frequency domain information is of interest

$$G(j\omega) = \int_{0}^{\infty} w(t) e^{-j\omega t} dt.$$

Hence, since w(t) = 0 for t < 0

$$G(j\omega) = \int_{-\infty}^{\infty} w(t) e^{-j\omega t} dt$$
 7.18

Now also

$$\int_{-\infty}^{\infty} \varphi_{xx}(\tau + \dot{\lambda}_1 - \dot{\lambda}_2) e^{-j\omega(\tau + \lambda_1 - \dot{\lambda}_2)} d\tau = \int_{-\infty}^{\infty} \varphi_{xx}(\tau) e^{-j\omega\tau} d\tau = \Phi_{xx}(\omega) \quad 7.19$$

Using Eq. 7.18 and Eq. 7.19 to simplify Eq. 7.17

$$\Phi_{yy}(\omega) = G(-j\omega)G(j\omega)\Phi_{xx}(\omega)$$

But $G(j\omega)$ is the conjugate of $G(-j\omega)$

$$\Phi_{ve}(\omega) = |G(j\omega)|^2 \Phi_{ve}(\omega)$$
7.20

Hence the output spectral density is obtained from the input spectral density by multiplying by the square of $|G(j\omega)|$. This is to be expected since the spectral density represents signal power, or amplitude squared. For system identification purposes, knowing $\Phi_{xx}(\omega)$ and $\Phi_{yy}(\omega)$, $|G(j\omega)|$ can be found, and thus the magnitude curve of the Bode plot can be drawn.

The derivation of the corresponding phase relationship is somewhat lengthy; hence it will be stated without proof:

$$\frac{\Phi_{xy}(\omega)}{\Phi_{yx}(\omega)} = \frac{G(j\omega)}{G(-j\omega)} = \exp\left(-j2 \angle G(j\omega)\right)$$
 7.21

Using this the phase curve of the Bode plot can be obtained.

In principle this is a useful method for determining experimentally the transfer function of a system component. It has the advantage that provided the normally occurring input signal has a wide enough frequency spectrum then

7.5 Digital evaluation of correlation functions and power spectral density 137

an estimate of the transfer function can be made using only the information contained in the normal operating data. The value of the method, however, decreases rapidly as the level of internally generated system noise increases. If this inherent noise n(t) affects the recorded system output as shown in Fig. 7.4b then the power spectrum of the noise contaminated output is that of the clean system output added to that of the noise; i.e.

$$\Phi_{\rm ev}(\omega) = \Phi_{\rm cr}(\omega) + \Phi_{\rm ev}(\omega)$$

which means that Eq. 7.20 must be written as

$$\Phi_{yy}(\omega) = |G(j\omega)|^2 \Phi_{xx}(\omega) + \Phi_{yy}(\omega)$$
 7.22

Unfortunately, the noise power spectrum $\Phi_{nn}(\omega)$ cannot be measured directly. However, an expression called the *coherence function* can be evaluated to estimate the degree of distortion of the clean output caused by the presence of noise. The coherence function is defined as

$$\gamma^{2}_{xy}(\omega) = \frac{|\Phi_{xy}(\omega)|^{2}}{\Phi_{xx}(\omega)\Phi_{yy}(\omega)}$$
7.23

and has a value of unity for a noise-free output. It is common practice to accept the data of the spectral analysis if the coherence function has a value greater than 0.8.

7.5 Digital evaluation of correlation functions and power spectral density

The present approach to signal analysis is to convert the continuous function x(t) to a discrete *time series representation* $x_1, x_2, x_3, \ldots, x_r, \ldots, x_N$ by sampling at regular intervals Δt (Fig. 7.9a) and then to obtain an estimate of the correlation function or spectral density by digital computation. The choice of the sampling frequency and the length of the sample record to be analysed must be made with care since poor selection of either adversely affects the accuracy of the results, and both directly influence the computation time required. Intuitively one can see that in Fig. 7.9b the sampling interval is too large, detail is missing and results will be lacking in reliable data for the high frequency content of the signal, and conversely in Fig. 7.9c no information would be lost and computation time decreased if the sampling interval were increased significantly.

For the samples to be truly representative of the original signal it must be possible to recreate the signal accurately from the samples. It is well known in communication theory that this is possible only when the sampling frequency is at least twice the highest frequency component present in the signal. To be able to determine the dynamic characteristics of a process it is necessary that the time series shall contain all components of the signal at frequencies up to one or two orders of magnitude greater than the reciprocal of the smallest significant time constant of the process. This will determine the value of the minimum acceptable sampling rate. In practice the signal from the process almost always contains components extending to much higher frequencies



Fig. 7.9 Sampling of a continuous function (a) derivation of time series representation (b) Δt too large (c) Δt unnecessarily small

than the wanted data because of noise pick-up, and this noise must be taken into consideration. It would be computationally wasteful to process the excessively large amount of data resulting from sampling at a rate sufficiently high to record the noise accurately. Merely reducing the sampling rate can give misleading results since the high frequency components then appear as though they are of lower frequency. This phenomenon, known as *aliasing*, is illustrated in Fig. 7.10, where a signal of constant magnitude but contaminated by noise at 50 Hz would appear to have a 2 Hz oscillation if sampled at 16 samples/ second. The normal way to avoid this is to use an analogue filter to remove





7.5 Digital evaluation of correlation functions and power spectral density 139

the high frequency noise before sampling. It is not possible to remove aliased components after sampling since they are indistinguishable from signal components at the aliased frequencies.

Consider a continuous signal x(t) which has been sampled at regular intervals Δt to yield a time series $x_1, x_2, x_3, \dots, x_N$ as in Fig. 7.9a. An *estimate of the autocorrelation function* of x(t) can be obtained from the N sample values by time shifting, multiplying and averaging as follows:

$$\varphi_{xx}(0) = \frac{1}{N} (x_1^2 + x_2^2 + x_3^2 + \dots + x_{N-1}^2 + x_N^2)$$

$$\varphi_{xx}(\Delta t) = \frac{1}{N-1} (x_1 x_2 + x_2 x_3 + x_3 x_4 + \dots + x_{N-2} x_{N-1} + x_{N-1} x_N)$$

$$\varphi_{xx}(2\Delta t) = \frac{1}{N-2} (x_1 x_3 + x_2 x_4 + x_3 x_5 + \dots + x_{N-3} x_{N-1} + x_{N-2} x_N)$$

etc.

etc

Values are estimated for $\tau = 0$, $\tau = \Delta t$, $\tau = 2\Delta t$, $\tau = 3\Delta t$ etc., but there is no information about intermediate values of τ . It is clear that N should be large and that the estimates of $\varphi_{xx}(\tau)$ can only be realistic if $\tau \ll N\Delta t$. It may be that the signal is periodic such that $x_{N+1} = x_1$, $x_{N+2} = x_2$, etc. in which case each value of $\varphi_{xx}(\tau)$ can be evaluated from N products rather than from a decreasing number of products as above, the computations then being of the form

$$\varphi_{xx}(2\Delta t) = \frac{1}{N} \left(x_1 x_3 + x_2 x_4 + \dots + x_{N-2} x_N + x_{N-1} x_1 + x_N x_2 \right)$$

and only one complete cycle need be considered. The resulting plot of $\varphi_{xx}(\tau)$ against τ is often referred to as a *correlogram*. In practice a smooth curve is often drawn through the points, or alternatively they are joined by straight lines, to highlight more clearly the general form of $\varphi_{xx}(\tau)$. For a signal which is not periodic an increase in the number of samples N should yield a better estimate of $\varphi_{xx}(\tau)$ at the expense of increased computation time. Increasing the number of samples N and simultaneously decreasing the sampling interval Δt gives a correlogram with more closely spaced points, which is desirable when the correlation function changes rapidly in magnitude from one value to the next.

As was stated in Section 7.3 the power spectral density of a signal is now usually determined digitally from a discrete time series representation of the signal, and generally by utilizing a Fast Fourier Transform (FFT) algorithm since it is very much faster than normal Fourier transformation on a digital computer. Spectral density plots for analogue signals can be most easily obtained if a proprietary spectrum analyser can be used, or if special software is available on a general purpose digital computer with analogue-to-digital converter. Failing this a program can be written incorporating the FFT algorithm from a software library, or the FFT routine can be obtained from a textbook. Any reader interested in the operation of the algorithm is referred to a textbook such as reference 29 in Appendix B which gives a clear description.

An appreciation of the artefacts associated with spectral density evaluation is essential, and the main object of the next few paragraphs is to explain some of these with the aid of simple pictorial representation so that spectral analysis can be undertaken sensibly and results interpreted correctly.

The FFT algorithm requires a number of samples which is a power of 2, i.e. $N = 2^n$ (say 2048, 4096, 8192, or 16384) and yields a discrete spectrum with 2^{n-1} frequency lines. The length of data record analysed is therefore $N\Delta t$, and spectral information is evaluated for the frequencies $1/N\Delta t$, $2/N\Delta t$, $3/N\Delta t$, ... $1/2\Delta t$, but not for any intermediate frequencies or higher frequencies. (Points can be joined by straight lines or by a smooth curve to highlight the trend.) The sample function of the signal or the autocorrelation function is thus truncated to a specific finite length prior to processing and this introduces some distortion in the spectral density curve estimated from the data. The signal x(t) to be transformed is viewed through a window or gate, normally a rectangular window function hence

$$x_T(t) = x(t)W(t) \quad \text{where} \quad W(t) = \begin{cases} 1 \text{ for } |t| \le \frac{T}{2} \\ 0 \text{ for } |t| > \frac{T}{2} \end{cases}$$

The window function and its effect on the signal x(t) are shown in Fig. 7.11a together with the Fourier transform of the window function

$$F(\omega) = T\left(\frac{\sin \omega T/2}{\omega T/2}\right)$$

The finite width of the window with its sharp edges and discontinuity between the ends introduces spurious harmonic information and thus a pure sine wave would be seen to have side lobes. The amplitude spectrum of a sine wave would then have the form shown in Fig. 7.11a instead of being a single line, and the power spectrum would be the square of this function.

Consider in a little more detail the effect of evaluating the power spectral density of a portion of sine wave using the FFT algorithm. If $N\Delta t$, the width of the window, corresponds to an integer number of cycles of the sine wave then the spectral line spacing is $2\pi/N\Delta t = 2\pi/T$ radians per second and one of these frequency values will correspond exactly with the frequency of the sine wave. In this case a single spectral line results, all other values being zero-if say there are 5 complete cycles then the 5th spectral line will have amplitude A, power A^2 and all others will be zero. Almost invariably, however, since N must be a power of 2 and Δt is a rounded time interval such as 10 ms or 0.05 ms, say, the product $N\Delta t$ will not correspond to an integer number of cycles. None of the frequencies for which spectral information is obtained will then coincide with the frequency of the sine wave, and the signal power will be shared across a number of frequencies, with the highest value corresponding to the frequency closest to that of the sine wave. Fig. 7.12 illustrates the effect. The shape and the peak amplitude change markedly with change in frequency. Note that an increase in N for the same Δt increases T and hence compresses



Fig. 7.11 Window function, its Fourier transform, and effect on spectral density of a signal (a) rectangular window (b) triangular (Bartlett) window (and Hanning window, shown dotted)

the effect to a narrower range of frequency but does not eliminate it. A square wave can be represented by the Fourier series

$$x(t) = \frac{4}{\pi} (\cos \omega t - \frac{1}{3} \cos 3\omega t + \frac{1}{5} \cos 5\omega t - \frac{1}{7} \cos 7\omega t + \dots)$$

Hence there should be spectral lines for odd multiples of the fundamental frequency, and the successive power spectral values will be in the ratios $1, \frac{1}{9}, \frac{1}{25}$,



Fig. 7.12 Line spectrum for truncated sine wave (5.3 cycles, unit amplitude)

 $\frac{1}{49}$, $\frac{1}{81}$, $\frac{1}{121}$ etc. or 0, -9.5, -14.0, -16.9, -19.1 ... on a decibel scale. If an integer number of cycles of the square wave fills the window $N\Delta t$ then these lines and only these will appear on the spectral plot. Normally, however, each shows the windowing effect (Fig. 7.13) in which case there is a leakage of power to adjacent frequencies.

If a signal contains frequency components which are close together then the spectra tend to merge blurring the distinction between them (Fig. 7.11a). The resolvability is improved with a wider window (longer length of sample function analysed). Other window shapes, such as the triangular or Bartlett window and the raised cosine Hanning window, which reduce the emphasis on parts of the function towards the ends are sometimes used especially where a smoother spectrum is desired. The Fourier transform of the Bartlett window,

$$F(\omega) = T\left(\frac{\sin \omega T/4}{\omega T/4}\right)^2$$

is shown with its effect on a pure sine wave in Fig. 7.11b. The rectangular window is simple but suffers from large amplitude side lobes and poor amplitude accuracy. At the expense of an increase in bandwidth both the Bartlett and the widely used Hanning window have smaller side lobes and better amplitude accuracy.

7.6 Pseudo random binary sequences (PRBS)

These are very useful approximations to periodic white noise and are the



Fig. 7.13 Discrete spectrum for truncated square wave (5.3 cycles, unit amplitude)

forcing functions most widely used in statistical system testing. Their general form is shown in Fig. 7.3 since they are special cases of the random binary function of Example 7.2. They are signals which can take on only two possible states, say +a and -a, the state can change only at discrete intervals of time Δt , the change occurs in a deterministic pseudo random manner, and the sequence is periodic with period $T = N\Delta t$ where N is an integer.

These deterministic repeatable signals all satisfy a set of 'conditions of randomness':

- (i) Balance property: in any period of the sequence the number of logic ones (+a) should not differ from the number of logic zeros (-a) by more than one.
- (ii) Run property: among the runs of one, two, three etc. in the period half should be of length 1, a quarter of length 2, an eighth of length 3 etc. and there should be as many of each run of logic one as logic zero state (these requirements cannot be satisfied exactly since sequence lengths are not powers of 2).
- (iii) Correlation property: if a period of the sequence is compared term by term with any cyclic shift of itself then the number of agreements and disagreements should not differ by more than one.

When these conditions are satisfied the sequence can be thought of as being as random as is possible for any given length.

The most commonly used type of PRBS is the maximum length sequence, which is of length $N = 2^n - 1$, where n is an integer (i.e. N = 15, 31, 63, 127, 255...). This can be generated by an n stage shift register with the first stage determined by feedback of the appropriate modulo two sum of the last stage and one or two earlier stages (Fig. 7.14). Modulo 2 addition is the logic function 'exclusive or' i.e. if the inputs are the same the output is logic 0; if the



Fig. 7.14 Generation of PRBS by shift register

inputs are different the output is logic 1; alternatively it can be thought of as binary addition where only the least significant digit is recorded. The logic contents of the shift register are moved one stage to the right every Δt seconds by simultaneous triggering by a clock pulse. All possible states of the shift register are passed through except that of all zeros. The output can be taken from any stage and is a serial sequence of logic states having cyclic period $N\Delta t$. If feedback is taken from the modulo 2 sum of the wrong register stages then the resulting cyclic sequence has a length less than the maximum length, and will not be suitable. The correct stages for the most commonly used lengths are:

п	4	5	6	7	8	9	
N	15	31	63	127	255	511	
Feedback from modulo two addition of stages	3 & 4	3845	586	4 & 7	2.3.4 & 8	5 & 9	

Delayed versions of the sequence can easily be obtained by modulo 2 addition of appropriate stages making use of a shift and add property which states that 'if a binary maximum length sequence is added modulo two to the same sequence delayed from the original then the resulting sequence is also a delayed version'.

Example 7.7. Consider as illustration of the above a 4 stage shift register with feedback from stages 3 & 4. Successive states of the shift register, starting all ones, are:

Stage

1	1	0	0	0	1	0	0	1	1	0	1	0	1	1	1	1	
2	1	1	0	0	0	1	0	0	1	1	0	1	0	1	1	1	and the pattern
3	1	1	1	0	0	0	1	0	0	1	1	0	I	0	1	1	repeats
4	1	1	1	1	0	0	0	ĩ	0	0	1	1	0	1	0	1	

Hence the sequence length is 15, which is $2^n - 1$ with n = 4. With feedback from stages 2 and 4 the states would be:

Stage

1	1	0	0	1	1	1	1	
2	1	1	0	0	1	1	1	and the nattern reneats after only 6 states
3	1	1	1	0	0	1	1	and the pattern repeats, after only o states.
4	1	1	1	1	0	0	-11	

Investigate now the three properties of randomness when applied to the full 15 bit sequence:

(1)	Balance property	£				
	Number of ones	= 8				
	Number of zeros	= 7	.:. di	fferen	ce = 1	
(2)	Run property:					
	Length of run	1	2	3	4	
	Number of runs	4	2	1	1	Total = 8
	Actual ratio	4/8	2/8	1/8	1/8	
	Ideal ratio	1/2	1/4	1/8	1/16	

There are equal numbers of runs of 1 and 0 except 3 and 4.

 (3) Correlation property: Compare stages 1 and 4, say, Number of agreements = 7 Number of disagreements = 8 ..., difference = 1.

If the output is from stage 1 then stages 2-4 give delays 1-3

 $(4 \oplus 1) \oplus 3$ gives 1 0 0 1 1 0 1 0 1 1 1 1 0 0 0 i.e. delay 11, etc.

There also exist other PRBS sequences e.g. quadratic residue codes which exist for N = 4k - 1 bit sequences, where k is an integer and N is a prime number (i.e. 11, 19, 23, 31, 43, 47, ...). These are difficult to generate using logic circuitry, but the sequences can be precomputed and read from paper tape, say, for slower applications. An advantage is that successive sequence lengths are much closer together.

Autocorrelation function and power spectral density of PRBS. Consider, first, values at $\tau = k\Delta t$ where k is an integer. Let values of the sequence for successive intervals Δt be $x(1), x(2), x(3), \dots x(N)$. The ACF is

$$\varphi_{xx}(k) = \frac{1}{N} \sum_{j=1}^{N} x(j)x(j+k)$$
$$= \frac{a^2}{N} \times (\text{Number of matching digits} - \text{number of differing digits})$$
$$\int_{0}^{\infty} a^2$$

7.24

$$\therefore \quad \varphi_{xx}(k) = \begin{cases} -\frac{a^2}{N} \text{ if } k \neq 0\\ +a^2 \text{ if } k = 0 \end{cases}$$

It can be shown by considering area changes that the ACF is linear between these points. Hence the form of the ACF is as shown in Fig. 7.15. As $\Delta t \rightarrow 0$ and N becomes large the ACF tends closer to that of true periodic white noise (Fig. 7.6).

The power spectral density, also shown in Fig. 7.15, differs from Fig. 7.8 in one respect—it is a line spectrum and not a continuous spectrum. This occurs



Fig. 7.15 Autocorrelation function and power spectral density of PRBS

because the lowest frequency component in the PRBS signal is that corresponding to the period, i.e. $2\pi/N\Delta t$ radians/second, and all other frequencies present are integer multiples of this value.

Advantages of PRBS and practical considerations

(a) The binary nature of the signal simplifies the cross correlation calculation since the multiplication can be replaced by simple gating of the output time function and its inverse.

(b) The binary signal is easy to generate and introduce into a system (using say a solenoid), and the constant Δt avoids the distortion which can occur with attempting too rapid switching which can be required with a completely random binary signal.

(c) The signal intensity is low, with energy spread over a wide frequency range, hence it is a suitable forcing function for a plant operating normally as it causes little disturbance from the operating condition.

(d) The power of the noise can be arranged to be in the band of frequencies

of interest by appropriate choice of Δt and N. Choose $\Delta t \ll$ smallest system time constant, and $N\Delta t$ > settling time (some prior knowledge of the order of magnitude of system parameters is helpful).

(e) The input ACF is calculated to its full accuracy by correlation over one period; hence, to average out the effects of system noise, the cross correlation should be carried out for an integral number of input sequences. The smaller the PRBS amplitude relative to system noise, the longer the averaging time required. One sequence must be input to set the initial conditions correctly.

7.7 Illustrative example

To conclude the chapter, and to highlight the significance of its contents, this section presents and discusses briefly a typical set of graphs of the various



Fig. 7.16 Block diagram of system being identified

functions described earlier. The curves have been obtained from tests on a simulated second order system in the presence of noise, using a PRBS signal as forcing function. The system has been arbitrarily chosen to have $\omega_n = 6 \text{ rad/}$ second and $\zeta = 0.3$; the results of statistical testing should therefore confirm these values. It is assumed that the normal input may include a noise component, and that a disturbance signal introduces further noise within the system, both contributing to give a noisy output signal y(t), in the manner depicted schematically in Fig. 7.16.

In the absence of noise, the system can be subjected to an input change in the form of a step or an impulse function, and the resulting response recorded. Fig. 7.17 shows the response curves for a step of magnitude unity and an impulse of magnitude 1/4. Comparison of these curves with those of Fig. 4.7 would enable values to be estimated for ω_n and ζ , and the closeness of fit of each experimental response curve and the appropriate standard curve would indicate how well the system can be represented by a second order transfer function. In the presence of noise on the system output, these waveforms would be masked unless the input changes were of very large amplitude; with a real system a large enough step change to allow even an approximate estimate to be made would probably be unacceptable, and an impulse change would certainly not be acceptable. In this situation statistical testing can yield an impulse response curve with minimum disturbance to the system.



Fig. 7.17 Step response and impulse response of noise free system

Fig. 7.18b shows a typical sample trace of the output response of the system in the presence of noise for a nominally constant input (the noise characteristics have been arbitrarily chosen). A 63 bit PRBS signal of bit interval 0.1 second is used as input excitation for system identification; Fig. 7.18a shows the form of the PRBS input and the resulting system output in the absence of noise. The bit interval and the sequence length can be chosen on the basis of a preliminary estimate of the order of magnitude of the dominant roots of a system, and later can be confirmed as being satisfactory by inspection of the correlation functions. The chosen PRBS amplitude value gives a response amplitude which is of the same order of magnitude as the noise amplitude, although it can be seen to have quite different harmonic characteristics. The



Fig. 7.18 Typical traces of output response of system (a) forced by PRBS in the absence of noise (b) without PRBS in the presence of noise (c) forced by PRBS in the presence of noise response of the system to the PRBS signal in the presence of noise is shown in Fig. 7.18c and, although a clear difference can be seen between this and the normal noise output of Fig. 7.18b, the difference is not very marked. Smaller PRBS amplitudes could be used resulting in a less obvious effect on the output response, but a longer period of correlation would be required, and for very low signal-noise ratios results would tend to be poor.



Fig. 7.19 Autocorrelation functions of input and output signals

The autocorrelation functions of these input and output signals are shown in Fig. 7.19. The ACF of the PRBS signal has the form theoretically expected, whilst that of the system output in the absence of noise shows a reduction in signal power to somewhat less than half of the input power, and a marked tendency in the short term towards harmonic change at about 1 Hz. The ACF of the noise signal shows that there is a significant component of the signal which approximates to white noise with, in addition, some increase in power at frequencies around 1 to 1.5 Hz. The system is linear and hence the ACF of the noisy output is broadly the sum of the previous two. Such information from interpretation of the ACF curves confirms and quantifies trends predictable in this case from visual inspection of the response traces, and can be supplemented by information obtained from the corresponding power spectral density curves. Correlation for each curve shown was carried out for a time duration of about 130 seconds, which corresponds to about 20 periods of the PRBS; a longer period of correlation would help to smooth out the curves at the larger values of time shift r, provided the dynamic characteristics of the system being tested remained unchanged over the longer time span involved.

Cross correlation of the system output signal with the PRBS input should yield a good approximation to the impulse response of the system, provided that the bit interval and sequence length are wisely chosen. Fig. 7.20 shows



Fig. 7.20 Cross correlation functions

typical traces for the CCF with and without system noise; the former curve approaches the latter more closely when the correlation is carried out for longer time durations. The chosen bit interval of $\Delta t = 0.1$ second can now be seen to be a value which gives an adequate approximation to white noise for this system, and the period of 6.3 seconds correctly exceeds the system settling time (a sequence of length 31 bits could have been used instead). $\varphi_{xy}(\tau)$ can be seen to start from a small positive value for $\tau = 0$, consistent with the curve



Fig. 7.21 Power spectral density curves of input and output curves

being the response to a narrow triangular pulse starting at time $-\Delta t$, and not a true impulse. This is confirmed by noting the form of $\varphi_{yx}(\tau)$, where the output response curve is shifted in the opposite direction when correlating and which thus gives values of $\phi_{xy}(\tau)$ for negative values of τ .

The power spectral density curves for the input and output signals, plotted against a logarithmic frequency scale, are shown in Fig. 7.21. It can be seen that with a PRBS input almost the entire power of the output signal is contained in the frequency range 0 to 2 Hz, and the curve for $\Phi_{xx}(\omega)$ shows that over this frequency range the input PRBS signal has a substantially constant



Fig. 7.22 Bode plot and polar plot obtained from CCF

power spectral density. This confirms that the bit interval used gives an excitation signal which is a good approximation to true white noise for the system tested. In the absence of noise the magnitude of the transfer function can be calculated from the input and output power spectral densities using Eq. 7.20. The power spectrum of the noise signal shows it to have a bandwidth of about 5 Hz, and confirms that it has white noise characteristics with additional power at frequencies around 1.2 Hz. The presence of this noise in the system output would clearly introduce large errors if Eq. 7.20 were to be used to determine $|G(j\omega)|$. The final figure, Fig. 7.22, shows the Bode and polar plots for the system obtained by evaluating the cross spectrum from the cross correlation function for the noisy system shown in Fig. 7.20. The results are clearly consistent with those expected for a second order system with $\omega_n = 6$ rad/ second and $\zeta = 0.3$. The main evidence of errors arising from the presence of the noise shows up in the poor results for frequencies beyond about 3 Hz.

8 Feedback Systems—Accuracy and Stability

The earlier chapters of this book have been primarily concerned with describing how a dynamic physical system can be represented in certain convenient mathematical ways, and with showing how different types of system react to certain forms of forcing function. The main aim has been to present a clear picture of the nature of the dynamic behaviour of linear systems. This chapter and those following discuss the way in which the principle of feedback can be used to achieve a desired performance specification; some of the design methods available for synthesizing closed loop control systems are described.

Section 8.1 considers some general characteristics of systems incorporating a single feedback loop. It is shown that requirements for high accuracy of control and for fast well damped behaviour conflict, and that often a compromise must be made. Section 8.2 discusses the nature of steady state error and shows how it is a function both of the type of input and of the dynamic characteristics of the system. Sections 8.3 and 8.4 outline two widely used methods available for stability analysis, which are respectively relevant to time domain and frequency domain analysis; Section 8.5 defines gain and phase margins which are used to describe the degree of stability. The chapter concludes by considering the relationship between open loop and closed loop frequency response, and the graphical presentation of this information on a Nichols chart.

8.1 Closed loop or feedback control

An open loop or scheduling control, of which some everyday examples were given in Chapter 1, takes the form represented by Fig. 8.1. There is no comparison between the actual output variable of the system, usually referred to as the system response or controlled output c(t), and the desired output of the system which would be some function of the input variable, usually referred to



Fig. 8.1 Block diagram of open loop control system

as the reference input r(t). The output, using Laplace transform notation, is given by the relationship

$$C(s) = R(s) G(s)$$
8.1

and is thus determined by both the input R(s) and the transfer function G(s) of the system. External disturbances or deterioration with increasing age may well cause changes in the parameters of G(s); hence even in the steady state the system will only give the desired value of output at the design point.

Introduction of a negative feedback path allows the actual system output to be compared with the desired value, as shown in Fig. 8.2a, and an error signal



Fig. 8.2 Block diagram of simple feedback system (a) with unity feedback (b) with elements in the feedback loop

to be generated, this then forming the system input. This is termed *closed loop* control or feedback control. By inspection it can be seen that

$$C(s) = G(s) E(s)$$
$$E(s) = R(s) - C(s)$$

and

Elimination of E(s) from these equations gives the output as

$$C(s) = \frac{G(s)}{1 + G(s)} R(s)$$
 8.2

Again the output is dependent on the system transfer function and on the input function, but the effect of a change in G(s) is less than for the open loop system, and by appropriate choice of G(s) the error can be made small and the effects of external disturbances can be largely cancelled out.

If there is an element with transfer function H(s) in the feedback path, representing perhaps the dynamic characteristics of the transducer measuring

154 Feedback Systems—Accuracy and Stability

the output signal and converting it to the same form as the reference input signal, Fig. 8.2b, then

$$C(s) = G(s) E(s) = G(s)[R(s) - C(s) H(s)] \frac{C(s)}{R(s)} = \frac{G(s)}{1 + G(s) H(s)}$$
8.3

The characteristic equation as defined in Section 2.3 is the denominator of the transfer function equated to zero, and is thus

$$1 + G(s) H(s) = 0$$
 8.4

Although closing the loop allows the steady state error to be reduced or eliminated completely, the system can become very oscillatory or even unstable. This arises as a result of the lags that occur within the loop; these lags cause a delayed response to the corrective action and often also a delayed sensing of the error signal. In an extreme case this results in pronounced overcorrection leading to large overshoots and undershoots. The requirements of accuracy and stability conflict; the engineer must be aware of this and must be able to design the feedback system to achieve a satisfactory compromise. Basically, design can be regarded as a problem of arranging the location in the *s*-plane of the roots of the characteristic equation in such a way that the corresponding system performs according to prescribed specifications.

Figure 8.3 shows the general form taken by a feedback control system. Most commonly the error signal obtained from the error detector is amplified and





then fed to a power device which effects the necessary changes to the system, as shown in the figure. Often there are also compensating elements, auxiliary feedback loops, or other modifications designed to achieve the desired system performance. Some of these features will be illustrated in Chapter 11.

Servomechanisms or servos are, strictly speaking, a special group of feedback control systems where the output is a position or velocity. The term is, however, used much more widely and in many cases synonymously with feedback control systems. The input (or desired value) will in general be varying, and the output is designed to follow the input.

Regulators form another group of feedback control systems in which the reference input, although adjustable, is held constant for long periods of time (e.g. as in most temperature controllers). The output is maintained approximately constant at the value corresponding to the reference input, irrespective of external disturbances to the system. In process control, where regulating systems are widely used, the reference input is generally referred to as the *set point*.

Closed loop control action is frequently achieved in everyday life and in industrial situations by incorporating a human being within the feedback loop. Consider the action of moving to switch on a light, in relation to the generalized diagram. Fig. 8.3. The location of the switch can be thought of as the reference input, the position of the hand as the controlled output; there is no monitoring transducer, and the eyes act as the error detector. The eyes and brain generate the error signal and amplify it to give inputs to the muscles, the power device, which move the body and hand towards the switch and cause the hand to actuate the switch in the required manner. The importance of the feedback path can be seen by considering what occurs when it is broken by blindfolding or by complete darkness. It is worth noting that other feedback systems are also involved in this apparently simple action; for example, servomechanism action positions the pupil of the eve to focus on the hand and the switch, and regulator action alters the size of the iris to make adjustment for varying light intensity. The human being, although having a number of physical limitations and bringing with him problems of reliability, can act as a very sophisticated controller.

8.2 Steady state error

The prime reason for using feedback is to minimize the error between the actual system output and the desired system output. The magnitude of the steady state error, the value to which the error signal tends as the transient disturbance from any input change dies out, is of importance since it is a measure of system accuracy. To demonstrate what factors affect the value of the steady state error consider first the unity feedback system of Fig. 8.2a. The error signal e(t) is the difference between the reference input and the controlled output:

$$e(t) = r(t) - c(t)$$

The reference input signal can thus be thought of as the desired output. The steady state error e_{ss} is the limiting value of the error e(t) as time t becomes very large:

steady state error
$$= e_{ss} = \frac{\lim_{t \to \infty} e(t)}{t \to \infty} e(t) = \frac{\lim_{s \to 0} sE(s)}{s \to 0} e(s)$$
 8.5

by the final value theorem of Laplace transform analysis.

But

$$E(s) = R(s) - C(s)$$
$$= R(s) - G(s) E(s)$$
$$\therefore E(s) = \frac{R(s)}{1 + G(s)}$$

156 Feedback Systems—Accuracy and Stability

Hence the steady state error is

$$e_{\rm ss} = \lim_{s \to 0} \left(\frac{sR(s)}{1 + G(s)} \right)$$
8.6

This equation shows that the steady state error is a function both of the type of system as described by the transfer function G(s) and of the type of input R(s). Consider in turn three types of input.

(a) Step input function. i.e. a constant input for values of time t > 0. Let the magnitude of the step be k. The Laplace transform of the input is then

$$R(s) = \frac{k}{s}$$

Inserting this in Eq. 8.6 gives

$$e_{ss} = \lim_{s \to 0} \left(\frac{k}{1 + G(s)} \right) = \frac{k}{1 + \lim_{s \to 0} G(s)} = \frac{k}{1 + K_p}$$

where $K_p = \lim_{s \to 0} G(s)$ is called the *positional error coefficient*, or *positional* error constant. Rearranging the expression for e_{ss} gives

$$K_{\rm p} = \frac{k - e_{\rm ss}}{e_{\rm ss}}$$

Hence $K_p = \lim_{s \to 0} G(s) = \frac{\text{desired output} - \text{allowable steady state error}}{\text{allowable steady state error}}$ 8.7

Thus for the steady state error to be zero it is necessary that $K_p = \lim_{s \to 0} G(s) = \infty$,

which requires G(s) to have included in it a factor s in the denominator, an integral term. If no integral term is present then, for example, for an allowable steady state error of 1% of the step size, K_p must be at least $\frac{1-0.01}{0.01} = 99$; for

an error of 5% it must be at least 19. Hence, if an allowable steady state error is specified for a constant input, a restriction is placed on the system transfer function G(s) by requiring the gain to have a certain minimum value. In the field of process control this steady state error is termed the offset.

(b) *Ramp input function*. i.e. input changing at a constant rate. Let the input be a ramp increasing at k' units/second

i.e.
$$r(t) = k't$$

The Laplace transform of the input is then

$$R(s) = \frac{k'}{s^2}$$

$$e_{ss} = \lim_{s \to 0} \left(\frac{k'}{s + sG(s)} \right) = \frac{k'}{\lim_{s \to 0} sG(s)} = \frac{k'}{K_v}$$

where K, is called the velocity error coefficient or velocity error.

$$K_{v} = \lim_{s \to 0} sG(s) = \frac{\text{desired output velocity}}{\text{allowable steady state error}}$$
8.8

It should be noted that this does not refer to an error in velocity, but rather a positional error in following a velocity input. K, has units of seconds⁻¹.

For the steady state error to be zero with a ramp input, K_v must be infinite; this requires G(s) to have a factor s^2 on the denominator. If the error must be kept within 1% of k', then $K_v = 100$, but if an error of 5% is allowable then the gain can be reduced until $K_v = 20$.

(c) Acceleration input function. In this case

$$K_a = \text{acceleration error coefficient} = \frac{\lim_{s \to 0} s^2 G(s)}{s \to 0}$$

$$\frac{\text{desired output acceleration}}{\text{allowable steady state error}} (\text{seconds}^{-2}) \qquad 8.9$$

The analysis of these three types of input function shows that whether or not there is a steady state error for a given type of input depends on α , the power of s in the factored denominator of the open loop transfer function

$$G(s) = \frac{K(s - z_1)(s - z_2) \dots}{s^{z}(s - p_1)(s - p_2)(s - p_3) \dots}$$
8.10

Systems are classified as being Type 0, 1, or 2, where the type number is the value of α , which corresponds to the number of open loop poles at the origin. The values of the steady state error for these system types are summarized in Table 8.1.

	Steady state error with:								
System	steady input	ramp input	acceleration input						
Type 0	finite	infinite	infinite						
Type 1	0	finite	infinite						
Type 2	0	0	finite						

Table 8.1 Steady state error for type 0, 1, and 2 systems

When there is an element with transfer function H(s) in the feedback loop the error signal must be defined as E(s) = R(s) - C(s) H(s). This causes certain differences in the above evaluation of steady state errors.

8.3 Routh-Hurwitz stability criterion

It has been shown in Section 4.3 that the form of the transient response of a dynamic system is largely dependent on the location in the s-plane of the roots of the characteristic equation. For a system to be useful it must clearly be

158 Feedback Systems—Accuracy and Stability

stable at all times, where, considered in simple terms, it is defined as being stable if the output response to any bounded input function is finite. This implies that no roots of the characteristic equation can lie in the right half of the complex s-plane. A complex conjugate pair of roots with positive real part would give rise to an oscillation with amplitude increasing progressively and limited only by physical failure or saturation in part of the system. With a positive real root the output would increase exponentially and be limited for the same reasons. The problem of determining the stability of a linear system can thus be viewed as one of finding the roots of the characteristic equation.

If the characteristic equation is of high order then the task of determining the values of the roots can be tedious and time consuming. It is desirable to be able to determine more quickly and easily whether a system of known transfer function is stable or not. Around 1880 Routh and Hurwitz independently developed somewhat similar methods of determining whether any roots of a linear equation in the complex variable s have positive real parts, without having to solve the equation to find the values of the roots. The procedure is now used primarily as a rapid check on stability when all system parameters are fixed, and as a means of determining the limiting value for a variable parameter beyond which the system would become unstable.

Consider the general form of the characteristic equation, a polynomial in s:

$$1 + G(s) H(s) = a_0 s^n + a_1 s^{n-1} + a_2 s^{n-2} + \dots + a_{n-1} s + a_n = 0 \qquad 8.11$$

This has n roots, and these may be located anywhere in the complex s-plane.

For there to be no roots with positive real parts there is a necessary but not sufficient condition that all coefficients have the same sign and that none are zero. Hence, provided coefficient a_0 is positive, if inspection shows that one or more coefficients is negative or that one of the powers of s is absent, the equation is known to have at least one root in the right half of the s-plane. The system represented by that characteristic equation can then be said to be unstable without any further analysis being required. This can give a useful warning if an error of sign has been made in the theoretical derivation of a transfer function for a physical arrangement which is intuitively expected to be stable.

If this condition is satisfied then the necessary and sufficient condition that none of the roots has positive real parts is that the Hurwitz determinants of the polynomial must all be positive, where the determinants are given by

$$D_1 = a_1, D_2 = \begin{vmatrix} a_1 & a_3 \\ a_0 & a_2 \end{vmatrix}, D_3 = \begin{vmatrix} a_1 & a_3 & a_5 \\ a_0 & a_2 & a_4 \\ 0 & a_1 & a_3 \end{vmatrix}, D_4 = \begin{vmatrix} a_1 & a_3 & a_5 & a_7 \\ a_0 & a_2 & a_4 & a_6 \\ 0 & a_1 & a_3 & a_5 \\ 0 & 0 & a_2 & a_4 \end{vmatrix}, \text{ etc.}$$

The arithmetic involved in evaluating these determinants can largely be avoided since the Routh technique effectively does this more simply. An array of the following form is produced:

The first two rows are formed by writing down alternate coefficients of the polynomial equation. Each value in the subsequent rows is calculated from four of the previous values according to the following pattern

$$b_{1} = a_{2} - \frac{a_{0}a_{3}}{a_{1}}, \quad b_{2} = a_{4} - \frac{a_{0}a_{5}}{a_{1}}, \quad b_{3} = a_{6} - \frac{a_{0}a_{7}}{a_{1}}$$

$$c_{1} = a_{3} - \frac{a_{1}b_{2}}{b_{1}}, \quad c_{2} = a_{5} - \frac{a_{1}b_{3}}{b_{1}},$$

$$d_{1} = b_{2} - \frac{b_{1}c_{2}}{c_{1}}, \quad \text{etc.}$$

Coefficients are calculated until only zeros are obtained, the rows shortening until the s⁰ row contains only one value.

Every change of sign in the first column of this array signifies the presence of a root with positive real part. For stability, therefore, all values in the first column of this array must be positive.

Example 8.1. Consider the characteristic equation

 $s^4 + 2s^3 + s^2 + 4s + 2 = 0$

All coefficients are present and positive; hence, to determine whether the system is stable, form the Routh array by writing down these coefficients in the first two rows, and then evaluating from them the subsequent rows.

s4	1	1	2	
s ³	2	4		
s^2	-1	2		$b_1 = 1 - \frac{(1)(4)}{2} = -1$ $b_2 = 2 - \frac{(1)(0)}{2} = 2$
s ¹	8	0		(2)(2)
s ⁰	2			$c_1 = 4 - \frac{1}{(-1)} = 8$, etc.

There are two sign changes in the first column, hence there are 2 roots with positive real parts and the system represented by this characteristic equation is unstable. Note that the method does not give the value of these roots.

160 Feedback Systems-Accuracy and Stability

Special cases. Two kinds of difficulty can occur as a result of zeros in the array.

(a) If a zero appears in the first column and the remaining terms in the same row are not all zero, then the terms in the subsequent row would all be infinite. The procedure for applying the Routh criterion is then to replace s by a new variable $1/\sigma$ and thus obtain a new equation where the order of the coefficients has been reversed. If this equation has no roots with positive real parts then neither will the original equation have any.

Example 8.2

The Routh array is:

 $s^{5} + s^{4} + 4s^{3} + 4s^{2} + 2s + 1 = 0$ $s^{5} | 1 \ 4 \ 2$ $s^{4} | 1 \ 4 \ 1$ $s^{3} | 0 \ 1 \ 0$ $s^{2} | -\infty$

... and the table cannot be completed.

Letting $s = \frac{1}{\sigma}$, the equation becomes

 $\sigma^{5} + 2\sigma^{4} + 4\sigma^{3} + 4\sigma^{2} + \sigma + 1 = 0$

and the array is:

There are two changes of sign; hence each equation has two roots with positive real parts.

(b) If a complete row of zeros occurs then again the table cannot be completed. This condition indicates that one or more pairs of roots (real, imaginary, or complex) are an equal radial distance from the origin but diametrically opposite. The condition where one pair of conjugate roots lies on the imaginary axis is of most interest, since then the system will oscillate with constant amplitude, a condition of marginal stability. The equation corresponding to the coefficients just above the row of zeros is called the *auxiliary equation*, and has as its highest order term the power of s indicated in the reference column to the left of the row. The order is always even and indicates the number of root pairs, e.g. second order indicates presence of two equal and opposite roots. The value of these roots can be obtained by solving the auxiliary equation, and these roots are roots of the characteristic equation.

To complete the Routh array, differentiate the auxiliary equation with

respect to s, insert the coefficients of the resulting equation in place of the row of zeros, and then calculate the remaining coefficients as before.

Example 8.3

s ³	$+10s^{2}+16$	5s + 160	= 0
The table begins:	$ \begin{array}{c c} s^{3} \\ s^{2} \\ s^{1} \\ 0 \end{array} $	16 160 0	and cannot be completed.
The auxiliary equation is: Differentiate:	$\frac{10s^2 + 1}{20s}$		

This coefficient is inserted as the s1 row, and the table completed

s3	1	16
s2	10	160
s1	20	0
s ⁰	160	

There are no sign changes in the first column, hence there are no roots in the right half of the s-plane. The pair of roots is obtained from the auxiliary equation $10s^2 + 160 = 0$, or $s = \pm j4$, i.e. a pair of conjugate imaginary roots, indicating undamped oscillations of constant amplitude at a natural frequency of 4 rad/second.

Note: When analysing practical systems the coefficients often vary greatly in order of magnitude, in which case a substitution of the form $\sigma = 10s$ can help to ease the arithmetic.

8.4 Nyquist stability criterion

The Routh-Hurwitz criterion determines stability only in a binary sense (i.e. either stable, or unstable) and gives no information about the degree of stability; even although a system may be theoretically stable, oscillations may take too long to die out. Consider now the frequency domain. On the closed loop magnitude frequency plot (see Fig. 6.12) the absolutely unstable condition theoretically appears as a discontinuity, with amplitude tending to infinity at the frequency of instability. In practice, however, the amplitude is either limited to a steady value due to non-linearities, such as saturation within system components, or oscillation builds up to a point at which failure occurs. With sustained oscillations energy is transferred back and forth between two energy storage media (e.g. inertia and spring, or inductance and capacitance) and the external energy source only needs to make up the losses.

The Nyquist diagram, defined as an open loop polar plot, a polar plot of the system with the loop opened and hence a plot of $G(j\omega)H(j\omega)$, enables one to find whether any roots have positive real parts, without actually evaluating them. It also shows how near to instability that a system is, and thus can be used as one way of determining how best to improve the system stability.

162 Feedback Systems—Accuracy and Stability

Harmonic information experimentally obtained can also be used, without any need to determine the transfer functions.

In general the characteristic function will have the form

$$1 + G(s)H(s) = \frac{K(s - z_1)(s - z_2) \dots}{s^2(s - p_1)(s - p_2)(s - p_3) \dots}$$
8.12

For stability, none of the zeros of the characteristic function (roots of the characteristic equation) can have positive real parts. There is no particular restriction on the poles of 1 + G(s)H(s), which are also the poles of G(s)H(s). If any of the poles of G(s)H(s) lie in the right half of the s-plane (as can occur if there is a secondary feedback loop within the forward path) then the open loop system will be unstable, but the closed loop system can nevertheless still be stable.

The Nyquist stability criterion in its most comprehensive form is somewhat complex, and to understand it requires familiarity with the mathematical process of conformal mapping. For most practical systems, those in which the open loop system is itself stable, a simplified form of the criterion can be applied; the full criterion is only needed when the open loop system is unstable.

The simplified Nyquist stability criterion states that if an open loop system is stable then the system with the loop closed is also stable provided that the $G(j\omega)H(j\omega)$ locus on the polar plot does not enclose the (-1, j0) point. This is illustrated by Fig. 8.4. If the locus passes through the critical (-1, j0) point,



Fig. 8.4 Open loop polar plots of stable and unstable systems

this corresponds to a system of marginal stability, one with a pair of wholly imaginary roots.

The simplified Nyquist criterion can be described in an alternative way: a system is unstable if the open loop magnitude exceeds unity when the open loop phase lag is 180°. Physically, the condition of instability can be visualized as follows. If the reference input to a closed loop system is a sine wave, then the signal returning to the error detector will have a different amplitude and phase. If the phase lag is 180°, then the returning signal, when inverted and added to the reference input, will reinforce the signal. If the amplitude of the returning signal is less than that of the input signal at this phase lag, a steady condition will be reached, but if the amplitude is greater then the amplitude

will build up continuously until the system saturates. Even if the input signal is removed the system continues to oscillate.

8.5 Gain margin and phase margin

The previous section makes it clear that the position of the $G(j\omega)H(j\omega)$ open loop polar plot relative to the (-1, j0) point has great significance when considering system stability. It is desirable to be able to quantify the degree of stability or relative stability of a system, and this can be done by indicating how close the Nyquist plot passes to the critical point. The measures used are the gain margin and the phase margin; these are defined below and illustrated in Fig. 8.5.





The gain margin is defined as the amount by which the system gain can be increased before instability occurs, and is normally quoted in decibels.

Hence gain margin = $20 \log_{10} \frac{1}{|GH|_c} dB$ 8.13

where $|GH|_c$ is the open loop magnitude at the crossover point on the negative real axis, the magnitude corresponding to a phase lag of 180°. Thus for the plot of Fig. 8.5 which crosses at around the value 0.4 the gain margin would be 20 log₁₀ 2.5 = 8 dB. For first and second order systems the plot never crosses the negative real axis, and hence the gain margin is infinite. If the plot passes through the critical point the gain margin is zero, while if it encloses the critical point the gain margin is negative and gain must be reduced to attain stability of the feedback system. The gain margin by itself may not be sufficient to indicate relative stability, as can be seen from Fig. 8.6a where two plots each have an infinite gain margin although one passes very much closer to the critical point than the other, and Fig. 8.6b where the plot with the larger gain margin passes closer to the critical point.



Fig. 8.6 Polar plots illustrating limitation of gain margin on its own as a measure of relative stability

The phase margin is defined as the angle through which the Nyquist locus must be rotated in order that the unity magnitude point on the locus passes through the critical point. It is thus the amount by which the open loop lag falls short of 180° at the frequency where the open loop magnitude is unity. It is particularly significant when investigating the effect on stability of system changes which primarily affect the phase of $G(j\omega)H(j\omega)$.

The gain and phase margins can be read directly from a Bode plot (Fig. 8.7). The gain margin is the attenuation at the *phase crossover frequency*, while the phase margin is the phase lag at the *gain crossover frequency* deducted from 180°. When a numerical value is specified for the minimum acceptable gain margin or phase margin, then the required value of loop gain can be determined by shifting the magnitude plot up or down on the graph until the





8.6 Loci of constant closed loop magnitude and phase 165

specified value is achieved, and then calculating the corresponding value of gain. This frequently results in a value of gain which is satisfactory for stability but too low to give an acceptable steady state error. Where this is the case it is, in principle, possible to increase the gain to satisfy the accuracy requirements, and then to improve the correspondingly low stability margin by introducing additional components into the loop to reshape the harmonic locus in the vicinity of the gain and phase crossover points in such a way as to increase the gain and phase margins. This process of system compensation is outlined in Chapter 11.

It must be stressed that when investigating the stability of a closed loop system by frequency domain methods it is the open loop polar diagram which is studied, and it is this plot of $G(j\omega)H(j\omega)$, referred to as a Nyquist plot, which must not pass too close to the critical (-1, j0) point. It is, however, the closed loop and not the open loop which is potentially unstable. As a consequence, a practical system can be response tested with the feedback loop left open, and the margin of stability can be determined and any necessary alterations made before the loop is first closed.

8.6 Loci of constant closed loop magnitude and phase

The values of gain margin and phase margin form only one aspect of the description of the dynamic response of a feedback system, and there will frequently be a requirement to determine completely the variation of overall magnitude and phase with frequency, the closed loop frequency response. This can readily be evaluated from the open loop frequency response, whether obtained experimentally or analytically, with a small amount of calculation, and can be presented as a polar plot or as separate curves of closed loop magnitude and phase plotted against frequency.

Consider a unity feedback system. The overall transfer function is given by Eq. 8.2 and is

$$\frac{C(s)}{R(s)} = \frac{G(s)}{1 + G(s)}$$

or, in the frequency domain,

$$\frac{C(j\omega)}{R(j\omega)} = \frac{G(j\omega)}{1 + G(j\omega)}$$

Hence the closed loop magnitude, by convention given the symbol M and often referred to as the magnification, is

$$M = \frac{|C(j\omega)|}{|R(j\omega)|} = \left|\frac{C(j\omega)}{R(j\omega)}\right| = \frac{|G(j\omega)|}{|1 + G(j\omega)|}$$
8.14

For any given frequency the magnitudes $|G(j\omega)|$ and $|1 + G(j\omega)|$ can be measured from the plot as illustrated in Fig. 8.8, and the value of M calculated by dividing one by the other. It is clear that if the open loop plot approaches close to the (-1, j0) point, then for a certain range of ω , $|1 + G(j\omega)| \le |G(j\omega)|$



Fig. 8.8 Open and closed loop polar plots for unity feedback system with G(s) of the form $\frac{1}{s(1 + \tau s)}$

and the closed loop magnification M rises to a maximum in the vicinity of one particular frequency. This is the peak value M_p occurring at frequency ω_p (see Section 6.5), and at the limiting point of stability M_p becomes infinite. The closed loop phase is given by

$$\angle C(j\omega) - \angle R(j\omega) = \angle [C(j\omega)/R(j\omega)] = \angle G(j\omega) - \angle [1 + G(j\omega)] \quad 8.15$$

and this angle can be measured directly from the plot.

Since every point in the complex plane of a Nyquist diagram has associated with it a value of closed loop magnitude and one of phase, then points with the same value of one or the other can be joined to form loci of constant closed loop magnitude and phase. These are referred to as *M* contours and *N* contours respectively, and can be shown to be two families of circles.

Let the coordinates of a point on the plot $G(j\omega)$ be represented by x + jy.

Then
$$M = \left| \frac{C(j\omega)}{R(j\omega)} \right| = \left| \frac{G(j\omega)}{1 + G(j\omega)} \right| = \left| \frac{x + jy}{1 + x + jy} \right| = \frac{\sqrt{(x^2 + y^2)}}{\sqrt{\{(1 + x)^2 + y^2\}}}$$

 $\therefore M^2[(1 + x)^2 + y^2] = x^2 + y^2$
 $\therefore (1 - M^2)x^2 - 2M^2x + (1 - M^2)y^2 = M^2$

Dividing through by $(1 - M^2)$, and adding the common term $\left(\frac{M^2}{1 - M^2}\right)^2$ to both sides gives

$$x^{2} - \left(\frac{2M^{2}}{1 - M^{2}}\right)x + \left(\frac{M^{2}}{1 - M^{2}}\right)^{2} + y^{2} = \frac{M^{2}}{1 - M^{2}} + \left(\frac{M^{2}}{1 - M^{2}}\right)^{2}$$
$$\left(x - \frac{M^{2}}{1 - M^{2}}\right)^{2} + y^{2} = \left(\frac{M}{1 - M^{2}}\right)^{2}$$
8.16

or
This is the equation of a circle with centre $\left(\frac{M^2}{1-M^2}, j0\right)$ and radius $\left|\frac{M}{1-M^2}\right|$. A family of circles can thus be drawn on the Nyquist diagram; these circles are loci of constant M, as shown in Fig. 8.9. M = 1 is a special



Fig. 8.9 M contours on polar diagram

case giving a locus which is a straight line at x = -0.5; as M becomes very large the circles become small and tend towards the critical (-1, j0) point. When an open loop polar plot for a unity feedback system is superimposed on this, each intersection with an M circle gives a point on the M against ω plot, the value of ω being obtained by interpolation on the $G(j\omega)$ locus. The M circle which is just tangential to the $G(j\omega)$ locus gives the value of M_p , and ω_p is read off at the tangent point.

In a similar way it can be shown that contours of constant phase shift, the N loci, also form a family of circles. These loci have centres at $\left(-0.5, -j\frac{0.5}{N}\right)$ and have radii $0.5\sqrt{\left(\frac{N^2+1}{N^2}\right)}$ where N is the tangent of the phase angle; they thus take the form shown in Fig. 8 10

thus take the form shown in Fig. 8.10.



Fig. 8.10 N contours on polar diagram

For analysis and design it is more convenient to have the M and N loci plotted in the gain-phase plane rather than the polar plane. Such a plot is called a *Nichols chart* and is shown in Fig. 8.11 for the region of most interest to the engineer. The co-ordinates are open loop magnitude (in decibels) and open loop phase (in degrees), and the M and N circles when transferred to this plot take the form respectively of one set of contours encircling and another set radiating from the critical 0 dB, -180 degree point. When the open loop frequency response information is plotted on a Nichols chart, then the closed loop harmonic response information can be read off directly from the intersections of the locus with the M and N contours. Gain margin, phase margin, M_p , ω_p , and bandwidth can be determined as shown in Fig. 8.12. To find the required value of gain to meet a specified value of any of these parameters the $G(j\omega)$ locus is drawn for unity gain say, and the resulting locus shifted as required in the direction of the lines of constant open loop phase, the amount of shift then giving the value of open loop gain needed.



Open loop phase (degrees)

Fig. 8.11 Nichols chart



The transient response of a linear system to deterministic forcing functions has been shown to be dependent on the roots of the characteristic equation. The root locus method is a graphical procedure which was devised for determining the changing values of the roots of a characteristic equation for variation in a given system parameter. This parameter, often a simple multiplying factor proportional to the loop gain and normally labelled K, can be considered an independent variable and all roots of the characteristic equation become dependent on the parameter K. A set of loci drawn in the s-plane, showing how the root positions move in the plane as functions of the parameter K, is called a root locus plot.

Appropriate digital computer programs are available for root evaluation, for plotting of the loci, and for Laplace inversion if time responses are required, and these programs ease the computational effort required. However, plots showing the approximate form of the root loci can be produced manually in a comparatively short time; frequently these will suffice to give a useful qualitative understanding of system behaviour and of the influence of parameter and system variations. For example, if a system has several roots, then those which move towards the imaginary axis with an increasing value of K will become more dominant, while those moving away have progressively less influence on the transient behaviour. The influence of root position on the transient response has been described in Section 4.3. A value of K that gives a negative real root close to the origin can result in a sluggish exponential type of response. Roots lying along the same horizontal line have the same damped natural frequency of oscillation, while roots lying along a particular vertical line constrain the amplitude of the decaying response to the same exponential envelope exp $(-\zeta \omega_n t)$. Every pair of complex roots yields an oscillatory term in the response but the pair nearest the origin dominates the response.

After defining in detail what is meant by a root locus plot and showing how one can be produced, Sections 9.1 and 9.2, a number of aids to construction which simplify the manual plotting of a diagram are listed and explained in Section 9.3. Examples are given to illustrate the use of these aids, then Section 9.4 discusses the information obtainable from completed plots and the way in which such plots can be used. Finally, to cater for variations in more than one parameter, and to include situations where the independent variable is not a simple multiplying factor, the technique is generalized, Section 9.5, and the term *root contours* is then used to represent the loci of the roots of the characteristic equation.

9.1 Root locus plots

Consider a simple design situation in which the response of the mass-springdamper system shown in Fig. 2.6 is to be investigated for a range of spring stiffness constants K. Variations in the numerical value of this parameter cause changes in the values of the roots of the characteristic equation, and thus an alteration in the position of the roots in the *s*-plane.

For arithmetic simplicity, let M = C = 1; the transfer function, Eq. 2.15, then reduces to

$$\frac{X(s)}{F(s)} = \frac{1}{s^2 + s + K}$$
9.1

The roots of the characteristic equation

$$s^2 + s + K = 0$$
 9.2

can in this case be evaluated algebraically as

$$p_{1,2} = \frac{1}{2} \{-1 \pm \sqrt{(1-4K)}\}$$

When K = 0 the two roots are both real, $p_1 = 0$ and $p_2 = -1$. If K is gradually increased through the range 0 < K < 0.25, the two roots move towards each other along the negative real axis, starting from $p_1 = 0$ and $p_2 = -1$ until they coincide for K = 0.25 resulting in a double root at s = -0.5. Further increase in the value of K results in the roots becoming a complex conjugate pair, i.e.

$$p_{1,2} = \frac{1}{2} \{ -1 \pm j \sqrt{(4K-1)} \}$$

and the response then becomes oscillatory in nature. For $K = \infty$ the values of these roots are

$$p_{1,2} = -\frac{1}{2} \pm j\infty$$

which corresponds to an oscillatory response with infinite natural frequency, but nevertheless constrained within an amplitude decay envelope of exp -t/2.

The loci of the two roots, when K varies from 0 to ∞ , can now be drawn as shown in Fig. 9.1a. The step responses in Fig. 9.1b demonstrate how the system response changes with changing root position and how dependent the damped sinusoidal responses are on the parameter K. This result, of course, is not unexpected since for the second order transfer function, Eq. 9.1, the undamped natural frequency has the value \sqrt{K} and the damping factor has the value $1/(2\sqrt{K})$. Both the transfer function and the nature of the physical system lead one to expect the change of steady state value with K as shown.

It will be appreciated that although theoretically such a system is inherently stable for all values of K, an acceptable engineering response is likely to require a value of K of the order of unity. A value of K = 10 is likely to be unacceptable because of the large initial overshoot which in certain situations would lead to mechanical failure.



Fig. 9.1 Second order system $G(s) = \frac{1}{s^2 + s + \kappa}$ (a) root locus diagram (b) step response for variation in κ

By inspection of the root locus plot, the following information about the transient and frequency response is obtained for the system of Fig. 2.6 with transfer function given by Eq. 9.1.

- (i) Stability: The system is inherently stable for all values of stiffness K, no roots appearing in the right half of the s-plane.
- (ii) Transient response: For all values 0 < K < 0.25, the system is overdamped, as there is no imaginary part contributing to the time solution. For $0.25 < K < \infty$, the system is under-damped, critical damping occurring at K = 0.25. The undamped natural frequency ω_n increases and the damping factor ζ decreases with increase in K, but the amplitude of the decaying response is constrained to the same exponential envelope.
- (iii) Frequency response: For any given value of K the roots p_1 and p_2 are directly available from the root locus plot enabling the transfer function, Eq. 9.1, to be written as

$$\frac{X(s)}{F(s)} = \frac{1}{(s - p_1)(s - p_2)}$$
9.3

The Bode diagram can now be constructed by the methods of Section 6.3, enabling frequency response studies to be made.

To illustrate these ideas further consider a simple control engineering situation where feedback is used to maintain near constant speed of an internal combustion engine under varying load conditions. The arrangement assumed is that shown in block diagram form in Fig. 9.2a. The governor is a simple one with proportional gain term k_1 only, the actual engine speed is sensed by an electrical tachometer with a first order transfer function, and a reciprocating engine can for such a general study be adequately described by a second order transfer function (Eq. 2.17). At the design stage, one problem would be to evaluate the variation in dynamic performance resulting from changes in governor gain setting k_1 . If, for this example, the numerical values of the system parameters are arbitrarily chosen to be $\tau = 1$ second, $\omega_n = 5$ rad/second, and $\zeta = 0.9$, then the open loop transfer function can be written as

$$G(s)H(s) = \frac{C(s)}{E(s)} = \frac{K}{(s+1)(s^2+9s+25)}$$
9.4

where $K = k_1 k_2 k_3$, and H(s) = 1, as it is a unity feedback system. The closed loop transfer function is then

$$\frac{C(s)}{R(s)} = \frac{K}{(s+1)(s^2+9s+25)+K}$$
9.5

from which the characteristic equation is

$$(s+1)(s^2+9s+25)+K=0$$
 9.6

For K = 0, it can be seen that the roots of Eq. 9.6 are the poles of the transfer

function G(s)H(s), Eq. 9.4, and these poles are the starting points for the separate loci. For the numerical values selected there is one real root at -1 and a complex conjugate pair at $-4.5 \pm j2.18$. As K increases the values of these roots alter and the roots trace out the paths shown in Fig. 9.2b.

Transient response curves for varying values of K can be obtained by simulation of the system represented by the block diagram. Inspection of the measured engine speed response, Fig. 9.2c, obtained from simulation, to a unit step change in set point for a value of K near zero, shows a response similar to that of a first order system indicating that the real root close to s = -1 is dominant. As K increases the position of the real root moves to the left along the real axis while the complex roots move towards the right half of the s-plane and thus become more dominant. For a value of K = 315 the complex roots lie on the imaginary axis, the condition of marginal stability. As K increases from zero to infinity the system response varies from a rather sluggish overdamped response for small values of K, through what might be considered a 'good' response at a value of around K = 35 when the damping factor of the dominant roots is 0.7, to the unstable condition which exists for K > 315.

It is seen from Fig. 9.1a and Fig. 9.2b that a root locus plot consists of distinct loci each plotting the variation in the value of one root as the independent variable K is changed in value from zero to infinity. The number of root loci is equal to the order of the characteristic equation, and the plot is symmetrical with respect to the real axis, since complex roots must always occur in conjugate pairs for linear rational functions. Each locus starts (K = 0) at an open loop pole and finishes ($K = \infty$) at an open loop zero or else moves off to infinity.

9.2 Construction of root loci

The most general form of the characteristic equation for a feedback system is (Section 8.1)

$$1 + G(s)H(s) = 0$$
 9.7

Since G(s)H(s) is generally derived from the grouping of several system elements, each with relatively simple transfer function, it normally appears in factored form

$$G(s)H(s) = \frac{K(s-z_1)(s-z_2)\dots(s-z_m)}{(s-p_1)(s-p_2)\dots(s-p_n)}$$
9.8

where K is a constant. Any point s in the s-plane which satisfies Eq. 9.7 is a point on the root locus plot; the procedure for finding such points is the basis of the root locus method.

The test to determine whether any point s lies on a locus is developed from Eq. 9.7 by substitution of Eq. 9.8, resulting in the equation

$$\frac{K(s-z_1)(s-z_2)\dots(s-z_m)}{(s-p_1)(s-p_2)\dots(s-p_n)} = -1$$
9.9

As explained earlier, the standard terminology used in the literature is to



9.2 Construction of root loci 175

Fig. 9.2 Engine speed regulating system (a) block diagram representation (b) root locus diagram (c) measured engine speed responses to unit step input for variation in K

define the roots of the numerator labelled z as zeros, since they are the values of s that make the value of G(s)H(s) zero, while the roots of the denominator labelled p are known as poles since they make G(s)H(s) infinite in value. The left-hand side of Eq. 9.9 is a complex expression, therefore the equation is fully defined if the modulus and phase angle conditions are both satisfied. Writing Eq. 9.9 in modulus and angle form yields the equations

$$\frac{K|s-z_1||s-z_2|\dots|s-z_m|}{|s-p_1||s-p_2|\dots|s-p_n|} = 1$$
9.10

$$\begin{bmatrix} \angle (s-z_1) + \angle (s-z_2) \dots \angle (s-z_m) \end{bmatrix} - \begin{bmatrix} \angle (s-p_1) + \angle (s-p_2) \dots \\ \angle (s-p_n) \end{bmatrix} = \text{odd multiple of } \pi \quad 9.11$$

The terms |s - z| and |s - p| will be recognized as the lengths of vectors drawn from either a zero or a pole to a point s, whilst $\angle (s - z)$ and $\angle (s - p)$ are the arguments of these vectors measured relative to the positive real axis in the s-plane. By repeated application of Eq. 9.10 and Eq. 9.11 a root locus diagram can be drawn. The angle condition, Eq. 9.11, is used to locate points which are on the loci and hence determine the shape of the root locus plot. The magnitude condition, Eq. 9.10, then enables values of K to be assigned to specific points on each locus.

As an illustration of the application of Eq. 9.10 and Eq. 9.11, consider the open loop transfer function

$$G(s)H(s) = \frac{K(s+1)(s+2)}{s(s^2+2s+5)}$$

which has poles marked \times , and zeros marked o at the locations shown in Fig. 9.3. If an arbitrary point s_1 in the s-plane is chosen and vectors a, b, c, d and e are drawn to the zeros and poles, Fig. 9.3, then s_1 is a point on a root locus if, and only if, the angle condition defined by Eq. 9.11 is satisfied. Using the symbols given in the figure, this requires that

$$(\beta_1 + \beta_2) - (\alpha_1 + \alpha_2 + \alpha_3) = \text{odd multiple of } \pi$$

If this is so, then the point s_1 lies on one of the loci and the value of K at this s_1 location can be evaluated from the magnitude condition, Eq. 9.10,

$$\frac{Kab}{cde} = 1 \qquad \therefore \quad K = \frac{cde}{ab}$$

Given the pole-zero configuration of the open loop transfer function G(s)H(s), the construction procedure for manual plotting of a root locus diagram can be summarized as follows:

- (i) find, by trial and error, points in the s-plane that satisfy the angle condition given by Eq. 9.11, and join them to form the root loci.
- (ii) calculate the K values at points on the root loci by using the magnitude condition, Eq. 9.10.

It can now be seen that the root locus technique enables the position of the roots of the characteristic equation in the s-plane to be determined without



Fig. 9.3 Poles and zeros for $G(s)H(s) = \frac{K(s+1)(s+2)}{s(s^2+2s+5)}$

actually solving analytically for the roots of the characteristic equation for varying values of an independent variable. At first sight the application of (i) would appear to be a most impractical procedure to implement manually for all but the simplest of systems. However, with the aid of a series of rules which are described in the next section, the field of search can be reduced markedly and a root locus diagram can be drawn more readily.

9.3 Aids to construction of root locus diagram

A number of rules developed from Eq. 9.10 and Eq. 9.11 significantly aid the manual plotting of a root locus diagram and are of real value since, when methodically applied, they give a very good idea of the shape of the loci. The rules are given without rigorous proofs, but simple justification for each rule is provided whenever possible.

1. Starting point of loci: The loci start, K = 0, at the *n* poles of the open loop transfer function G(s)H(s).

Using the relationship $G(s)H(s) = \frac{KP(s)}{Q(s)}$, then the characteristic equation is

Q(s) + KP(s) = 0. From this equation it can be seen that the values of s that satisfy this equation when K = 0 are the factors of Q(s), the poles of the open loop transfer function.

Number of loci: The number of loci is equal to the order of the characteristic equation.

Each root traces out a locus as K varies from 0 to infinity. The loci are

continuous curves and, since complex roots must occur in conjugate pairs, the plot is symmetrical about the real axis.

3. Termini of loci: The root loci end at the *m* zeros of G(s)H(s), and if m < n as is usually the case the remaining (n - m) loci end at infinity.

The loci terminate as $K \to \infty$, and the magnitude condition, Eq. 9.10, can then only be satisfied either if one of the terms |s - z| is zero or one of the terms |s - p| is infinite.

4. Loci on real axis: Portions of the real axis are sections of root locus if the number of poles and zeros lying on the axis to the right is odd.

Consider a trial point on the real axis. The angle contribution from any pair of complex conjugate poles is 2π (in Fig. 9.4 when s_1 is on the real axis then



Fig. 9.4 Root loci on real axis

 $\alpha_2 + \alpha_3 = 360^\circ$), the angle contribution from a real pole or zero to the right is $-\pi$ or $+\pi$ respectively, while that from a real pole or zero to the left is 0. Hence the total angle is an odd multiple of π only if there is an odd number of poles and zeros lying on the real axis to the right of the trial point.

5. Angles of asymptotes: Those loci terminating at infinity tend towards asymptotes at angles relative to the positive real axis given by

$$\frac{\pi}{n-m}, \frac{3\pi}{n-m}, \frac{5\pi}{n-m}, \dots, \frac{\{2(n-m)-1\}\pi}{n-m}$$

This can be shown by considering the angle condition as applied to a point far from the group of open loop poles and zeros. The angle contribution from each pole and zero is then numerically equal. The effect of each zero is cancelled by that of a pole, and the sum of the angles for the remaining (n - m) poles is then an odd multiple of π , provided points lie along lines at the above angles. If one locus goes to infinity it does so at 180°, i.e. along the negative real axis, if two loci go to infinity they approach asymptotes at angles 90° and 270°, if three at 60°, 180°, and 300° etc.

6. Intersection of asymptotes on real axis: This occurs at the 'centre of

gravity' of the G(s)H(s) pole zero configuration, where the centre of gravity is determined from the following expression:

\sum (numerical values of G(s)H(s) poles) – \sum (numerical values of G(s)H(s) zeros) n-m

7. Intersection of root loci with imaginary axis: This is the limiting condition for stability; the value ω on the imaginary axis at intersection, together with the value of K, can be determined by application of the Routh-Hurwitz criterion as presented in Section 8.3.

8. Breakaway from real axis: A point in the s-plane where multiple roots exist is called a breakaway point; it occurs when two or more loci meet at the point and subsequently break away again along separate paths. The location of breakaway points can be found analytically by solving the equation dK/ds = 0.

Practical computational difficulties limit the ease of application of this rule, and it should be noted that not all of the roots of the equation dK/ds = 0correspond to breakaway points. The actual breakaway points are those roots of the equation at which the root locus angle condition is satisfied. If the polynomial is of high order it may be easier to find the breakaway points graphically by use of the angle condition than by solution of this equation. In many engineering situations the poles and zeros of G(s)H(s) all lie on the real axis, making it possible to use a graphical approach to find the breakaway point. Consider a case where there are three real poles at s = 0, s = -1, and s = -2 respectively, Fig. 9.5. A breakaway point must exist on the real axis



Fig. 9.5 Breakaway from real axis-multiple root condition

somewhere between s = 0 and s = -1 (rules 2 and 4). Choose a trial point which is very close to the axis and whose distance along the negative real axis is b, then application of Eq. 9.11 requires that

$$-(\alpha_1 + \alpha_2 + \alpha_3) = \text{odd multiple of } \pi$$

$$\left[\left(\pi - \frac{a}{b} \right) + \left(\frac{a}{1-b} \right) + \left(\frac{a}{2-b} \right) \right] = -\pi$$

$$3b^2 - 6b + 2 = 0$$

hence

or

b = 0.42 (or 1.58 which does not represent a breakaway point)

This procedure of constructing an algebraic equation by use of the angle condition and solving to find the breakaway point is a more direct and practical method. The root loci leave such a breakaway point in a direction normal to the real axis.

9. Angle of departure from complex pole: An indication of the initial direction in which a locus leaves a complex pole can be determined by applying Eq. 9.11 to a point very close to the pole of interest.

To illustrate, consider a situation in which a pair of complex poles is situated at $-1 \pm j$ with a third pole at the origin as shown in Fig. 9.6. Let s_1 be the trial point of interest, then



 $-(\alpha_1 + \alpha_2 + \alpha_3) = \text{odd multiple of } \pi$



Since s_1 is very close to the pole, $\alpha_1 = 135^\circ$ and $\alpha_3 = 90^\circ$ irrespective of the value of α_2 ,

hence $-(135 + \alpha_2 + 90) = -540^{\circ}$ and $\alpha_2 = 540 - 225 = 315^{\circ}$

which indicates that the locus will initially leave the complex poles in the direction of the origin. This approach is also useful for finding the angle of arrival at complex zeros.

By application of these nine simple rules a good approximation to the shape of each locus can be found, enabling the general form of a root locus diagram to be sketched. If a more accurate location of certain intermediate points is desired, then these sketched locus paths will give guidance as to the location in which the search to satisfy the angle condition, Eq. 9.11, should be conducted. Finally, numerical values of K are then directly obtained by application of the magnitude condition, Eq. 9.10.

Example 9.1. As an exercise to illustrate the use of these rules, consider the speed regulating system shown in Fig. 9.2a which has the characteristic equation defined by Eq. 9.6, i.e.

$$(s+1)(s^2+9s+25)+K=0$$
9.12

Details are inserted on the s-plane (Fig. 9.7) as they are determined from the aids:

1. Starting point of loci: The loci start at the poles of G(s)H(s), i.e. the roots of Eq. 9.12 when K = 0. These are

$$s = -1, s = -4.5 \pm j2.18$$

2. Number of loci: The number of loci is equal to the order of the characteristic equation, and Eq. 9.12 is of order 3.

3. Termini of loci: There are no zeros of G(s)H(s), hence all loci move towards infinity as K becomes very large.

4. Loci on real axis: There will be a section of root locus between s = -1 and $s = -\infty$, since along this section of the real axis the total number of poles and zeros to the right is one, and hence is odd. The remainder of the real axis does not form part of a root locus.

5. Angles of asymptotes: Since all 3 loci move to infinity, they approach asymptotes at 60° , 180° , and 300° to the positive real axis.

6. Intersection of asymptotes on real axis: The position of this point is at the centre of gravity of the 3 poles, which is given by





7. Intersection of root loci with imaginary axis: This is found by construction of the Routh array for the characteristic equation, Eq. 9.12, viz $s^3 + 10s^2 + 34s + 25 + K = 0$

$$s^{3} \quad 1 \qquad \qquad 34$$

$$s^{2} \quad 10 \qquad \qquad 25 + K$$

$$s^{1} \quad 34 - \left(\frac{25 + K}{10}\right) \quad 0$$

$$s^{0} \quad 25 + K$$

The limiting condition for a positive value of K is 315, and the frequency value is found from solution of the auxiliary equation

$$10s^2 + 25 + K = 0$$

On the imaginary axis $s = j\omega$, which yields

$$\omega = \sqrt{\left(\frac{25+K}{10}\right)} \approx \sqrt{34} = 5.84 \text{ rad/second}$$

8. Breakaway from real axis: There is a single locus on the real axis moving from the pole at -1 to $-\infty$ and no multiple roots occur on the real axis.

9. Angle of departure from complex poles: Use of the angle condition, Eq. 9.11, and measuring angles from the diagram, yields

$$-(\alpha_1 + \alpha_2 + \alpha_3) = \text{odd multiple of } \pi$$

-(149 + \alpha_2 + 90) = -540
$$\therefore \quad \alpha_2 = 540 - 239 = 301 \text{ degrees}$$

The exact location of a number of intermediate points on the dominant locus can now be found graphically by application of Eq. 9.11. The magnitude condition, Eq. 9.10, can be applied to find values of K at specific points on the loci. For example for the point s_1 , Fig. 9.7, a = 2.3, b = 1.9, and c = 3.5, obtained by measurement, hence

$$K = 2.3 \times 1.9 \times 3.5 = 15.3$$

In many engineering design situations this approximate root locus diagram would be adequate for preliminary design purposes.

9.4 Interpretation of the root locus diagram

To help understand the significance of the root locus technique, and to use it as a design tool, it is necessary to know what effect altering the position of poles and zeros or introducing new poles and zeros has on a root locus plot and hence on the system dynamic behaviour. A control engineer must build up a background of knowledge that enables him to relate root position to transient behaviour if he wishes to use the root locus approach as a guide for better system design. For example, he must be fully aware that introduction of an additional pole or moving a pole towards the right pushes the dominant complex loci towards the imaginary axis and by so doing reduces the system relative stability. Conversely, the introduction of a zero pulls the dominant loci away from the imaginary axis, shortens the dominant portion of real locus and thus tends to make the response more sluggish for a given value of system gain. The root locus diagram shows where a dominant branch comes from and how it is affected by open loop poles and zeros; a designer's attention is usually focused on the specific aspect of how close this dominant branch runs to the imaginary axis rather than on the overall pattern of the loci. A final choice of root position, however, can only be made with the aid of physical understanding of the system and sound engineering judgement and skill.

To highlight some of the more important points of interpretation, the performance changes occurring in a system with a second order process and three different controllers will be studied. The system is shown in block diagram form in Fig. 9.8a for a controller with the simple transfer function of unity, and a process with a pair of complex poles $-2 \pm j2$, which are the factors of $s^2 + 4s + 8$. Fig. 9.8b, the root locus diagram for the overall system, shows that the closed loop system is always stable for all positive values of K. The inherent weakness of a system having these controller/process characteristics can be seen from Fig. 9.8c, in that although the system exhibits a typical second order step response of acceptable form when K = 10, it also shows a steady state error from the set point of 45%. This magnitude of set point error normally would not be acceptable in engineering practice, making it necessary to modify the controller characteristics. A steady state error can be avoided by the introduction of a pole at s = 0, which is achieved by using an integral action controller, as shown in the block diagram Fig. 9.9a. The root locus diagram is shown in Fig. 9.9b. For small values of K the response would be sluggish, as can be seen from Fig. 9.9c for a value of K = 1, and hence unacceptable in engineering practice. For values of K in excess of 32, instability occurs. Again a value of K = 10 gives a transient response that would be acceptable, still resembling the response of a second order system with damping factor 0.4, but with this controller the steady state positional error is now zero.

Guaranteed absolute stability can be restored for all positive values of K by the introduction of an open loop zero which pulls the complex loci of Fig. 9.9b away from the imaginary axis. An open loop zero results if the characteristics of each controller are added to create the controller shown in Fig. 9.10a. This type of controller is referred to as a proportional plus integral (P + 1) controller and finds extensive use in engineering. There are still three loci but one now terminates at the zero at -1, and two move to ∞ as shown in Fig. 9.10b. The asymptotes are at 90° and 270° and they intersect the real axis at -1.5; on the real axis there is a portion of locus only between 0 and -1. This demonstrates that the introduction of the lead term (1 + s), in the numerator of G(s)H(s), has made the system stable for all values of K but has changed the system characteristics in such a way that the slow exponential decay term is more prominent for any given value of K. This undesirable condition must be minimized during design by appropriate positioning of the zero achieved by









Fig. 9.8 System with proportional control action (a) block diagram (b) root locus diagram (c) step response of system





introducing a second controlled variable into the controller equations, i.e. $M(s) = k_1 \left(1 + \frac{1}{T_1 s}\right) E(s)$, a design situation discussed in Chapter 10. Transient responses for K = 10, 20, and 100 are shown in Fig. 9.10c, indicating that the best value of K would be in the region of 20, an increase of a factor of 2 over



Fig. 9.10 System with proportional + integral control action (a) block diagram (b) upper half root locus diagram (c) step response with variation in K

9.4 Interpretation of the root locus diagram 187

the previous two controllers, resulting from the need to partially counteract the effect of the root on the shortened locus on the real axis.

The value of studying changes in the root locus pattern resulting from changes in the controller characteristics, and the design ideas that might emerge from consideration of these, are limited by the fact that there is no generally known acceptable specification for the optimum location of the roots of 1 + G(s)H(s) = 0. Were this not so, the problem would be reduced to adjusting loop parameters or changing components in the loop to obtain the desired root positions.

It must be emphasized that the root locus plot shows the locus of the roots of the system characteristic equation only, and for the general case, Section 2.3, the transient response is also influenced by any closed loop zeros which are present. Therefore, when attempting to predict the transient response from a root locus pattern, the influence of both closed loop poles and zeros must be taken into account either by use of simulation studies or by direct solution of the equations. Having obtained a root locus plot, extension of the ideas presented in Section 4.3 enables an analytical expression for a closed loop transient response curve to be derived for any given value of K. The method is illustrated in the next example.

Example 9.2. For the system shown in block diagram form in Fig. 9.10a, find an analytical expression for the output response c(t) for a unit step change in the set point r(t).

The closed loop transfer function for this unity feedback system is

$$\frac{C(s)}{R(s)} = \frac{G(s)}{1 + G(s)}$$

where

G(s) = forward path transfer function.

Hence

$$\frac{C(s)}{R(s)} = \frac{k_1k_2(s+1)}{s^3 + 4s^2 + 8s + k_1k_2(s+1)}$$

K(s+1)

$$=\frac{1}{s^3+4s^2+(8+K)s+K}$$

 $K = k_1 k_2$

where

Examination of the root locus diagram, Fig. 9.10b, might suggest that a value of K = 20 is likely to give an acceptable transient response. The roots of the denominator of Eq. 9.13 for this value of K can be found from the plot or analytically and, with the root values rounded off to one decimal place, the equation can then be written in factored form as

$$\frac{C(s)}{R(s)} = \frac{20(s+1)}{(s+0.8)(s+1.6+4.8j)(s+1.6-4.8j)}$$
9.14

9.13

Letting $R(s) = \frac{1}{s}$, a unit step input, and expanding C(s) into partial fractions,

as in Section 4.3, enables the output response to be written as

$$c(t) = \mathscr{L}^{-1}\left\{\frac{A}{s} + \frac{B}{(s+0.8)} + \frac{Ds+E}{s^2+3.2s+25.6}\right\}$$

Solving for the constants A, B, D, and E gives A = 1, B = -0.24, D = -0.76, and E = -2.64, hence

$$c(t) = \mathcal{L}^{-1}\left\{\frac{1}{s} - \frac{0.24}{(s+0.8)} - \frac{0.76s+2.64}{(s^2+3.2s+25.6)}\right\}$$

Expressing c(t) in a form that enables the Laplace inverse of each term to be found from Table 2.1 gives

$$c(t) = \mathscr{L}^{-1} \left\{ \frac{1}{s} - \frac{0.24}{(s+0.8)} - \frac{0.76(s+1.6)}{(s+1.6)^2 + (4.8)^2} - \frac{0.30 \times 4.8}{(s+1.6)^2 + (4.8)^2} \right\}$$

= 1 - 0.24e^{-0.8t} - 0.76e^{-1.6t} cos (4.8t) - 0.30e^{-1.6t} sin (4.8t)
= 1 - 0.24e^{-0.8t} - 0.82e^{-1.6t} sin (4.8t + tan⁻¹ 2.54)

By inserting discrete values of t, output values can be calculated and the step response curve drawn.

If the open loop zero had appeared in the feedback path instead of the forward path

$$G(s) = \frac{K}{s(s^2 + 4s + 8)}, \quad H(s) = s + 1$$

then the characteristic equation and the root locus plot would have been identical to the above. There would, however, be no zero in the closed loop transfer function, Eq. 9.14, and the resulting step response would be given by

$$c(t) = 1 - 1.08e^{-0.8t} - 0.17e^{-1.6t} \sin(4.8t - \tan^{-1} 0.53)$$

which is clearly not the same as that evaluated above. This demonstrates the importance of not overlooking closed loop zeros, as can happen when using a root locus diagram alone for design purposes.

Evaluation of a transient response equation can be a lengthy procedure for systems with high order characteristic equations. However, some simplification to the characteristic equation can be made by neglecting any root whose distance from the origin is more than 5 to 6 times that of the dominant roots, the resulting error being very small. It is more convenient to simulate a system on a computer, either analogue or digital, to obtain data on transient behaviour, and to use the root locus plot as a guide to the importance of, and likely changes resulting from, variation in system parameters. This approach has the advantage of avoiding the need for lengthy computer trial and observation periods.

i.e.

9.5 Root contours

To enable the root locus technique to be applied to engineering design problems in which a number of independent variables are to be chosen, the technique was extended to allow additional loci to be added to a root locus diagram. Such loci representing variation of roots with a second independent variable, and each corresponding to a given value of the first independent variable (K), are known as *root contours*. The approach also enables a root locus diagram to be plotted for an independent variable which is not normally a simple multiplying factor. The method employed to construct root contours necessitates the rearrangement of the system equations to form an equivalent transfer function in which the independent variable does in fact appear as a simple multiplying factor; after this the plot can be drawn as described in Sections 9.2 and 9.3 above.

To illustrate the method of approach, consider the block diagram shown in Fig. 9.11a, where T is taken as the independent variable of interest, and the system gain constant K on this occasion has been previously specified. A root locus diagram is required to show the variation in the roots of the characteristic equation as T increases in value from 0 to ∞ for certain fixed values of K. To be able to study this effect of a variable amount of derivative action in the feedback loop on the roots of the characteristic equation, the closed loop transfer function must be rearranged as follows. The closed loop transfer function for the system shown in Fig. 9.11a can be written as

$$\frac{C(s)}{R(s)} = \frac{K}{s^3 + 4s^2 + 8s + KTs + K}$$
9.15

Divide through by terms not containing the independent variable T to yield

$$\frac{C(s)}{R(s)} = \frac{\frac{K}{s^3 + 4s^2 + 8s + K}}{1 + \frac{KTs}{s^3 + 4s^2 + 8s + K}}$$
9.16

The transfer function written in this form corresponds to an equivalent system with different values of G(s) and H(s) but with the same characteristic equation. By inspection it can be seen that

$$[G(s)H(s)]_{equiv} = \frac{KTs}{s^3 + 4s^2 + 8s + K}$$
9.17

where the independent variable T is now a simple multiplying factor and hence a root locus diagram can be plotted in the normal way. If a family of curves is constructed from Eq. 9.17 for a range of values of K, then the root contours begin at the poles of $[G(s)H(s)]_{equiv}$ and end at its zeros. A closer inspection of Eq. 9.17 reveals that the poles in this equation are the roots of the characteristic equation of Eq. 9.15 when T = 0. Hence the starting points

of the root contours are points on the root loci for $G(s) = \frac{K}{s(s^2 + 4s + 8)}$ for





specific values of K. This is shown in Fig. 9.11b for a pair of values of K, and the independent parameter T. The introduction within the feedback loop of the transfer function H(s) = 1 + Ts has made the system inherently stable but, for increasing values of T the response becomes sluggish, as can be seen from Fig. 9.11c, and these responses would, in general, not be satisfactory in an engineering situation.

The dynamic behaviour of this system would be affected by the presence of a simple time constant τ in the feedback loop, as shown in Fig. 9.12a. In a practical system this simple time constant could represent the dynamic characteristics of a monitoring transducer. Alternatively the combined feedback blocks can be viewed as a compensation network introduced into the system loop in an attempt to achieve a specific change in the system dynamic behaviour. Part of any design task would be to study the effect on system performance of variation in this parameter.

The approach is again to find from the actual closed loop transfer function an analytically equivalent open loop transfer function which enables the root locus diagram to be constructed. From inspection of Fig. 9.12a the closed loop transfer function is

$$\frac{C(s)}{R(s)} = \frac{K(\tau s + 1)}{s(s^2 + 4s + 8)(\tau s + 1) + K(1 + Ts)}$$
9.18

Choosing a value of K = 8

$$\frac{C(s)}{R(s)} = \frac{8(\tau s + 1)}{s^3 + 4s^2 + 8s + 8Ts + 8 + \tau(s^4 + 4s^3 + 8s^2)}$$
$$= \frac{8(\tau s + 1)}{\frac{s^3 + 4s^2 + (8 + 8T)s + 8}{1 + \frac{\tau s^2(s^2 + 4s + 8)}{s^3 + 4s^2 + (8 + 8T)s + 8}}$$

which yields an equivalent open loop transfer function

$$[G(s)H(s)]_{equiv} = \frac{\tau s^2(s^2 + 4s + 8)}{s^3 + 4s^2 + 8(1 + T)s + 8}$$
9.19

For any chosen value of T, the root locus is plotted in the normal way, Fig. 9.12b. The loci start at the roots of the characteristic equation defined in Eq. 9.15 and end at the zeros $-2 \pm 2j$, 0 and 0 of Eq. 9.19. It will be noticed that since the order of the numerator of Eq. 9.19 is greater than that of the denominator, one of the three loci starts at infinity. In this case the portion of the real axis to the left of -1.5 forms part of a root locus; this is a necessary condition to satisfy Rule 4 of Section 9.3, and there will be a breakaway point to the left of the -1.5 point.

A most interesting result has emerged for the value of $T = \frac{1}{8}$; a fast response system has resulted with guaranteed stability as can be seen from inspection of Fig. 9.12b and Fig. 9.12c.

This illustration demonstrates clearly that the simple rules employed for





constructing a root locus diagram can be extended to root contours for variation of any poles or zeros. If a pole and zero are both to be varied, then two sets of root contours are necessary, the poles of the second being roots on the first. This makes possible the design of compensation networks of the type shown in the feedback loop of Fig. 9.12a, via a root locus plot. Further illustration of the use of the method for design purposes is given in the concluding chapter.

10 The Sampled-Data Process

Comparisons are made in this chapter between the discrete data and the continuous data system. These are used to illustrate the impact on system performance resulting from the introduction into the closed loop control scheme of a *sampler*, a device that converts continuous data into a discrete data form. Typically this might simply be to replace an analogue control device by its digital equivalent. However, the wider use of computer control resulting from increased availability of micro-electronic devices does extend the range of control algorithms that can be physically realized by the design engineer without significantly increasing equipment cost. Expanding the number of control algorithms readily available will enable more efficient 'start-up' and 'shut-down' procedures to be designed and so ensure less product waste. The improvement possible in servomechanism and regulator system performance will lead to better process dynamics and the manufacture of higher quality product, particularly when frequent quality changes are necessary on the same plant.

In its most basic form the sampled-data scheme can be defined as one in which the error signal is intermittently sampled at a constant rate. The error signal is then transformed into a sequence of pulses which are amplitude modulated in accordance with the continuous function of the signal from which the samples are taken. This basic form of sampled-data system is illustrated in Fig. 10.1. The symbols r(t), c(t) and e(t) represent the set-point disturbance (reference input for servo system), the output response and the



Fig. 10.1 Block diagram of sampled-data controller and process

actuating error signal respectively and each is a continuous time function. A starred symbol is used to indicate that a time function is in sampled form, thus the error variable denoted $e^*(t)$ is the discontinuous time function which is the input disturbance to the controller.

10.1 Mathematical description of sampling process

Although the process of sampling can be performed at a constant rate, at a variable rate, or at random, in the following analysis and in most practical applications constant rate periodic sampling is used. The sampling device permits the input disturbance to be sensed only during the short interval (Δ) of a sampling period and ignores this disturbance during the much longer time interval ($T - \Delta$) until the next sampling instant, Fig. 10.2. In most engineering



Fig. 10.2 The process of sampling a continuous signal, Eq. 10.1

applications the width Δ of the sampled pulse is small in comparison with the dominant process time constant and hence the sampler output can be represented without serious loss of accuracy as a train of rectangular pulses. The information from the continuous error signal is then contained in the amplitude variations that occur in the pulse train. The assumption of rectangular pulses of negligible width simplifies significantly the synthesis and analysis of sampled-data control schemes.

With e(t) as the continuous input disturbance and $e^*(t)$ as the sampler output, the input and output variables for the sampler can be related by

$$e^*(t) = \delta_T(t)e(t) \tag{10.1}$$

where $\delta_T(t)$ is the *ideal sampling function* which is a train of unit strength impulses described by

$$\delta_T(t) = \sum_{n=0}^{\infty} \delta(t - nT)$$

196 The Sampled-Data Process

The nomenclature $\delta(t - nT)$ by convention implies a unit impulse at time nT. To aid the understanding of the fundamental concept of sampling, each of the terms used in Eq. 10.1 is shown in Fig. 10.2.

Thus, Eq. 10.1 can be written as:

$$e^{*}(t) = e(t) \sum_{n=0}^{\infty} \delta(t - nT)$$
$$= \sum_{n=0}^{\infty} e(nT) \delta(t - nT)$$
10.2

where e(nT) is the amplitude of e(t) when t = nT. Applying Laplace transformation gives

$$\begin{split} E^*(s) &= \mathcal{L}[e^*(t)] = \mathcal{L}\sum_{n=0}^{\infty} e(nT) \ \delta(t-nT) \\ &= \mathcal{L}[e(0) \ \delta(t) + e(T) \ \delta(t-T) + e(2T) \ \delta(t-2T) + \dots] \end{split}$$

Using Eq. 2.8 and Table 2.1 yields

$$E^*(s) = e(0) + e(T)e^{-Ts} + e(2T)e^{-2Ts} + \dots$$

i.e.
$$E^*(s) = \sum_{n=0}^{\infty} e(nT)e^{-nTs}$$
 10.3

Jury (reference 24) shows that an equivalent harmonic representation is

$$E^{*}(s) = \frac{1}{T} \sum_{n=-\infty}^{\infty} E(s + jn\omega_{s})$$
 10.4

where ω_s denotes the sampling frequency $2\pi/T_{\perp}$

The first of these two equations (Eq. 10.3) is a convenient representation for analysis in the time domain and allows the introduction of z-transform theory which is the discrete counterpart to the Laplace transform of continuous data system analysis. The second equation (Eq. 10.4) is a theoretical basis for the extension of the basic frequency analysis methods to sampled-data systems.

10.2 Transfer function of sampled-data element

In the sampler and process arrangement of Fig. 10.3 the response c(t) is sampled synchronously with the manipulated variable m(t) to yield the pulsed time responses $c^*(t)$ and $m^*(t)$; the function g(t) is the impulse response function of a linear process with transfer function G(s).

The process input disturbance $m^*(t)$ is a train of ideal narrow pulses of





varying amplitude and for the linear process the output will be the sum of the individual impulse responses appropriately displaced in time. That is, for the time period $0 \le t \le nT$

 $\sigma(x) = \sigma(x) + \sigma(x) +$

$$c(t) = g(t)m(0) + g(t - T)m(T) + g(t - 2T)m(2T) + \dots$$
 10.5

Taking the Laplace transform of Eq. 10.5 and using Eq. 2.8 gives

$$C(s) = G(s)m(0) + G(s)m(T)e^{-nTs}$$

= $G(s) \sum_{n=0}^{\infty} m(nT)e^{-nTs}$
 $C(s) = G(s)M^*(s)$ 10.6

Equation 10.6 implies that the Laplace transform of the output variable c(t) is equal to the product of the Laplace transform of the pulsed input variable and the transfer function of the process. Note the similarity between Eq. 10.6 and that for the same arrangement but without the samplers given as Eq. 8.1.

By a similar argument it can be demonstrated that

$$C^*(s) = G^*(s)M^*(s)$$
 10.7

where $G^*(s)$ is the overall transfer function of the sampled-data process and is defined (Eq. 10.4) as

$$G^*(s) = \frac{1}{T} \sum_{n=-\infty}^{\infty} G(s + jn\omega_s)$$

It should be noted that the sampled-data transfer function gives information only at sampling instants.

10.3 Closed-loop transfer function

Drawing an analogy with Fig. 8.2b, the basic sampled-data system would be as shown in Fig. 10.4a. To help in the development of the transfer function relationships, Eq. 10.3 allows $E^*(s)$ to be written as

$$E^{*}(s) = \sum_{n=0}^{\infty} [r(nT) - b(nT)]e^{-nTs}$$

= R*(s) - B*(s)

and hence the equivalent circuit shown in Fig. 10.4b can be introduced for analytical purposes.

Inspection of Fig. 10.4b reveals that

$$C(s) = G(s)E^*(s) \tag{10.8}$$

 $E^*(s) = R^*(s) - B^*(s)$ 10.9

$$B^*(s) = GH^*(s)E^*(s)$$
 10.10

where $GH^*(s)$ is the pulsed transfer relationship of the product G(s)H(s). These





expressions can be treated as algebraic equations in an identical manner to those in Sections 2.3 and 8.1.

Substituting for $B^*(s)$ from Eq. 10.10 into Eq. 10.9 gives

$$E^*(s) = \frac{R^*(s)}{1 + GH^*(s)}$$
 10.11

Inserting Eq. 10.11 into Eq. 10.8 yields the Laplace transform for the output variable c(t) of the sampled-data system

$$C(s) = \frac{G(s)R^*(s)}{1 + GH^*(s)}$$
 10.12

Using Eq. 10.4 to replace the starred terms in Eq. 10.12 and substituting $s = j\omega$ yields the frequency function

$$C(j\omega) = \frac{G(j\omega)T^{-1}\sum_{\substack{n=-\infty\\n=-\infty}}^{\infty} R(j\omega + jn\omega_s)}{1 + T^{-1}\sum_{\substack{n=-\infty\\n=-\infty}}^{\infty} GH(j\omega + jn\omega_s)}$$
10.13

The function $GH^*(s)$ is defined as the open-loop transfer function of the sampled-data control system.

10.4 Polar plots

Graphical representation of the frequency response information for $GH^*(s)$ can be obtained from the Argand diagram as described in Section 6.2 if the frequency response function is written as

$$GH^*(j\omega) = T^{-1} \sum_{n=-\infty} GH[j(\omega + n\omega_s)]$$

= $T^{-1} \left\{ GH[j\omega] + \sum_{n=1}^{\infty} GH[j(\omega + n\omega_s)] + \sum_{n=1}^{\infty} GH[j(\omega - n\omega_s)] \right\}$
= $T^{-1} \{ GH[j\omega] + GH[j(\omega + \omega_s)] + GH[j(\omega - \omega_s)]$
+ $GH[j(\omega + 2\omega_s)] + GH[j(\omega - 2\omega_s)] + \dots \}$ 10.14

and then the vectors for the component functions in this expression are combined graphically or numerically. Since the sampling frequency is usually chosen to be at least 4 or 5 times the system bandwidth, Eq. 10.14 can be approximated by the first three terms without serious loss of accuracy in the polar plot, i.e.

$$GH^{*}(j\omega) \simeq T^{-1}GH[j\omega] + T^{-1}GH[j(\omega + \omega_{s})] + T^{-1}GH[j(\omega - \omega_{s})]$$
 10.15

The contribution made by the second term with increasing values of ω becomes less significant and can often be omitted from Eq. 10.15. The



Fig. 10.5 Approximate polar plot for sampled-data system

200 The Sampled-Data Process

sequence of events for the construction of $GH^*(j\omega)$ using the remaining two terms, illustrated in Fig. 10.5, is:

1. Draw the polar plots for both positive and negative frequency values, as described in Section 6.2.

2. Label the points for specific frequency values $\omega_1, \omega_2, ..., \omega_n$, as $A_1, A_2, ..., A_n$ on the positive plot and the points corresponding to $(\omega_1 - \omega_s), (\omega_2 - \omega_s), ..., (\omega_n - \omega_s)$ as $B_1, B_2, ..., B_n$ on the negative plot.

3. Add graphically the vectors OB_1 , OB_2 , ..., OB_n to the vectors OA_1 , OA_2 , ..., OA_n respectively to give points labelled $C_1, C_2, ..., C_n$.

4. The line trajectory through these points is a first approximation to the sampled-data polar plot, as shown by the broken line in Fig. 10.5.

5. A better estimate would result if the vector $T^{-1}GH[j(\omega + \omega_s)]$ were included, particularly at low values of ω .

Example 10.1. To demonstrate the application of this procedure, consider the second order transfer function represented graphically by Fig. 6.3. For the largest of the polar plots shown, with $\omega_n = 1$ rad/second and $\zeta = 0.5$, the equation is:

$$G(j\omega) = \frac{1}{(1-\omega^2) + j\omega}$$

Using the three terms of Eq. 10.15 the pulsed polar plot is described by

$$TG^*(j\omega) = G[j\omega] + G[j(\omega - \omega_s)] + G[j(\omega + \omega_s)]$$
$$= \frac{1}{(1 - \omega^2) + j\omega} + \frac{1}{[1 - (\omega - \omega_s)^2] + j(\omega - \omega_s)}$$
$$+ \frac{1}{[1 - (\omega + \omega_s)^2] + j(\omega + \omega_s)}$$

The continuous process has a natural frequency of 1 rad/second and Fig. 6.9 indicates the process bandwidth to be approximately 1.5 rad/second. A sampling frequency of 6 rad/second will be adequate to obtain representative process data. This corresponds to a sample interval of $\pi/3$ seconds (\simeq 1), hence

$$G^{*}(j\omega) = \frac{1}{(1-\omega^{2})+j\omega} + \frac{1}{[1-(\omega-6)^{2}]+j(\omega-6)} + \frac{1}{[1-(\omega+6)^{2}]+j(\omega+6)}$$

Values obtained from this equation are plotted in Fig. 10.6 for $0 \le \omega \le 4$, and show the contribution of each term of Eq. 10.15 to the shape and size of the polar plot. The most significant point to observe is that whilst the locus of $G(j\omega)$ indicates a process which is stable for any value of gain, the introduction of the sampler destroys this property (the simplified Nyquist criterion of Section 8.4 can still be applied). For frequencies in excess of $\omega_s/2$, 3 rad/second in this example, the polar plot becomes a mirror image of itself about the real



Fig. 10.6 Effect of sampler on polar plot of second order system $G(s) = 1/(s^2 + s + 1)$, $\omega_s = 6 \text{ rad/second}$

axis, a consequence of the harmonic distortion introduced by the process of sampling (described in Section 7.5).

This illustrates that the sampling operation undermines system stability and that the harmonic characteristics as given by the shape and size of the polar plot are a function of both process gain and sampling frequency. Therefore, in system design, variation in the parameter T (sample interval) must be included when dynamic behaviour is investigated.

10.5 Pulse transfer function

The requirement to evaluate the infinite series specified by Eq. 10.3 can be avoided by the introduction of a new variable z defined by

 $z = e^{sT}$

Application of this transformation to Eq. 10.3 yields

$$E^*\left(\frac{1}{T}\ln z\right) = E(z) = \sum_{n=0}^{\infty} e(nT)z^{-n}$$
 10.16

where E(z) is defined as the z-transform of $e^{*}(t)$. In general, any continuous function that possesses an s-transform will have an equivalent z-transform and

202 The Sampled-Data Process

it is common to write

$$E(z) = \mathscr{Z}[e(t)]$$

To derive a pulse transfer function this concept will be applied to Fig. 10.3. The output response for this arrangement, using the definition implied by Eq. 10.16, is

$$C(z) = \sum_{n=0}^{\infty} c(nT) z^{-n}$$
 10.17

The value of the process output at the sampling instant t = nT is, from Eq. 10.5,

$$c(nT) = g(nT)m(0) + g(nT - T)m(T) + g(nT - 2T)m(2T) + \dots$$
$$= \sum_{k=0}^{n} g(nT - kT)m(kT)$$
10.18

Since g(nT - kT) = 0 for values of k > n, Eq. 10.18 can be written

$$c(nT) = \sum_{k=0}^{\infty} g(nT - kT)m(kT)$$
 10.19

Substituting Eq. 10.19 into Eq. 10.17 yields the output response function

$$C(z) = \sum_{n=0}^{\infty} \left[\sum_{k=0}^{n} g(nT - kT)m(kT) \right] z^{-n}$$

In view of the fact that g(nT - kT) = 0 when k > n this expression is the same as

$$C(z) = \sum_{p=0}^{\infty} g(pT) z^{-p} \sum_{k=0}^{\infty} m(kT) z^{-k}$$

Using the definition of Eq. 10.16, the right hand side of the above equation can be abbreviated to

$$C(z) = G(z)M(z)$$
 10.20

Equation 10.20 is the z-transform equivalent of the Laplace transform equation given as Eq. 10.7, and G(z) is known as the process *pulse transfer function*.

Although z-transformation is only applicable to discrete signals, it can be applied to a continuous process response if signal values are required at the sampling instants only and no information is needed between sample instants.

To demonstrate the use of Eq. 10.17 for deriving z-transforms two simple examples are given.

Example 10.2. Consider a step disturbance of magnitude K applied at time t = 0

i.e.
$$f(t) = K$$
 for $t \ge 0$
The output of the sampler is then

$$f(nT) = K$$
 for $n = 0, 1, 2, ...$

Hence

$$F(z) = \sum_{n=0}^{\infty} f(nT) z^{-n} = \sum_{n=0}^{\infty} K z^{-n}$$
$$= K(1 + z^{-1} + z^{-2} + z^{-3} + \dots)$$

where the term in brackets is the binomial expansion of
$$(1 - z^{-1})^{-1}$$
 for $|z^{-1}| < 1$ Hence the z transform of a step disturbance is

2 -transform of a ste

$$F(z) = \frac{Kz}{z-1}$$

The Laplace transform counterpart is

$$F(s) = \frac{K}{s}$$

Example 10.3. Consider the Laplace transform

$$F(s) = \frac{K}{s^2 + 4s + 8}$$

From Table 2.1 the time response f(t) is

$$f(t) = \frac{1}{2}Ke^{-2t}\sin 2t$$

At t = nT

and hence

$$=\frac{K}{4j}\left[\frac{1}{1-e^{-2T}e^{j2T}z^{-1}}-\frac{1}{1-e^{-2T}e^{-j2T}z^{-1}}\right]$$

by inspection of Example 10.2.

Hence
$$F(z) = \frac{K}{4j} \left[\frac{e^{-2T} z^{-1} (e^{j2T} - e^{-j2T})}{1 - e^{-2T} z^{-1} (e^{j2T} + e^{-j2T}) + e^{-4T} z^{-2}} \right]$$
$$= \frac{K}{2} \left[\frac{z e^{-2T} \sin 2T}{z^2 - 2z e^{-2T} \cos 2T + e^{-4T}} \right]$$

Some widely used z-transform pairs are listed in Table 10.1.

204 The Sampled-Data Process

Time function $f(t)$	Laplace transform $F(s)$	z-transform F(z)				
$\delta(t)$	1	1				
$\delta(t-nT)$	e ^{-nTs}	$\frac{1}{z^{\alpha}}$				
unit step	$\frac{1}{s}$	$\frac{z}{z-1}$				
t	$\frac{1}{s^2}$	$\frac{Tz}{(z-1)^2}$				
ē- <i>a</i> ≀	$\frac{1}{s+a}$	$\frac{Z}{Z - e^{-\alpha T}}$				
te ^{-at}	$\frac{1}{(s+a)^2}$	$\frac{T_{ZB}^{-aT}}{(z-e^{-aT})^2}$				
sin <i>wt</i>	$\frac{\omega}{s^2 + \omega^2}$	$\frac{z\sin\omega T}{z^2 - 2z\cos\omega T + 1}$				
cos wt	$\frac{s}{s^2 + \omega^2}$	$\frac{z^2 - z \cos \omega T}{z^2 - 2z \cos \omega T + 1}$				
e ^{-at} sin <i>wt</i>	$\frac{\omega}{(s+a)^2+\omega^2}$	$\frac{z e^{-aT} \sin \omega T}{z^2 - 2z e^{-aT} \cos \omega T + e^{-2aT}}$				
e ^{-at} cos <i>wt</i>	$\frac{s+a}{(s+a)^2+\omega^2}$	$\frac{z^2 - z e^{-aT} \cos \omega T}{z^2 - 2z e^{-aT} \cos \omega T + e^{-2aT}}$				

Table 10.1 Laplace and z-transforms for basic time functions

10.6 Block diagrams

To illustrate the use of Eq. 10.7 and Eq. 10.20 in the derivation and manipulation of block diagrams, the three principal loop arrangements of Fig. 10.7a, b, and c will be studied; other configurations are reducible to combinations of these loop arrangements.

(a) Processes in cascade—Fig. 10.7a

The output responses from the two sections are

$$U_{2}^{*}(s) = G_{1}^{*}(s)U_{1}^{*}(s)$$
$$C^{*}(s) = G_{2}^{*}(s)U_{2}^{*}(s)$$

Eliminating $U_2^*(s)$ gives the starred transfer function

$$\frac{C^{*}(s)}{U_{1}^{*}(s)} = G_{1}^{*}(s)G_{2}^{*}(s)$$

and



Fig. 10.7 Block diagrams for systems with sampling (a) processes in cascade (b) closed loop with one sampler (c) closed loop with two samplers

and the corresponding pulsed transfer function is

$$\frac{C(z)}{U_1(z)} = G_1(z)G_2(z)$$

i.e. the product of the individual pulsed transfer functions.

(b) Closed loop with one sampler—Fig. 10.7b The starred error signal is

$$E^*(s) = R^*(s) - B^*(s)$$

where

$$B^*(s) = GH^*(s)E^*(s)$$

Eliminating B*(s) gives on rearrangement

$$E^*(s) = \frac{R^*(s)}{1 + GH^*(s)}$$

which is the Laplace transform of the sampler output response. The output response from the system at the sample instants is:

$$C^{*}(s) = G^{*}(s)E^{*}(s)$$

206 The Sampled-Data Process

Substituting for $E^*(s)$ yields

$$C^*(s) = \frac{G^*(s)R^*(s)}{1 + GH^*(s)}$$

and in terms of the z-transform

$$C(z) = \frac{G(z)R(z)}{1 + GH(z)}$$

The equation only describes the values of the output response at the sampling instants and the term GH(z) is obtained by first multiplying together the G(s) and H(s) functions, then the z-transform of the product is obtained.

(c) Closed loop with two samplers—Fig. 10.7c

The starred error signal is

$$E^{*}(s) = R^{*}(s) - B^{*}(s)$$
Also
$$C^{*}(s) = G^{*}(s) E^{*}(s)$$
and
$$B^{*}(s) = H^{*}(s) C^{*}(s)$$

Eliminating $E^*(s)$ and $B^*(s)$ gives the starred Laplace transform of the system response as:

$$C^{*}(s) = \frac{G^{*}(s) R^{*}(s)}{1 + G^{*}(s) H^{*}(s)}$$

The corresponding z-transform is

$$C(z) = \frac{G(z) R(z)}{1 + G(z) H(z)}$$

Note that G(z)H(z) is not the same as GH(z), but is the product of the individual z-transforms of the functions G(s) and H(s).

10.7 Inverse operation

To obtain the time response $f^{*}(t)$ of a process where F(z) has been evaluated by an equation in the form of Eq. 10.20, a mathematical procedure known as inverse z-transformation must be carried out. The most direct route available to this end is to adopt either a partial fraction or a power series expansion approach.

The former method requires the expansion of $z^{-1}F(z)$ into partial fractions such that the inverse z-transformation of each term multiplied by z is recognizable from z-transform tables in an analogous manner to Section 2.2 and Chapter 4. In the latter, the easier of the two methods to implement, F(z) is expanded into a power series of z^{-1} by long division; the coefficient of the z^{-n} term corresponds to the value of the time function $f^*(t)$ at the *n*th sampling instant.

In order to obtain the value of $f^{*}(t)$ as time becomes infinite, assuming F(s)

has no poles on the imaginary axis or to the right of it in the s-plane, use is made of the final-value theorem:

$$\lim_{t \to \infty} \{f^*(t)\} = \lim_{z \to 1} \left\{ \frac{z-1}{z} F(z) \right\}$$
 10.21

i.e. it is given by the value of $\frac{z-1}{z} F(z)$ as z approaches unity.

10.8 Data reconstruction

Reconstruction of continuous data from a sampled record is performed by starting with known data about the signal up to a given sample instant and then estimating by extrapolation data values up to the next sample instant. A mathematical approximation for the signal in the interval $nT \le t < (n + 1)T$ can be derived from the Taylor's power series expansion

$$f_n(t) = f(nT) + f(nT)(t - nT) + \frac{f(nT)}{2!}(t - nT)^2 + \dots$$

where $f_n(t)$ is the estimate for f(t) between the two measured consecutive samples. The most widely used data reconstruction device is the *zero-order* hold in which f(t) across the interval is approximated by the constant value

$$f_n(t) = f(nT)$$
 for $nT \le t < (n+1)T$

For the zero-order hold device the output is thus a set of flat-top pulses of width T, as illustrated in Fig. 10.8a. In contrast a *first order hold* device **loout** 1



Fig. 10.8 Data reconstruction (a) zero order hold (b) 1st order hold

208 The Sampled-Data Process

incorporates the first two terms in the Taylor's power series expansion, Fig. 10.8b.

Mathematically, the zero-order hold device can be described by use of Eq. 2.4 applied to the signal of Fig. 10.8a, which results in the Laplace transform of the output function

$$F(s) = \int_{0}^{T} f(0)e^{-st} dt + \int_{T}^{2T} f(T)e^{-st} dt + \dots + \int_{nT}^{(n+1)T} f(nT)e^{-st} dt + \dots$$
$$= \frac{1 - e^{-sT}}{s} \sum_{i=0}^{\infty} f(iT)e^{-isT}$$
$$= \frac{1 - e^{-sT}}{s} F^{*}(s)$$

The transfer function of a zero-order hold circuit is thus

$$G_{\rm ho}(s) = \frac{1 - e^{-sT}}{s}$$
 10.22

The frequency response function $G_{ho}(j\omega)$ is obtained by effecting the substitution $s = j\omega$ in Eq. 10.22.

Example 10.4. The open-loop transfer function used in Example 10.1 will be studied further to illustrate the power series expansion method as a means of obtaining the inverse z-transformation, and the use of Eq. 10.21 to determine the steady state value.

The combined transfer function is

$$G_{\text{ho}}(s) \ G(s) = \frac{(1 - e^{-sT})}{s} \left(\frac{1}{s^2 + s + 1}\right)$$

and hence

$$\begin{aligned} G_{\text{ho}}G(z) &= (1 - z^{-1})\mathscr{Z}\left[\frac{1}{s(s^2 + s + 1)}\right] \\ &= (1 - z^{-1})\mathscr{Z}\left[\frac{1}{s} - \frac{s + 1}{s^2 + s + 1}\right] \\ &= (1 - z^{-1})\mathscr{Z}\left[\frac{1}{s} - \frac{s + 0.5}{(s + 0.5)^2 + 0.866^2} - \frac{0.5}{(s + 0.5)^2 + 0.866^2}\right] \end{aligned}$$

From Table 10.1

$$G_{bo}G(z) = (1 - z^{-1}) \left[\frac{z}{z - 1} - \frac{z^2 + e^{-0.5T}(0.5774 \sin 0.866T - \cos 0.866T)z}{z^2 - 2ze^{-0.5T} \cos 0.866T + e^{-T}} \right]$$

If the sampling interval is chosen to be 1 second, for the reasons given in Example 10.1

$$G_{\rm ho}G(z) = \frac{0.3404z + 0.2415}{z^2 - 0.7859z + 0.3679} = \frac{C(z)}{U(z)}$$

The output response for a unit step input is

$$C(z) = \left(\frac{z}{z-1}\right) \left[\frac{0.3404z + 0.2415}{z^2 - 0.7859z + 0.3679}\right]$$
$$= \frac{0.3404z^2 + 0.2415z}{z^3 - 1.7859z^2 + 1.1538z - 0.3679}$$

This equation can now be expressed in the form

$$C(z) = a_0 + a_1 z^{-1} + a_2 z^{-2} + \dots$$

by use of manual long division (or a simple computer program) as follows:

$z^3 - 1.7859z^2$	0.3404z	$+ 0.8494z^{-1}$	$^{2} + 1.1241z^{2}$	-3 +	
+ 1.1538z - 0.3679	$0.3404z^2$	+0.2415z			
	$0.3404z^2$	-0.6079z	+0.3928	$-0.1252z^{-1}$	
		+0.8494z	- 0.3928	$+ 0.1252z^{-1}$	
		0.8494z	- 1.5169	$+ 0.9800z^{-1}$	÷
			+ 1.1241	$-0.8548z^{-1}$	0.01
			1.1241		

Continuing in this way will yield successive coefficients for z^{-n} . The time function at the instants of sampling is therefore given by

$$c^{*}(t) = 0.34 \ \delta(t - T) + 0.85 \ \delta(t - 2T) + 1.12 \ \delta(t - 3T) + \dots$$

Using Eq. 10.21 will give the value of $c^*(t)$ as t becomes infinite for a step disturbance in u(t)

$$\lim_{t \to \infty} \{c^*(t)\} = \lim_{z \to 1} \left\{ \frac{0.3404z + 0.2415}{z^2 - 0.7859z + 0.3679} \right\}$$
$$= 1$$

For comparison the time response to a unit step input disturbance for the continuous data process has been computed, and the continuous and the sampled system responses are both shown in Fig. 10.9b. Because the input disturbance is a step function, no approximation error exists at the output from the hold circuit, hence the sampled-data points are coincident with those from the continuous data process. This will be the case whatever value of sampling interval is chosen. Also included in this figure are typical response



Fig. 10.9 Effect of sampling. Example 10.4 (a) block diagram (b) unit step response curves

curves for the same system but with the hold circuit omitted. These show that the system output signal increases in magnitude for reducing sampling interval, as would be expected as a consequence of the sampler output for a unit step input being a train of unit strength impulses (Eq. 10.1).

Example 10.5. For the closed loop arrangement shown in Fig. 10.10a with sampling interval 1 second estimate the process sampled-data time response to a unit step disturbance for 10 sample intervals, with and without the hold circuit present. Calculate the final steady state value of this output response for both cases.

For Fig. 10.7b it has been established that

$$C(z) = \frac{G(z)R(z)}{1 + GH(z)}$$

hence for Fig. 10.10a the closed loop pulsed transfer function is

$$\frac{C(z)}{R(z)} = \frac{G_{\rm ho}G(z)}{1 + G_{\rm ho}G(z)}$$





and in Example 10.4 where this open loop system is analysed it is shown that

$$G_{\rm ho}G(z) = \frac{0.3404z + 0.2415}{z^2 - 0.7859z + 0.3679}$$

The closed loop pulsed transfer function is therefore

$$\frac{C(z)}{R(z)} = \frac{0.3404z + 0.2415}{z^2 - 0.4455z + 0.6094}$$
$$R(z) = \frac{z}{z^2 - 0.4455z + 0.6094}$$

and hence for

212 The Sampled-Data Process

$$C(z) = \frac{0.3404z^2 + 0.2415z}{z^3 - 1.4455z^2 + 1.0549z - 0.6094}$$

Using Eq. 10.21 yields the steady state value

$$\lim_{t \to \infty} \{c^*(t)\} = \frac{0.5819}{1.1639} = 0.5$$

With the hold device omitted from the loop the output is

$$C(z) = \frac{0.5335z^2}{z^3 - 1.2524z^2 + 0.6203z - 0.3679}$$

and the steady state value is

$$\lim_{t \to \infty} \{c^*(t)\} = 0.48$$

For each of these C(z) expressions, the power series expansion method gives the coefficients of the terms $\delta(t - nT)$ and these are listed together with the steady values in Table 10.2. For clarity the system output responses are

Coefficient	1		2	-		,		F		E	
	<i>n</i> 1		2		a		4		,	0	
$c^{*}(t)$ with hold $c^{*}(t)$ without	0.34	07	34	0.7	01	0.4	47	0.3	54	0.467	
hold	0.534	0.6	68	0.5	06	0.4	15	0.4	52	0.494	
Continuous data system	0.739	0.6	65	0.4	61	0.4	28	0.4	93	0.524	
	1		_								
Coefficient n		7		8		9	1	0	sta	Steady ate value	
c*(t) with hold	0.5	574	0.5	553	0.4	478	0.4	458	0.1	5	
hold	0.4	492	0.4	475	0.4	472	0.4	477	0.4	48	
Continuous data system	0.	510	0.4	494	0.4	495	0.9	ō	0.	5	

Table 10.2 Results of Example 10.5, T = 1 second

plotted in Fig. 10.10b. In this figure the continuous output response trace for the sampled system is created by connecting the discrete output sequence by a smooth curve. Such an approximation can be used with confidence if the sampling frequency is substantially higher than the natural frequency of the system. The response curves which correspond to the smaller sampling interval of 0.5 second are also shown. These traces, together with those of Fig. 10.9b, focus attention on the need for adequate data reconstruction in control system design and it is now established practice to introduce a hold circuit after each sampler in a loop to guarantee this end.

10.9 The z-plane

The Laplace transform method is the basic tool for controller design in linear continuous data systems. In a similar manner the z-transform method offers scope for analysis and synthesis of sampled-data systems. A study of the pole-zero configuration of the characteristic equation is one common design approach and requires the facility of being able to transfer root locations in the s-plane to their corresponding positions in the z-plane. This mapping process is achieved by employing the transformation $z = e^{Ts}$, where $T = 2\pi/\omega_s$ is the sampling interval and s the Laplace operator.

If the s-plane is divided into horizontal strips ω_s wide and symmetrical about the real axis, the perimeter of the first of these strips, shown in Fig. 10.11 and referred to as the primary strip, can be mapped into the z-plane as follows. The path from the origin along the positive imaginary axis in the s-plane becomes

$$z = e^{j(2\pi\omega/\omega_s)}$$

= cos 2\pi \omega/\omega_s + j sin (2\pi \omega/\omega_s)

and substituting all values for ω in the range $0 \le \omega \le \frac{1}{2}\omega_s$ gives the locus (1) to (2) in Fig. 10.11b.





214 The Sampled-Data Process

$$z = e^{2\pi/\omega_s (\sigma + (1/2))\omega_s)}$$
$$= -e^{(2\pi/\omega_s)\sigma}$$

- - + Lin

and for values of $0 \le \sigma \le -\infty$ gives values for z in the range -1 to 0, which determines the route from (2) to (3). Continuing in this way around the perimeter enables the route (3)-(4)-(5)-(1) to be mapped, illustrating that this portion of the left half of the s-plane is mapped into a circle of unit radius centred at the origin in the z-plane. Any root lying in the primary strip must lie within the unit circle.

Transferring the boundary of each of the other strips, referred to as complementary strips, from the s-plane to the z-plane generates the same unit circle. Hence, the boundary for stability is the perimeter of this circle and the circle of unit radius in the z-plane is thus equivalent to the imaginary axis in the s-plane. Roots within the circle give rise to decaying motions while the opposite is true for those that lie outside the unit circle.

10.10 Routh-Hurwitz stability test

The conventional procedure described in Section 8.3 was devised to determine whether the roots of a polynomial equation lie in the left or right half of the s-plane. The method must be adapted to permit testing whether roots lie within or outside the unit circle in the z-plane. To achieve this, a modification can be introduced that maps the interior of the unit circle onto the left half of a third complex plane. This transformation is achieved by use of the mapping function

$$r = \frac{z - 1}{z + 1}$$
 10.23

where the complex variable

$$r = \sigma_r + j\omega_r \qquad 10.24$$

If this transformation in r is used, the conventional Routh–Hurwitz test can be applied to the polynomial equation in the variable r.

As was illustrated in Section 10.8 the stability of a sampled data closed loop system is a function of both loop gain K and sampling interval T. The Routh-Hurwitz test is a convenient procedure for locating the stability boundary linking T and K, and the following example illustrates this use.

Example 10.6. Consider the arrangement of Fig. 9.8 modified to include a single sampler and a hold circuit as shown in Fig. 10.12. The open loop



Fig. 10.12 Block diagram for Example 10.6

continuous data transfer function is

$$G(s) = \frac{K}{s^2 + 4s + 8}$$

and the z-transform of the process together with the hold circuit is

$$G_{bo}G(z) = (1 - z^{-1})\mathscr{L}\left\{\frac{K}{s(s^2 + 4s + 8)}\right\}$$
$$= (1 - z^{-1})\mathscr{L}\left\{\frac{K}{8}\left(\frac{1}{s} - \frac{(s + 2)}{(s + 2)^2 + 2^2} - \frac{2}{(s + 2)^2 + 2^2}\right)\right\}$$

Putting $k = \frac{K}{8}$, replacing the groups in 's' by the equivalent groups in 'z' using Table 10.1, and rearranging gives:

$$G_{ho}G(z) = k \left\{ \frac{z e^{-2T} (e^{2T} - \sin 2T - \cos 2T) + e^{-2T} (e^{-2T} + \sin 2T - \cos 2T)}{z^2 - 2z e^{-2T} \cos 2T + e^{-4T}} \right\}$$

By definition the characteristic equation is

$$1 + G_{ho}G(z) = 0$$

i.e. $z^2 + ae^{-2T}z + be^{-2T} = 0$

where

$$a = k(e^{2T} - \sin 2T - \cos 2T) - 2\cos 2T$$
$$b = k(e^{-2T} + \sin 2T - \cos 2T) + e^{-2T}$$

and

Using the mapping function
$$z = \frac{1+r}{1-r}$$
 yields

$$(1 + be^{-2T} - ae^{-2T})r^2 + 2(1 - be^{-2T})r + (1 + ae^{-2T} + be^{-2T}) = 0$$

from which it can be seen that if loop stability is to be achieved

$$1 + he^{-2T} - ae^{-2T} > 0$$

i.e. $k < k_1$

where

$$k_1 = \frac{e^{2T} + e^{-2T} + 2\cos 2T}{e^{2T} - e^{-2T} - 2\sin 2T}$$

and

$$1 - be^{-2T} > 0$$

i.e. $k < k_2$

where

$$k_2 = \frac{e^{2T} - e^{-2T}}{e^{-2T} + \sin 2T - \cos 2T}$$

and

$$1 + ae^{-2T} + be^{-2T} > 0$$

or

$$(e^{2T} + e^{-2T} - 2\cos 2T) > -(e^{2T} + e^{-2T} - 2\cos 2T)$$

i.e. $k > k_3$

where

$$k_3 = -1$$

The profile of the stability region defined for variations in K and T is presented in Fig. 10.13 and this shows very clearly the loss of inherent stability present with the continuous-data system as a direct result of introducing the sample and hold device into the loop.

Time response curves computed using the power series expansion procedure together with Eq. 10.21 to evaluate the steady state value are shown in Fig. 10.14 for a unit step input disturbance. These do confirm the expected change in dynamic behaviour with increasing values of T, as can also be predicted from Fig. 10.13 for the fixed gain value K = 10. The corresponding root locus diagram is given in Fig. 10.15. This figure can be made more informative if the contour lines for constant damping factor ζ are mapped across from the s-plane.





Using Fig. 4.6, it can be seen that a root on a constant damping factor (ζ) line is given by

 $s = -\omega \tan \beta + j\omega$

where $\beta = 90 - \varphi$.

The mapping function from the s-plane to the z-plane is

 $z = e^{(-\omega \tan \beta + j\omega)T}$ = $e^{-\omega T \tan \beta} (\cos \omega T + j \sin \omega T)$

Expressing this in polar coordinates, gives

$$z = e^{-(2\pi \tan \beta)(\omega/\omega_s)}/2\pi(\omega/\omega_s)$$
 10.25

after substitution of $T = 2\pi/\omega_s$. Equation 4.22 gives $\zeta = \cos \varphi$, hence $\zeta = \sin \beta$ and by using Eq. 10.25 the contour lines for constant damping factor can be mapped onto the z-plane. For most practical systems a sampling frequency is chosen such that the forward path transfer function introduces significant attenuation at frequencies higher than $\omega_s/2$, hence only the poles in the strip between $+j\omega_s/2$ and $-j\omega_s/2$ in the s-plane need be mapped across.







Fig. 10.15 Root locus plot—K = 10, T varying

With the contour lines drawn for constant damping factor ζ , Fig. 10.15 indicates the change in transient response to be expected for increasing values of T and does highlight the usefulness of this figure in control system design.

10.11 Frequency response of sampled-data system

All the properties and rules for the continuous-data system are still valid when applied to the open loop pulsed transfer function GH(z), if the bilinear transformation in r is employed. The plot of GH(z) is made in terms of the magnitude and phase as a function of ω_r , as defined by Eq. 10.24.

For example, consider the transfer function of Example 6.2, i.e.

$$G(s) = \frac{10}{s(1+0.5s)(1+0.1s)}$$
 10.26

For the sampled process incorporating zero-order hold

$$G_{\text{ho}}G(z) = (1 - z^{-1})\mathscr{Z}\left(\frac{10}{s^2(1 + 0.5s)(1 + 0.1s)}\right)$$
$$= (1 - z^{-1})\mathscr{Z}\left(\frac{10.0}{s^2} - \frac{6.0}{s} + \frac{2.85}{(1 + 0.5s)} + \frac{0.15}{(1 + 0.1s)}\right)$$

From Table 10.1

$$G_{ho}G(z) = (1 - z^{-1}) \left(\frac{10Tz}{(z - 1)^2} - \frac{6z}{(z - 1)} + \frac{5.7z}{(z - e^{-2T})} + \frac{1.5z}{(z - e^{-10T})} \right)$$

To ensure that the sampling frequency is at least 4 times the bandwidth of the continuous system, choose a value T of 0.3 seconds. Hence, rearrangement of the above equation gives

$$G_{\rm ho}G(z) = \frac{1.2z^3 - 2.91z^2 + 3.86z - 0.86}{(z-1)(z-0.55)(z-0.05)}$$

Employing the bilinear transformation yields

$$G_{\rm ho}G\left(z = \frac{1+j\omega_r}{1-j\omega_r}\right) = \frac{(1.29 - 0.07\omega_r^2) - j\omega_r(8.83\omega_r^2 + 0.59)}{0.86j\omega_r(1+3.44j\omega_r)(1+1.11j\omega_r)} \qquad 10.27$$

If the gain of Eq. 10.26 is reduced by a factor of 5 the polar plot drawn in Fig. 10.16 for this reduced gain value yields a phase margin of 44° and a gain margin of 16 db. These values suggest a marginally acceptable dynamic behaviour from the closed loop system incorporating this process, and this is



Fig. 10.16 Polar plots for continuous system and equivalent sampled system

220 The Sampled-Data Process

confirmed from the step response shown in Fig. 10.17. By reducing the gain of Eq. 10.27 by a factor of 8.2 a phase margin of 45° and a gain margin of 9 db can be achieved. With these values, a step response for the closed loop sampled system should be very similar to that of the continuous system, and this is confirmed in Fig. 10.17.





This result for an arbitrarily selected process demonstrates that the design rules used with the continuous-data system can be extended to the sampleddata system if the bilinear transformation is employed. Hence expertise gained with continuous-data systems can be used in design studies of sampled-data systems.

Similarly, the Bode diagram can be produced and used in the way outlined in Section 6.3 for the continuous-data system.

The pseudo-frequency ω_r can be transformed into the real frequency ω by using the mapping function of Eq. 10.23 and the transform definition $z = e^{sT}$

i.e.
$$j\omega_r = \frac{e^{j\omega T} - 1}{e^{j\omega T} + 1}$$

Using the exponential values of sin θ and cos θ yields

j tan
$$\theta = \frac{e^{j\theta} - e^{-j\theta}}{e^{j\theta} + e^{-j\theta}} = \frac{e^{j2\theta} - 1}{e^{j2\theta} + 1}$$

then by inspection

$$j\omega_r = j \tan \frac{\omega T}{2}$$
$$\omega = \frac{2}{T} \tan^{-1} \omega_r \qquad 10.28$$

and hence

11 Design of Closed Loop Systems

The engineer concerned with the design of a control loop will know from the project specifications the prime objectives which are to be attained, and the requirements for acceptable behaviour depend largely on the applications for which the system is to be designed. The aim of the earlier chapters of this book has been to impart an understanding of the nature of dynamic behaviour, to show the general advantages which can be gained by the use of feedback, and to describe some of the analytical techniques available to the design engineer. This final chapter is concerned with synthesis, where the requirement is one of determining what form of closed loop arrangement is necessary to yield a specific form of output response for a given input excitation.

The first section discusses in general terms the principal ways by which systems can be designed in order to achieve specific performance requirements. There are two aspects, that of choosing a form of controller within the loop, and that of determining suitable parameter values for the controller. Sections 11.2 to 11.4 describe the use of proportional, integral and derivative action; the effect of each is explained and guidance is given on the selection of controller settings. The following three sections describe how performance specifications can be achieved by the use of passive compensation networks. Section 11.8 outlines some other forms of control scheme for continuous systems. This is followed by a section to discuss the design of sampled-data systems and digital controllers, and a section to introduce the ideas of state vector feedback control. Section 11.11 describes relay control, a form of control action which is widely used because of its simplicity and hence low cost, and which introduces non-linearity into the feedback loop and thus requires techniques that extend the design methods described earlier. The chapter concludes with a case study illustrating the application to a specific system of the many techniques presented in this book. It describes, in outline, the derivation of a suitable mathematical model for the physical system, the analysis of the dynamic behaviour, and the design of appropriate compensation for the feedback loop.

11.1 The general approach to design

The requirements for a system will be described by some appropriate performance specification, expressed either as time domain requirements (defining the transient and steady state response for a step change or other forcing function) or as frequency domain requirements (phase margin, gain margin, bandwidth, peak magnification, etc.). Although there is no direct analytical relationship between the two sets of performance characteristics for systems of order higher than two there is a broad correspondence between them. A system designed with a value of $M_p = 1.3$ should have a relatively small rise time without large overshoot when subjected to a step input.

There are two basic approaches to design. The older, but still very widely used one is an orderly trial and observation intuitive approach aimed at finding an acceptable, but not necessarily the best possible, design solution. Based on past experience a control loop configuration and form of controller is chosen at the outset and one or more of the analytical techniques described earlier is used to try to determine controller parameters which allow the system to meet the specifications. If the chosen arrangement is not satisfactory then other forms of controller must be investigated. The more complex the controller the greater the number of parameters which can be varied. Previous experience helps in obtaining a satisfactory solution within a reasonable time and within the economic constraints that normally exist in a design project. The alternative approach is one of true synthesis, where an attempt is made to determine a unique solution in accord with a rigidly defined specification in some optimal way. To aim to achieve this is attractive in principle, but it requires a sound knowledge of the analytical methods of modern control theory. Also, the cost of the equipment necessary to implement the resulting control law may prove to be difficult to justify economically, on the grounds of dynamic performance alone.

The simplest and most widely used arrangement is *series compensation* where the controller or compensation device is positioned in the forward loop as shown in Fig. 11.1a. *Parallel* or *feedback compensation* by means of a



Fig. 11.1 Control loop configurations (a) series or cascade compensation (b) feedback compensation (c) feedforward compensation (d) state feedback control

subsidiary feedback loop (Fig. 11.1b) offers certain advantages; it may also be used in conjunction with series compensation. The next section collects together ideas presented earlier for studying the response of a closed loop system where the loop gain can be adjusted but where otherwise the dynamic

224 Design of Closed Loop Systems

characteristics are fixed. Such an arrangement with a series controller which is simply a gain element gives what is termed proportional control action. Where no suitable value of gain can be found to achieve the specification, additional loop elements are needed. Section 11.3 looks at the use of integral action and derivative action, and describes how these modify the response and how the coefficients in the resulting 2-term or 3-term controller may be chosen. Section 11.4 considers the practical side of control by introducing empirical methods used by engineers for selecting the correct amount of control action necessary to achieve a required performance on existing plant. The topic of system compensation is then discussed; this is a procedure whereby additional electrical networks or mechanical elements are incorporated within the loop to modify the system behaviour in such a way that it more nearly satisfies the required specification. As an example to illustrate the general features of system compensation, Section 11.6 considers in some detail one of these additional control actions, phase lead series compensation, and the next section discusses some of the differing characteristics of other compensation methods.

The steady state accuracy requirement dictates the form of open loop transfer function and the value of loop gain needed, as described in Section 8.2. Design in the time domain to achieve specific transient response characteristics is facilitated by studies of root locus plots for the system. The aim is to select physically achievable numerical values for the system parameters in such a way that the system poles are placed in suitable regions of the *s*-plane. The dominant poles are the most critical, and considerations of settling time and maximum overshoot in response to a step change dictate the area which should give acceptable behaviour as indicated in Fig. 11.2. The response is, however, influenced both by secondary poles and by any system zeros which are present, and hence root locus studies must be supplemented by simulation studies to confirm that the specifications are satisfied.

Where performance specifications are given as frequency domain characteristics design will employ the analytical methods of Chapter 6. Although they are largely trial and error methods and the earliest ones devised these remain important since they have a number of advantages. They are easy to understand and to apply, they are independent of the order of the system transfer function, and indeed do not require the mathematical model to be known in transfer function form since experimentally obtained harmonic response data is equally acceptable. The Bode diagram is the most useful, with the Nichols chart being used to relate open loop and closed loop data when the system is a simple unity feedback system.

11.2 Proportional control

Consider the control of the output of a process with transfer function G(s) which utilizes feedback principles, and assume that the process characteristics cannot be altered by the designer (for example the speed regulating system shown in Fig. 9.2a). The simplest form of control is one in which the error signal e(t) is multiplied by a constant k_1 to yield a signal called the *manipulated variable* m(t) which is the input to the process; the numerical value of this constant k_1 determines the amount of corrective effort which is applied for a



Fig. 11.2 Acceptable region for dominant roots

given magnitude of error. This arrangement is called *proportional control*, and by varying the value of k_1 , the dynamic behaviour of the overall system can be altered. For very low values of k_1 the corrective effort is small, and hence the response is likely to be sluggish (Fig. 9.2c); as k_1 increases the response of the system for the same magnitude of error becomes more rapid and, if k_1 is very large, instability is likely to result, or the oscillatory response would be so lightly damped that it would be unsatisfactory for all practical purposes. The significant variable is actually the loop gain K_p , which is the product of k_1 and the steady state gain of the process.

If G(s) has no poles at the origin of the complex s-plane, then the overall system will always have a steady state positional error. This error, as shown in Section 8.2, is proportional to $\frac{1}{1+K_p}$, hence it can be reduced by increasing the loop gain K_p , and can be evaluated by application of the final value theorem for a constant input excitation. Increase in loop gain, however, causes the dominant complex roots (Fig. 9.2b) to move closer to the imaginary axis and to the instability associated with root positions in the right half of the complex s-plane. Provided a value of loop gain K_p can be chosen which gives both an acceptable transient response and a small enough steady state error then the design problem is solved. If these requirements cannot be satisfied simultaneously then the loop must be modified by the inclusion of some other form of control action, of a compensation network, or of a subsidiary feedback

226 Design of Closed Loop Systems

loop. If, however, G(s) does have a pole at the origin (a factor s in the denominator) or if additional control action is included to introduce a pole at the origin of the root locus diagram, this reduces the positional error to zero and reduces the velocity error to a value which is inversely proportional to the loop gain. Similarly, a double pole at the origin would yield a positional error and a velocity error which are both zero, and an acceleration error which is inversely proportional to loop gain (Section 8.2).

Example 11.1. For the feedback system shown in Fig. 9.2a, can a suitable value of loop gain be found?

At the limit of stability, the positional error is at the lowest value that it can attain, namely $\frac{1}{1+315/25} = 0.073$, 7%; this is at the limit of what would be acceptable for a simple speed regulating system. A value of $K_p = 35$ gives a suitable transient response with damping factor $\zeta = 0.64$ for the dominant roots; however, the steady state error is excessive at a value of 42%. For this simple speed control system the design engineer would have to think seriously about the introduction of additional control action.

11.3 Integral and derivative action

(a) Integral action. A prime requirement of many control systems is that there should be no error or at worst a very small error in the steady state. It was shown in Section 8.2 that for a type 0 system, one with no factors of s in the denominator of the transfer function, a steady state error always exists for a steady input. This error can be decreased at the expense of a more oscillatory response by an increase in gain, but it may not be possible to attain simultaneously satisfactory steady state and dynamic behaviour (as discussed in Section 11.2). Zero steady state positional error would require the system to be of Type 1, and this can be achieved by introducing integral action within the controller. To the proportional term is added a signal proportional to the time integral of the error; i.e. the controller output m(t) is $[k_1e(t) + k_2[e(t)dt]]$ and it is this signal which actuates the system. The block diagram is then of the form shown in Fig. 11.3. Since the error signal is integrated within the control-



Fig. 11.3 Feedback system with P + I control action

ler, even the smallest error eventually produces a corrective signal of sufficient magnitude to actuate the system to eliminate the error. The system will, theoretically, only come to rest when the error has been reduced to zero.

Consider analytically the effect of integral action on the system of Fig. 11.3 when the process and measuring transducer have transfer function

$$G_1(s)G_2(s) = \frac{1}{(s^2 + 4s + 8)(1 + s)}$$

With proportional control $(k_2 = 0)$ the closed loop transfer function is

$$\frac{C(s)}{R(s)} = \frac{k_1}{s^3 + 5s^2 + 12s + 8 + k_1}$$
11.1

Hence the steady state error for a unit input is

$$1 - \lim_{s \to 0} s \left[\frac{1}{s} \cdot \frac{k_1}{s^3 + 5s^2 + 12s + 8 + k_1} \right] = \frac{8}{8 + k_1}$$

With proportional plus integral (P + I) control

$$\frac{C(s)}{R(s)} = \frac{k_1 s + k_2}{s^4 + 5s^3 + 12s^2 + (8 + k_1)s + k_2}$$
11.2

and the steady state error for a unit input is

$$1 - \lim_{s \to 0} s \left[\frac{1}{s} \frac{k_1 s + k_2}{s^4 + 5s^3 + 12s^2 + (8 + k_1)s + k_2} \right] = 0$$

One method by which numerical values for the parameters k_1 and k_2 can be selected is by use of a root locus or root contour plot. If a performance index based on the transient response characteristics, such as rise time, initial overshoot, etc. as described in Section 4.4 is used, then the roots on the dominant loci must lie in some region such as that bounded by the lines drawn for a damping factor of $\zeta = 0.4$ and $\zeta = 0.5$ (corresponding to transient responses shown in Fig. 4.7). This is the hatched region shown in Fig. 11.4, the root contour plot for the system with P + I control. Inspection of this figure shows that values of $k_1 = 8$ and $k_2 = 11$ position the dominant roots in the centre of the hatched area; hence these values of k_1 and k_2 are likely to give the desired transient response. A simulation study would now be made to select better values for k_1 and k_2 that would more nearly give the response shown in Fig. 4.7 for $\zeta = 0.5$, since further adjustment of both these parameters may be necessary to make allowance for the effect of the secondary locus shown in Fig. 11.4 and for the zero in Eq. 11.2. Because of this zero, the root locus pattern cannot be used in isolation during a design study, but must be supplemented by simulation data. The transient response for the parameters given above is shown in Fig. 11.7b. With P + I, the response shown for this typical speed control system would in engineering practice normally be considered 'satisfactory', having a maximum overshoot of 23%, a rise time of 1.5 seconds, and a settling time of 6 seconds. For the process alone, the overshoot would be zero, and the rise time and settling time would each be around 3 to 4 seconds.



Fig. 11.4 Root locus diagram for variation of proportional action (k_1) and integral action (k_2) . $G_1(s)G_2(s) = \frac{1}{(s^2 + 4s + 8)(s + 1)}$

(b) Derivative action. A form of control action which can increase the effective damping is derivative action; this is not used by itself but in conjunction with proportional or proportional plus integral action. To the normal error signal is added a signal proportional to its derivative, giving a 2-term or 3-term controller (Fig. 11.5). The 3-term controller has a transfer function $G_c(s) = (k_1 + k_2/s + k_3s)$. Alternatively, this is often expressed as

$$G_e(s) = k_e \left(1 + \frac{1}{T_i s} + T_d s \right)$$

in which $k_1 = k_c$, $k_2 = k_c/T_i$ and $k_3 = k_cT_d$. The derivative term contributes an anticipatory type of control action, where the output of the controller is modified when the error is changing rapidly, thus anticipating a large overshoot and making some correction before it occurs. When a system is moving







Fig. 11.6 Root locus diagram for variation of integral action (k_2) and derivative action (k_3) . $G_1(s)G_2(s) = \frac{1}{(s^2 + 4s + 8)(s + 1)}$



Fig. 11.7 Step responses for system shown in Fig. 11.3 and Fig. 11.5: $G_1(s)G_2(s) = \frac{1}{(s^2 + 4s + 8)(s + 1)}$ (a) P controller (b) P + I controller (c) P + I + D controller

towards a state of zero error e(t), Fig. 11.7b, then e(t) and $\dot{e}(t)$ have opposite signs; hence the derivative term reduces the magnitude of m(t) and thus reduces the signal that is accelerating the output response c(t) towards the zero error state. When the output response has overshot or undershot and is moving away from the zero error condition, then e(t) and $\dot{e}(t)$ have the same sign and the derivative term augments the decelerating signal. When the system comes to rest then $\dot{e}(t)$ is zero and the derivative term has no further influence.

The effect of derivative action on the position of the roots of the characteristic equation can be seen in Fig. 11.6, which shows that the addition of derivative action has as expected improved the relative stability of the system. This is always a highly desirable feature in the design of a control system, since any change in the values of plant parameters over a period of time is less likely to cause the system to drift into instability. The effect on the dynamic behaviour resulting from the additional closed loop zeros must, however, be investigated before any decision on the suitability of, or the need for, derivative action can be taken with confidence. Inspection of Fig. 11.6 suggests that a useful value for k_3 to reduce the overshoot, with minimum effect on the other dynamic characteristics, might be 0.9. However, a simulation study, Fig. 11.7c, indicates that $k_3 = 2.5$ would be a better choice. This required change in the value of k_3 is the direct result of the presence of the closed loop zeros. The poles alone dictate the stability boundary, but both the poles and zeros contribute to the dynamic behaviour of the closed loop system.

(c) Rate feedback or negative velocity feedback. To avoid mechanical failure or system malfunctioning, it is essential in all engineering situations to safeguard against the occurrence of large initial transient overshoots. The suppression of these overshoots must not, in general, be at the expense of the system accuracy; therefore it becomes necessary in most designs to introduce additional control action to prevent this. An action similar to that of proportional plus derivative control can be achieved by incorporating within the control loop a minor feedback path which introduces control action known as velocity feedback. The block diagram, with velocity feedback included, is then as shown in Fig. 11.8, from which it can be seen that a signal proportional to the derivative of the output rather than the derivative of the error signal is used. The numerical value of these two derivatives will be the same for all unity feedback systems except when adjustment is being made to the set point r(t). The change in the closed loop transfer function can be seen from the following two equations:

$$\frac{C(s)}{R(s)} = \frac{k_1 + k_3 s}{s^3 + 5s^2 + (12 + k_3)s + (8 + k_1)}$$
11.3

for the system considered earlier with proportional plus derivative control, and

$$\frac{C(s)}{R(s)} = \frac{k_1}{s^3 + 5s^2 + (12 + k_4)s + (8 + k_1)}$$
11.4

for the system incorporating negative velocity feedback. In practice, Eq. 11.3 would be modified as a result of a small additional time constant associated with the differentiation, necessary to avoid problems of noise amplification and of saturation with step changes of error. Fig. 11.9 shows that the use of



Fig. 11.8 System with negative velocity feedback



Fig. 11.9 Step responses for change in control action: $G_{1}(s)G_{2}(s) = \frac{1}{12}$ (a) P controller (b) P control

 $G_1(s)G_2(s) = \frac{1}{(s^2 + 4s + 8)(s + 1)}$ (a) P controller (b) P controller and negative velocity feedback (c) P + D controller

negative velocity feedback instead of derivative action can give a smaller initial overshoot but would do so at the expense of an increase in rise time.

Velocity feedback is chiefly of advantage for servomechanisms in which the output velocity can be measured directly by a tachogenerator or velocity transducer, thus avoiding the problem of noise amplification that can arise when trying to differentiate an error signal. It is always good engineering practice to avoid the use of differentiation, if at all possible, when choosing the type of control action to be incorporated in a design.

11.4 Selecting controller settings on existing process plant

As a result of empirical tests on a wide variety of process plant, Ziegler and Nichols (see Ref. 1, p. 278) propose a simple rule of thumb procedure for estimating the values of controller settings k_1 , k_2 , and k_3 for existing operating plant in order to achieve an optimum transient response. There are two methods, one based on the step response of the open loop system and the other based on information obtained at the stability limit of the process under proportional control.

In the first method, with the loop opened, the plant is subjected to a step change of manipulated variable and the resulting output response curve is characterized by two measured parameters N and L, shown in Fig. 11.10. N is the maximum slope of the curve for a change M of manipulated variable, and L is the time at which the line of maximum slope intersects the time axis. The recommendations which Ziegler and Nichols put forward for the controller settings are:

$$k_{c} = \frac{M}{NL} \text{ for P control}$$

$$k_{c} = 0.9 \frac{M}{NL}, T_{i} = 3.3L \text{ for P + I control}$$

$$k_{c} = 1.2 \frac{M}{NL}, T_{i} = 2L, T_{d} = 0.5L \text{ for P + I + D control}$$

$$11.5$$



Fig. 11.10 Open loop response parameters for Ziegler-Nichols first method

where k_c , T_i , and T_d are the parameter values of controller gain, integral action time and derivative action time respectively, as they appear in the control law

$$G_{\rm c}(s) = k_{\rm c} \left(1 + \frac{1}{T_{\rm i} s} + T_{\rm d} s \right)$$
 11.6

The procedure of the second method is to determine experimentally the limiting condition of stability of the closed loop system under proportional control only, and to use the resulting information to calculate controller settings. If the limiting value of gain for stability is $k_{\rm crit}$ and the time period of oscillation is $P_{\rm crit}$, Fig. 11.11, then the Ziegler-Nichols recommended controller settings are:

$$k_{c} = 0.5 k_{crit} \text{ for P control}$$

$$k_{c} = 0.45 k_{crit}, T_{i} = 0.83 P_{crit} \text{ for P + I control}$$

$$k_{r} = 0.6 k_{crit}, T_{i} = 0.5 P_{crit}, T_{d} = 0.125 P_{crit} \text{ for P + I + D control}$$

$$11.7$$

Other workers (see Ref. 2, p. 278) have extended these ideas to show how the commonly used analogue controllers can be replaced by their digital equivalents, and how the analogue settings of Ziegler–Nichols and others can be translated into settings for digital controllers, to achieve satisfactory loop tuning. Most controller manufacturers also provide more specific instruction for the adjustment of their instruments based on these ideas. The settings obtained using these instructions, however, only give a good first estimate and



Fig. 11.11 Oscillatory response parameters for Ziegler-Nichols second method

further adjustment is still necessary to meet the control requirements of specific plant.

11.5 System compensation

Most feedback systems are required simultaneously to match performance specifications both for steady state accuracy and for relative stability. The former requires that the steady state error with a given type of input excitation should not exceed some specified value, and this defines a certain minimum value of loop gain, say K_1 . To ensure that the system has adequate relative stability, it must have a specified minimum value of phase margin or gain margin, or a specified maximum value of M_p , and this defines a certain maximum value of loop gain, say K_2 . If $K_1 > K_2$, then the two requirements are not compatible, and the specifications cannot both be satisfied unless some form of *phase compensation* is introduced.

The way in which the addition of compensating networks can result in compatibility can be illustrated by considering the Nyquist diagram, Fig. 11.12. The general objective is one of reshaping the open loop harmonic response plot so that the low frequency gain is high enough and, in addition, the plot avoids the critical (-1, j0) point with an adequate safety margin. The locus can be reshaped in the manner shown by

- (i) starting with gain K_1 for the system and introducing phase lead at high frequencies in order to attain the specified phase margin, gain margin, or M_p (phase lead compensation), or
- (ii) starting with gain K_2 for the system and introducing phase lag at low frequencies to meet steady state accuracy requirements (*phase lag compensation*) or
- (iii) starting with a gain between K_1 and K_2 and introducing some phase lead at high frequencies and some phase lag at low frequencies (*lag-lead* compensation).

The design task of determining the transfer function required for a suitable compensating network can be carried out in the time domain or in the frequency domain using one or more of the techniques of analysis described in



Fig. 11.12 Nyquist diagram for $G(s) = \frac{h}{s(s^2 + 4s + 8)}$ illustrating principle of compensation

earlier chapters. Design by means of the Bode plot is particularly useful because of the ease with which the effect of compensation can be evaluated by noting the improvement to stability margins resulting from the addition of magnitude and phase curves for the compensating network being investigated. The design procedures are orderly graphical trial and error procedures, and although this approach is among the earliest of design techniques it is probably still the best for systems with one or two feedback loops. The time domain characteristics are not directly apparent from the Bode plot but are related in a general way to the shape of the plot in the vicinity of the critical 0 dB, -180 degree point. Although a phase margin of 45°, with a gain margin of around 6 to 8dB, provides no guarantee that the transient behaviour will be acceptable, conditions are not often encountered where it is not. The procedure is usually to establish a tentative design with the aid of a Bode plot, then to simulate the resulting system and, by trial and observation, make adjustments if necessary to achieve an appropriate transient response based on rise time, initial overshoot, etc. The design technique gives no clear guidance at the outset as to which type of compensation would be the best for any given system; hence the designer must use past experience to decide on a method, try it, and if no satisfactory design emerges try another.

11.6 Phase lead series compensation

This section outlines the general features of phase lead series compensation in order to illustrate the procedure of designing a compensation network. The approach with phase lag, lag-lead, or parallel compensation is broadly similar, although the details vary. The phase lead is provided by a compensating device which has a transfer function

$$G_c(s) = \frac{1}{\alpha} \frac{1 + \alpha T s}{1 + T s}$$
, where $\alpha > 1$.

It should be noted that with a passive device the phase lead provided by the zero at $-1/\alpha T$ cannot be obtained without including in addition a pole at -1/T. Inherent also is the attenuation $1/\alpha$; hence, when used for compensation, additional amplification is needed to restore the loop gain to the required value.



Fig. 11.13 Phase lead compensation networks. (a) electrical (b) mechanical

Such a transfer function is valid for the electrical circuit of Fig. 11.13a. That this is so with $\alpha = \frac{R_1 + R_2}{2}$, $T = \frac{R_1 R_2 C}{R_1 + R_2}$, can be shown by the analytical methods of Chapter 2, provided assumptions are made that any input impedance presented to the circuit will be small, and that the circuit will always be connected to a high impedance output, thus maintaining the input-output properties of an open circuit. The mechanical arrangement sketched in Fig. 11.13b can be shown to be analogous, with $\alpha = (K_1 + K_2)/K_2$, and $T = C/(K_1 + K_2)$. The amount of phase lead, and the frequency band where it is effective, can be selected by appropriate choice of circuit components to obtain the required values of α and T.

What are the characteristics of such a phase lead arrangement? In the s-plane a pole and a zero are introduced on the negative real axis with the zero

lying closer to the imaginary axis than the pole (Fig. 11.14a); the position and spacing of the roots is determined by the numerical values of α and T. The effect on a root locus diagram of introducing such a pair of roots, with the zero dominant, is to pull the dominant loci towards the left of the diagram and, as a consequence, improve the relative stability of the system. The extent to which the dominant loci are moved to the left of the diagram will be dictated by the values of α and T chosen at the design stage. On the polar plot,

the locus of the unity gain function $\frac{1 + j\omega\alpha T}{1 + j\omega T}$ can readily be shown to be a

semicircle in the 4th quadrant with a magnitude of 1 at low frequencies, rising to a maximum value of α at high frequencies (Fig. 11.14b). The phase is a lead which increases with increase of frequency from zero to a maximum value determined by the chosen value of α , and then reduces to zero again as the frequency approaches infinity. The tangent drawn to this semicircle from the origin determines the maximum phase lead φ_m that is obtainable, and the frequency ω_m at which it occurs. The angle φ_m increases with increase of α , and tends towards a maximum lead of 90° as α approaches infinity. On the Bode plot (Fig. 11.14c), the lead term has a break point at $\omega = 1/\alpha T$ and thus starts to influence the response at about one tenth of this frequency; as frequency increases the effect of the lead term grows but is gradually opposed by the lag term which has its break point at $\omega = 1/T$. As a consequence of symmetry of the phase plot, the frequency ω_m at which the maximum phase lead occurs lies midway between the corner frequencies on the logarithmically scaled frequency axis, hence

$$\log_{10} \omega_{\rm m} = \frac{1}{2} \left(\log_{10} \frac{1}{\alpha T} + \log_{10} \frac{1}{T} \right)$$
$$\omega_{\rm m} = \frac{1}{T \sqrt{\alpha}}$$

or

The maximum value of phase lead is given by

$$\varphi_{m} = \tan^{-1} \omega_{m} \alpha T - \tan^{-1} \omega_{m}$$
$$\omega_{m} \alpha T - \omega_{m} T$$

$$\tan \varphi_{\rm m} = \frac{1}{1 + (\omega_{\rm m} \alpha T)(\omega_{\rm m} T)}$$

$$\omega_{\rm m}=\frac{1}{T\sqrt{\alpha}},$$

Substituting

$$4\pi \phi_m = 2\sqrt{\alpha}$$

or more conveniently

$$\sin \varphi_{\rm m} = \frac{\alpha - 1}{\alpha + 1}$$
 11.8

This equation is used to calculate the value of α needed to provide any specific value of phase lead φ_m .




The design procedure using the Bode diagram is as follows:

1. Plot the Bode diagram for the uncompensated system with the gain chosen to achieve the steady state error requirements.

2. Read from the plot the phase margin, and estimate the phase lead required to give acceptable system response. To make allowance for the increase in gain crossover frequency, caused by the magnitude contribution of the compensation network, approximately 5 degrees should be added to determine a trial design value for $\varphi_{\rm m}$. Calculate the required value of α using Eq. 11.8.

3. To ensure that φ_m is located at the new gain crossover frequency, calculate the high frequency magnification of the function $\frac{1+j\omega\alpha T}{1+j\omega T}$, find the frequency at which the uncompensated system has an attenuation of half this value, and make ω_m equal to that frequency. Hence $T = \frac{1}{\omega_m \sqrt{\alpha}}$, and the break points for the compensation network are at frequencies $\omega_m/\sqrt{\alpha}$ and $\omega_m\sqrt{\alpha}$.

4. Increase the gain by a factor α , to correct for the attenuation inherent in a

practical compensation network, and draw the Bode diagram for the compensated system.

5. Check that the required performance specifications are satisfied and, if not, select a larger value of φ_m and repeat the procedure.

Example 11.2. A unity feedback system with open loop transfer function $\frac{K}{(1 + 10s)(1 + s)}$ is required to have a phase margin of at least 45 degrees and a

steady state positional error which does not exceed 1%. Determine the required value of K and, if the performance specifications cannot be satisfied by appropriate choice of K, design suitable phase lead series compensation for the loop.

The value of K must be chosen to satisfy the steady state error requirements. For a unit step input the steady state error is given by

$$e_{ss} = 0.01 = \lim_{s \to 0} sE(s) = \lim_{s \to 0} 0 \left[\frac{1}{1 + \frac{K}{(1 + 10s)(1 + s)}} \right]$$

 $K = 100, \, say.$

The Bode diagram for the uncompensated system is drawn by summing the contributions to magnitude and phase of the gain term, and of the two simple lag terms which have break points at frequencies of 0.1 and 1 rad/second:

Magnitude (dB) =
$$20 \log_{10} 100 - 20 \log_{10} \sqrt{(1 + (10\omega)^2) - 20 \log_{10} \sqrt{(1 + \omega^2)}}$$

Phase = $-\tan^{-1} 10\omega - \tan^{-1} \omega$.

From the Bode diagram (Fig. 11.15) the phase margin is measured to be 19°; hence at least 26° of phase lead is required. To make some allowance for phase



Fig. 11.15 Phase lead compensation—Bode diagram for Example 11.2.

reduction resulting from increase of the gain crossover frequency, choose a trial value of $\phi_m = 30^\circ$. Applying Eq. 11.8:

$$\sin 30^\circ = \frac{\alpha - 1}{\alpha + 1} = 0.5$$
$$\therefore \quad \alpha = 3$$

The transfer function of the phase lead network is therefore

$$G_c(s) = \frac{1}{3} \frac{1+3Ts}{1+Ts}$$

and the value of T must now be determined to ensure that the phase lead of 30° is effective at the new gain crossover frequency. At high frequencies, the

magnitude of the transfer function $\frac{1+3Ts}{1+Ts}$ is

 $20 \log_{10} 3 = 9.6 \text{ dB}.$

To ensure that φ_m is effective at the new gain crossover frequency, place ω_m at the frequency where the uncompensated system magnitude is -0.5 (9.6) = -4.8 dB. From Fig. 11.15 this is seen to give $\omega_m = 4.2$ rad/second. Hence the corner frequencies for the compensation network are

$$\frac{1}{\alpha T} = \frac{\omega_{\rm m}}{\sqrt{3}} = 2.4$$
 rad/second, and $\frac{1}{T} = \omega_{\rm m}\sqrt{3} = 7.3$ rad/second

Hence the transfer functions for the compensation network and the system are respectively

$$G_c(s) = \frac{1}{3} \frac{1+0.42s}{1+0.14s}$$
 and $G(s) = \frac{300}{(1+10s)(1+s)}$

Addition of the magnitude and phase curves for the compensation network and the uncompensated system in Fig. 11.15 confirms that this compensation network increases the phase margin to 45° , which is the value required. The gain margin is infinite since the system is of second order.

The improvement to the transient behaviour resulting from compensation can be seen in the step response traces of Fig. 11.16.

This example illustrates that phase lead compensation generally improves the rise time and reduces the amplitude of transient oscillations, but increases the bandwidth, and so may introduce undesirable effects resulting from noise transmission through the system. The compensation network acts as a high pass filter. Much noise is, however, usually suppressed due to the low pass characteristics of most physical systems. It should be noted that by no means all systems can be satisfactorily compensated by means of a phase lead network. A common situation in which the design method fails is where the transfer function of the uncompensated system is of a form which causes the phase lag to increase rapidly near the gain crossover frequency; any phase lead added is then nullified to a large extent by the marked decrease in phase of the uncompensated system at the new gain crossover frequency. The designer may also be constrained by the physical nature of components to be used, or limited by factors such as cost, weight, and space. It may not be possible, because of engineering difficulties, to achieve the large gain required to meet the steady state accuracy specification. Phase lead compensation is not advisable where the system gain must be so high that the uncompensated system is badly unstable.

11.7 Phase lag, and lag-lead series compensation

A phase lag compensation element has the transfer function

$$G_{c}(s) = \frac{1 + \alpha T s}{1 + T s}$$
, where $\alpha < 1$



Fig. 11.16 Effect of phase lead compensation (a) root locus diagram (b) step responses

This can be realized by the passive electrical network of Fig. 11.17a provided that the input impedance is small and the output impedance is large; $\alpha = R_2/(R_1 + R_2)$, which is less than unity, and $T = C(R_1 + R_2)$. A spring and damper arrangement as in Fig. 11.17b is equivalent, with $\alpha = K_1/(K_1 + K_2)$ and $T = C_2(K_1 + K_2)/K_1K_2$. At low frequencies the phase lag element has unity gain and at high frequencies there is an attenuation α . Introduction of such an element adds a real pole and a real zero in the s-plane, the pole being dominant and their position and spacing being determined by the numerical



Fig. 11.17 Phase lag compensation networks (a) electrical (b) mechanical



Fig. 11.18 Characteristics of phase lag network $1 + \alpha Ts/1 + Ts$, $\alpha < 1$ (a) root locus plot (b) polar plot (c) Bode plot

values of α and T. The dominant loci of the root locus plot are pushed towards the right, but the values of K for the compensated system are lower giving improved relative stability. The characteristics of a phase lag element, shown in Fig. 11.18, are the inverse of those for a phase lead element. Design of phase lag compensation does not rely on the phase shift of the network, rather it utilizes the characteristic of attenuation at high frequencies.

As with phase lead compensation the starting point in the design process is the uncompensated system with the value of the steady state gain chosen to meet the steady state error requirements. The phase curve of the Bode plot in the region of the gain crossover frequency is kept relatively unchanged and the magnitude is decreased in this region (and at higher frequencies) by means of the lag network so that the gain crossover frequency decreases to give an improved phase margin. (This contrasts with phase lead compensation where the magnitude curve is kept relatively unchanged in the region of the gain crossover frequency and phase lead is introduced to increase the phase margin, making due allowance for the effect on phase margin of the accompanying increase in gain crossover frequency). The design procedure using the Bode diagram is as follows:

1. Plot the Bode diagram for the uncompensated system with the gain chosen to achieve the steady state error requirements.

2. Find the frequency ω_c corresponding to the desired phase margin plus about 5°. The magnitude plot must be altered so that it passes through 0 dB near this frequency. Measure the amplitude G_c of the uncompensated system at this frequency ω_{e} . The phase lag network must provide an attenuation of G_c .

3. Calculate the value of α to provide this attenuation from the relationship $G_c = -20 \log_{10} \alpha \, dB$. Select T so that the phase lag effect is well below ω_c . say $1/\alpha T = 0.1 \omega_e$, in which case the phase lag introduced by the network at the gain crossover frequency is about 5°, which has already been allowed for in step 2.

4. From the plots for the uncompensated system and the compensation network with α and T chosen in step 3 draw the Bode plot for the compensated system.

Check that the required performance specifications are satisfied and if not adjust the selected values to try to attain them.

Example 11.3. A unity feedback system with open loop transfer function

 $\frac{1}{(1+10s)(1+s)(1+0.5s)}$ is required to have a phase margin of at least 45° and

a steady state positional error which does not exceed 1%. Determine the required value of K and design suitable phase compensation for the loop. (Note that this is the system of Example 11.2 with one additional time constant of 0.5 seconds).

The value of K must be chosen to satisfy the steady state requirements, and remains identical to that for Example 11.2, i.e. K = 100. The Bode diagram for the uncompensated system is drawn, as in the previous example, by summing the contributions to the magnitude and phase of the constituent factors of the transfer function:

Magnitude (dB) = $20 \log_{10} 100 - 20 \log_{10} \sqrt{(1 + (10\omega)^2)}$ $-20 \log_{10} \sqrt{(1+\omega^2)} - 20 \log_{10} \sqrt{(1+(0.5\omega)^2)}$ Phase = $-\tan^{-1}(10\omega) - \tan^{-1}\omega - \tan^{-1}(0.5\omega)$

The magnitude and phase can be easily calculated with the aid of a hand calculator. The gain margin and phase margin are both negative $(-9 \text{ dB} \text{ and } -23^\circ, \text{ Fig. 11.19})$ indicating that the uncompensated system with the gain set to achieve the required steady state error is unstable. It can readily be shown by applying the Routh-Hurwitz criterion that the limiting value of K before instability occurs is 34.65.

Phase lead compensation is unlikely to succeed in meeting the stability requirements since the phase in the region of the gain crossover frequency is decreasing rapidly with increase in frequency.

Investigate therefore the benefits of phase lag compensation. The phase is $(180^{\circ} - 50^{\circ}) = 130^{\circ}$ for the frequency $\omega_c = 0.62$ rad/second. The magnitude at this frequency is found from the plot to be 22 dB.

$$\therefore 20 \log_{10} \alpha = -22 \qquad \therefore \quad \alpha = 0.08$$

also $\frac{1}{\alpha T} = 0.1 \ (0.62) \qquad \therefore \quad T = 202 \text{ second}$

A suitable phase lag compensation network should therefore be

$$G_c(s) = \frac{1+16s}{1+202s}$$

which has magnitude and phase characteristics as shown in Fig. 11.19. Addition of these and the plots for the uncompensated system gives a plot for the



Fig. 11.19 Phase lag compensation—Bode diagram for Example 11.3



Fig. 11.20 Effect of phase lag compensation (a) root locus diagram (b) step response curves

compensated system, and from this the phase margin is found to be 45° as required and the gain margin is now 13 dB.

The resulting step response can be seen in Fig. 11.20b. The change to the root locus diagram, and to the position of the roots can be seen in Fig. 11.20a—the dominant complex roots are brought from the right half of the *s*-plane to a position corresponding to $\zeta = 0.36$ and $\omega_n = 0.82$ rad/second at the expense of introducing a real root close to the origin, which makes the response sluggish.

By replotting the harmonic response characteristics on a Nichols chart one can find the closed loop characteristics to be $M_p = 1.4$, $\omega_p = 0.7$ rad/second and bandwidth = 1.1 rad/second.

Study of the root locus plot suggests how this may be used for the design of phase lag compensation. Establish on the root locus for the uncompensated system a desirable root position, compare the value of K there with that required to satisfy steady state requirements, and call the ratio α . Choose $1/\alpha T$ to be one order of magnitude at least smaller than the smallest pole of the uncompensated system; the exact location of the additional pole and zero is not critical, only the distance between them.

This study illustrates that phase lag compensation improves the relative stability and reduces overshoot but usually at the expense of a longer rise time since the values of ω_p and bandwidth are decreased. This contrasts with phase lead compensation where the bandwidth is increased thus improving rise time, and perhaps giving problems of noise transmission. Each method has advantages and disadvantages, and where a system cannot be designed to satisfy several requirements simultaneously by using one or the other the desired performance may be attainable by using the two together to gain advantages from each one. This is referred to as *lag-lead compensation*, and can be achieved either by use of separate lag and lead circuits in series, with a buffer amplifier between, or by the use of the single circuit of Fig. 11.21.





11.8 Pole cancellation and feedforward compensation

Where the dominant roots of a system comprise a pair of complex conjugate poles which are close to the imaginary axis the behaviour will be excessively oscillatory and the amount of improvement available by phase compensation methods is limited. A technique which can sometimes be used is that of *pole cancellation*. In principle the aim is to introduce a compensation network with zeros which cancel the system poles, together with poles which are positioned more suitably in the s-plane. Figure 11.22 shows two compensation networks



Fig. 11.22 Compensation networks for pole cancellation

referred to as 'bridged T networks' which can be used for this purpose. Cancellation can seldom be exact since the system poles will not usually be known exactly due to practical limitations in modelling and due to slow changes of system parameters with time. The effect of inexact cancellation is that a pair of roots remains near to the system poles, but the effect is small since the coefficient associated with the roots has a small numerical value.

A different design approach which may be useful if the expected unwanted disturbances are known to act at a specific point in the system is by using *feedforward compensation* in conjunction with the feedback loop (Fig. 11.1c). The disturbance signal is monitored and a control signal derived to cancel out much of the disturbance before it affects the system. The computation of the control action needed requires that a model of the system is available. The more



Fig. 11.23 Additional example of feedforward control

closely the disturbance can be monitored and the more accurate the system model, the better will be the control. The arrangement of Fig. 11.23, although apparently containing a minor feedback loop, is in fact another example of feedforward control, the disturbance D here entering at a different point in the system.

11.9 Compensation of sampled-data system

Compensation network design for a sampled-data system is similar to that described in Sections 11.5 to 11.7 for a continuous-data system, and can conveniently be carried out using a Bode diagram.



Fig. 11.24 Sampled-data system with continuous-data compensation

Example 11.4. Consider a system with block diagram as in Fig. 11.24 where, for comparison, the open loop transfer function is as used in the continuous data system of Example 11.2.

The z-transform of the uncompensated open loop can be written as

$$G_{ho}G(z) = (1 - z^{-1})\mathscr{P}\left[\frac{K}{s(1 + 10s)(1 + s)}\right]$$
$$= \frac{K}{9}\left(\frac{z - 1}{z}\right)\mathscr{P}\left[\frac{9}{s} - \frac{10}{(0, 1 + s)} + \frac{1}{(1 + s)}\right]$$

From Table 10.1

$$G_{he}G(z) = \frac{K}{9} \left(\frac{z-1}{z}\right) \left(\frac{9z}{z-1} - \frac{10z}{z-e^{-0.1T}} + \frac{z}{z-e^{-T}}\right)$$

Inspection of Fig. 11.16b would suggest a sampling time of T = 0.2 seconds to give a good sampled-data representation of the continuous signal.

$$G_{ho}G(z) = \frac{K}{9} (z-1) \left(\frac{9}{z-1} - \frac{10}{z-0.9802} + \frac{1}{z-0.8187} \right)$$
$$= \frac{K}{9} \left(\frac{0.017z + 0.016}{(z-0.98)(z-0.819)} \right)$$

Rearranging Eq. 10.23 to make z the subject gives

$$z = \frac{1+r}{1-r}$$

Using this mapping function for z, the corresponding open loop r-transform of $G_{ho}G(z)$ is

$$G_{\text{ho}}G(r) = \frac{K}{9} \left[\frac{0.017 \left(\frac{1+r}{1-r}\right) + 0.016}{\left(\left(\frac{1+r}{1-r}\right) - 0.98\right) \left(\left(\frac{1+r}{1-r}\right) - 0.819\right)} \right]$$

$$\simeq \frac{K(1 - 0.97r)}{(1 + 99r)(1 + 10r)}$$

and $G_{ha}G(j\omega_r) = \frac{K(1 - 0.97j\omega_r)}{(1 + 99i\omega_r)(1 + 10i\omega_r)}$

The break points needed to plot the Bode diagram for $G_{ho}G(j\omega_r)$ occur at $\omega_r = 0.1$ and very close to 0.01 and 1.0 radians per second. If a value of K = 40 is selected the phase margin is 19° and a direct comparison with Example 11.2 can be made. The Bode diagram for $G_{ha}G(j\omega_r)$ is drawn in Fig. 11.25. For a phase lead network

$$G_{c}(r) = \frac{1}{\alpha} \frac{1 + \alpha \tau r}{1 + \tau r}$$
 where $\alpha > 1$

As explained in Example 11.2, a suitable trial value for φ_m is 30°, resulting in the value $\alpha = 3$. The new crossover frequency is positioned at a magnitude value of -4.8 dB and this yields a value for $\omega_m = 0.26$ rad/second. Hence the corner frequencies for the network are located at

and

$$1/\alpha \tau = 0.26/\sqrt{3} = 0.15$$
 rad/second
 $1/\tau = 0.26\sqrt{3} = 0.45$ rad/second



Fig. 11.25 Bode diagram for Example 11.4

For the compensated system

$$\frac{C(r)}{E(r)} = \frac{40(1-0.97r)(1+6.7r)}{(1+99r)(1+10r)(1+2.2r)}$$

and by making the appropriate substitution for r

$$\frac{C(z)}{E(z)} = \frac{40(0.03z + 1.97)(7.7z - 5.7)(z + 1)}{(100z - 98)(11z - 9)(3.2z - 1.2)}$$

Multiplying the top and bottom of this equation by (z-1) and separating into partial fractions gives

$$\frac{C(z)}{E(z)} = \left(\frac{z-1}{z}\right) \left(-\frac{42.5z}{(z-0.98)} + \frac{1.9z}{(z-0.818)} + \frac{0.52z}{(z-0.375)} + \frac{40z}{(z-1)}\right)$$

Using Table 10.1, and rearranging gives

$$\frac{C(s)}{E(s)} = \left(\frac{1 - e^{-sT}}{s}\right) \left(\frac{40(1 + 0.65s)}{(1 + 10s)(1 + s)(1 + 0.2s)}\right)$$

from which

$$G_{\rm c}(s) = \frac{(1+0.65s)}{(1+0.2s)}$$

The realization of this transfer function can be achieved by use of the compensation networks shown in Fig. 11.13.

The closed loop pulse transfer function is

$$\frac{C(z)}{R(z)} = \frac{9.24z^3 + 609z^2 + 151z - 449}{3529z^3 - 7041z^2 + 5347z - 1507}$$

For a unit step disturbance in R(z), i.e. $R(z) = \frac{z}{z-1}$, the output response is

$$C(z) = \frac{9.24z^4 + 609z^3 + 151z^2 - 449z}{3529z^4 - 10\,570z^3 + 12\,388z^2 - 6854z + 1507}$$

From this equation the steady state value of c(t) can be computed as 0.976 by use of Eq. 10.21, and by means of long division the time response can be estimated. This output response is shown in Fig. 11.26 and illustrates the improvement achieved by the addition of the phase lead compensation element.

The compensated Bode diagram, Fig. 11.25, shows that a phase margin of 39° and a gain margin of 10 dB have been achieved. However, the steep slope



Fig. 11.26 Effect of compensation on step response. Example 11.4

of the phase angle curve suggests that the sampled process is less amenable to phase lead compensation than the original continuous-data system.

If alternatively a phase lag network is incorporated and the procedure described in Section 11.7 is followed, it can be seen that a phase margin of 45° can be realized if the gain crossover frequency is moved to 0.1 rad/second. Since the network does influence marginally the final phase lag, a lower frequency of 0.08 will be used as the crossover value. Thus the network must produce 13 dB of attenuation at this frequency.

Hence

$$20 \log \alpha = -13 \text{ dB}$$
$$\alpha = 10^{-13/20} = 0.22$$

and this fixes the distance between the two corner frequencies of the lag network. The upper corner frequency is located at 1/10 of the new crossover frequency

i.e.
$$\frac{1}{\alpha \tau} = \frac{0.08}{10} = 8.0 \times 10^{-3} \text{ rad/second}$$

and $\frac{1}{\tau} = 0.22 \times 8.0 \times 10^{-3} = 1.76 \times 10^{-3} \text{ rad/second}$

Thus

$$\frac{C(r)}{E(r)} = \frac{40(1-0.97r)(1+125r)}{(1+99r)(1+10r)(1+568r)}$$

and

$$\frac{C(z)}{R(z)} = \frac{3.78z^3 + 248.28z^2 + 0.22z - 244.28}{15\,651.28z^3 - 43\,481.27z^2 + 40\,584.82z - 12\,746.63}$$

As would be expected the stability margin, Table 11.1, has been improved by use of the lag network but the reduced bandwidth results in a longer rise time as shown by the step response given in Fig. 11.26.

Table 11.1 Harmonic response characteristics.

PM	GM	Bandwidth
19	9	0.20
38	10	0.37
48	20	0.11
43	16	0.20
	PM 19 38 48 43	PM GM 19 9 38 10 48 20 43 16

A more practical and possibly more versatile approach to sampled-data system compensation may be found in the use of the lag-lead network (or its equivalent) shown in Fig. 11.21 since it contains advantages of both the phase lead and phase lag networks.

The transfer function of a lag-lead network can be written as

$$G_c(s) = \left(\frac{1+\alpha_1\tau_1 s}{1+\tau_1 s}\right) \times \left(\frac{1+\alpha_2\tau_2 s}{1+\tau_2 s}\right)$$

where $\alpha_1 \alpha_2 = 1$, $\alpha_1 > 1$ and $\alpha_2 < 1$

The lag portion will first be established by selecting values for α_2 and τ_2 , a purely arbitrary choice in this case.

If the crossover be moved to 0.15 rad/second a phase margin of 30° will result.

i.e.
$$5 = -20 \log \alpha_2$$

 $\alpha_2 = 10^{-5/20} = 0.5623$
 $\frac{1}{\alpha_2 \tau_2} = \frac{0.15}{10} = 0.015 \text{ rad/second}$
 $\frac{1}{\tau_2} = 8.4 \times 10^{-3} \text{ rad/second}$

and

Using the constraint

$$\alpha_1 \alpha_2 = 1$$

 $\alpha_1 = \frac{1}{0.5623} = 1.8$

gives

The phase lead portion is positioned on the Bode diagram in the same way as described earlier. The attenuation of the lead network is

$$20 \log_{10} 1.8 = 5 dB$$

hence

$$\omega_m = 0.23$$
$$\frac{1}{\tau_1} = 0.23 \sqrt{1.8} = 0.3086$$
$$\frac{1}{\alpha_1 \tau_1} = \frac{0.3086}{1.8} = 0.1714$$

For the compensated system

$$\frac{C(r)}{E(r)} = \frac{40(1-0.97r)(1+5.8r)(1+67r)}{(1+99r)(1+10r)(1+3.2r)(1+119r)}$$

and the closed loop pulse transfer function is

$$\frac{C(z)}{R(z)} = \frac{13.87z^4 + 90\,154z^3 - 629.97z^2 - 893.54z + 624.1}{13\,873.87z^4 - 44\,910.26z^3 + 55\,184.45z^2 - 30\,479.94z + 6348.28}$$

The time response for this system is shown in Fig. 11.26 and although an improvement is shown further trials would be necessary to determine if the gain margin could be increased while still maintaining a similar phase margin and bandwidth (Table 11.1).

This example illustrates that the ideas normally associated with continuousdata system design can be extended in all aspects to the sampled-data system by use of z-transformation and the r-transform method.

11.10 State vector feedback control

An exposition of the design of linear systems using the classical methods of Bode diagrams and root locus plots has been given in the earlier sections of this chapter. It will have been observed that in the classical method of design feedback is most generally obtained from one variable, the output. Only when inner feedback loops are used for system compensation is more than one variable employed for feedback control. The conventional controller incorporating P + D, P + I, phase lead or phase lag algorithms is generally unable to control independently all system poles, Fig. 11.6, since the number of free parameters available for adjustment is restricted to two or three in most cases. Therefore, if the process can be described in state vector form, it is logical to extend the power of the classical design approach by providing full state



Fig. 11.27 Block diagram for state vector feedback control

feedback. This leads to the concept of *state variable feedback* and is the basis of most design techniques in modern control theory.

Many of the analytical design procedures reduce to problems of (a) finding a suitable variable to manipulate to enable the system output to be changed from some initial state to any other desired state in a finite time, and (b) being able to estimate, with confidence, the process state from observations made over a finite time of the output response. These two properties are classified in the literature as *controllability* and *observability* respectively.

The arrangement shown in Fig. 11.27 is a block diagram representation of a system in which each state variable is fed back through a fixed gain. The state vector $\{x(t)\}$ for the *n*-dimensional process is fed back through a constant $1 \times n$ matrix **H**, and the system describing equations can be written

$$u(t) = r(t) - H\{x(t)\}$$

Substituting this in Eq. 5.16, yields

$${\dot{x}(t)} = (A - BH){x(t)} + Br(t)$$

From the analytical results of Section 5.3, the characteristic equation for this closed loop system is

$$|s\boldsymbol{I} - \boldsymbol{A} + \boldsymbol{B}\boldsymbol{H}| = 0 \qquad 11.9$$

If the system can be shown to be controllable, i.e. if it is possible to change the system state from some initial state $\{x(0)\}$ to a desired equilibrium state in a finite time interval by means of the process input u(t), then the roots of Eq. 11.9 can be chosen arbitrarily. This gives rise to a design procedure known as *pole-placement*.

Although most physical systems are controllable, there are some exceptions and care must be taken in making all-inclusive general statements. In particular, lack of controllability may not be apparent when a transfer function description is used to model a process. The necessary and sufficient condition for controllability is that the composite matrix

$$S = [B A B A^2 B \dots A^{n-1} B]$$

has a rank of n. The rank of matrix S is the order of the largest non-singular matrix (defined in Appendix C) contained in S. For example, if the matrix is

$$S = \begin{bmatrix} 1 & 2 & 3 \\ 2 & 3 & 4 \\ 3 & 5 & 7 \end{bmatrix}$$

then S is of rank 2 because

$$\begin{vmatrix} 1 & 2 \\ 2 & 3 \end{vmatrix} = -1$$
 and det $S = 0$

Consider a process described by the state variable Eq. 5.7, i.e.

$$\{\dot{\mathbf{x}}(t)\} = \begin{bmatrix} 0 & 1 & 0 \\ 0 & 0 & 1 \\ 0 & -8 & -6 \end{bmatrix} \{\mathbf{x}(t)\} + \begin{bmatrix} 0 \\ 0 \\ 1 \end{bmatrix} u(t)$$
(5.7)

then

$$S = [B A B A^2 B] = \begin{bmatrix} 0 & 0 & 1 \\ 0 & 1 & -6 \\ 1 & -6 & 28 \end{bmatrix} = -1$$

which satisfies the controllability requirement of this process, i.e. that S is of order 3 since n = 3. It should be noted that Eq. 5.7 is of a special type, known as 'phase-variable canonical form' and as such will always be state controllable (Appendix C, Section (j)).

Example 11.5. For the engine speed regulating system of Fig. 9.2, re-drawn in modified form in Fig. 11.28, derive a state-vector control law that will give





zero steady state error for a step change in selected engine speed and that will also give minimal initial overshoot to this step-input disturbance.

To quicken the process response and reduce the steady state error a gain value of 100 is taken, hence for this example the forward loop transfer function is

$$\frac{Y(s)}{U(s)} = \frac{100}{(s^2 + 9s + 25)(1 + s)}$$



Fig. 11.29 State variable diagram for $100/(s^2 + 9s + 25)(1 + s)$

and the state variable diagram can be drawn as shown in Fig. 11.29. From inspection of Fig. 11.29 the state variable equations are:

$$\begin{aligned} \dot{x}_1(t) &= x_2(t) \\ \dot{x}_2(t) &= x_3(t) \\ \dot{x}_3(t) &= -25x_1(t) - 34x_2(t) - 10x_3(t) + 100u(t) \\ y(t) &= x_1(t) \end{aligned}$$

that is

$$\boldsymbol{A} = \begin{bmatrix} 0 & 1 & 0 \\ 0 & 0 & 1 \\ -25 & -34 & -10 \end{bmatrix}, \quad \boldsymbol{B} = \begin{bmatrix} 0 \\ 0 \\ 100 \end{bmatrix}, \quad \boldsymbol{C} = \begin{bmatrix} 1 & 0 & 0 \end{bmatrix}$$

The constant feedback matrix H can be written

$$H = [h_1 h_2 h_3]$$

Using Eq. 11.9, the characteristic equation is

$$|sI - A + BH| = s^3 + (10 + 100h_3)s^2 + (34 + 100h_2)s + (25 + 100h_1) = 0$$

Since zero steady state error is a design requirement

$$25 + 100h_1 = 100$$

 $h_1 = 0.75$

The second requirement of minimal overshoot suggests, by inspection of Fig. 9.2b, that two roots positioned at $-2 \pm 2j$ will help achieve this end. Hence

 $s^{3} + (10 + 100h_{3})s^{2} + (34 + 100h_{2})s + 100 = (s + 2 - 2j)(s + 2 + 2j)(s + p)$ = $s^{3} + (4 + p)s^{2} + 4(p + 2)s + 8p$

Inspection of this equation and equating coefficients yields

$$p = \frac{100}{8} = 12.5$$

$$34 + 100h_2 = 4(12.5 + 2)$$

$$\therefore h_2 = 0.24$$

$$10 + 100h_3 = 4 + 12.5$$

$$\therefore h_3 = 0.065$$

The closed loop transfer function is then

$$\frac{Y(s)}{R(s)} = \frac{100}{s^3 + 16.5s^2 + 58s + 100}$$

The complex roots of the characteristic equation have a damping factor of 0.707 and are dominant since the third root is far to the left. Hence the required specifications have been satisfied and the response (shown in Fig. 11.30) is very similar to that of a second order system.



Fig. 11.30 Step response

A weakness of state-vector feedback control in practice is that not all of the state variables will be readily accessible directly from the system, or the cost of monitoring these might be prohibitive. For a high order system there are many state variables and hence many transducers are needed. These problems can be surmounted by estimating individual unavailable state values by use of the *state observer algorithm*. Luenberger (Ref. 3, p. 278) showed that an *n*th order process with *q* independent outputs can be observed by using an (n - q) order linear dynamic system. This condition of observability is defined in a manner analogous to controllability and indicates that an unobservable system will have dynamic modes of operation which do not influence the measured output response in any way.

If the matrix

$$\boldsymbol{O} = \begin{bmatrix} \boldsymbol{C}^{\mathrm{T}} \boldsymbol{A}^{\mathrm{T}} \boldsymbol{C}^{\mathrm{T}} (\boldsymbol{A}^{\mathrm{T}})^{2} \boldsymbol{C}^{\mathrm{T}} \dots (\boldsymbol{A}^{\mathrm{T}})^{n-1} \boldsymbol{C}^{\mathrm{T}} \end{bmatrix}$$

is of rank *n*, the system is observable and the state vector can be constructed from linear combinations of the output $\{y(t)\}$, input $\{u(t)\}$ and derivatives of these variables. Intuitively, the observer should have the same form of state equation as the original process. The estimated state vector is designated as



Fig. 11.31 Control loop incorporating observer

 $\{\hat{x}(t)\}\$ and is used to generate the control variable $\{u(t)\}\$ through the feedback matrix H, Fig. 11.31, since the observer will be shown to have the capacity to minimize the error between the actual states and the observed states. These requirements can be embodied into a mathematical algorithm that enables the observer block, Fig. 11.31, to be specified in detail. Based on this premise, the appropriate mathematical equations are now presented together with an example to illustrate how these might be used in the design of a state observer.

Defining the observer algorithm as

$$\{\dot{z}(t)\} = E\{z(t)\} + F\{y(t)\} + G\{u(t)\}$$
 11.10

an estimate of the process states can be achieved by use of the linear relation

$$\{z(t)\} = L\{\hat{x}(t)\}$$
 11.11

The orders of the vectors and coefficient matrices, yet to be chosen, are:

 $\{z(t)\} = (n-q) \times 1$ vector; $\{y(t)\} = q \times 1$ vector; $\{u(t)\} = p \times 1$ vector and

$$E = (n - q) \times (n - q)$$
 matrix; $F = (n - q) \times q$ matrix;

$$G = (n - q) \times p$$
 matrix; $L = (n - q) \times n$ matrix.

To be able to use an observer with confidence it is essential that $\{\hat{x}(t)\}\$ be driven as close to $\{x(t)\}$ as possible, hence an error vector $\{\varepsilon_{x}(t)\}$ is defined as,

$$\{\varepsilon_{x}(t)\} = \{\hat{x}(t)\} - \{x(t)\}$$
 11.12

Substituting from Eq. 11.12 into Eq. 11.11 gives

$$z(t) = L\{x(t)\} + L\{\varepsilon_x(t)\}$$
 11.13

Differentiating each term with respect to time in Eq. 11.13 gives

$$\dot{z}(t)$$
 = $L{\dot{x}(t)} + L{\dot{\varepsilon}_{x}(t)}$ 11.14

Replacing $\{\dot{x}(t)\}$ from Eq. 5.16 yields

$$\{\dot{z}(t)\} = LA\{x(t)\} + LB\{u(t)\} + L\{\dot{\varepsilon}_{x}(t)\}$$
 11.15

Substituting Eq. 11.13 and Eq. 11.15 into Eq. 11.10, and replacing { y(t)} with Eq. 5.17 (matrix D = 0), gives on rearrangement

$$L\{\hat{\mathbf{z}}_{\mathbf{x}}(t)\} = EL\{\mathbf{z}_{\mathbf{x}}(t)\} + (EL + FC - LA)\{\mathbf{x}(t)\} + (G - LB)\{\mathbf{u}(t)\} \quad 11.16$$

The solution of this equation is dictated by the constraints imposed, therefore to ensure that $\{\varepsilon_x(t)\}$ decays with time, Eq. 11.16 will be constrained to be

$$L\{\dot{\mathbf{\varepsilon}}_{\mathbf{x}}(t)\} = EL\{\mathbf{\varepsilon}_{\mathbf{x}}(t)\}$$
 11.17

and hence the estimated vector $\{\hat{x}(t)\}\$ will converge onto the actual vector $\{x(t)\}\$. From Eq. 5.34 the solution of Eq. 11.17 is

$$L\{\varepsilon_{\mathbf{x}}(t)\} = e^{Et}L\{\varepsilon_{\mathbf{x}}(0)\}$$
 11.18

Now the value of the observer output at time t = 0 cannot readily be evaluated because the observer output at the time instant immediately before is not known. It is most simple, therefore, to assume that $\{\hat{x}(0)\}$ is zero, hence

$$\{\varepsilon_{\mathbf{x}}(0)\} = -\{\mathbf{x}(0)\}\$$

Thus, the error vector at any time instant can be obtained from a knowledge of the initial states of the plant.

For Eq. 11.17 to be true

$$LA - EL = FC 11.19$$

$$G = LB 11.20$$

and

In order to implement the observer, values must be assigned to the matrices E, F, G and L which are related according to Eq. 11.19 and Eq. 11.20. The approach adopted here is to specify E and F and solve Eq. 11.19 for L and Eq. 11.20 for G.

The observer described is of a reduced order because the information contained in the plant output is utilized to formulate the estimated state vector. Hence by forming an adjoined equation from Eq. 11.13 and Eq. 5.17, with matrix D = 0, yields

$$\left[\frac{\{\mathbf{z}(t)\}}{\{\mathbf{y}(t)\}}\right] = \left[\frac{L}{C}\right] \{\mathbf{x}(t)\} + \left[\frac{L}{0}\right] \{\mathbf{\varepsilon}_{\mathbf{x}}(t)\}$$

and since $\{\varepsilon_x(t)\}\$ decays with time, an estimate of the state vector is obtained from

$$\left\{\hat{\mathbf{x}}(t)\right\} = \left[\frac{L}{C}\right]^{-1} \left[\frac{\left\{\mathbf{z}(t)\right\}}{\left\{\mathbf{y}(t)\right\}}\right]$$
 11.21

Example 11.6. Design a state observer suitable for use with the speed regulating system shown in Fig. 11.28.

To confirm that the system is observable the matrix $Q = [C^T A^T C^T (A^T)^2 C^T]$ must have rank 3.

Now

$$C^{\mathrm{T}} = \begin{bmatrix} 1 & 0 & 0 \end{bmatrix}^{\mathrm{T}}$$
$$(A^{\mathrm{T}})^{2} = \begin{bmatrix} 0 & 0 & -25\\ 1 & 0 & -34\\ 0 & 1 & -10 \end{bmatrix}^{2} = \begin{bmatrix} 0 & -25 & 250\\ 0 & -34 & 315\\ 1 & -10 & 66 \end{bmatrix}$$

and

det
$$Q = \begin{vmatrix} 1 & 0 & 0 \\ 0 & 1 & 0 \\ 0 & 0 & 1 \end{vmatrix} = 1,$$

hence the necessary condition is satisfied.

If the dependence of the estimated errors upon the observer dynamics is to be accounted for, a number of restrictions must always be imposed on matrices E and F in order to reduce the number of independent variables. In this example, these restrictions will be:

- (a) E is a diagonal matrix
- (b) the elements of F are made equal to unity, that is the plant output is fed directly into the observer.

Since the roots of the characteristic equation are at $-2 \pm 2j$ and -12.5, the elements of *E* will be arbitrarily chosen to be -3 and -2. From Eq. 5.37

$$\mathcal{L}(\mathbf{e}^{Et}) = \frac{\operatorname{adj} (sI - E)}{|sI - E|}$$
$$= \begin{bmatrix} \frac{1}{(s+3)} & 0\\ 0 & \frac{1}{(s+2)} \end{bmatrix}$$
$$\mathbf{e}^{Et} = \mathcal{L}^{-1} \begin{bmatrix} \frac{1}{(s+3)} & 0\\ 0 & \frac{1}{(s+2)} \end{bmatrix}$$
$$= \begin{bmatrix} \mathbf{e}^{-3t} & 0\\ 0 & \mathbf{e}^{-2t} \end{bmatrix}$$

hence

hence Eq. 11.18 gives the error vector

 $\boldsymbol{L}\{\boldsymbol{\varepsilon}_{\boldsymbol{x}}(t)\} = \begin{bmatrix} \mathrm{e}^{-3t} & 0\\ 0 & \mathrm{e}^{-2t} \end{bmatrix} \boldsymbol{L}\{\boldsymbol{\varepsilon}_{\boldsymbol{x}}(0)\}$

The largest state error is less than 2% of the initial value after 2 seconds, therefore these values for the elements of E will be used in the observer design for this speed regulating system.

Expressing

$$L = \begin{bmatrix} l_1 & l_2 & l_3 \\ l_4 & l_5 & l_6 \end{bmatrix},$$

then

$$LA = \begin{bmatrix} l_1 & l_2 & l_3 \\ l_4 & l_5 & l_6 \end{bmatrix} \begin{bmatrix} 0 & 1 & 0 \\ 0 & 0 & 1 \\ -25 & -34 & -10 \end{bmatrix}$$
$$= \begin{bmatrix} -25 \ l_3 & l_1 - 34 \ l_3 & l_2 - 10 \ l_3 \\ -25 \ l_6 & l_4 - 34 \ l_6 & l_5 - 10 \ l_6 \end{bmatrix}$$

Similarly

$$EL = \begin{bmatrix} -3 l_1 & -3 l_2 & -3 l_3 \\ -2 l_4 & -2 l_5 & -2 l_6 \end{bmatrix}$$

and

$$LA - EL = \begin{bmatrix} -25 \ l_3 + 3 \ l_1 & l_1 - 34 \ l_3 + 3 \ l_2 & l_2 - 10 \ l_3 + 3 \ l_3 \\ -25 \ l_6 + 2 \ l_4 & l_4 - 34 \ l_6 + 2 \ l_5 & l_5 - 10 \ l_6 + 2 \ l_6 \end{bmatrix}$$
$$FC = \begin{bmatrix} 1 \\ 1 \end{bmatrix} \begin{bmatrix} 1 & 0 & 0 \end{bmatrix} = \begin{bmatrix} 1 & 0 & 0 \\ 1 & 0 & 0 \end{bmatrix}$$

By equating the individual elements of Eq. 11.19

$$\boldsymbol{L} = \begin{bmatrix} 0.93 & 0.5 & 0.07 \\ 1.64 & 0.73 & 0.09 \end{bmatrix}$$

From Eq. 11.20

$$G = \begin{bmatrix} 7 \\ 9 \end{bmatrix}$$

Now

$$\begin{bmatrix} \boldsymbol{L} \\ -\boldsymbol{C} \end{bmatrix} = \begin{bmatrix} 0.93 & 0.5 & 0.07 \\ 1.64 & 0.73 & 0.09 \\ 1 & 0 & 0 \end{bmatrix}$$

From Appendix C

$$\operatorname{Adj} \begin{bmatrix} L \\ C \end{bmatrix} = \begin{bmatrix} 0 & 0 & -0.0061 \\ 0.09 & -0.07 & 0.0311 \\ -0.73 & 0.50 & -0.1461 \end{bmatrix}$$
$$\operatorname{det} \begin{bmatrix} L \\ C \end{bmatrix} = -0.0061$$
$$\begin{bmatrix} L \\ C \end{bmatrix}^{-1} = \begin{bmatrix} 0 & 0 & 1 \\ -14.75 & 11.48 & -5.1 \\ 119.67 & -81.97 & 24 \end{bmatrix}$$

and



Fig. 11.32 Block diagram for complete system

From Eq. 11.21, the estimated state vector is

$\hat{x}_1(t)$	1	- 0	0	17	$\left[z_{1}(t) \right]$
$\hat{x}_2(t)$	=	-14.75	11.48	-5.1	$z_2(t)$
$\hat{x}_{3}(t)$		119.67	- 81.97	24	y(t)

With this information the complete system diagram can be drawn as shown in Fig. 11.32. The time response for this arrangement is almost identical to that shown in Fig. 11.30 for the system with state vector feedback indicating the power of this approach in control algorithm design.

11.11 Relay control

A simple low cost form of feedback control action that makes use of a switched relay has found wide industrial application. The magnitude of the corrective action is independent of the size of the error, but the sign of this constant corrective action is directly dependent on the sign of the error signal. The most familiar relay control system is a room temperature control (Fig. 11.33); this employs a thermostat to switch the heat on when the temperature is too low and off again when the temperature exceeds the desired value (cooling would be needed where the desired temperature is below that of the prevailing ambient temperature).

Inherent within all relay elements is a certain amount of dead-band, and this is used to ensure that as long as the error magnitude is less than some defined



Fig. 11.33 Relay control of room temperature

value there is no corrective action. In an electrical relay this dead-band arises because the coils require a finite amount of current to actuate the relay contacts; in a hydraulic system valve overlap may be present to reduce fluid leakage at porting, and this creates a dead-band. The presence of dead-band may cause the system to exhibit self-sustained oscillations of constant amplitude and frequency, referred to as *limit cycles*. Control over the size of the dead-band to prevent limit cycles can only be exercised if the magnitudes of the signals within a system are known. A change to a set-point could, for example, be responsible for the onset of a limit cycle oscillation. However, analytical methods do exist which enable the engineer to predict the deadband widths for given signal magnitudes which would cause limit cycle conditions. The limit cycle phenomenon can be used to advantage in certain industrial situations to overcome problems of valve stiction which may otherwise lead to malfunction and component failure.

The dynamic behaviour of systems such as those above, which include straight-forward non-linearities, can be analysed by three techniques; these are extensions of the linear techniques described earlier in this book, and are no more difficult to understand. The system can be simulated on an analogue computer using appropriate non-linear units, and the output response noted for different forcing functions. In the time domain, the step response can be studied by means of the phase plane technique, where a phase trajectory (similar to Fig. 2.10, but with discontinuities of slope arising from the switching) is drawn by graphical or analytical means. This method is limited to second order systems since higher order trajectories cannot be drawn on paper. Phase trajectories can be obtained directly from an analogue simulation and are particularly useful for giving a physical insight into the form of the transient behaviour and the effect of the non-linearity. The third technique is a frequency domain method, suitable for any order of system but restricted to a single group of non-linearities, in which the non-linearity is approximated by a describing function. This is a function analogous to $G(j\omega)$ but whose magnitude and phase are, in general, functions of input amplitude in addition to frequency (they are the amplitude and phase of the first harmonic component of the output, it being assumed that higher harmonics are attenuated in their passage round the loop).

In contrast to linear systems, the response of a non-linear system to an input excitation of known magnitude and form is no guide to its behaviour for other input signals, since the principle of superposition no longer holds. The stability of linear systems is determined solely by the location of the roots of the

11.12 Case study of electrohydraulic servomechanism 265

characteristic equation; for non-linear systems the situation is not so clear and is very much dependent on the input signal and the initial system state.

It can be concluded that in the design of a nonlinear system, information is required about the type and amplitude of all anticipated inputs, and the initial operating condition of the system about which the design study takes place, in addition to the usual detailed knowledge of the physical process from which the mathematical model is derived. With this information a full analytical and simulation study can be conducted and results obtained which can be used with confidence.

11.12 Case study of electrohydraulic servomechanism

As a conclusion to the book, this final section attempts to improve the reader's physical understanding of the rather abstract concepts of system modelling, analysis, and control, and to further illustrate the interrelationship between the many different topics by describing a case study of a specific practical system. The study outlines the analysis and design of an electrohydraulic servo-mechanism used for positioning the slideway of a numerically controlled machine tool. The first stage is concerned with obtaining a mathematical model suitable for analysis. This is followed by the determination of system accuracy and dynamic response, and the design of appropriate compensation to achieve a performance which is deemed to be 'satisfactory'.

(a) Modelling of system. With a numerically controlled machine tool, the required machining operations are specified by a set of coded instructions read from paper tape or other storage medium. The machining is then carried out by moving the workpiece relative to the cutting tool in the appropriate direction at the required feed rate by means of positional servomechanisms; the servomechanisms convert electrical signals specifying the desired position into an actual position. The subject of this study is a milling machine (shown schematically in Fig. 11.34) in which the cutter rotates about a fixed axis and the workpiece is moved relative to it. The workpiece is mounted on a slideway which has three axes of movement, and the movement is effected for each axis by a ram (or rams) controlled by an electrohydraulic servovalve. To achieve accurate positioning and hence accurate machining, feedback is essential; thus the actual position of the slideway is compared with the desired position to generate an error signal; this is amplified and acts as input signal to the servovalve, A simple block diagram representation is shown in Fig. 11.35, omitting at this stage any minor loop or other compensation feature. The control loops for all three axes are similar, except that the masses to be moved and the actuating ram areas have different numerical values; hence for illustrative purposes only one of the loops need be studied.

The system chosen is one in which, for a preliminary study, the nonlinearities inherently present can be ignored, and small perturbation analysis can be employed to derive a linear model. Once a general understanding of the form of the behaviour of the linearized system has been obtained, then the model can if required be refined to include certain of the non-linearities and the less dominant effects. To enable an analysis to be carried out, a transfer



Fig. 11.34 Schematic arrangement of servomechanisms for positioning of machine tool slideway on three axes





function description must be determined for each of the blocks in the loop, the transfer functions being obtained on the basis of theoretical considerations, experimental testing, or a combination of the two. It is likely that the dynamic characteristics of the mechanical components will be dominant, and that the transducer and amplifier can effectively be thought of as having constant gains

and negligible phase shifts within the bandwidth of the remainder of the system.

The form of the transfer function for the hydraulic ram and slideway can be determined theoretically by application of the appropriate physical equations—in this case Newton's second law of motion, and a flow continuity equation. Allowing for the presence of leakage, and for a fluid which is not completely incompressible, the transfer function can be derived (Section 2.5, Eq. 2.36) in the form

$$\frac{X(s)}{Q(s)} = \frac{1}{s\left\{\frac{Mv}{K_{\rm B}A}s^2 + \left(\frac{K_{\rm L}M}{A} + \frac{\mu v}{K_{\rm B}A}\right)s + \left(\frac{K_{\rm L}\mu}{A} + A\right)\right\}}$$
 11.22

where X(s) and Q(s) are the Laplace transforms of the actual position and the fluid volumetric flow rate respectively, M is the total mass being moved, A the effective ram area, K_B the bulk modulus of the fluid, v the volume of fluid between the servovalve and the ram, K_L a leakage coefficient (flow/unit pressure difference), and μ a friction coefficient (force/unit velocity). A is constant and M nearly so (varying only by virtue of the different masses of workpieces); K_B is dependent on the type of fluid used and the amount of entrained air which is assumed to be present; v varies with ram position; K_L and μ are likely to be unknown, but order of magnitude estimates can probably be made. If leakage is assumed to be negligible for a first study, then the transfer function simplifies to

$$\frac{X(s)}{Q(s)} = \frac{1}{s\left\{\frac{Mv}{K_{\rm B}A}s^2 + \frac{\mu v}{K_{\rm B}A}s + A\right\}} = \frac{\frac{1}{A}\left(\frac{K_{\rm B}A^2}{Mv}\right)}{s\left\{s^2 + \frac{\mu}{M}s + \frac{K_{\rm B}A^2}{Mv}\right\}}$$
 11.23

The relationship between slideway velocity and ram input flowrate is thus of second order with undamped natural frequency $\omega_n = \sqrt{\frac{K_B A^2}{Mv}}$, damping factor $\zeta = \frac{\mu}{2A} \sqrt{\frac{v}{MK_B}}$, and gain $= \frac{1}{A}$. The 'spring' effect arises from the compressibility of the fluid giving an effective stiffness $= \frac{K_B A^2}{v}$. It can also be seen that

 ω_n is a function of ram position and will be a minimum when v is a maximum, i.e. when the ram is in the midposition with equal volumes of fluid at each side. The damping which is assumed to arise primarily from viscous drag in the lubricating film will be small, and an accurate assessment of ζ can only follow from experimental testing. Comparing the two transfer functions, the effect of any leakage can be seen to be to increase ω_n , increase the effective ζ , and reduce the gain.

The flow to the ram is controlled by an electrohydraulic servovalve. A typical valve is shown schematically in Fig. 11.36, and its mode of operation is described briefly in this paragraph, with terminology as used in the diagram. The valve is designed to give an output flow rate proportional to input current



Fig. 11.36 Schematic arrangement of servovalve (Dowty Series 4551)

and to have a 'good' dynamic response (one that is fast relative to the system); it utilizes the principle of feedback to achieve this. The input signal, the current flowing through the coil, induces an electromagnetic force on the armature which tilts the armature-flapper assembly slightly about an effective pivot provided by the flexure tube, thus causing an increase of flow area at one nozzle and a decrease at the other. This creates a differential pressure across the ends of the spool valve, and results in spool displacement which, in turn, causes a restoring torque to be applied to the armature assembly via the feedback spring. Spool movement continues until the feedback torque balances the input signal torque, and the armature-flapper assembly, with forces in equilibrium, returns to its null position with the flapper mid-way between the nozzles, the pressure difference across the spool dropping to zero. The resulting spool position is then proportional to the input current. If the pressure difference between the supply pressure and the ram pressure is substantially constant, then the output flow rate is proportional to the input current.

Any attempt at deriving a transfer function for such a valve, using the fundamental physical equations, would require many assumptions to be made about magnitudes of effective inertias, damping forces, etc. It would therefore be essential to validate the theoretical model by experimental testing. A model derived by this means would probably be more complex than necessary, and would include dynamic effects that are only of significance at frequencies beyond the system bandwidth. For such a component, therefore, direct practical testing and fitting of an appropriate low order transfer function to the test results would probably be more relevant. Typical harmonic response information (Fig. 11.37), together with parameters for an equivalent second order transfer function, are provided on the manufacturer's data sheet. A first order transfer function can give a reasonable approximation for frequencies up to about 40 Hz (Fig. 11.37a) which could be useful if the system bandwidth is less than this and if a very simple model is needed. The second order transfer



Fig. 11.37 Typical servovalve frequency response with first and second order transfer function fits. (a) first order fit (b) second order fit

function suggested by the manufacturer ($\omega_n = 140$ Hz, $\zeta = 0.9$) can be seen to provide a very good fit to both the magnitude and the phase curves over the full frequency range for which harmonic information is given (Fig. 11.37b). The transfer function of the servovalve can thus be written as

$$\frac{Q(s)}{l(s)} = k_{\rm v} \left(\frac{1}{1 + 0.0016s} \right)$$
 11.24

$$\frac{Q(s)}{I(s)} = \frac{k_v(775\,000)}{s^2 + 1580s + 775\,000}$$
11.25

where k_v is the valve static gain at zero load pressure, also available from the data sheet as 2.5 in³/second/mA. The valve harmonic information and the

ог

transfer function derived from it relates to a current input. If the error amplifier is a voltage amplifier, then an additional block relating current I(s) to voltage V(s) must be included in the block diagram. The dynamics of this arise from the inductance of the coil of the servovalve torque motor, and it can easily be shown that the transfer function is

$$\frac{I(s)}{V(s)} = \frac{\frac{1}{R}}{1 + \frac{L}{R}s}$$
 11.26

where R and L are the resistance and inductance respectively. For the valve used, $R = 200 \Omega$ and L = 1 H, hence the gain term is 5 mA/volt and the time constant is 0.005 seconds. This is larger than the primary time constant of the valve, Eq. 11.24, and would thus dominate. In practice, the error amplifier would normally be a power amplifier in which case the time constant relating current output and voltage input is likely to be very small in comparison to the effective time constant of the ram and slideway; thus the amplifier will have a flat response to a frequency well above the system bandwidth, will have negligible phase shift, and can be considered to have a transfer function which is a constant k_a . Similarly the position transducer can be assumed to be a pure gain term k_a .

To confirm the form of the ram and slideway transfer function, to check the estimated value of ω_n , and to determine the value of ζ it is necessary to carry out some practical testing. Testing can be carried out with the loop opened, by applying a forcing voltage to the amplifier input or directly to the servovalve using a separate drive amplifier, and recording slideway position or velocity with the existing transducer or with a test transducer. To eliminate errors arising from non-linearities caused by Coulomb friction, and to avoid the danger of damage to the bearing surface which might result from long testing with small amplitudes close to any given position, the testing is best carried out by superimposing steps and sine waves on slow ramp inputs, and taking measurements at a fixed point as the slideway passes it, while moving between two limiting positions on either side. Typical results for step response tests (Fig. 11.38) and frequency response tests (Fig. 11.39) confirm the existence of







Fig. 11.39 Typical harmonic response of velocity

the very lightly damped second order component of response expected from the considerations above. A good estimate of the value of ω_n for the ram and slideway can be obtained directly from the step response trace and, as the damping is very small, the damping factor ζ can be determined by application of the logarithmic decrement method. As a consequence of the averaging inherent in frequency response testing, these values can be estimated more accurately from the harmonic response curves, and the value of the servovalve time constant confirmed, by seeking asymptotes on the magnitude plot and by trial and error curve fitting. At this stage, a digital computer program for evaluating the harmonic response for a known transfer function can be of great help, and would be used in conjunction with some appropriate criterion of 'goodness of fit', to determine a transfer function whose harmonic response is a good fit to the experimental curves (e.g. Levy's method (see ref. 4, p. 278)).



Fig. 11.40 Polar plot corresponding to Fig. 11.39

Study of the phase plot in relation to the gain plot will show whether any significant nonminimum phase effects are present. The corresponding polar diagram (Fig. 11.40) is of interest and highlights the rapid phase change which occurs in the region of the resonant frequency, and the nominally flat response up to about 20 Hz.

(b) Analysis of response and design of compensation. From a combination of theoretical analysis, experimental testing and manufacturers' published data, a linearized mathematical model can thus be developed. If it is assumed that the amplifier is a power amplifier with negligible time constant, that the servovalve can initially be represented by the first order approximation derived above, and that the transducer gain k_t is lumped with the amplifier gain (i.e. the desired position signal becomes the actual desired position) then the block diagram Fig. 11.35 takes the form shown in Fig. 11.41. The dynamic behaviour of the system can now be investigated theoretically with a view to determining how parameter variations or higher order transfer function representations affect the system behaviour, and how different forms of compensation might change this behaviour.

Thought must be given at this stage to what type of response is required, bearing in mind that the end result is a machining operation. Qualitatively, the desired form of behaviour can be described as one which responds rapidly to



Fig. 11.41 Simplified block diagram of system

any change of desired position, with negligible overshoot and with a very small steady state error. The positional error should be zero and the velocity error as small as conveniently possible to allow accurate contouring at fast feed rates. It is thus desirable to undertake an investigation into transient response and steady state accuracy for the basic system, and for the system with various forms of compensation. The transient behaviour can conveniently be determined by means of an analogue computer simulation or a digital computer simulation (Chapter 3), using root locus plots to aid the interpretation of the results (Chapter 9). The corresponding steady state accuracy can be determined by application of the final value theorem of Laplace transform theory (Chapter 8).

Consider first the basic system (Fig. 11.41) in which there is scope for alteration of the value of the loop gain. The positional error is zero as a consequence of the presence of the integration term inherent in the operation of the ram, and the velocity error is inversely proportional to $K_{\rm p}$. The root locus diagram (Fig. 11.42a) can be sketched relatively quickly by hand using the aids to construction described in Section 9.3 or can be computed accurately if a digital computer package is available. Its form suggests that for very low values of K_n (and hence large velocity errors) the response will be very sluggish because of the dominance of the real root very close to the origin. As $K_{\rm p}$ increases, the very lightly damped superimposed oscillation soon becomes dominant, and for a relatively small loop gain the response will become unstable (Fig. 11.43). The limiting value of K_p can be found by using the Routh-Hurwitz criterion; it determines the smallest achievable velocity error. No value of gain appears to be suitable. It is clear that the very lightly damped ram and slideway poles present a problem, and that any compensation used should have the effect of moving these portions of locus further from the imaginary axis towards the left of the s plane. Representation of the servovalve by the second order transfer function, which is valid over a wider frequency range (Fig. 11.42b), makes negligible difference to the result, since the appropriate poles are so far to the left that they cause no marked alteration to the dominant portions of the root locus.

There is little scope, even at the design stage, for altering the physical parameters of the ram and slideway in order to move the complex poles










Fig. 11.44 System with minor loop compensation



Fig. 11.45 Root locus plot for velocity loop with (a) Negative velocity feedback (b) Negative acceleration feedback

farther to the left, and hence increase the damping of the oscillation which dominates the transient response, and allow the loop gain and hence accuracy to be increased. The poles can, however, effectively be moved by introducing a minor feedback loop around the block whose transfer function contains them: this can conveniently be done by feeding back some function of velocity as shown in Fig. 11.44. The aim is to choose the form of inner loop compensation H(s), the parameter values of H(s), and the inner loop gain variable K_{ς} to be such that the roots of the inner loop (velocity loop) which lie near to the ram and slideway poles, and which are poles of the main loop (position loop), are positioned as far to the left as possible. For any given form of transfer function H(s), a root locus diagram for the velocity loop will show whether any improvement is possible, and step response traces for the simulated velocity loop can be used to confirm the expected changes in transient behaviour.

Three forms of transfer function H(s) are probably worth investigating negative velocity feedback, $H(s) = \text{constant} = k_1$, negative acceleration feed-



Fig. 11.46 Root locus plot for velocity loop with negative transient acceleration feedback



Fig. 11.47 Root locus plot for position loop with NTA compensation

back, $H(s) = k_2 s$, and negative transient acceleration feedback, $H(s) = \frac{k_3 s^2}{1 + \beta s}$. A root locus diagram for the velocity loop immediately shows that, with negative velocity feedback, the dominant roots move towards the right in the *s*-plane (Fig. 11.45a), causing the behaviour to be even more oscillatory. For negative acceleration feedback, however, the corresponding portions of root locus move towards the left, suggesting that some degree of improvement is possible (Fig. 11.45b). The root locus plot for negative transient acceleration feedback is more complicated, since a different set of loci arises for each value of the time constant β (Fig. 11.46). There appears to be an optimum value of β where the damping factor of the dominant roots increases most rapidly with increase in gain. For any of the forms of compensation with given parameter values, the root locus plot for the position loop can now be drawn, the poles of this plot being the roots of the velocity loop instead of those used in Fig. 11.42 (e.g. Fig. 11.47). If H(s) has been chosen well, then the result is a much improved transient response (Fig. 11.48). Other forms of compensation can also be investigated, such as the possible use of phase compensation within the minor loop or in the forward loop, and the potential available in a cancellation compensation approach.



Fig. 11.48 Step response curve for system with NTA compensation with parameters chosen to give fast well damped response

For any of the compensation methods studied, some study of the effect on gain and phase margin will improve understanding of system harmonic response, and the relationship between transient and harmonic response. Analysis of system accuracy in terms of loop gains and forms of compensation will fill in a further part of the picture, and will highlight the conflict between achieving good dynamic behaviour and good steady state accuracy. In a design situation it would then be necessary to attempt to choose a good and realizable compensation method, and to optimize the system parameters with respect to the chosen performance criteria. The large numerical values, which tend to occur here, highlight the scaling which commonly is necessary or desirable when analysing a real system, as opposed to a hypothetical system, where convenient numerical values are chosen.

A full and detailed description of the analysis of this system, comparison of different compensation methods, and design of suitable compensation, requires more space than is available here. The reader is therefore left to carry out such an investigation, for the model given in Fig. 11.41, as an exercise to consolidate the material learnt in this book.

278 Design of Closed Loop Systems

Specific references in Chapter 11.

- 1 Ziegler, J. G., and Nichols, N. B. Trans A.S.M.E. Vol. 64, pages 759-68, 1942.
- 2 Quarterly Journal of Automatic Control, Journal 4, Vol. XVI, No. 2, pages 53-67, 1975.
- 3 Luenberger, D. G. 'Observing the State of a Linear System', IEEE Transactions of Military Electronics, Vol. MIL-8, pages 74-80, 1964.
- 4 Levy, E. C. 'Complex Curve Fitting', IRE Trans Vol. AC-4, pages 37-43, 1959.

Appendix A Problems

1 In a hydraulic spool valve leakage occurs in an axial direction through the narrow gap between the valve and the housing. Such flow is normally laminar and the flow rate q can be evaluated by applying the Poiseuille equation. This can be written as

$$q = \frac{\pi dh^3}{12\eta} \frac{p_1 - p_2}{l}$$

where $p_1 - p_2$ is the pressure difference between the inlet and outlet of the leakage gap, η is the fluid viscosity, *l* the length of the leakage path, *d* the valve diameter and *h* the leakage gap (the radial clearance). If *d* and *h* are constant obtain a linearized equation for the flow rate in terms of the remaining variables and the constants. What is the significance of this equation, and why is it used?

2 Determine the transfer functions relating the applied force f to the position y, and the applied torque T to the angular position θ for the translational and rotational mechanical systems shown in Fig. P1. K_1 , K_2 , K_3 , and K are stiffness coefficients, C is a damping coefficient, M is a mass, and J_1 and J_2 are moments of inertia. State what assumptions are made.



Fig. P1

3 Fig. P2 shows three passive electrical networks which can be used for system compensation. On the assumption that the input impedance is zero and the output impedance is infinite derive the transfer function relating input and output voltages (relative to earth potential) for each circuit.



4 A thermometer of thermal capacity W_1 (joules/°C) is inserted in a mercury filled protective pocket of thermal capacity W_2 (joules/°C). If the overall coefficient of heat transfer from the mercury in the pocket to the fluid in the thermometer is H_1 (joules/second °C) and from the external fluid to the mercury in the pocket is H_2 (joules/second °C), determine the transfer function relating indicated temperature to actual temperature.

5 Fig. P3 shows schematically a component of a hydraulic system in which oil at pressure p is used to obtain angular movement θ of a lever. Explain the significance of the expression 'small perturbation analysis' with particular reference to this arrangement. Hence derive the transfer function, assuming the piston to be frictionless and leak-free, and the fluid to be incompressible. What effect would leakage past the piston and viscous drag on the piston have on the derived transfer function?



Fig. P3

6 With the aid of a schematic diagram describe briefly the principle of operation of a hydraulic positional servomechanism. Derive the form of the transfer function assuming that account must be taken of the effects of fluid compressibility, fluid leakage, and inertia. Derive also the simplified form of transfer function which results if these effects are neglected.

7 Fig. P4 shows schematically an arrangement for controlling the temperature of a steam-heated oven, utilizing temperature sensitive bellows and a hydraulic servomechanism. The desired temperature is set by adjustment of the position x of a pointer attached to one end of the bellows. An increase in oven temperature θ causes the bellows to expand, the movement being the input signal

Problems 281

to the hydraulic servomechanism whose output y actuates a valve to decrease the steam inlet flow. Determine the transfer function relating valve position y to oven temperature θ . Assume servo valve flow for unit displacement is Q_v , area of ram is A, overall coefficient of heat transfer across bellows wall is H, specific heat and mass of bellows fluid are c and m respectively, and bellows extension is L for unit temperature rise of bellows fluid. Neglect fluid compressibility, fluid leakage, and inertia of moving parts. Suggest briefly the way in which the transfer function would differ if secondary effects such as the above were included.



Fig. P4

8 What is an analogue computer, and in what main ways does it differ from a digital computer? Two variables y and x are both functions of time and are related by the differential equation

$$\frac{d^4y}{dt^4} + 5.8 \frac{d^3y}{dt^3} + 0.03 \frac{d^2y}{dt^2} + 12 \frac{dy}{dt} + 120y = 20x$$

Derive an analogue computer circuit diagram for this equation, and explain how it could be used. Assume that the operational amplifiers available have two inputs with a gain of 10 and two inputs with a gain of unity.

9 The output of a system component whose transfer function is known to be

$$G_1(s) = \frac{20}{s^2 + 1.2s + 0.2}$$

is monitored by means of a transducer with transfer function

$$G_2(s) = \frac{0.06}{1+0.1s}$$

as shown in Fig. P5.

Input System component

$$U(s)$$
 $G_1(s) = \frac{20}{s^2 + 1.2s + 0.2}$
 $C(s)$
 $Transducer$
 $G_2(s) = \frac{0.06}{1 + 0.1s}$
 $C_m(s)$

282 Appendix A

Write down the differential equations for the system component and for the transducer, and derive an analogue computer circuit diagram which could be used to study the response of the system component and the effect of the transducer dynamic behaviour on the measured response. Explain how the circuit might be used. How would the circuit differ to permit investigation of the effect of a change of transducer time constant to 0.04 seconds?

10 The feedback system shown in block diagram form in Fig. P6 controls the output variable c(t) of a plant. The plant dynamic behaviour is thought to be adequately described by the linear differential equation

$$\ddot{c}(t) + 5.2\ddot{c}(t) + 17.5\dot{c}(t) + 0.77c(t) = u(t)$$

The amplifier has an adjustable gain K and a negligible time constant, and the transducer is known to approximate to a second order linear system component with unity gain, damping factor 0.3, and undamped natural frequency 5 rad/second. Draw an analogue computer circuit diagram which could be used to investigate the dynamic behaviour of this system. What studies might be carried out, and what checks could be made to ensure the validity of the results? Use operational amplifier notation and assume that amplifier input gains of 1 and 10 are available.



Fig. P6

11 Draw an analogue computer circuit diagram for a dynamic process represented by the overall transfer function

$$\frac{Y(s)}{U(s)} = \frac{4.2(s+3.4)}{s^5+8.6s^4+0.15s^3+10.3s^2+100s+5}$$

Assume that the amplifiers available have three inputs with gain 1 and two inputs with gain 10. What are the limitations of this diagram, and what additional information is needed to enable a more useful diagram to be prepared?

12 For the block diagram shown in Fig. P7 obtain an analogue computer circuit diagram which could be used to investigate the effect on transient behaviour of the system of changes in settings of the proportional plus integral plus derivative governor. Use operational amplifier notation and assume that amplifiers are available only with inputs of gain 1 or 10.

13 Write a digital computer program which could be used to investigate the dynamic behaviour of the system of Problem 9. (This requires familiarity with Fortran, Pascal or some other high level programming language.)



Fig. P7

14 Write a program using the simulation language CSMP to simulate the system of Problem 10. (A program with the correct general form can be written with the information in Section 3.9, but to ensure freedom from errors a CSMP reference manual or textbook is required.)

15 'A component part of a physical system when tested experimentally was found to be closely approximated by a first order transfer function with time constant 10 seconds.'

Explain clearly the significance of this statement, in particular

- (i) what is meant by a 'first order transfer function with time constant 10 seconds',
- (ii) in what ways the system component could have been tested,
- (iii) how it would have been shown to be approximated by a first order transfer function,
- (iv) how the numerical value of time constant could have been obtained.
- 16 A step change of magnitude unity is applied to a dynamic system consist-

ing of two elements in series with transfer functions $\frac{1}{1+s}$ and $\frac{1}{s^2+7s+10}$

respectively. Obtain an expression for the output response as a function of time. Hence find the steady state value of the output, and check the result using the final value theorem. Find also the maximum overshoot and the approximate time taken to settle within 5% of the steady value.

17 A system component has a transfer function

$$G(s) = \frac{1}{(s+1)(s^2+s+1)}$$

Obtain an expression for the output as a function of time resulting from a unit step change of input. Plot the response curve for the transient. Explain the significance of the result with reference to the values of the poles of the transfer function. Sketch also the likely form of the step response for

$$G(s) = \frac{1}{(s+1)(s^2+4s+16)}$$

18 For the simple closed loop system represented by the block diagram of Fig. P8 determine the response c as a function of time t for (a) a unit step change of input (b) a steadily changing input r = 4t for $t \ge 0$. Hence determine the initial rates of change of c(t), the approximate times to settle within 5% of steady



Fig. P8

state, and the steady state errors. How would the response differ if the forward loop gain were increased?

19 What is the 'principle of superposition' and why is it important in control systems analysis?

A thermocouple mechanically strengthened by encapsulation is used to measure the temperature of the fluid flowing in a pipe. It has been established by experimental testing that the response of the encapsulated thermocouple is closely represented by a first order transfer function with time constant of magnitude 2 seconds. The fluid temperature rises suddenly by 20 °C from a constant temperature, and thereafter rises at a constant rate of 2 °C per second for 20 seconds before becoming constant again. Sketch the form of the output response of the thermocouple. Obtain an equation expressing indicated temperature as a function of time, and hence determine the maximum error in the reading and the time at which the error first becomes less than 1 °C and remains less than this value.

20 A system component is known to be represented by the transfer function

$$G(s) = \frac{10}{(1+s)(1+4s)}$$

The input signal to the component increases suddenly from a datum value to a new value 10 units higher, and then 4 seconds later changes suddenly back to the datum value. Sketch approximately to scale the general form of the response of the output which you would expect and explain in some detail why you have drawn it as you have. Derive an analytical expression for the output as a function of time.

21 Fig. P9 shows the output responses recorded when two different system components were subjected in turn to a unit step input function. Estimate the transfer functions of the components, and indicate on a sketch the position of the roots in the s-plane,

22 Describe in state variable form a system characterized by the differential equation

$$\frac{d^3y}{dt^3} + 4.6 \frac{d^2y}{dt^2} + 39.6 \frac{dy}{dt} + 36y = 36u$$

Draw the block diagram representation of the state model. 23 A feedback control system has transfer function

$$G(s) = \frac{s^2 + 3s + 2}{s(s^2 + 7s + 12)}$$



Fig. P9

Derive two different state models for this system and give the state variable diagram for each.

24 A multitank system consists of four tanks with cross-sectional areas a_1, a_2 . a_3 , and a_4 interconnected by pipes with flow resistances R_1 , R_2 , and R_3 . Fluid is normally supplied to tanks 1 and 4 at flow rates $q_1(t)$ and $q_4(t)$, and drawn from tanks 2 and 3 at flow rates $q_2(t)$ and $q_3(t)$ as shown in Fig. P10. These



Fig. P10

flow rates are independent of the liquid levels in the tanks, while the flow rate from one tank to another is equal to the difference in liquid levels divided by the flow resistance value. If the controlled or output variables are stated to be $h_2(t)$ and $h_3(t)$ derive the state equations for the three situations when the manipulated or input variables are defined as

- (a) $q_1(t)$, $q_2(t)$, $q_3(t)$ and $q_4(t)$
- (b) $h_1(t)$, $h_4(t)$, $q_2(t)$ and $q_3(t)$
- (c) $h_1(t)$ and $h_4(t)$, with $q_2(t) = q_3(t) = 0$

25 What is the transfer function matrix of a process? Derive the transfer function matrix of the multitank arrangement of Problem 24(c). Assume that $a_1 = a_2 = a_3 = a_4 = a$ and that $R_1 = R_2 = 2R_3 = R$.

26 Describe the system represented by the transfer function

$$G(s) = \frac{10(1+s)}{s^2 + 7s + 10}$$

using a first order set of state equations in matrix form. Discuss briefly the uniqueness of your realization and derive the analytical expression for the system output response when a unit step is applied to the input.

27 Explain what you understand by a transition matrix and how it is used in the solution of a state vector differential equation. Derive the transition matrix and hence the solution for a unit step input for the following state vector equation

$$\{\dot{\mathbf{x}}(t)\} = \begin{bmatrix} 0 & 1\\ -1 & -1 \end{bmatrix} \{\mathbf{x}(t)\} + \begin{bmatrix} 0\\ 1 \end{bmatrix} u(t)$$

28 A linear time invariant system is represented by the state equation

$$\{\dot{\mathbf{x}}(t)\} = \begin{bmatrix} -1 & 0 & 0 \\ 0 & -4 & 0 \\ 0 & 0 & -2 \end{bmatrix} \{\mathbf{x}(t)\} + \begin{bmatrix} 1 \\ 1 \\ 1 \end{bmatrix} u(t)$$

Evaluate the solution matrix, and hence obtain the solution of the state equation for an input u(t) = 1 and an initial state

$$\langle \boldsymbol{x}(0) \rangle = \begin{bmatrix} 0\\0\\1 \end{bmatrix}$$

Confirm the result by means of the Laplace transform method of solution. If the output is

$$\{y(t)\} = \begin{bmatrix} 1 & 2 & 1 \\ 0 & 1 & 1 \end{bmatrix} \{x(t)\}$$

what is the system matrix transfer function? 29 Given that the coefficient matrix of a process is

$$A = \begin{bmatrix} 0 & 1 \\ -4 & -2 \end{bmatrix}$$

evaluate the solution matrix e^{At} using the first five terms of the series expansion. Calculate numerical values for the elements of the solution matrix for the discrete cases when t = 1 second and when t = 0.5 second, and say whether the series expansion approximation seems sufficiently accurate.

30 Obtain by calculation the harmonic response characteristics for system components with transfer functions

(a)
$$G(s) = \frac{100}{(s+1)(s+2)(s^2+3s+16)}$$
 (b) $G(s) = \frac{1+0.1s}{s(1+0.02s)(1+0.01s)}$

Plot the results on a polar diagram, and explain their significance.

31 Using straight line approximations draw Bode diagrams for the transfer functions of Problem 30. Sketch also more accurate estimates of the harmonic response curves.

32 The information given in Table P1 is harmonic data obtained by frequency response testing a practical system. It is known that the transducer used to measure the output has second order characteristics with $\omega_n = 100$ rad/second, damping factor = 0.1, and gain = 30. Magnitude and phase information for a second order system with damping factor 0.1 is given in Table P2.

Display the test data on a Bode diagram and obtain the harmonic information for the system itself. Hence derive an approximate transfer function for the system. Explain clearly the significance of any curves drawn.

Table P1

LL

01	0.2	0.4	0.7	1.0	2	4
98.5	47 3	24.5	138	8 92	3 5 5	1 08
90	95	103	111	119	135	155
7	10	20	40	60	80	
0.38	0.19	0.05	0014	0.008	0.0075	
166	170	180	185	192	205	
	01 985 90 7 038 166	01 02 98.5 47.3 90 95 7 10 0.38 0.19 166 170	0.1 0.2 0.4 98.5 47.3 24.5 90 95 103 7 10 20 0.38 0.19 0.05 166 170 180	0.1 0.2 0.4 0.7 98.5 47.3 24.5 13.8 90 95 103 111 7 10 20 40 0.38 0.19 0.05 0.014 166 170 180 185	0.1 0.2 0.4 0.7 1.0 98.5 47.3 24.5 13.8 8.92 90 95 103 111 119 7 10 20 40 60 0.38 0.19 0.05 0.014 0.008 166 170 180 185 192	0.1 0.2 0.4 0.7 1.0 2 98.5 47.3 24.5 13.8 8.92 3.55 90 95 103 111 119 135 7 10 20 40 60 80 0.38 0.19 0.05 0.014 0.008 0.0075 166 170 180 185 192 205

$\frac{\omega}{\omega_n}$	0.2	0.4	0.6	0.8	1.0
Magnitude	1.04	119	1 49	2.54	5
Phase (deg)	0	-5	-14	-24	-90

33 Table P3 lists the experimentally obtained harmonic response information for a component part of a system. Estimate from this the component transfer function, explaining clearly how you obtain your result. Describe the significance of the data tabulated and of the transfer function obtained.

Table PS									
Frequency (Hz)	0.01	0.02	0.04	0.07	0.1	0.15	0.2	0.3	0.5
Magnitude (dB)	-8.5	-8.1	-7.4	-6.0	-4.4	-2.0	-0.5	10	20
Phase (deg)	11	16	24	29	33	34	31	22	12
Frequency (Hz)	0.7	1	2	3	5	7	10	20	30
Magnitude (dB)	26	30	5.3	60	2.3	-5.0	-12.7	-25	-32
Phase (deg)	7	2	-23	-53	-113	-134	-150	-158	-161

34 Define the term 'autocorrelation function' and explain its role in system identification. List the most important properties of such a function and outline the significance of each.

Obtain the autocorrelation function of the variable

$$y = A \sin \omega t + B \cos \left(3\omega t + \frac{\pi}{4} \right) + C$$

where A, B, and C are constants.

35 Derive expressions for the autocorrelation function of

(a) a sine wave $A \sin(\omega t + \varphi)$

(b) white noise with a band limit of ω_c

Hence sketch the form of autocorrelation function which you would expect when signal (a) is contaminated by signal (b), indicating qualitatively the differences which would arise with different ratios of ω_c/ω .

36 Determine the waveform of a pseudo random binary sequence of length 31 bits, and confirm that it satisfies certain laws of randomness. Sketch the form of the autocorrelation function and of the power spectral density for such a sequence with period 6.2 seconds, and explain the significance of each plot.

37 Describe clearly what is meant by the term 'pseudo random binary sequence', and why it is useful for system identification. A practical system has been tested using a PRBS input signal consisting of a 15 bit sequence with a bit interval of 10 seconds. After the initial transient had died out, the input and output traces were recorded; the values of the output measured at the midpoint of each bit interval are given in Table P4. Determine from these the impulse response of the system, and suggest the form of transfer function which the system is likely to have, with values of parameters where these can be estimated.

Table P4

Input	± 1	± 1	± 1	± 1	-1	-1	-1	+1	-1	-1	+)	+1	-1	± 1	-1
Output	11	10	9	8	8	11	14	14	10	3	0	4	11	11	10

38 Determine the steady state error which would be present with different forms of input excitation for a unity feedback system in which the forward loop transfer function is

(a)
$$G(s) = \frac{15}{s(s+3)(s^2+5s+10)}$$
 (b) $G(s) = \frac{1}{s^3+5s^2+6s+10}$

Describe how this error could be reduced in each case and explain what side effects result.

39 A feedback control system incorporating a 3-term controller has the block diagram representation shown in Fig. P11. Determine by means of the Routh–Hurwitz criterion the ranges of values of the gain K for which the system is stable

(i) when $a = b = 0$	(proportional control)
(ii) when $a = 1, b = 0$	(proportional + integral control)
(iii) when $a = b = 1$	(proportional + integral + derivative control)

Obtain also the frequencies of sustained oscillations where these occur. What would be the magnitude of the lowest achievable steady state error in each of the above cases for a step input and for a ramp input?



Fig. P11

40 A temperature controller is represented schematically by the block diagram of Fig. P12. All components are assumed to be linear and to have transfer functions as shown. Draw a Bode diagram for the system and hence determine the values of gain margin and phase margin. Straight line approximations may be used but, in assessing the gain and phase margin, an estimate should be made of the inaccuracy arising from the difference between the true curves and the straight line approximations. Discuss the significance of the values obtained.



Fig. P12

41 The open loop harmonic response data obtained experimentally for the forward path of a unity feedback control system are given in Table P5. By plotting the magnitude information on a Bode diagram, estimate the likely form of the system transfer function. Plot a Nyquist diagram for the range 2 to 20 rad/s and use it to obtain a plot of closed loop magnification against frequency. Hence determine the values of peak magnification and bandwidth, and explain the significance of the values obtained.

Table P5						-		
Frequency (rad/s) Magnitude (dB)	0.3 26.5	0.6 20.5	1.0 16.0	2 9 5	3 6.0	4	5 -22	6 -5.0
Phase lag (deg)	95	100	109	124	135	147	156	164
Frequency (rad/s)	8	10	15	20	30	40	60	80
Magnitude (dB)	-90	-13.0	-200	-27.0	-36.0	-42 5	-51.5	-59.0
Phase lag (deg)	176	185	-	230		252	_	-

290 Appendix A

42 A feedback system with H(s) = 1 has forward loop transfer function given by

$$G(s) = \frac{35}{s(3s+1)(s^2+2s+10)}$$

Determine the open loop frequency response by any appropriate method and hence estimate the values of gain margin and phase margin. Sketch the form of the closed loop magnitude against frequency relationship. Comment on the significance of the results.

43 The open loop transfer function of a unity-feedback control system is given by

$$G(s) = \frac{K}{s(1+0.1s)(1+0.01s)}$$

Evaluate the gain margin and phase margin in terms of K, and hence find the limiting value of K for stability. Check the result by means of the Routh-Hurwitz criterion. What are the magnitudes of the gain and phase margins when K is half of the limiting value?

44 A unity feedback servomechanism has an open loop harmonic response described by

$$G(j\omega) = \frac{-200(1+0.1j\omega)}{\omega^2(1+0.01j\omega)(1+0.02j\omega)}$$

By plotting the harmonic response information on a Nichols chart evaluate the gain and phase margins, and the values of M_p , ω_p , and bandwidth. Plot also the overall magnitude and phase of the servomechanism against an abscissa of frequency. Explain the significance of the results.

45 The regulator system shown in Fig. P13 controls the output response c(t) of a process plant. The harmonic response characteristics of the plant alone,



Fig. P13

obtained experimentally, are given by the values listed in Table P6. There is a simple lag with time constant 0.5 second associated with the measurement of the system output c(t), and a gain term K operating on the error signal. Draw

Table P6

w(rad/second)	0.3	0.5	0.8	1.0	1.2	1.5	2	3	5
$ G(j\omega) $	1 48	1.65	1.88	1.75	1.25	0.73	0.41	Q.18	0.06
LG(jw)	-15°	-28°	-61	-93°	-115°	-136°	-152°	-162°	-170°

the open loop Nyquist diagram for the system and from the plot determine the value of K required for a gain margin of 8 dB. For this value of K determine the variation of closed loop magnification with frequency. Discuss what effect the transducer time constant has on the steady state and dynamic behaviour of the overall system.

46 Sketch the form of the root locus plot corresponding to each of the following open loop transfer functions G(s)H(s), determining as appropriate the numerical values of the salient features which indicate the shape of each plot. Describe in a short paragraph for each how the transient response of the closed loop system might be expected to change with variation of gain constant K.

(i)
$$\frac{K}{s(s+4)(s^2+16s+100)}$$
 (ii) $\frac{K}{s(s+20)(s^2+16s+100)}$
(iii) $\frac{K(s+3)}{(s+1)(s+2)(s+6)}$ (iv) $\frac{K(1+10s)}{s^2(1+5s)(1+s)}$
(v) $\frac{K(1+2s+3/s)}{(1+0.5s)(1+s)(1+3s)}$ (vi) $\frac{K}{s(1+0.1s)^2}$
(vii) $\frac{K(s+5)}{s^2(s+20)}$ (viii) $\frac{K(s^2+2s+16)}{s(s+1)(s^2+4s+16)}$

47 Sketch the general form of the root locus plot for the temperature control system shown in Fig. P12 for variation of the amplifier gain K, indicating clearly the important features. Draw accurately sufficient of the plot to permit determination of the value K required to give a damping factor of 0.7 for the dominant roots, and to determine the value of ω_n for these roots. Mark on the plot the approximate positions of the remaining roots, and comment on the significance of their contribution to the transient response.

48 A feedback system has

$$G(s) = \frac{K}{(1+s)(1+0.08s)(s^2+4s+5)}, \quad H(s) = \frac{1}{1+0.1s}$$

By plotting a root locus diagram, determine the approximate value of K required to ensure that the dominant mode of oscillation has a damping factor of 0.7. For this value of K write down an expression for C(s), the Laplace transform of the output response, when the input is subjected to a step change of magnitude unity. By studying the root locus plot to decide which roots have negligible influence, write down a simplified expression for C(s) from which a good approximation to c(t), the output response, could be obtained by Laplace inversion. Estimate the value of the settling time.

49 Fig. P14 shows the block diagram representation of a temperature control loop for an exothermal reaction process. Draw a root locus diagram indicating clearly the important features of the plot. Describe fully what the plot shows about the transient behaviour and stability of the system, and hence determine approximately what is likely to be the most suitable value of controller gain $K_{\rm e}$. State clearly your criterion for suitability.

292 Appendix A



Fig. P14

50 The open loop transfer function of a unity feedback positional servomechanism with the gain adjusted so that the maximum allowable steady state error is not exceeded is

$$G(s) = \frac{20}{s(1+0.1s)(1+0.2s)}$$

Sketch the root locus plot and determine the undamped natural frequency and damping factor of the dominant roots. Show how the plot is modified by the introduction in the forward loop of a series compensating network with transfer function $\frac{1+4s}{1+30s}$. What are the new values of natural frequency and damping factor for the dominant roots? Explain the significance of the results. **51** A feedback system, Fig. P15, has open loop transfer function

$$G(s)H(s) = \frac{K(1+Ts)}{s(1+0.1s)(1+0.02s)}$$

Sketch a root locus diagram to show the variation of the values of the roots of the characteristic equation as the magnitude of the lead term time constant T is altered from zero to a very large value, for a constant value of gain K = 50. Draw on the same plot the corresponding loci for values of K = 20 and K = 100.

52 A closed loop system (Fig. P15) has feedforward and feedback transfer functions given by

$$G(s) = \frac{100(1+s)}{(1+2s)(1+4s)}$$
 and $H(s) = \frac{1}{1+\tau s}$

Sketch the form of the root locus plot for variation of the time constant τ . Comment on the significance of the plot.



53 What role does the z-transform serve in the analysis of sampled-data systems? Obtain the z-transform of the transfer function

$$G(s) = \frac{s+2}{s^2 + 4s + 8}$$

54 For the error sampled unity feedback arrangement shown in Fig. P16 obtain

(i) the z-transform of the output response, C(z)

(ii) the output response $c^{*}(t)$ up to the 5th sampling instant, and

(iii) the final value of the output response

when the input disturbance is a unit step.



Fig. P16

55 Explain why the transformation $z = \frac{r+1}{r-1}$ is used in the analysis of sampled-data systems. Show by application of the Routh-Hurwitz criterion that the limiting conditions for stability of the system shown in Fig. P17 are given by

$$K > 0, \quad K < \frac{2(1 + e^{-T})}{1 - e^{-T}}$$



Fig. P17

56 The open loop pulse transfer function of a unity feedback sampled-data control system is given by

$$G(z) = \frac{K(z+3)}{z(z^2+2z+2)(z+5)(z+6)}$$

Sketch the root loci of the system indicating clearly all important features and determine the marginal value of K for stability.

57 The open loop pulsed transfer function for an error sampled control system is

$$GH(z) = \frac{0.792zK}{(z-1)(z-0.208)}$$

where K = 1.57. Use the bilinear transformation $z = \frac{r+1}{r-1}$ and plot the polar

diagram for this system. From the diagram estimate the phase margin, the gain margin, and the limiting value of K for stability. Comment on the suitability of the system.

58 Explain why proportional, integral and derivative action are of use in a controller in a closed loop system, and how each affects the response of the system. A process consists of four non-interacting stages in series, these being closely represented by simple lags of time constant 3, 5, 1, and 5 minutes and gains of 1, 0.4, 1, and 5 respectively. The process is controlled by a proportional controller of gain 2.5 in a unity feedback system. Is the system stable or not? If integral action time with $T_i = 10$ minutes is added, will the system be stable? If for simulation purposes the time constants and integral action time are considered to be in seconds, what is the effect on stability?

59 Sketch the root locus plot for the process of Problem 58 under the action of a proportional controller of gain K. Show how the plot differs when a 2-term or 3-term controller is used, by sketching qualitatively the form of the plot for various orders of magnitude of integral action time T_i and derivative action time T_d .

60 A regulator system employs rate feedback, as shown in block diagram form in Fig. P18. For the transfer functions shown, and assuming in the first instance that the time constant T is negligible, determine the values of K and k required for the following specification: the steady state error for a step change of input should not exceed 4%, and any dominant complex roots should have a damping factor as large as possible, but not greater than 0.8. How different are the results if T = 0.05 seconds?

61 A closed loop system employing minor loop compensation has a block diagram of the form shown in Fig. P18 but with

$$G_1(s) = \frac{K}{s(1+0.1s)}$$
 and $H_1(s) = \frac{0.02s}{1+0.02s}$

Determine the value of forward loop gain K required to ensure that the peak magnification has a value in the range 1.3 to 1.4. What are the values of resonant frequency, bandwidth, gain margin and phase margin?

62 A feedback system incorporates acceleration feedback, and has a block diagram of the form shown in Fig. P18 with transfer functions

$$G_1(s) = \frac{80}{s(s+1)(s+2)(s+4)}, \quad H_1(s) = ks^2$$

Draw a root locus diagram which shows the variation in the roots of the characteristic equation with variation of the constant k. What information does the plot give about the dynamic performance of the system?

Problems 295



Fig. P18

63 Draw the Bode diagram for the servomechanism of Problem 50 both without and with the compensation element in the loop. What effect does the given phase lag series compensation network have on the values of gain margin and phase margin?

64 Design a phase lead series compensation network in order that the closed loop system shown in Fig. P15, with

$$G(s) = \frac{K}{s(1+0.25s)}, \quad H(s) = 1$$

has a velocity error constant of at least 100 second $^{-1}$, and a phase margin of at least 45°. Draw a block diagram for the compensated system, showing the transfer function of each block.

65 A servomechanism (Fig. P15) has forward path transfer function

$$G(s) = \frac{0.05}{s(1+1.25s)}$$

and feedback path transfer function

$$H(s) = \frac{10}{1+0.01s}.$$

It is required that the velocity error coefficient K_v be 15 second⁻¹ and the phase margin be 45° minimum. Design a phase lead series network to enable this specification to be met. Compare the values of gain margin and phase margin before and after compensation, and comment on the significance.

66 For the system used in Example 11.3, design a state variable feedback loop so that the maximum overshoot does not exceed 20% of a step input disturbance and the steady state error does not exceed 1%. Take a value of K = 10.

67 A regulator system using state variable feedback is described by the following state equations

$$\begin{aligned} \{\dot{\mathbf{x}}\} &= \begin{bmatrix} 1 & -3\\ 5 & 0 \end{bmatrix} \{\mathbf{x}\} + \begin{bmatrix} 0\\ 1 \end{bmatrix} u \\ u &= -[h_1 h_2] \{\mathbf{x}\} \end{aligned}$$

296 Appendix A

Find the equation relating h_1 to h_2 when

- (i) the system has an undamped natural frequency of $\sqrt{2}$ rad/second
 - (ii) the system has a damping factor of 0.707.

Evaluate the coefficients of the feedback matrix if the system has a natural frequency of $\sqrt{2}$ rad/second and a damping factor of 0.707.

68 Describe the advantages to be gained by employing feedback in control systems, give the reasons why analysis is necessary in the design stage, and outline the general form which such analysis could take for any given physical system.

Appendix B Bibliography

A selected list of books for further reading

General texts on feedback control theory which are complementary to this book

 Auslander, D. M., Takahashi, Y., and Rabins, M. J. Introducing Systems and Control, McGraw-Hill, 1974.
 Broadly similar coverage, with emphasis on state representation and

digital computer solution; very useful, and presents ideas in a different way.

- 2 Kuo, B. C. Automatic Control Systems, Prentice Hall, (4th Ed.), 1982. Both classical and state variable approach; the 4th edition has increased emphasis on digital control and design.
- 3 Raven, F. H. Automatic Control Engineering, McGraw-Hill, (3rd Ed.), 1978.

Mainly classical approach; additional material on practical components, state space, digital control, and non-linear systems.

4 D'Azzo, J. J., and Houpis, C. H. Feedback Control System Analysis and Synthesis, McGraw-Hill, (2nd Ed.), 1966.

Very detailed treatment of classical linear control theory.

5 D'Azzo, J. J., and Houpis, C. H. Linear Control System Analysis and Design, McGraw-Hill, 1975.

Both the classical and modern approach.

- 6 Shinners, S. M. Modern Control System Theory and Application, Addison Wesley, 1972.
 - Aims to bridge the gap between the classical and modern approaches.
- 7 Towill, D. R. Transfer Function Techniques for Control Engineers, Iliffe, 1970.

Very useful for clarifying the relationship between root locus plots and transient response.

- 8 Healey, M. Principles of Automatic Control, EUP, (3rd Ed.), 1975. Introduces the main principles of a wide range of control engineering topics.
 - 9 Truxall, J. G. Automatic Feedback Control System Synthesis, McGraw-Hill, 1955.

A classic early text with much useful information.

 Eveleigh, V. W. Introduction to Control Systems Design, McGraw-Hill, 1972.

A comprehensive basic text.

298 Appendix B

- Emanuel, P., and Leff, E. Introduction to Feedback Control Systems, McGraw-Hill, 1979.
 An elementary introduction by the classical approach, with inclusion of data acquisition and interfacing between computer and control system.
 - Sante, D. P. Automatic Control System Technology, Prentice Hall, 1980. Elementary text, with particular reference to electrical systems.
 - 13 Miller, R. W. Servomechanisms, Devices and Fundamentals, Reston, 1977, Elementary, and largely hardware oriented.

Texts of particular relevance to specific chapters

- 14 Mayr, O. The Origins of Feedback Control, MIT Press, 1970. History of the early development of practical feedback control systems.
- 15 Johnson, C. L. Analog Computer Techniques, McGraw-Hill, (2nd Ed.), 1963.
- 16 Wilkins, B. R. Analogue and Iterative Methods, Chapman and Hall, 1970. This and the previous book give a more detailed treatment of basic analogue computation.
- 17 Dorn, W. S., and McCracken, D. D. Numerical Methods with Fortran IV Case Studies, John Wiley, 1972.
- 18 Korn, G. A., and Wait, J. V. Digital Continuous System Simulation, Prentice Hall, 1978.
- 19 Speckhart, F. H., and Green, W. L. A Guide to Using CSMP, Prentice Hall, 1976.
- 20 Davies, W. D. T. System Identification for Self-Adaptive Control, Wiley Interscience, 1970.
 - Statistical system identification and its use for automatic identification.
- 21 Thaler, G. J. Design of Feedback Systems, Dowden Hutchinson and Ross, 1973.

An excellent treatment of system design.

- 22 Hohn, F. E. Elementary Matrix Algebra, Macmillan, (3rd Ed.), 1973.
- 23 Hildebrand, F. B. Methods of Applied Mathematics, Prentice Hall, (2nd Ed.), 1965.
 - These two books contain a more complete treatment of matrix algebra.
- 24 Jury, E. I. Sampled-Data Control Systems, John Wiley, 1958.
- 25 Kuo, B. C. Analysis and Synthesis of Sampled-Data Control Systems, Prentice Hall, 1963.

More advanced texts which extend the basic theory included here

- 26 MacFarlane, A. G. J. Dynamical Systems Models, Harrap, 1970.
- 27 Nicholson, H. (Ed.) Modelling of Dynamical Systems, Institution of Electrical Engineers, 1980.
- 28 Bishop, A. B. Introduction to Discrete Linear Controls, Academic Press, 1975.
- 29 Newland, D. E. Random Vibrations and Spectral Analysis, Longman, 1975.
- 30 Rosenbrock, H. H. State-space and Multivariable Theory, Nelson, 1970.
- 31 Mishkin, E., and Braun, L. Adaptive Control Systems, McGraw-Hill, 1961.
- 32 Graupe, D. Identification of Systems, Van Nostrand Reinhold, 1972.
- 33 Kirk, D. E. Optimal Control Theory, Prentice Hall, 1970.

- 34 Chu, Y. Digital Simulation of Continuous Systems, McGraw-Hill, 1969.
- 35 Ord-Smith, R. J., and Stephenson, J. Computer Simulation of Continuous Systems, Cambridge University Press, 1975.
- 36 Chestnut, H. Systems Engineering Tools, John Wiley, 1965.
- 37 Lee, T. H., Adams, G. E., and Gaines, W. M. Computer Process Control: Modelling and Optimisation, John Wiley, 1968.
- 38 Kuo, B. C. Digital Control Systems, Holt, Rinehart and Winston, 1980.
- 39 Rosenbrock, H. H. Computer Aided Control System Design, Academic Press, 1974.

Appendix C Introduction to Matrix Algebra

This appendix presents definitions of some terms used in matrix algebra and of elementary matrix operations, and provides a foundation for understanding the mathematical material included in Chapter 5 and Section 11.10, for those readers unfamiliar with matrix algebra.

(a) Matrix. A matrix is a set of elements consisting of real or complex numbers, functions or operators, arranged in a rectangular formation of rows and columns which is denoted by square brackets and is of the form

$$A = \begin{bmatrix} a_{11} & a_{12} & \dots & a_{1n} \\ a_{21} & & & a_{2n} \\ \dots & & & \ddots \\ \vdots & & & \ddots \\ a_{m1} & a_{m2} & \dots & a_{mn} \end{bmatrix}$$

This matrix with *m* rows and *n* columns is said to be a matrix of order $m \times n$, and the symbol a_{ij} is used to denote the element located in the *i*th row and the *j*th column. If n = 1, the matrix has only one column and is known as a column vector; if m = 1 it is called a row vector. When m = n the matrix is described as square, and of order *n*.

For the special case when m = n, and all the off-diagonal elements are equal to zero $(a_{ij} = 0 \text{ for } i \neq j)$, the matrix reduces to a diagonal matrix of the form

a11	0	0	0		0
0	a22	0	0		0
0	0	a33	0	1.4	0
0	0	0	a_{44}	4	0
5	6	1.4		÷.	1
0	0	0	0		ann

A diagonal matrix, where all the diagonal elements have the value unity $(a_{ii} = 1)$, is called a unit matrix or an identity matrix and given the symbol **I**.

(b) Determinant. For any square matrix, a determinant can be evaluated from the elements of the matrix. For example, if

$$A = \begin{bmatrix} 3 & -2 & 4 \\ 1 & 5 & 6 \\ 2 & 1 & 7 \end{bmatrix}$$

then

Introduction to matrix algebra 301

det
$$A = |A| = 3 \begin{vmatrix} 5 & 6 \\ 1 & 7 \end{vmatrix} - (-2) \begin{vmatrix} 1 & 6 \\ 2 & 7 \end{vmatrix} + 4 \begin{vmatrix} 1 & 5 \\ 2 & 1 \end{vmatrix}$$

= 3(29) + 2(-5) + 4(-9) = 41

When the determinant of a matrix is zero, the matrix A is called singular.

(c) Transpose of a matrix. The transpose of matrix A, denoted by A^T , is the matrix formed by interchanging rows and columns of A. If the original matrix is an $m \times n$ matrix, its transpose is an $n \times m$ matrix.

$$A^{T} = \begin{bmatrix} a_{11} & a_{12} & \cdots & a_{1n} \\ a_{21} & a_{22} & \cdots & a_{2n} \\ \vdots & \vdots & \vdots \\ a_{m1} & a_{m2} & \cdots & a_{mn} \end{bmatrix}^{T} = \begin{bmatrix} a_{11} & a_{21} & \cdots & a_{m1} \\ a_{12} & a_{22} & \cdots & a_{m2} \\ \vdots & \vdots & \vdots \\ a_{1n} & a_{2n} & \cdots & a_{mn} \end{bmatrix}$$

The transpose of a column vector is a row vector and vice versa.

(d) Multiplication of a matrix by a scalar quantity. A matrix is said to be multiplied by a scalar K if all elements a_{ij} are multiplied by K.

(e) Addition and subtraction of matrices. Addition and subtraction of two matrices can only be performed if the two matrices have the same order. Addition of two matrices A and B results in a new matrix C with its elements c_{ii} equal to the sum of the corresponding elements a_{ii} and b_{ii} . Further

$$(A+B)^T = A^T + B^T$$

Similar arguments apply for subtraction of matrices.

(f) Multiplication of matrices. The multiplication of two matrices is possible only if the number of columns of the first matrix is equal to the number of rows of the second. If an $m \times n$ matrix A is post multiplied by an $n \times p$ matrix B, then the result will be a matrix C of order $m \times p$. For example,

 $\begin{bmatrix} a_{11} & a_{12} & a_{13} \\ a_{21} & a_{22} & a_{23} \\ a_{31} & a_{32} & a_{33} \\ a_{41} & a_{42} & a_{43} \end{bmatrix} \begin{bmatrix} b_{11} & b_{12} \\ b_{21} & b_{22} \\ b_{31} & b_{32} \end{bmatrix} =$ $(4 \times 3 \text{ matrix}) \qquad (3 \times 2 \text{ matrix})$ $\begin{bmatrix} a_{11}b_{11} + a_{12}b_{21} + a_{13}b_{31} & a_{11}b_{12} + a_{12}b_{22} + a_{13}b_{32} \\ a_{21}b_{11} + a_{22}b_{21} + a_{23}b_{31} & a_{21}b_{12} + a_{22}b_{22} + a_{23}b_{32} \\ a_{31}b_{11} + a_{32}b_{21} + a_{33}b_{31} & a_{31}b_{12} + a_{32}b_{22} + a_{33}b_{32} \\ a_{41}b_{11} + a_{42}b_{21} + a_{43}b_{31} & a_{41}b_{12} + a_{42}b_{22} + a_{43}b_{32} \end{bmatrix}$

 $(4 \times 2 \text{ matrix})$

That is, the elements c_{ij} are found by multiplying the elements of the *i*th row of A with the elements of the *j*th column of B and then summing these element

302 Appendix C

products. It is important to note that generally in matrix multiplication

$$AB \neq BA$$

hence it is always necessary to specify the relative position of the matrices to be multiplied. Multiplication of any matrix by a unit matrix results in the original matrix (AI = A). The transpose of the product of two matrices is the product of their transposes in reverse order, i.e.

$$(AB)^T = B^T A^T$$

(g) Cofactor and adjoint of a matrix. The cofactor A_{ij} of a matrix A is defined as

$$A_{ij} = (-1)^{i+j} M_{ij}$$

where M_{ij} is the minor determinant of the *A* matrix. The minor M_{ij} of an $n \times n$ matrix is the determinant of the $(n-1) \times (n-1)$ matrix formed by deleting the *i*th row and *j*th column of the $n \times n$ matrix.

The adjoint matrix is found by replacing each element a_{ij} of matrix A by its cofactor A_{ij} and then transposing. For example,

$$\operatorname{Adj} \begin{bmatrix} 3 & -2 & 4 \\ 1 & 5 & 6 \\ 2 & 1 & 7 \end{bmatrix} = \begin{bmatrix} 5 & 6 \\ 1 & 7 \\ - 2 & 4 \\ 1 & 7 \\ - 2 & 4 \\ - 3 & 2 & 7 \\ - 3 & 2 & 7 \\ - 3 & 2 & 7 \\ - 2 & 4 \\ - 3 & 4 \\ - 3 & 4 \\ - 3 & 4 \\ - 3 & 4 \\ - 3 & 4 \\ - 3 & 4 \\ - 3 & 4 \\ - 3 & 4 \\ - 3 & 4 \\ - 3 & 4 \\ - 3 & 4 \\ - 3 & 4 \\ - 3 & -2 \\ - 2 & -1 \\ - 3 & -2 \\ - 3 & -2 \\ - 2 & -1 \\ - 3 & -2 \\ - 2 & -1 \\ - 3 & -2 \\ - 2 & -1 \\ - 3 & -2 \\ - 3 & -2 \\ - 2 & -1 \\ - 3 & -2 \\ - 3 & -2 \\ - 2 & -1 \\ - 3 & -2 \\ - 3 & -2 \\ - 2 & -1 \\ - 3 & -2 \\ - 2 & -1 \\ - 3 & -2 \\ - 3 & -2 \\ - 2 & -1 \\ - 3 & -2 \\ - 2 & -1 \\ - 3 & -2 \\ - 3 & -2 \\ - 2 & -1 \\ - 3 & -2 \\ - 3 & -2 \\ - 2 & -1 \\ - 3 & -2 \\ - 2 & -1 \\ - 3 & -2 \\ - 2 & -1 \\ - 3 & -2 \\ - 2 & -1 \\ - 3 & -2 \\ - 2 & -1 \\ - 3 & -2 \\ - 2 & -1 \\ - 3 & -2 \\ - 2 & -1 \\ - 3 & -2 \\ - 2 & -1 \\ - 3 & -2 \\ - 3 & -$$

(h) Inverse, or reciprocal, of a matrix. The inverse of a square matrix A is written as A^{-1} and defined by

$$AA^{-1} = A^{-1}A = I$$

The inverse of matrix A is evaluated numerically by dividing its adjoint matrix by its determinant A

$$A^{-1} = \frac{\operatorname{adj} A}{\operatorname{det} A}$$

(i) Matrix Calculus. The derivative of an $m \times n$ matrix A(t) is defined to be

$$\frac{d}{dt} A(t) = \begin{vmatrix} \frac{d}{dt} a_{11}(t) & \frac{d}{dt} a_{12}(t) & \dots & \frac{d}{dt} a_{1n}(t) \\ \frac{d}{dt} a_{21}(t) & \frac{d}{dt} a_{22}(t) & \dots & \frac{d}{dt} a_{2n}(t) \\ \dots & \dots & \dots \\ \frac{d}{dt} a_{m1}(t) & \frac{d}{dt} a_{m2}(t) & \dots & \frac{d}{dt} a_{mn}(t) \end{vmatrix}$$

Similarly the integral of an $m \times n$ matrix A(t) is defined as

$$\int \mathbf{A}(t) \, \mathrm{d}t = \begin{bmatrix} \int a_{11}(t) \, \mathrm{d}t & \int a_{12}(t) \, \mathrm{d}t & \dots & \int a_{1n}(t) \, \mathrm{d}t \\ \int a_{21}(t) \, \mathrm{d}t & \int a_{22}(t) \, \mathrm{d}t & \dots & \int a_{2n}(t) \, \mathrm{d}t \\ \dots & \dots & \dots \\ \int a_{m1}(t) \, \mathrm{d}t & \int a_{m2}(t) \, \mathrm{d}t & \dots & \int a_{mn}(t) \, \mathrm{d}t \end{bmatrix}$$

(j) Phase-variable canonical form

Consider the linear time invariant system described by the *n*th order differential equation:

$$\frac{d^{n}c(t)}{dt^{n}} + a_{1} \frac{d^{n-1}c(t)}{dt^{n-1}} + \ldots + a_{n}c(t) = r(t)$$

For this equation it is possible to define the state variables as:

$$x_1(t) = c(t)$$

$$x_2(t) = \frac{dc(t)}{dt}$$

$$x_n(t) = \frac{d^{n-1}c(t)}{dt^{n-1}}$$

and write the nth order differential equation as a set of 1st order equations, viz

 $\dot{x}_1(t) = x_2(t)$ $\dot{x}_2(t) = x_3(t)$

$$\dot{x}_{n-1}(t) = x_n(t) \dot{x}_n(t) = -a_n x_1(t) - a_{n-1} x_2(t) \dots - a_1 x_n(t) + r(t)$$

Writing these in the concise form of Eq. 5.16 yields

	0	1	0	0		0			0
	0	0	1	0		0			0
	0	0	0	F		0	1.5		0
<i>A</i> =		× -		-	***	14	4	B =	
		35		2			m -		2
	0	0	0	0	1000	1			0
	$-a_n$	$-a_{n-1}$	$-a_{n-2}$	$-a_{n-3}$		$-a_1$		- C	1

With the coefficient matrix of the process and its driving matrix arranged in this way, Eq. 5.16 is said to be written in 'phase-variable canonical form'.

The matrix operations presented in this appendix can readily be carried out on a digital computer using the standard sub-routine programs available on most commercial machines.

Appendix D Answers to Problems

$$q' = C_1(p'_1 - p'_2) + C_2\eta' + C_2\eta$$

where all variables are relative to a datum, and with constants

$$C_{1} = \left[\frac{\pi dh^{3}}{12\eta l}\right]_{0}, \quad C_{2} = \left[\frac{\pi dh^{3}(p_{1} - p_{2})}{12\eta^{2}l}\right]_{0}, \quad C_{3} = \left[\frac{\pi dh^{3}(p_{1} - p_{2})}{12\eta l^{2}}\right]_{0}$$

where []0 indicates that the expressions are evaluated using absolute values of d, h, η etc. for the datum condition.

4

5

$$\frac{Cs + (K_2 + 2K_3)}{CMs^3 + (K_2M + 2K_3M)s^2 + K_2Cs + 2K_2K_3},$$

$$\frac{J_2s^2 + K}{s(J_1J_2s^3 + CJ_2s^2 + K(J_1 + J_2)s + CK)}$$

 $\frac{3}{1+R_2Cs} = \frac{(1+R_1C_1s)(1+R_2C_2s)}{1+(R_1+R_2)Cs}, \quad \frac{(1+R_1C_1s)(1+R_2C_2s)}{1+(R_1C_1+R_2C_2+R_1C_2)s+R_1R_2C_1C_2s^2},$ $1 + C_1(R_1 + R_2)s + R_1R_2C_1C_2s^2$ $1 + (R_1C_1 + R_1C_2 + R_2C_1)s + R_1R_2C_1C_2s^2$

$$\frac{1}{\frac{W_1W_2}{H_1H_2}s^2 + \left(\frac{W_1}{H_1} + \frac{W_1}{H_2} + \frac{W_2}{H_2}\right)s + 1}$$

$$\frac{\frac{AC_{1}C_{2}}{K(C_{1}+K_{L})}}{\frac{M}{K}s^{2} + \left(\frac{\mu}{K} + \frac{A^{2}}{K(C_{1}+K_{L})}\right)s + 1}$$

where A is piston area,

K is spring stiffness,

 K_L and μ are leakage and viscous drag coefficients,

and C_1 and C_2 are coefficients in linearized orifice and lever equations.

Answers to problems 305

$$\frac{1}{\frac{vM}{AK_{\rm B}C}s^3 + \left(\frac{2K_{\rm L}M}{AC} + \frac{v\mu}{AK_{\rm B}C}\right)s^2 + \left(\frac{2A}{C} + \frac{2K_{\rm L}\mu}{AC}\right)s + 1}$$

where the symbols are as defined on pages 22-26.





6





16

 $c(t) = \frac{1}{10} - \frac{1}{4}e^{-t} + \frac{1}{6}e^{-2t} - \frac{1}{60}e^{-5t}$ $c(\infty) = 0.1. \text{ No overshoot. } 3.9 \text{ seconds}$

17

 $c(t) = 1 - e^{-t} - 1.155e^{-0.5t} \sin 0.866t$



18

(a) $c(t) = 1 - \frac{4}{3}e^{-t} + \frac{1}{3}e^{-4t}$ (b) $c(t) = 4t - 5 + \frac{16}{3}e^{-t} - \frac{1}{3}e^{-4t}$ zero; 3.3 seconds; 0 and 5.

$$6 + 2t - 6e^{-0.5t}$$
 for $t \le 20$
 $50 - 4e^{-0.5(t-20)}$ for $t > 20$

Maximum error is 10 at t = 0. Error $<1^{\circ}$ when t > 22.8 seconds.

20
$$100(1 - 1.33e^{-0.25t} + 0.33e^{-t})$$
 for $0 \le t \le 4$
 $100(2.29e^{-0.25t} + 17.87e^{-t})$ for $t \ge 4$

21
$$\frac{4}{(s+1)(s^2+0.6s+4)}, \frac{0.8}{(1+3.5s)(1+0.8s)}$$

22

19



23



 $y = \begin{bmatrix} 2 & 3 & 1 \end{bmatrix} \{x\}$



$$\boldsymbol{B} = \begin{bmatrix} 1 & 0 & 0 & 0 \\ 0 & -1 & 0 & 0 \\ 0 & 0 & -1 & 0 \\ 0 & 0 & 0 & 1 \end{bmatrix}, \quad \boldsymbol{C} = \begin{bmatrix} 0 & 1 & 0 & 0 \\ 0 & 0 & 1 & 0 \end{bmatrix}, \quad \boldsymbol{D} = 0$$

$$\{\mathbf{x}(t)\} = \begin{cases} h_2(t) \\ h_3(t) \end{cases}, \quad \{\mathbf{u}(t)\} = \begin{cases} h_1(t) \\ q_2(t) \\ q_3(t) \\ h_4(t) \end{cases},$$

$$\boldsymbol{A} = \begin{bmatrix} -\left(\frac{1}{a_2R_1} + \frac{1}{a_2R_2}\right) & \frac{1}{a_2R_2} \\ \frac{1}{a_3R_1} & -\left(\frac{1}{a_3R_1} + \frac{1}{a_3R_3}\right) \end{bmatrix}$$
$$\boldsymbol{B} = \begin{bmatrix} \frac{1}{a_2R_1} & -1 & 0 & 0 \\ 0 & 0 & -1 & \frac{1}{a_3R_3} \end{bmatrix}, \quad \boldsymbol{C} = \begin{bmatrix} 1 & 0 \\ 0 & 1 \end{bmatrix}, \quad \boldsymbol{D} = 0$$

$$\{\mathbf{x}(t)\} = \begin{cases} h_2(t) \\ h_3(t) \end{cases}, \quad \{\mathbf{u}(t)\} = \begin{cases} h_1(t) \\ h_4(t) \end{cases}$$
$$A = \begin{bmatrix} -\left(\frac{1}{a_2 R_1} + \frac{1}{a_2 R_2}\right) & \frac{1}{a_2 R_2} \\ \frac{1}{a_3 R_1} & -\left(\frac{1}{a_3 R_1} + \frac{1}{a_3 R_3}\right) \end{bmatrix}$$
$$B = \begin{bmatrix} \frac{1}{a_2 R_1} & 0 \\ 0 & \frac{1}{a_3 R_3} \end{bmatrix}, \quad C = \begin{bmatrix} 1 & 0 \\ 0 & 1 \end{bmatrix}, \quad D = 0$$

$$G(s) = \begin{bmatrix} \frac{aRs+3}{a^2R^2s^2+5aRs+5} & \frac{2}{a^2R^2s^2+5aRs+5} \\ \frac{1}{a^2R^2s^2+5aRs+5} & \frac{2aRs+4}{a^2R^2s^2+5aRs+5} \end{bmatrix}$$

(b)

(c)

25

310 Appendix D

$$\{\dot{\mathbf{x}}(t)\} = \begin{bmatrix} 0 & 1 \\ -10 & -7 \end{bmatrix} \{\mathbf{x}(t)\} + \begin{bmatrix} 0 \\ 1 \end{bmatrix} u(t), \quad y(t) = \begin{bmatrix} 10 & 10 \end{bmatrix} \{\mathbf{x}(t)\}$$

or
$$\{\dot{\mathbf{x}}(t)\} = \begin{bmatrix} -5 & 0 \\ 0 & -2 \end{bmatrix} \{\mathbf{x}(t)\} + \begin{bmatrix} 1 \\ 1 \end{bmatrix} u(t), \quad y(t) = \begin{bmatrix} \frac{40}{3} & -\frac{10}{3} \end{bmatrix} \{\mathbf{x}(t)\}$$

or
$$\{\dot{\mathbf{x}}(t)\} = \begin{bmatrix} -5 & 10 \\ 0 & -2 \end{bmatrix} \{\mathbf{x}(t)\} + \begin{bmatrix} 0 \\ 1 \end{bmatrix} u(t), \quad y(t) = \begin{bmatrix} -4 & 10 \end{bmatrix} \{\mathbf{x}(t)\}$$

or
$$\{\dot{\mathbf{x}}(t)\} = \begin{bmatrix} -2 & 10 \\ 0 & -5 \end{bmatrix} \{\mathbf{x}(t)\} + \begin{bmatrix} 0 \\ 1 \end{bmatrix} u(t), \quad y(t) = \begin{bmatrix} -1 & 10 \end{bmatrix} \{\mathbf{x}(t)\}$$

$$c(t) = 1 + \frac{5}{3}e^{-2t} - \frac{8}{3}e^{-5t}$$

27
$$\varphi(t) = \begin{bmatrix} e^{-0.5t}(\cos \theta + 0.577 \sin \theta) & e^{-0.5t}(1.155 \sin \theta) \\ e^{-0.5t}(-1.555 \sin \theta) & e^{-0.5t}(\cos \theta - 0.577 \sin \theta) \end{bmatrix}$$

where $\theta = 0.866t$

$$x_1(t) = 1 - 1.155e^{-0.5t} \sin\left(0.866 + \frac{\pi}{3}\right); \ x_2(t) = \dot{x}_1(t)$$

$$\varphi(t) = \begin{bmatrix} e^{-t} & 0 & 0\\ 0 & e^{-4t} & 0\\ 0 & 0 & e^{-2t} \end{bmatrix} \qquad \begin{array}{c} x_1(t) = 1 - e^{-t}\\ x_2(t) = \frac{1}{4} - \frac{1}{4}e^{-4t}\\ x_3(t) = \frac{1}{2} + \frac{1}{2}e^{-2t} \end{array}$$

$$G(s) = \begin{bmatrix} \frac{.4s^2 + 14s + 14}{(s+1)(s+2)(s+4)} \\ \frac{.2(s+3)}{(s+2)(s+4)} \end{bmatrix}$$

28

$$\varphi(t) = e^{4} = \begin{bmatrix} 1 - 2t^{2} + \frac{4}{3}t^{3} + \dots & t - t^{2} + \frac{1}{3}t^{4} + \dots \\ -4t + 4t^{2} - \frac{4}{3}t^{4} + \dots & 1 - 2t + \frac{4}{3}t^{3} - \frac{2}{3}t^{4} + \dots \end{bmatrix}$$

For t = 1 second this gives poor accuracy since further terms could make a significant difference, but for t = 0.5 second it is a reasonable approximation, since the terms reduce rapidly in magnitude.

26


31 Same information as Problem 30 plotted on a Bode diagram.

32 $G(s) = \frac{0.33}{s(1+0.5s)}$ The transducer bandwidth is significantly higher than that of the system.

33
$$G(s) = \frac{220(1+2.6s)}{(1+0.65s)(s^2+20s+580)}$$

34
$$\phi_{yy}(\tau) = \frac{A^2}{2} \cos \omega t + \frac{B^2}{2} \cos 3\omega \tau + C^2$$

35 (a)
$$\frac{A^2}{2}\cos\omega\tau$$
 (b) $\frac{B}{\pi\tau}\sin\omega_c\tau$

36 11111000110111010100001001011100



37
$$G(s) = Ke^{-0.5s} \left(\frac{40}{s^2 + 2s + 40} \right)$$
, with time in minutes.

312 Appendix D

38 0 and $\frac{10}{11}$ with steady unit input; 2 and ∞ with unit ramp; ∞ for both with acceleration input.

- **39** (i) $0 < K < 75, 3.74, 0.107, \infty$ (ii) 0 < K < 45, 3, 0, 0.2(iii) $0 < K < \infty, -, 0, 0$
- 40 11 dB, 42° (from straight line approximation) 13.4 dB, 50.4° (by calculation)

41
$$G(s) = \frac{6.3}{s(1+0.25s)(1+0.05s)}$$
, $M_p = 1.9$, bandwidth = 7 rad/second.

42 1 dB, 2.5°. Close to instability.

43
$$20 \log_{10} \frac{110}{K}; \ \tan^{-1} \left(\frac{1 - 0.001 \omega_1^2}{0.11 \omega} \right)$$

where ω_1 is given by

$$0.000\,001\omega_1^4 + 0.0101\omega_1^2 + 1 = \frac{K^2}{\omega_1^2};$$
 110; 6 dB, 12°.

44 14.5 dB, 30°; 6 dB, 0.2 rad/second, 0.45 rad/second. Rather poor stability, little improvement possible by change in gain alone.

45 K = 0.8. Transducer time constant is significant. With lag GM = 8 dB, PM = 45° for K = 0.8, e_{ss} is 50%. Without lag GM = ∞ , PM = 45° for K = 1.37, e_{ss} is 37%.



47 60; 0.52 rad/second; -2,3, -10, negligible effect.

48 1.78;
$$C(s) = \frac{225}{s(s^2 + 2.17s + 2.4)(s + 2.9)(s + 10)(s + 12.5)}$$

$$\approx \frac{1.78}{s(s^2 + 2.17s + 2.4)(s + 2.9)}$$
3.7 seconds.

49 Conditionally stable; stable for $0.5 < K_c < 1.8$, approximately second order. For $K_c = 0.57$ dominant roots have $\zeta \approx 0.7$.

50 Unstable. 3.3 rad/second, 0.47 (with compensation).



 $c^{*}(t) = 0.02(t - T) + 0.07(t - 2T) + 0.14(t - 3T) + 0.224(t - 4T) + 0.317(t - 5T) + \dots; 1.$

314 Appendix D





57 $PM = 38^\circ$, GM = 5.76 dB; limiting K = 3.05.

58 Stable; unstable; no effect.

59



60 For T=0, K=480 for 4% error, k=0.26 gives a maximum ζ of 0.40 for the dominant roots. For T=0.05, K=480 as before, but maximum attainable ζ is 0.16 for k=0.18.

61 K = 50 gives $M_p \simeq 1.22$, and 20 rad/second, 33 rad/second, ∞ , 47° . Required M_p is not attainable by change in K alone.



63 -1 dB, -5° (unstable) improved to 16 dB, 45° .

64
$$G_c(s) = \frac{1}{4.2} \left(\frac{1+0.070s}{1+0.017s} \right), \quad G(s) = \frac{420}{s(1+0.25s)}$$

65
$$G_c(s) = \frac{1}{3.7} \left(\frac{1+0.41s}{1+0.11s} \right), \quad G(s) = \frac{5.5}{s(1+1.25s)}$$

17 dB, 11° improved to 24 dB, 45

66

67

$$\{\dot{x}\} = \begin{bmatrix} 0 & 1 & 0 \\ 0 & 0 & 1 \\ -0.2 & -2.3 & -3.1 \end{bmatrix} \{x\} + \begin{bmatrix} 0 \\ 0 \\ 2 \end{bmatrix} u$$
$$H = \begin{bmatrix} 0.89 & 0.94 & 0.55 \end{bmatrix}, \text{ Overshoot} = 16\%$$
$$h_2 = 13 - 3h_1; \quad h_2^2 = 29 - 6h_1; \quad H = \begin{bmatrix} 3\frac{1}{3} & 3 \end{bmatrix}$$

Index

acceleration error coefficient, 157 acceleration feedback, 275-8 acceleration input, 60, 157 accuracy of simulation, 29, 30, 44, 49, 50 steady state, 4, 154-7 ACF, see autocorrelation function adaptive control, 71 addition of matrices, 89, 301 of poles and zeros, 182-3 adjoint matrix, 89, 92, 261, 302 aids to construction of root locus diagram, 177-82 algorithm, 48, 50, 94, 128, 139, 258 aliasing, 138 allowable error, 156-7 amplifier. 30-33, 270 amplifier gain. 30-32, 43-4 amplitude ratio, 99 see also magnitude amplitude scaling, 42-7 analogues, 7, 29 analogue computer, 7, 29, 126, 264 elements, 30-33, 40 analysis need for. 4 methods of, 6-8 angle condition for root loci, 176, 180 angle of departure of root loci, 180 asymptotes to root loci, 178 asymptotic approximation, 106, 109, 111. 116 attenuation, 236, 239 autocorrelation function (ACF), 121 properties, 124-5 of PRBS, 145-6, 149 of various signals, 122-6, 129, 137 auxiliary equation, 160-61 auxiliary feedback loop, 154, 162, 275

backlash, 14 band limited white noise, 131-2 bandwidth, 117, 128, 168, 199, 241 Bartlett window, 141-2 bit interval, 148, 150 black box problem, 6, 7, 72 block diagram, 4-5, 10, 152-4, 204-5, 265 reduction, 17-18 Bode plot (or diagram). 106-14, 116, 136. 237-9.244 breakaway point of root loci, 179 break point on Bode plot, 109, 111, 113 bridged T network, 246 bulk modulus, 26, 266 CCF, see cross correlation function characteristic equation, 18, 66, 69, 154. 158, 170-71, 174 characteristic function, 162 circuit diagram, 33-4, 42-7 classical methods of analysis, 6, 8 classification of systems, see type of feedback system closed loop control, 3-4, 153-5 closed loop frequency response, 165-9 closed loop magnitude. see magnification closed loop zeros, 187, 188, 230-31 coefficient matrix, 86 coefficient potentiometer. 32, 38 coherence function, 137 column matrix, 83 column vector, 86, 298 comparator relay, 40 compensation network. 236, 241-3. 247 compensation, 165, 234-5, 249 complementary function, 63, 100 complex s plane. 66-7. 69-70 compressibility, 26 compute mode, 39 computer time, see machine time

318 Index

conditional frequency, 67 conditions of randomness, 143 conjugate roots, 69-70, 110, 178 controllability, 255-6 controlled output, 152 controller settings, 232-4 control vector, 86 convolution integral, 77-9, 92, 126, 135 corner frequency, see break point correlation, 120-21 correlation function. see autocorrelation function, cross correlation function correlator, 128 correlogram, 139 covariance, 121 covariance function, 121 critically damped system, 65 critical point, 162-3, 171 cross correlation function, 125, 126-9, 150 - 51crossover frequency, 164 cross power spectral density, 131 CSMP. 54-9 cut-off rate, 118

damped natural frequency, 67 damper, see hydraulic damper damping factor, 20, 65-7, 103-4, 110-11 data reconstruction, 207 datum condition (datum point), 11-14 dB, 106 deadband, 40, 263 dead time, 73 decade of frequency. 109-11 decibel. 106 degree of stability, 161 derivative action, 224, 228-30 derivative of matrix, 302 derivative term, 108 describing function, 264 design, 154, 182-3, 220 desired output, 3-4, 152-3, 155 determinant of matrix, 90, 93, 300 deterministic signals, 119 difference equations, 48, 94 differential equation, 6, 7-8, 10, 16, 33 differentiation, on analogue computer, 40 digital computer, 30, 47, 94, 128, 137, 170 digital simulation, 47 diode function generator, 40 discrete system, 94, 97 discrete time model, 94 discrete time series, 137

discrete time solution, 96 distributed parameter system, 21 disturbance, 7, 60, 77, 119 dominant locus (loci), 183, 227, 237 dominant pole, 69 dominant root, 67, 170, 174, 227 dominant time constant, 73, 75, 195 driving matrix, 86 energy distribution of signal, 131 environment, 1, 4 error criterion, 73 error signal, 153-4 error vector, 259-60 errors with straight line approximations. 109, 111, 113 estimated state vector, 258, 259 Euler method, 48, 50 even function, 125, 130

exclusive or, 144 expected value, 121 experimental testing, 6, 61, 114, 266, 268 experimental verification, 7 exponential lag, *see* simple lag

Fast Fourier Transform, 134, 139-40 feedback control, see closed loop control feedback matrix, 255, 259 feedforward compensation, 247-8 FFT, see Fast Fourier Transform filter, 138, 240 final value theorem, 15, 69, 155, 207 first order hold, 207 first order system, 61, 71, 73, 90, 102 see also simple lag flow continuity equation, 24-6, 87, 267 forcing function, 27, 60-61, 264 Fourier analysis, 133 Fourier transform, 130, 140 frequency response, 99, 218 testing, 114-16 functions, in simulation language, 55-6

gain, 69, 99, 107 gain crossover frequency, 164, 239 gain margin, 163–5, 168–9 gain phase plot, *see* Nichols chart gate function, 140 generalized state equations, 86 governor, 12, 173

Hanning window, 141-2 harmonic locus, 102-5 harmonic response, *see* frequency response higher order systems, 68, 75, 105 HOLD mode, 39 human being, 47, 155 Hurwitz, *see* Routh–Hurwitz criterion hybrid computer, 30, 39 hydraulic damper, 19 hydraulic ram, 24–6, 104, 267 hydraulic servomechanism, 22, 61, 63 hysteresis, 14, 40

IC mode, 38-9 ideal sampling function, 195 identification, see system identification impulse function, 60, 77 impulse response, 64, 66, 77-8, 120, 126-9, 197 inertia, 24-5 initial condition vector, 89 initial condition voltages, 33, 34, 38 input function, see forcing function input vector, 86 instability, 69, 158, 161-2 integral action, 183, 185, 226-8, 232-4 integral functions of error, 72, 116 integral term, harmonic response, 108 integrating amplifier or integrator, 31-2 integration algorithm, 48-51 integration step size (or interval), 49-51 integrator, ideal, 81-6, 104 interpretation of root loci, 182-8 inverse Laplace transform, 14, 16, 68, 92 - 3inverse matrix, 89, 92, 302 inverse z transformation, 206 lag, 109, 110, 154 see also simple lag lag-lead compensation, 247, 253

Laplace transforms, table of, 15 Laplace transform technique, 14 lead, 109, 110, 183 leakage, 24 limit cycles, 264 linearity, 10, 72, 99 linearization, 11–14 line spectrum, 146 loci on real axis, 178 logarithmic decrement method, 74, 271 loop gain, 47, 225–6, 236 lumped parameters, 21 machine time, 43-4 magnification, 117, 165 magnitude (magnitude ratio), 99-101. 106magnitude condition for root loci, 176, 182 manipulated variable, 224, 232 marginal stability, 174, 179 mass-spring-damper, 19-20, 64, 171 mathematical model, 6, 10, 72, 265-72 matrix addition and subtraction, 301 matrix inversion, 97, 302 matrix multiplication, 86, 301 matrix notation, 83, 86 matrix transfer function, 89 maximum length sequence, 144 M contours, 166-9 mean power (mean square value). 125. 131 minor feedback loop, 231, 275, 277 mode control, 40 model, see mathematical model modern control theory, 8, 223 modulo two addition, 144 $M_{\rm p}$, see peak magnification multi-input-multi-output system, 86-8 multiplication of matrices, 86, 301 multiplier, 40 multivariable system, 81, 86, 97

natural frequency, see undamped natural frequency N contours, 166-9 Nichols, 232-4 Nichols chart, 168-9 noise, 7, 41, 76, 77, 119, 127-9, 137, 138 non-deterministic function, 119 non-linearity, 11-14, 40, 72, 99, 264 non-minimum phase, 116 non-parametric model, 6, 73 Nyquist diagram (Nyquist plot), 161, 165. 234 Nyquist stability criterion, 161-2 observability, 255 observer. see state observer octave of frequency, 109 offset, 156 on-line testing, 119

open loop control, 3, 152 operating modes of analogue computer, 38-9 operational amplifier, 32, 35 optimum response, 232 see also performance criteria order of matrix, 83, 86 of system, 18, 68, 83 orifice, 11, 13 oscillations, number of, 71 oscillatory response, 51, 53, 58, 132 output matrix, 71 output response, 6-7, 54, 67, 102 output vector, 71 overall transfer function, 16 overcorrection, 4, 132 overdamped response, 51, 60 overshoots, 58-60, 132, 178 P + I, see proportional plus integral control P + I + D, see three term controller parabolic function, 60 parameter variations, 7, 34, 153, 189, 272 parametric model, 6, 73 partial fraction expansion, 16, 62, 68, 82, 100, 206, 250 particular integral, 63, 100 patch panel, 34 peak magnification, 118 performance characteristics, time domain, 69, 71, 220 performance criteria, frequency domain 117-18, 220 performance index, 72, 227 periodic white noise, 129, 142 phase, 99, 101, 106 phase crossover frequency, 164 phase lag, 101, 103, 105 phase lag compensation, 234, 241-7 phase lead, 110 phase lead compensation, 236-41, 250 phase margin, 163-5, 168-9 phase plane technique, 264 phase variable canonical form, 256, 303 polar plot, 101-6, 116, 199-201 pole cancellation, 245 pole placement, 255 pole position in s plane, 69-70 poles of transfer function, 18, 69, 107-11. 176 positional error, 157, 226, 273 positional error coefficient (constant). 156potentiometer, 32, 34, 38 POT SET mode, 38

power density spectrum, 130 power device, 154 power series expansion, 206, 209 power spectral density, 137-42 of PRBS, 146, 151 practical testing, 72, 114-16, 119, 270 PRBS, see pseudo random binary sequence principle of superposition, 10-11, 77, 91, 264 problem time, 43 proportional control, 224-6 proportional plus integral control, 35, 183, 226-8, 232-4 pseudo random binary sequence (PRBS). 8, 142-7, 148 pulse (rectangular), 60-61, 195 pulse transfer function, 202, 205 pulsed time response, 196 quadratic lag (lead), 110-11 quadratic residue code, 145 r, complex variable, 214 ram, see hydraulic ram ramp function, 60, 63, 65, 77, 156 random binary function, 123, 132-3 see also pseudo random binary sequence rate feedback, 31, 231-2 real time simulation, 43, 47 record length, 137, 140-42 reference input, 152-4 reference voltage, 38, 43, 44 regulator, 154-5 relative stability, 163 relay control, 263-4 repetitive operation ('REP OP'), 39, 44 RESET mode, 38-9 response vector, 86 rise time, 71, 241 root contours, 189-93, 227-8 root locus plots (diagrams), 170-88, 227-8, 242, 273-7 root position, 182 roots of characteristic equation, 18, 66, 69, 157-9, 170 Routh-Hurwitz criterion, 157-61, 179, 214-15, 273 Runge-Kutta method, 50, 55

sample function, 140 sampler, 194 sampled data process (or scheme), 194 sampling, 95, 137-8, 195 sampling interval (or frequency), 48. 95-7.137 saturation, 11, 40 scaling, 35, 42-7 scheduling control, 3, 152 schematic diagram, 5, 266 secondary factors, 23-6 secondary feedback path. 162 secondary loci, 227 second order system, 64-8, 90, 103-4, 147.200.267 servomechanism (servo), 22, 154-5. 265 - 77servovalve, 267-70 set point, 155, 183, 194 settling time, 71 shift register, 144 signal flow graph, 6 signal-noise ratio. 35, 42, 129 simple lag. 23, 62, 109 see also first order system simulation, 29, 35, 47, 174, 187, 188 simulation languages, 53 single-input-single-output system, 3, 4, 81, 127 sinusoidal signals, 7, 99, 122-3 small perturbation analysis, 11, 119, 265 solution of differential equation, 28, 33. 60 solution of state equation, 90-94 solution matrix, 92-4 solution time, 43-4 spectral density, see power spectral density s-plane, 66, 69-70 spool valve, 22-3. 268 spring, 11, 18-19, 267 stability. 154, 157-65 stability boundary, 69, 214 staircase approximation, 95-6 starting points of root loci, 177 state equations, 27, 81-8 state observer, 258-60 state space technique, 6, 80 state variable diagrams, 81-6 state variable feedback control, 255 state variables, 26, 80-81 state vector, 27-8, 83, 86 state vector feedback control, 254-5 statements, in simulation language, 54-6 statistical testing, 7, 8, 127

steady state error, 63, 66, 71, 155-7 steady state gain. 69 steady state response, 62, 100 step change (step function), 7, 60, 76, 156 step response, 62-3, 65, 68-9 step response testing, 72-6 straight line approximation, 107, 109, 111 summing amplifier (summer). 30-32, 34 summing junction, 31, 33, 38 superposition integral, 78 synthesis, 221 system, 1-2, 4 system identification. 119-20, 126-9. 134-7 see also practical testing system representation, 4-6, 10 system response, 6-8 termini of root loci. 178 testing, see practical testing, system identification thermometer, 20-21, 61, 63 thermostat, 3, 263 three term controller (P+I+D). 224. 228-9, 232-4 time constant, 23, 62, 71, 73, 75-6 time scaling, 42-3, 47 time series representation. 48, 137, 139 transducer, 38, 118, 153, 191, 266 transfer function, 6, 16, 68, 100, 136-7. 268 - 9of sampled-data element. 196 transfer function analyser, 115 transient response, 62-3, 66-7, 69, 100 transition matrix, 92 transmission matrix, 86 transportation lag. 73, 76 transpose of a matrix, 299 two term controller, 223, 226-7 type of feedback system, 157, 225 undamped natural frequency (natural frequency), 20, 66-7, 111 underdamped response, 65, 71, 73 unit circle, 214 unit impulse, 64, 66, 195 unit matrix, 88, 89, 300

unit ramp, 63, 65, 96

unit step, 60, 61, 64

variables, absolute and relative, 12, 14 velocity error coefficient (velocity error), 157, 226, 273 322 Index

velocity feedback, 231–3, 275–6 virtual earth, 31, 33 viscous leakage, 24 viscous load, 19, 25

weighting function, 78, 126–7 white noise, 124, 127–9, 131, 142 window function, 140-42

zeros, 100, 107–10, 174–80 zero order hold, 207, 214, 218 Ziegler, 232–4 z-plane, 213, 214 z transform, 95, 201–4

